

# JEFFERSON COUNTY DRAINAGE DISTRICT NO. 6 2016 HAZARD MITIGATION PLAN UPDATE

## TABLE OF CONTENTS

EXECUTIVE SUN	/IMARYi
SECTION 1	THE PLANNING PROCESS1
	INTRODUCTION1
	AUTHORITY1
	THE PLANNING PROCESS2
	PUBLIC INVOLVEMENT4
	REVIEW AND INCORPORATION OF EXISTING PLANS, STUDIES, REPORTS AND TECHNICAL INFORMATION5
	INCORPORATION OF THE HAZARD MITIGATION PLAN INTO OTHER PLANNING MECHANISMS6
	PLAN MAINTENANCE AND CONTINUED PUBLIC INVOLVEMENT
	PLAN MONITORING, EVALUATING AND UPDATING
SECTION 2	HAZARD ASSESSMENT9
	INTRODUCTION9
	GEOGRAPHY, CLIMATE, AND POPULATION11
	OVERVIEW OF RISKS
	TORNADO19
	TORNADO LOCATION
	TORNADO EXTENT
	PREVIOUS OCCURRENCES OF TORNADO
	TORNADO PROBABILITY
	TORNADO IMPACT25
	HURRICANES AND TROPICAL STORMS25
	HURRICANE AND TROPICAL STORM LOCATION
	HURRICANE AND TROPICAL STORM EXTENT
	PREVIOUS OCCURRENCES OF HURRICANE AND TROPICAL STORMS

HURRICANE AND TROPICAL STORM PROBABILITY	28
HURRICANE AND TROPICAL STORM IMPACT	28
SEVERE THUNDERSTORMS AND HIGH WINDS	29
SEVERE THUNDERSTORMS AND HIGH WINDS LOCATION	
SEVERE THUNDERSTORMS AND HIGH WINDS EXTENT	31
SEVERE THUNDERSTORMS AND HIGH WINDS PREVIOUS OCCURRENCE	33
SEVERE THUNDERSTORMS AND HIGH WINDS PROBABILITY	34
SEVERE THUNDERSTORMS AND HIGH WINDS IMPACT	
FLOOD	35
FLOOD LOCATION	
FLOOD EXTENT	
FLOOD PREVIOUS OCCURRENCE	40
FLOOD PROBABILITY	42
FLOOD IMPACT	42
NFIP REPETITIVE LOSS PROPERTIES	43
FLOOD RISK TO RESIDENTIAL REPETITIVE LOSS PROPERTIES	46
NON-RESIDENTIAL REPETITIVE LOSS PROPERTIES	50
NFIP SEVERE REPETITIVE LOSS PROPERTIES	52
FLOOD RISKS – PUBLIC BUILDINGS	56
FLOOD RISKS – DISTRICT ASSESTS	57
FLOOD RISKS – ROADS	57
FLOOD RISKS – LOCAL DRAINAGE	59
SECTION 3 – MITIGATION STRATEGY	60
DD6'S MITIGATION GOALS	60
DD6'S MITIGATION GOAL STATEMENT	60

STATE OF TEXAS MITIGATION GOALS	. 60
FEMA'S MITIGATION GOAL	.61
IDENTIFYING PRIORITY ACTIONS	.61
DD6'S MITIGATION ACTIONS	. 64
DD6'S NEW ACTIONS	. 71
CONTINUED COMPLIANCE WITH NFIP	. 84
CHANGES IN DEVELOPMENT AND REVIEW OF	
LOCAL REGULATION AND DD6 RESOURCES	.85
APPENDIX 1 – MITIGATION PLANNING COMMITTEE MEETING MINUTES	. 87
APPENDIX 2 – PUBLIC MEETINGS AND PRESENTATION	. 98
APPENDIX 3 – CORRESPONDENCE WITH STAKEHOLDERS	. 111
APPENDIX 4 – SOURCES	. 131
APPENDIX 5 – ADOPTION RESOLUTION	. 133
APPENDIX 6 – FEMA APPROVAL LETTER	. 136

### **EXECUTIVE SUMMARY**

Jefferson County Drainage District No. Six (DD6) undertook development of this Mitigation Plan to protect public health, safety and welfare. The purpose of this Plan is to reduce or avoid the impacts of hazards by identifying and analyzing hazards and outlining mitigation actions that will help the District reach this goal. This Mitigation Plan is a single jurisdiction Plan. This Mitigation Plan also enables DD6 to apply for disaster mitigation funding sources that are otherwise unavailable without an approved Mitigation Plan.

DD6 staff has shown their commitment to hazard mitigation by writing a Hazard Mitigation Plan in 2005 and updating that plan in 2010. The 2010 Plan was approved by FEMA on June 26, 2011. DD6 staff further demonstrated their commitment to hazard mitigation by applying for and administering FEMA grants to complete numerous projects as well as self-funding projects that help protect public health, safety and welfare.

For this Mitigation Plan update, DD6 secured funding to hire a private firm to guide the planning process and Plan development. DD6 organized a Mitigation Planning Committee (MPC) consisting of members from DD6, Jefferson County and the City of Beaumont. The MPC is comprised of both members from the previous MPCs as well as new members.

The majority of this Plan is focused on the flood hazard because DD6's mission and jurisdictional authority are explicitly limited to activities related to controlling floods (although DD6 does have the authority to complete actions to protect and mitigate damage to its own facilities, assets and protect personnel).

The Plan has been prepared in compliance with Section 322 of the Robert T. Stafford Disaster Relief and Emergency Assistance Act (Stafford Act), as amended; The National Flood Insurance Act of 1968 (NFIA), as amended; and Title 44 Code of Federal Regulations Section 201.6 (44 CFR 201.6).

# **SECTION 1 – THE PLANNING PROCESS**

# INTRODUCTION

Jefferson County Drainage District No. Six (DD6) undertook development of the original Hazard Mitigation Plan because of increasing awareness that natural and man-made hazards, especially flood hazards, may affect people and property in the area. The Hazard Mitigation Plan was written to find District vulnerabilities to hazards and outline mitigation actions that help to reduce or avoid the impacts of hazards. To help reach those goals, mitigation funds are made available to jurisdictions with updated Hazard Mitigation Plans.

In accordance with 44 Code of Federal Regulations (CFR) 201.6(d)(3), local mitigation plans must be "reviewed, revised if appropriate, and resubmitted for approval within five years in order to continue to be eligible for... grant project funding."

In this update, the Mitigation Planning Committee looked at what has changed in the District since the last version of the plan was approved and what new actions need to be taken based on those changes.

# AUTHORITY

Jefferson County Drainage District No. Six (DD6) is a conservation and reclamation district and a political subdivision of the State of Texas. DD6 was established January 21, 1920, after a favorable vote on January 10, 1920. It was extended and enlarged (Vol. 63, P.478) according to the authority of the 57th Legislature, Chapter 349, and Chapter 7, Title 128, Revised Civil Statutes of Texas, Art. 8129. Enlargement came about in 1961 through legislation (HB 1063), which also established DD6 as a Conservation and Reclamation District under Section 59, Article XVI, Texas Constitution. DD6 was created primarily to provide drainage of overflow lands within DD6. DD6 is governed by a five-member Board of Directors, appointed by the County Commissioners Court of Jefferson County, Texas.

Authority for the preparation of the Hazard Mitigation Plan is derived from Section 322 of the Robert T. Stafford Disaster Relief and Emergency Assistance Act (Stafford Act), as amended; The National Flood Insurance Act of 1968 (NFIA), as amended; and Title 44 Code of Federal Regulations Section 201.6 (44 CFR 201.6). These required State and local governments to develop and formally adopt Hazard Mitigation Plans in order to be eligible for certain disaster mitigation grant funding sources.

In June, 2003 Drainage districts within the State of Texas were granted additional authority via Chapter 49.211 of the Texas Water Code and House Bill 919. Specifically, Chapter 49.211 of the Texas Water Code required districts to adopt master plans before they can adopt rules relating to review and approval of proposed development drainage plans. Further, HB 919 granted districts the authority to require developers to submit drainage plans for approval during the platting process. The bill also provided legal authority for DD6 to halt subdivision plans based on floodplain development control and regulation. DD6 was created primarily to provide drainage of over flow lands within DD6, including the construction and maintenance of drains, ditches and levees, and other improvements of the District.

Emergency response is the responsibility of the City of Beaumont and Jefferson County. The City owns and maintains several roadside ditches, however DD6 owns the majority of ditches within DD6 and is responsible for routine maintenance. After an event, it is a cooperative effort between the City, County Precincts, and DD6 to identify ditches that need cleaning (as well as crossings). There are known problem areas that are regularly checked during and after an event.

Both the City and the County have early warning capability. Citizens in the area rely mostly on local weather, which is reported to be very capable. DD6 has over 59 stream and rainfall gauges throughout DD6. These stream gauges provide data that is used by DD6 and the Lake Charles branch of the National Weather Service to predict potential flooding. DD6 uploads stream gauge data to the National Weather Service every 15 minutes.

Further discussion on existing policies and programs are addressed in "review and incorporation of existing plans, studies, reports and technical information".

# THE PLANNING PROCESS

The process used to develop this Plan was guided by a Mitigation Planning Committee (MPC), which carried out most of the planning duties. The MPC determined that in addition to the small committee that would steer the planning process, a larger group of interested individuals called Stakeholders would be included in the planning process to review drafts and provide comments at critical points in the plan development. At the first Plan update meeting held on February 25, 2016, the MPC finalized who would comprise the MPC (below). The MPC Meeting Minutes can be found in Appendix 1 of this Plan. The MPC was responsible for data collection and update, review and update each section of the plan, provide status of the mitigation actions from the previous plan and provide any new actions for review by the Stakeholders and the public. The Stakeholders group was larger and comprised of individuals and organizations from both inside and outside Jefferson County DD6. The MPC was comprised of the following individuals:

Team Member	Job Title	Organization
Ms. Karen Stewart	Business Manager-Purchasing Agent, Grant Manager, Director of HR & Risk Management	DD6
Mr. Doug Canant	District Engineer	DD6
Mr. Chuck Oakley	CFO	DD6

### Table 1 – Mitigation Planning Committee

Ms. Adina Ward	Floodplain Administrator	City of Beaumont
Mr. Thomas Gill	Streets and Drainage Manager	City of Beaumont
Mr. Don Rao	Jefferson County Engineering Director	Jefferson County
Mr. Jeff Ward	Mitigation Planning Consultant	JSWA
Mr. Dan Ward	Mitigation Planning Consultant	JSWA

As drafts of the Updated Plan were prepared, the MPC used email to distribute them to Stakeholders, and requested that they provide comments. Stakeholders were also invited to public meetings and were requested to provide feedback through email or by telephoning Jefferson County DD6 or a member of the consultant team. When the final draft was completed by the MPC, the Stakeholders were notified by mail and provided a website to download and review the plan for any updates or comments. The consultant was responsible for archiving the comments and including them in edited versions of the Plan update. The Stakeholders Group was comprised of the following individuals and entities:

Group Member	Job Title	Organization
Mr. Richard LeBlanc	General Manager	DD6
Mr. Gilbert Ward	C.P.G. Hydrologist Water Resources	Texas Water
	Planning Water Supplies Section	Development Board
Mr. Randall Reese	General Manager	Sabine Neches
		Navigation District
Dr. John W. Frossard	Beaumont Independent School	Beaumont Independent
	District Superintendent	School District
Mr. Shannon Holmes	Hardin Jefferson Independent	Hardin Jefferson
	School District Superintendent	Independent School
		District
Ms. Pamela Lechler	Hamshire Fannett Independent	Hamshire Fannett
	School District Superintendent	Independent School
		District
Dr. Kenneth Evans	President	Lamar University
Dr. Paul J. Szuch	President	Lamar Institute of
		Technology
Ms. Kim Moncla	Executive Director - Foundation,	Baptist Beaumont
	Baptist Hospitals of Southeast Texas	Hospital
Mr. Paul Trevino	CEO, Christus Southeast Texas	Christus St. Elizabeth
		Hospital
Mr. Phil Kelley	Manager	Jefferson County
		Drainage District 7
Mr. Leroy Mc Call Jr	Manager	Jefferson County
		Drainage District 3

### Table 2 – Stakeholders Group Members

Mr. Kenneth Wiemers	Area Engineer Beaumont	Texas Department of
P.E.		Transportation
Mr. Scott Hall, P.E.	General Manager	Lower Neches Valley
		Authority
Mr. Patrick Trahan	Government Relations Director	Exxon Mobil Oil
		Corporation
Mr. Michael Lockwood	Plant Manager	Goodyear Tire and
		Rubber
Ms. Sue Landry	Homeland Security Director	Southeast Texas Regional
		Planning Commission
Mr. Kerry Abney	Mayor	City of Nome, TX
Mr. John Walker	Mayor	City of China, TX
Ms. Becky Ford	Mayor	City of Bevil Oaks, TX

### **PUBLIC INVOLVEMENT**

Consistent with DD6's standard objective to inform and involve citizens, and to fulfill the public involvement requirements of the mitigation planning programs, during the plan development process, DD6 notified and invited residents to review the document and attend two public meetings. For the first public meeting, a public notice was published informing the public about the Hazard Mitigation Planning process and urged the public to be involved in this process.

The first public meeting was held on April 14, 2016 to review and comment on a preliminary draft of the Plan update. In accordance with legal requirements, DD6 published public notices about the presentation in the Beaumont Enterprise prior to the meeting (See Appendix 2, Public Notice Documents). The notice explained the purpose of the meeting and provided the date, time, and location of the meeting. The meeting minutes (and attendee lists) for the public meeting is included in Appendix 2 of the Plan update.

The public had a second opportunity to review the final draft Plan when the document was posted on the Jefferson County DD6 website at www.dd6.org/plan.pdf and placed at the City of Beaumont's City Hall and at DD6's office on Walden Rd. Prior to placing the document online, DD6 announced the availability of the final draft plan for review and provided a way for the public to provide comments. The public notice in the Beaumont Enterprise and the letters to all stakeholders explaining that the District's Hazard Mitigation Plan update was in the final draft stages and available for review are attached in Appendix 2). The stakeholder and public comments were incorporated and the plan was presented at the second public meeting on June 28<sup>th</sup> for final review and comment before submission. No one from the public attended either meeting.

# REVIEW AND INCORPORATION OF EXISTING PLANS, STUDIES, REPORTS AND TECHNICAL INFORMATION

A wide variety of existing plans, studies, reports and technical information were reviewed and incorporated into this Hazard Mitigation Plan. The following Plans are available: The Jefferson County DD6 Drainage and Flood Damage Reduction Plan (Master Drainage Plan), Drainage Regulations; Jefferson County Drainage District No. 6, Drainage Criteria Manual for Drainage District No. 6 and the Texas State Mitigation Plan. In addition, Flood Insurance Rate Maps (FIRM) and Flood Insurance Study were reviewed, though no changes were made since the last version of the Plan. The specific Plans, Studies and Reports are listed below along with a discussion on how they were incorporated into the Plan update.

- Jefferson County DD6 Drainage and Flood Damage Reduction Plan (Master Drainage Plan). Jefferson County DD6 prepared the Flood Damage Reduction Plan to examine how development is reviewed and to satisfy the requirements of HB 919 so that DD6 could develop, adopt, implement, and enforce regulations relating to its review and approval of development proposals. DD6 meets on an annual basis to review this plan, specifically to select the best way to expand on the District's capability to enforce development restrictions throughout the service area. The District completes periodic reviews of the Master Drainage Plan to identify mitigation actions that can be incorporated in the Hazard Mitigation Plan.
- Drainage Regulations; Jefferson County Drainage District No. 6, Jefferson County DD6 completed Drainage Regulations within DD6. The regulations were adopted by DD6 to protect, maintain and enhance public health, safety and general welfare, and to minimize the impacts of increases in stormwater runoff and flooding. The District completes periodic reviews of the Drainage Regulations to identify mitigation actions that can be incorporated in the Hazard Mitigation Plan. DD6 meets on an annual basis to review these regulations, specifically to identify ways to enhance and/or tighten drainage regulations to further minimize impacts of increases in stormwater runoff and flooding.
- Drainage Criteria Manual for Drainage District No. 6, DD6 completed the Drainage Criteria Manual. This manual was completed to support the Master Drainage Plan and Drainage Regulations that were adopted by Jefferson County DD6 pursuant to the authority set forth in the Texas Water Code §49.211. The purpose of the Drainage Criteria Manual is to outline criteria and guidance to be used by developers, engineers, and land surveyors in the design of drainage measures to manage runoff. The District completes periodic reviews of the Criteria Manual to identify mitigation actions that can be incorporated in the Hazard Mitigation Plan. DD6 meets on an annual basis to review these regulations, specifically to identify ways to expand criteria and guidance to be used by developers, engineers, and land surveyors in the design of drainage measures to manage runoff.

- **2013 State of Texas Mitigation Plan Update**. The mitigation strategies and goals from the State Plan were reviewed. This plan also incorporated useful tables, figures and information from the State of Texas Mitigation Plan.
- Flood Protection Planning Study; City of Beaumont and Jefferson County Texas; This study focuses on the Hillebrandt Bayou Watershed and shares some of the same actions and potential projects as this Hazard Mitigation Plan.
- Jefferson County Flood Insurance Rate Map (FIRM). The Flood Insurance Rate Maps (FIRMs) prepared by FEMA offer the best overview of flood risks. FIRMs are used to regulate new development and to control the substantial improvement and repair of substantially damaged buildings. The Jefferson County and City of Beaumont FIRMs were reviewed and included in the Plan update to develop a floodplain map identifying the 100-year floodplain within DD6.
- Jefferson County and the City of Beaumont Flood Insurance Study (FIS). The most recent FIS's for both the City of Beaumont and Jefferson County are dated August 6, 2002. These studies were reviewed again as part of the Plan update.
- Jefferson County Hazard Mitigation Plan. The plan is currently undergoing its update but reviewed the actions and risk assessment.

# INCORPORATION OF THE HAZARD MITIGATION PLAN INTO OTHER PLANNING MECHANISMS

As part of the original HMP development, DD6 integrated components of the Plan into other planning mechanisms. In addition to incorporating some of the Plan requirements in the DD6 Master Drainage Plan, elements of the previous versions of this Plan have been incorporated into the Flood Protection Planning Study; as well as City of Beaumont and Jefferson County Texas hazard mitigation plan. The MPC is currently reviewing the plans listed above and looking for opportunities where components of this HMP update can be integrated into these other plans and studies as well as new plans or studies. Mitigation Plan requirements have been incorporated into DD6's annual project planning and budgeting process. A member from DD6 attends Planning Committee Meetings for both the City and Beaumont and Jefferson County. During these meetings hazard mitigation projects, goals and priorities identified in the DD6 HMP are discussed and then considered by all Planning Committee Members for incorporation into both the City and county level HMP's. Many of the goals and priorities identified in this HMP are similar to the goals and priorities identified in this HMP are similar to the goals and priorities identified in this HMP are similar to the goals and priorities for the City of Beaumont and Jefferson County which will help to ensue this plan is incorporated into those planning mechanisms.

### PLAN MAINTENANCE AND CONTINUED PUBLIC INVOLVEMENT

Upon adoption of this Plan update, the public will be notified of any substantial changes to the document between 2016 and the next scheduled Plan update in 2021. Any changes

proposed by the MPC considered significant will be distributed to the list of Stakeholders. The Stakeholders will be encouraged to review the changes and provide comments on any proposed plan revisions.

DD6 will involve the public in the plan maintenance process and during the next Plan Update in 2021, using the same methods as the plan development. The public will be notified when the revision process is started and provided the opportunity to review and comment on changes to the plan and priority action items. It is expected that a combination of informational public meetings, draft documents posted on the web site, and public Board of Director meetings will be undertaken.

The DD6 Hazard Mitigation Plan update will be posted on the District's Web site and notices of its availability will be distributed to the federal and State agencies that were notified and invited to participate in Plan development, Jefferson County, City of Beaumont, City of Bevil Oaks, City of Nome, City of China, adjacent counties and cities, Citizens who attended public meetings and provided contact information; and the organizations, agencies, and elected officials who received notices of public meetings.

# PLAN MONITORING, EVALUATING AND UPDATING

The Mitigation Planning Committee determined that progress would be best monitored by annual meetings of the MPC. Upon adoption in 2016, the MPC will meet on an annual basis to discuss the status of the Plan and determine if any significant changes are warranted. As part of the meeting, the Business Manager of DD6 will note progress made on the prior mitigation action items listed in Table 30. To this end, the Business Manager may convene a meeting of the appropriate District, City of Beaumont and Jefferson County Departments to discuss and determine progress, and to identify obstacles to progress, if any.

In addition to annual meetings, the Business Manager will convene meetings after damagecausing natural hazard events to review the effects of such events. Based on those effects, adjustments to the mitigation priorities listed in Table 30 may be made or additional eventspecific actions identified.

DD6 will initiate Plan reviews and updates based on the following:

1. On the recommendation of the Business Manager or on its own initiative, DD6 Board may initiate a Plan review at any time.

2. At approximately the one-year anniversary of the Plan's re-adoption, and every year thereafter.

3. After natural hazard events that appear to significantly change the apparent risk to District assets, operations and/or citizens.

4. When activities of DD6, County, or the State significantly alter the potential effects of natural hazards on District assets, operations and/or citizen. Examples include completed mitigation projects that reduce risk, or actions or circumstances that increase risk.

5. When new mitigation opportunities or sources of funding are identified.

In addition to the circumstances listed above, revisions that warrant changing the text of this Plan or incorporating new information may be prompted by a number of circumstances, including identification of specific new mitigation projects, completion of several mitigation actions, or requirements for qualifying for specific funding.

Major comprehensive review of and revisions to this Hazard Mitigation Plan update will be considered on a five-year cycle. To be adopted in 2016, the Plan will enter its next review cycle sometime in 2021. The Mitigation Planning Committee will be convened to conduct the comprehensive evaluation and revision.

The 2021 Update of this plan will begin 2 years prior to expiration of this plan. The Mitigation Planning Committee will begin by reviewing the meeting notes from the Annual review and evaluation meetings that will be taking place throughout the next five years. The planning committee will also review any changes in development and disasters that have occurred within the District since the last version of this Plan. This information will help determine hazards to be included in the Update of this plan and possibly identify mitigation actions needed to address hazards based on the changes in new development. The next Update will follow the same planning process to allow the public input on hazards and prioritization of actions.

# **SECTION 2 – HAZARD ASSESSMENT**

# INTRODUCTION

DD6 is a conservation and reclamation district and a political subdivision of the State of Texas. DD6 was established January 21, 1920, after favorable vote on January 10, 1920. It was created primarily to provide drainage of over flow lands within DD6, including the construction and maintenance of drains, ditches and levees, and other improvements of the District.

Although DD6 is subject to a range of hazards typical of the northern Gulf Coast, for the reasons outlined below, DD6 has determined that the most appropriate and useful approach to developing its mitigation plan is to eliminate certain hazards from detailed risk assessment in its Hazard Mitigation Plan. There are three reasons for this: (1) DD6's mission and jurisdictional authority is explicitly limited to activities related to controlling floods (although the organization does have the authority to complete actions to protect and mitigate damage to its own facilities); (2) non-DD6 assets and populations that are potentially exposed to hazards are part of another mitigation plan, and hence including them in the present document would be redundant and serve no meaningful purpose – Jefferson County and the City of Beaumont have both the authority and the responsibility to sponsor mitigation activities for their constituent populations and communities, and; (3) the hazards were determined not to affect District Facilities. DD6 will continue to coordinate with the County and City to ensure that mitigation actions are developed and implemented in a rational manner, reducing or eliminating conflict and overlap between the jurisdictions.

During the Crosswalk review process in the 2010 version of this Plan, the FEMA reviewer highlighted the confusion and inconsistency of addressing a hazard and completing a risk assessment for a hazard that DD6 had no jurisdictional authority to mitigate against and that has no impact on DD6 owned facilities. As opposed to removing any discussion of the hazard, which would have warranted an entire re-write of the plan, it was agreed to provide a narrative discussion for each of these hazard indicating that there is no negative impact to DD6 operations or facilities. As such, it has been determined that the planning area, based on DD6 jurisdictional authority, and DD6 owned facilities will not be negatively impacted from the below hazards. For this reason, the hazards listed below have been eliminated from further consideration and there are no mitigation action items associated with them.

- Extreme Heat This hazard does not affect District-owned facilities and DD6 has no authority to mitigate against this hazard.
- Drought This hazard does not affect District owned facilities and DD6 has no authority to mitigate against this hazard.
- Winter Storm While winter storm can cause pipes to freeze, the need for ice and snow to be removed, and downed power lines, the District facilities have been built to insulate the pipes, have backup generators for downed power lines and have the

necessary equipment to remove ice and snow. This hazard does not affect District owned facilities and DD6 has no authority to mitigate against this hazard.

- Earthquake Due to the extremely low probability of an earthquake within the planning area, and the fact that there is no record of any historical building damage as a result of seismic activity in the planning area, this hazard does not affect District owned facilities and DD6 has no authority to mitigate against this hazard.
- Land Subsidence Due to the extremely low probability of a land subsidence within the planning area and the fact that there is no record of any historical occurrences of land subsidence in the planning area, this hazard does not affect District owned facilities and DD6 has no authority to mitigate against this hazard.
- Wildfire Due to the low probability of Wildfire in the Planning area and the fact that DD6 owned facilities are located in an urban area, this hazard does not affect District owned facilities and DD6 has no authority to mitigate against this hazard.
- Coastal Erosion Jefferson County, not DD6, maintains beaches and dune systems and the District has no authority to mitigate against this hazard.
- Expansive Soils Severe damage from expansive soils is not well documented so its occurrence as catastrophically damaging is not documented. This hazard does not affect District owned facilities and DD6 has no authority to mitigate against this hazard.
- Hailstorms DD6 buildings are built to withstand hail damage and the District has covered parking garages and storage areas to protect all assets from hail damage. This hazard does not affect District owned facilities and DD6 has no authority to mitigate against this hazard.
- Lightning DD6 facilities are all built to be protected from lightning and therefore, lightning does not affect District owned facilities and DD6 has no authority to mitigate against this hazard.
- Dam/Levee Failure FEMA and the U.S. Army Corps of Engineers (USACE) maintain the National Inventory of Dams, a database of high and significant hazard dams. For the most part, data are provided by State agencies responsible for regulation and inspection of dams or by the USACE. Based on that inventory, there are no high hazard dams that affect the watersheds in or draining through DD6.

The hazards that DD6 will address as part of this plan update are:

- Tornado
- Thunderstorm/High Wind
- Hurricane and Tropical Storm
- Flood

# **GEOGRAPHY, CLIMATE, AND POPULATION**

The area covered by Jefferson County Drainage District No. Six (DD6) is located in southeast Texas. The City of Beaumont is the County seat and the largest City of Jefferson County. The City is situated approximately 85 miles east of Houston, approximately 70 miles northeast of Galveston, and 275 miles southeast of Dallas (Figure 1). Ground surface elevations across DD6 vary from 37 feet to 3 feet above mean sea level. The topography is described as nearly flat prairie and the geologic structure is nearly flat strata. The bedrock types are comprised of deltaic sands and muds. Data from the Bureau of Economic Geology, at the University of Texas at Austin, identifies the land as "expansive clay and mud – locally silty, locally calcareous, flat to low; hilly prairie; commonly tilled".



### Figure 1 - Vicinity Map: State of Texas (Source: Mapquest)

The climate of the region is humid subtropical, with warm summers and moderate winters. Rainfall is abundant and on the average, evenly distributed throughout the year. The heaviest rains usually occur during the hurricane season, which extends from June through October. Average annual precipitation for the area is approximately 56 inches and the average annual temperature is about 69 degrees. Jefferson County Drainage DD6 consists of approximately 487 square miles and lies entirely within Jefferson County and the City of Beaumont. Figure 2 is a map identifying the boundary area (shown in pink) for Jefferson County DD6. The DD6 Hazard Mitigation Plan Update is prepared for the entire District.



According to the United States Census Bureau, Jefferson County as a whole had an estimated total population of 252,235 in 2014. This is a 3.7 percent increase from the 2008 census data, which estimated the Jefferson County population at 243,090. In 2014, the population of the labor force in Jefferson County was 109,639, approximately a one percent increase from the estimated labor force in year 2000. The original Plan indicated the top three industries in the County were education, health, and social services. As of 2013, the top three industries were health care, educational services and public administration as indicated by Table 3.



Table 3 – Most Common Industries, Beaumont, Texas (Source: www.City-data.com)

Jefferson County includes both incorporated and unincorporated areas. The population totals for the eight incorporated areas within the County are identified in Table 4. As indicated in the table, the cities of Beaumont, Bevil Oaks, China, and Nome are located within Jefferson County DD6. The population of the four cities within the planning area consists of over half the County population. The remaining incorporated areas are located outside of the planning area. The population of unincorporated Jefferson County totals 31,562.

City	Overall Population	Within DD6 Planning	Population within
		Area	Planning area
Beaumont	117,585	Yes	117,585
Bevil Oaks	1,244	Yes	1,244
China	1,127	Yes	1,127
Nome	561	Yes	561
Groves	15,753	No	N/A
Nederland	17,108	No	N/A
Port Arthur	54,540	No	N/A
Port Neches	12,755	No	N/A
Total	220,673		120,517

Table 4 - Incorporated Areas of Jefferson County (Source: US Census Bureau,	2014 -
Estimates)	

The City of Beaumont is the largest municipality in the planning area and as of 2014 had an estimated population of 117,585.

Table 5 identifies the total number and estimated value of buildings/infrastructure within Jefferson County DD6. The table indicates there are 50,266 residential buildings and 7,666 commercial buildings. As shown in Table 4 of the Plan update, the total population of the incorporated areas within DD6 is 120,517. The total population in DD6 is slightly higher than this figure when considering the additional residents living within the unincorporated areas. The data in Table 4 is used periodically throughout this section to identify the overall District-wide exposure for certain hazards that equally impact the entire planning area such as hurricanes/tropical storms.

 Table 5 – Buildings/Infrastructure within Jefferson County Drainage District Six (Sources:

 Jefferson County Central Appraisal District)

Туре	Number of Structures	Estimated Value
Residential Buildings*	50,266	\$4,933,674,187
Commercial Buildings*	7,666	\$16,065,585,012
District owned Buildings or	19	\$3,342,142
structures**		
Total	57,951	\$21,202,601,341

\* – Value and number of structures based on percent of County population in the Planning Area.

\*\* –Value based on insured value of District owned structures

### **OVERVIEW OF RISKS**

Numerous federal agencies maintain a variety of records regarding losses associated with natural hazards. Unfortunately, no single source is considered to offer a definitive accounting of all losses. FEMA maintains records on federal expenditures associated with declared major disasters. The U.S. Army Corps of Engineers (USACE) and the Natural Resources Conservation Service collect data on losses during the course of some of their ongoing projects and studies. National Oceanic and Atmospheric Administration's (NOAA) National Climatic Data Center (NCDC) database is another source where data statistics such as injuries, deaths, and damage estimates are maintained for a variety of natural hazards. The data is maintained at the county-wide level, with more recent entries listing the specific location within the county. Although not always specific to DD6, this county-wide hazard data from the NCDC is often the best available resource for documenting historical events. For the hazards profiled, the query results from the NCDC database are provided in the hazard specific subsections.

In the absence of definitive data on some of the natural hazards that may occur in DD6, illustrative examples are useful. Table 6 provides brief descriptions of particularly significant natural hazard events occurring in DD6's recent history. This list is not meant to capture every

event that has affected the area, rather lists one or two examples of the types of events that have affected the area in the past.

Data on Presidential Disaster Declarations characterize some natural disasters that have affected the area. In 1965, the federal government began to maintain records of events determined to be significant enough to warrant declaration of a major disaster by the President of the United States. Presidential Disaster Declarations are made at the county level and are not specific to any one city or sub-area, such as DD6. Given that DD6 is responsible for drainage in a large portion of Jefferson County, it is likely that a disaster declaration for Jefferson County affected DD6 in some way. Between 1965 and 2009 twelve such disasters have been declared in Jefferson County and are identified in Table 6. In addition to the declared events, the table also includes several incidents which did not receive a Presidential Disaster Declaration. Since the last update, while the State of Texas has received 9 more Presidential Disaster Declaration, none of those have impacted or included Jefferson County.

Date & Disaster (DR)	Nature of Event
November 7, 1957	TORNADO (F3) – An F3 tornado touched down in Jefferson County. This tornado was 200 yards wide and stayed on the ground for 4 miles causing \$2.5M in damages, 2 deaths, and 59 injuries.
June 29, 1973 (DR-393)	SEVERE STORMS AND FLOODING – a massive storm hit the Houston Texas area dumping 10 – 15 inches of rain. In total the storm resulted in 10 deaths and over \$50M in damage.
April 26, 1979 (DR-580)	SEVERE STORMS, TORNADOES, AND FLOODING – (Nearly 300 Jefferson County/City of Beaumont policy holders filed flood claims resulting in over \$2.8 M in payments). Rainfall reported in amounts between 9.56 to 10.7 inches in the Beaumont area and 11.5 inches in Bevil Oaks are, flooded many communities along the Neches river and Taylor, Pine Island, and Hillebrandt Bayous. Pine Island crested at 34.29 feet at Sour Lake, surpassing a record 31 feet set in 1917. Many homes, businesses and roads in the area were damaged.

#### (Sources: Public Entity Risk Institute (PERI) website, FEMA, NCDC database)

Date & Disaster (DR)	Nature of Event
July 28, 1979 (DR-595)	STORMS AND FLASH FLOODS - (Over 100 Jefferson County/City of Beaumont policy holders filed flood claims resulting in over \$700K in payments). Tropical Storm Claudette formed in the Central Atlantic the morning of July 15, 1979. It never reached hurricane intensity as it wandered across the northern Caribbean, and the Gulf of Mexico 10 days, making landfall near Port Arthur the evening of the 24th. Rainfall was estimated at 11 inches in the Beaumont area. The area suffered severe wind damage to utilities.
September 26, 1980 (DR-632)	TROPICAL STORM DANIELLE - (Over 200 Jefferson County/City of Beaumont policy holders filed flood claims resulting in over \$1.5M in payments). Rains of 8-9 in. fell on most of Texas. Particularly hard hit were Fisher, Mitchell, Nolan, and Scurry Counties.
May 31, 1989 (DR-828)	SEVERE STORMS, TORNADOES AND FLOODING - (28 Jefferson County/City of Beaumont policy holders filed flood claims resulting in over \$500K in payments). Widespread rains caused flooding that resulted in five deaths and total damages of about \$50 million. The storm dumped between 10 and 15 inches of rain in the southeast Texas area. Homes in Bevil Oaks flooded.
July 18, 1989 (DR-836)	TROPICAL STORM ALLISON - (Over 400 Jefferson County/City of Beaumont policy holders filed flood claims resulting in over \$3.8M in payments). Tropical Storm Allison caused torrential rains of 10-15 in. from Houston to Beaumont. Houston Intercontinental Airport recorded 10.34 in. during 24 hours. The storm resulted in three deaths and over \$60M in damages.
November 15, 1994 (DR-1041)	SEVERE THUNDERSTORMS AND FLOODING - (Over 200 Jefferson County/City of Beaumont policy holders filed flood claims resulting in over \$5.5M in payments). A tropical, mid-latitude rainfall of unusual proportion on a 30- to 35-county area of southeast Texas resulted in catastrophic flooding. The intense rainfalls totaled more than 25 in. at several locations and more than 8 in. on much of southeast Texas. The storm resulted in 18 deaths and an estimated \$700M in damages.

Date & Disaster (DR)	Nature of Event
May, 1996	DROUGHT - Drought conditions continued across southeast Texas. Rainfall totals from January through May averaged 10 to 15 inches below normal. The main areas affected include farming and fire protection. Crop damage across the entire region exceeded 1 million dollars.
August 12, 1996	SEVERE LIGHTNING - As many as 9,000 lightning strikes this evening resulted in one man injured, one house fire, and several telephone poles damaged.
January 14, 1997	ICE STORM - A record ice storm paralyzed southeast Texas and southwest Louisiana. Around 90,000 electric customers across southeast Texas were without power for up to six days. Emergency shelters were opened for several nights due to the cold weather following the ice storm. More trees and power lines were knocked down in this ice storm than what came down during Hurricane Bonnie in 1986. Hundreds of homes received minor damage due to trees or tree limbs falling on roofs. Several house fires were directly or indirectly related to the ice storm, but fortunately there were only no injuries. Numerous traffic accidents attributed to icy roads led to several minor injuries. One death was indirectly attributed to the ice storm. Two men were electrocuted on Tuesday, January 21st, while doing cleanup work for a local electric company. One 48 year old man died, and a 19 year old man was seriously injured in the accident
August, 26 1998 (DR-1239)	TROPICAL STORM CHARLEY – (Limited damage in Jefferson County) Up to 16 in. of rainfall in south-central Texas caused flooding in many counties, to include Jefferson
October, 14 1998 (DR-1245 & 1257)	HURRICANE GEORGES - (23 Jefferson County/City of Beaumont policy holders filed flood claims resulting in over \$200K in payments). Tropical Storm Frances, and a localized thunderstorm that followed later in the same month, resulted in widespread flooding.
August 31, 2000	EXTREME HEAT - Record heat occurred in late August across southeast Texas. At the Southeast Texas Regional Airport, the all-time record high of 108 was tied on August 31st. Previously it had been achieved on July 14 1902.

Date & Disaster (DR)	Nature of Event
June 9, 2001 (DR-1379)	TROPICAL STORM ALLISON - (Nearly 500 Jefferson County/City of Beaumont policy holders filed flood claims resulting in over \$12 M in payments). Tropical Storm Allison produced flooding throughout Southeast Texas, Louisiana, and across the eastern United States. Damages were estimated at \$5 Billion and prompted a Presidential disaster declaration for 30 counties in Texas.
October 29, 2002 (DR-1439)	SEVERE STORMS, TORNADOES AND FLOODING – (Over 400 Jefferson County/City of Beaumont policy holders filed flood claims resulting in over \$8.7M in payments). This unnamed storm produced heavy rains, causing flooding throughout Jefferson County.
September, 24, 2005 (DR-1606)	HURRICANE RITA – Hurricane Rita made landfall just east of the Texas-Louisiana border. Along the coast of Jefferson County, storm surges near 10 feet occurred near Sabine Pass, where over 90 percent of the homes were severely damaged or destroyed. The storm surge backed up the Sabine River, and flooded a small section of downtown Orange with around four to five feet of storm surge. High winds estimated at over 100 mph snapped and uprooting trees, and damaged over 125,000 homes and businesses.
September 13, 2008 (DR-1791)	HURRICANE IKE - Ike delivered a 17.5-foot storm surge on Jefferson County's coastal plain and dropped anywhere from 6 to 20 inches of rain, depending on where in the County it was measured. The surge caused flooding in the county's sparsely developed coastal areas, though no flooding occurred as a result of heavy rain. In total, at least 4,000 homes were flooded in Jefferson County. Within DD6, the event caused no flood related property damages, mainly due to recently completed mitigation efforts.

Jefferson County Drainage District Six has the authority to mitigate tornadoes, hurricanes and thunderstorm/high winds when they threaten DD6 buildings and assets. They also have the authority to mitigate against the flood hazard in all aspects. The National Oceanic and Atmospheric Administration's (NOAA) National Climatic Data Center (NCDC) collects and maintains certain hazard data in summary format, indicating injuries, deaths, and estimated damages.

For each hazard profiled in the present section, the planning team assigned a highly likely, likely, occasional or unlikely probability of future occurrences. The hazard probability was assigned in accordance with table 7 below.

### Table 7 – Frequency of Hazard Occurrence

### Frequency of Occurrence:

Highly likely: Event probable in next year.

Likely; Event probable in next three years

 $\Box$ Occasional; Event possible in next five years

Unlikely; Event possible in next ten years

To deduce which hazards leave DD6 most vulnerable, the MPC ranked each hazard the potential to cause damage, disrupt continuity of operations or shutdown facilities by providing a classification. Definitions for overall vulnerability are subjective based primarily on future probability, impact and severity, with additional considerations for potential impacts locations of buildings, critical facilities and infrastructure. Vulnerability classification criteria are general and involve some degree of overlap amongst classes. Definitions for overall vulnerability classifications used are listed in Table 8.

Level of Vulnerability	Description
Very High	High probability of future occurrence and potential catastrophic severity
High	Moderate/high probability of future occurrence and potential critical severity
Moderate	Moderate probability of future occurrence and limited potential severity
Low	Low/moderate probability of future occurrence and limited/negligible potential severity

Table 8 – Level of Vulnerability

### TORNADO

A tornado is defined as a rapidly rotating vortex or funnel of air extending ground-ward from a cumulonimbus cloud. Most of the time, vortices remain suspended in the atmosphere and are visible as a funnel cloud. However, when the lower tip of a vortex touches the ground, the tornado becomes a force of destruction.

### **TORNADO LOCATION**



#### Figure 3 – Vicinity Map: State of Texas (Source: Mapquest)

Figure 4 illustrates the frequency of tornado strikes in Texas per 1,000 square miles. Texas is considered the U.S. "tornado capital." While Texas tornadoes can occur in any month and at all hours of the day or night, they occur with greatest frequency during the late spring and early summer months during late afternoon and early evening hours. Northern Texas is most vulnerable, but the area around DD6 experiences 1 - 5 tornadoes per 1,000 square miles. The tornado hazard affects the entire planning area equally.



Figure 4 – Tornado Activity in the U.S. (Source: NOAA – Storm Prediction Center)

### **TORNADO EXTENT**

Tornado damage severity is measured by the Enhanced Fujita Tornado Scale (EF-Scale). The Enhanced Fujita Scale assigns numerical values based on wind speed and categorizes tornadoes from zero to five representing increased degrees of damage. Tornadoes are related to larger vortex formations, and therefore often form in convective cells such as thunderstorms or in the right forward quadrant of a hurricane or tropical storm, far from the hurricane eye. Table 9 describes the categories for the Enhanced Fujita Tornado Scale. We can expect to experience a tornado ranging from EFO to EF5 in the planning area.

### Table 9 – Enhanced Fujita (EF) Scale

Enhanced Fujita (EF) Scale						
Enhanced Fujita Category	Wind Speed (mph)	Potential Damage				
EFO	65-85	Light damage. Peels surface off some roofs; some damage to gutters or siding; branches broken off trees; shallow- rooted trees pushed over.				
EF1	86-110	Moderate damage. Roofs severely stripped; mobile homes overturned or badly damaged; loss of exterior doors; windows and other glass broken.				
EF2	111-135	Considerable damage. Roofs torn off well-constructed houses; foundations of frame homes shifted; mobile homes completely destroyed; large trees snapped or uprooted; light-object missiles generated; cars lifted off ground.				
EF3	136-165	Severe damage. Entire stories of well-constructed houses destroyed; severe damage to large buildings such as shopping malls; trains overturned; trees debarked; heavy cars lifted off the ground and thrown; structures with weak foundations blown away some distance.				
EF4	166-200	Devastating damage. Well- constructed houses and whole frame houses completely leveled; cars thrown and small missiles generated.				
EF5	>200	Incredible damage. Strong frame houses leveled off foundations and swept away; automobile-sized missiles fly through the air in excess of 100 m (109 yd); high-rise buildings have significant structural deformation; incredible phenomena will occur.				

## PREVIOUS OCCURRENCES OF TORNADO

According to the NCDC database, Jefferson County experienced 101 tornadoes (42 F0s, 29 F1s, 2EF1s, 20 F2s, and 6 F3s) between 1950 and 2016 (experiencing just one event since the last planning effort was underway). There is not specific data for just Jefferson County DD6, but the hazard is reasonably predicted to have uniform probability of occurrence across the entire planning area. Note that prior to 2007, the Fujita (F) Scale was used. Again, note that the boundaries of the DD6 planning area do not correspond exactly to Jefferson County, which is the reporting area for the NCDC, so the figure should be regarded as a general indication of event history area-wide. For these events, the NCDC database reported three deaths, 142 injuries and just \$55.979 Million in damages. Table 10 summarizes the 31 tornadoes that resulted in at least \$50,000 in damages.

Table 10 – Tornado Events in Jefferson County with at Least \$50,000 in Property Damage
(Source: NCDC Storm Events Database)

Location	County/ Zone	St	Date	Time	T.Z.	Mag	Dth	Inj	PrD
Totals:							3	142	55.979M
GILLBURG	JEFFERSON CO.	ТХ	08/18/2009	12:57	CST-6	EF1	0	10	20.000M
GROVES	JEFFERSON CO.	ТХ	04/03/2000	03:20	CST	F1	0	0	3.000M
JEFFERSON CO.	JEFFERSON CO.	ТХ	11/07/1957	21:15	CST	F3	2	59	2.500M
JEFFERSON CO.	JEFFERSON CO.	ТХ	10/11/1970	14:47	CST	F2	0	0	2.500M
JEFFERSON CO.	JEFFERSON CO.	ТХ	10/11/1970	15:30	CST	F3	0	19	2.500M
JEFFERSON CO.	JEFFERSON CO.	ТХ	11/13/1972	05:25	CST	F2	0	0	2.500M
JEFFERSON CO.	JEFFERSON CO.	ТХ	06/11/1973	21:05	CST	F1	0	0	2.500M
JEFFERSON CO.	JEFFERSON CO.	ТХ	10/22/1979	04:44	CST	F2	0	0	2.500M
JEFFERSON CO.	JEFFERSON CO.	ТХ	01/31/1983	14:00	CST	F3	0	1	2.500M
JEFFERSON CO.	JEFFERSON CO.	ТХ	05/20/1983	13:15	CST	FO	0	0	2.500M
JEFFERSON CO.	JEFFERSON CO.	ТХ	05/20/1983	13:40	CST	FO	0	0	2.500M
JEFFERSON CO.	JEFFERSON CO.	ТХ	05/20/1983	13:45	CST	F2	1	9	2.500M
JEFFERSON CO.	JEFFERSON CO.	ТХ	05/20/1983	14:01	CST	F1	0	3	2.500M

BEAUMON T	JEFFERSON CO.	ТХ	10/13/2001	03:55	CST	F1	0	0	1.000M
NOME	JEFFERSON CO.	ТХ	01/01/1999	23:45	CST	F3	0	5	500.00K
CHINA	JEFFERSON CO.	ТХ	10/16/2006	05:00	CST-6	F1	0	0	300.00K
JEFFERSON CO.	JEFFERSON CO.	ТХ	11/07/1957	21:23	CST	F3	0	1	250.00K
JEFFERSON CO.	JEFFERSON CO.	ТХ	08/05/1964	18:20	CST	F2	0	6	250.00K
JEFFERSON CO.	JEFFERSON CO.	ТХ	07/15/1969	16:10	CST	F1	0	0	250.00K
JEFFERSON CO.	JEFFERSON CO.	ТХ	05/12/1972	07:23	CST	F2	0	2	250.00K
JEFFERSON CO.	JEFFERSON CO.	ТХ	07/29/1972	15:00	CST	F1	0	0	250.00K
JEFFERSON CO.	JEFFERSON CO.	TX	05/26/1973	03:45	CDT	F2	0	3	250.00K
JEFFERSON CO.	JEFFERSON CO.	ТХ	10/28/1974	13:55	CST	F2	0	3	250.00K
JEFFERSON CO.	JEFFERSON CO.	ТХ	09/05/1980	16:45	CST	F1	0	0	250.00K
JEFFERSON CO.	JEFFERSON CO.	ТХ	04/23/1981	11:15	CST	F2	0	2	250.00K
JEFFERSON CO.	JEFFERSON CO.	ТХ	04/10/1984	23:30	CST	FO	0	0	250.00K
JEFFERSON CO.	JEFFERSON CO.	ТХ	02/18/1987	04:50	CST	F1	0	0	250.00K
HAMSHIRE	JEFFERSON CO.	ТХ	04/03/2000	03:08	CST	F1	0	1	100.00K
BEAUMON T	JEFFERSON CO.	ТХ	11/18/2003	01:00	CST	FO	0	0	100.00K
NOME	JEFFERSON CO.	ТХ	06/09/2010	04:32	CST-6	EF1	0	0	100.00K
PORT ARTHUR	JEFFERSON CO.	ТХ	07/14/1997	15:05	CST	FO	0	3	50.00K

### TORNADO PROBABILITY

Though Jefferson County DD6 has experienced 101 tornadoes between 1950 and 2016, many of these events reported occurred on the same day and are considered part of the same storm system. Most of the tornado events are from EF0 to EF2, with the catastrophic tornado events occurring with far less chance. An average of damaging tornadoes across the planning area provides the probability. With 31 significantly damaging events over 66 years, the frequency of an event is:

### Table 11 – Frequency of Tornado Occurrence

### Frequency of Occurrence:

Highly likely: Event probable in next year

⊠Likely; Event probable in next three years

Occasional; Event possible in next five years

Unlikely; Event possible in next ten years

It should be noted that due to DD6's missions and jurisdictional authority being explicitly limited to activities related to controlling floods, they only have the authority to mitigate the effect of tornadoes on District owned facilities and personnel.

# TORNADO IMPACT AND VULNERABILITY

DD6's missions and jurisdictional authority being explicitly limited to activities related to controlling floods, they only have the authority to mitigate the effect of tornadoes on District owned facilities and personnel. The FEMA software used for assessing tornado risk is based exclusively on life safety. Since the last version of the plan, DD6 built a 3,000 sq. ft. tornado shelter built to house District Staff during a tornado event, in accordance with FEMA 361 - Design and Construction Guidance for Community Shelters. This building can easily house all 150 employees for the duration of a tornado event.

Considering the analysis is based entirely on avoided injuries and fatalities, the tornado risk for DD6 is considered \$0. Even though District facilities and personnel are not vulnerable to tornadoes, based on our analysis, other District assets such as tractors, bulldozers, dump trucks, excavators and many other vehicles totaling to \$18,551,880 in insured value, may still have some risk of being damaged by tornadoes either while in storage or on project sites. However, the size and number of vehicles owned by the District make trying to protect all of them from tornadoes infeasible. The District's vulnerability is considered Low as defined in Table 8.

### HURRICANES AND TROPICAL STORMS

Hurricanes, tropical storms, and typhoons, collectively known as tropical cyclones, are among the most devastating naturally occurring hazards in the United States. They present flooding, storm surge, and high wind hazards to the communities that they impact.

A hurricane is defined as a low-pressure area of closed circulation winds that originates over tropical waters.

Hurricanes bring high winds and heavy rains and are usually accompanied by high storm surge; a rapid rise of offshore water elevation primarily caused by the combination of extremely high winds over a large stretch of open water and low barometric pressure which accompany a hurricane, together working to create a dome of water near the eye of the hurricane. As the hurricane nears land, its winds push the dome toward the shore while the slope of the sea floor blocks the water's escape and it comes ashore as a rising surge.

### HURRICANE AND TROPICAL STORM LOCATION

A hurricane or tropical storm occurs every year during hurricane season which begins in May and ends in November. Some of these storms dissipate before making landfall, but have the chance of striking anywhere on the Golf Coast or eastern seaboard. According to the National Hurricane Center's Historical Hurricane Tracker, from 1950 to 2016, there have been ten hurricanes and 11 tropical storms within a 65 nautical mile radius of Jefferson County Texas. Figure 5 shows the planning area, indicated by the white arrow, and the ten hurricanes that came within 65 nautical miles. As shown by the figure, there were six H1s, two H2s, one H3 and one H4.



Figure 5 – Historical Hurricane Tracks for Jefferson County TX (National Hurricane Center)

### HURRICANE AND TROPICAL STORM EXTENT

A hurricane begins as a tropical depression with wind speeds below 39 mph. As it intensifies, it may develop into a tropical storm, with further development producing a hurricane. Hurricane winds blow in a large spiral around a relative calm center known as the "eye." The "eye", the storm's core, is an area of low barometric pressure and is generally 20 to 30 miles wide. The storm may extend outward 100 - 400 miles in diameter. As a hurricane approaches, the skies will begin to darken and winds will grow in strength. As a hurricane nears land, it can bring torrential rains, high winds, storm surges, and severe flooding. Table 12 and Table 13 below identify the criteria for each stage of development. The Saffir / Simpson Hurricane Scale is used to classify storms by numbered categories. Hurricanes are classified as Categories 1 through 5 based on central pressure, wind speed, storm surge height, and damage potential. We can

expect to experience a storm ranging from a tropical depression to a category 5 hurricane in the planning area.

Stage of Development	Criteria
Tropical Depression (development)	Maximum sustained surface wind speed is <
	39 mph
Tropical Storm	Maximum sustained wind speed ranges 39 -
	<74 mph
Hurricane	Maximum sustained surface wind speed 74
	mph+
Tropical Depression (dissipation)	Decaying stages of a cyclone in which
	maximum sustained surface wind speed has
	dropped below 39 mph

#### Table 13 – Saffir/Simpson Hurricane Scale

Storm Category	Central Pressure	Sustained Winds	Storm Surge	Potential Damage
1	> 980 mbar	74 - 95 mph	4 – 5 ft	Minimal
2	965 – 979 mbar	96 - 110 mph	6 – 8 ft	Moderate
3	945 – 964 mbar	111 – 130 mph	9 – 12 ft	Extensive
4	920 – 944 mbar	131 – 155 mph	13 – 18 ft	Extreme
5	< 920 mbar	> 155 mph	> 18 ft	Catastrophic

### PREVIOUS OCCURRENCES OF HURRICANES AND TROPICAL STORMS

The NCDC database was queried for previous events, but only six were shown, so the National Hurricane Center's (NHC) Historical Hurricane Tracker was used. This hurricane tracker showed ten hurricanes and 11 tropical storms within 65 nautical miles of Jefferson County between 1950 and 2016. There have been no hurricanes or tropical storms experienced by DD6 since the last version of this plan. There is not specific data for just DD6, but it is reasonable to assume that if a hurricane or tropical storm effected the county, it would also effect DD6 because of the size and magnitude of the hazard. The six events shown on the NCDC database were from 1998 to 2008 and caused \$1.255 Billion in property damage. The 21 events from the NHC are shown below in Figure 6.



### Figure 6 – Historical Hurricane Tracks for Jefferson County TX (National Hurricane Center)

### HURRICANE AND TROPICAL STORM PROBABILITY

DD6 has experienced 21 tropical storms and hurricanes between 1950 and 2016. Even though DD6 has not experienced a hurricane or tropical storm in over seven years, it is reasonable to assume that one such storm will affect the planning area once every three years or so.

### Table 14 – Frequency of Tropical Storm and Hurricane Occurrence

### **Frequency of Occurrence:**

□ Highly likely: Event probable in next year

 $\boxtimes$ Likely; Event probable in next three years

Occasional; Event possible in next five years

Unlikely; Event possible in next ten years

It should be noted that due to DD6's missions and jurisdictional authority being explicitly limited to activities related to controlling floods, they only have the authority to mitigate the flood portion of the hurricane and tropical storm hazard except where it is related to District owned facilities and property.

### HURRICANE AND TROPICAL STORM IMPACT AND VULNERABILITY

DD6's missions and jurisdictional authority being explicitly limited to activities related to controlling floods, they only have the authority to mitigate the effect of hurricanes and tropical storms on District owned facilities and personnel. The FEMA software used for assessing hurricane and tropical storm wind risk is based exclusively on life safety. Since the last version of the plan, DD6 built a 3,000 sq. ft. tornado and hurricane shelter built to house District Staff

during a tornado or hurricane event, in accordance with FEMA 361 - Design and Construction Guidance for Community Shelters. DD6 also installed hurricane shutters on their administrative building and their engineering building. Hurricane and tropical storm events have a very long warning time, so when an event is expected to hit, all employees will be evacuated other than essential personnel. That essential personal can easily stay within the hurricane shelter throughout the duration of an event. Considering the analysis is based entirely on avoided injuries and fatalities, the hurricane risk for DD6 is considered \$0. Even though District facilities and personnel are not vulnerable to hurricanes, based on our analysis, other District assets such as tractors, bulldozers, dump trucks, excavators and many other vehicles totaling to \$18,551,880 in insured value, may still have some risk of being damaged by hurricanes either while in storage or on project sites. However, the size and number of vehicles owned by the District make trying to protect all of them from hurricanes infeasible. DD6 closely monitors the weather and takes proactive steps, when possible, to move vulnerable equipment to higher ground when equipment is being operated or staged in a floodprone area. The District's vulnerability is considered Low as defined in Table 8.

### SEVERE THUNDERSTORMS AND HIGH WINDS

Several meteorological conditions can result in winds severe enough to cause property damage. High winds have been associated with extreme hurricanes traveling inland, tornadoes, and locally strong thunderstorms. Thunderstorms are the by-products of atmospheric instability, which promotes vigorous rising of air particles. A typical thunderstorm may cover an area three miles wide. The National Weather Service considers a thunderstorm "severe" if it produces tornadoes or winds of 58 miles per hour or more. Structural wind damage may imply the occurrence of a severe thunderstorm.

### SEVERE THUNDERSTORM AND HIGH WIND LOCATION



Figure 7 - Vicinity Map: State of Texas (Source: Mapquest)

Figure 8 illustrates the minimum wind speed that buildings should be designed to withstand for buildings in Texas according to the International Building Code. As you can see below, new construction in DD6 should be built to withstand three-second gusts up to at least 140 MPH in some places and 150 MPH in others.

Figure 8 – Basic Design Wind Speed (Source: International Building Code)

Notes:

1. Values are nominal design 3-second gust wind speeds in miles per hour (m/s) at 33 ft (10m) above ground for Exposure C category.

- 2. Linear interpolation between contours is permitted.
- 3. Islands and coastal areas outside the last contour shall use the last wind speed contour of the coastal area.

4. Mountainous terrain, gorges, ocean promontories, and special wind regions shall be examined for unusual wind conditions.

5. Wind speeds correspond to approximately a 7% probability of exceedance in 50 years (Annual Exceedance Probability = 0.00143, MRI = 700 Years).

### SEVERE THUNDERSTORM AND HIGH WIND EXTENT

The most widely accepted extent scale for wind is the Beaufort Wind Scale. The table below shows the force of the storm and the wind speed, classification and appearance that that is associated with each force. In the planning area we can expect to experience wind events ranging from light winds to hurricane force winds.
Force	Wind	WMO	Appearance of	Wind Effects				
	(Knots)	Classification	On the Water	On Land				
0	Less than 1	Calm	Sea surface smooth and mirror- like	Calm, smoke rises vertically				
1	1-3	Light Air	Scaly ripples, no foam crests	Smoke drift indicates wind direction, still wind vanes				
2	4-6	Light Breeze	Small wavelets, crests glassy, no breaking	Wind felt on face, leaves rustle, vanes begin to move				
3	7-10	Gentle Breeze	Large wavelets, crests begin to break, scattered whitecaps	Leaves and small twigs constantly moving, light flags extended				
4	11-16	Moderate Breeze	Small waves 1-4 ft. becoming longer, numerous whitecaps	Dust, leaves, and loose paper lifted, small tree branches move				
5	17-21	Fresh Breeze	Moderate waves 4-8 ft taking longer form, many whitecaps, some spray	Small trees in leaf begin to sway				
6	22-27	Strong Breeze	Larger waves 8-13 ft, whitecaps common, more spray	Larger tree branches moving, whistling in wires				
7	28-33	Near Gale	Sea heaps up, waves 13-19 ft, white foam streaks off breakers	Whole trees moving, resistance felt walking against wind				
8	34-40	Gale	Moderately high (18-25 ft) waves of greater length, edges of crests begin to break into spindrift, foam blown in streaks	Twigs breaking off trees, generally impedes progress				
9	41-47	Strong Gale	High waves (23-32 ft), sea begins to roll, dense streaks of foam, spray may reduce visibility	Slight structural damage occurs, slate blows off roofs				
10	48-55	Storm	Very high waves (29-41 ft) with overhanging crests, sea white with densely blown foam, heavy rolling, lowered visibility	Seldom experienced on land, trees broken or uprooted, "considerable structural damage"				
11	56-63	Violent Storm	Exceptionally high (37-52 ft) waves, foam patches cover sea, visibility more reduced					
12	64+	Hurricane	Air filled with foam, waves over 45 ft., sea completely white with driving spray, visibility greatly reduced					

Table 15 Dedulor Wind Sedie (Sourcer Hadonal Security and Sedie (Sourcer Hadonal Se
--

### SEVERE THUNDERSTORM AND HIGH WIND PREVIOUS OCCURRENCE

According to the NCDC database, Jefferson County experienced 201 events between 1950 and 2016, 83 of which had wind above 50 kts (experiencing 11 such events since the last planning effort was underway). There is not specific data for just Jefferson County DD6, but the hazard is reasonably predicted to have uniform probability of occurrence across the entire planning area. For these events, the NCDC database reported one death, 12 injuries and just \$3.128 Million in damages. There have been 83 events over 50 kts and 21 events that resulted in at least \$25,000 in property damage. Table 16 summarizes the 21 events that resulted in at least \$25,000 in damages.

Location	County/Zone	Date	Туре	Mag	Dth	Inj	PrD
Totals:					1	12	3.128M*
GROVES	JEFFERSON CO.	07/16/2002	Thunderstorm Wind	65 kts. E	0	0	1.500M
JEFFERSON (ZONE)	JEFFERSON (ZONE)	10/09/2009	High Wind	56 kts. EG	0	0	200.00K
CHINA	JEFFERSON CO.	07/14/1998	Thunderstorm Wind		0	0	150.00K
PORT ARTHUR	JEFFERSON CO.	03/16/1998	Thunderstorm Wind		0	0	100.00K
BEAUMONT	JEFFERSON CO.	05/10/1999	Thunderstorm Wind		0	0	100.00K
GROVES	JEFFERSON CO.	08/03/1999	Thunderstorm Wind		0	0	100.00K
JEFFERSON (ZONE)	JEFFERSON (ZONE)	12/16/2000	High Wind		0	0	100.00K
BEAUMONT	JEFFERSON CO.	08/14/1998	Thunderstorm Wind		1	1	75.00K
Port Arthur	JEFFERSON CO.	03/13/1995	Thunderstorm Wind	70 kts.	0	2	70.00K
Beaumont	JEFFERSON CO.	03/09/1994	Thunderstorm Wind	0 kts.	0	0	50.00K
NEDERLAND	JEFFERSON CO.	12/03/1997	Thunderstorm Wind		0	1	50.00K
BEAUMONT	JEFFERSON CO.	08/29/1998	Thunderstorm Wind		0	0	50.00K
BEAUMONT	JEFFERSON CO.	08/20/1999	Thunderstorm Wind		0	0	50.00K

## Table 16 – Severe Thunderstorm and High Wind Events in Jefferson County with at Least \$25,000 in Property Damage (Source: NCDC Storm Events Database)

HAMSHIRE	JEFFERSON CO.	04/29/2006	Thunderstorm Wind	50 kts. EG	0	0	50.00K
BEAUMONT	JEFFERSON CO.	08/31/1999	Thunderstorm Wind		0	0	25.00K
NOME	JEFFERSON CO.	02/28/2001	Thunderstorm Wind		0	0	25.00K
LA BELLE	JEFFERSON CO.	05/17/2002	Thunderstorm Wind		0	0	25.00K
BEAUMONT	JEFFERSON CO.	08/26/2002	Thunderstorm Wind		0	0	25.00K
BEAUMONT	JEFFERSON CO.	05/11/2004	Thunderstorm Wind	50 kts. EG	0	0	25.00K
BEAUMONT	JEFFERSON CO.	05/29/2005	Thunderstorm Wind	50 kts. EG	0	0	25.00K
BEAUMONT	JEFFERSON CO.	08/16/2010	Thunderstorm Wind	52 kts. EG	0	1	25.00K
Totals:					1	12	3.128M*

\*Property damage total is the total for all events, not just those with at least \$25,000 worth of damage.

### SEVERE THUNDERSTORM AND HIGH WIND PROBABILITY

Jefferson County DD6 has experienced 201 severe thunderstorm and high wind events between 1950 and 2016, 83 of which had wind speeds of at least 50 kts. With so many events occurring, a severe thunderstorm or high wind event is:

#### Table 17 – Frequency of Severe Thunderstorms and High Wind

#### **Frequency of Occurrence:**

Highly likely: Event probable in next year

Likely; Event probable in next three years

 $\Box$ Occasional; Event possible in next five years

Unlikely; Event possible in next ten years

It should be noted that due to DD6's missions and jurisdictional authority being explicitly limited to activities related to controlling floods, they only have the authority to mitigate high wind on District owned facilities and property.

# SEVERE THUNDERSTORM AND HIGH WIND IMPACT AND VULNERABILITY

DD6's missions and jurisdictional authority being explicitly limited to activities related to controlling floods, they only have the authority to mitigate the effects of severe thunderstorms and high wind on District owned facilities and personnel. The FEMA software used for assessing

high wind risk is based exclusively on life safety. Since the last version of the plan, DD6 built a 3,000 sq. ft. tornado and hurricane shelter built to house District Staff during a tornado or other high wind event, in accordance with FEMA 361 - Design and Construction Guidance for Community Shelters. DD6 also installed hurricane shutters on their administrative building and their engineering building. Between these three buildings, all District staff can remain inside and safe during a severe thunderstorm or high wind event. Considering the analysis is based entirely on avoided injuries and fatalities, the severe thunderstorm and high wind risk for DD6 is considered \$0. Even though District facilities and personnel are not vulnerable to severe thunderstorms and high winds, based on our analysis, other District assets such as tractors, buildozers, dump trucks, excavators and many other vehicles totaling to \$18,551,880 in insured value, may still have some risk of being damaged by severe thunderstorms and high winds either while in storage or on project sites. However, the size and number of vehicles owned by the District make trying to protect all of them from severe thunderstorms and high winds infeasible. The District's vulnerability is considered Low as defined in Table 8.

### FLOOD

Flooding is the accumulation of water within a water body (e.g., stream, river, lake, or reservoir) and the overflow of excess water onto adjacent floodplains. Floodplains are usually lowlands adjacent to water bodies that are subject to recurring floods. Floods are natural events that are considered hazards only when people and property are affected. Nationwide, hundreds of floods occur each year, making them one of the most common hazards in the U.S.

Floods result from rain events, whether short and intense, or long and gentle. Flood hazards are categorized as follows:

- Flash floods not only occur suddenly, but also involve forceful flows that can destroy buildings and bridges, uproot trees, and scour out new channels. Most flash flooding is caused by slow-moving thunderstorms, repeated thunderstorms in a local area, or heavy rains from hurricanes and tropical storms. Although flash flooding occurs often along mountain streams, it is also common in urban areas, where much of the ground is covered by impervious surfaces and drainage ways are designed for smaller flows. Flood Insurance Rate Maps typically show the 1%-annual-chance (100-year) floodplain for waterways with at least 1 square mile of drainage area. The flood hazard areas for waterways with less than one square mile of drainage area typically are not shown.
- Riverine floods are a function of precipitation levels and water runoff volumes, and occur when water rises out of the banks of the waterway. Flooding along waterways that drain larger watersheds often can be predicted in advance, especially where it takes 24 hours or more for the flood crest (maximum depth of flooding) to pass. In Jefferson County, riverine flooding is caused by large rainfall systems and thunderstorm activity associated with seasonal cold fronts. These systems can take as long as a day to pass,

giving ample opportunity for large amounts of rain to fall over large areas. The Flood Insurance Rate Maps show the 1%-annual-chance floodplains.

• Urban drainage flooding occurs where development has altered hydrology through changes in the ground surface and modification of natural drainage ways. Urbanization increases the magnitude and frequency of floods by increasing impervious surfaces, increasing the speed of drainage collection, reducing the carrying capacity of the land, and, occasionally, overwhelming sewer systems. Localized urban flooding is not usually shown on the Flood Insurance Rate Maps in areas with less than one square mile of contributing drainage area.

The Flood Insurance Rate Maps (FIRMs) prepared by FEMA offer the best overview of flood risks. FIRMs are used to regulate new development and to control the substantial improvement and repair of substantially damaged buildings. Flood Insurance Studies (FISs) are often developed in conjunction with FIRMs. The FIS typically contains a narrative of the flood history of a community and discusses the engineering methods used to develop the FIRMs. The study also contains flood profiles for studied flooding sources and can be used to determine Base Flood Elevations for some areas.

The revised FIS' for both the City of Beaumont and Jefferson County are dated August 6, 2002. These FIS' compile all previous flood information and include data collected on numerous waterways. Both FIS' indicate that riverine flooding results primarily from overflow of the streams and drainage ditches caused by rainfall runoff, ponding, and sheet flow. Storms occurring during the summer months are often associated with tropical storms moving inland from the Gulf of Mexico. Thunderstorms are common throughout the spring, summer, and fall months. The frequent hurricanes and tropical storms interrupt the summer with high winds, heavy rainfalls, and high storm surges. FIRM maps for the City of Beaumont and Jefferson County show flood zones:

- AE Zones along rivers and streams for which detailed engineering methods were used to determine Base Flood Elevations (BFEs). AE Zones (or A1-30 Zones) are shaded in gray.
- A Zones, which are areas inundated by the 100-year flood for which BFEs and Flood Hazard Factors (FHFs) have not been determined
- AH Zones, which are areas inundated by types of 100-year shallow flooding where depths are between one and three feet, and for which BFEs are shown, but no FHFs are determined.
- V Zones are areas along coasts subject to inundation by the 1-percent-annual-chance flood event with additional hazards associated with storm-induced waves. Because detailed hydraulic analyses have not been performed, no Base Flood Elevations (BFEs) or flood depths are shown. Mandatory flood insurance purchase requirements and floodplain management standards apply.
- VE Zones are areas subject to inundation by the 1-percent-annual-chance flood event with additional hazards due to storm-induced velocity wave action. Base Flood

Elevations (BFEs) derived from detailed hydraulic analyses are shown. Mandatory flood insurance purchase requirements and floodplain management standards apply.

- B Zones and Shaded X Zones, which are areas of "moderate" flood hazard, typically associated with the 500-year flood (or 0.2% annual chance).
- C Zones and Unshaded X Zones are areas of "minimal" flood hazard, typically considered to be "out of the floodplain." Although local drainage problems and ponding may still occur, these minor flood problems typically are not shown on the FIRM.

### **FLOOD LOCATION**

Figure 9 identifies the 100-year floodplain (shaded light blue) for Jefferson County DD6. The map shows the 100-year floodplain is predominately found along the southern half of DD6 near the Gulf of Mexico, the western edge of Sabine Lake and the tributaries leading into the Gulf.

**Figure 9 – Jefferson County DD6 – 100-year Floodplain Map** (Source: FEMA National Flood Insurance Program (NFIP) December 2009)



Figure 10 – Jefferson County TX Effective Flood Insurance Rate Map (Source: RiskMap6 Effective FIRM)



### **FLOOD EXTENT**

Flood severity is measured in various ways, including frequency, depth, velocity, duration and contamination, among others. In DD6, characterizing the severity of the flood hazard depends on what part of the District is being considered, but generally speaking the issues relate to how often floods occur. Historically, floods are and continue to be the most frequent, destructive, and costly natural hazard facing the State of Texas. This is also the case within the District.

In DD6, the kind of rainfall that causes flash flooding almost always comes from. This area receives the second greatest frequency of thunderstorms in the United States and is also favorable to frequent heavy rainfall, supporting an annual rainfall of approximately 60 inches. The flooding problems in the County are considered severe in some areas. The flat terrain and clay soils (which do not readily absorb water) found in this area contribute to the flood problem. In the District, there are nearly 8,000 active flood insurance policies, many of which sit within the floodplain.

Flooding can occur during any month of the year in DD6; however, the greatest likelihood of the occurrence is mid-summer to early winter. Mid-summer flooding (July, August, and September) is most likely to result from tropical storm and hurricane development. Flooding in the fall to early winter (October, November and December) usually results from stationary weak cold fronts.

DD6 has been actively pursuing projects to reduce the severity of flooding in the area. The majority of these projects have been drainage projects including detention basins, ditch improvements and floodwater diversions. Many of these projects have already reduced the 100-year flood levels in the project areas.





Depth of precipitation for 50-year storm for 1-hour duration in Texas.

Based on the above USGS map, the planning (in the red circle) area can expect, on average, an increase of 4.0" of water in one hour on the ground in a 50-year event.



Figure 12 – Depth of Precipitation for 100-year Storm for 1-hour duration in Texas (Source: USGS)

Depth of precipitation for 100-year storm for 1-hour duration in Texas.

Based on the above USGS map, the planning area (in the red circle) can expect, on average, an increase of 4.4" of water in one hour on the ground in a 100-year event.

### FLOOD PREVIOUS OCCURRENCE

The NCDC indicates that Jefferson County and DD6 have experienced 61 flood events between 1996 and 2016. Of this total, 23 flood events have occurred since the last planning effort was underway. The NCDC database provides no indication as to why there are no events prior to 1996, although presumably occurrences follow the same pattern and frequency as shown in the NCDC list. Property damages for these events totaled just over \$18.504 million. The NCDC reported two deaths and no injuries from the 61 flood events. The 23 flood events that have occurred since the last planning effort was under way are listed below.

Table 18 – Jefferson County Flood Events since Last Planning Effort (Source: NCDC Storm Events Database)

Location	County/Zone	Date	Туре	Mag	Dth	Inj	PrD
Totals:					2*	0	18.504M*
CHINA	JEFFERSON	01/04/2009	Flash		0	0	5.00K
	CO.		Flood				
FANNETT	JEFFERSON	04/18/2009	Flash		0	0	20.00K
	CO.		Flood				
(BPT)BEAUMONT-	JEFFERSON	04/27/2009	Flash		0	0 5.00K	
PT ART	CO.		Flood				
GILLBURG	JEFFERSON	09/09/2009	Flood		0	0	0.00K
	CO.						
GILLBURG	JEFFERSON	10/22/2009	Flash		0	0	100.00K
	CO.		Flood				
FT ACRES	JEFFERSON	10/22/2009	Flash		0	0	10.00K
	CO.		Flood				
AMELIA	JEFFERSON	10/26/2009	Flood		0	0	10.00K
	CO.						
GUFFEY	JEFFERSON	08/17/2010	Flash		0	0	1.00K
	CO.		Flood				
PEAR RIDGE	JEFFERSON	07/19/2011	Flash		0	0	10.00K
	CO.		Flood				
GUFFEY	JEFFERSON	01/25/2012	Flash		0	0	1.00K
	CO.		Flood				
BEVIL OAKS	JEFFERSON	03/20/2012	Flash		0	0	10.00K
	CO.		Flood				
GALLOWAY	JEFFERSON	07/13/2012	Flash		0	0	2.00K
	CO.		Flood				
HOLLYWOOD	JEFFERSON	01/09/2013	Flash		0	0	0.00K
	CO.		Flood				
GILLBURG	JEFFERSON	05/10/2013	Flash		0	0	50.00K
	CO.		Flood				
GILLBURG	JEFFERSON	10/31/2013	Flash		0	0	50.00K
	CO.		Flood				
PORT NECHES	JEFFERSON	07/18/2014	Flash		0	0	0.00K
	CO.		Flood				
(BPT)BEAUMONT-	JEFFERSON	03/21/2015	Flash		0	0	15.00K
PT ART	CO.		Flood				
FT ACRES	JEFFERSON	04/16/2015	Flash		0	0	0.00K
	CO.		Flood				
BEAUMONT	JEFFERSON	05/12/2015	Flash		0	0	10.00K
	CO.		Flood				
AMELIA	JEFFERSON	05/21/2015	Flash		0	0	5.00K
	CO.		Flood				
AMELIA	JEFFERSON	05/27/2015	Flash		0	0	0.00K
	CO.		Flood				

GUFFEY	JEFFERSON	06/17/2015	Flash	0	0	1.00K
	CO.		Flood			
PORT ARTHUR	JEFFERSON	11/07/2015	Flash	0	0	0.00K
	со.		Flood			
Totals:				2*	0	18.504M*

\*Property damage and death total is the total for all events, not just those since the last planning effort.

### **FLOOD PROBABILITY**

Jefferson County and DD6 have experienced 61 floods between 1996 to 2016. With so many events occurring, future probability of a flood is:

#### Table 19 – Flood Frequency of Occurrence

### Frequency of Occurrence:

 $\boxtimes$  Highly likely: Event probable in next year

Likely; Event probable in next three years

Occasional; Event possible in next five years

Unlikely; Event possible in next ten years

### FLOOD IMPACT AND VULNERABILITY

To develop more specific data about flood-prone buildings, as part of the original Plan development DD6 worked with Jefferson County Engineering, Jefferson County Appraisal District (JCAD) and the City of Beaumont, who have access to a Geographic Information System (GIS) database. The tool that makes this possible is the GIS computer software application that relates physical features on the ground in mapping applications and analyses. The number of flood-prone residential and commercial buildings was re-evaluated in January, 2010, and again in April 2016. Updated figures are included in the building characterizations described below.

Flood insurance policies and claims information can be used to identify buildings in mapped floodplains (where lenders require insurance) and where flooding has occurred (where owners are sufficiently concerned that they purchase flood insurance even if not required). This characterization of flood risk is described below.

Data provided by FEMA indicate that as of January 1, 2016, federal flood insurance policies were in-force on 7,896 buildings in the City of Beaumont, Bevil Oaks, Nome, China and unincorporated Jefferson County. These insurance policies are administered by the National Flood Insurance Program (NFIP). Of those 7,896 buildings with NFIP policies still in force (active policies), 61 are Repetitive Loss structures and 19 (there are 37 SRLs in total but only 19 are currently insured) are Severe Repetitive Loss structures. The District's vulnerability to flood is considered very high according to table 8.

### **NFIP REPETITIVE LOSS PROPERTIES**

In recent years, FEMA has focused considerable attention on the Repetitive Loss (RL) subset of insured buildings. These properties have received two or more claim payments of at least \$1,000 over a ten-year period. FEMA's database identifies 293 properties as Repetitive Loss properties in DD6 (this number includes properties with active flood insurance policies as well as those with inactive policies). Collectively, they had received claim payments of almost \$20 million (includes payments for building damage and contents damage).

As of January 1, 2016, repetitive loss statistics for areas within DD6 (including unincorporated Jefferson County as a whole) showed 293 Repetitive Loss properties. Of this total, 264 were categorized as residential properties and 29 were non-residential.

The RL data for Jefferson County was broken down by eliminating the properties located in the incorporated areas outside of DD6. Although it is not possible to extract only the Jefferson County DD6 RL properties, by removing the incorporated areas outside DD6 this leaves only the properties located within the incorporated areas of DD6 and the unincorporated areas for all of Jefferson County. The population data indicates that the majority of the population within Jefferson County is located within DD6, and therefore this approach provides the closest method for estimating the RL properties in DD6. Table 20 summarizes the residential and non-residential properties for each municipality within DD6.

Municipality	Properties	Building	Contents	Total	# of claims	Average
Beaumont	191	\$7,713,913.72	\$2,791,355.04	\$10,505,459.76	620	\$16,944.29
Bevil Oaks	11	\$1,193,123.49	\$499 <i>,</i> 684.68	\$1,692,819.17	33	\$51,297.22
Jefferson County	91	\$4,806,661.38	\$1,661,362.53	\$6,468,023.91	327	\$19,780.17
Grand Total	293	\$13,713,698.59	\$4,952,402.25	\$18,666,393.84	980	\$19,047.34

 Table 20 - Summary of Residential and Non-Residential NFIP Repetitive Loss Statistics,

 Jefferson County DD6, ordered by Municipality (Source: FEMA NFIP query January 1, 2016)

As indicated above, it is estimated there are 264 residential RL properties in Jefferson County DD6. Table 21 summarizes the RL claims data by municipality. The table shows that the majority of the residential RL properties are located within the City of Beaumont. As of January 1, 2016, claim payments for all 264 residential properties totaled over \$13 million.

Table 21 - Summary of Residential NFIP Repetitive Loss Statistics, Jefferson County DD6, ordered by Municipality (Source: FEMA NFIP query January 1, 2016)

Municipality	Properties	Building	Contents	Total	# of claims	Average
Beaumont	167	\$4,288,851.97	\$1,116,889.52	\$5,405,741.49	513	\$6,278.45
Bevil Oaks	11	\$1,193,123.49	\$499,684.68	\$1,692,808.17	33	\$51,297.22
Jefferson County	86	\$4,583,726.25	\$1,546,725.90	\$6,130,452.15	315	\$19,461.75
Grand Total	264	\$10,065,701.71	\$3,163,300.10	\$13,229,001.81	861	\$15,364.69

Table 22 - Summary of Non-Residential NFIP Repetitive Loss Statistics, Jefferson County DD6,ordered by Municipality (Source: FEMA NFIP query January 1, 2016)

Municipality	Properties	Building	Contents	Total	# of claims	Average
Beaumont	24	\$3,425,059.75	\$1,674,465.52	\$5,099,525.27	67	\$76,112.32
Jefferson County	5	\$222,935.13	\$114,646.63	\$337,571.76	12	\$28,130.98
Grand Total	29	\$3,647,994.88	\$1,789,102.15	\$5,437,097.03	119	\$45,689.89

The RL claims can be further broken down from listing by municipality to focusing on individual street level data. Table 23 provides a summary of residential repetitive flood insurance claims for individual streets within Jefferson County DD6 that include two or more repetitive loss properties. The data displayed in the table summarizes the NFIP repetitive loss data for 37 individual streets in DD6 that include two or more repetitive loss property. For each street, the building, contents, and total claims data has been combined. Note that by selecting only streets with two or more repetitive loss properties, the table only includes 146 of the 264 residential RL properties estimated within Jefferson County DD6.

The table shows that for these 146 RL properties, claim payments totaled approximately \$9.5 million as of January 1, 2016. The data shows that Crow Road clearly has the street with the most repetitive loss properties in Jefferson County DD6. Address data about individual sites is omitted for reasons of confidentiality.

Street Name	Municipality	Properties	Building	Contents	Total	# of Claims	Average (Per Claim)
Alford Oaks Dr	Jefferson County	3	\$415,867.20	\$155,164.83	\$571,032.03	15	\$38,068.80
Blossom Dr	Beaumont, TX	3	\$24,165.48	\$11,160.40	\$35,325.88	8	\$4,415.74
Brockman St	Beaumont, TX	3	\$121,898.90	\$42,397.59	\$164,296.49	10	\$16,429.65
CARNAHAN PL	Beaumont, TX	3	\$54,773.31	\$9,550.02	\$64,323.33	11	\$5 <i>,</i> 847.58
Central Blvd	Jefferson County	10	\$322,946.58	\$165,295.43	\$488,242.01	26	\$18,778.54
Cherokee Ln	Beaumont, TX	3	\$127,952.86	\$55,199.47	\$183,152.33	8	\$22,894.04
Concord Rd	Beaumont, TX	2	\$26,965.30	\$10,019.50	\$36,984.80	5	\$7,396.96
Coolidge St	Beaumont, TX	7	\$329,727.28	\$51,969.70	\$381,696.98	26	\$14,680.65
Crow Rd	Beaumont, TX	24	\$427,472.08	\$11,860.89	\$439,332.97	60	\$7,322.22
Downs Rd	Beaumont, TX	2	\$19,443.27	\$0	\$19,443.27	4	\$4,860.82
Elinor St	Beaumont, TX	2	\$62,496.90	\$15,299.51	\$77,796.41	8	\$9,724.55
FM 365	Jefferson County	2	\$127,160.97	\$29,679.21	\$156,840.18	4	\$39,210.05
Folsom Dr	Beaumont, TX	2	\$67,453.36	\$7,914.05	\$75,367.41	6	\$12,561.24
Forsythe St	Beaumont, TX	2	\$43,262.55	\$18,734.76	\$61,997.31	6	\$10,332.89
Galveston St	Beaumont, TX	4	\$128,254.58	\$29,126.32	\$157,380.90	16	\$9,836.31
Heartfield Ln	Beaumont, TX	2	\$37,001.13	\$1,564.62	\$38,565.75	4	\$9,641.44
Hillebrandt Acres	Jefferson County	11	\$624,102.15	\$310,498.36	\$934,600.51	70	\$13,351.44
Jerry Dr	Jefferson County	3	\$405,481.89	\$151,959.96	\$557,441.85	9	\$61,937.98
Kenner Rd	Jefferson County	2	\$64,365.35	\$2,827.20	\$67,192.55	4	\$16,798.14
Madison St	Jefferson County	2	\$51,468.59	\$20,847.34	\$72,315.93	7	\$10,330.85
Marsh Rd	Jefferson County	3	\$259,317.65	\$92,517.30	\$351,834.95	13	\$27,064.23

## Table 23 – Summary of Residential NFIP Repetitive Loss Statistics, Jefferson County DD6,ordered by number of Properties on Each Street (Source: FEMA NFIP query January 1, 2016)

Street Name	Municipality	Properties	Building	Contents	Total	# of Claims	Average (Per Claim)
N 2 <sup>nd</sup> Ave	Jefferson County	2	\$95,624.21	\$25,484.44	\$121,108.65	4	\$30,277.16
N Killarney Dr	Jefferson County	2	\$65,395.25	\$4,485.73	\$69,880.98	10	\$6,988.10
Park St	Beaumont, TX	8	\$232,030.82	\$88,930.65	\$320,961.47	29	\$11,067.64
Phelan Blvd	Beaumont, TX	2	\$13,914.07	\$2,308.37	\$16,222.44	6	\$2,703.74
Pipkin St	Beaumont, TX	2	\$62,685.20	\$20,832.14	\$83,517.34	7	\$11,931.05
Rainbow Ln	Jefferson County	2	\$182,443.62	\$48,343.30	\$230,786.92	6	\$38,464.49
Redwood Dr	Beaumont, TX	2	\$45,184.68	\$6,772.04	\$51,956.72	4	\$12,989.18
River Bend Dr	Bevil Oaks, TX	6	\$1,040,460.30	\$392,753.84	\$1,433,214.14	21	\$68,248.29
River Rd	Bevil Oaks, TX	2	\$111,927.91	\$59,805.04	\$171,732.95	4	\$42,933.24
Roberts St	Beaumont, TX	4	\$129,777.03	\$59,057.41	\$188,834.44	10	\$18,883.44
Rockwell St	Beaumont, TX	6	\$135,719.39	\$44,328.45	\$180,047.84	17	\$10,591.05
S Major Dr	Beaumont, TX	2	\$47,396.22	\$9,263.98	\$56,660.20	8	\$7,082.53
Thomas Rd	Beaumont, TX	2	\$37,802.92	\$550	\$38,352.92	5	\$7,670.58
Vernon St	Jefferson County	3	\$207,742.50	\$40,615.91	\$248,358.41	18	\$13,797.69
W Euclid St	Beaumont, TX	4	\$157,623.00	\$79,591.79	\$237,214.79	15	\$15,814.32
W Lucas Dr	Beaumont, TX	2	\$18,008.74	\$0	\$18,008.74	4	\$4,502.19
Total		146	\$6,325,313.24	\$2,076,709.55	\$8,402,022.79	488	\$17,217.26

### FLOOD RISK TO RESIDENTIAL REPETITIVE LOSS PROPERTIES

Jefferson County DD6 has an extensive history of repetitive loss flood claims, so it is possible to perform a relatively simple statistical risk assessment using average annual losses and a present value coefficient calculation to project losses over a planning horizon. Residential flood risk is calculated by a simple methodology that uses the FEMA default present-value coefficients from the benefit-cost analysis software modules. To perform this calculation, the repetitive loss data were reviewed to determine an approximate period over which the claims occurred. This method should not be used for risk assessments for individual properties because of the generalizations that are used, but the method is appropriate for larger numbers of properties and policies that are spread over an entire jurisdiction. It is presumed that more accurate

figures would be somewhat higher because the underlying statistics are for properties that had flood insurance, were flooded, and had paid claims. There are nearly always some properties in a jurisdiction that are flooded in big events, and do not have flood insurance (or did not make claims), and are thus not represented in the sample.

Most of the flood claims in this query occurred between 1979 and 2015, a period of 36 years. As shown in Table 24, there have been 901 claims in the 36-year period, for an average number of 25.03 claims per year. Based on a 100-year horizon and a present value coefficient of 14.27 (the coefficient for 100 years using the mandatory Office of Management and Budget (OMB) discount rate of 7.0 percent), the projected flood risk to these properties is shown at the bottom of the table. It must be understood that individuals can obtain and cancel flood insurance policies, and the flood hazard depends on many variables, including the weather, so this projection is simply an estimate of potential damages. Nevertheless, it offers a useful metric that can be used in assessing the potential cost effectiveness of mitigation actions.

Table 24 – Projected 100-year Flood Risk in Jefferson County DD6 Repetitive Loss Area
(Source: FEMA NFIP query January 1, 2016)

Data	Value
Period in years	36
Number of claims	901
Average claims per year	25.03
Total value of claims	\$14,335,517.19
Average value of claims per year	\$398,208.81
Projected risk, 100-year horizon	\$5,682,439.73

The risk assessment for residential repetitive loss properties can be further broken down to the street level. Streets in Jefferson County with ten or more repetitive loss claims were considered to have sufficient claims history to perform a risk assessment. Table 25 displays the annual and 100-year risk for streets in Jefferson County with ten or more NFIP claims. The table shows that Hillebrandt Acres in Jefferson County is the street with the highest number of claims. Although Hillebrandt Acres has the highest number of claims, River Bend Dr in Bevil Oaks has the highest 100-year risk totaling \$568,110.16.

## Table 25 – Projected Future Damages (Risk) to NFIP Repetitive Flood Loss Properties inJefferson County DD6, ordered by 100-year Risk (Source: FEMA NFIP query January 1, 2016)

Street Name	Municipality	Properties	# of Claims	Total Paid	Average (Per Policy)	Annual Risk	100-year Risk
Alford Oaks Dr	Jefferson County	3	15	\$571,032.03	\$38,068.80	\$15,862.00	\$226,350.75
Anita St	Beaumont, TX	1	12	\$155,860.06	\$12,988.34	\$4,329.45	\$61,781.20
Brockman St	Beaumont, TX	3	10	\$164,296.49	\$16,429.65	\$4,563.79	\$65,125.30
Carnahan Pl	Beaumont, TX	3	11	\$64,323.33	\$5,847.58	\$1,786.76	\$25,497.05
Central Blvd	Jefferson County	10	26	\$488,242.01	\$18,778.54	\$13,562.28	\$193,533.71
Coolidge St	Beaumont, TX	7	26	\$381,696.98	\$14,680.65	\$10,602.69	\$151,300.44
Crow Rd	Beaumont, TX	24	60	\$439,332.97	\$7,322.22	\$12,203.69	\$174,146.71
Galveston St	Beaumont, TX	4	16	\$157,380.90	\$9,836.31	\$4,371.69	\$62,384.04
Hillebrandt Acres	Jefferson County	11	70	\$934,600.51	\$13,351.44	\$25,961.13	\$370,465.26
Josey St	Beaumont, TX	1	11	\$69,926.81	\$6,356.98	\$1,942.41	\$27,718.21
Marsh Rd	Jefferson County	3	13	\$351,834.95	\$27,064.23	\$9,773.19	\$139,463.46
N Kilarney	Jefferson County	2	10	\$69,880.98	\$6,988.10	\$1,941.14	\$27,700.04
Park St	Beaumont, TX	8	29	\$320,961.47	\$11,067.64	\$8,915.60	\$127,225.56
River Bend Dr	Bevil Oaks, TX	6	21	\$1,433,214.14	\$68,248.29	\$39,811.50	\$568,110.16
Roberts St	Beaumont, TX	4	10	\$188,834.44	\$18,883.44	\$5,245.40	\$74,851.87
Rockwell St	Beaumont, TX	6	17	\$180,047.84	\$10,591.05	\$5,001.33	\$71,368.96
Vernon St	Jefferson County	3	18	\$248,358.41	\$13,797.69	\$6,898.84	\$98,446.51
W Euclid St	Beaumont, TX	4	15	\$237,214.79	\$15,814.32	\$6,589.30	\$94,029.31
Total		103	390	\$6,457,039.11	\$16,556.51	\$179,362.20	\$2,559,498.56

Figure 13 shows the locations of the residential repetitive loss properties in Jefferson County DD6. The map highlights the total number of residential repetitive loss flood insurance claims per property in DD6. The map shows that the residential RL properties in DD6 are mainly concentrated within the City of Beaumont.



Figure 13 – Number of NFIP Flood Insurance Claims Per Residential Repetitive Loss Property in Jefferson County DD6 (Source: FEMA/NFIP, Query January 1, 2016; Plotted by DD6)

### NON-RESIDENTIAL REPETITIVE LOSS PROPERTIES

As noted earlier, as of January 1, 2016, Jefferson County DD6 had an estimated 29 nonresidential repetitive loss properties in the NFIP database. Table 26 provides a summary of nonresidential repetitive loss claims for individual streets in Jefferson County with at least five claims. The building, contents, and total claims data has been combined for streets that include more than one repetitive loss property. Similar to the residential repetitive loss data, address data about individual sites is omitted for reasons of confidentiality.

The Table shows that Calder and 11th Streets have the most non-residential repetitive losses in Jefferson County DD6. The data shows that these two streets rank highest in total paid claims (both building and contents combined) and number of past claims, indicating a significant history of flooding at these sites. Jefferson County DD6 has completed a drainage improvement project in the Hillebrandt Bayou watershed that includes areas of Calder Street and is nearing completion of a drainage improvement project on 11<sup>th</sup> St. Upon completion, the future risk in this area will be substantially reduced. Additional details about this project can be found in later in this Plan.

Street Name	Municipality	Claims	Properties	Total Claims (\$)	Annual Risk	100-year risk
College St	Beaumont, TX	6	1	\$37,456.78	\$1,040.47	\$14,847.45
Calder St	Beaumont, TX	26	1	\$1,464,159.10	\$40,671.09	\$580,376.40
Milam St	Beaumont, TX	6	1	\$60,751.68	\$1,687.55	\$24,081.29
Park St	Beaumont, TX	5	1	\$24,201.32	\$672.26	\$9,593.13
N 11 <sup>th</sup> St	Beaumont, TX	19	3	\$1,461,014.50	\$40,583.74	\$579,129.91
Cheek St	Beaumont, TX	5	1	\$300,016.08	\$8,333.78	\$118,923.04
Woodrow St	Beaumont, TX	6	1	\$247,669.06	\$6,879.70	\$98,173.26
Total		73	9	\$3,595,268.52	\$99,868.57	\$1,425,124.49

## Table 26 – Projected 100-year Flood Risk, Non-Residential Repetitive Loss Properties in Jefferson County DD6 (Source: FEMA NFIP query January 1, 2016)

It should be noted that some of the non-residential properties on this list may be at far greater flood risk than indicated, because there may be have been periods where the owner(s) did not carry flood insurance, with the result that they may have been damaged but there is no record of it. This type of analysis is not totally conclusive. It would be possible to perform relatively simple engineering studies to better assess risks for properties with just a few claims, but where data suggests that sites may be vulnerable to additional flood-related losses.

The information in this section should be used for planning purposes only, i.e. as the basis for additional steps in risk assessment, and eventually (where warranted) targeted mitigation

actions to reduce the risk. For example, a property that has received a number of claim payments not much higher than \$1,000 would be considered an unlikely candidate for mitigation using public funds. It may, however, be an excellent candidate for damage-reduction actions taken by the owner.

#### Figure 14 – Number of NFIP Flood Insurance Claims Per Non-Residential Repetitive Loss Property in Jefferson County DD6 (Source: FEMA/NFIP, Query January 1, 2016; Plotted by DD6)



### **NFIP SEVERE REPETITIVE LOSS PROPERTIES**

In 2004 FEMA began to develop the Severe Repetitive Loss (SRL) Grant Program in an effort to reduce or eliminate flood damages to residential properties that met certain minimum requirements. FEMA initiated the program early in 2008. The SRL Grant Program has since been included in the FMA Grant Program, with SRL properties being a top priority. An SRL property is defined as a residential property that is covered under an NFIP flood insurance policy and:

- has at least four NFIP claim payments (including building and contents) over \$5,000 each, and the cumulative amount of such claims payments exceeds \$20,000; or
- for which at least two separate claims payments (building payments only) have been made with the cumulative amount of the building portion of such claims exceeding the market value of the building.

SRL properties are a subset of the RL list and include only residential structures. As of January 1, 2016, Jefferson County had 37 properties on the SRL list all of which are located in either unincorporated Jefferson County, the City of Beaumont or the City of Bevil Oaks.

Table 27 provides loss estimates for SRL properties in DD6 summarized at the street level, as calculated by FEMA and the NFIP. The table shows that within Jefferson County, Hillebrandt Acres has the highest number of SRL properties.

## Table 27 – Projected 100-year Flood Risk, Severe Repetitive Loss Properties in JeffersonCounty DD6 (Source: FEMA/NFIP, Query January 1, 2016)

Street Name	Municipality	Claims	Properties	Total Claims (\$)	Annual Risk	100-year risk
Alford Oaks Dr	Jefferson County	9	1	\$368,135.66	\$10,225.99	\$145,924.89
Anita St	Beaumont, TX	12	1	\$155,860.06	\$4,329.45	\$61,781.20
Boussard St	Beaumont, TX	6	1	\$120,317.15	\$3,342.14	\$47,692.38
Brockman St	Beaumont, TX	8	2	\$157,817.39	\$4,383.82	\$62,557.06
Cherokee Ln	Beaumont, TX	4	1	\$103,456.78	\$2,873.80	\$41,009.12
Coolidge St	Beaumont, TX	6	1	\$127,564.16	\$3,543.45	\$50,565.02
Corley St	Beaumont, TX	7	1	\$100,985.69	\$2,805.16	\$40,029.61
Crow Rd	Jefferson County	4	1	\$43,251.86	\$1,201.44	\$17,144.56

~						
Elinor St	Beaumont, TX	5	1	\$54,052.98	\$1,501.47	\$21,426.00
Euclid St	Beaumont, TX	7	1	\$67,212.73	\$1,867.02	\$26,642.38
Galveston St	Beaumont, TX	6	1	\$49,013.39	\$1,361.48	\$19,428.36
Gross St	Beaumont, TX	4	1	\$78,260.48	\$2,173.90	\$31,021.58
Highway 105	Jefferson County	5	1	\$84,857.66	\$2,357.16	\$33,636.63
Hillebrandt Acres	Jefferson County	51	5	\$739,156.93	\$20,532.14	\$292,993.59
Iola St	Beaumont, TX	6	1	\$41,589.97	\$1,155.28	\$16,485.80
Josey St	Beaumont, TX	11	1	\$69,926.81	\$1,942.41	\$27,718.21
Marsh Rd	Jefferson County	4	1	\$142,388.59	\$3,955.24	\$56,441.25
N 23Rd	Beaumont, TX	7	1	\$100,269.41	\$2,785.26	\$39,745.68
N Kilarney	Jefferson County	8	1	\$66,827.36	\$1,856.32	\$26,489.62
Ogden Ave	Beaumont, TX	7	1	\$64,850.79	\$1,801.41	\$25,706.13
Park St	Beaumont, TX	11	2	\$221,265.97	\$6,146.28	\$87,707.37
Pinkstaff	Beaumont, TX	2	1	\$96,334.04	\$2,675.95	\$38,185.74
River Bend	Bevil Oaks, TX	15	3	\$1,178,926.34	\$32,747.95	\$467,313.30
Roberts St	Beaumont, TX	3	1	\$101,464.40	\$2,818.46	\$40,219.36
Saratoga Cir	Beaumont, TX	4	1	\$49,100.82	\$1,363.91	\$19,463.02
Sour Lake	Jefferson County	5	1	\$78,867.25	\$2,190.76	\$31,262.10
Stratton Ln	Jefferson County	6	1	\$166,605.66	\$4,627.94	\$66,040.63
W Lynwood	Beaumont, TX	4	1	\$49,085.19	\$1,363.48	\$19,456.82
۵		*	*	*	*	***************************************

Westmoreland St	Beaumont, TX	5	1	\$101,539.25	\$2,820.53	\$40,249.03
Total		232	37	\$4,778,984.77	\$132,749.58	\$1,894,336.46

It should be noted that some of the properties on this list may be at far greater flood risk than indicated, because there may be have been periods where the owner(s) did not carry flood insurance, with the result that they may have been damaged but there is no record of it. This type of analysis is not totally conclusive. It would be possible to perform relatively simple engineering studies to better assess risks for properties with just a few claims, but where data suggests that sites may be vulnerable to additional flood-related losses.

The information in this section should be used for planning purposes only, i.e. as the basis for additional steps in risk assessment, and eventually (where warranted) targeted mitigation actions to reduce the risk.

The SRL properties can also be mapped to identify the floodprone areas of DD6. Figure 15 highlights the total number of NFIP severe repetitive loss flood insurance claims per property in Jefferson County DD6. The map shows that the SRL properties in DD6 are mainly concentrated within the City of Beaumont.



Figure 15 – Number of NFIP Flood Insurance Claims Per Severe Repetitive Loss Property in Jefferson County DD6 (Source: FEMA/NFIP, Query January 1, 2016; Plotted by DD6)

### FLOOD RISKS – PUBLIC BUILDINGS

DD6 owns just one complex of buildings, on Walden Road. These buildings are not located in the Special Flood Hazard Area and have never experienced flooding. The other plotted structure is a salt water intrusion.





Public Schools. The Beaumont Independent School District (BISD) owns all of the areas 32 public schools. A review of the FIRM indicates none of these are in the mapped floodplain. As part of the 2016 Plan update, the FIRM maps were again reviewed and verified that none of the 32 schools within the BISD are located within the floodplain.

### **FLOOD RISKS – DISTRICT ASSESTS**

Aside from District facilities, DD6 also owns other assets such as tractors, bulldozers, dump trucks, excavators and many other vehicles totaling to \$18,551,880 in insured value. Those vehicles are mainly stored on District property, far from the floodplain. However, some of these vehicles are often in use and at various project sites that may sit in a floodprone area. DD6 closely monitors the weather and takes proactive steps, when possible, to move vulnerable equipment to higher ground when equipment is being operated or staged in a

#### floodprone area. FLOOD RISKS – ROADS

Nationwide, flooded roads pose the greatest threat to people during floods. Most of the more than 200 people who die in floods each year are lost when they try to drive across flooded roads. Driving into water is the number one weather-related cause of death in Central Texas. Statewide, between 1960 and 1996, 76% of flood-related deaths were vehicle-related.

As illustrated in Figure 17, flood hazards for cars vary with both velocity and depth of floodwaters. Many cars will float in less than 24 inches of water. Fast-moving water can quickly wash cars off the road or wash out a low section of road.

## Figure 17 – Flood Hazard Chart for Cars (Source: Downstream Hazard Classification Guidelines)



Although most roads in the area are unlikely to have deep or fast-moving water during flood conditions up to the level of the 100-year flood, many are still known to flood regularly. Within the City of Beaumont and Jefferson County there are approximately 1,165 miles of roads (750 miles within the City, and 415 within the County).

The Texas Department of Transportation (TXDOT) maintains the freeways that run through the City and County. These major roadways include the following

- Cardinal
- I-10
- East Tex
- College (90) (from I-10 to the west)
- Fannett Rd (from Cardinal to the west)
- Martin Luther King (from I-10 to the south)

Due to the extensive and common road flooding in DD6, it would be nearly impossible to generate a list of flood-prone roads. Members of the planning committee responded to the question of which roads in the area are flood-prone with the answer, "all of them". Due to this reason, the City and County do not close roads due to flooding. However, the City does close major underpasses where water tends to get much deeper. This is accomplished by waiting until the water is deep enough to warrant the closure. There are water depth signs at these major underpasses.

When building new State roads or upgrading existing roads, TxDOT considers the NFIP's floodplain and floodway requirements to evaluate the impact of new and replacement structures. The City and County consider floodplain and floodway impacts in its planning and design for area roads. Within the City of Beaumont, developers must satisfy the City's drainage criteria and other aspects of road designs in order for the City to accept ownership.

Replacing roads and bridges damaged or washed out by floods costs millions of dollars each year. If the damage is caused by a Presidentially-declared disaster, FEMA may pay up to 75% of the repair or replacement costs, with the remaining 25% covered by the State and local governments. The full costs of a damaging event that is not declared a major disaster must be borne by the State and local communities.

TXDOT inspects State bridges for structural integrity and to determine if erosion is a risk. Where erosion has been identified, stabilization measures have been put into place.

Roads and drainage structures in the area have sustained limited erosion damage due to flooding. Damage has occurred to two bridges in area, the bridge on Phelan, and the Bridge on Longhorn Rd. Staff interviews resulted in the following characterizations of past road flooding:

• Most roads in the area are designed to carry water and, therefore, flood even in small events.

• The worst street flooding tends to be on feeder roads.

### FLOOD RISKS – LOCAL DRAINAGE

Many areas and streets experience accumulations of rainfall that are slow to drain away, which may cause disruption of normal traffic, soil erosion, and water quality problems. Local drainage problems contribute to the frequency of flooding, increase ditch maintenance costs, and are perceived to adversely affect the quality of life in some neighborhoods.

Many areas prone to shallow, local drainage flooding are not shown on the City or County's Flood Insurance Rate Maps. One measure of the magnitude of this problem is the number of flood insurance policies in-force on buildings that are outside of the mapped floodplain. Local drainage flooding throughout some subdivisions in DD6 is a problem, even during frequent rainstorms. It is a concern because access for emergency services (fire, emergency medical) can be limited. While the depth of water generally is relatively shallow, a number of homes have been flooded repetitively and are identified by FEMA as repetitive loss properties.

### **SECTION 3 – MITIGATION STRATEGY**

### **DD6'S MITIGATION GOALS**

State and federal guidance and regulations pertaining to mitigation planning require the development of a mitigation goal statement that is consistent with other goals, mission statements and vision statements. To do so, the MPC reviewed FEMA's national mitigation goals, several examples of goal statements from other states and communities, and the State of Texas' Mitigation Goal. The committee also considered information about natural hazards that may occur in the area and their potential consequences and losses.

As part of the Plan update, DD6's mitigation goal statement from the previous HMP was reviewed by the MPC during the initial meeting held on February 26, 2016. The MPC determined that the mitigation goal statement remains current as is with no changes or modifications. The mitigation goal statement remains as follows:

## DD6'S MITIGATION GOAL STATEMENT

The mitigation goals of DD6 are:

- To protect public health, safety, and welfare;
- To reduce losses due to hazards by identifying hazards, minimizing exposure of citizens and property to hazards, and increasing public awareness and involvement;
- To facilitate the development review and approval process to accommodate growth in a practical way that recognizes existing stormwater and floodplain problems while avoiding creating new problems or worsening existing problems; and
- To seek solutions to existing problems.

### STATE OF TEXAS MITIGATION GOALS

The Texas Division of Emergency Management (TDEM) is designated by the Governor as the state's coordinating agency for disaster preparedness, emergency response, and disaster recovery assistance. TDEM also is tasked to coordinate the state's natural disaster mitigation initiatives and administer grant funding provided by FEMA. A key element in that task is the preparation of the State of Texas Hazard Mitigation Plan. The State's 2007 plan includes a series of mitigation goals. As part of the 2013 State of Texas HMP update, the goals from the previous State Plan were re-assessed by the planning committee. TDEM reviewed the goals and added Goal 5 and Goal 6 to their Plan Update.

Goal 1 Reduce or eliminate hazardous conditions that may cause loss of life

Goal 2 Reduce or eliminate hazardous conditions that may inflict injuries

Goal 3 Reduce or eliminate hazardous conditions that can cause property damages

Goal 4 Reduce or eliminate hazardous conditions that degrade important natural resources

Goal 5 Reduce or eliminate repetitive losses due to frequent probability of occurrence

Goal 6 Lessen economic impact within communities when hazards occur

### FEMA'S MITIGATION GOAL

FEMA's mitigation strategy is set forth in a document originally prepared in the late 1990s. This strategy is the basis on which FEMA implements mitigation programs authorized and funded by the U.S. Congress. The national mitigation goal Statement is as follows:

To engender fundamental changes in perception so that the public demands safer environments in which to live and work; and

To reduce, by at least half, the loss of life, injuries, economic costs, and destruction of natural and cultural resources that result from natural disasters.

### **IDENTIFYING PRIORITY ACTIONS**

The 2011 DD6 Plan had 26 mitigation actions. At the time of this Plan Update, many of those actions have been completed and their statuses are listed below. As part of this Plan update, the mitigation actions items from the 2011 Plan were updated to reflect DD6's current priorities for specific activities to achieve the goals discussed above. Each action item identifies an appropriate lead person for each action, cost effectiveness, a schedule for completion and suggested funding sources. For this Plan update, the MPC kept the same priorities and used the (STAPLEE) methodology to prioritize mitigation actions. STAPLEE assesses actions based on six general criteria: Social, Technical, Administrative, Political, Legal, Economic, and Environmental. Table 28 describes the criteria used in the STAPLEE methodology.

#### Table 28 – STAPLEE Methodology Criteria

STAPLEE	Criteria Explanation
S – Social	Mitigation actions are acceptable to the community if they do not adversely affect a particular segment of the population, do not cause relocation of lower income people, and if they are compatible with the community's social and cultural values.
T – Technical	Mitigation actions are technically most effective if they provide long- term reduction of losses and have minimal secondary adverse impacts.
A – Administrative	Mitigation actions are easier to implement if the jurisdiction has the necessary staffing and funding.
P – Political	Mitigation actions can truly be successful if all stakeholders have been offered an opportunity to participate in the planning process and if there is public support for the action.
L – Legal	It is critical that the jurisdiction or implementing agency have the legal authority to implement and enforce a mitigation action.
E – Economic	Budget constraints can significantly deter the implementation of mitigation actions. Hence, it is important to evaluate whether an action is cost- effective, as determined by a cost benefit review, and possible to fund.
E - Environmental	Sustainable mitigation actions that do not have an adverse effect on the environment, that comply with Federal, State, and local environmental regulations, and that are consistent with the community's environmental goals, have mitigation benefits while being environmentally sound.

The Mitigation Planning Committee members developed and prioritized the actions using the STAPLEE criteria. As part of the Plan update, the action tables from the 2011 version were distributed to the MPC and members were requested to update and provide comments. The updates and comments received were integrated into the Action Table for the Plan update. The updated high priority action items in Table 29 were prioritized by the MPC based on the STAPLEE criteria and their potential to reduce risk to DD6, including its operations, and physical

assets. The highest priority actions are generally those that are most effective in reducing risks to multiple assets simultaneously.

The Planning Committee defined High, Medium, and Low priorities in the Action Plan as follows:

- High: Meets five of the seven STAPLEE criteria
- Medium: Meets four of the seven STAPLEE criteria
- Low: Meets three of the seven STAPLEE criteria

These priorities were applied to update the action items. In addition, new actions were identified. The STAPLEE criteria to prioritize also was used but they were not incorporated into the existing list as those projects are completed or ongoing. The new action were prioritized, ranked, with an estimated cost and impact on new buildings and infrastructure (Table 30). The items were sorted by high, medium and low priority. A key criterion in DD6's prioritization of actions was the cost-effectiveness of actions and projects. Cost effectiveness will continue to be central to DD6's decision-making processes in identifying and funding mitigation actions.

### DD6'S CURRENT MITIGATION ACTIONS

Table 29 – Status of Mitigation Actions from the 2011 Plan

No.	Action Item Description / Benefits	Hazard	Status as of 2016
1	Work with National Weather Service to augment and perfect Pine Island Bayou Modeling.	Flood	Complete. DD6 coordinated closely with the NWS to correlate known flood water surfaces in Bevil Oaks to the NWS forecasting station in Sour Lake. The NWS now has a monitoring station on their website that uses DD6 data at Bevil Oaks which is the populated area affected. Additionally, DD6 has installed a site gauge accessible to the residents to aid in their interpretation of the NWS forecast.
2	Hurricane Shutters DD6 desires to harden their administration and engineering building to make a safe harbor for any person that so chooses to stay in these buildings during an event.	Hurricanes and tropical Storms, Thunderstor ms/High Winds	Complete.
3	Tyrell Park Drainage Project The proposed project is to construct two small detention basins (14 acre feet), and increase the size of a road crossing. The net result of this effort will be a lower 100-year water surface in the area, and a significant reduction in flooding.	Flood	Complete.

4	Lawhon Detention	Flood	Complete.
	In order to relieve flooding, a detention basin is proposed to be constructed along Bayou Din.		
5	Hillebrandt Floodwater Diversion Under Calder	Flood	Complete.
6	Replace 10 wooden flood gates at the very bottom of Taylor's Bayou watershed with 4 concrete and steel tainter gates.	Flood	Complete. In addition, four additional tainter gates have been added to an existing seven gate structure. This brings the capacity of the total outfall conveyance structures up to the capacity of the receiving streams (19 gates and a navigation lock).
7	Improvements on Ditch 100-D (Cartwright Corley Area) This mitigation project will remove the existing box culverts and excavate two detention basins on the land that the box culverts cross and adjacent lands. In addition, in order to bring water more efficiently to the new detention basins, an existing man-made ditch will be enlarged, a culvert will be bored under an existing crossing, and a 2,600-foot culvert will be placed along Corley Street.	Flood	Complete.
8	Ditch 104B Improvement Project (Park St. and Saxe Ave. of the City of Beaumont and surrounding subdivisions)	Flood	Complete.

	This mitigation alternative is to bore two 84" pipes under an existing crossing and enlarge 7,200 linear feet of an existing man-made ditch for erosion control.		
9	Ditch 104 Detention Basin – (Highland Park Addition and surrounding subdivisions) This mitigation project is to construct a detention basin project whereby 117 acre- feet of detention will be excavated in a series of basins that are connected by culverts.	Flood	Complete.
10	Upgrade / Repair Floodgates at Taylors Bayou Navigation District Facility.	Flood	Complete. Addressed as part of item 6.
11	Construction of Disaster Shelter at DD6 Facility This shelter will be 3,000 sf in area, house up to 30 people and will be built in accordance with FEMA 361 - Design and Construction Guidance for Community Shelters.	Flood, hurricanes and tropical storms, and, tornado	Complete.
12	Control/Shelter Room at the Flood Control Gates on Taylors Bayou.	Flood	Complete. The Sabine Neches Navigation District has constructed state of the art computer operated controls on all 19 gates and the navigation lock which are housed in a hurricane proof elevated room. Emergency back-up power is included. The SNND is currently working to install a fiber optic line to their office whereby all of the gates can be remotely operated from there also.

13	Implement 800 MHz radio system District- wide.	Hurricanes and Tropical Storms, Tornadoes, Thunderstor ms/High Winds	Complete.
14	Enhance DD6's internal GIS capabilities.	Flood and Hurricanes/ Tropical Storms	Ongoing. The District has been researching the best, most cost effective way to enhancing our GIS capabilities. They have purchased approximately \$20,000.00 worth of scanner and computer equipment to scan maps and begin our GIS database. They are currently researching programs and hardware options.
15	Greenpond Gully Drainage Project Ditch 600 needs to be widened in order to convey the flood flows delivered by the fields and tributaries, and the crossings need to be replaced with longer bridges that are constructed up and out of the flood flows.	Flood	Almost Complete. A \$13.5 million FMA grant was awarded. Construction on this very complicated project has been going well, and the project will be completed this August. This project will help protect approximately 100 homes, thousands of acres of farmland, and miles of roads.
16	Create severe weather action plan, conduct drills, identify and promulgate evacuation and sheltering options.	Floods, Hurricanes and Tropical Storms, Tornadoes, Severe Thunderstor ms/High Winds	Ongoing. A 2,800 sq. ft. building has been constructed at Jefferson County Drainage District No. 6's facility. This building was constructed using all concrete construction that will withstand 150 mph winds. The building will be the team evacuation shelter and used as a command point to work from immediately after hurricanes and storm events. Storm shutters have been installed on all the operation buildings, a diesel generator has been purchased and installed to run all operations during time of power
17	Increase coordination with the City and County regarding flood predictions and post	Flood	outages. This generator will also power the underground fuel tanks. Ongoing. Started to coordinate in 2015. DD6 coordinated with the Jefferson County Emergency Management during a flood in Bevil Oaks last summer in order to plan relief
----	---	--	--
18	Increase flood predictive capability for streams and creeks that affect DD6 (stream gauges, to include adding prior flood levels to current gauges).	Flood	efforts and warning. Ongoing. At least 20 ALERT stations have been added throughout our Jefferson County Drainage District No. 6's district to monitor rainfall and water levels. Jefferson County Drainage District No. 6 has also worked with the National Weather Service to help citizens of the Bevil Oaks community better understand the flood warnings and predictions.
19	Develop distribution centers in local libraries, DD6 facilities, DD6 website and other public buildings where information and safety guidance on natural and manmade hazards as well as ways to mitigate hazards can be provided to citizens	Flood, Hurricanes and Tropical Storms, Tornadoes, and Severe Thunderstor ms/High Winds	To be completed.
20	Ditch No. 901 Re-routing	Flood	Ongoing. Subdivision road flooding. The City of Beaumont study is underway for this and other area flooding to determine how best to mitigate. Once the study is complete, will work with City to determine next steps for a project.
21	Periodically perform engineering and structural surveys on DD6 Facilities (e.g. command and control facilities) to ensure		Ongoing. Structures are surveys after an event comes through if there are issues, they are fixed to continue to protect against the effects of wind and rain.

	that they are sufficiently protected from effects of hazards		
22	Undertake periodic informational mailings to at risk property owners (flood insurance, maintain drainage, flood safety, easy mitigation measures, permit requirements) Include information on other relevant hazards as appropriate	Flood, Hurricanes and Tropical Storms, Tornadoes, and Severe Thunderstor ms/High Winds	Remove. Will be removed from actions in the next plan update. DD6 determined that this work should be Cities and not Drainage District outreach.
23	Conduct homeowner workshops on retrofitting and low cost measures	Flood, Hurricanes and Tropical Storms, Tornadoes, and Severe Thunderstor ms/High Winds	Remove. Will be removed from actions in the next plan update. DD6 determined that this work should be Cities and not Drainage District outreach.
24	Collect sunny day data for at risk buildings	Flood	Remove. Will be removed from actions in the next plan update. Specific project identifications calls for this kind of data collection and there are not enough resources to collect data for every building, rather when a building is to be included in a project.
25	Identify whether hazardous materials handlers/waste sites are in mapped flood plain; notify company and encourage protective measures	Flood	Remove. Will be removed from actions in the next plan update. DD6 determined that this work should be Cities, County or private entities responsibility and not Drainage District outreach.

Winds	26	Formalize procedures on DD6 roles and responsibilities before, during and after a hazard event.	Flood, Hurricanes and Tropical Storms, Tornadoes, and Severe Thunderstor ms/High	Ongoing. Determining SOPs and roles and responsibilities are extremely helpful when an event occurs. DD6 will continue to work on formalizing SOPs and roles and responsibilities to be ready before, during and after a hazard event.
-------	----	---	---	--

#### Table 30 – DD6 NEW 2016 Mitigation Actions

No	Action Item Description / Benefits	Lead Manager	Schedule and Staplee Priority	Hazard	Est. Cost and Rank	Cost Benefit and Effect on Hazard on Existing or New Buildings and Infrastructure
1	Detention project to help mitigate flooding on Delaware Street Implementation lead: DD6 Funding Sources: DD6 operating budget, grants, City support	DD6 Engineering/ Administration	Schedule 2016- 2018 Staplee High/ Medium	Flood and hurricanes and tropical storms	Very Cost Effective	The last major rainfall highlighted an area of Delaware Street that suffers flooding and a detention project is being considered, along with the City of Beaumont, in a City park known as Wuthering Heights Park. High level engineering indicates very cost effective and would help protect structures on Delaware Street and in the Park which includes 50 homes, apartments, schools, a church, and approximately 10 businesses.

No	Action Item Description / Benefits	Lead Manager	Schedule and Staplee Priority	Hazard	Est. Cost and Rank	Cost Benefit and Effect on Hazard on Existing or New Buildings and Infrastructure
2	Ditch 609 (South China Relief) 20,000 linear feet of open channel is planned to be enlarged as well as replacement of six crossings including an inverted siphon for a major Lower Neches Valley Authority (LNVA) canal. Also, a portion of a canal will be relocated to provide space for a much needed drainage ditch. Funding Sources: DD6 operating budget, grants	DD6 Engineering	Schedule 2017 - four years to complete Staple High	Flood and hurricanes and tropical storms	Very Cost Effective	Ditch No. 609 S. China Relief has now been studied in great detail, with project plans nearly completed, and a project cost estimate of \$5,235,015.00 has been finalized. A Flood Mitigation Assistance(FMA) Grant Application has been completed and will be submitted this year. If the grant application is not successful, Jefferson County Drainage District No. 6 will attempt to complete this project over the next four years. Would protect 68 homes and some businesses.

No	Action Item Description / Benefits	Lead Manager	Schedule and Staplee Priority	Hazard	Est. Cost and Rank	Cost Benefit and Effect on Hazard on Existing or New Buildings and Infrastructure
3	Ditch 100 A (East Calwood) 2,200 feet of unmaintainable channel is planned to be retrofitted with an underground culvert to allow for shaping and resizing the ditch to allow for continued maintenance. Funding Sources: DD6 operating budget, grants	DD6 Engineering	Schedule 2016- 2017 Staplee High	Flood and hurricanes and tropical storms	Cost Effective	Right-of-way tracts have been purchased for the channel work, as well as an access and work area. This is an estimated \$300,000.00 project that will begin this year and completed next year. Would protect approximately 40 homes and a Church.
4	Amelia Cutoff Diversion Funding Sources: DD6 operating budget, grants	DD6 Engineering	Schedule 2018 Staplee Medium	Flood and hurricanes and tropical storms	Cost Effective	This \$2.4 million project has been planned and is waiting environmental permitting and funding. Since this project has environmental permitting issues, it has been tabled while other permit applications are processed. This project will be reinstated in 2018. Would protect approximately 408 homes.

No	Action Item Description / Benefits	Lead Manager	Schedule and Staplee Priority	Hazard	Est. Cost and Rank	Cost Benefit and Effect on Hazard on Existing or New Buildings and Infrastructure
5	Taylor's Bayou Funding Sources: DD6 operating budget, grants	DD6 Engineering	Schedule 2017 Staplee High	Flood and hurricanes and tropical storms,	Cost Effective	This \$13 million project has been permitted through the U.S. Army Corps of Engineers, 80% of the necessary right-of-way has been purchased, and a grant application has been submitted and will be resubmitted this year. If we are unsuccessful in obtaining a grant, Jefferson County Drainage District No. 6 will begin this 3-year project next year. Would protect approximately 227 homes and many businesses.
6	Ditch 607 Channel work and crossings Funding Sources: DD6 operating budget, grants	DD6 Engineering	Schedule 2019 Staplee Medium	Flood and hurricanes and tropical storms	Cost Effective	This \$1.5 million estimated project is a rural project for the not too distant future, and it will likely begin in 2019. Would protect approximately 10 homes and many acres of agricultural land the miles of roads.

No	Action Item Description / Benefits	Lead Manager	Schedule and Staplee Priority	Hazard	Est. Cost and Rank	Cost Benefit and Effect on Hazard on Existing or New Buildings and Infrastructure
7	Whites Ranch outfall structures four @ \$250,000 each. One a year for four years. Funding Sources: DD6 operating budget, grants	DD6 Engineering/ Administration	Schedule 2016 Staplee High	Flood and hurricanes and tropical storms	Cost Effective	An engineering firm has been hired to design these structures for \$50,000.00. The design is underway. One structure will be installed this year and three additional structures will be installed in the next three years. Would protect valuable agricultural land and miles of roadways.
8	Ditch 119 Drossings at Yount and Edson Funding Sources: DD6 operating budget, grants, City	DD6 Engineering	Schedule 2017 Staplee Medium/ High	Flood and hurricanes and tropical storms	Cost Effective	This will be a joint project with the City of Beaumont. The City will purchase the box culverts and Jefferson County Drainage District No. 6 will install them, along with the erosion control. Then, the City of Beaumont will reconstruct the street over the box culverts. The City of Beaumont's Engineering Department is currently considering eliminating the Yount Street crossing altogether and leaving an open channel with dead-end cul-de-sacs on each side. The City is performing traffic analysis and taking public input. This project will begin in 2017 and is currently estimated at \$340,000.00. Would protect approximately 50 homes as well as help make a dangerous road that floods significantly become safer.

No	Action Item Description / Benefits	Lead Manager	Schedule and Staplee Priority	Hazard	Est. Cost and Rank	Cost Benefit and Effect on Hazard on Existing or New Buildings and Infrastructure
9	JD Murphree Outfall Funding Sources: DD6 operating budget, grants	DD6 Engineering/ Administration	Schedule 2016- 2017 Staplee High	Flood and hurricanes and tropical storms	Cost Effective	This project has required a tremendous amount of coordination with landowners, Texas Parks & Wildlife Department, US Fish & Wildlife Service, Ducks Unlimited, and Jefferson County. All agreements have been finalized and a \$1 million grant has been obtained by Ducks Unlimited. A consulting engineer has been hired to design the structure for \$80,000.00. The terms of the Ducks Unlimited grant state that the project will be completed by the summer of 2017, so this project is on the fast-tract to be designed, a contractor hired, and construction completed by that time. Would protect thousands of acres of fresh water marshland that is drowning from the floods.
10	Enhance DD6's internal GIS capabilities.	DD6 Engineering	Began in 2015. STAPLEE High	Flood and Hurricanes / Tropical Storms	Cost Effective	Much thought has been given and research has been accomplished to obtain a direction with regard to enhancing our GIS capabilities. We have purchased approximately \$20,000.00 worth of scanner and computer equipment to scan maps and begin our GIS database. We are currently researching programs and hardware options.

No	Action Item Description / Benefits	Lead Manager	Schedule and Staplee Priority	Hazard	Est. Cost and Rank	Cost Benefit and Effect on Hazard on Existing or New Buildings and Infrastructure
11	Greenpond Gully Drainage Project Ditch 600 needs to be widened in order to convey the flood flows delivered by the fields and tributaries, and the crossings need to be replaced with longer bridges that are constructed up and out of the flood flows.	DD6 Engineering	Began in 2015. STAPLEE High	Flood	2.9 BCA	A \$13.5 million FMA grant was awarded. Construction on this very complicated project has been going well, and the project will be completed this August.
12	Create severe weather action plan, conduct drills, identify and promulgate evacuation and sheltering options.	DD6 Administration	2016 STAPLEE High	Floods, Hurricanes and Tropical Storms, Tornadoes, Severe Thundersto rms/High Winds	Cost effective	A 3,000 sq. ft. building has been constructed at Jefferson County Drainage District No. 6's facility. This building was constructed using all concrete construction that will withstand 150 mph winds. The building will be the team evacuation shelter and used as a command point to work from immediately after hurricanes and storm events. Storm shutters have been installed on all the operation buildings, a diesel generator has been purchased and installed to run all operations during time of power outages. This generator will also power the underground fuel tanks.

No	Action Item Description / Benefits	Lead Manager	Schedule and Staplee Priority	Hazard	Est. Cost and Rank	Cost Benefit and Effect on Hazard on Existing or New Buildings and Infrastructure
13	Increase coordination with the City and County regarding flood predictions and post event recovery.	DD6 Administration	Began in 2015. High	Flood	Cost Effective	Ongoing. DD6 coordinated with the Jefferson County Emergency Management during a flood in Bevil Oaks last summer in order to plan relief efforts and warning.
14	Increase flood predictive capability for streams and creeks that affect DD6 (stream gauges, to include adding prior flood levels to current gauges).	DD6 Engineering	Began in 2015. High	Flood	Cost Effective	At least 20 ALERT stations have been added throughout our Jefferson County Drainage District No. 6's district to monitor rainfall and water levels. Jefferson County Drainage District No. 6 has also worked with the National Weather Service to help citizens of the Bevil Oaks community better understand the flood warnings and predictions.

No	Action Item Description / Benefits	Lead Manager	Schedule and Staplee Priority	Hazard	Est. Cost and Rank	Cost Benefit and Effect on Hazard on Existing or New Buildings and Infrastructure
15	Develop distribution centers in local libraries, DD6 facilities, DD6 website and other public buildings where information and safety guidance on natural and manmade hazards as well as ways to mitigate hazards can be provided to citizens	DD6 Administration	Six months from beginning. STAPLEE High	Flood, Hurricanes and Tropical Storms, Tornadoes, and Severe Thundersto rms/High Winds	Cost Effective	To be completed.
16	Procurement of NOAA All Hazard Radios and distribute them to key personnel	DD6 Administration	Six months from beginning. STAPLEE High	Flood, Hurricanes and Tropical Storms, Tornadoes, and Severe Thundersto rms/High Winds	Cost Effective	Acquiring NOAA All Hazard Radios will allow the District to prepare for and respond to hazards as efficiently as possible.

In addition to the actions above, after Hurricane Harvey, additional actions have been added:

* A	A= Actions	reducing risk to existing buildings and	infrastruct	ture		
	* B=	Actions reducing risk to new develop Action #17	oment			
Title: Borley	Heights (	Dutfall Channelization and NLVA	canal cros	ssing	addition	
Hazard		Description/Issue	Implen	entin	ng Department	
Flood Hurricane/Tropical Storms Severe Thunderstorm/High Winds	Project wi Borley Hei Lower Ne into Griffin FMA gran and all of i	l accelerate the floodwater out of the ights subdivision and across the ches Valley Authority BI Canal and Ditch which was improved with an t (all of GD ditches were widened ts crossing enlarged)	JCDD6			
Cost Estimate/Fundir	ıg	Time Frame	Priority	Risk	x Focus (A/B) *	
Cost Estimate: \$6,00 Funding: Grants and operating budget	00,000 general	2020-2025	Н		A	
		Cost and Benefits Considerations				
There are 250 homes future flooding. How	s in the area vever, a full	A that have flooded in the past and this BCA would need to be done to deter Action #18	project co	s and	help mitigate l benefits.	
	<u> </u>	itle: Neches River Diversion Pr	roject			
Hazard Flood Hurricane/Tropical		Description/Issue		olem	enting Department	
Storms Severe Thunderstorm/High Winds	Divert fle and the l River to relieve H	ood flows out of the City of Beaum Hillbrant watershed into the Neches relieve flooding in Beaumont and lillebrant Bayou downstream.	JCDE	D6 ar	nd City of Beaumont	
Cost Estimate/Fund	ling	Time Frame	Prior	rity	Risk Focus (A/B) <sup>3</sup>	
Cost Estimate: 500 Funding: Federal C CDBG/FEMA HM	),000,000 brants (e.g. IGP)	. 2020-2025	H A/B			
		Cost and Benefits Consideration	ns			
This will protect hal	f of the en	ire City of Beaumont from repetitiv	e flooding	<u>.</u>		

#### Table 30-A– Post Harvey – Additional New Action Items

* A= .	Actions red	ucing risk to existing buildings and	d infrastruc	ture
	* B= Ac	tions reducing risk to new develop	pment	
		Action #19		
	Title:	The Bayou Dinn Detention Ba	sin	
Hazard		Description/Issue	Imple	ementing Department
	Detain floo its tributar downstrea	odwater on Hillebrant Bayou and ies to relieve flooding im and allow channelization		
Flood	projects u	pstream. This would Include		
Hurricane/Tropical Storms	Detention	on Bayou Dinn as identified in		
Severe Thunderstorm/High	the Bernar	d Johnson Incorporated Master		
Winds	Drainage I	Plan 1986.	JCDD6 ar	nd Jefferson County
Cost Estimate/Funding		Time Frame	Priority	Risk Focus (A/B) *
Cost Estimate: \$70,000,000 Funding: Federal Grants (e. CDBG/FEMA HMGP)	) g	2020-2025	Н	A/B
		Cost and Benefits Considerations		
Many houses will benefit in the and Cheek.	he south en	d of the Beaumont Texas as well	as the rura	l area Fannett, Labelle,

Action #20						
		Title: Nome Relief				
Hazard		Description/Issue	Implem	enting Department		
Flood	Detain flo	odwaters on Taylors				
Hurricane/Tropical	Bayou trib	outaries 804B and 804D to				
Storms	relieve floo	oding downstream and				
Severe	allow char	nelization projects				
Thunderstorm/High	upstream	to relieve flooding in Nome				
Winds	Texas. Wi	ill include a detention basin.	JCDD6			
Cost Estimate/Fundi	ng	Time Frame	Priority	Risk Focus (A/B) *		
Cost Estimate: \$6,0	00,000					
Funding: Grants and	general					
operating budget		2020-2025	Н	А		
	Co	ost and Benefits Consideration	ons			
Deligues the frequence	wofflood	ng in the City of Norma, DC	A would m	and to be completed		
Relieves the frequence	y of hoodi	ng in the City of Northe, BC.	A would lie	eeu to be completed.		

* A= A	ctions redu	cing risk to existing buildings	s and infras	structure			
	* B= Acti	ons reducing risk to new de	velopment				
Action #21							
		Title: China Relief					
Hazard		Description/Issue	Implem	enting Depar	rtment		
Flood	Detain floo	odwaters on Taylors Bayou					
Hurricane/Tropical	tributaries	600 and 609 to relieve					
Storms	flooding d	ownstream and allow					
Severe	channeliza	tion projects upstream to					
Thunderstorm/High	relieve floo	oding in China Texas. Will					
Winds	include a c	letention basin.	JCDD6				
Cost Estimate/Funding	ng	Time Frame	Priority	Risk Focus	(A/B) *		
Cost Estimate: \$10,0	000,000						
Funding: Grants and	general						
operating budget	C	2020-2025	Н	А			
	Сс	ost and Benefits Consideration	ons				
Relieves the frequence	cy of floodi	ng in the City of China. BCA	A would ne	ed to be con	mpleted.		

Action #22							
	Ti	tle: Study Ditch 505 Detention	on				
Hazard		Description/Issue	Implementing Department				
FIOOU	Dotoin flor	duratara on Taulora Payou					
Humcane/ mopical	Detain not	itak 505 in ander to provide					
Storms	tributary d	itch 505 in order to provide					
Severe	vere flood relief downstream and allow						
Thunderstorm/High	Thunderstorm/High channelization projects to relieve flooding						
Winds	in the Fan	net area	JCDD6				
Cost Estimate/Fundi	ng	Time Frame	Priority	Risk Focus (A/B) *			
Cost Estimate: \$25,	000,000						
Funding: Grants and	general						
operating budget	-						
		2020-2025	H	Α			
Cost and Benefits Considerations							
Relieves the frequence	cy of floodi	ng in Fannet. BCA would need	to be com	pleted.			

* A= Actions reducing risk to existing buildings and infrastructure						
	*	B= Actions reducing risk to new deve	lopment			
Action #23						
	Titl	e: Concrete line ditch assessment a	and repair			
Hazard		Description/Issue	Implementing Department			
Flood						
Hurricane/Tropical	Evaluate a	nd characterize concrete lined ditch				
Storms	damage th	roughout the district to estimate repair	r			
Severe	costs and	pursue opportunities for funding for				
Thunderstorm/High	rehabilitati	on of these channels in order to				
Winds	provide in	proved flood flow conveyance	JCDD6			
Cost Estimate/Fundia	ng	Time Frame	Priority Risk Focus (A/B)			
Cost Estimate: \$100	),000,000					
Funding: Grants, op	erating					
budget	_	2020-2025	Н	A/B		
		Cost and Benefits Consideration	S			
Will provide addition	al capacity	to remove floodwaters out the popula	ted areas c	of the City of Beaumont.		

## **CONTINUED COMPLIANCE WITH THE NFIP**

Participation in the National Flood Insurance Program (NFIP) is important to DD6 and its residents. This is evidenced by the Cities in the planning area, and the County's commitment to regulating development and redevelopment, by adoption of provisions that exceed the minimum requirements, and by its active pursuit of mitigation opportunities. The Cities and Jefferson County, with support from DD6, are firmly committed to continued compliance with the NFIP. It is important to note that DD6 cannot participate in the NFIP as Cities and Counties do. It cannot not apply for NFIP (Cities and County do) or CRS (Cities and County do) status. However, it supports the communities within its planning area in any way it can to keep its standing in the NFIP and CRS.

DD6 is a conservation and reclamation district and a political subdivision of the State of Texas. Considering DD6 is a separate entity and does not directly participate in the NFIP, specific actions will be determined by representatives and officials with the incorporated areas and Jefferson County within DD6. With this in mind, DD6 did not identify and prioritize NFIP actions as part of the planning process. DD6 will continue to work closely with the cities and Jefferson County to identify and recommend actions that will ensure continued compliance with the NFIP.

The City of Beaumont satisfied requirements for initial participation in the NFIP and joined the Emergency Program and ultimately the regular program in 1970. The City of China satisfied requirements for initial participation in the NFIP and joined the Emergency Program and ultimately the regular program in 2008. The City of Bevil Oaks satisfied requirements for initial participation in the Emergency Program and ultimately the regular program in 208. The City of Bevil Oaks satisfied requirements for initial participation in the NFIP and joined the Emergency Program and ultimately the regular program in 1983. The City of Nome satisfied requirements for initial participation in the NFIP and joined the Emergency Program and ultimately the regular program in 1983.

Jefferson County satisfied requirements for initial participation in the NFIP and joined the Emergency Program. Upon issuance and final approval of the Flood Insurance Rate Map in June of 1983, the County joined the Regular Program. The effective Flood Insurance Rate Map for the County has been revised a number of times to reflect more detailed information and changes to the floodplain, and is now used as the minimum flood hazard area within which development must conform to floodplain management regulations.

As mentioned at the beginning of this Section, DD6 is a conservation and reclamation district and a political subdivision of the State of Texas. Considering DD6 is a separate entity and does not directly participate in the NFIP, specific actions will be determined by representatives and officials with the incorporated areas and Jefferson County within DD6. With this in mind, GCCDD did not identify and prioritize NFIP actions as part of the planning process. DD 6will continue to work closely with the Cities and County to identify and recommend actions that will ensure continued compliance with the NFIP.

# CHANGES IN DEVELOPMENT AND REVIEW OF LOCAL REGULATION AND DD6 RESOURCES

DD6 has no direct responsibility for oversight of development in the floodplain. When development is proposed within the Cities or County, within the floodplain, DD6 is asked to review and comment on the subdivision plans. The Cities have strong development and permitting requirements for development in and out of the floodplain. Since 2011, DD6 has reviewed approximately ten proposals.

Since the last plan, The City of China has passed a new permit ordinance, effective April 2016.

Construction permits. Table 31 lists the amount of building permits received by jurisdiction, by type of building (commercial/residential) demolition or construction.

Beaumont Permits						
Туре	2011	2012	2013	2014	2015	2016
Residential Building Permit	298	209	162	358	204	76
Commercial Building Permit	42	36	35	44	35	23
Demolition Residential Permit	39	320	120	342	351	73
Demolition Commerical Permit	282	32	20	36	24	9
China						2016
Residential						1
Commercial						0
Demolition						0
New Permit Ordinance effective	April 2016					
Bevil Oaks						
Туре	2011	2012	2013	2014	2015	2016
Residential Building Permit	3	2	4	3	3	0
Commerical Builidng Permit	0	0	0	0	1	0
Demolition Residential	1	0	0	1	1	0
Demolition Commerical	0	0	0	0	0	0

#### Table 30 Permits

Inspections. DD6 has no inspectors and has no jurisdiction over inspection. However, DD6 relies on the Cities to provide the necessary inspections. To manage development in the floodplain, DD6 relies on the City engineers.

Per the US Census Bureau Quick Facts, the population percent change from 2010 until 2015 have gone up 0.7% in Beaumont and 0.8% in Jefferson County. The Census Bureau quick facts

only reports for cities with a population of 5,000 or more, so China and Bevil Oaks were not reported. Reviewing the population change with the permits, the development trend for the planning areas is only slightly growing and the Cities have strong development and permitting requirements for development in and out of the floodplain.

DD6 is comprised of Engineering, Operations and Administration. Through these departments, the master drainage plan is administered to protect the area. DD6 has GIS capabilities, engineering capabilities, financial and grant capabilities and operational and equipment capabilities to either complete projects completely or largely with its own resources. It has a close working relationship with the Cities and Counties to help get needed resources and projects complete to help protect the residents, infrastructure, businesses and property from future flooding.

These small changes in development along with the Jurisdictions' permitting processes and the many projects that DD6 has engaged in, have led to a decrease in the overall flood vulnerability to the Jurisdiction. The majority of these projects have been drainage projects including detention basins, ditch improvements and floodwater diversions. Many of these projects have already reduced the 100-year flood levels in the project areas.

## APPENDIX 1 – MITIGATION PLANNING COMMITTEE MEETING MINUTES

Jefferson County Drainage District 6

#### **MPC Meeting Minutes**

#### February 25, 2016

Agenda for the February 25, 2016 Mitigation Planning Committee (MPC) meeting

MPC Meeting Number 1

#### Introductions

Karen Stewart Business Manager – DD6

Chuck Oakley CFO – DD6

Doug Canant District Engineer – DD6

Thomas Gill – City of Beaumont Streets and Drainage Manager and Debris Remover

Jeff Ward – JSWA

Dan Ward – JSWA

Don Rao - Engineering Director for Jefferson County

#### Background and purpose of mitigation planning

Updating the HMP has two purposes, it keeps us eligible for FEMA mitigation grant funds and it helps us to understand risk and think through the planning process as well as come up with other mitigation project ideas.

The original HMP written in 2005 and updated in 2010. The 2010 update was approved by FEMA on 6/26/2011. The current plan expires on 6/26/2016.

#### The plan update process

The Plan update will follow the same process as before. Things that have changed within the District must be accounted for like the new admin building which is not in current plan. We need to address how was it built and any risk that comes with the new building and how to mitigate that risk.

We will take the old plan and look at new FEMA requirements. We have to look at any changes in planning area and assets at risk over the past five years as well as any actions to protect those areas.

We have to look at any hazards that have occurred over the past five years.

We will have to look at the status of old action items and add new actions that were not in the plan previously. We also want to involve anyone that interfaces with the district and may have input in what actions should be taken

The following are the Tasked to be followed/completed as part of this planning process:

Task 1: Determine the Planning Area and Resources

Task 2: Build the Planning Team

Task 3: Create an Outreach Strategy

Task 4: Review Community Capabilities - update previous and make sure it hasn't changed

Task 5: Conduct a Risk Assessment – look at any changes to the planning area like new population, new buildings and what has changed

Task 6: Develop a Mitigation Strategy -

Task 7: Keep the Plan Current

Task 8: Review and Adopt the Plan

Communications – communications between the team will occur via email and/or phone

#### Schedule

Task	Start Date	Completion Date
1st Committee Meeting	2/25/2016	2/25/2016
Data Collection	3/1/2016	5/1/2016
Draft plan development	3/1/2016	5/1/2016
Public Presentation	Mar-16	Mar-16
2nd Committee Meeting	May/June 2016	May/June 2016
Rough Draft Complete	6/1/2016	6/1/2016
Draft for Committee to review	6/1/2016	6/1/2016
3rd Committee Meeting	Mid June 2016	Mid June 2016
Committee comments incorporated	6/25/2016	6/25/2016
Submit to the State/FEMA for review	6/30/2016	6/30/2016
Final Public Meeting - associated with		
a board meeting		TBD

#### To Dos:

Verify point of contact for the mitigation plan:

Karen Stewart will be the point of contact for the District.

Verify authority for plan development. The authority for plan development/update is the same.

The status as a drainage district and their authorities have not changed.

#### Verify no change to planning area

ACTION: Karen needs to send an update to the map with a current color version of the planning area showing the City, County and District boundaries. – revised map provided on 2/26/16



#### Verify and update planning committee membership

Mitigation Planning Committee

- Ms. Karen Stewart, Jefferson County Drainage District 6
- Mr. Doug Canant, Jefferson County Drainage District 6
- Ms. Adina Ward, City of Beaumont
- Thomas Gill, City of Beaumont
- Mr. Chuck Oakley, Jefferson County Drainage District 6
- Other Jefferson County Drainage District 6?
- Don Rao Jefferson County Engineering Department

#### Verify Stakeholders Group

- Mr. Richard LeBlanc, Jefferson County Drainage District 6
- Other Members of Jefferson County Drainage District 6
- Mr. Gilbert Ward, Texas Water Development Board
- Sabine Neches Navigation District
- Texas Department of Emergency Management
- BISD, Harden Jefferson ISD (HJISD); Hamshire Fannett ISD (HFISD);
- Lamar University
- Lamar Institute of Technology
- Baptist Memorial Hospital; Christus St. Elizabeth Hospital
- <u>Industry ACTION to Karen and Doug to follow up on whether there is we should</u> <u>include a certain industry as a stakeholder</u>
- Jefferson County Drainage District 3
- TXDOT
- LNVA (Lower Neches Valley Authority

#### Check updates to FIS/FIRMs:

The FIS and FIRMs have the same effective date as the previous Plan.

#### Public involvement

Previously we sent out surveys with the Water Bill, getting about 5,000 responses the first time, but far less last time. The data this time from a survey will not be as useful as previously, so we will not send one out.

#### Set First public meeting

We have to set a public meeting and advertise in the newspaper and on the District Website when and where the public meeting will be held.

#### ACTION – Jeff to propose to Karen some dates for a public meeting.

Review and re-verify mitigation goal is current

### DD6's Mitigation Goal Statement The mitigation goals of DD6 are:

- To protect public health, safety, and welfare;
- To reduce losses due to hazards by identifying hazards, minimizing exposure of citizens and property to hazards, and increasing public awareness and involvement;
- To facilitate the development review and approval process to accommodate growth in a practical way that recognizes existing stormwater and floodplain problems while avoiding creating new problems or worsening existing problems; and
- To seek solutions to existing problems.

The Mitigation Goals are still current and will be kept as is.

We will need an update on the number of buildings. Building permits issued by Jefferson County and the City of Beaumont will help us get an accurate number. <u>ACTION – We need the</u> <u>number of permits issued by the County and the City for new construction in the past 5 years;</u> <u>residential and commercial.</u>

Talk to Boyd Meyer about permits in the City of Beaumont

ACTION – Find the number of permits for demolished buildings

<u>ACTION – Jeff to call City of China and Bevil Oaks for building permits issued for new</u> <u>construction and demolition.</u>

Verify Jefferson Co and Beaumont still submit subdivision proposals to DD6 – how many? Beaumont only submits proposals for subdivisions and commercial buildings.

Change in local ordinances and/or procedures related to building permits/floodplain.

There have been no changes in local ordinances and/or procedures related to building permits/floodplain

Any recent CAVs

<u>ACTION – check with City of Beaumont on recent Community Assisted Visit</u> – JW spoke with AW on 2/26/16 - None in the past 5 years.

Any recent mitigation projects completed since update?

ACTION work with Doug/Chuck/Karen on a list of recent mitigation projects completed since the last update

New projects currently being contemplated?

ACTION work with Doug/Chuck/Karen on a list of new projects being contemplated.

#### Status of prior actions

<u>ACTION – Karen work with Doug to send written response of status of actions from last plan</u>. – DC provided a response on 2/25/16

#### ACTION – Jeff provide to Doug a list of completed Mitigation Projects since last plan

DD6 is completing a new project; there should be an action relating to this in the plan.

ACTION – Jeff send Doug entire repetitive loss list including properties that have been mitigated. – Sent to DC on 2/25/16

Status of prior actions

- Continue to pursue cost effective mitigation projects; apply for federal funding, as appropriate. For flood mitigation projects, focus on areas known to be floodprone/Repetitive Loss areas. For other hazard mitigation projects, coordinate with Jefferson County and incorporated areas within DD6 as they will be required to take the lead on non-flood related projects.
- 2. Formalize procedures on DD6 roles and responsibilities before, during, and after a hazard event
- 3. Work with National Weather Service to augment and perfect Pine Island Bayou Modeling.
- 4. Undertake periodic informational mailings to at-risk property owners (flood insurance, importance of maintaining drainage, flood safety, easy mitigation measures, permit requirements). Include information on other relevant hazards, as appropriate.
- 5. Enhance DD6's internal GIS capabilities
- 6. Hurricane Shutters DD6 desires harden their administration and engineering building to make a safe harbor for any person that so chooses to stay in these buildings during an event.
- 7. Greenpond Gully Drainage Project Ditch 600 needs to be widened in order to convey the flood flows delivered by the fields and tributaries, and the crossings need to be replaced with longer bridges that are constructed up and out of the flood flows.
- 8. Tyrell Park Drainage Project The proposed project is to construct two small detention basins (14 acre feet), and increase the size of a road crossing. The net result of this effort will be a lower 100-year water surface in the area, and a significant reduction in flooding.
- 9. Lawhon Detention In order to relieve flooding, a detention basin is proposed to be constructed along Bayou Din.
- 10. Hillebrandt Floodwater Diversion Under Calder
- 11. Replace 10 wooden flood gates at the very bottom of Taylor 's Bayou watershed with 4 concrete and steel tainter gates.

- 12. Improvements on Ditch 100-D (Cartwright Corley Area) This mitigation project will remove the existing box culverts and excavate two detention basins on the land that the box culverts cross and adjacent lands. In addition, in order to bring water more efficiently to the new detention basins, an existing man-made ditch will be enlarged, a culvert will be bored under an existing crossing, and a 2,600 foot culvert will be placed along Corley Street.
- 13. Ditch 104B Improvement Project (Park St. and Saxe Ave. of the City of Beaumont and surrounding subdivisions) - This mitigation alternative is to bore two 84" pipes under an existing crossing and enlarge 7,200 linear feet of an existing man-made ditch for erosion control
- 14. Ditch 104 Detention Basin (Highland Park Addition and surrounding subdivisions) This mitigation project is to construct a detention basin project whereby 117 acre-feet of detention will be excavated in a series of basins that are connected by culverts
- 15. Upgrade / Repair Floodgates at Taylors Bayou Navigation District Facility
- 16. Construction of Disaster Shelter at DD6 Facility This shelter will be 3,000 s.f. in area, house up to 30 people and will be built in accordance with FEMA 361 Design and Construction Guidance for Community Shelters.
- 17. Ditch No. 901 Re-routing
- 18. Control/Shelter Room at the Flood Control Gates on Taylors Bayou
- 19. Periodically perform engineering and structural surveys of DD-6 facilities (in particular, command and control facilities) to ensure that they are sufficiently protected from effects of hazards, especially wind
- 20. Create severe weather action plan, conduct drills, identify and promulgate evacuation and sheltering options.
- 21. Implement 800 MHz radio system District-wide
- 22. Conduct homeowner workshops on retrofitting & low cost measures
- 23. Increase coordination with the City and County regarding flood predictions and post event recovery
- 24. Increase flood predictive capability for streams and creeks that affect DD6 (stream gages).
- 25. Collect "sunny day" data for at-risk buildings (photographs, elevation information/certificates)
- 26. Identify whether hazardous materials handlers/waste sites are in the mapped floodplain; if flood-prone, notify company and encourage protective measures.

Action Item	Owner
Jeff to revise the submittal date so the plan is submitted to the State before 6/28/16	JM
Karen needs to send an update to the map with a current color version of the planning area showing the City, County and District boundaries.	KS
Follow up on whether there is we should include a certain industry as a stakeholder	MPC
Jeff to propose to Karen some dates for a public meeting	JW
We need the number of permits issued by the County and the City for new construction in the past 5 years; residential and commercial	MPC
Talk to Boyd Meyer about permits in the city of Beaumont	JW
Find the number of permits for demolished buildings	MPC
Jeff to call City of China – 409-752-5403 and Bevil Oaks for building permits issued for new construction and demolition.	JM
check with Adina on City of Beaumont on recent Community Assisted Visit	JW
Jeff to work with Doug/Chuck/Karen on a list of recent mitigation projects completed since the last update	JW, DC, CO, KS
Jeff to work with Doug/Chuck/Karen on a list of new projects being contemplated	JW, DC, CO, KS
Karen work with Doug to send written response of status of actions from last plan	KS, DC
Jeff provide to Doug a list of completed Mitigation Projects since last plan	JW
Jeff send Doug entire repetitive loss list including properties that have been mitigated.	JM

#### Jefferson County Drainage District 6

#### **MPC Meeting Minutes**

#### May 25, 2016

#### JCDD6 Hazard Mitigation Planning Committee (MPC) Meeting 2

#### May 25, 2016

#### AGENDA

- 1. Confirm MPC attendees on Call
- 2. Review remaining data needs
- 3. Review draft mitigation action status from 2011 plan
- 4. Review mitigation action plans
- 5. Revised schedule:

Task	Start Date	<b>Completion Date</b>
1st Committee Meeting	2/25/2016	2/25/2016
Data Collection	3/1/2016	6/1/2016
Draft plan development	3/1/2016	6/15/2016
Public Presentation	4/14/2016	4/14/2016
2nd Committee Meeting	5/25/2016	5/25/2016
Rough Draft Complete	6/6/2016	6/6/2016
Public Notice for Public to review draft	5/27/2016	6/20/2016
Stakeholders requested to Review	5/27/2016	6/20/2016
Committee comments/stakeholder commments incorporated	6/6/2016	6/22/2016
3rd Committee Meeting	6/23/2016	6/23/2016
Public Presentation - Board meeting	6/28/2016	6/28/2016
Submit to State/FEMA for review	6/28/2016	6/28/2016
Final Public Meeting for approval of plan	TBD	TBD

- 6. Review Draft letter to stakeholders
  - a. Who can place draft plan on DD6 website and can it be placed on for 6-6-16?
  - b. MPC must review and provide comments back by 6-20-16 for incorporation by 6-25-16.
- 7. Next call will be June 23rd at 10 am for final review of draft
- Second public meeting will be at JCDD6's Board meeting on draft plan to be submitted to State, 10 am June 28<sup>th</sup>.
- 9. JCDD6 to overnight plan to TDEM on June 28<sup>th</sup> for the review to begin

#### Introductions

Karen Stewart Business Manager – DD6 Chuck Oakley CFO – DD6 Doug Canant District Engineer – DD6 Kristen Thatcher – JSWA Dan Ward – JSWA

#### **Review remaining data needs**

The RL and SRL maps with properties plotted within the district, decide on new mitigation actions and complete risk assessment.

#### Review draft mitigation action status from 2011 plan

Doug to review the list of ongoing actions and give a status summary.

#### **Review mitigation action plans**

The new mitigation actions were reviewed and agreed upon.

#### **Review Draft letter to stakeholders**

The draft letter to stakeholders was reviewed and six new stakeholders added .Letters will be mailed out and the draft plan needs to be put up on the District website by 6/6/16.

The next MPC meeting will be June 23rd at 10 am for final review of draft

## **APPENDIX 2 – PUBLIC MEETINGS AND PRESENTATION**

DD6 held a public meeting on April 14, 2016. Information about this meeting was put in the Beaumont Enterprise, however, no one from the public attended. The publisher's affidavit and the notice in the Beaumont Enterprise are included below.

APR 1 4 2016 Publishers Affidavit

Acct #089393102 Job =199717601 Tear Sheet Attached Name JEFFERSON COUNTY DRAINAGE DI 3910673 B24270750

STATE OF TEXAS COUNTY OF JEFFERSON

BEFORE ME, THE UNDERSIGNED AUTHORITY, ON THIS DAY PERSONALLY APPEARED

WHO BEING BY ME DULY SWORN, DEPOSES AND SAYS THAT HE/SHE IS A NEWSPAPER REPRESENTATIVE FOR THE BEAUMONT ENTERPRISE; THAT SAID NEWSPAPER REGULARLY PUBLISHED IN JEFFERSON COUNTY CIRCULATED IN JEFFERSON, HARDIN, TYLER, NEWTON, ORANGE, JASPER, LIBERTY, SABINE, CHAMBERS, SAN AUGUSTINE, ANGELINA AND GALVESTON COUNTY (COUNTIES), TEXAS; THAT THE ATTACHED NOTICE

IN SAID NEWSPAPER ON THE FOLLOWING DATE(S), TO WIT: 04-07-16

NEWSPAPER REPRESENTATIVE

SWORN AND SUBSCRIBED TO BEFORE ME, THIS 8TH DAY OF APRIL 2016. TO CERTIFY WHICH WITNESS MY HAND AND SEAL OF OFFICE.

NOTARY PIN mu PUBLIC IN AND FOR

NUMBER M T A BRATE OF THE POPULATION

THÈ STATE OF TEXAS

ronica 0 PRINT OR TYPE NAME OF NOTARY MY COMMISSION EXPIRES

#### Legal Notices

matter has been set for a final hearing on, Thursday, May 12, 2016 at 9 o'clock a.m. in Room 330 of the Pulaski County Court House located at 401 West Markham, Little Rock, AR.

In Witness Whereof, I have hereunto set my hand and seal as Clerk of the Court on this 25th day of March, 2016.

Larry Crane, Circuit Clerk

By: Isl Nancy Sadler Deputy Clerk

Prepared by: Jimmy C. Morris, Jr. Attorney for Plaintiff AR Bar No. 2007134 2020 S. Broadway Little Rock, AR 72206 Phone: (501) 319-7647 Fax: (501) 353-0287 Email: j.c.morrisatty@gmail.com

Notice of Plan Availability for Public Review and Comment Jefferson County Drainage District No. Six (DD6) Hazard Mitigation Plan

The public is invited to a meeting where DD6 will present an overview of a planning process recently started by the District to update DD5 current Hazard Mitigation

#### Legal Notices

Plan. This planning process will lead to an updated plan of will lead to an updated plan of action to reduce the long-term impacts of flooding that impact citizens residing in the DD6 planning area and other hazards that impact DD6 owned facilities. Members of the public are en-couraged to attend, especially those with property located in flood-prone areas.

This meeting will be held on April 14, 2016 at 4:00 PM, at DD6 Offices located at:

6550 Walden Road Beaumont, TX 77707-5510

The plan update is required by the Federal Emergency Management Agency (FEMA) for DD5 to remain eligible for certain types of federal grants. The plan update will provide an operation of patients provide an overview of natural hazards in the District, a sum-mary of past hazard events, and describe how the District recognizes and addresses

#### Legal Notices

hazards in the development process along with other Dis-trict functions. The actions i-dentified in this plan are intended to reduce the long-term impact of flooding on the District and its citi-zens. Members of the public will be encouraged to com-ment on the draft plan update as it progresses.

Once the plan update is com-plete, the final hazard mitigation plan will be presented to the DD6 Board for adoption later this year. Any additional public meetings will be scheduled before then to present the draft recommendations and request addi-tional comment from interested citizens.

Questions about the plan should be directed to Mitiga-tion Plan Consultant, Jeff Ward at 888-208-6695 or jward@rstarmail.com. You can also contact Karen J. Stewart at (409) 842-1818.



## **Public Presentation**



April 14, 2016

## Purpose of a Hazard Mitigation Plan

Maintains eligibility for FEMA mitigation grants Keeps understanding of risk current Helps identify new mitigation projects FEMA requires updates every 5 years

## **Mitigation Plan Update**

Risk assessment – Flooding and all hazards that could impact DD6 owned facilities

Establish priorities for actions and projects to reduce risk

## A quick summary of what's in the plan

Executive Summary Background Approval and Adoption Planning Process Hazard Identification Risk Assessment Mitigation Strategy Plan Monitoring and Maintenance Appendices

## Plan Update Process

Form team Update Hazard Assessment Update Risk Assessment Provide status of prior mitigation actions Generate new mitigation actions Prioritize Actions Invite public review and comment This is why we are here tonight

ARD MITIGATION	HAZ/		FEDERAL	
GRANT AMOUNT		SASTER DESCRIPTION	DISASTER #	DATE
<\$10M		re Storms and Flooding	4266	3/19/2016
<\$10M		re Storms and Flooding	4255	2/9/2016
<\$10M		re Storms and Flooding	4245	11/25/2015
<\$10M		re Storms and Flooding	4223	5/29/2015
<\$10M		re Storms and Flooding	4159	12/20/2013
<\$10M		Explosion	4136	8/2/2013
<\$10M		Wildfires	4029	9/9/2011
<\$10M		Wildfires	1999	7/1/2011
6,200,000.00	\$	Hurricane Alex	1931	8/3/2010
400,266,164.00	Ś	Hurricane Ike	1791	9/13/2008
15,535,593.00	s	Hurricane Dolly	1780	7/24/2008
800,367.00	\$	Tropical Storm Erin	1730	10/2/2007
13,119,696.00	\$	re Storms and Flooding	1709	6/29/2007
2,003,052.00	\$	e Storms and Tornadoes	1697	5/1/2007
1,164,959.00	\$	Flooding	1658	8/15/2006
3,731,248.00	\$	reme Wildfire Threat	1624	1/11/2006
103,449,143.00	\$	Hurricane Rita	1606	9/24/2005
2,644,965.00	\$	lurricane Claudette	1479	7/17/2003
7,879,153.00	\$	re Storms and Flooding	1439	11/5/2002
2,080,994.00	\$	Tropical Storm Fay	1434	9/26/2002
11,388,366.00	\$	re Storms and Flooding	1425	7/4/2002
231,827,393.00	\$	Allison	1379	6/9/2001
5,523,264.00	\$	evere Winter Storm	1356	1/8/2001
Approx \$900M		Total		

## Statistics for insured properties and losses

- In the Jefferson County and Cities within
  - 9,000 insured properties
  - >\$91M in paid claims
- 600 RL properties (2 or more claims)
  - \$41M in paid claims
- 85 SRL properties (4 or more claims)
  - \$15M in paid claims

## What is a Mitigation Project?

Reduces risk from natural hazards Other FEMA criteria (later) Projects potentially funded through one of FEMA's grant programs

Project Types (focus on wind and flooding)

 Flood control projects (improved drainage, pumping, etc.)
Primary DD6 projects

- Acquisition/Demolition
- Elevation
- Mitigation-reconstruction
### Project Types (focus on wind and flooding)

Window protection (shutters, filming, etc.)
Improved connections between mechanical equipment/non-structural elements and structure
Roof and structural load path

improvements

## Progress to date

- Planning Grant awarded
- Contract in place
- Initial Structure of revised plan drafted
- Request for Information provided
  - Will provide data for the plan
- Public outreach initiated
  - Today's meeting is the first
  - One more closer to when draft is complete

### Path forward...

- Finalize drafts
- Continued Public outreach (draft to be made available)
- Notify stakeholders
- · Submit draft to state and FEMA
- · Incorporate changes
- Resubmit
- · Adopt in final



Acct #089393102 Job =199969501 Tear Sheet Attached Name JEFFERSON COUNTY DRAINAGE DT 3910673 B24271662

## STATE OF TEXAS COUNTY OF JEFFERSON

bond-BEFORE ME, THE UNDERSIGNED AUTHORITY, ON THIS DAY PERSONALLY AFFFARED Victoria. NHO BEING BY ME DULY SWORN, DEPOSES AND SAYS THAT HE/SHE IS A NEWSPAPER REPRESENTATIVE FOR THE BEADMONT ENTERPRISE; THAT SAID NEWSPAPER REGULARLY FURLISHED IN DEFERRENCE COUNTY CIRCULATED IN JEFFERSON, HARDIN, TYLER, NEWTON, ORANGE, JASPER, LIBERTY, SABINE, CHAMBERS, SAN AUGUSTINE, ANGELINA AND GALVESTON COUNTY(COUNTIES), TEXAS; THAT THE ATTACHED NOTICE IN SAID NEWSPAPER ON THE FOLLOWING DATE(S), TO WIT: 05-27-16

.

NEWSPAPER REPRESENTATIVE

SWORN AND SUBSCRIBED TO BEFORE ME, THIS 30TH DAY OF MAY 2016,

TO CERTIFY WHICH WITNESS MY HAND AND SEAL OF OFFICE

Ma NOTARY PUBLIC IN AND FOR

THE STATE OF TEXAS

ERIKA ACEVEDO FRINT OR TYPE NAME OF NOTARY FUBLIC

MY COMMISSION EXPIRES MAY 16,2020

ERIKA ACEVEDO 128948353 MAY 16, 2020





### Notice of Plan Availability for Public Review and Comment Jefferson County Drainage District No. Six (DD6) Hazard Mitigation Plan

The public is invited to a meeting where DD6 will present an overview of a planning process recently started by the District to update DD6' current Hazard Mitigation Plan. This planning process will lead to an updated plan of action to reduce the long-term impacts of flooding that impact citizens residing in the DD6 planning area and other hazards that impact DD6 owned facilities. Members of the public are encouraged to attend, especially those with property located in flood-prone areas.

This meeting will be held on April 14, 2016 at 4:00 PM, at DD6 Offices located at:

6550 Walden Road Beaumont, TX 77707-5510

The plan update is required by the Federal Emergency Management Agency (FEMA) for DD6 to remain eligible for certain types of federal grants. The plan update will provide an overview of natural hazards in the District, a summary of past hazard events, and describe how the District recognizes and addresses hazards in the development process along with other District functions. The actions identified in this plan are intended to reduce the long-term impact of flooding on the District and its citizens. Members of the public will be encouraged to comment on the draft plan update as it progresses.

Once the plan update is complete, the final hazard mitigation plan will be presented to the DD6 Board for adoption later this year. Any additional public meetings will be scheduled before then to present the draft recommendations and request additional comment from interested citizens.

Questions about the plan should be directed to Mitigation Plan Consultant, Jeff Ward at 888-208-6695 or jward@rstarmail.com. You can also contact Karen J. Stewart at (409) 842-1818.

KETURATO:

karen J. Stewart 6550 Walden RD Beaumont TX 77707 OFFICIAL PUBLIC RECORDED

Carolyn X Marichay

Carolyn L. Guidry, County Clerk Jefferson County, Texas March 31, 2016 02:41:56 PM FEE: \$0.00 COLBERT 20160

2016009951

#### NFL Violating practice rule costs Ravens

By Tribune News Service By filture Ress Service The NFL endered the lowers to forfait there one how the service of the service and finese (cose) (only las-haugh more than \$400,000 combined for violating the long of collective bargain-ter of the service of the service confirmed in the service of the long of collective bargain-ter of the service of the service confirmed in the service of the long of the service of the service tryout participants in faith tryout participants in faith tryout participants in faith service of the service of the service or have 6, which is a clear violation of the CDA. Teams aren't allowed to have full-padded practice until train-meron.

And Burtle of Mierrer 100 Mier

SOUTHEAST

<text><text><text><text><text><text><text><text><text><text>

County wailing tail, clearly spectry watch msQu and neuron-man and the second second second second second second second be submitted to the Office of the Cale Cale A (10, 8), TCR). A second second

CALL (409) 838-2888 to advertise EMAIL Classifieds@BeaumontEnterprise.com Legal Notices Legal Notices Legal Notices Legal Notices An original Applica-tion has been made with the Texas Alco-holis Boverage Com-mission for a Wine and Seer Retailer's in Dy Dolgencorp of texas Inc. divid Dol-lar General Store WT21 located at WT21 locat

NCE NO. 16-028

 Mart Alexandre J. Z. 2015
Mart J. Mart B. ALEXAND L. S. 2015
Mart J. 2015</l на нало страто слока полнати нало на состато на состат

Hazard Milligation Plan Update The public is insited to review County Design District No. Sol (2005) Milligetion, Plan op-date. Dols is required by the dots Dols required by the DOUDAARES OF THE 2006 Har Dols required by the DOUDAARES OF THE 2006 DOUDAARES OF THE 2006 DOUDAARES OF THE 2006

Legal Notices Federat Energieser, Monorgi, tataketing to Swelly of the Construction of the Construction and the Construction of the Construction of the Construction and the Construction of the Construction the Construction of the Construction the Construction of the Construction of the Construction the Construction of the Construction of the Construction the Construction of the Construction the Construction of the Construction of the Construction of the Construction the Construction of the Construction of the Construction of the Construction the Construction of the Con een Read, Baaument, TX 77707-5510. Comments to the plan should be directed to the Millpation Plan Consultant, Jeff Ward at 688-203-6055 method Barton Birom by June 200h. Your can also centact Kareen J. Stewart at (400) 842-1518 or <u>kstewart@ctd0.orp.</u> GREETING: GREETING: You are hereby commanded to approve the series of the courty of the series of the courty of the series of the courts and courty in the City of Besumont, Jefferson Courts, the series of the series of the explicition of the first Monday after the date of publication of this series of the series of the series of the first Monday after the explicition of the first Monday after the date of publication of this series to a petition first in said Court on May 177, 2056, Court of the Series of the series of the court of the series of the series of the court of the series of the series of the court of the series of the series of the court of the series of the series of the court of the series of the series of the court of the series of the series of the servers of the series of the series of the series of the servers of the series of the series of the series of the servers of the series of the series of the series of the servers of the series of the series of the series of the servers of the series of the series of the series of the servers of the series of the series of the series of the servers of the series of the series of the series of the series of the servers of the series of the servers of the series of Hand Barbertoge Voring
Hand Barbertoge V NOTICE OF PUBLIC AUCTION Uncle Bob's Self Storage

**CLASSIFIEDS** 



IN TWE BEAUMONT

Eesler Olition's E1 & up or Lever Nationappen, exy 5 1 beforem apartments, Apply a 1200 French Rd. Ball (Cel) NSZ ABB Equal Neuroling Opportunity CORRENTLY ACCEPTING Equal Neuroling Opportunity Section 5 Elicinety Apartments. Rent Nationa Utilities. Constantiation Statistics Section 5 Elicinety Apartments. Rent Nationa Utilities. Constantiation Statistics Constantiation Statistics Section 5 Elicinety Apartments. Rent National Utilities. Constantiation Statistics Constantiation Statisticon Statistics Constantiation Statistics Constan

FIND YOUR FUTURE

ENTERPRISE CLASSIFIERS.

We're the aren's Job Marketplace

109



### WELCOME TO THE JEFFERSON COUNTY DD6 WEBSITE

It is short for Jefferson County Drainage District Number Six. Established in 1920, DD6 serves Beaumont, Bevil Oaks, China, Nome and the communities of Fannett, Northwest Forest, Hillebrandt Acres, Cheek and LaBelle as well as farm and timber land in between. DD6 controls storm and flood waters from rivers, streams and ditches, and drains and reclaims overflowed lands. Drainage District Six services about 40% of northern Jefferson County, an area almost 1/2 the size of the state of Rhode Island.

Within this area are from 750 to goo linear miles of streams, ditches and outfalls, all under the jurisdiction and control of DD6. Since the inception of the District, Jefferson County has relinquished most of its flood control and drainage activities to DD6 for those properties located within the District boundaries. All incorporated and unincorporated areas rely heavily on DD6 to provide outfall drainage and flood relief.

#### 2016 HAZARD MITIGATION PLAN UPDATE

Jefferson County Drainage District No. Six (DD6) undertook development of this Mitigation Plan to protect public health, safety and welfare. The purpose of this Plan is to reduce or avoid the impacts of hazards by identifying and analyzing hazards and outlining mitigation actions that will help the District reach this goal. This Mitigation Plan also enables DD6 to apply for disaster mitigation funding sources that are otherwise unavailable without an approved Mitigation Plan.

Download the Full PDF - Click Here



### **APPENDIX 3 – CORRESPONDENCE WITH STAKEHOLDERS**

Jefferson County

Drainage District No. 6

6550 Walden R.d., Beaumont, Texas 77707 Telephone (409) 842-1818 Fax (409) 842-2729 Established in 1920

Board of Directors: Joshua W. Allen, Sr. James D. McNicholas William F. Miranda Charles "Chuck" Guillory Miriam K. Johnson Richard P. LeBlanc, Jr. General Manager Jim Broussard Operations Manager Doug S. Canant, Jr., PE, RPLS, CFM Engineer Chuck Oakley, CPA Chief Financial Officer Karen J. Stewart, MBA, CTP Business Manager Andrew J. Jones

Superintendent

May 26, 2016

Mr. Richard LeBlanc General Manager Jefferson County Drainage District 6 6550 Walden Road Beaumont, TX 77707 Re: Jefferson County Drainage District 6, Hazard Mitigation Plan Update

Dear Mr. Richard LeBlanc :

As you may be aware, Jefferson County Drainage District 6 (the District) is currently in the process of updating its FEMA Hazard Mitigation Plan. FEMA requires local jurisdictions to update their plans every five years, and one of the update and re-approval requirements is to have a stakeholder group and the public review and provide input to the plan. Members of a stakeholder group are individuals or organizations that are affected by a mitigation action or policy and can provide specific information on a topic or provide input from a different point of view in the community. These organizations should include business, academia and other private and non-profit interests.

Your organization has been identified by the District as one that could be impacted by the mitigation actions and strategy and therefore, the District would like to add you to its stakeholder group. The District respectfully requests your organization review the draft and provide any comments to the draft. The District will place the draft plan on their website by June 6th at: HTTP://www.dd6.org. Comments may be provided via email to Jeff Ward, the District's Plan Consultant at jward@rstarmail.com or you may reach him by phone at 540-668-6945. Comments will be considered by the Mitigation Planning Committee and incorporated as appropriate. The District asks that you please review and provide your comments back <u>by June 20, 2016</u> in order to give enough time to incorporate the comments into the draft.

Thank you very much for considering this request. It is important that stakeholders and the public have an opportunity to review and comment.

Sincerely,

Haren Stewart

Drainage District No. 6

6550 Walden R.d., Beaumont, Texas 77707 Telephone (409) 842-1818 Fax (409) 842-2729 Established in 1920

**Board of Directors**:

Joshua W. Allen, Sr. James D. McNicholas William F. Miranda Charles "Chuck" Guillory Miriam K. Johnson Richard P. LeBlanc, Jr. General Manager Jim Broussard Operations Manager Doug S. Canant, Jr., PE, RPLS, CFM Engineer Chuck Oakley, CPA Chief Financial Officer Karen J. Stewart, MBA, CTP Business Manager Andrew J. Jones Superintendent

May 26, 2016

Mr. Gilbert Ward Team Leader, State Flood Protection Planning Program Texas Water Development Board 1700 North Congress Avenue Austin, TX 78701 Re: Jefferson County Drainage District 6, Hazard Mitigation Plan Update

Dear Mr. Gilbert Ward :

As you may be aware, Jefferson County Drainage District 6 (the District) is currently in the process of updating its FEMA Hazard Mitigation Plan. FEMA requires local jurisdictions to update their plans every five years, and one of the update and re-approval requirements is to have a stakeholder group and the public review and provide input to the plan. Members of a stakeholder group are individuals or organizations that are affected by a mitigation action or policy and can provide specific information on a topic or provide input from a different point of view in the community. These organizations should include business, academia and other private and non-profit interests.

Your organization has been identified by the District as one that could be impacted by the mitigation actions and strategy and therefore, the District would like to add you to its stakeholder group. The District respectfully requests your organization review the draft and provide any comments to the draft. The District will place the draft plan on their website by June 6th at: HTTP://www.dd6.org. Comments may be provided via email to Jeff Ward, the District's Plan Consultant at jward@rstarmail.com or you may reach him by phone at 540-668-6945. Comments will be considered by the Mitigation Planning Committee and incorporated as appropriate. The District asks that you please review and provide your comments back <u>by June 20, 2016</u> in order to give enough time to incorporate the comments into the draft.

Thank you very much for considering this request. It is important that stakeholders and the public have an opportunity to review and comment.

Haren Stewart

Karen J. Stewart, MBA, CTP Business Manager-JCDD6

Drainage District No. 6 6550 Walden Rd., Beaumont, Texas 77707 Telephone (409) 842-1818 Fax (409) 842-2729 Established in 1920

Board of Directors: Joshua W. Allen, Sr. James D. McNicholas William F. Miranda Charles "Chuck" Guillory Miriam K. Johnson Richard P. LeBlanc, Jr. General Manager Jim Broussard Operations Manager Doug S. Canant, Jr., PE, RPLS, CFM Engineer Chuck Oakley, CPA Chief Financial Officer Karen J. Stewart, MBA, CTP Business Manager Andrew J. Jones Superintendent

May 26, 2016

Mr. Randall Reese General Manager Sabine Neches Navigation District 8180 Anchor Drive Port Arthur, TX 77642 Re: Jefferson County Drainage District 6, Hazard Mitigation Plan Update

Dear Mr. Randall Reese :

As you may be aware, Jefferson County Drainage District 6 (the District) is currently in the process of updating its FEMA Hazard Mitigation Plan. FEMA requires local jurisdictions to update their plans every five years, and one of the update and re-approval requirements is to have a stakeholder group and the public review and provide input to the plan. Members of a stakeholder group are individuals or organizations that are affected by a mitigation action or policy and can provide specific information on a topic or provide input from a different point of view in the community. These organizations should include business, academia and other private and non-profit interests.

Your organization has been identified by the District as one that could be impacted by the mitigation actions and strategy and therefore, the District would like to add you to its stakeholder group. The District respectfully requests your organization review the draft and provide any comments to the draft. The District will place the draft plan on their website by June 6th at: HTTP://www.dd6.org. Comments may be provided via email to Jeff Ward, the District's Plan Consultant at jward@rstarmail.com or you may reach him by phone at 540-668-6945. Comments will be considered by the Mitigation Planning Committee and incorporated as appropriate. The District asks that you please review and provide your comments back <u>by June 20, 2016</u> in order to give enough time to incorporate the comments into the draft.

Thank you very much for considering this request. It is important that stakeholders and the public have an opportunity to review and comment.

Monen Stewart

Karen J. Stewart, MBA, CTP Business Manager-JCDD6

Drainage District No. 6

6550 Walden R.d., Beaumont, Texas 77707 Telephone (409) 842-1818 Fax (409) 842-2729 Established in 1920

Board of Directors: Joshua W. Allen, Sr. James D. McNicholas William F. Miranda Charles "Chuck" Guillory Miriam K. Johnson Richard P. LeBlanc, Jr. General Manager Jim Broussard Operations Manager Doug S. Canant, Jr., PE, RPLS, CFM Engineer Chuck Oakley, CPA Chief Financial Officer Karen J. Stewart, MBA, CTP Business Manager Andrew J. Jones Superintendent

May 26, 2016

Dr. John W. Frossard Superindentent Beaumont Independent School District 3395 Harrison Avenue Beaumont, TX 77706 Re: Jefferson County Drainage District 6, Hazard Mitigation Plan Update

Dear Dr. John W. Frossard :

As you may be aware, Jefferson County Drainage District 6 (the District) is currently in the process of updating its FEMA Hazard Mitigation Plan. FEMA requires local jurisdictions to update their plans every five years, and one of the update and re-approval requirements is to have a stakeholder group and the public review and provide input to the plan. Members of a stakeholder group are individuals or organizations that are affected by a mitigation action or policy and can provide specific information on a topic or provide input from a different point of view in the community. These organizations should include business, academia and other private and non-profit interests.

Your organization has been identified by the District as one that could be impacted by the mitigation actions and strategy and therefore, the District would like to add you to its stakeholder group. The District respectfully requests your organization review the draft and provide any comments to the draft. The District will place the draft plan on their website by June 6th at: HTTP://www.dd6.org. Comments may be provided via email to Jeff Ward, the District's Plan Consultant at jward@rstarmail.com or you may reach him by phone at 540-668-6945. Comments will be considered by the Mitigation Planning Committee and incorporated as appropriate. The District asks that you please review and provide your comments back by June 20, 2016 in order to give enough time to incorporate the comments into the draft.

Thank you very much for considering this request. It is important that stakeholders and the public have an opportunity to review and comment.

Sincerely,

Haren Stewart

Drainage District No. 6

6550 Walden Rd., Beaumont, Texas 77707 Telephone (409) 842-1818 Fax (409) 842-2729 Established in 1920

Board of Directors: Joshua W. Allen, Sr. James D. McNicholas William F. Miranda Charles "Chuck" Guillory Miriam K. Johnson Richard P. LeBlanc, Jr. General Manager Jim Broussard Operations Manager Doug S. Canant, Jr., PE, RPLS, CFM Engineer Chuck Oakley, CPA Chief Financial Officer Karen J. Stewart, MBA, CTP Business Manager Andrew J. Jones Superintendent

May 26, 2016

Ms. Shannon Holmes Superindentent Hardin - Jefferson ISD 520 W. Herring Street Sour Lake, TX 77659 Re: Jefferson County Drainage District 6, Hazard Mitigation Plan Update

Dear Ms. Shannon Holmes :

As you may be aware, Jefferson County Drainage District 6 (the District) is currently in the process of updating its FEMA Hazard Mitigation Plan. FEMA requires local jurisdictions to update their plans every five years, and one of the update and re-approval requirements is to have a stakeholder group and the public review and provide input to the plan. Members of a stakeholder group are individuals or organizations that are affected by a mitigation action or policy and can provide specific information on a topic or provide input from a different point of view in the community. These organizations should include business, academia and other private and non-profit interests.

Your organization has been identified by the District as one that could be impacted by the mitigation actions and strategy and therefore, the District would like to add you to its stakeholder group. The District respectfully requests your organization review the draft and provide any comments to the draft. The District will place the draft plan on their website by June 6th at: HTTP://www.dd6.org. Comments may be provided via email to Jeff Ward, the District's Plan Consultant at jward@rstarmail.com or you may reach him by phone at 540-668-6945. Comments will be considered by the Mitigation Planning Committee and incorporated as appropriate. The District asks that you please review and provide your comments back by June 20, 2016 in order to give enough time to incorporate the comments into the draft.

Thank you very much for considering this request. It is important that stakeholders and the public have an opportunity to review and comment.

Sincerely,

Haren Stewart

Drainage District No. 6 6550 Walden Rd., Beaumont, Texas 77707

Telephone (409) 842-1818 Fax (409) 842-2729 Established in 1920

Board of Directors: Joshua W. Allen, Sr. James D. McNicholas William F. Miranda Charles "Chuck" Guillory Miriam K. Johnson Richard P. LeBlanc, Jr. General Manager Jim Broussard Operations Manager Doug S. Canant, Jr., PE, RPLS, CFM Engineer Chuck Oakley, CPA Chief Financial Officer Karen J. Stewart, MBA, CTP Business Manager Andrew J. Jones Superintendent

May 26, 2016

Ms. Pamela Lechler Superindentent Hamshire Fannett ISD P.O. Box 223 Hamshire, TX 77622 Re: Jefferson County Drainage District 6, Hazard Mitigation Plan Update

Dear Ms. Pamela Lechler :

As you may be aware, Jefferson County Drainage District 6 (the District) is currently in the process of updating its FEMA Hazard Mitigation Plan. FEMA requires local jurisdictions to update their plans every five years, and one of the update and re-approval requirements is to have a stakeholder group and the public review and provide input to the plan. Members of a stakeholder group are individuals or organizations that are affected by a mitigation action or policy and can provide specific information on a topic or provide input from a different point of view in the community. These organizations should include business, academia and other private and non-profit interests.

Your organization has been identified by the District as one that could be impacted by the mitigation actions and strategy and therefore, the District would like to add you to its stakeholder group. The District respectfully requests your organization review the draft and provide any comments to the draft. The District will place the draft plan on their website by June 6th at: HTTP://www.dd6.org. Comments may be provided via email to Jeff Ward, the District's Plan Consultant at jward@rstarmail.com or you may reach him by phone at 540-668-6945. Comments will be considered by the Mitigation Planning Committee and incorporated as appropriate. The District asks that you please review and provide your comments back by June 20, 2016 in order to give enough time to incorporate the comments into the draft.

Thank you very much for considering this request. It is important that stakeholders and the public have an opportunity to review and comment.

Karen J. Stewart, MBA, CTP Business Manager-JCDD6

Drainage District No. 6 6550 Walden Rd., Beaumont, Texas 77707 Telephone (409) 842-1818

> Fax (409) 842-2729 Established in 1920

#### **Board of Directors:**

Joshua W. Allen, Sr. James D. McNicholas William F. Miranda Charles "Chuck" Guillory Miriam K. Johnson Richard P. LeBlanc, Jr. General Manager Jim Broussard Operations Manager Doug S. Canant, Jr., PE, RPLS, CFM Engineer Chuck Oakley, CPA Chief Financial Officer Karen J. Stewart, MBA, CTP Business Manager Andrew J. Jones Superintendent

May 26, 2016

Dr. Kenneth Evans President Lamar University 4400 MLK Boulevard, PO Box 10009 Beaumont, TX 77710 Re: Jefferson County Drainage District 6, Hazard Mitigation Plan Update

Dear Dr. Kenneth Evans :

As you may be aware, Jefferson County Drainage District 6 (the District) is currently in the process of updating its FEMA Hazard Mitigation Plan. FEMA requires local jurisdictions to update their plans every five years, and one of the update and re-approval requirements is to have a stakeholder group and the public review and provide input to the plan. Members of a stakeholder group are individuals or organizations that are affected by a mitigation action or policy and can provide specific information on a topic or provide input from a different point of view in the community. These organizations should include business, academia and other private and non-profit interests.

Your organization has been identified by the District as one that could be impacted by the mitigation actions and strategy and therefore, the District would like to add you to its stakeholder group. The District respectfully requests your organization review the draft and provide any comments to the draft. The District will place the draft plan on their website by June 6th at: HTTP://www.dd6.org. Comments may be provided via email to Jeff Ward, the District's Plan Consultant at jward@rstarmail.com or you may reach him by phone at 540-668-6945. Comments will be considered by the Mitigation Planning Committee and incorporated as appropriate. The District asks that you please review and provide your comments back by June 20, 2016 in order to give enough time to incorporate the comments into the draft.

Thank you very much for considering this request. It is important that stakeholders and the public have an opportunity to review and comment.

Sincerely,

apprend Stewart

Drainage District No. 6 6550 Walden Rd., Beaumont, Texas 77707

> Telephone (409) 842-1818 Fax (409) 842-2729 Established in 1920

Board of Directors: Joshua W. Allen, Sr. James D. McNicholas William F. Miranda Charles "Chuck" Guillory Miriam K. Johnson Richard P. LeBlanc, Jr. General Manager Jim Broussard Operations Manager Doug S. Canant, Jr., PE, RPLS, CFM Engineer Chuck Oakley, CPA Chief Financial Officer Karen J. Stewart, MBA, CTP Business Manager Andrew J. Jones Superintendent

May 26, 2016

Dr. Paul Sszuch President Lamar Institute of Technology 855 East Lavaca Beaumont, TX 77705 Re: Jefferson County Drainage District 6, Hazard Mitigation Plan Update

Dear Dr. Paul Sszuch :

As you may be aware, Jefferson County Drainage District 6 (the District) is currently in the process of updating its FEMA Hazard Mitigation Plan. FEMA requires local jurisdictions to update their plans every five years, and one of the update and re-approval requirements is to have a stakeholder group and the public review and provide input to the plan. Members of a stakeholder group are individuals or organizations that are affected by a mitigation action or policy and can provide specific information on a topic or provide input from a different point of view in the community. These organizations should include business, academia and other private and non-profit interests.

Your organization has been identified by the District as one that could be impacted by the mitigation actions and strategy and therefore, the District would like to add you to its stakeholder group. The District respectfully requests your organization review the draft and provide any comments to the draft. The District will place the draft plan on their website by June 6th at: HTTP://www.dd6.org. Comments may be provided via email to Jeff Ward, the District's Plan Consultant at jward@rstarmail.com or you may reach him by phone at 540-668-6945. Comments will be considered by the Mitigation Planning Committee and incorporated as appropriate. The District asks that you please review and provide your comments back <u>by June 20, 2016</u> in order to give enough time to incorporate the comments into the draft.

Thank you very much for considering this request. It is important that stakeholders and the public have an opportunity to review and comment.

Sincerely,

Haren Stewart

Drainage District No. 6

6550 Walden R.d., Beaumont, Texas 77707 Telephone (409) 842-1818 Fax (409) 842-2729 Established in 1920

Board of Directors: Joshua W. Allen, Sr. James D. McNicholas William F. Miranda Charles "Chuck" Guillory Miriam K. Johnson Richard P. LeBlanc, Jr. General Manager Jim Broussard Operations Manager Doug S. Canant, Jr., PE, RPLS, CFM Engineer Chuck Oakley, CPA Chief Financial Officer Karen J. Stewart, MBA, CTP Business Manager Andrew J. Jones Superintendent

May 26, 2016

Ms. Kim Moncla Executive Director - Foundation Baptist Hospitals of Southeast Texas 3080 College Street Beaumont, TX 77702 Re: Jefferson County Drainage District 6, Hazard Mitigation Plan Update

Dear Ms. Kim Moncla :

As you may be aware, Jefferson County Drainage District 6 (the District) is currently in the process of updating its FEMA Hazard Mitigation Plan. FEMA requires local jurisdictions to update their plans every five years, and one of the update and re-approval requirements is to have a stakeholder group and the public review and provide input to the plan. Members of a stakeholder group are individuals or organizations that are affected by a mitigation action or policy and can provide specific information on a topic or provide input from a different point of view in the community. These organizations should include business, academia and other private and non-profit interests.

Your organization has been identified by the District as one that could be impacted by the mitigation actions and strategy and therefore, the District would like to add you to its stakeholder group. The District respectfully requests your organization review the draft and provide any comments to the draft. The District will place the draft plan on their website by June 6th at: HTTP://www.dd6.org. Comments may be provided via email to Jeff Ward, the District's Plan Consultant at jward@rstarmail.com or you may reach him by phone at 540-668-6945. Comments will be considered by the Mitigation Planning Committee and incorporated as appropriate. The District asks that you please review and provide your comments back by June 20, 2016 in order to give enough time to incorporate the comments into the draft.

Thank you very much for considering this request. It is important that stakeholders and the public have an opportunity to review and comment.

Sincerely,

Haren Stewart

Drainage District No. 6

6550 Walden Rd., Beaumont, Texas 77707 Telephone (409) 842-1818 Fax (409) 842-2729 Established in 1920

Board of Directors:

Joshua W. Allen, Sr. James D. McNicholas William F. Miranda Charles "Chuck" Guillory Miriam K. Johnson Richard P. LeBlanc, Jr. General Manager Jim Broussard Operations Manager Doug S. Canant, Jr., PE, RPLS, CFM Engineer Chuck Oakley, CPA Chief Financial Officer Karen J. Stewart, MBA, CTP Business Manager Andrew J. Jones Superintendent

May 26, 2016

Mr. Paul Trevino CEO Christus St. Elizabeth Hospital 2830 Calder Street Beaumont, TX 77702 Re: Jefferson County Drainage District 6, Hazard Mitigation Plan Update

Dear Mr. Paul Trevino :

As you may be aware, Jefferson County Drainage District 6 (the District) is currently in the process of updating its FEMA Hazard Mitigation Plan. FEMA requires local jurisdictions to update their plans every five years, and one of the update and re-approval requirements is to have a stakeholder group and the public review and provide input to the plan. Members of a stakeholder group are individuals or organizations that are affected by a mitigation action or policy and can provide specific information on a topic or provide input from a different point of view in the community. These organizations should include business, academia and other private and non-profit interests.

Your organization has been identified by the District as one that could be impacted by the mitigation actions and strategy and therefore, the District would like to add you to its stakeholder group. The District respectfully requests your organization review the draft and provide any comments to the draft. The District will place the draft plan on their website by June 6th at: HTTP://www.dd6.org. Comments may be provided via email to Jeff Ward, the District's Plan Consultant at jward@rstarmail.com or you may reach him by phone at 540-668-6945. Comments will be considered by the Mitigation Planning Committee and incorporated as appropriate. The District asks that you please review and provide your comments back by June 20, 2016 in order to give enough time to incorporate the comments into the draft.

Thank you very much for considering this request. It is important that stakeholders and the public have an opportunity to review and comment.

Haren Stewart

Karen J. Stewart, MBA, CTP Business Manager-JCDD6

Drainage District No. 6 6550 Walden Rd., Beaumont, Texas 77707

Telephone (409) 842-1818 Fax (409) 842-2729 Established in 1920

**Board of Directors**:

Joshua W. Allen, Sr. James D. McNicholas William F. Miranda Charles "Chuck" Guillory Miriam K. Johnson Richard P. LeBlanc, Jr. General Manager Jim Broussard Operations Manager Doug S. Canant, Jr., PE, RPLS, CFM Engineer Chuck Oakley, CPA Chief Financial Officer Karen J. Stewart, MBA, CTP Business Manager Andrew J. Jones Superintendent

May 26, 2016

Mr. Phil Kelley Manager JCDD7 4401 9th Avenue Port Aruthur, TX 0 Re: Jefferson County Drainage District 6, Hazard Mitigation Plan Update

Dear Mr. Phil Kelley :

As you may be aware, Jefferson County Drainage District 6 (the District) is currently in the process of updating its FEMA Hazard Mitigation Plan. FEMA requires local jurisdictions to update their plans every five years, and one of the update and re-approval requirements is to have a stakeholder group and the public review and provide input to the plan. Members of a stakeholder group are individuals or organizations that are affected by a mitigation action or policy and can provide specific information on a topic or provide input from a different point of view in the community. These organizations should include business, academia and other private and non-profit interests.

Your organization has been identified by the District as one that could be impacted by the mitigation actions and strategy and therefore, the District would like to add you to its stakeholder group. The District respectfully requests your organization review the draft and provide any comments to the draft. The District will place the draft plan on their website by June 6th at: HTTP://www.dd6.org. Comments may be provided via email to Jeff Ward, the District's Plan Consultant at jward@rstarmail.com or you may reach him by phone at 540-668-6945. Comments will be considered by the Mitigation Planning Committee and incorporated as appropriate. The District asks that you please review and provide your comments back <u>by June 20, 2016</u> in order to give enough time to incorporate the comments into the draft.

Thank you very much for considering this request. It is important that stakeholders and the public have an opportunity to review and comment.

Harm Stewart

Karen J. Stewart, MBA, CTP Business Manager-JCDD6

### Drainage District No. 6

6550 Walden R.d., Beaumont, Texas 77707 Telephone (409) 842-1818 Fax (409) 842-2729 Established in 1920

#### Board of Directors: Joshua W. Allen, Sr. James D. McNicholas William F. Miranda Charles "Chuck" Guillory Miriam K. Johnson

Richard P. LeBlanc, Jr. General Manager Jim Broussard Operations Manager Doug S. Canant, Jr., PE, RPLS, CFM Engineer Chuck Oakley, CPA Chief Financial Officer Karen J. Stewart, MBA, CTP Business Manager Andrew J. Jones Superintendent

May 26, 2016

Mr. Leroy McCall Jr. Manager JCDD3 1149 Pearl Street Beaumont, TX 77701 Re: Jefferson County Drainage District 6, Hazard Mitigation Plan Update

Dear Mr. Leroy McCall Jr. :

As you may be aware, Jefferson County Drainage District 6 (the District) is currently in the process of updating its FEMA Hazard Mitigation Plan. FEMA requires local jurisdictions to update their plans every five years, and one of the update and re-approval requirements is to have a stakeholder group and the public review and provide input to the plan. Members of a stakeholder group are individuals or organizations that are affected by a mitigation action or policy and can provide specific information on a topic or provide input from a different point of view in the community. These organizations should include business, academia and other private and non-profit interests.

Your organization has been identified by the District as one that could be impacted by the mitigation actions and strategy and therefore, the District would like to add you to its stakeholder group. The District respectfully requests your organization review the draft and provide any comments to the draft. The District will place the draft plan on their website by June 6th at: HTTP://www.dd6.org. Comments may be provided via email to Jeff Ward, the District's Plan Consultant at jward@rstarmail.com or you may reach him by phone at 540-668-6945. Comments will be considered by the Mitigation Planning Committee and incorporated as appropriate. The District asks that you please review and provide your comments back by June 20, 2016 in order to give enough time to incorporate the comments into the draft.

Thank you very much for considering this request. It is important that stakeholders and the public have an opportunity to review and comment.

Karen J. Stewart, MBA, CTP Business Manager-JCDD6

Drainage District No. 6 6550 Walden Rd., Beaumont, Texas 77707

Telephone (409) 842-1818 Fax (409) 842-2729 Established in 1920

Board of Directors: Joshua W. Allen, Sr. James D. McNicholas William F. Miranda Charles "Chuck" Guillory Miriam K. Johnson Richard P. LeBlanc, Jr. General Manager Jim Broussard Operations Manager Doug S. Canant, Jr., PE, RPLS, CFM Engineer Chuck Oakley, CPA Chief Financial Officer Karen J. Stewart, MBA, CTP Business Manager Andrew J. Jones Superintendent

May 26, 2016

Mr. Kenneth Wiemers P.E. Area Engineer Beaumont TXDOT 8350 Eastex Freeway Beaumont, TX 77708 Re: Jefferson County Drainage District 6, Hazard Mitigation Plan Update

Dear Mr. Kenneth Wiemers P.E. :

As you may be aware, Jefferson County Drainage District 6 (the District) is currently in the process of updating its FEMA Hazard Mitigation Plan. FEMA requires local jurisdictions to update their plans every five years, and one of the update and re-approval requirements is to have a stakeholder group and the public review and provide input to the plan. Members of a stakeholder group are individuals or organizations that are affected by a mitigation action or policy and can provide specific information on a topic or provide input from a different point of view in the community. These organizations should include business, academia and other private and non-profit interests.

Your organization has been identified by the District as one that could be impacted by the mitigation actions and strategy and therefore, the District would like to add you to its stakeholder group. The District respectfully requests your organization review the draft and provide any comments to the draft. The District will place the draft plan on their website by June 6th at: HTTP://www.dd6.org. Comments may be provided via email to Jeff Ward, the District's Plan Consultant at jward@rstarmail.com or you may reach him by phone at 540-668-6945. Comments will be considered by the Mitigation Planning Committee and incorporated as appropriate. The District asks that you please review and provide your comments back <u>by June 20, 2016</u> in order to give enough time to incorporate the comments into the draft.

Thank you very much for considering this request. It is important that stakeholders and the public have an opportunity to review and comment.

Sincerely,

opprond Stewart

Drainage District No. 6

6550 Walden R.d., Beaumont, Texas 77707 Telephone (409) 842-1818 Fax (409) 842-2729 Established in 1920

Board of Directors: Joshua W. Allen, Sr. James D. McNicholas William F. Miranda Charles "Chuck" Guillory Miriam K. Johnson Richard P. LeBlanc, Jr. General Manager Jim Broussard Operations Manager Doug S. Canant, Jr., PE, RPLS, CFM Engineer Chuck Oakley, CPA Chief Financial Officer Karen J. Stewart, MBA, CTP Business Manager Andrew J. Jones Superintendent

May 26, 2016

Mr. Scott Hall, P.E. General Manager Lower Neches Valley Authority 7850 Eastex Freeway Beaumont, TX 77708 Re: Jefferson County Drainage District 6, Hazard Mitigation Plan Update

Dear Mr. Scott Hall, P.E. :

As you may be aware, Jefferson County Drainage District 6 (the District) is currently in the process of updating its FEMA Hazard Mitigation Plan. FEMA requires local jurisdictions to update their plans every five years, and one of the update and re-approval requirements is to have a stakeholder group and the public review and provide input to the plan. Members of a stakeholder group are individuals or organizations that are affected by a mitigation action or policy and can provide specific information on a topic or provide input from a different point of view in the community. These organizations should include business, academia and other private and non-profit interests.

Your organization has been identified by the District as one that could be impacted by the mitigation actions and strategy and therefore, the District would like to add you to its stakeholder group. The District respectfully requests your organization review the draft and provide any comments to the draft. The District will place the draft plan on their website by June 6th at: HTTP://www.dd6.org. Comments may be provided via email to Jeff Ward, the District's Plan Consultant at jward@rstarmail.com or you may reach him by phone at 540-668-6945. Comments will be considered by the Mitigation Planning Committee and incorporated as appropriate. The District asks that you please review and provide your comments back by June 20, 2016 in order to give enough time to incorporate the comments into the draft.

Thank you very much for considering this request. It is important that stakeholders and the public have an opportunity to review and comment.

Sincerely,

Haron Stewart

Drainage District No. 6

6550 Walden Rd., Beaumont, Texas 77707 Telephone (409) 842-1818 Fax (409) 842-2729 Established in 1920

Board of Directors: Joshua W. Allen, Sr. James D. McNicholas William F. Miranda Charles "Chuck" Guillory Miriam K. Johnson Richard P. LeBlanc, Jr. General Manager Jim Broussard Operations Manager Doug S. Canant, Jr., PE, RPLS, CFM Engineer Chuck Oakley, CPA Chief Financial Officer Karen J. Stewart, MBA, CTP Business Manager Andrew J. Jones Superintendent

May 26, 2016

Mr. Patrick Trahan Government Relations Director Exxon Mobil Oil Corporation 1795 Burt Steet Beaumont, TX 77701 Re: Jefferson County Drainage District 6, Hazard Mitigation Plan Update

Dear Mr. Patrick Trahan :

As you may be aware, Jefferson County Drainage District 6 (the District) is currently in the process of updating its FEMA Hazard Mitigation Plan. FEMA requires local jurisdictions to update their plans every five years, and one of the update and re-approval requirements is to have a stakeholder group and the public review and provide input to the plan. Members of a stakeholder group are individuals or organizations that are affected by a mitigation action or policy and can provide specific information on a topic or provide input from a different point of view in the community. These organizations should include business, academia and other private and non-profit interests.

Your organization has been identified by the District as one that could be impacted by the mitigation actions and strategy and therefore, the District would like to add you to its stakeholder group. The District respectfully requests your organization review the draft and provide any comments to the draft. The District will place the draft plan on their website by June 6th at: HTTP://www.dd6.org. Comments may be provided via email to Jeff Ward, the District's Plan Consultant at jward@rstarmail.com or you may reach him by phone at 540-668-6945. Comments will be considered by the Mitigation Planning Committee and incorporated as appropriate. The District asks that you please review and provide your comments back by June 20, 2016 in order to give enough time to incorporate the comments into the draft.

Thank you very much for considering this request. It is important that stakeholders and the public have an opportunity to review and comment.

Sincerely,

Maring Stewart

### Drainage District No. 6

6550 Walden Rd., Beaumont, Texas 77707 Telephone (409) 842-1818 Fax (409) 842-2729 Established in 1920

Board of Directors: Joshua W. Allen, Sr. James D. McNicholas William F. Miranda Charles "Chuck" Guillory Miriam K. Johnson Richard P. LeBlanc, Jr. General Manager Jim Broussard Operations Manager Doug S. Canant, Jr., PE, RPLS, CFM Engineer Chuck Oakley, CPA Chief Financial Officer Karen J. Stewart, MBA, CTP Business Manager Andrew J. Jones Superintendent

May 26, 2016

Mr. Michael Lockwood Plant Manager Goodyear Tire and Rubber 11357 I-10 Beaumont, TX 77705 Re: Jefferson County Drainage District 6, Hazard Mitigation Plan Update

Dear Mr. Michael Lockwood :

As you may be aware, Jefferson County Drainage District 6 (the District) is currently in the process of updating its FEMA Hazard Mitigation Plan. FEMA requires local jurisdictions to update their plans every five years, and one of the update and re-approval requirements is to have a stakeholder group and the public review and provide input to the plan. Members of a stakeholder group are individuals or organizations that are affected by a mitigation action or policy and can provide specific information on a topic or provide input from a different point of view in the community. These organizations should include business, academia and other private and non-profit interests.

Your organization has been identified by the District as one that could be impacted by the mitigation actions and strategy and therefore, the District would like to add you to its stakeholder group. The District respectfully requests your organization review the draft and provide any comments to the draft. The District will place the draft plan on their website by June 6th at: HTTP://www.dd6.org. Comments may be provided via email to Jeff Ward, the District's Plan Consultant at jward@rstarmail.com or you may reach him by phone at 540-668-6945. Comments will be considered by the Mitigation Planning Committee and incorporated as appropriate. The District asks that you please review and provide your comments back <u>by June 20, 2016</u> in order to give enough time to incorporate the comments into the draft.

Thank you very much for considering this request. It is important that stakeholders and the public have an opportunity to review and comment.

Sincerely,

Haren Stewart

Drainage District No. 6

6550 Walden R.d., Beaumont, Texas 77707 Telephone (409) 842-1818 Fax (409) 842-2729 Established in 1920

**Board of Directors**:

Joshua W. Allen, Sr. James D. McNicholas William F. Miranda Charles "Chuck" Guillory Miriam K. Johnson Richard P. LeBlanc, Jr. General Manager Jim Broussard Operations Manager Doug S. Canant, Jr., PE, RPLS, CFM Engineer Chuck Oakley, CPA Chief Financial Officer Karen J. Stewart, MBA, CTP Business Manager Andrew J. Jones Superintendent

May 26, 2016

Ms. Sue Landry Homeland Security Director Southeast Texas Regional Planning Commission 2210 Eastex Freeway Beaumont, TX 77703 Re: Jefferson County Drainage District 6, Hazard Mitigation Plan Update

Dear Ms. Sue Landry :

As you may be aware, Jefferson County Drainage District 6 (the District) is currently in the process of updating its FEMA Hazard Mitigation Plan. FEMA requires local jurisdictions to update their plans every five years, and one of the update and re-approval requirements is to have a stakeholder group and the public review and provide input to the plan. Members of a stakeholder group are individuals or organizations that are affected by a mitigation action or policy and can provide specific information on a topic or provide input from a different point of view in the community. These organizations should include business, academia and other private and non-profit interests.

Your organization has been identified by the District as one that could be impacted by the mitigation actions and strategy and therefore, the District would like to add you to its stakeholder group. The District respectfully requests your organization review the draft and provide any comments to the draft. The District will place the draft plan on their website by June 6th at: HTTP://www.dd6.org. Comments may be provided via email to Jeff Ward, the District's Plan Consultant at jward@rstarmail.com or you may reach him by phone at 540-668-6945. Comments will be considered by the Mitigation Planning Committee and incorporated as appropriate. The District asks that you please review and provide your comments back by June 20, 2016 in order to give enough time to incorporate the comments into the draft.

Thank you very much for considering this request. It is important that stakeholders and the public have an opportunity to review and comment.

Sincerely,

Hanny Stewart

Drainage District No. 6 6550 Walden Rd., Beaumont, Texas 77707 Welephone (109) 812-1818 Fac(109) 812-2729 8 activity of the 150

Board of Directors: Joshua W. Allen, Sr. James D. McNicholas William F. Miranda Charles "Chuck" Guillory Miriam K. Johnson Richard P. LeBiane, Jr. General Kanager Jim Broussard Operations Nersager Doug S. Canani, Jr., PE, RPLS, CFM Engineer Chuck Oakley, CPA Chief Financial Officer Haren J. Stewart, MBA, CTP Business Managor Andrew J. Jones Superinzedeat

June 1, 2016

Ms. Becky Ford Mayor City of Bevil Oaks 7525 Sweetgum Road Beaumont, TX 77713

Re: Jefferson County Drainage District 6, Hazard Mitigation Plan Update

Dear Ms. Becky Ford :

As you may be aware, Jefferson County Drainage District 6 (the District) is currently in the process of updating its FEMA Hazard Mitigation Plan. FEMA requires local jurisdictions to update their plans every five years, and one of the update and re-approval requirements is to have a stakeholder group and the public review and provide input to the plan. Members of a stakeholder group are individuals or organizations that are affected by a mitigation action or policy and can provide specific information on a topic or provide input from a different point of view in the community. These organizations should include business, academia and other private and non-profit interests.

Your organization has been identified by the District as one that could be impacted by the mitigation actions and strategy and therefore, the District would like to add you to its stakeholder group. The District respectfully requests your organization review the draft and provide any comments to the draft. The District will place the draft plan on their website by June 6th at: HTTP://www.dd6.org/plan. Comments may be provided via email to Jeff Ward, the District's Plan Consultant at jward@estarmail.com or you may reach him by phone at 540-668-6945. Comments will be considered by the Mitigation Planning Committee and incorporated as appropriate. The District asks that you please review and provide your comments back by June 20, 2016 in order to give enough time to incorporate the comments into the draft.

Thank you very much for considering this request. It is important that stakeholders and the public have an opportunity to review and comment.

Sincerely,

April Hains



6550 Walden Rd., (Beaumont, Texas 77707 Telephone (109) 842-1818 Tax (109) 842-2729 Bool Note: In: 500

Board of Directors: Joshua W. Allen, Sr. James D. McNicholas William F. Miranda Charles "Chuck" Guillory Miriam K. Johnson Richard P. LeBlane, Jr. Connel Managet Jim Bronssard Operations Natager Doug S. Canant, Jr., PE, RPLS, CFM Engineer Chuck Oakley, CPA Chief Financial Concer Karen J. Stewart, MBA, CTP Dumnes Manager Andrew J. Jonnes Superimendant

June 1, 2016

Mr. John Walker Mayor City of China P.O. Box 248 China, TX 77613

Re: Jefferson County Drainage District 6, Hazard Mitigation Plan Update

Dear Mr. John Walker :

As you may be aware, Jefferson County Drainage District 6 (the District) is currently in the process of updating its FEMA Hazard Mitigation Plan. FEMA requires local jurisdictions to update their plans every five years, and one of the update and re-approval requirements is to have a stakeholder group and the public review and provide input to the plan. Members of a stakeholder group are individuals or organizations that are affected by a mitigation action or policy and can provide specific information on a topic or provide input from a different point of view in the community. These organizations should include business, academia and other private and non-profit interests.

Your organization has been identified by the District as one that could be impacted by the mitigation actions and strategy and therefore, the District would like to add you to its stakeholder group. The District respectfully requests your organization review the draft and provide any comments to the draft. The District will place the draft plan on their website by June 6th at: IHTTP://www.dd6.org/plan. Comments may be provided via email to Jeff Ward, the District's Plan Consultant at jward@rstarmail.com or you may reach him by phene at 540-668-6945. Comments will be considered by the Mitigation Planning Committee and incorporated as appropriate. The District asks that you please review and provide your comments back by June 20, 2016 in order to give enough time to incorporate the comments into the draft.

Thank you very much for considering this request. It is important that stakeholders and the public have an opportunity to review and comment.

Sincerely,

frank Stant



6550 Walden Rd., (Beaumont, Texas 77707 Telephone (109) 842-1818 Tax (109) 842-2729 Bool Note: In: 500

Board of Directors: Joshua W. Allen, Sr. James D. McNicholas William F. Miranda Charles "Chuck" Guillory Miriam K. Johnson Richard P. LeBlane, Jr. Connel Managet Jim Bronssard Operations Natager Doug S. Canant, Jr., PE, RPLS, CFM Engineer Chuck Oakley, CPA Chief Financial Concer Karen J. Stewart, MBA, CTP Dumnes Manager Andrew J. Jonnes Superimendant

June 1, 2016

Mr. John Walker Mayor City of China P.O. Box 248 China, TX 77613

Re: Jefferson County Drainage District 6, Hazard Mitigation Plan Update

Dear Mr. John Walker :

As you may be aware, Jefferson County Drainage District 6 (the District) is currently in the process of updating its FEMA Hazard Mitigation Plan. FEMA requires local jurisdictions to update their plans every five years, and one of the update and re-approval requirements is to have a stakeholder group and the public review and provide input to the plan. Members of a stakeholder group are individuals or organizations that are affected by a mitigation action or policy and can provide specific information on a topic or provide input from a different point of view in the community. These organizations should include business, academia and other private and non-profit interests.

Your organization has been identified by the District as one that could be impacted by the mitigation actions and strategy and therefore, the District would like to add you to its stakeholder group. The District respectfully requests your organization review the draft and provide any comments to the draft. The District will place the draft plan on their website by June 6th at: IHTTP://www.dd6.org/plan. Comments may be provided via email to Jeff Ward, the District's Plan Consultant at jward@rstarmail.com or you may reach him by phene at 540-668-6945. Comments will be considered by the Mitigation Planning Committee and incorporated as appropriate. The District asks that you please review and provide your comments back by June 20, 2016 in order to give enough time to incorporate the comments into the draft.

Thank you very much for considering this request. It is important that stakeholders and the public have an opportunity to review and comment.

Sincerely,

frank Stant

## **APPENDIX 4 – SOURCES**

Figure 1 - Vicinity Map: State of Texas (Source: www.Mapquest.com)

Figure 2 – DD6 Boundary Map (Source: DD6)

Table 3 – Most Common Industries, Beaumont, Texas (Source: www.City-data.com)

Table 4 - Incorporated Areas of Jefferson County (Source: US Census Bureau, 2014 - Estimates)

Table 5 – Buildings/Infrastructure within Jefferson County Drainage District Six (Sources: Jefferson County Central Appraisal District)

Table 6 – Natural Hazard Events and Declared Major Disasters in Jefferson County (Sources: Public Entity Risk Institute (PERI) website, FEMA, NCDC database)

Table 9 – Tornado Events in Jefferson County with at Least \$50,000 in Property Damage (Source: NCDC Storm Events Database)

Figure 4 – Historical Hurricane Tracks for Jefferson County TX (National Hurricane Center)

Figure 5 – Historical Hurricane Tracks for Jefferson County TX (National Hurricane Center)

Figure 6 – Basic Design Wind Speed (Source: International Building Code)

Table 14 – Beaufort Wind Scale (Source: National Oceanic and Atmospheric Administration)

Table 15 – Severe Thunderstorm and High Wind Events in Jefferson County with at Least \$25,000 in Property Damage (Source: NCDC Storm Events Database)

Figure 7 – Jefferson County DD6 – 100-year Floodplain Map (Source: FEMA National Flood Insurance Program (NFIP) December 2009)

Figure 8 – Jefferson County TX Effective Flood Insurance Rate Map (Source: RiskMap6 Effective FIRM)

Figure 9 – Depth of Precipitation for 50-year Storm for 1-hour duration in Texas (Source: USGS)

Figure 10 – Depth of Precipitation for 100-year Storm for 1-hour duration in Texas (Source: USGS)

Table 17 – Jefferson County Flood Events since Last Planning Effort (Source: NCDC Storm Events Database)

Table 19 - Summary of Residential and Non-Residential NFIP Repetitive Loss Statistics, Jefferson County DD6, ordered by Municipality (Source: FEMA NFIP query January 1, 2016)

Table 20 - Summary of Residential NFIP Repetitive Loss Statistics, Jefferson County DD6, ordered by Municipality (Source: FEMA NFIP query January 1, 2016)

Table 21 - Summary of Non-Residential NFIP Repetitive Loss Statistics, Jefferson County DD6, ordered by Municipality (Source: FEMA NFIP query January 1, 2016)

Table 22 – Summary of Residential NFIP Repetitive Loss Statistics, Jefferson County DD6, ordered by number of Properties on Each Street (Source: FEMA NFIP query January 1, 2016)

Table 23 – Projected 100-year Flood Risk in Jefferson County DD6 Repetitive Loss Areas (Source: FEMA NFIP query January 1, 2016)

Table 24 – Projected Future Damages (Risk) to NFIP Repetitive Flood Loss Properties in Jefferson County DD6, ordered by 100-year Risk (Source: FEMA NFIP query January 1, 2016)

Figure 11 – Number of NFIP Flood Insurance Claims Per Residential Repetitive Loss Property in Jefferson County DD6 (Source: FEMA/NFIP, Query January 1, 2016; Plotted by DD6)

Table 25 – Projected 100-year Flood Risk, Non-Residential Repetitive Loss Properties in Jefferson County DD6 (Source: FEMA NFIP query January 1, 2016)

Figure 12 – Number of NFIP Flood Insurance Claims Per Non-Residential Repetitive Loss Property in Jefferson County DD6 (Source: FEMA/NFIP, Query January 1, 2016; Plotted by DD6)

Table 26 – Projected 100-year Flood Risk, Severe Repetitive Loss Properties in Jefferson County DD6 (Source: FEMA/NFIP, Query January 1, 2016)

Figure 13 – Number of NFIP Flood Insurance Claims Per Severe Repetitive Loss Property in Jefferson County DD6 (Source: FEMA/NFIP, Query January 1, 2016; Plotted by DD6)

Figure 14 – Flood Hazard Chart for Cars (Source: Downstream Hazard Classification Guidelines)

## **APPENDIX 5 – ADOPTION RESOLUTION**

## **APPENDIX 6 – FEMA AND OR TDEM APPROVAL LETTER**



**JEFFERSON COUNTY** 

## Hazard Mitigation Plan Update 2017



## Maintaining a Safe, Secure, and Sustainable Community

August 29, 2017





For more information, visit our website at:

www.co.jefferson.tx.us/em

Written comments should be forwarded to:

H2O Partners, Inc.

P. O. Box 160130

Austin, Texas 78716

info@h2opartnersusa.com

www.h2opartnersusa.com

## TABLE OF CONTENTS

## Section 1 – Introduction

Background	1-1
Scope and Participation	1-2
Purpose	1-3
Authority	1-3
Summary of Sections	1-4

## Section 2 – Planning Process

Plan Preparation and Development	2-1
Review and Incorporation of Existing Plans	2-7
Timeline for Implementing Mitigation Actions	2-10
Public and Stakeholder Involvement	2-10

## Section 3 – County Profile

Overview	3-1
Population and Demographics	3-4
Future Development	3-6
Economic Impact	3-6
Existing and Future Land Use and Development Trentds	3-6

### Section 4 – Risk Overview

Hazard Identification	4-1
Natural Hazards and Climate Change	4-3
Overview of Hazard Analysis	4-4
Hazard Ranking	4-6

## Section 5 – Flood

Hazard Description	5-1
Location	5-2

# TABLE OF CONTENTS

Extent	5-12
Historical Occurrences	5-15
Probability of Future Events	5-19
Vulnerability and Impact	5-19
NFIP Participation	5-23
NFIP Compliance and Maintenance	5-25
Repetitive Loss	5-25

## Section 6 – Lightning

Hazard Description	6-1
Location	6-1
Extent	6-1
Historical Occurrences	6-2
Probability of Future Events	6-4
Vulnerability and Impact	6-4

## Section 7 – Hurricane

Hazard Description	7-1
Location	7-2
Extent	7-4
Historical Occurrences	7-5
Probability of Future Events	7-7
Vulnerability and Impact	7-7

## Section 8 – Extreme Heat

Hazard Description	8-1
Location	8-1
Extent	8-2
Historical Occurrences	8-4

# TABLE OF CONTENTS

Probability of Future Events	8-6
Vulnerability and Impact	8-6

## Section 9 – Hail

Hazard Description	9-1
ocation	9-1
Extent	9-1
Historical Occurrences	9-2
Probability of Future Events	9-7
/ulnerability and Impact	9-7

## Section 10 – Thunderstorm Wind

Hazard Description	10-1
Location	10-1
Extent	10-2
Historical Occurrences	10-3
Probability of Future Events	10-8
Vulnerability and Impact	10-8

## Section 11 – Tornado

Hazard Description	
Location	11-2
Extent	11-2
Historical Occurrences	11-5
Probability of Future Events	11-8
Vulnerability and Impact	11-8

## Section 12 – Drought

Hazard Description	
Location	
# TABLE OF CONTENTS

Extent	
Historical Occurrences	
Probability of Future Events	
Vulnerability and Impact	12-6

# Section 13 – Wildfire

Hazard Description	13-1
Location	13-1
Extent	13-11
Historical Occurrences	13-23
Probability of Future Events	
Vulnerability and Impact	

# Section 14 – Winter Storm

Hazard Description	14-1
Location	14-3
Extent	14-3
Historical Occurrences	14-4
Probability of Future Events	14-6
Vulnerability and Impact	14-6

# Section 15 – Coastal Erosion

Hazard Description	15-1
Location	15-1
Extent	15-2
Historical Occurrences	15-4
Probability of Future Events	15-4
Vulnerability and Impact	15-4

# TABLE OF CONTENTS

# Section 16 – Mitigation Strategy

Mitigation Goals	
Goal 1	
Goal 2	16-1
Goal 3	16-2
Goal 4	
Goal 5	
Goal 6	

# Section 17 – Previous Actions

Summary	17-1
Jefferson County	17-2
Beaumont	17-19
Bevil Oaks	17-47
China	17-60
Groves	17-71
Nederland	17-86
Nome	
Port Arthur	
Port Neches	
Southeast Texas Regional Planning Commission (SETRPC)	

# Section 18 – Mitigation Actions

Summary	
Jefferson County	
Beaumont	
Bevil Oaks	
China	
Groves	

# TABLE OF CONTENTS

Nederland	
Nome	
Port Arthur	
Port Neches	
SETRPC	

# Section 19 – Plan Maintenance

Plan Maintenance Procedures	19-1
Incorporation	19-1
Monitoring and Evaluation	19-5
Updating	19-6
Continued Public Involvement	19-7

Appendix A – Low Risk and Manmade Hazards Appendix B – Planning Team Appendix C – Public Survey Results Appendix D – Critical Facilities Appendix E – Dam Locations Appendix F – Meeting Documentation Appendix G – Capability Assessment

# **SECTION 1: INTRODUCTION**

Background	1
Scope and Participation	2
Purpose	3
Authority	3
Summary of Sections	4

## Background

Jefferson County is located in far southeastern Texas along the Texas-Louisiana border and the Gulf of Mexico. Jefferson County was formed in 1836 and organized in 1837, and was named for the municipality that preceded it, which was in turn named for Thomas Jefferson.

Jefferson County is bounded by Hardin and Orange Counties to the north, the Sabine River and Lake Sabine and Cameron Parish, Louisiana on the east, Chamber County is to the west and Liberty County is to the northwest. The county seat is the City of Beaumont, located 75 miles east of Houston and 17 miles northwest of Port Arthur.

Texas is prone to extremely heavy rains and flooding with half of the world record rainfall rates (48 hours or less).<sup>1</sup> While flooding is a well-known risk, Jefferson County is susceptible to a wide range of natural hazards, including but not limited to extreme heat, tornadoes, hail, and wildfires. These life-threatening hazards can destroy property, disrupt the economy, and lower the overall quality of life for individuals.

While it is impossible to prevent an event from occurring, the effect from many hazards to people and property can be lessened. This concept is known as hazard mitigation, which is defined by the Federal Emergency Management Agency (FEMA) as *sustained actions taken to reduce or eliminate long-term risk to people and property from hazards and their effects.*<sup>2</sup> Communities participate in hazard mitigation by developing hazard mitigation plans. The Texas Division of Emergency Management (TDEM) and FEMA have the authority to review and approve hazard mitigation plans through the Disaster Mitigation Act of 2000.

In 2005-2006, Jefferson County and the participating cities originally developed their Hazard Mitigation Action Plan (HMAP). Then in 2011, information about the planning area and hazard events were updated and incorporated into their HMAP update titled, "Jefferson County Hazard Mitigation Action Plan Version 1.9". This plan was developed by the Jefferson County Hazard Mitigation Team with assistance from Metro Planning, Inc.

<sup>&</sup>lt;sup>1</sup> http://floodsafety.com/texas/regional\_info/regional\_info/dallas\_zone.htm

<sup>&</sup>lt;sup>2</sup> http://www.fema.gov/hazard-mitigation-planning-resources

#### Section 1: Introduction

The Disaster Mitigation Act requires that hazard mitigation plans be reviewed and revised every five years to maintain eligibility for Hazard Mitigation Assistance (HMA) grant funding. Since FEMA originally approved the Jefferson County HMAP in 2006, and then approved an update in 2011, the County began the process of developing a HMAP Update in order to maintain eligibility for grant funding within the five-year window. The South East Texas Regional Planning Commission (SETRPC) coordinated among Orange County, Hardin County, and Jefferson County to update each of their HMAP plans and selected the consultant team of H2O Partners, Inc. to write and develop the HMAP Update 2017 for each of the three counties, including Jefferson County. The HMAP Update planning process provided an opportunity for Jefferson County to evaluate successful mitigation actions and explore opportunities to avoid



future disaster loss. The 2011 HMAP Update will expire in 2016; therefore, the SETRPC and Jefferson County has selected H2O Partners, Inc. to write and develop the 2017 HMAP Update, hereinafter titled: "Jefferson County Hazard Mitigation Plan Update 2017: Maintaining a Safe, Secure and Sustainable Community" (Plan or Plan Update).

Hazard mitigation activities are an investment in a community's safety and sustainability. It is widely accepted that the most effective hazard mitigation measures are implemented at the local government level, where decisions on the regulation and control of development are ultimately made. A comprehensive update to a hazard mitigation plan addresses hazard vulnerabilities that exist today and in the foreseeable future. Therefore, it is essential that a plan identify projected patterns of how future development will increase or decrease a community's overall hazard vulnerability.

### Scope and Participation

Jefferson County's 2017 Plan Update is a multi-jurisdictional Plan. The participating jurisdictions include Jefferson County, the City of Beaumont, the City of Bevil Oaks, the City of China, the City of Groves, the City of Nederland, the City of Nome, the City of Port Arthur, the City of Port Neches, and the South East Texas Regional Planning Commission (SETRPC). These jurisdictions provided valuable input into the planning process.

Throughout the Plan "Jefferson County planning area" refers to the entire planning area including all participating jurisdictions. Similarly, the term "countywide" refers to the entire planning area including all participating jurisdictions.

The focus of the 2017 Plan Update is to identify activities to mitigate hazards classified as "high" or "moderate" risk, as determined through a detailed hazard risk assessment conducted for Jefferson County and the participating jurisdictions. Hazards that pose a "low" or "negligible" risk will continue to be

#### Section 1: Introduction

evaluated during future updates to the Plan, but may be included in the appendices and not be fully addressed until they are determined to be a high or moderate risk. The hazard classification enables the County and participating jurisdictions to prioritize mitigation actions based on hazards which can present the greatest risk to lives and property in the geographic scope (i.e., planning area).

### Purpose

The 2017 Plan Update was prepared by Jefferson County, participating jurisdictions, and H2O Partners, Inc. The purpose of the Plan Update is to protect people and structures, and to minimize the costs of disaster response and recovery. The goal of the Plan Update is to minimize or eliminate long-term risks to human life and property from known hazards by identifying and implementing cost-effective hazard mitigation actions. The planning process is an opportunity for Jefferson County, the participating jurisdictions, stakeholders, and the general public to evaluate and develop successful hazard mitigation actions to reduce future risk of loss of life, and damage to property resulting from a disaster in the Jefferson County planning area.

The Mission Statement of the Plan Update is, "Maintaining a secure and sustainable future through the revision and development of targeted hazard mitigation actions to protect life and property."

Jefferson County, participating jurisdictions, and planning participants identified eleven natural hazards to be addressed by the Plan Update. Additional hazards that have a very low risk or no risk to the planning area are included in Appendix A. The specific goals of the Plan Update are to:

- Provide a comprehensive update to the 2011 HMAP;
- Minimize disruption to Jefferson County and the participating jurisdictions following a disaster;
- Streamline disaster recovery by articulating actions to be taken before a disaster strikes to reduce or eliminate future damage;
- Demonstrate a firm local commitment to hazard mitigation principles;
- Serve as a basis for future funding that may become available through grant and technical assistance programs offered by the State or Federal government. The Plan Update will enable Jefferson County and participating jurisdictions to take advantage of rapidly developing mitigation grant opportunities as they arise; and
- Ensure that Jefferson County and participating jurisdictions maintain eligibility for the full range of future Federal disaster relief.

## Authority



The Plan Update is tailored specifically for Jefferson County, participating jurisdictions, and plan participants including Planning Team members, stakeholders, and the general public who participated in the Plan Update development process. The Plan Update complies with all requirements

promulgated by the Texas Division of Emergency Management (TDEM) and all applicable provisions of the Robert T. Stafford Disaster Relief and Emergency Assistance Act, Section 104 of the Disaster Mitigation Act of 2000 (DMA 2000) (P.L. 106-390), and the Bunning-Bereuter-Blumenauer Flood Insurance Reform Act of 2004 (P.L. 108–264), which amended the National Flood Insurance Act (NFIA) of 1968 (42 U.S.C.

#### Section 1: Introduction

4001, et al). Additionally, the Plan complies with the Interim Final Rules for the Hazard Mitigation Planning and Hazard Mitigation Grant Program (44 CFR, Part 201), which specify the criteria for approval of mitigation plans required in Section 322 of the DMA 2000 and standards found in FEMA's "Local Mitigation Plan Review Guide" (October 2011), and the "Local Mitigation Planning Handbook" (March 2013). Additionally, the Plan is developed in accordance with FEMA's Community Rating System (CRS) Floodplain Management Plan standards and policies.

### Summary of Sections

Sections 1 and 2 of the Plan Update outline the Plan's purpose and development, including how Planning Team members, stakeholders, and members of the general public were involved in the planning process. Section 3 profiles the planning area's population and economy. Sections 4 through 15 present a hazard overview and information on individual natural hazards in the planning area. The hazards generally appear in order of priority based on potential losses to life and property, and other community concerns. For each hazard, the Plan Update presents a description of the hazard, a list of historical hazard events, and the results of the vulnerability and risk assessment process. Section 16 presents hazard mitigation goals and objectives; Section 17 gives an analysis for the previous actions; and Section 18 presents hazard mitigation actions for Jefferson County and the participating jurisdictions. Section 19 identifies Plan maintenance mechanisms.

Several hazards that were included in the previous plans that have very low or no risk to the planning area are included in Appendix A and are updated with any occurrence that have occurred in the past five years. A list of Planning Team members is located in Appendix B. Public survey results are analyzed and presented in Appendix C. Appendix D contains a detailed list of critical facilities for the planning area, and Appendix E provides a list of dam locations. Appendix F contains information regarding workshops, and meeting documentation. The Capability Assessment for Jefferson County and the participating jurisdictions is located in Appendix G.<sup>3</sup>

<sup>&</sup>lt;sup>3</sup> Information contained in some of these appendices are exempt from public release under the Freedom of Information Act (FOIA).

# **SECTION 2: PLANNING PROCESS**

Plan Preparation and Development	1
Overview of the Plan	1
Planning Team	2
Planning Process	4
Kickoff Workshop	5
Hazard Identification	5
Risk Assessment	5
Mitigation Review and Development	6
Review and Incorporation of Existing Plans	7
Review	7
Incorporation of Existing Plans into the HMAP Process	7
Incorporation of the HMAP into Other Planning Mechanisms	8
Plan Review and Plan Update	10
Timeline for Implementing Mitigation Actions	10
Public and Stakeholder Involvement	10
Stakeholder Involvement	11
Public Meetings	12

### **Plan Preparation and Development**

Hazard mitigation planning involves coordination with various constituents and stakeholders to develop a more disaster-resistant community. Section 2 provides an overview of the planning process including the identification of key steps, and a detailed description of how stakeholders and the public were involved.

#### Overview of the Plan

The Southeast Texas Regional Planning Commission (SETRPC) hired H2O Partners, Inc. (Consultant Team), to provide technical support and oversee the development of the Plan Update 2017 for Jefferson County. The Consultant Team used the FEMA "Local Mitigation Plan Review Guide" (October 1, 2011), and the Local Mitigation Planning Handbook" (March 2013) to develop the Plan. The overall planning process is shown in Figure 2-1 below.



Jefferson County, participating jurisdictions, and the Consultant Team met in March 2016 to begin organizing resources, identify Planning Team members, and conduct a Capability Assessment.

#### Planning Team

Key members of H2O Partners, Inc. developed the Plan Update in conjunction with the Planning Team. The Planning Team was established using a direct representation model. Some of the responsibilities of the Planning Team included: completing Capability Assessment surveys, providing input regarding the identification of hazards, identifying mitigation goals, and developing mitigation strategies. An Executive Planning Team consisting of key personnel from each of the participating jurisdictions as well as Jefferson County, shown in Table 2-1, was formed to coordinate planning efforts, and request input and participation in the planning process. Table 2-2 reflects the Advisory Planning Team, consisting of additional representatives from area organizations and departments from the participating jurisdictions and Jefferson County that participated throughout the planning process.

ORGANIZATION/JURISDICTION	TITLE
Jefferson County	Emergency Management Coordinator
City of Beaumont	Emergency Management Coordinator
City of Bevil Oaks	Mayor/ Floodplain Manager
City of China	Mayor
City of Groves	Emergency Management Coordinator

#### Table 2-1. Executive Planning Team

ORGANIZATION/JURISDICTION	TITLE
City of Nederland	Emergency Management Coordinator
City of Nome	Mayor
City of Port Arthur	Senior Planner
City of Port Neches	Emergency Management Coordinator
SETRPC	Homeland Security and Emergency Management Planning Director

#### Table 2-2. Advisory Planning Team

ORGANIZATION/JURISDICTION	TITLE
City of Beaumont	Communications Manager
City of Beaumont	Emergency Management Assistant
City of Beaumont	Emergency Management Specialist
City of Beaumont	Director of Technology Services
City of Beaumont Police Department	Assistant Chief
City of China	City Secretary
City of Nederland Police Department	Assistant Chief
City of Nome	City Secretary
City of Port Arthur	Senior Planner
City of Port Arthur Development Services	Director
City of Port Arthur Fire Department	Emergency Management Coordinator
City of Port Arthur Police Department	Emergency Management Coordinator
City of Port Neches Fire Department	Emergency Management Coordinator
City of Taylor Landing	Mayor
Jefferson County	Assistant Emergency Management Coordinator
SETRPC	Regional Emergency Planner

Additionally, a Stakeholder Group was invited to participate in the planning process via e-mail. The Consultant Team, Planning Team, and Stakeholder Group coordinated to identify mitigation goals, and develop mitigation strategies and actions for the Plan Update. Appendix B, provides a complete listing of all participating Planning Team members and stakeholders by organization and title.

Based on results of completed Capability Assessment, Jefferson County and participating jurisdictions described methods for achieving future hazard mitigation measures by expanding existing capabilities. For example, the City of Bevil Oaks and the City of Groves each have an emergency manager, but no emergency operations plan in place. Other options for improving capabilities include the following:

- Establishing Planning Team members with the authority to monitor the Plan Update and identify grant funding opportunities for expanding staff.
- Identifying opportunities for cross-training or increasing the technical expertise of staff by attending free training available through FEMA and the Texas Division of Emergency Management (TDEM) by monitoring classes and availability through preparetexas.org.
- Reviewing current floodplain ordinances for opportunities to increase resiliency such as modifying permitting or building codes.
- Developing ordinances that will require all new developments to conform to the highest mitigation standards.

Sample hazard mitigation actions developed with similar hazard risk were shared at the meetings. These important discussions resulted in development of multiple mitigation actions that are included in the Plan Update to further mitigate risk from natural hazards in the future.

The Planning Team developed hazard mitigation actions for mitigating risk from potential flooding and hurricanes, including storm-hardening or retrofitting critical facilities, regional communication sites and infrastructure throughout the County to mitigate hazard damage from water and wind, and practicing hazard mitigation techniques. In order to reduce the damage resulting from county-wide flooding that occurs during heavy rain periods, the Plan Update also includes county-wide actions to elevate or upgrade bridges, culverts and other crossings throughout Jefferson County to reduce damages to infrastructure and reduce flooding caused by undersized crossing and culverts.

#### **Planning Process**

The process used to prepare the 2017 Plan Update followed the four major steps included at Figure 2-1. After the Planning Team was organized, a capability assessment was developed and distributed at the Kick-Off Workshop. Hazards were identified and assessed, and results associated with each of the hazards were provided at the Risk Assessment Workshop. Based on Jefferson County's identified vulnerabilities, specific mitigation strategies were discussed and developed at the Mitigation Strategy Workshop. Finally, Plan maintenance and implementation procedures were developed and are included in Section 19. Participation of Planning Team members, stakeholders, and the public at each of the workshops is documented in Appendix F.

At the Plan Update development workshops held throughout the planning process described herein, the following factors were taken into consideration:

- The nature and magnitude of risks currently affecting the community;
- Hazard mitigation goals to address current and expected conditions;
- Whether current resources will be sufficient for implementing the Plan Update;
- Implementation problems, such as technical, political, legal, and coordination issues that may hinder development;

- Anticipated outcomes; and
- How Jefferson County, participating jurisdictions, agencies, and partners will participate in implementing the Plan Update.

#### Kickoff Workshop

The Kickoff Workshop was held at the SETRPC Offices on March 30, 2016. The initial workshop informed County officials and key department personnel about how the planning process pertained to their distinct roles and responsibilities, and engaged stakeholder groups such as Lamar University. In addition to the kickoff presentation, participants received the following information:

- Project overview regarding the planning process;
- Public survey access information;
- Hazard Ranking form; and
- Capability Assessment survey for completion.

A risk ranking exercise was conducted at the Kickoff Workshop to get input from the Planning Team and stakeholders pertaining to various risks from a list of natural hazards affecting the planning area. Participants ranked hazards high to low in terms of perceived level of risk, frequency of occurrence, and potential impact.

#### Hazard Identification

At the Kickoff Workshop, and through e-mail and phone correspondence, the Planning Team conducted preliminary hazard identification. The Planning Team in coordination with the Consultant Team reviewed and considered a full range of natural hazards. Once identified, the teams narrowed the list to significant hazards by reviewing hazards affecting the area as a whole, the 2013 State of Texas Hazard Mitigation Plan Update, and initial study results from reputable sources such as federal and state agencies. Based on this initial analysis, the teams identified a total of eleven natural hazards which pose a significant threat to the planning area.

#### Risk Assessment

An initial risk assessment for Jefferson County and the participating jurisdictions was completed in May 2016 and results were presented to Planning Team members at the Risk Assessment Workshop held on June 1, 2016. At the workshop, the characteristics and consequences of each hazard were evaluated to determine the extent to which the planning area would be affected in terms of potential danger to property and citizens.

Potential dollar losses from each hazard were estimated using the Federal Emergency Management Agency's National Centers for Environmental Information. The resulting risk assessment profiled hazard events, provided information on previous occurrences, estimated probability of future events, and detailed the spatial extent and magnitude of impact on people and property. Each participant at the Risk Assessment Workshop was provided a risk ranking sheet that asked participants to rank hazards in terms of the probability or frequency of occurrence, extent of spatial impact, and the magnitude of impact. The results of the ranking sheets identified unique perspectives on varied risks throughout the planning area.

The assessments were also used to set priorities for hazard mitigation actions based on potential loss of lives and dollar losses. A hazard profile and vulnerability analysis for each of the hazards can be found in Sections 4 through 15.

#### Mitigation Review and Development

Developing the Mitigation Strategy for the Plan Update involved identifying mitigation goals and new mitigation actions. A Mitigation Workshop was held at the SETRPC Offices on August 24, 2016. In addition to the Planning Team, stakeholder groups were invited to attend the workshop. Regarding hazard mitigation actions, Workshop participants emphasized the desire for flood and hurricane projects. Additionally, the County and participating jurisdictions were proactive in identifying mitigation actions to lessen the risk of all the identified hazards included in the Plan Update.

An inclusive and structured process was used to develop and prioritize new hazard mitigation actions for the 2017 Plan Update. The prioritization method was based on FEMA's STAPLE+E criteria and included social, technical, administrative, political, legal, economic and environmental considerations. As a result, each Planning Team Member assigned an overall priority to each hazard mitigation action. The overall priority of each action is reflected in the hazard mitigation actions found in Section 18.

Planning Team Members then developed action plans identifying proposed actions, costs and benefits, the responsible organization(s), effects on new and existing buildings, implementation schedules, priorities, and potential funding sources.

Specifically the process involved:

- Listing optional hazard mitigation actions based on information collected from previous plan reviews, studies, and interviews with federal, state and local officials. Workshop participants reviewed the optional mitigation actions and selected actions that were most applicable to their area of responsibility, cost-effective in reducing risk, easily implemented, and likely to receive institutional and community support.
- Workshop participants inventoried federal and state funding sources that could assist in implementing the proposed hazard mitigation actions. Information was collected, including the program name, authority, purpose of the program, types of assistance and eligible projects, conditions on funding, types of hazards covered, matching requirements, application deadlines, and a point of contact.
- Planning Team Members considered the benefits that would result from implementing the hazard mitigation actions compared to the cost of those projects. Although detailed cost-benefit analyses were beyond the scope of the Plan Update, Planning Team Members utilized economic evaluation as a determining factor between hazard mitigation actions.
- Planning Team Members then selected and prioritized mitigation actions.

Hazard mitigation actions identified in the process were made available to the Planning Team for review. The draft 2017 Plan Update was made available to the general public for review on Jefferson County's website with the chance to comment via responding to Jefferson County's Assistant Emergency Management Coordinator's email at mwhite@co.jefferson.tx.us.

# Review and Incorporation of Existing Plans

#### Review

Background information utilized during the planning process included various studies, plans, reports, and technical information from sources such as FEMA, the United States Army Corps of Engineers (USACE), the U.S. Fire Administration, National Oceanic and Atmospheric Administration (NOAA), the Texas Water Development Board (TWDB), the Texas Commission on Environmental Quality (TCEQ), the Texas State Data Center, Texas Forest Service, the Texas Division of Emergency Management (TDEM), and local hazard assessments and plans. Section 4 and the hazard-specific sections of the Plan (Sections 5-15) summarize the relevant background information.

Specific background documents, including those from FEMA, provided information on hazard risk, hazard mitigation actions currently being implemented, and potential mitigation actions. Previous hazard events, occurrences and descriptions were identified through NOAA's National Center for Environmental Information (NCEI). Results of past hazard events were found through searching the NCEI. The USACE studies were reviewed for their assessment of risk and potential projects in the region. State Data Center documents were used to obtain population projections. The State Demographer webpages were reviewed for population and other projections and included in Section 3 of the Plan Update. Information from the Texas Forest Service was used to appropriately rank the wildfire hazard, and to help identify potential grant opportunities. Materials from FEMA and TDEM were reviewed for guidance on Plan Update development requirements.

#### Incorporation of Existing Plans into the HMAP Process

A Capability Assessment was completed by key Jefferson County and participating jurisdictions' departments which provided information pertaining to existing plans, policies, ordinances and regulations to be integrated into the goals and objectives of the Plan Update. The relevant information was included in a master Capability Assessment, Appendix G.

Existing projects and studies were utilized as a starting point for discussing hazard mitigation actions among Planning and Consultant Team members. For example, the City of Beaumont had a study completed in 2005 that suggested moving several fire stations and the health department in order to improve neighborhood coverage. This was included as an action for the City of Beaumont. Additionally the Continuity of Operations plan from several participating jurisdictions is incorporated into the Plan Update as many critical facilities were identified to install generators with hardwired quick connections to ensure continuity of operations during a hazard event, along with retrofitting and storm-hardening these facilities. Other plans were reviewed, such as Floodplain Management Plans and Storm water Management Plans, to identify any additional mitigation actions. Finally, the 2013 State of Texas Mitigation Plan Update, developed by TDEM, was discussed in the initial planning meeting in order to develop a specific group of hazards to address in the planning effort. The 2013 State Plan Update was also used as a guidance document, along with FEMA materials, in the development of the Jefferson County Plan Update.

#### Incorporation of the HMAP into Other Planning Mechanisms

Planning Team members will integrate implementation of the Plan Update with other planning mechanisms for Jefferson County, such as the Floodplain Management Plan. Existing plans for Jefferson County will be reviewed, and incorporated into the Plan Update, as appropriate. This section discusses how the Plan Update will be implemented by Jefferson County and the participating jurisdictions. It also addresses how the Plan Update will be evaluated and improved over time, and how the public will continue to be involved in the hazard mitigation planning process.

Jefferson County and the participating jurisdictions will be responsible for implementing hazard mitigation actions contained in Section 18. Each hazard mitigation action has been assigned to a specific County and City department that is responsible for tracking and implementing the action.

A funding source has been listed for each identified hazard mitigation action and may be utilized to implement the action. An implementation time period has also been assigned to each hazard mitigation action as an incentive and to determine whether actions are implemented on a timely basis.

Jefferson County and the participating jurisdictions will integrate hazard mitigation actions contained in the Plan Update with existing planning mechanisms such as Stormwater Management Plans and ordinances, Emergency Operations or Management Plans, Evacuation Plans and other local and area planning efforts. Jefferson County will work closely with area organizations to coordinate implementation of hazard mitigation actions that benefit the planning area in terms of financial and economic impact.

Upon formal adoption of the 2017 Plan Update, Planning Team members from Jefferson County and the participating jurisdictions will review existing plans, along with building codes to guide development and ensure that hazard mitigation actions are implemented. Each of the jurisdictions will be responsible for coordinating periodic review of the Plan Update with members of the Advisory Planning Team to ensure integration of hazard mitigation strategies into these planning mechanisms and codes. The Planning Team will also conduct periodic reviews of various existing planning mechanisms and analyze the need for any amendments or updates in light of the approved Plan Update. Jefferson County and the participating jurisdictions will ensure that future long-term planning objectives will contribute to the goals of the Plan Update to reduce the long-term risk to life and property from moderate and high risk hazards. Within one year of formal adoption of the Plan Update.

Planning Team members will review and revise, as necessary, the long-range goals and objectives in its strategic plan and budgets to ensure that they are consistent with the Plan Update.

Further, Jefferson County will work with neighboring jurisdictions to advance the goals of the Plan Update as it applies to ongoing, long-range planning goals and actions for mitigating risk to natural hazards throughout the planning area.

Table 2-3 identifies types of planning mechanisms and examples of methods for incorporating the Plan Update into other planning efforts.

Planning Mechanism	Incorporation of Plan
Grant Applications	The Plan Update will be evaluated by Jefferson County and participating jurisdictions when grant funding is sought for mitigation projects. If a project is not in the Plan Update, an amendment may be necessary to include the action in the Plan Update.
Annual Budget Review	Various departments and key personnel that participated in the planning process for Jefferson County and participating jurisdictions will review the Plan Update and mitigation actions therein when conducting their annual budget review. Allowances will be made in accordance with grant applications sought, and mitigation actions that will be undertaken, according to the implementation schedule of the specific action.
Regulatory Plans	Currently, Jefferson County and participating jurisdictions have regulatory plans in place, such as Emergency Management Plans, Continuity of Operations Plans, Economic Development, and Evacuation Plans. The Plan Update will be consulted when County and City departments review or revise their current regulatory planning mechanisms, or in the development of regulatory plans that are not currently in place.
Capital Improvement Plans	Jefferson County and participating jurisdictions have a Capital Improvement Plan (CIP) in place. Prior to any revisions to the CIP, County and City departments will review the risk assessment and mitigation strategy sections of the HMAP, as limiting public spending in hazardous zones is one of the most effective long-term mitigation actions available to local governments.
Floodplain Management Plans	Floodplain management plans include preventative and corrective actions to address the flood hazard. Therefore, the actions for flooding, and information found in Section 5 of this Plan Update discussing the people and property at risk to flood, will be reviewed and revised when Jefferson County updates their management plans or develops new plans.

#### Table 2-3. Examples of Methods of Incorporation

Appendix G provides an overview of Planning Team members' existing planning and regulatory capabilities to support implementation of mitigation strategy objectives. Appendix G also provides further analysis of

how each intends to incorporate hazard mitigation actions into existing plans, policies, and the annual budget review as it pertains to prioritizing grant applications for funding and implementation of identified hazard mitigation projects.

It should be noted for the purposes of the plan update that the HMAP has been used as a reference when reviewing and updating all plans and ordinances for the entire planning area, including all participating jurisdictions. The Emergency Management Plan developed independently by all participating jurisdictions is updated every 5 years and incorporates goals, objectives and actions identified in the mitigation plan.

#### Plan Review and Plan Update

As with the development of Plan Update, Jefferson County will oversee the review and update process for relevance and to necessary make adjustments. At the beginning of each fiscal year, Planning Team Members will meet to evaluate the Plan Update and review other planning mechanisms to ensure consistency with long-range planning efforts. In addition, planning participants will also meet twice a year, by conference call or presentation, to re-evaluate prioritization of the hazard mitigation actions.

# **Timeline for Implementing Mitigation Actions**

Both the Executive Planning Team (Table B-1, Appendix B), and the Advisory Planning Team (Table B-2, Appendix B), will engage in discussions regarding a timeframe for how and when to implement each hazard mitigation action. Considerations include when the action will be started, how existing planning mechanisms' timelines affect implementation, and when the action should be fully implemented. Timeframes may be general, and there will be short, medium, and long term goals for implementation based on prioritization of each action, as identified on individual Hazard Mitigation Action worksheets included in the Plan Update for Jefferson County and participating jurisdictions.

Both the Executive and Advisory Planning Team will evaluate and prioritize the most suitable hazard mitigation actions for the community to implement. The timeline for implementation of actions will partially be directed by Jefferson County's comprehensive planning process, budgetary constraints, and community needs. Jefferson County and the participating jurisdictions are committed to addressing and implementing hazard mitigation actions that may be aligned with and integrated into the Plan Update.

Overall, the Planning Team is in agreement that goals and actions of the Plan Update shall be aligned with the timeframe for implementation of hazard mitigation actions with respect to annual review and updates of existing plans and policies.

# Public and Stakeholder Involvement

An important component of hazard mitigation planning is public participation and stakeholder involvement. Input from individual citizens and the community as a whole provides the Planning Team with a greater understanding of local concerns, and increases the likelihood of successfully implemented hazard mitigation actions. If citizens and stakeholders, such as local businesses, non-profits, hospitals, and schools are involved, they are more likely to gain a greater appreciation of the risks that hazards may present in their community and take steps to reduce or mitigate their impact.

The public was involved in the development of Jefferson County's 2017 Plan Update at different stages prior to official Plan Update approval and adoption. Public input was sought using three methods: (1) open public meetings; (2) survey instruments; and (3) making the draft Plan Update available for public review at Jefferson County's website.

The draft 2017 Plan Update was made available to the general public for review and comment on the Jefferson County's website. The public was notified at the public meetings that the draft Plan Update would be available for review. No feedback was received on the draft 2017 Plan Update, although it was given on the public survey, and all relevant information was incorporated into the Plan Update.

The 2017 Plan Update will be advertised and a copy available at the SETRPC office and the County's Engineering office upon approval from FEMA.

#### Stakeholder Involvement

Stakeholder involvement is essential to hazard mitigation planning since a wide range of stakeholders can provide input on specific topics and input from various points of view. Throughout the planning process, members of community groups, local businesses, neighboring jurisdictions, schools, and hospitals were invited to participate in development of the 2017 Plan Update. The Stakeholder Group (Table B-3 in Appendix B, and Table 2-4, below), included a broad range of representatives from both the public and private sector, and served as a key component in Jefferson County's outreach efforts for development of the Plan Update. Documentation of stakeholder meetings is found in Appendix F. A list of organizations invited to attend via e-mail is found in Table 2-4.

AGENCY	TITLE	PARTICIPATED
Colonial Pipeline	Manager	Х
Lamar University	Assistant Professor	Х
Local Emergency Planning Committee	Chairperson	Х
Jefferson County Drainage District 7	Graduate Engineer	Х
Jefferson County Drainage District 7	Supervisor	Х
RPS	Senior Consulting Engineer	Х
South East Texas Disaster Recovery Group	Executive Director	Х
Texas House of Representatives	Texas US Representative	Х
Texas State Senate	Texas State Senator	
United Way	Executive Director	Х
City of Kountze	Emergency Management Coordinator	Х
City of Lumberton	City Manager	Х

#### Table 2-4. Stakeholder Working Group

Jefferson County | Hazard Mitigation Plan Update 2017 | Page 11

AGENCY	TITLE	PARTICIPATED
City of Rose Hill Acres	Mayor	Х
City of Silsbee	Emergency Management Coordinator	Х
City of Silsbee	Assistant Emergency Management Coordinator	х
City of Sour Lake	City Manager	Х
City of Sour Lake	Police Chief	Х
Hardin County	Emergency Management Coordinator	Х
Hardin County	Floodplain Administrator	Х
South East Texas Regional Planning Commission	Homeland Security and Emergency Management Planning Director	х
City of Bridge City	Emergency Management Coordinator	Х
City of Orange	Deputy Chief/Emergency Management Coordinator	Х
City of Pinehurst	Emergency Management Coordinator	Х
City of Pine Forest	Emergency Management Coordinator	Х
City of Rose City	City Secretary	Х
City of Vidor Police Department	Emergency Management Coordinator	Х
City of West Orange	Emergency Management Coordinator	Х
Orange County	Tax Assessor-Collector	Х
Orange County Office of Emergency Management	Deputy Emergency Management Coordinator	х
Orange County Office of Emergency Management	Emergency Management Coordinator	Х

Stakeholders and participants from neighboring communities that attended the Planning Team and public meetings played a key role in the planning process. For example, hurricanes and flooding were major concerns to the stakeholders, so many of the participating jurisdictions included mitigation actions to improve their drainage systems to reduce flood damages to structures and infrastructure in the area, as well as revising their evacuation routes and plans to ensure safety to the residents during times when an evacuation is necessary.

#### **Public Meetings**

A series of public meetings were held throughout the planning area, to collect public and stakeholder input. Topics of discussion included the purpose of hazard mitigation, discussion of the planning process, and types of natural hazards. Representatives from area neighborhood associations, and area residents were invited to participate. Additionally, Jefferson County utilized social media sources including

Facebook, Twitter, and the local media to increase public participation in the Plan Update development process. Documentation on the public meetings are found in Appendix F.

Public meetings were held on the following dates and locations:

- March 30, 2016, SETRPC Homer E Nagel Conference Room
- June 1, 2016, Hardin County Courthouse Commissioners' Courtroom
- August 24, 2016, Orange County Expo Center

#### Public Participation Survey

In addition to public meetings, the Planning and Consultant Teams developed a public survey designed to solicit public input during the planning process from citizens and stakeholders, and to obtain data regarding the identification of any potential hazard mitigation actions or problem areas. The survey was promoted by local officials and a link to the survey was posted on Jefferson County's website. A total of 69 surveys were completed online. The survey results are analyzed in Appendix C. Jefferson County reviewed the input from the surveys and decided which information to incorporate into the Plan Update as hazard mitigation actions. For example, many citizens mention concerns about flooding, and suggested levee/drainage improvements as potential steps the jurisdictions could take. In response to public input several hazard mitigation actions were added to the Plan Update to pursue funding and implement drainage improvements through the County to include installing/upgrading culverts and headwalls as well as enlarging storm water ditches and canals.

# **SECTION 3: COUNTY PROFILE**

Overview	. 1
Population and Demographics	.4
Population Growth	.5
Future Development	.6
Economic Impact	.6
Existing and Future Land Use and Development Trends	.6
Building Permits	.7

### Overview

Jefferson County was established in 1836 as a municipality of Mexico and was organized as a county in 1837. It is named for U.S. president Thomas Jefferson, and Beaumont is the county seat. Jefferson County has a total area of 1,113 square miles, of which 876 square miles is land and 236 square miles (21%) is water. The County consists of several cities, a few census-designated places, and unincorporated areas. The following cities are participating within this plan and are considered part of the planning area: the City of Beaumont, the City of Bevil Oaks, the City of China, the City of Groves, the City of Nederland, the City of Nome, the City of Port Arthur, and the City of Port Neches. The other unincorporated communities will be considered under Jefferson County.

Primary waterways within Jefferson County include the Neches River, Pine Island Bayou and its tributaries, Walker Branch and Walker Branch Tributary, Taylor Bayou and its tributaries, Rhodair Gully, Mayhaw Bayou, Hillebrandt Bayou (a major tributary of Taylor Bayou) with its tributaries, Willow Marsh Bayou, Bayou Din and its tributaries, Bayou Din Tributary, Kidd Gully and Cotton Creek. Soils in Jefferson County have high concentrations of clay and silt, with low infiltration rates and high runoff potential.

Jefferson County is characterized by flat, featureless terrain that slopes gently to the Gulf of Mexico. Elevations range from sea level to approximately forty-five (45) feet. Belts of hardwood and pine are found in the upland portions of the county. Swamps exist in the floodplains of the Neches River and the major bayous. The swamps support vegetation such as cypress trees, and water tolerant grass and sedges in a few cleared areas. Most open land is prairie land used for grazing and rice production. Large areas of tidal marsh along the Gulf of Mexico/Sabine Lake support a dense growth of salt-water vegetation, principally cord grass and marsh cane.



The majority of developed land in the county is primarily agricultural, although many parts are highly industrialized. Commercial, residential, and recreational areas are generally located in the eastern portion of the county. The Gulf of Mexico shore is mostly undeveloped and is used extensively for public recreation. Leading industries in the area produce petroleum and natural gas, sulfur, petrochemicals, and petroleum and natural gas products.

Figure 3-1 shows the general location of Jefferson County, along with the Cities that are located within the County.



Figure 3-1. Location of Jefferson County Planning Area

Figure 3-2 shows the Jefferson County Study Area, including the participating jurisdictions that are covered in the risk assessment analysis of the Plan.



Figure 3-2. Jefferson County Study Area

Provided in Table 3-1 below is a listing of the jurisdictions in Jefferson County that participated in the Hazard Mitigation Plan Update.



#### Table 3-1. Participating Jurisdictions

## **Population and Demographics**

In the official Census population count, as of April 1, 2010, Jefferson County had a population of 252,273 residents. By July 2014, the number had grown to 252,439, and by July 2015, the population was 254,308. Table 3-2 provides the population distribution by jurisdiction within Jefferson County.<sup>1</sup>

Between official U.S. Census population counts, the estimate uses a formula based on new residential building permits and household size. It is simply an estimate and there are many variables involved in achieving an accurate estimation of people living in a given area at a given time.

JURISDICTION	TOTAL 2010 POPULATION	TOTAL 2010 POPULATION		ESTIMATED VULNERABLE OR SENSITIVE POPULATIONS Elderly Below (Over 65) Poverty Level		
Beaumont	118,296	46.9%	14,432	25,789		
Bevil Oaks	1,274	0.5%	280	41		
China	1,160	0.4%	186	263		

Table 3-2.	Population	Distribution	bv	Jurisdiction
			~,	

<sup>1</sup> http://www.census.gov/quickfacts/table/PST045215/48245,00

	TOTAL 2010		ESTIMATED VULNERABLE OR SENSITIVE POPULATIONS		
JURISDICTION	POPULATION	PERCENTAGE	Elderly (Over 65)	Below Poverty Level	
Groves	16,144	6.4%	2,518	2,147	
Nederland	17,547	7.0%	2,527	1,491	
Nome	588	0.2%	67	106	
Port Arthur	53,818	21.3%	7,158	14,692	
Port Neches	13,040	5.2%	2,034	1,630	
Unincorporated Jefferson County	30,406	12.1%	2,837	7,323	
JEFFERSON COUNTY TOTAL	252,273	100%	32,039	53,482	

#### **Population Growth**

The official 2010 Jefferson County population is 252,273. Overall, Jefferson County experiences a slight increase in population between 1980 and 2010 by 1.46%, or an increase by 3,621 people. Beaumont, Nederland, Nome, and the unincorporated areas of Jefferson County experienced a population growth between 1980 and 2010, while the rest of the cities experienced a decrease in their population. Beaumont, China, Groves, Nederland, and Nome experienced an increase in population between 2000 and 2010, while the rest of the cities and the unincorporated areas of the County exhibited a decrease in population during this time period. Table 3-2 provides historic growth rates in Jefferson County.

JURISDICTIONS	1980	1990	2000	2010	POP CHANGE 1980-2010	PERCENT OF CHANGE	POP CHANGE 2000-2010	PERCENT OF CHANGE
Beaumont	118,102	114,323	113,866	118,296	194	0.164%	4,430	3.89%
Bevil Oaks	1,306	1,350	1,346	1,274	-32	-2.45%	-72	-5.35%
China	1,351	1,144	1,112	1,160	-191	-14.14%	48	4.32%
Groves	17,090	16,744	15,733	16,144	-964	-5.54%	411	2.61%
Nederland	16,855	16,192	17,422	17,547	692	4.11%	125	0.72%
Nome	550	448	515	588	38	6.91%	73	14.17%
Port Arthur	61,251	58,724	57,755	53,818	-7,433	-12.14%	-3,937	-6.82%
Port Neches	13,944	12,908	13,601	13,040	-904	-6.48%	-561	-4.12%
Unincorporated Jefferson County	18,203	17,556	30,701	30,406	12,203	67.04%	-295	-0.96%
COUNTY TOTAL	248,652	239,389	252,051	252,273	3,621	1.46%	222	0.08%

#### Table 3-2. Population for Jefferson County, 1980-2010

# Future Development

To better understand how future growth and development in the County might affect hazard vulnerability, it is useful to consider population growth, occupied and vacant land, the potential for future development in hazard areas, and current planning and growth management efforts. This section includes an analysis of the projected population change, the number of permits that have been issued throughout the county, and economic impacts.

Population projections from 2010 to 2040 are listed in Table 3-3, as provided by the Office of the State Demographer, Texas State Data Center, and the Institute for Demographic and Socioeconomic Research. Population projections are based on a 0.5 scenario growth rate, which is 50 percent of the population growth rate that occurred during 2000-2010. This information is only available at the County level; however, the population projection shows an increase in population density for the County, which would mean overall growth for the County.

		20	010	20	020	20 20		20	040
	LAND	Population							
County	AREA (SQ MI)	Total Number	Density (Land Area, SQ MI)						
Jefferson	1,113	252,273	226.7	267, 88	240.1	283,813	255.0	300,728	270.2

Table 3-3. Jefferson County Population Projects

# Economic Impact

Building and maintaining infrastructure depends on the economy; therefore, protecting infrastructure from risk due to natural hazards in the planning area is important to Jefferson County. Whether it's expanding culverts under a road that washes out during flash flooding, shuttering a fire station, or flood-proofing a wastewater facility, infrastructure must be mitigated from natural hazards in order to continue providing essential utility and emergency response services in a fast-growing planning area.

Major employers in the area are critical to the health of the economy, as well as effective transportation connectivity.

# Existing and Future Land Use and Development Trends

Jefferson County is part of the South East Texas Regional Planning Commission (SETRPC) which has many departments to promote intergovernmental cooperation and coordination, conduct comprehensive regional planning, and provide a forum for the discussion and study of area issues. The Community Development Department focus on building a stronger more prosperous region through the focus on an individual community, while the Transportation and Environmental Resources department provides assistance through grants and resources regarding the environment and working with state, city, and

county entities to coordinate transportation planning for the Jefferson-Orange-Hardin Regional Transportation Study area.

Additionally, the City of Beaumont, the City of Groves, the City of Nederland and the City of Port Arthur all have a Comprehensive Land Use Plan in place. These plans, along with the planning department are responsible for the comprehensive planning activities of the city and for administering the subdivision regulations.

#### **Building Permits**

Building permits indicate what types of buildings are being constructed and their relative uses. Table 3-4 lists the number of residential building permits for Jefferson County that have been granted between 1996 and 2015. The data includes all sizes of family homes for reported permits, as well as the construction costs, to show the potential increase in vulnerability of structures to the various hazards reviewed in the risk assessment. The increase in vulnerability can be attributed to the higher construction costs that would be factored into repairing or replacing a structure using current market values. Permits are reported annually in September; data reflects permits for years 2010, 2011, 2012, 2013, 2014, and 2015 to demonstrate growth rates.

Jefferson County							
Year	Buildings	Units	Construction Cost				
1996	428	453	\$40,661,459				
2000	602	615	\$73,176,250				
2005	450	819	\$83,568,395				
2010	965	1,111	\$104,441,681				
2011	588	743	\$81,941,185				
2012	552	995	\$106,460,700				
2013	362	364	\$70,423,743				
2014	839	1,061	\$122,734,784				
2015	513	599	\$95,208,119				

#### Table 3-4. County Residential Building Permits<sup>2</sup>

<sup>&</sup>lt;sup>2</sup> http://censtats.census.gov/cgi-bin/bldgprmt/bldgdisp.pl

# **SECTION 4: RISK OVERVIEW**

Hazard Description	1
Natural Hazards and Climate Change	3
Overview of Hazard Analysis	4
Hazard Ranking	6

## Hazard Description

Section 4 is the first phase of the Risk Assessment, providing background information for the hazard identification process, and descriptions for the hazards identified. The Risk Assessment continues with Sections 5 through 15, which include hazard descriptions and vulnerability assessments.

Upon a review of the full range of natural hazards suggested under FEMA planning guidance, Jefferson County and the participating jurisdictions identified eleven natural hazards that are addressed in the 2017 Hazard Mitigation Plan Update (Plan or Plan Update). Of the hazards identified, eleven natural hazards) were identified as significant, as shown in Table 4-1. The hazards were identified through input from Planning Team members, and a review of the current 2013 State of Texas Hazard Mitigation Plan Update (State Plan Update). Readily available online information from reputable sources such as federal and state agencies were also evaluated and utilized to supplement information as needed.

In general, there are three main categories of hazards including atmospheric, hydrologic, and technological. Atmospheric hazards, are events or incidents associated with weather generated phenomenon. Atmospheric hazards that have been identified as significant for the Jefferson County Planning area include extreme heat, hail, hurricane, lightning, thunderstorm wind, tornado, and winter storm (Table 4-1).

Hydrologic hazards, are events or incidents associated with water related damage and account for over 75 percent of Federal disaster declarations in the United States. Hydrologic hazards identified as significant for the planning area include flood and drought.

Technological hazards, refers to the origins of incidents that can arise from human activities, such as the construction and maintenance of dams. Technological hazards are distinct from natural hazards primarily because they originate from human activity. The risks presented by natural hazards may be increased or decreased as a result of human activity, however they are not inherently human-induced.

For the Risk Assessment, the wildfire hazard is considered "other," since a wildfire may be natural or human-caused, and is not considered atmospheric or hydrologic.

#### Table 4-1. Hazard Descriptions

HAZARD	DESCRIPTION		
ATMOSPHERIC			
Extreme Heat	Extreme heat is the condition whereby temperatures hover ten degrees or more above the average high temperature in a region for an extended period of time.		
Hail	Hailstorms are a potentially damaging outgrowth of severe thunderstorms. Early in the developmental stages of a hailstorm, ice crystals form within a low-pressure front due to the rapid rising of warm air into the upper atmosphere and subsequent cooling of the air mass.		
Hurricane	A hurricane is an intense tropical weather system of strong thunderstorms with a well-defined surface circulation and maximum sustained winds of 74 mph or higher.		
Lightning	Lightning is a sudden electrostatic discharge that occurs during an electrical storm. This discharge occurs between electrically charged regions of a cloud, between two clouds, or between a cloud and the ground.		
Thunderstorm Wind	A thunderstorm occurs when an observer hears thunder. Radar observers use the intensity of the radar echo to distinguish between rain showers and thunderstorms. Lightning detection networks routinely track cloud-to-ground flashes, and therefore thunderstorms.		
Tornado	A tornado is a violently rotating column of air that has contact with the ground and is often visible as a funnel cloud. Its vortex rotates cyclonically with wind speeds ranging from as low as 40 mph to as high as 300 mph. The destruction caused by tornadoes ranges from light to catastrophic, depending on the location, intensity, size, and duration of the storm.		
Winter Storm	Severe winter storms may include snow, sleet, freezing rain, or a mix of these wintry forms of precipitation. Blizzards, the most dangerous of all winter storms, combine low temperatures, heavy snowfall, and winds of at least 35 miles per hour, reducing visibility to only a few yards. Ice storms occur when moisture falls and freezes immediately upon impact on trees, power lines, communication towers, structures, roads, and other hard surfaces. Winter storms and ice storms can down trees, cause widespread power outages, damage property, and cause fatalities and injuries to human life.		
	HYDROLOGIC		
Drought	A prolonged period of less than normal precipitation such that the lack of water causes a serious hydrologic imbalance. Common effects of drought include crop failure, water supply shortages, and fish and wildlife mortality.		

HAZARD	DESCRIPTION		
Flood	The accumulation of water within a body of water, which results in the overflow of excess water onto adjacent lands, usually floodplains. The floodplain is the land adjoining the channel of a river, stream, ocean, lake, or other watercourse or water body that is susceptible to flooding. Most floods fall into the following three categories: riverine flooding, coastal flooding, and shallow flooding.		
Coastal Erosion	Coastal erosion is a hydrologic hazard defined as the wearing away of land and loss of beach, shoreline, or dune material as a result of natural coastal processes or manmade influences. Coastal Erosion occurrences and damages are not well documented, however, team members indicate that coastal erosion pose little to no risk for the area based on local knowledge and experience. Coastal Erosion is ranked as a minimal hazard risk in the State's HMAP and for the purposes of this Plan, is addressed as a sub-hazard of Hurricanes.		
OTHER			
Wildfire	A wildfire is an uncontrolled fire burning in an area of vegetative fuels such as grasslands, brush, or woodlands. Heavier fuels with high continuity, steep slopes, high temperatures, low humidity, low rainfall, and high winds all work to increase the risk for people and property located within wildfire hazard areas or along the urban/wildland interface. Wildfires are part of the natural management of forest ecosystems, but most are caused by human factors.		

Hazards that weren't considered significant and were not included in the Plan are located in Table 4-2, along with the evaluation process used for determining the significance of each of these hazards. These natural hazards are not addressed in detail due to their no to minimal level of risk within the Jefferson County planning area. Hazards not identified for inclusion at this time may be addressed during future evaluations and updates.

HAZARD	DESCRIPTION		
ATMOSPHERIC			
Expansive Soils	Expansive soils occurrences and damages are not well documented. There are no historical occurrences of expansive soils for the Jefferson County planning area and it is located in an area where occurrences are considered rare. Expansive Soils poses little to no risk for the area and was not addressed further in the plan.		
Earthquakes	According to the State Plan, an earthquake occurrence for the Jefferson County planning area is considered exceedingly rare. Although a small event is possible, it would pose little to no risk for the area. There are no recorded earthquake events or damages for		

Jefferson County | Hazard Mitigation Plan Update 2017 | Page 3

HAZARD	DESCRIPTION		
	the planning area. Due to the low frequency of the hazard and limited impact, the hazard was not addressed further in the plan.		
Land Subsidence	There are no historical occurrences of land subsidence for the Jefferson County planning area and it is located in an area where occurrences are considered rare. The impact would be limited and the frequency of occurrence is unlikely according to the State Plan. Land Subsidence poses little to no risk for the area and was not addressed further in the plan.		
Dam Failure	There are 3 dams in the Jefferson County planning area. However, there are no historical occurrences of these dams failing and there is none expected in the future. If the dams were to fail Jefferson County would not experience any impact to life, property, or services provided by the community.		

# Natural Hazards and Climate Change

Climate change is defined as a long-term hazard which can increase or decrease the risk of other weather hazards; and directly endangers property due to sea level rise, and biological organisms due to habitat destruction.

Global climate change is expected to exacerbate the risks of certain types of natural hazards impacted through rising sea levels, warmer ocean temperatures, higher humidity, the possibility of stronger storms and an increase in wind and flood damages due to storm surges. While sea level rise is a natural phenomenon and has been occurring for several thousand years, the general scientific consensus is that the rate has increased in the past 200 years, from 0.5 millimeters per year to 2 millimeters per year.

Texas is considered one of the more vulnerable states in the U.S. to both abrupt climate changes and to the impact of gradual climate changes to the natural and built environments. Mega-droughts can trigger abrupt changes to regional ecosystems and the water cycle, drastically increase extreme summer temperature and fire risk, and reduce availability of water resources, as Texas experienced during 2011-2012.

Paleoclimate records also show that the climate over Texas had large changes between periods of frequent mega-droughts and the periods of mild droughts that Texas is currently experiencing. While the cause of these fluctuations is unclear, it would be wise to anticipate that such changes could occur again, and may even be occurring now.

## **Overview of Hazard Analysis**

The methodologies utilized to develop the Risk Assessment are a historical analysis and a statistical approach. Both methodologies provide an estimate of potential impact by using a common, systematic framework for evaluation.

Records retrieved from the National Centers for Environmental Information (NCEI) and the National Oceanic and Atmospheric Administration (NOAA) were reported for the Jefferson County planning area, including the participating jurisdictions. Remaining records identifying the occurrence of hazard events in the planning area and the maximum recorded magnitude of each event were also evaluated.

The use of geographic information system (GIS) technology to identify and assess risks for the Jefferson County planning area, and evaluate community assets and their vulnerability to the hazards.

The four general parameters that are described for each hazard in the Risk Assessment include frequency of return, approximate annualized losses, a description of general vulnerability, and a statement of the hazard's impact.

Frequency of return was calculated by dividing the number of events in the recorded time period for each hazard by the overall time period that the resource database was recording events. Frequency of return statements are defined in Table 4-3, and impact statements are defined in Table 4-4 below.

PROBABILITY	DESCRIPTION
Highly Likely	Event is probable in the next year.
Likely	Event is probable in the next three years.
Occasional	Event is probable in the next five years.
Unlikely	Event is probable in the next ten years.

#### Table 4-3. Frequency of Return Statements

#### Table 4-4. Impact Statements

POTENTIAL SEVERITY	DESCRIPTION
Substantial	Multiple deaths. Complete shutdown of facilities for 30 days or more. More than 50 percent of property destroyed or with major damage.
Major	Injuries and illnesses resulting in permanent disability. Complete shutdown of critical facilities for at least two weeks. More than 25 percent of property destroyed or with major damage.
Minor	Injuries and illnesses do not result in permanent disability. Complete shutdown of critical facilities for more than one week. More than 10 percent of property destroyed or with major damage.
Limited	Injuries and illnesses are treatable with first aid. Shutdown of critical facilities and services for 24 hours or less. Less than 10 percent of property destroyed or with major damage.

Each of the hazard profiles includes a description of a general Vulnerability Assessment. Vulnerability is the total of assets that are subject to damages from a hazard, based on historic recorded damages. Assets in the region were inventoried and defined in hazard zones where appropriate. The total amount of damages, including property and crop damages, for each hazard is divided by the total number of assets (building value totals) in that community to determine the percentage of damage that each hazard can cause to the community.

To better understand how future growth and development in the County might affect hazard vulnerability, it is useful to consider population growth, occupied and vacant land, the potential for future development in hazard areas, and current planning and growth management efforts. Hazard Vulnerability for Jefferson County, and all participating jurisdictions, was reviewed based on recent development changes that occurred throughout the County. The Jefferson County planning area has grown slightly between 2010 and 2015 according to the U.S. Census Bureau, therefore there has been no significant factors or development trends with a consequential effect or increase in vulnerability to the population, infrastructure, and buildings for hazards.

Once loss estimates and vulnerability were known, an impact statement was applied to relate the potential impact of the hazard on the assets within the area of impact.

## Hazard Ranking

Table 4-5 portrays the results of the County's self-assessment for hazard ranking, based on the preliminary results of the risk assessment presented at the Risk Assessment Workshop. This table also takes into account local knowledge regarding frequency of occurrence and the potential impact of each hazard.

HAZARD	FREQUENCY OF OCCURENCE	POTENTIAL SEVERITY	RANKING
Flood	Highly Likely	Limited/Major <sup>1</sup>	High
Hurricane	Highly Likely	Substantial	High
Lightning	Highly Likely	Substantial	Moderate
Extreme Heat	Occasional	Minor	Moderate
Thunderstorm Wind	Highly Likely	Substantial	Moderate
Drought	Occasional	Limited	Moderate
Hail	Highly Likely	Limited	Low
Tornado	Highly Likely	Major	Low
Wildfire	Highly Likely	Minor	Low

Table 4-5.	Hazard	Risk	Ranking
------------	--------	------	---------

<sup>&</sup>lt;sup>1</sup> The Potential Severity for the City of Beaumont is considered "Major" while the rest of the jurisdictions, including Jefferson County have a Potential Severity of "Limited".

HAZARD	FREQUENCY OF OCCURENCE	POTENTIAL SEVERITY	RANKING
Winter Storm	Highly Likely	Limited	Low
Coastal Erosion	Highly Likely	Limited	Low

# **SECTION 5: FLOOD**

Hazard Description	1
Location	2
Extent	
Historical Occurrences	
Significant Events	19
Probability of Future Events	
Vulnerability and Impact	
Assessment of Impacts	22
National Flood Insurance Program (NFIP) Participation	23
NFIP Compliance and Maintenance	25
Repetitive Loss	25

# Hazard Description

Floods generally result from excessive precipitation. The severity of a flood event is determined by a combination of several major factors, including: stream and river basin topography and physiography; precipitation and weather patterns; recent soil moisture conditions; and the degree of vegetative clearing and impervious surface. Typically, floods are long-term events that may last for several days.

The primary types of general flooding are inland and coastal flooding. Inland or riverine flooding is a result of excessive precipitation levels and water runoff volumes within the watershed of a stream or river. Inland or riverine flooding is overbank flooding of rivers and streams, typically resulting from large-scale weather systems that generate prolonged rainfall over a wide geographic area; thus it is a naturally occurring and inevitable event. Some river floods occur seasonally when winter or spring rainfalls fill river basins with too much water, too quickly. Torrential rains from decaying hurricanes or tropical systems can also produce river flooding.

Coastal flooding occurs when normally dry, low-lying land is flooded by seawater. The extent of coastal flooding is a function of the elevation inland flood waters penetrate, which is controlled by the topography of the coastal land exposed to flooding.

Coastal flooding is largely a natural event, however human influence on the coastal environment can exacerbate coastal flooding. Extraction of water from groundwater reservoirs in the coastal zone can enhance subsidence of the land, increasing the risk of flooding. Engineered protection structures along the coast, such as sea walls, alter the natural processes of the beach, often leading to erosion on adjacent stretches of the coast, which also increases the risk of flooding. Coastal flooding is covered in detail under the profile for Hurricanes, located in Section 7.

## Location

The Digital Flood Insurance Rate Map (DFIRM) data provided by FEMA for Jefferson County shows the following flood hazard areas:

- Zone A: Areas subject to inundation by the 1-percent-annual-chance flood event generally determined using approximate methodologies. Because detailed hydraulic analyses have not been performed, no Base Flood Elevations (BFEs) or flood depths are shown. Mandatory flood insurance requirements and floodplain management standards apply.
- Zone AE: Areas subject to inundation by 1-percent-annual-chance shallow flooding. It is the base floodplain where BFEs are provided. AE zones are now used on new format FIRMs instead of A1-30 zones.
- Zone AH: Areas subject to inundation by the 1-percent-annual-chance shallow flooding, usually areas of ponding, where average depths range from 1 to 3 feet. BFEs derived from detailed hydraulic analyses are shown. Mandatory flood insurance purchase requirements and floodplain management standards apply.
- Zone VE: Areas subject to inundation by the 1-percent-annual-chance flood event with additional hazards due to storm-induced velocity wave action. BFEs derived from detailed hydraulic analyses are shown. Mandatory flood insurance purchase requirements and floodplain management standards apply.
- Zone X: Moderate risk areas within the 0.2-percent-annual-chance floodplain, areas of 1-percentannual-chance flooding where average depths are less than 1 foot, areas of 1-percent-annualchance flooding where the contributing drainage area is less than 1 square mile, and areas protected from the 1-percent-annual-chance flood by a levee. No BFEs or base flood depths are shown within these zones.

Locations of flood zones in Jefferson County based on the DFIRM from FEMA are illustrated in Figures 5-1 to 5-9.


Figure 5-1. Estimated Flood Zones in Jefferson County



Figure 5-2. Estimated Flood Zones in the City of Beaumont, SETRPC



Figure 5-3. Estimated Flood Zones in the City of Bevil Oaks



Figure 5-4. Estimated Flood Zones in the City of China



Figure 5-5. Estimated Flood Zones in the City of Groves



Figure 5-6. Estimated Flood Zones in the City of Nederland



Figure 5-7. Estimated Flood Zones in the City of Nome







Figure 5-9. Estimated Flood Zones in the City of Port Neches

## Extent

The severity of a flood event is determined by a combination of several factors including: stream and river basin topography and physiography; precipitation and weather patterns; recent soil moisture conditions; and degree of vegetative clearing and impervious surface. Typically, floods are long-term events that may last for several days.

Determining the intensity and magnitude of a flood event is dependent upon the flood zone and location of the flood hazard area, in addition to depths of flood waters. The extent of flood damages can be expected to be more damaging in the areas that will convey a base flood. FEMA categorizes areas on the terrain according to how the area will convey flood water. Flood zones are the categories that are mapped on Flood Insurance Rate Maps. Table 5-1 provides a description of FEMA flood zones and the flood impact in terms of severity or potential harm. Flood Zones A, AE, and X are the only hazard areas mapped in the region. Figures 5-1 through 5-9 (above) should be read in conjunction with the extent for flooding in Tables 5-1, 5-2, and 5-3 in order to determine the intensity of a potential flood event.

INTENSITY	ZONE	DESCRIPTION			
	ZONE A	Areas with a 1-percent-annual-chance of flooding and a 26-percent- chance of flooding over the life of a 30-year mortgage. Because detailed analyses are not performed for such areas, no depths or base flood elevations are shown within these zones.			
	ZONE A1-30	These are known as numbered A Zones (e.g., A7 or A14). This is the base floodplain where the FIRM shows a Base Flood Elevation (BFE) (old format).			
	ZONE AE	The base floodplain where base flood elevations are provided. AE Zones are now used on the new format FIRMs instead of A1-A30 Zones.			
HIGH ZONE AO		River or stream flood hazard areas and areas with a 1-percent or greater chance of shallow flooding each year, usually in the form of sheet flow, with an average depth ranging from 1 to 3 feet. These areas have a 26-percent-chance of flooding over the life of a 30-year mortgage. Average flood depths derived from detailed analyses are shown within these zones.			
	ZONE AH	Areas with a 1-percent-annual-chance of shallow flooding, usually in the form of a pond, with an average depth ranging from 1 to 3 feet. These areas have a 26-percent chance of flooding over the life of a 30- year mortgage. Base flood elevations derived from detailed analyses are shown at selected intervals within these zones.			

#### Table 5-1. Flood Zones

INTENSITY	ZONE	DESCRIPTION				
	ZONE A99	Areas with a 1-percent-annual-chance of flooding that will be protected by a federal flood control system where construction has reached specified legal requirements. No depths or base flood elevations are shown within these zones.				
	ZONE AR	Areas with a temporarily increased flood risk due to the building or restoration of a flood control system (such as a levee or a dam). Mandatory flood insurance purchase requirements will apply, but rates will not exceed the rates for unnumbered A zones if the structure is built or restored in compliance with Zone AR floodplain management regulations.				
	ZONE V	Areas along coasts subject to inundation by the 1-percent-annual- chance flood event with additional hazards associated with storm- induced waves. Because detailed hydraulic analyses have not been performed, no BFEs or flood depths are shown. Mandatory flood insurance purchase requirements and floodplain management standards apply.				
	ZONE VE	Areas subject to inundation by the 1-percent-annual-chance flood event with additional hazards due to storm-induced velocity wave action. BFEs derived from detailed hydraulic analyses are shown. Mandatory flood insurance purchase requirements and floodplain management standards apply.				
MODERATE to LOW	ZONE X 500	An area inundated by 500-year flooding; an area inundated by 100-year flooding with average depths of less than 1 foot or with drainage areas less than 1 square mile; or an area protected by levees from 100-year flooding.				

Zone A is interchangeably referred to as the 100-year flood, the 1-percent-annual chance flood, the Special Flood Hazard Area (SFHA), or more commonly, the base flood. This is the area that will convey the base flood and constitute a threat to the planning area. The impact from a flood event can be more damaging in areas that will convey a base flood.

Structures built in the SFHA are subject to damage by rising waters and floating debris. Moving flood water exerts pressure on everything in its path and causes erosion of soil and solid objects. Utility systems, such as heating, ventilation, air conditioning, fuel, electrical systems, sewage maintenance systems, and water systems, if not elevated above BFE, may also be damaged.

The intensity and magnitude of a flood event is also determined by the depth of flood waters. Table 5-2 below describes the category of risk and potential magnitude of an event in correlation to water depth. The water depths depicted in Table 5-2 are approximations based on elevation data. Table 5-3 describes the extent associated with stream gauge data provided by the United States Geological Survey (USGS).

#### Table 5-2. Extent Scale – Water Depth

SEVERITY	DEPTH (in feet)	DESCRIPTION		
BELOW FLOOD STAGE	0 to 15	Water begins to exceed low sections of banks and the lowest sections of the floodplain.		
ACTION STAGE	16 to 23	Flow is well into the floodplain, minor lowland flooding reaches low areas of the floodplain. Livestock should be moved from low lying areas.		
FLOOD STAGE	24 to 28	Homes are threatened and properties downstream of river flows or in low lying areas begin to flood.		
MODERATE FLOOD STAGE	29 to 32	At this stage the lowest homes downstream flood. Roads and bridges in the floodplain flood severely and are dangerous to motorists.		
MAJOR FLOOD STAGE 33 and above		Major flooding approaches homes in the floodplain. Primary and secondary roads and bridges are severely flooded and very dangerous. Major flooding extends well into the floodplain, destroying property, equipment, and livestock.		

#### Table 5-3. Extent for Jefferson County<sup>1</sup>

JURISDICTION <sup>2</sup>	ESTIMATED SEVERITY PER FLOOD EVENT	PEAK FLOOD EVENT
Jefferson County	Below Flood Stage, 0 to 15 feet	Below Flood Stage: Taylor Bayou reached an overflow elevation of 11.29 feet in September 1963 near the City of LaBelle.
Jefferson County	Below Flood Stage, 0 to 15 feet	Below Flood Stage: Hillebrandt Bayou reached an overflow elevation of 12.30 feet in September 1963 near Lovell Lake.
Jefferson County	Below Flood Stage, 0 to 15 feet	Action Stage: Pine Island Bayou reached an overflow elevation of 16.18 feet in October 2006 near Sour Lake, Texas.
City of Beaumont	Below Flood Stage, 0 to 15 feet	Below Flood Stage: Neches River reached an overflow elevation of 11.71 feet in October 2006 in Beaumont, Texas.

<sup>&</sup>lt;sup>1</sup> Severity estimated by averaging floods at certain stage level over the history of flood events. Severity and peak events are based on U.S. Geological Survey data.

<sup>&</sup>lt;sup>2</sup> Severity is provided for jurisdictions where peak data was provided.

The range of flood intensity that the County can experience is high, or Zone A. Based on reporting from the USGS, a flood event can place the County at the extent of "Below Flood Stage" as shown in Tables 5-2 and 5-3. Based on historical occurrences, on average, the entire planning area, including all participating jurisdictions, could expect to experience 7 inches of water within a 5 hour period due to flooding.

The data described in Tables 5-1 through 5-3, together with Figures 5-1 through 5-9, and historical occurrences for the area, provides an estimated potential magnitude and severity for the County. For example the City of Port Arthur, as shown in Figure 5-8, has areas designated as Zone AE and Zone VE. Reading this figure in conjunction with Table 5-1 means the area is of high risk for flood. It is noted that the SETRPC is not located in a special flood hazard area and has no known localized flood risk.

## **Historical Occurrences**

Historical evidence indicates that areas within Jefferson County are susceptible to flooding, especially in the form of flash flooding. It is important to note that only flood events that have been reported have been factored into this risk assessment; therefore it is likely that additional flood occurrences have gone unreported before and during the recording period. Table 5-4 identifies historical flood events that resulted in damages, injuries, or fatalities within the Jefferson County planning area. Table 5-5 provides the historical flood event summary by jurisdiction. Historical Data is provided by the National Oceanic and Atmospheric Administration and the National Weather Service's (NOAA/NWS) Storm Prediction Center, in addition to the National Center for Environmental Information (NCEI) database for Jefferson County.

JURISDICTION	DATE	TIME	DEATHS	INJURIES	PROPERTY DAMAGE	CROP DAMAGE
Beaumont	1/26/1996	10:00 AM	0	0	\$0	\$0
Jefferson County	9/27/1996	9:00 AM	0	0	\$137,243	\$0
Port Arthur	7/30/1997	3:00 PM	0	0	\$29,814	\$0
Port Arthur	9/23/1997	3:45 PM	0	0	\$14,907	\$0
Nederland	1/21/1998	11:00 PM	0	0	\$58,714	\$0
Beaumont	8/14/1998	1:00 PM	0	0	\$14,679	\$0
Beaumont	9/11/1998	1:00 PM	0	0	\$146,786	\$0
Jefferson County	9/13/1998	10:00 AM	0	0	\$146,786	\$0
Beaumont	10/6/1998	8:30 AM	0	0	\$29,357	\$0
Nederland	4/12/2000	10:00 AM	0	0	\$347,359	\$0

#### Table 5-4. Historical Flood Events, 1996-2016<sup>3</sup>

<sup>3</sup> Values are in 2016 dollars.

JURISDICTION	DATE	TIME	DEATHS	INJURIES	PROPERTY DAMAGE	CROP DAMAGE
Beaumont	6/7/2001	3:30 AM	0	0	\$13,509,938	\$0
Jefferson County	9/1/2001	10:00 AM	0	0	\$33,775	\$0
Jefferson County	9/2/2001	10:00 AM	0	0	\$101,325	\$0
Beaumont	11/28/2001	5:30 PM	0	0	\$27,020	\$0
Jefferson County	6/27/2002	3:00 AM	0	0	\$6,650	\$0
Jefferson County	8/15/2002	4:00 AM	0	0	\$26,599	\$0
Beaumont	10/29/2002	12:30 AM	1	0	\$6,649,833	\$0
Beaumont	11/3/2002	11:10 AM	0	0	\$1,329,967	\$0
Port Arthur	12/4/2002	7:00 AM	0	0	\$66,498	\$0
Beaumont	2/21/2003	1:34 AM	0	0	\$32,508	\$0
Jefferson County	9/12/2003	7:00 AM	0	0	\$32,508	\$0
Beaumont	10/9/2003	3:45 PM	0	0	\$1,300,332	\$0
Port Neches	10/25/2003	2:00 PM	0	0	\$130,033	\$0
Port Arthur	5/11/2004	8:40 PM	0	0	\$2,533	\$0
Nederland	5/13/2004	3:45 PM	0	0	\$6,333	\$0
Beaumont	6/26/2004	4:10 PM	0	0	\$6,333	\$0
Nederland	9/23/2004	8:50 PM	0	0	\$12,666	\$0
Jefferson County	5/29/2006	10:25 AM	1	0	\$11,868	\$0
Beaumont	5/29/2006	3:09 AM	0	0	\$59,341	\$0
Beaumont	5/29/2006	6:18 AM	0	0	\$29,670	\$0
Beaumont	7/23/2006	4:30 PM	0	0	\$29,670	\$0
Beaumont	7/26/2006	1:00 PM	0	0	\$2,374	\$0
Beaumont	10/16/2006	7:00 AM	0	0	\$0	\$0
Beaumont	10/16/2006	4:15 PM	0	0	\$11,868	\$0
Port Arthur	10/16/2006	6:00 PM	0	0	\$5,934	\$0
Bevil Oaks	2/12/2007	6:00 PM	0	0	\$11,539	\$0

JURISDICTION	DATE	TIME	DEATHS	INJURIES	PROPERTY DAMAGE	CROP DAMAGE
Jefferson County	7/6/2007	9:30 AM	0	0	\$5,770	\$0
Jefferson County	9/13/2007	1:00 AM	0	0	\$57,697	\$0
China	1/4/2009	3:25 AM	0	0	\$5,576	\$0
Jefferson County	4/18/2009	12:00 AM	0	0	\$22,305	\$0
Beaumont	4/27/2009	8:15 PM	0	0	\$5,576	\$0
Jefferson County	9/9/2009	10:30 AM	0	0	\$0	\$0
Beaumont	10/22/2009	8:00 AM	0	0	\$111,524	\$0
Port Arthur	10/22/2009	9:00 AM	0	0	\$11,152	\$0
Beaumont	10/26/2009	9:00 AM	0	0	\$11,152	\$0
Jefferson County	8/17/2010	3:00 PM	0	0	\$1,097	\$0
Port Arthur	7/19/2011	6:30 AM	0	0	\$10,637	\$0
Jefferson County	1/25/2012	3:20 PM	0	0	\$1,042	\$0
Jefferson County	3/20/2012	12:00 AM	0	0	\$0	\$0
Bevil Oaks	3/20/2012	12:00 PM	0	0	\$10,421	\$0
Jefferson County	7/13/2012	6:00 AM	0	0	\$2,084	\$0
Jefferson County	1/9/2013	2:58 PM	0	0	\$0	\$0
Beaumont	5/10/2013	4:00 AM	0	0	\$51,353	\$0
Beaumont	10/31/2013	8:15 AM	0	0	\$51,353	\$0
Port Neches	7/18/2014	6:52 AM	0	0	\$0	\$0
Beaumont	3/21/2015	9:35 AM	0	0	\$15,142	\$0
Port Arthur	4/16/2015	9:57 PM	0	0	\$0	\$0
Beaumont	5/12/2015	5:25 PM	0	0	\$10,095	\$0
Beaumont	5/21/2015	1:23 PM	0	0	\$5,047	\$0
Beaumont	5/27/2015	4:07 AM	0	0	\$0	\$0
Jefferson County	6/17/2015	7:29 PM	0	0	\$1,009	\$0
Jefferson County	10/25/2015	8:00 AM	0	0	\$0	\$0

JURISDICTION	DATE	TIME	DEATHS	INJURIES	PROPERTY DAMAGE	CROP DAMAGE
Port Arthur	11/7/2015	6:35 AM	0	0	\$0	\$0
Jefferson County	3/9/2016	7:30 AM	0	0	\$0	\$0
Beaumont	3/10/2016	12:50 AM	0	0	\$0	\$0
Jefferson County	3/30/2016	11:00 PM	0	0	\$0	\$0
Jefferson County	4/18/2016	10:20 AM	0	0	\$0	\$0
Beaumont	5/1/2016	11:30 AM	0	0	\$0	\$0
Jefferson County	5/22/2016	8:13 AM	0	0	\$0	\$0
Beaumont	6/2/2016	11:10 PM	0	0	\$0	\$0

Table 5-5. Summary of Historical Flood Events, 1996-2016<sup>4</sup>

JURISDICTION	NUMBER OF EVENTS	DEATHS	INJURIES	PROPERTY DAMAGE	CROP DAMAGE
Jefferson County	23	1	0	\$587,759	\$0
Beaumont	29	1	0	\$23,440,918	\$0
Bevil Oaks	2	0	0	\$21,960	\$0
China	1	0	0	\$5,676	\$0
Groves	0	0	0	\$0	\$0
Nederland	4	0	0	\$425,073	\$0
Nome	0	0	0	\$0	\$0
Port Arthur	9	0	0	\$141,476	\$0
Port Neches	2	0	0	\$130,033	\$0
TOTAL LOSSES	70	2	0	\$24,75	2,796

Based on the list of historical flood events for the Jefferson County planning area (listed above), including the SETRPC and all participating jurisdictions, 24 events have occurred since the 2011 Plan.

<sup>&</sup>lt;sup>4</sup> Values are in 2016 dollars.

### **Significant Events**

#### Flash Flood on June 7, 2001 – Jefferson County

Tropical Storm Allison caused minor problems along coastal sections of southeast Texas, but eventually resulted in catastrophic flood losses further inland. Wind gusts of 30 to 40 miles per hour (mph) resulted in minor roof damage to less than 10 homes along the coast in Jefferson County between the evening of June 5, 2001 and the early morning hours of June 6, 2001. A 2 foot storm surge resulted in minor beach erosion and portions of Highway 82 between Sabine Pass and Port Arthur to go underwater during the nighttime high tide of June 5 to 6, 2001. The specific flood events that occurred between June 7 and 9, 2001 were a result of the remnants of Tropical Storm Allison, as it meandered across southeast and east Texas. Around ten inches of rain fell in less than 6 hours, resulting in widespread flooding across northern Jefferson County. Nearly 900 homes were damaged, but only ten were classified as destroyed. Roads and bridges also received some damage from the flood waters.

#### Flash Flood on October 29, 2002 – Jefferson County, City of Beaumont

Excessive rainfall in a short period of time impacted portions of Jefferson County on October 29, 2002. Between 6 and 8 inches of rain fell in less than 6 hours. Water filled up to 8 feet deep in underpasses. A woman drove her car into the deep water and drowned. A hospital had the emergency room floor flooded. Over 500 homes in Beaumont had water enter them. Damages were estimated at approximately \$5,000,000.

#### Flood on October 9, 2003 – Jefferson County

Excessive rainfall in a short period of time impacted portions of Jefferson County on October 9, 2003. Around 6 to 8 inches of rain fell in less than 2 hours, causing significant flooding to sections of Beaumont. Water entered homes, vehicles were left stranded on the roads, but fortunately no injuries or deaths were reported. Damages were estimated at approximately \$1,000,000.

## Probability of Future Events

Based on recorded historical occurrences and extent within the Jefferson County planning area including all participating jurisdictions, flooding is highly likely and an event will likely occur within the next year. The SETRPC facilities has no history of flood events and an event impacting the SETRPC is unlikely.

## Vulnerability and Impact

A property's vulnerability to a flood depends on its location and proximity to the floodplain. Structures that lie along banks of a waterway are the most vulnerable and are often repetitive loss structures.

All participating jurisdictions encourage development outside of the floodplain, although there are some critical facilities, homes, and businesses already located in the floodplain. Table 5-6 includes critical facilities in the planning area that are located in the floodplain and are vulnerable to flooding.

JURISDICTION	CRITICAL FACILITIES
Jefferson County	1 School, Port Authority Facility
Beaumont	3 Water District Facilities, 1 Drainage District Facility, 3 Schools
Bevil Oaks	1 Fire Station
China	None
Groves	None
Nederland	None
Nome	None
Port Arthur	1 School
Port Neches	1 School
SETRPC	None

#### Table 5-6. Critical Facilities in the Floodplain by Jurisdiction

Historic loss estimates due to flood are presented in Table 5-7 below. Considering 70 flood events over a 21-year period, frequency is approximately 2 to 3 events every year.

#### Table 5-7. Potential Annualized Losses by Jurisdiction, 1996-2016<sup>5</sup>

JURISDICTION	NUMBER OF EVENTS	PROPERTY & CROP LOSS	ANNUAL LOSS ESTIMATES
Jefferson County	23	\$587,759	\$27,989
Beaumont	29	\$23,440,918	\$1,116,234
Bevil Oaks	2	\$21,960	\$1,046
China	1	\$5,576	\$266
Groves	0	\$0	\$0
Nederland	4	\$425,073	\$20,242
Nome	0	\$0	\$0
Port Arthur	9	\$141,476	\$6,737
Port Neches	2	\$130,033	\$6,192

<sup>&</sup>lt;sup>5</sup> Values are in 2016 dollars.

JURISDICTION	NUMBER OF EVENTS	PROPERTY & CROP LOSS	ANNUAL LOSS ESTIMATES
SETRPC	0	\$0	\$0
Total Losses	70	\$24,752,796	\$1,178,705

The severity of a flooding event varies depending on the relative risk to citizens and structures located within each city. Table 5-8 depicts the level of impact for Jefferson County and each participating city.

#### Table 5-8. Impact by Jurisdiction

JURISDICTION	IMPACT	DESCRIPTION
Jefferson County	Limited	Any injuries or illnesses would be treatable with first aid, with minor quality of life lost. If critical facilities are shut down it would be for 24 hours or less, and it is expected that less than 10 percent of property would be destroyed or damaged in the county.
Beaumont	Major	Injuries or illnesses result in permanent disability. Complete shutdown of critical facilities for at least 2 weeks, and it is expected that more than 25 percent of property would be destroyed or with major damage in the city.
Bevil Oaks	Limited	Any injuries or illnesses would be treatable with first aid, with minor quality of life lost. If critical facilities are shut down it would be for 24 hours or less, and it is expected that less than 10 percent of property would be destroyed or damaged in the city.
China	Limited	Any injuries or illnesses would be treatable with first aid, with minor quality of life lost. If critical facilities are shut down it would be for 24 hours or less, and it is expected that less than 10 percent of property would be destroyed or damaged in the city.
Groves	Limited	Any injuries or illnesses would be treatable with first aid, with minor quality of life lost. If critical facilities are shut down it would be for 24 hours or less, and it is expected that less than 10 percent of property would be destroyed or damaged in the city.
Nederland	Limited	Any injuries or illnesses would be treatable with first aid, with minor quality of life lost. If critical facilities are shut down it would be for 24 hours or less, and it is expected that less than 10 percent of property would be destroyed or damaged in the city.
Nome	Limited	Any injuries or illnesses would be treatable with first aid, with minor quality of life lost. If critical facilities are shut down it would be for 24 hours or less, and it is expected that less than 10 percent of property would be destroyed or damaged in the city.
Port Arthur	Limited	Any injuries or illnesses would be treatable with first aid, with minor quality of life lost. If critical facilities are shut down it would be for 24 hours or less, and it is expected that less than 10 percent of property would be destroyed or damaged in the city.

JURISDICTION	IMPACT	DESCRIPTION
Port Neches	Limited	Any injuries or illnesses would be treatable with first aid, with minor quality of life lost. If critical facilities are shut down it would be for 24 hours or less, and it is expected that less than 10 percent of property would be destroyed or damaged in the city.
SETRPC	Limited	Facilities are unlikely to be impacted. If critical facilities are shut down it would be for 24 hours or less, and it is expected that less than 10 percent of property would be destroyed or damaged at the SETRPC.

### Assessment of Impacts

Flooding is the deadliest natural disaster that occurs in the U.S. each year, and it poses a constant and significant threat to the health and safety of the people in the planning area. Impacts to the planning area can include:

- Recreation activities at Sabine Lake and Sea Rim State Park may be unavailable and tourism can be unappealing for years following a large flood event, devastating directly related local businesses and negatively impacting economic recovery.
- The McFaddin National Wildlife Refuge may suffer significant wildlife mortality during and following a flood due to damaged or destroyed ecosystems and water contamination.
- The Port Arthur-Orange Bridge may be damaged or inaccessible, cutting off critical emergency services to Port Arthur.
- Flood-related rescues may be necessary at swift water and low water crossings or in flooded neighborhoods where roads have become impassable, placing first responders in harm's way.
- Evacuations may be required for entire neighborhoods because of rising floodwaters, further taxing limited response capabilities and increasing sheltering needs for displaced residents.
- Health risks and threats to residents are elevated after the flood waters have receded due to contaminated flood waters (untreated sewage and hazardous chemicals) and mold growth typical in flooded buildings and homes.
- Significant flood events often result in widespread power outages, increasing the risk to more vulnerable portions of the population who rely on power for health and/or life safety.
- Extended power outage can result in an increase in structure fires and/or carbon monoxide poisoning, as individuals attempt to cook or heat their home with alternate, unsafe cooking or heating devices, such as grills.
- Floods can destroy or make residential structures uninhabitable, requiring shelter or relocation of residents in the aftermath of the event.
- First responders are exposed to downed power lines, contaminated and potentially unstable debris, hazardous materials, and generally unsafe conditions, elevating the risk of injury to first responders and potentially diminishing emergency response capabilities.
- Emergency operations and services may be significantly impacted due to damaged facilities.
- Significant flooding can result in the inability of emergency response vehicles to access areas of the community.
- Critical staff may suffer personal losses or be otherwise impacted by a flood event and unable to report for duty, limiting response capabilities.

- City or County departments may be flooded, delaying response and recovery efforts for the entire community.
- Private sector entities that the City and its residents rely on, such as utility providers, financial institutions, and medical care providers may not be fully operational and may require assistance from neighboring communities until full services can be restored.
- Damage to infrastructure may slow economic recovery since repairs may be extensive and lengthy.
- Some businesses not directly damaged by the flood may be negatively impacted while utilities are being restored or water recedes, further slowing economic recovery.
- When the community is affected by significant property damage it is anticipated that funding would be required for infrastructure repair and restoration, temporary services and facilities, overtime pay for responders, as well as normal day-to-day operating expenses.
- Displaced residents may not be able to immediately return to work, further slowing economic recovery.
- Residential structures substantially damaged by a flood may not be rebuilt for years and uninsured or underinsured residential structures may never be rebuilt, reducing the tax base for the community.
- Large floods may result in a dramatic population fluctuation, as people are unable to return to their homes or jobs and must seek shelter and/or work outside of the affected area.
- Businesses that are uninsured or underinsured may have difficulty reopening, which results in a net loss of jobs for the community and a potential increase in the unemployment rate.
- Flooding may cause significant disruptions of clean water and sewer services, elevating health risks and delaying recovery efforts.
- The psycho-social effects on flood victims and their families can traumatize them for long periods of time, creating long term increases in medical treatment and services.
- Extensive or repetitive flooding can lead to decreases in property value for the affected community.
- Flood poses a potential catastrophic risk to annual and perennial crop production and overall crop quality, leading to higher food costs.
- Flood related declines in production may lead to an increase in unemployment.
- Large floods may result in loss of livestock, potential increased livestock mortality due to stress and water borne disease, and increased cost for feed.

The overall extent of damages caused by floods is dependent on the extent, depth, and duration of flooding, and the velocities of flows in the flooded areas. The level of preparedness and pre-event planning done by government, businesses, and citizens will contribute to the overall economic and financial conditions in the aftermath of a flood event.

# National Flood Insurance Program (NFIP) Participation

Flood insurance offered through the National Flood Insurance Program (NFIP) is the best way for home and business owners to protect themselves financially against the flood hazard. All of the jurisdictions located in Jefferson County participate in the NFIP. The SETRPC is not an eligible entity for participation in the NFIP and is not located in a special flood hazard area.

As an additional indicator of floodplain management responsibility, communities may choose to participate in FEMA's Community Rating System (CRS). This is an incentive-based program that allows communities to undertake flood mitigation activities that go beyond NFIP requirements. Currently, 3 of the communities in Jefferson County participate in CRS, including Beaumont (class 7), Bevil Oaks (class 7), and Port Arthur (class 9). The remaining jurisdictions in the planning area understand the value of participation in this program and have identified this as a goal and objective of the Plan that was discussed during Planning Team meetings.

Some participating jurisdictions in the NFIP currently have in place minimum NFIP standards for new construction and substantial improvements of structures; these jurisdictions include: China, Groves, Nome, and Port Neches. The remaining jurisdictions have adopted higher regulatory NFIP standards to limit floodplain development including: Jefferson County, Beaumont, Bevil Oaks, Nederland, and Port Arthur.

The flood hazard areas throughout Jefferson County are subject to periodic inundation, which may result in loss of life and property, health and safety hazards, disruption of commerce and governmental services, and extraordinary public expenditures for flood protection and relief, all of which adversely affect public safety.

These flood losses are created by the cumulative effect of obstructions in floodplains which cause an increase in flood heights and velocities, and by the occupancy of flood hazard areas by uses vulnerable to floods and hazardous to other lands because they are inadequately elevated, flood-proofed, or otherwise protected from flood damage. Mitigation actions are included to address flood maintenance issues as well, including routinely clearing debris from roadside ditches and bridges, and expanding drainage culverts and storm water structures to more adequately convey flood waters.

It is the purpose of Jefferson County and NFIP jurisdictions participating in the Hazard Mitigation plan to continue to promote the public health, safety, and general welfare by minimizing public and private losses due to flood conditions in specific areas. Each of the NFIP participating jurisdictions in the Plan are guided by their local Flood Damage Prevention Ordinance. These communities will continue to comply with NFIP requirements through their local permitting, inspection, and record-keeping requirements for new and substantially developed construction. Furthermore, the NFIP program for each of the participating jurisdictions designed to:

- Protect human life and health;
- Minimize expenditure of public money for costly flood control projects;
- Minimize the need for rescue and relief efforts associated with flooding and generally undertaken at the expense of the general public;
- Minimize prolonged business interruptions;
- Minimize damage to public facilities and utilities such as water and gas mains, electric, telephone and sewer lines, streets, and bridges located in floodplains;
- Help maintain a stable tax base by providing for the sound use and development of flood-prone areas in such a manner as to minimize future flood blight areas; and
- Ensure that potential buyers are notified that property is in a flood area.

In order to accomplish these tasks, Jefferson County and participating NFIP jurisdictions seek to follow the following guidelines to achieve flood mitigation:

- Restrict or prohibit uses that are dangerous to health, safety, or property in times of flood, such as filling or dumping, that may cause excessive increases in flood heights or velocities;
- Require that uses vulnerable to floods, including facilities, which serve such uses, be protected against flood damage at the time of initial construction, as a method of reducing flood losses;
- Control the alteration of natural floodplains, stream channels, and natural protective barriers, which are involved in the accommodation of floodwaters;
- Control filling, grading, dredging and other development, which may increase flood damage; and
- Prevent or regulate the construction of flood barriers which will unnaturally divert floodwaters or which may increase flood hazards to other lands.

# NFIP Compliance and Maintenance

As mentioned, Jefferson County and participating jurisdictions have developed mitigation actions that relate to either NFIP maintenance or compliance. Compliance and maintenance actions can be found in Section 18.

Flooding was identified by the majority of the communities as a high risk hazard during hazard ranking activities at the Risk Assessment Workshop. As a result, many of the mitigation actions were developed with flood mitigation in mind. A majority of these flood actions address compliance with the NFIP and implementing flood awareness programs. County-wide, communities recognize the need and are working towards adopting additional higher NFIP regulatory standards to further minimize flood risk in their community. Smaller no-growth communities that typically do not have personnel or funds to implement more stringent NFIP compliance measures are focusing on NFIP public awareness activities. This includes promoting the availability of flood insurance by placing NFIP brochures and flyers in public libraries or public meeting places.

# **Repetitive Loss**

The Severe Repetitive Loss (SRL) Grant Program under FEMA provides federal funding to assist states and communities in implementing mitigation measures to reduce or eliminate the long-term risk of flood damage to SRL residential structures insured under the NFIP. The Texas Water Development Board (TWDB) administers the SRL Grant Program for the State of Texas.

Severe Repetitive Loss properties are defined as residential properties that are:

- Covered under the NFIP and have at least 4 flood related damage claim payments (building and contents) over \$5,000 each, and the cumulative amount of such claims payments exceed \$20,000; or
- At least 2 separate claim payments (building payments only) have been made with the cumulative amount of the building portion of such claims exceeding the market value of the building.

In either scenario, at least 2 of the referenced claims must have occurred within any 10-year period, and must be greater than 10 days apart.<sup>6</sup> Table 5-9 shows repetitive loss and severe repetitive loss properties for Jefferson County and all participating jurisdictions.

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Jefferson County	0177501	Yes	Other-Nonres	2	\$18,447	-
Jefferson County	0016017	No	Single Fmly	5	\$84,858	VU
Jefferson County	0071531	No	Single Fmly	2	\$98,851	-
Jefferson County	0013029	Yes	Single Fmly	4	\$104,045	-
Jefferson County	0004178	SDF	Single Fmly	10	\$423,856	V
Jefferson County	0026644	Yes	Single Fmly	3	\$7,147	-
Jefferson County	0182118	Yes	Single Fmly	2	\$19,961	-
Jefferson County	0158314	No	Single Fmly	3	\$22,502	-
Jefferson County	0001457	No	Single Fmly	6	\$120,317	VU
Jefferson County	0180869	No	Single Fmly	2	\$118,698	-
Jefferson County	0137536	No	Single Fmly	2	\$13,204	-
Jefferson County	0241281	Yes	Single Fmly	2	\$45,054	-
Jefferson County	0017351	No	Single Fmly	4	\$43,252	PU
Jefferson County	0163518	No	Single Fmly	2	\$106,277	-
Jefferson County	0041486	No	Single Fmly	4	\$27,869	-
Jefferson County	0067919	No	Single Fmly	4	\$17,625	-
Jefferson County	0173735	No	Single Fmly	2	\$8,280	-
Jefferson County	0148780	Yes	Single Fmly	3	\$59,135	-
Jefferson County	0108329	No	Single Fmly	3	\$68,125	-
Jefferson County	0153950	Yes	Single Fmly	3	\$95,317	-
Jefferson County	0108430	SDF	Single Fmly	6	\$198,046	V
Jefferson County	0153949	Yes	Single Fmly	3	\$99,955	-
Jefferson County	0108476	No	Single Fmly	2	\$6,354	-

Table 5-9. Repetitive Loss and Severe Repetitive Loss Properties

<sup>6</sup> Source: Texas Water Development Board

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Jefferson County	0152782	No	Single Fmly	2	\$44,617	-
Jefferson County	0108527	Yes	Single Fmly	3	\$37,792	-
Jefferson County	0120084	No	Single Fmly	2	\$4,976	-
Jefferson County	0182072	No	Single Fmly	2	\$125,385	-
Jefferson County	0115988	No	Single Fmly	3	\$94,047	-
Jefferson County	0179239	No	Single Fmly	2	\$55,738	-
Jefferson County	0164706	No	Single Fmly	2	\$92,206	-
Jefferson County	0185403	No	Single Fmly	2	\$64,634	-
Jefferson County	0048506	No	Other-Nonres	2	\$6,542	-
Jefferson County	0179193	No	Other-Nonres	2	\$118,800	-
Jefferson County	0012918	No	Single Fmly	5	\$49,221	-
Jefferson County	0083532	No	Single Fmly	2	\$8,428	-
Jefferson County	0250441	Yes	Single Fmly	2	\$34,987	-
Jefferson County	0044550	No	Single Fmly	3	\$12,788	-
Jefferson County	0013026	No	Single Fmly	11	\$294,171	VU
Jefferson County	0182115	Yes	Single Fmly	2	\$66,076	-
Jefferson County	0044740	No	Single Fmly	2	\$12,748	-
Jefferson County	0045848	Yes	Single Fmly	6	\$75,679	-
Jefferson County	0001528	No	Single Fmly	18	\$158,394	VU
Jefferson County	0004169	No	Single Fmly	10	\$59,674	VU
Jefferson County	0042051	No	Single Fmly	2	\$5,826	-
Jefferson County	0003757	SDF	Single Fmly	8	\$156,202	V
Jefferson County	0017283	No	Assmd Condo	4	\$70,716	PU
Jefferson County	0017396	No	Single Fmly	4	\$22,327	-
Jefferson County	0173644	No	Single Fmly	2	\$56,737	-
Jefferson County	0180156	No	Single Fmly	2	\$18,766	-
Jefferson County	0083543	No	Single Fmly	5	\$148,547	VU
Jefferson County	0120315	No	Single Fmly	2	\$28,298	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Jefferson County	0153952	SDF	Single Fmly	2	\$380,597	V
Jefferson County	0001952	No	Single Fmly	8	\$66,827	VU
Jefferson County	0125137	No	Single Fmly	2	\$3,054	-
Jefferson County	0043339	No	Single Fmly	2	\$17,735	-
Jefferson County	0179752	Yes	Single Fmly	2	\$49,457	-
Jefferson County	0045117	No	Single Fmly	5	\$99,614	-
Jefferson County	0070472	No	Other-Nonres	3	\$60,171	-
Jefferson County	0128395	No	Single Fmly	3	\$45,981	-
Jefferson County	0165546	No	Single Fmly	2	\$40,529	-
Jefferson County	0108434	Yes	Single Fmly	5	\$58,179	-
Jefferson County	0108777	No	Single Fmly	2	\$14,137	-
Jefferson County	0163519	No	Single Fmly	2	\$96,223	-
Jefferson County	0173721	SDF	Single Fmly	4	\$142,389	V
Jefferson County	0002601	Yes	Single Fmly	5	\$87,823	-
Jefferson County	0002600	Yes	Single Fmly	4	\$121,623	-
Jefferson County	0041009	No	Single Fmly	4	\$7,203	-
Jefferson County	0071449	No	Single Fmly	2	\$5,032	-
Jefferson County	0042675	No	Single Fmly	3	\$9,768	-
Jefferson County	0004786	No	Single Fmly	6	\$33,939	-
Jefferson County	0113811	No	Single Fmly	3	\$272,767	PU
Jefferson County	0169147	Yes	Single Fmly	3	\$118,177	-
Jefferson County	0168573	Yes	Other Resid	3	\$112,610	-
Jefferson County	0004107	No	Single Fmly	4	\$21,688	-
Jefferson County	0073580	No	Single Fmly	4	\$87,941	VU
Jefferson County	0088112	No	Single Fmly	4	\$72,447	-
Jefferson County	0002864	No	Single Fmly	6	\$166,606	VU
Jefferson County	0165197	No	Single Fmly	2	\$5,852	-
Jefferson County	0044800	No	Single Fmly	2	\$7,410	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Jefferson County	0002678	SDF	Single Fmly	5	\$78,867	V
Jefferson County	0157861	No	Single Fmly	2	\$30,689	-
Jefferson County	0186703	Yes	Other-Nonres	4	\$153,427	VN
Jefferson County	0038223	No	Single Fmly	3	\$21,787	-
Jefferson County	0080468	SDF	Single Fmly	6	\$148,689	V
Jefferson County	0039714	No	Single Fmly	7	\$40,950	VU
Jefferson County	0108360	Yes	Single Fmly	6	\$72,399	-
Jefferson County	0178162	Yes	Single Fmly	2	\$50,732	-
Jefferson County	0122037	Yes	Single Fmly	2	\$41,202	-
Jefferson County	0108238	No	Single Fmly	4	\$35,291	-
Jefferson County	0108230	No	Single Fmly	2	\$36,963	-
Jefferson County	0214533	No	Single Fmly	2	\$45,558	-
Jefferson County	0160070	Yes	Single Fmly	2	\$75,551	-
Jefferson County	0026342	No	Single Fmly	2	\$26,231	-
Jefferson County	0025462	No	Single Fmly	2	\$4,420	-
Jefferson County	0044016	No	Single Fmly	5	\$153,408	MVU
Jefferson County	0040340	No	Single Fmly	8	\$233,738	MVU
Jefferson County	0040116	No	Single Fmly	4	\$137,523	-
Jefferson County	0026029	No	Assmd Condo	3	\$65,173	-
Jefferson County	0046295	No	Single Fmly	4	\$60,113	-
Jefferson County	0114070	No	Single Fmly	3	\$233,637	-
Jefferson County	0117301	No	Other-Nonres	2	\$6,344	-
Jefferson County	0067979	No	Single Fmly	2	\$12,305	-
Jefferson County	0041990	No	Single Fmly	5	\$54,108	-
Jefferson County	0048831	No	Single Fmly	3	\$152,011	-
Jefferson County	0186075	No	Single Fmly	2	\$131,591	-
Jefferson County	0180352	No	Single Fmly	2	\$191,252	-
Jefferson County	0183152	No	Assmd Condo	2	\$449,097	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Jefferson County	0026121	No	Single Fmly	6	\$118,247	MVU
Jefferson County	0121931	No	Single Fmly	2	\$19,911	-
Jefferson County	0179733	No	Single Fmly	2	\$68,076	-
Beaumont	0070066	No	Single Fmly	2	\$2,560	-
Beaumont	0182124	No	Single Fmly	2	\$11,266	-
Beaumont	0043492	No	Single Fmly	12	\$155,860	VU
Beaumont	0182135	No	Single Fmly	2	\$33,190	-
Beaumont	0038907	No	Single Fmly	2	\$12,491	-
Beaumont	0128504	No	Single Fmly	3	\$14,526	-
Beaumont	0043346	No	Single Fmly	3	\$15,097	-
Beaumont	0089788	No	2-4 Family	2	\$5,703	-
Beaumont	0115540	No	Single Fmly	5	\$55,255	-
Beaumont	0182764	Yes	Single Fmly	2	\$12,261	-
Beaumont	0017381	No	Single Fmly	4	\$11,309	-
Beaumont	0157279	Yes	Single Fmly	4	\$125,138	V
Beaumont	0067939	No	Single Fmly	2	\$6,479	-
Beaumont	0122426	No	Single Fmly	2	\$40,740	-
Beaumont	0067960	No	Single Fmly	4	\$36,206	-
Beaumont	0049134	No	Single Fmly	2	\$15,120	-
Beaumont	0002569	No	Assmd Condo	3	\$56,493	-
Beaumont	0002588	No	2-4 Family	3	\$14,187	-
Beaumont	0116431	SDF	Other-Nonres	5	\$300,016	VN
Beaumont	0042980	No	Single Fmly	5	\$39,482	-
Beaumont	0003237	No	Single Fmly	4	\$57,354	-
Beaumont	0067968	No	Single Fmly	2	\$16,138	-
Beaumont	0067969	No	Single Fmly	5	\$66,118	-
Beaumont	0125431	Yes	Single Fmly	2	\$35,184	-
Beaumont	0002599	SDF	Single Fmly	6	\$127,564	V

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Beaumont	0125433	Yes	Single Fmly	2	\$39,856	-
Beaumont	0002698	No	Other-Nonres	26	\$1,464,159	VNU
Beaumont	0082203	No	Single Fmly	3	\$10,138	-
Beaumont	0173620	Yes	Other-Nonres	2	\$15,453	-
Beaumont	0005811	Yes	Single Fmly	5	\$60,891	-
Beaumont	0026764	No	Single Fmly	3	\$29,717	-
Beaumont	0045178	No	Single Fmly	2	\$7,268	-
Beaumont	0121694	No	Single Fmly	2	\$11,687	-
Beaumont	0151747	No	Single Fmly	3	\$57,066	-
Beaumont	0002580	No	Single Fmly	2	\$8,304	-
Beaumont	0002575	No	Assmd Condo	3	\$27,883	-
Beaumont	0002585	No	Assmd Condo	2	\$21,922	-
Beaumont	0002590	No	2-4 Family	3	\$15,041	-
Beaumont	0002581	No	2-4 Family	2	\$7,227	-
Beaumont	0002589	No	2-4 Family	2	\$7,723	-
Beaumont	0002574	No	2-4 Family	2	\$7,240	-
Beaumont	0002583	No	2-4 Family	3	\$13,239	-
Beaumont	0002582	No	2-4 Family	2	\$7,027	-
Beaumont	0002576	No	2-4 Family	3	\$14,512	-
Beaumont	0002568	No	2-4 Family	3	\$14,083	-
Beaumont	0002596	No	Assmd Condo	3	\$58,348	-
Beaumont	0002592	No	Assmd Condo	2	\$21,866	-
Beaumont	0002593	No	2-4 Family	2	\$7,890	-
Beaumont	0002594	No	2-4 Family	2	\$10,722	-
Beaumont	0002595	Yes	2-4 Family	4	\$82,377	-
Beaumont	0122098	No	Single Fmly	2	\$29,978	-
Beaumont	0113248	SDF	Single Fmly	4	\$103,457	V
Beaumont	0120023	Yes	Single Fmly	2	\$49,718	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Beaumont	0004631	Yes	Single Fmly	4	\$60,627	-
Beaumont	0067978	SDF	Single Fmly	7	\$100,986	V
Beaumont	0122261	No	Single Fmly	2	\$9,172	-
Beaumont	0121260	No	Single Fmly	2	\$15,797	-
Beaumont	0046922	No	Single Fmly	7	\$39,355	-
Beaumont	0122366	Yes	Single Fmly	4	\$89,533	V
Beaumont	0132737	No	Single Fmly	2	\$12,416	-
Beaumont	0114740	No	Single Fmly	3	\$8,656	-
Beaumont	0120004	Yes	Single Fmly	2	\$26,303	-
Beaumont	0114774	Yes	Other-Nonres	2	\$16,534	-
Beaumont	0071539	Yes	Single Fmly	3	\$17,200	-
Beaumont	0183470	Yes	Single Fmly	2	\$3,887	-
Beaumont	0179716	Yes	Single Fmly	2	\$15,556	-
Beaumont	0120093	No	Single Fmly	2	\$7,223	-
Beaumont	0126222	No	Other-Nonres	2	\$47,645	-
Beaumont	0067949	No	Single Fmly	3	\$9,318	-
Beaumont	0240104	Yes	Single Fmly	2	\$111,011	-
Beaumont	0004650	SDF	Single Fmly	7	\$67,213	V
Beaumont	0046556	No	Single Fmly	5	\$49,800	-
Beaumont	0073389	No	Single Fmly	5	\$67,086	-
Beaumont	0128354	No	Single Fmly	3	\$9,027	-
Beaumont	0113792	Yes	Single Fmly	3	\$24,978	-
Beaumont	0067981	SDF	Single Fmly	5	\$54,053	V
Beaumont	0088111	No	Single Fmly	3	\$23,743	-
Beaumont	0166575	Yes	Single Fmly	2	\$6,595	-
Beaumont	0039928	No	Single Fmly	4	\$69,994	-
Beaumont	0121676	No	Single Fmly	2	\$5,373	-
Beaumont	0115539	No	Single Fmly	3	\$68,729	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Beaumont	0048616	No	Single Fmly	2	\$9,678	-
Beaumont	0003398	No	Single Fmly	3	\$24,920	-
Beaumont	0152109	No	Other-Nonres	2	\$32,421	-
Beaumont	0067959	No	Single Fmly	4	\$42,663	-
Beaumont	0083533	No	Single Fmly	2	\$19,334	-
Beaumont	0044211	No	Other-Nonres	2	\$2,847	-
Beaumont	0114917	SDF	Single Fmly	6	\$49,013	V
Beaumont	0168158	No	Single Fmly	2	\$26,616	-
Beaumont	0113804	No	Single Fmly	3	\$32,531	-
Beaumont	0067962	No	Single Fmly	2	\$4,805	-
Beaumont	0158046	No	Single Fmly	2	\$56,874	-
Beaumont	0127398	No	Single Fmly	2	\$13,922	-
Beaumont	0112832	SDF	Single Fmly	4	\$78,260	V
Beaumont	0002520	No	Single Fmly	3	\$6,077	-
Beaumont	0120305	No	Single Fmly	2	\$13,702	-
Beaumont	0237955	Yes	Single Fmly	2	\$26,324	-
Beaumont	0237956	Yes	Other Resid	2	\$12,242	-
Beaumont	0036099	No	Single Fmly	2	\$4,940	-
Beaumont	0116577	No	Other-Nonres	3	\$115,136	-
Beaumont	0067937	No	Single Fmly	6	\$41,590	VU
Beaumont	0121499	Yes	Other-Nonres	2	\$174,474	-
Beaumont	0002422	No	Single Fmly	11	\$69,927	VU
Beaumont	0067971	No	Single Fmly	6	\$51,994	-
Beaumont	0121689	No	Single Fmly	2	\$5,726	-
Beaumont	0157223	No	Single Fmly	2	\$87,574	-
Beaumont	0173160	No	Single Fmly	2	\$12,283	-
Beaumont	0181178	No	Single Fmly	2	\$7,898	-
Beaumont	0037562	No	Single Fmly	2	\$49,987	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Beaumont	0157056	No	Single Fmly	4	\$49,085	VU
Beaumont	0247942	Yes	Single Fmly	2	\$8,217	-
Beaumont	0239937	No	Other-Nonres	2	\$7,310	-
Beaumont	0036557	No	Single Fmly	2	\$5,332	-
Beaumont	0044946	No	Other-Nonres	2	\$28,596	-
Beaumont	0043814	No	Single Fmly	3	\$5,461	-
Beaumont	0067916	No	Single Fmly	5	\$51,199	-
Beaumont	0067952	No	Single Fmly	2	\$17,093	-
Beaumont	0013120	No	Single Fmly	2	\$6,402	-
Beaumont	0070494	No	Other-Nonres	6	\$60,752	VNU
Beaumont	0113806	No	Single Fmly	3	\$9,515	-
Beaumont	0003192	No	Single Fmly	3	\$56,695	-
Beaumont	0235028	No	Single Fmly	2	\$4,893	-
Beaumont	0045522	No	Single Fmly	7	\$64,851	VU
Beaumont	0070115	No	Single Fmly	2	\$28,288	-
Beaumont	0069904	No	Other-Nonres	2	\$63,383	-
Beaumont	0115894	No	Single Fmly	3	\$26,352	-
Beaumont	0043374	No	Single Fmly	2	\$3,173	-
Beaumont	0043197	No	Other-Nonres	3	\$16,137	-
Beaumont	0002673	No	Other-Nonres	5	\$24,201	PNU
Beaumont	0045889	No	Single Fmly	3	\$17,937	-
Beaumont	0047508	No	Assmd Condo	2	\$14,112	-
Beaumont	0044866	SDF	Single Fmly	5	\$65,046	V
Beaumont	0070303	No	Single Fmly	6	\$156,220	VU
Beaumont	0069941	No	Single Fmly	4	\$15,654	-
Beaumont	0048673	No	Single Fmly	3	\$46,442	-
Beaumont	0043698	No	Single Fmly	4	\$11,190	-
Beaumont	0013037	No	Single Fmly	3	\$12,556	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Beaumont	0044455	No	Single Fmly	4	\$100,040	-
Beaumont	0044919	No	Single Fmly	2	\$11,501	-
Beaumont	0125434	No	Single Fmly	3	\$24,944	-
Beaumont	0044382	No	Single Fmly	3	\$74,960	-
Beaumont	0067945	No	Single Fmly	2	\$7,393	-
Beaumont	0067980	No	Single Fmly	2	\$5,017	-
Beaumont	0043598	No	Single Fmly	3	\$101,464	VU
Beaumont	0237813	Yes	Single Fmly	2	\$10,543	-
Beaumont	0044997	No	Single Fmly	3	\$35,678	-
Beaumont	0173890	No	Single Fmly	3	\$11,880	-
Beaumont	0120003	No	Single Fmly	2	\$28,542	-
Beaumont	0086078	No	Single Fmly	3	\$68,338	-
Beaumont	0003353	No	Single Fmly	3	\$10,920	-
Beaumont	0120038	Yes	Single Fmly	4	\$40,735	-
Beaumont	0121912	No	Single Fmly	2	\$37,151	-
Beaumont	0162445	No	Single Fmly	2	\$14,806	-
Beaumont	0122004	Yes	Single Fmly	2	\$16,815	-
Beaumont	0026335	No	Other-Nonres	3	\$21,253	-
Beaumont	0088878	Yes	Single Fmly	2	\$10,648	-
Beaumont	0067970	No	Single Fmly	2	\$9,699	-
Beaumont	0120307	No	Single Fmly	2	\$20,326	-
Beaumont	0044573	No	Single Fmly	2	\$5,231	-
Beaumont	0237895	No	Single Fmly	2	\$13,111	-
Beaumont	0003413	No	Single Fmly	4	\$49,101	VU
Beaumont	0008406	No	Single Fmly	7	\$50,400	-
Beaumont	0067934	No	Single Fmly	2	\$7,265	-
Beaumont	0002559	Yes	Single Fmly	3	\$31,088	-
Beaumont	0121435	No	Single Fmly	2	\$10,639	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Beaumont	0084605	No	Single Fmly	2	\$40,252	-
Beaumont	0067947	No	Single Fmly	3	\$4,760	-
Beaumont	0084817	No	Single Fmly	2	\$12,875	-
Beaumont	0121914	No	Single Fmly	2	\$92,394	-
Beaumont	0044439	No	Single Fmly	5	\$101,539	VU
Beaumont	0067932	No	Single Fmly	2	\$13,353	-
Beaumont	0043922	SDF	Other-Nonres	6	\$247,669	VN
Beaumont	0186070	No	Other-Nonres	2	\$95,623	-
Beaumont	0003912	No	Other-Nonres	8	\$528,406	VNU
Beaumont	0122196	No	Other-Nonres	5	\$527,383	VNU
Beaumont	0115972	No	Other-Nonres	6	\$405,225	VNU
Beaumont	0002782	No	Other-Nonres	3	\$5,790	-
Beaumont	0113801	Yes	Other-Nonres	3	\$365,974	-
Beaumont	0043228	No	Single Fmly	7	\$100,269	VU
Beaumont	0046414	No	Single Fmly	2	\$5 <i>,</i> 809	-
Beaumont	0025461	No	Single Fmly	6	\$81,486	-
Beaumont	0042478	No	Single Fmly	2	\$10,791	-
Beaumont	0025410	No	Single Fmly	3	\$14,121	-
Beaumont	0004259	No	Other-Nonres	3	\$158,835	-
Beaumont	0005654	No	Single Fmly	4	\$32,680	-
Beaumont	0025298	No	Other-Nonres	6	\$37,457	-
Beaumont	0004293	No	Single Fmly	3	\$36,052	-
Beaumont	0046461	No	Single Fmly	5	\$92,986	MVU
Beaumont	0002744	No	Single Fmly	4	\$22,586	-
Beaumont	0014106	No	Assmd Condo	11	\$100,300	MVU
Beaumont	0048689	No	Other-Nonres	2	\$18,649	-
Beaumont	0067907	No	Single Fmly	2	\$14,745	-
Beaumont	0121005	No	Single Fmly	3	\$55,544	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Beaumont	0067909	No	Single Fmly	2	\$12,263	-
Beaumont	0067946	No	Single Fmly	3	\$37,077	-
Beaumont	0113793	No	Single Fmly	3	\$103,254	MVU
Beaumont	0157055	No	Single Fmly	4	\$45,132	-
Beaumont	0073385	No	Single Fmly	2	\$22,968	-
Beaumont	0116242	No	Single Fmly	4	\$32,899	-
Beaumont	0004779	No	Single Fmly	6	\$202,733	-
Beaumont	0121461	No	Single Fmly	4	\$37,076	-
Beaumont	0173760	No	Single Fmly	3	\$10,714	-
Beaumont	0080411	No	Single Fmly	3	\$4,728	-
Beaumont	0160048	No	Single Fmly	2	\$4,277	-
Beaumont	0005575	No	Single Fmly	12	\$66,171	-
Beaumont	0005574	No	Single Fmly	11	\$56,840	MVU
Beaumont	0005557	No	Single Fmly	10	\$61,424	-
Beaumont	0001250	No	Single Fmly	13	\$100,603	-
Beaumont	0007187	No	Other Resid	16	\$276,817	MVU
Beaumont	0007186	No	Other Resid	12	\$556,517	MVU
Beaumont	0007185	No	Other Resid	12	\$273,182	MVU
Beaumont	0069905	No	Single Fmly	2	\$10,082	-
Beaumont	0045245	No	Single Fmly	2	\$6,094	-
Beaumont	0184310	No	Other-Nonres	2	\$511,817	-
Beaumont	0088109	No	Other-Nonres	2	\$13,306	-
Beaumont	0122245	No	Other-Nonres	2	\$15,502	-
Beaumont	0017628	Yes	Other-Nonres	7	\$157,483	-
Beaumont	0001884	Yes	Single Fmly	4	\$44,018	-
Beaumont	0025380	No	Single Fmly	2	\$2,724	-
Beaumont	0120310	Yes	Single Fmly	2	\$55,601	-
Beaumont	0003781	No	Single Fmly	6	\$110,785	MVU

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Beaumont	0114644	No	Single Fmly	3	\$25,180	-
Beaumont	0067935	Yes	Single Fmly	3	\$29,184	-
Beaumont	0067957	No	Single Fmly	2	\$7,176	-
Beaumont	0043903	No	Single Fmly	2	\$12,548	-
Beaumont	0080717	Yes	Single Fmly	3	\$87,052	-
Beaumont	0122216	Yes	Single Fmly	2	\$41,954	-
Beaumont	0044574	No	Single Fmly	5	\$53,646	MVU
Beaumont	0151750	No	Single Fmly	2	\$34,330	-
Beaumont	0168670	No	Single Fmly	2	\$29,471	-
Beaumont	0167875	No	Single Fmly	2	\$20,334	-
Beaumont	0120321	No	Single Fmly	2	\$30,669	-
Beaumont	0120376	No	Single Fmly	2	\$12,790	-
Beaumont	0120298	No	Single Fmly	2	\$10,098	-
Beaumont	0173860	No	Single Fmly	2	\$46,961	-
Beaumont	0115537	No	Single Fmly	3	\$46,623	-
Beaumont	0116239	Yes	Single Fmly	3	\$43,080	-
Beaumont	0067914	No	Single Fmly	4	\$97,317	MVU
Beaumont	0120031	Yes	Single Fmly	2	\$34,800	-
Beaumont	0121437	No	Single Fmly	2	\$79,788	-
Beaumont	0067961	Yes	Single Fmly	2	\$2,801	-
Beaumont	0120309	No	Single Fmly	2	\$29,623	-
Beaumont	0122579	No	Other-Nonres	2	\$64,621	-
Beaumont	0121009	No	Single Fmly	2	\$47,570	-
Beaumont	0067902	No	Other-Nonres	2	\$7,040	-
Beaumont	0043261	No	Other-Nonres	2	\$4,387	-
Beaumont	0025267	Yes	Single Fmly	5	\$52,992	-
Beaumont	0004755	No	Other-Nonres	9	\$32,967	-
Beaumont	0004100	Yes	Other-Nonres	25	\$945,255	-
COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
-------------------	----------	---------	------------------	--------	------------	------------------
Beaumont	0122596	Yes	Other-Nonres	3	\$285,356	-
Beaumont	0070422	No	Single Fmly	3	\$26,138	-
Beaumont	0121465	No	Single Fmly	3	\$26,395	-
Beaumont	0121294	No	Single Fmly	2	\$46,399	-
Beaumont	0067965	No	Single Fmly	2	\$12,776	-
Beaumont	0067953	No	Single Fmly	3	\$101,925	-
Beaumont	0003442	Yes	Single Fmly	4	\$269,055	MV
Beaumont	0003991	No	Single Fmly	4	\$168,689	MVU
Beaumont	0005799	No	Single Fmly	5	\$296,966	MVU
Beaumont	0002563	No	Single Fmly	4	\$83,419	MVU
Beaumont	0013090	No	Single Fmly	4	\$6,583	-
Beaumont	0026742	No	Single Fmly	2	\$3,119	-
Beaumont	0124952	No	Single Fmly	2	\$31,328	-
Beaumont	0047488	Yes	Single Fmly	3	\$19,140	-
Beaumont	0003021	No	Single Fmly	7	\$164,791	MVU
Beaumont	0046268	No	Single Fmly	2	\$14,195	-
Beaumont	0067973	No	Single Fmly	4	\$28,698	-
Beaumont	0043305	No	Single Fmly	4	\$126,787	MVU
Beaumont	0067974	No	Single Fmly	4	\$29,586	-
Beaumont	0067975	No	Single Fmly	3	\$9,303	-
Beaumont	0043077	No	Single Fmly	2	\$10,092	-
Beaumont	0043164	No	Single Fmly	2	\$6,369	-
Beaumont	0067976	No	Single Fmly	2	\$30,867	-
Beaumont	0067943	Yes	Single Fmly	3	\$27,170	-
Beaumont	0044356	No	Single Fmly	3	\$21,774	-
Beaumont	0044589	No	Single Fmly	2	\$9,961	-
Beaumont	0044737	No	Single Fmly	5	\$67,122	MVU
Beaumont	0043997	No	Single Fmly	4	\$77,451	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Beaumont	0043171	No	Single Fmly	4	\$58,040	-
Beaumont	0067977	No	Single Fmly	4	\$77,196	MVU
Beaumont	0045181	No	Single Fmly	3	\$17,988	-
Beaumont	0073390	No	Single Fmly	4	\$43,273	-
Beaumont	0073391	No	Single Fmly	4	\$72,526	MVU
Beaumont	0046314	No	Single Fmly	5	\$70,852	MVU
Beaumont	0121014	No	Single Fmly	3	\$56,654	-
Beaumont	0080904	No	Single Fmly	3	\$47,722	-
Beaumont	0067944	No	Single Fmly	5	\$95,029	MVU
Beaumont	0121073	Yes	Single Fmly	3	\$50,164	-
Beaumont	0043038	Yes	Single Fmly	6	\$93,780	MV
Beaumont	0044931	No	Single Fmly	7	\$94,813	MVU
Beaumont	0036835	Yes	Single Fmly	4	\$31,974	-
Beaumont	0048688	No	Single Fmly	2	\$14,130	-
Beaumont	0025756	No	Single Fmly	8	\$126,000	MVU
Beaumont	0121067	Yes	Single Fmly	3	\$34,658	-
Beaumont	0113791	No	Single Fmly	3	\$55,762	-
Beaumont	0174600	Yes	Other Resid	5	\$71,252	-
Beaumont	0174601	Yes	Other Resid	5	\$63,814	-
Beaumont	0067982	Yes	Other Resid	5	\$63,541	-
Beaumont	0126221	Yes	Other-Nonres	2	\$46,207	-
Beaumont	0121631	Yes	Other-Nonres	3	\$227,994	-
Beaumont	0166425	Yes	Single Fmly	2	\$17,447	-
Beaumont	0039157	No	Single Fmly	5	\$27,840	-
Beaumont	0067922	Yes	Single Fmly	3	\$34,866	-
Beaumont	0067954	Yes	Single Fmly	3	\$162,635	-
Beaumont	0067955	Yes	Single Fmly	3	\$80,982	-
Beaumont	0067918	No	Single Fmly	2	\$65,187	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Beaumont	0067948	No	Assmd Condo	2	\$14,446	-
Beaumont	0035313	No	Single Fmly	2	\$4,458	-
Beaumont	0091338	Yes	Single Fmly	3	\$28,379	-
Beaumont	0025381	No	Single Fmly	4	\$59,806	MVU
Beaumont	0013023	No	2-4 Family	4	\$36,653	-
Beaumont	0002381	No	Single Fmly	5	\$117,743	MVU
Beaumont	0043237	Yes	Single Fmly	2	\$3,417	-
Beaumont	0025449	No	Single Fmly	4	\$42,639	-
Beaumont	0025340	No	Single Fmly	4	\$36,690	-
Beaumont	0168614	No	Single Fmly	2	\$29,785	-
Beaumont	0043390	No	Single Fmly	4	\$47,932	MVU
Beaumont	0041200	No	Single Fmly	3	\$31,644	-
Beaumont	0046367	No	Single Fmly	2	\$23,780	-
Beaumont	0067972	No	Single Fmly	2	\$11,792	-
Beaumont	0013095	No	Single Fmly	3	\$18,287	-
Beaumont	0003331	No	Single Fmly	5	\$101,558	MVU
Beaumont	0120347	No	Single Fmly	2	\$29,541	-
Beaumont	0002997	No	Single Fmly	3	\$33,492	-
Beaumont	0067942	No	Single Fmly	3	\$11,518	-
Beaumont	0013214	No	Single Fmly	2	\$16,988	-
Beaumont	0013040	No	Single Fmly	3	\$4,939	-
Beaumont	0044352	No	Single Fmly	4	\$21,222	-
Beaumont	0046170	Yes	Single Fmly	3	\$46,474	-
Beaumont	0067912	No	Single Fmly	6	\$53,576	MVU
Beaumont	0154186	No	Single Fmly	2	\$21,129	-
Beaumont	0043677	Yes	Single Fmly	2	\$11,346	-
Beaumont	0045631	No	Single Fmly	3	\$5,338	-
Beaumont	0067921	No	Single Fmly	2	\$9,667	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Beaumont	0001932	No	Single Fmly	4	\$38,988	-
Beaumont	0113236	No	Other-Nonres	3	\$88,993	-
Beaumont	0002610	Yes	Other-Nonres	4	\$107,347	-
Beaumont	0113250	Yes	Other Resid	3	\$81,049	-
Beaumont	0002611	No	Other-Nonres	3	\$147,906	-
Beaumont	0004227	No	Other-Nonres	8	\$1,566,257	-
Beaumont	0067903	No	Single Fmly	2	\$6,178	-
Beaumont	0067983	No	Single Fmly	2	\$12,677	-
Beaumont	0067984	No	Single Fmly	4	\$21,888	-
Beaumont	0122404	No	Single Fmly	2	\$47,406	-
Beaumont	0038812	Yes	Single Fmly	4	\$17,114	-
Beaumont	0067966	No	Single Fmly	4	\$17,912	-
Beaumont	0013232	No	Single Fmly	3	\$22,847	-
Beaumont	0046057	No	2-4 Family	2	\$7,296	-
Beaumont	0125740	No	Single Fmly	2	\$35,778	-
Beaumont	0067985	No	Single Fmly	4	\$65,285	MVU
Beaumont	0121368	Yes	Single Fmly	2	\$31,741	-
Beaumont	0120353	Yes	Single Fmly	2	\$72,755	-
Beaumont	0122067	Yes	Single Fmly	3	\$115,732	-
Beaumont	0002719	Yes	Single Fmly	6	\$222,093	MV
Beaumont	0120306	Yes	Single Fmly	3	\$87,160	-
Beaumont	0121454	Yes	Single Fmly	2	\$67,456	-
Beaumont	0125275	Yes	Single Fmly	2	\$50,214	-
Beaumont	0125274	No	Single Fmly	2	\$50,369	-
Beaumont	0122457	No	Single Fmly	6	\$136,673	MVU
Beaumont	0067926	No	Single Fmly	6	\$91,772	MVU
Beaumont	0121264	No	Single Fmly	3	\$53,584	-
Beaumont	0067950	No	Single Fmly	2	\$9,097	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Beaumont	0119972	No	Single Fmly	3	\$46,345	-
Beaumont	0043050	No	Single Fmly	5	\$90,867	MVU
Beaumont	0053309	No	Single Fmly	4	\$66,084	MVU
Beaumont	0067986	No	Single Fmly	3	\$46,710	-
Beaumont	0045446	Yes	Single Fmly	4	\$46,066	-
Beaumont	0002558	No	Other-Nonres	5	\$60,783	-
Beaumont	0122200	Yes	Other-Nonres	2	\$194,825	-
Beaumont	0121255	No	Single Fmly	2	\$60,386	-
Beaumont	0002670	No	Single Fmly	3	\$29,910	-
Beaumont	0067905	Yes	Single Fmly	2	\$35,991	-
Beaumont	0026687	No	Single Fmly	3	\$38,061	-
Beaumont	0037275	Yes	Single Fmly	5	\$226,525	MV
Beaumont	0067936	No	Other-Nonres	3	\$42,446	-
Beaumont	0002350	No	Other-Nonres	3	\$65,496	-
Beaumont	0013038	No	Single Fmly	6	\$48,263	-
Beaumont	0080899	No	Single Fmly	2	\$6,573	-
Beaumont	0067908	No	Single Fmly	2	\$14,111	-
Beaumont	0122479	No	Single Fmly	2	\$8,411	-
Beaumont	0017248	Yes	Single Fmly	4	\$119,606	-
Beaumont	0048680	No	Single Fmly	2	\$2,483	-
Beaumont	0044024	No	Assmd Condo	3	\$188,218	-
Beaumont	0043701	No	Single Fmly	5	\$71,699	MVU
Beaumont	0007714	No	Single Fmly	10	\$191,607	MVU
Beaumont	0043626	No	Single Fmly	3	\$25,071	-
Beaumont	0048067	No	Single Fmly	2	\$25,197	-
Beaumont	0043229	No	Single Fmly	2	\$15,163	-
Beaumont	0044116	No	Single Fmly	2	\$17,480	-
Beaumont	0153951	Yes	Single Fmly	2	\$31,256	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Beaumont	0067930	No	Single Fmly	2	\$4,594	-
Beaumont	0069691	No	Single Fmly	5	\$33,516	MVU
Beaumont	0122368	No	Single Fmly	3	\$27,716	-
Beaumont	0158059	Yes	Single Fmly	2	\$53,145	-
Beaumont	0043477	No	Other Resid	3	\$39,591	-
Beaumont	0067951	Yes	Single Fmly	3	\$36,681	-
Beaumont	0084043	No	Other-Nonres	5	\$44,749	-
Beaumont	0002564	Yes	Other Resid	6	\$67,028	MV
Beaumont	0017444	No	Single Fmly	4	\$75,195	-
Beaumont	0067958	No	Single Fmly	3	\$51,093	-
Beaumont	0042996	No	Single Fmly	4	\$79,633	-
Beaumont	0025505	Yes	Single Fmly	2	\$21,212	-
Beaumont	0067923	Yes	Single Fmly	3	\$61,622	-
Beaumont	0048683	No	Single Fmly	2	\$14,160	-
Beaumont	0017443	No	Single Fmly	4	\$48,259	-
Beaumont	0043642	No	Single Fmly	6	\$150,001	MVU
Beaumont	0116303	No	Single Fmly	2	\$2,180	-
Beaumont	0120017	No	Single Fmly	2	\$25,557	-
Beaumont	0044677	No	Single Fmly	2	\$30,927	-
Beaumont	0044377	No	Single Fmly	3	\$38,336	-
Beaumont	0067938	No	Single Fmly	5	\$151,489	MVU
Beaumont	0044563	No	Single Fmly	3	\$63,948	-
Beaumont	0121192	No	Single Fmly	3	\$47,322	-
Beaumont	0125281	No	Single Fmly	2	\$26,813	-
Beaumont	0121008	No	Single Fmly	4	\$164,751	MVU
Beaumont	0048024	No	Single Fmly	2	\$2,588	-
Beaumont	0001249	No	Single Fmly	16	\$150,821	-
Beaumont	0001350	No	Single Fmly	13	\$123,846	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Beaumont	0018763	No	Single Fmly	2	\$10,928	-
Beaumont	0067910	Yes	Single Fmly	5	\$54,614	-
Beaumont	0151749	No	Single Fmly	3	\$4,871	-
Beaumont	0120304	No	Single Fmly	3	\$72,328	-
Beaumont	0114423	No	Single Fmly	2	\$17,521	-
Beaumont	0125992	Yes	Single Fmly	2	\$31,592	-
Beaumont	0125273	Yes	Single Fmly	2	\$7,962	-
Beaumont	0072233	No	Single Fmly	5	\$57,943	MVU
Beaumont	0124682	No	Single Fmly	4	\$53,765	-
Beaumont	0126927	No	Single Fmly	2	\$15,831	-
Beaumont	0121953	Yes	Single Fmly	2	\$18,925	-
Beaumont	0067956	No	Single Fmly	2	\$18,742	-
Beaumont	0042987	No	Single Fmly	3	\$57,944	-
Beaumont	0119960	No	Single Fmly	2	\$3,483	-
Beaumont	0122244	Yes	Single Fmly	2	\$65,915	-
Beaumont	0121004	Yes	Single Fmly	2	\$180,070	-
Beaumont	0122032	No	Single Fmly	3	\$92,692	-
Beaumont	0115532	Yes	Single Fmly	3	\$246,558	-
Beaumont	0122242	Yes	Single Fmly	2	\$152,383	-
Beaumont	0067964	Yes	Single Fmly	7	\$47,265	MV
Beaumont	0148788	Yes	Single Fmly	2	\$3,933	-
Beaumont	0013146	No	Single Fmly	3	\$132,349	-
Beaumont	0054021	No	Single Fmly	4	\$174,452	-
Beaumont	0121193	Yes	Single Fmly	3	\$50,810	-
Beaumont	0067940	No	Single Fmly	3	\$52,300	-
Beaumont	0001915	No	Single Fmly	6	\$107,137	MVU
Beaumont	0121248	Yes	Single Fmly	2	\$27,770	-
Beaumont	0002664	No	Single Fmly	5	\$61,704	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Beaumont	0123219	Yes	Single Fmly	3	\$168,724	-
Beaumont	0044509	No	Single Fmly	4	\$78,377	-
Beaumont	0073387	No	Single Fmly	3	\$12,246	-
Beaumont	0121913	No	Single Fmly	3	\$31,783	-
Beaumont	0067963	Yes	Single Fmly	7	\$125,037	MV
Beaumont	0182144	No	Single Fmly	2	\$8,494	-
Beaumont	0181222	No	Single Fmly	2	\$47,385	-
Beaumont	0116244	No	Single Fmly	4	\$62,794	-
Beaumont	0160118	No	Single Fmly	2	\$5,804	-
Beaumont	0120037	No	Single Fmly	2	\$13,621	-
Beaumont	0122047	No	Single Fmly	2	\$25,367	-
Beaumont	0121406	No	Single Fmly	2	\$47,048	-
Beaumont	0121011	Yes	Single Fmly	3	\$7,303	-
Beaumont	0073388	No	Single Fmly	6	\$198,169	MVU
Beaumont	0043469	No	Single Fmly	5	\$43,863	-
Beaumont	0036436	No	Single Fmly	3	\$16,224	-
Beaumont	0025316	No	Single Fmly	4	\$56,616	-
Beaumont	0067941	No	Single Fmly	3	\$38,538	-
Beaumont	0121275	Yes	Single Fmly	2	\$21,064	-
Beaumont	0121372	No	Single Fmly	3	\$44,027	-
Beaumont	0122391	No	Single Fmly	3	\$77,942	-
Beaumont	0043174	No	Single Fmly	5	\$86,509	MVU
Beaumont	0025455	No	Single Fmly	7	\$208,942	MVU
Beaumont	0041309	Yes	Single Fmly	4	\$25,759	-
Beaumont	0160195	No	Single Fmly	2	\$39,507	-
Beaumont	0069886	No	Single Fmly	3	\$27,554	-
Beaumont	0157054	No	Single Fmly	2	\$67,667	-
Beaumont	0070674	No	Single Fmly	2	\$84,068	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Beaumont	0112944	Yes	Single Fmly	2	\$136,809	-
Beaumont	0113800	Yes	Single Fmly	2	\$71,415	-
Beaumont	0071441	Yes	Single Fmly	2	\$100,318	-
Beaumont	0070124	Yes	Single Fmly	2	\$86,669	-
Beaumont	0073393	No	Single Fmly	2	\$45,636	-
Beaumont	0070006	SDF	Single Fmly	8	\$533,292	V
Beaumont	0073386	SDF	Single Fmly	6	\$518,909	V
Beaumont	0013074	Yes	Single Fmly	2	\$121,983	-
Beaumont	0013261	Yes	Single Fmly	5	\$98,251	-
Beaumont	0115904	No	Single Fmly	2	\$41,501	-
Beaumont	0071593	Yes	Assmd Condo	2	\$66,356	-
Beaumont	0073394	No	Single Fmly	2	\$117,062	-
Groves	0158057	Yes	Single Fmly	2	\$6,580	-
Groves	0212638	No	Single Fmly	2	\$8,559	-
Groves	0166491	No	Single Fmly	2	\$8,371	-
Groves	0165547	No	Single Fmly	2	\$35,486	-
Groves	0240001	Yes	Single Fmly	3	\$67,437	-
Groves	0166488	Yes	Single Fmly	2	\$41,798	-
Groves	0178844	No	Single Fmly	2	\$24,219	-
Groves	0179760	No	Single Fmly	2	\$32,253	-
Groves	0073544	No	Single Fmly	5	\$52,562	-
Groves	0181067	No	Single Fmly	2	\$17,571	-
Groves	0240289	No	Single Fmly	2	\$13,882	-
Groves	0160604	No	Single Fmly	2	\$15,594	-
Groves	0153959	No	Single Fmly	2	\$28,097	-
Groves	0148796	No	Single Fmly	2	\$6,149	-
Groves	0239948	No	Single Fmly	2	\$13,847	-
Groves	0239950	No	Single Fmly	2	\$49,883	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Groves	0167200	No	Single Fmly	3	\$66,186	-
Groves	0179719	No	Single Fmly	2	\$2,726	-
Groves	0173542	Yes	Single Fmly	2	\$54,810	-
Groves	0148797	No	Single Fmly	2	\$25,011	-
Groves	0164893	No	Single Fmly	2	\$30,830	-
Groves	0025570	No	Single Fmly	2	\$5,916	-
Groves	0165766	Yes	Single Fmly	3	\$65,093	-
Groves	0018654	No	Single Fmly	2	\$23,364	-
Groves	0026439	No	Single Fmly	2	\$16,767	-
Groves	0122123	No	Single Fmly	4	\$61,325	-
Groves	0239758	No	Single Fmly	2	\$23,586	-
Groves	0165373	No	Single Fmly	3	\$50,691	-
Groves	0035811	No	Other Resid	2	\$24,221	-
Groves	0127798	Yes	Single Fmly	2	\$10,617	-
Groves	0166755	No	Other-Nonres	2	\$36,409	-
Groves	0165938	No	Other-Nonres	2	\$136,473	-
Groves	0048991	No	Single Fmly	2	\$4,069	-
Groves	0046237	No	Other-Nonres	3	\$14,951	-
Nederland	0080461	SDF	Single Fmly	4	\$135,837	V
Nederland	0151764	Yes	Single Fmly	2	\$60,411	-
Nederland	0177321	Yes	Single Fmly	2	\$3,018	-
Nederland	0167359	Yes	Single Fmly	2	\$9,853	-
Nederland	0184788	No	Single Fmly	2	\$42,169	-
Nederland	0080462	No	Single Fmly	3	\$126,812	-
Nederland	0151762	Yes	Single Fmly	2	\$28,833	-
Nederland	0108336	SDF	Single Fmly	3	\$166,368	V
Nederland	0148769	No	Single Fmly	2	\$60,463	-
Nederland	0120095	SDF	Other-Nonres	4	\$174,127	VN

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Nederland	0163517	No	Single Fmly	2	\$64,748	-
Nederland	0157073	Yes	Single Fmly	2	\$60,693	-
Nederland	0120317	No	Single Fmly	2	\$4,951	-
Nederland	0108428	Yes	Single Fmly	3	\$24,251	-
Nederland	0108431	No	Single Fmly	3	\$62,615	-
Nederland	0152853	No	Single Fmly	2	\$57,422	-
Nederland	0108351	No	Single Fmly	2	\$16,475	-
Nederland	0153448	No	Single Fmly	2	\$78,644	-
Nederland	0108429	No	Single Fmly	3	\$10,015	-
Nederland	0157076	No	Single Fmly	2	\$102,000	-
Nederland	0157074	No	Single Fmly	2	\$33,370	-
Nederland	0108352	No	Single Fmly	3	\$56,515	-
Nederland	0108547	Yes	Other-Nonres	5	\$32,927	-
Nederland	0167853	No	Other-Nonres	4	\$43,456	-
Nederland	0184364	No	Other-Nonres	2	\$221,712	-
Nederland	0112589	No	Single Fmly	3	\$30,912	-
Nederland	0073827	No	Single Fmly	7	\$102,499	VU
Nederland	0199335	Yes	Single Fmly	2	\$33,740	-
Nederland	0158682	No	Single Fmly	2	\$61,901	-
Nederland	0108243	No	Single Fmly	2	\$16,416	-
Nederland	0148779	Yes	Single Fmly	2	\$34,330	-
Nederland	0080463	Yes	Single Fmly	3	\$112,358	-
Nederland	0080464	No	Single Fmly	4	\$122,651	-
Nederland	0080465	Yes	Single Fmly	3	\$80,097	-
Nederland	0080466	SDF	Single Fmly	5	\$112,375	V
Nederland	0043005	No	Single Fmly	2	\$2,980	-
Nederland	0041191	No	Single Fmly	11	\$63,676	VU
Nederland	0025453	No	Other-Nonres	3	\$5,913	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Nederland	0148785	No	Single Fmly	2	\$67,781	-
Nederland	0108233	No	Single Fmly	2	\$13,664	-
Nederland	0144557	Yes	Single Fmly	2	\$51,731	-
Nederland	0148775	No	Single Fmly	2	\$61,064	-
Nederland	0154378	Yes	Other-Nonres	2	\$65,490	-
Nederland	0108229	Yes	Single Fmly	2	\$26,697	-
Nederland	0108235	No	Single Fmly	2	\$4,564	-
Nederland	0116427	Yes	Single Fmly	6	\$40,088	-
Nederland	0212439	No	Other-Nonres	2	\$20,627	-
Nederland	0181107	No	Single Fmly	2	\$54,439	-
Nederland	0153948	No	Single Fmly	2	\$9,861	-
Nederland	0181750	No	Single Fmly	3	\$42,981	-
Nederland	0080467	No	Single Fmly	4	\$44,718	-
Nederland	0073581	Yes	Single Fmly	4	\$36,154	-
Nederland	0108514	Yes	Single Fmly	3	\$68,400	-
Nederland	0073582	No	Single Fmly	3	\$27,735	-
Nederland	0108028	No	Single Fmly	2	\$53,577	-
Nederland	0152831	Yes	Single Fmly	2	\$77,440	-
Nederland	0109443	No	Single Fmly	2	\$17,940	-
Nederland	0153953	Yes	Single Fmly	2	\$34,696	-
Nederland	0108318	Yes	Single Fmly	3	\$91,043	-
Nederland	0158063	No	Single Fmly	2	\$95,355	-
Nederland	0182998	No	Single Fmly	2	\$95,097	-
Nederland	0108116	Yes	Single Fmly	3	\$58,059	-
Nederland	0157668	No	Single Fmly	2	\$68,321	-
Nederland	0108280	No	Single Fmly	2	\$38,020	-
Nederland	0108316	No	Single Fmly	3	\$89,909	-
Nederland	0148798	Yes	Single Fmly	2	\$54,943	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Nederland	0151765	Yes	Single Fmly	2	\$50,069	-
Nederland	0144556	Yes	Single Fmly	2	\$20,126	-
Nederland	0158062	No	Single Fmly	2	\$19,185	-
Nederland	0157848	No	Single Fmly	2	\$79,411	-
Nederland	0108433	Yes	Single Fmly	4	\$165,672	-
Nederland	0160073	No	Single Fmly	2	\$67,371	-
Nederland	0148791	No	Single Fmly	2	\$36,405	-
Nederland	0108237	No	Single Fmly	2	\$21,449	-
Nederland	0160071	Yes	Single Fmly	2	\$133,921	-
Port Arthur	0121692	No	Single Fmly	2	\$2,237	-
Port Arthur	0183693	Yes	Single Fmly	2	\$69,074	-
Port Arthur	0046046	No	Single Fmly	4	\$72,094	-
Port Arthur	0043993	No	Single Fmly	2	\$30,300	-
Port Arthur	0018413	No	Single Fmly	2	\$4,589	-
Port Arthur	0181881	Yes	Single Fmly	2	\$125,502	-
Port Arthur	0191554	Yes	Single Fmly	3	\$141,128	-
Port Arthur	0182778	No	Single Fmly	2	\$31,017	-
Port Arthur	0158061	SDF	Single Fmly	5	\$79,472	V
Port Arthur	0026126	Yes	Single Fmly	5	\$89,436	-
Port Arthur	0148789	SDF	Single Fmly	5	\$105,153	V
Port Arthur	0025772	SDF	Single Fmly	7	\$130,266	V
Port Arthur	0043531	No	Single Fmly	5	\$72,270	VU
Port Arthur	0017271	No	Single Fmly	2	\$14,959	-
Port Arthur	0148776	No	Single Fmly	2	\$100,676	-
Port Arthur	0025217	No	Single Fmly	3	\$29,308	-
Port Arthur	0239760	No	Single Fmly	2	\$49,578	-
Port Arthur	0025668	Yes	Single Fmly	5	\$35,885	-
Port Arthur	0016038	Yes	Single Fmly	4	\$20,229	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Port Arthur	0044089	No	Single Fmly	3	\$37,206	-
Port Arthur	0183097	Yes	Single Fmly	2	\$61,940	-
Port Arthur	0181246	No	Other-Nonres	2	\$82,900	-
Port Arthur	0121182	Yes	Single Fmly	3	\$110,547	-
Port Arthur	0045140	No	Single Fmly	3	\$11,081	-
Port Arthur	0157060	No	Single Fmly	3	\$49,839	-
Port Arthur	0040436	No	Single Fmly	2	\$11,254	-
Port Arthur	0240282	Yes	Single Fmly	2	\$133,359	-
Port Arthur	0181273	Yes	Single Fmly	2	\$78,582	-
Port Arthur	0043740	No	Single Fmly	2	\$7,085	-
Port Arthur	0015184	Yes	Single Fmly	5	\$91,835	-
Port Arthur	0239940	No	Single Fmly	2	\$13,266	-
Port Arthur	0067850	No	Single Fmly	4	\$14,094	-
Port Arthur	0162789	No	Other Resid	2	\$32,559	-
Port Arthur	0163773	No	Assmd Condo	2	\$106,620	-
Port Arthur	0163515	No	Other Resid	2	\$26,741	-
Port Arthur	0164169	No	Other Resid	2	\$45,394	-
Port Arthur	0128020	No	Single Fmly	3	\$21,417	-
Port Arthur	0018764	No	Single Fmly	2	\$17,876	-
Port Arthur	0033274	No	Single Fmly	2	\$11,377	-
Port Arthur	0039658	No	Single Fmly	2	\$11,629	-
Port Arthur	0039659	No	Assmd Condo	2	\$39,934	-
Port Arthur	0043279	No	Single Fmly	2	\$16,616	-
Port Arthur	0016071	No	Single Fmly	2	\$39,845	-
Port Arthur	0036493	No	Single Fmly	2	\$17,130	-
Port Arthur	0026537	No	Single Fmly	2	\$45,033	-
Port Arthur	0067833	No	Single Fmly	2	\$18,477	-
Port Arthur	0017502	No	Single Fmly	2	\$4,405	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Port Arthur	0017477	No	Single Fmly	2	\$14,964	-
Port Arthur	0026213	Yes	Single Fmly	3	\$101,381	-
Port Arthur	0240281	No	Single Fmly	2	\$7 <i>,</i> 869	-
Port Arthur	0244472	Yes	Single Fmly	2	\$69,408	-
Port Arthur	0179771	No	Single Fmly	2	\$44,488	-
Port Arthur	0018696	No	Single Fmly	2	\$14,235	-
Port Arthur	0045668	No	Other-Nonres	2	\$8,971	-
Port Arthur	0025188	No	Other-Nonres	5	\$73,368	-
Port Arthur	0044418	No	Other-Nonres	8	\$61,774	VNU
Port Arthur	0026111	No	Other-Nonres	2	\$4,369	-
Port Arthur	0239941	Yes	Other-Nonres	2	\$149,791	-
Port Arthur	0186628	Yes	Single Fmly	2	\$82,978	-
Port Arthur	0183915	No	Other-Nonres	2	\$397,735	-
Port Arthur	0179222	No	Single Fmly	2	\$116,900	-
Port Arthur	0190401	SDF	Other-Nonres	2	\$652,024	PN
Port Arthur	0179200	No	Assmd Condo	2	\$532,276	-
Port Arthur	0179440	No	Single Fmly	2	\$406,159	VU
Port Arthur	0181124	No	Single Fmly	2	\$117,500	-
Port Arthur	0049019	No	Other-Nonres	3	\$7,749	-
Port Arthur	0045629	No	Other-Nonres	2	\$10,315	-
Port Arthur	0025220	SDF	Other-Nonres	10	\$535,976	PN
Port Arthur	0025577	No	Other-Nonres	2	\$18,633	-
Port Arthur	0186523	No	Single Fmly	2	\$8,367	-
Port Arthur	0157061	No	Single Fmly	2	\$48,131	-
Port Arthur	0179663	Yes	Single Fmly	2	\$153,617	-
Port Arthur	0180138	Yes	Single Fmly	2	\$199,051	-
Port Arthur	0181767	Yes	Single Fmly	2	\$180,870	-
Port Arthur	0181851	No	Single Fmly	2	\$189,100	VU

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Port Arthur	0181125	No	Single Fmly	2	\$112,788	VU
Port Arthur	0026049	No	Single Fmly	2	\$19,873	-
Port Arthur	0017672	No	Single Fmly	2	\$11,569	-
Port Arthur	0158058	No	Single Fmly	2	\$19,005	-
Port Arthur	0122910	No	Single Fmly	3	\$41,241	-
Port Arthur	0179078	No	Other-Nonres	2	\$54,258	-
Port Arthur	0025172	No	Single Fmly	2	\$25,839	-
Port Arthur	0179781	Yes	Single Fmly	2	\$33,484	-
Port Arthur	0157059	Yes	Single Fmly	2	\$38,540	-
Port Arthur	0017297	No	Single Fmly	2	\$8,632	-
Port Arthur	0040469	No	Single Fmly	2	\$18,438	-
Port Arthur	0180881	No	Single Fmly	2	\$75,873	-
Port Arthur	0181228	Yes	Single Fmly	2	\$79,229	-
Port Arthur	0181314	No	Single Fmly	2	\$8,238	-
Port Arthur	0043554	No	Single Fmly	2	\$3,365	-
Port Arthur	0183963	Yes	Single Fmly	2	\$12,401	-
Port Arthur	0182783	No	Single Fmly	2	\$125,696	-
Port Arthur	0178112	Yes	Single Fmly	2	\$43,007	-
Port Arthur	0180880	No	Single Fmly	2	\$65,197	-
Port Arthur	0185394	No	Single Fmly	2	\$145,387	-
Port Arthur	0184075	Yes	Single Fmly	2	\$88,838	-
Port Arthur	0179666	No	Single Fmly	2	\$95,291	-
Port Arthur	0185462	Yes	Single Fmly	2	\$50,458	-
Port Arthur	0043706	No	Single Fmly	2	\$12,992	-
Port Arthur	0212548	Yes	Single Fmly	2	\$11,334	-
Port Arthur	0179778	No	Single Fmly	2	\$65,262	-
Port Arthur	0184986	SDF	Single Fmly	2	\$187,249	V
Port Arthur	0183694	Yes	Single Fmly	2	\$182,101	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Port Arthur	0179748	Yes	Single Fmly	2	\$28,369	-
Port Arthur	0026839	No	Single Fmly	4	\$102,819	-
Port Arthur	0180080	Yes	Single Fmly	2	\$46,149	-
Port Arthur	0184110	No	Single Fmly	2	\$105,200	-
Port Arthur	0191141	No	Single Fmly	2	\$37,588	-
Port Arthur	0026429	No	Single Fmly	3	\$5 <i>,</i> 824	-
Port Arthur	0043287	No	Other-Nonres	2	\$48,900	-
Port Arthur	0178175	Yes	Single Fmly	2	\$136,457	-
Port Arthur	0182111	Yes	Single Fmly	2	\$404,954	-
Port Arthur	0181445	Yes	Single Fmly	2	\$263,089	-
Port Arthur	0182136	Yes	Assmd Condo	2	\$586,755	-
Port Arthur	0183157	SDF	Single Fmly	2	\$369,716	V
Port Arthur	0181955	No	Single Fmly	2	\$246,306	VU
Port Arthur	0178197	Yes	Other-Nonres	2	\$102,752	-
Port Arthur	0177177	Yes	Single Fmly	2	\$66,748	-
Port Arthur	0181355	Yes	Single Fmly	2	\$191,496	-
Port Arthur	0177163	Yes	Single Fmly	2	\$24,803	-
Port Arthur	0183308	Yes	Single Fmly	2	\$135,645	-
Port Arthur	0179216	Yes	Single Fmly	2	\$66,637	-
Port Arthur	0178190	Yes	Single Fmly	2	\$126,081	-
Port Arthur	0160104	No	Single Fmly	2	\$32,629	-
Port Arthur	0116245	No	Single Fmly	2	\$14,691	-
Port Arthur	0035930	No	Single Fmly	5	\$124,516	VU
Port Arthur	0026465	No	Single Fmly	2	\$4,231	-
Port Arthur	0158043	No	Single Fmly	3	\$16,135	-
Port Arthur	0182779	No	Single Fmly	2	\$43,978	-
Port Arthur	0239759	No	Single Fmly	2	\$24,722	-
Port Arthur	0160056	No	Single Fmly	3	\$9,199	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Port Arthur	0047348	No	Single Fmly	6	\$36,195	-
Port Arthur	0176706	No	Single Fmly	2	\$29,034	-
Port Arthur	0180151	Yes	Single Fmly	2	\$36,221	-
Port Arthur	0049655	No	Single Fmly	13	\$81,463	VU
Port Arthur	0042615	No	Single Fmly	2	\$6,241	-
Port Arthur	0026746	No	Single Fmly	4	\$14,707	-
Port Arthur	0181363	Yes	Other-Nonres	2	\$179,744	-
Port Arthur	0017582	No	Other-Nonres	2	\$14,296	-
Port Arthur	0148790	Yes	Other-Nonres	2	\$49,255	-
Port Arthur	0180882	No	Single Fmly	2	\$9,856	-
Port Arthur	0179009	No	Single Fmly	2	\$32,967	-
Port Arthur	0238882	Yes	Single Fmly	2	\$16,097	-
Port Arthur	0180883	No	Single Fmly	2	\$4,530	-
Port Arthur	0153449	SDF	Single Fmly	5	\$45,145	V
Port Arthur	0026488	No	Single Fmly	2	\$9 <i>,</i> 676	-
Port Arthur	0157063	No	Single Fmly	3	\$12,409	-
Port Arthur	0026522	Yes	Single Fmly	3	\$13,412	-
Port Arthur	0035476	No	Single Fmly	2	\$9,544	-
Port Arthur	0017438	No	Other-Nonres	4	\$67,340	PNU
Port Arthur	0157058	No	Single Fmly	2	\$9 <i>,</i> 073	-
Port Arthur	0179717	Yes	2-4 Family	3	\$30,450	-
Port Arthur	0178772	No	Single Fmly	2	\$9,591	-
Port Arthur	0158053	No	Other-Nonres	3	\$168,900	-
Port Arthur	0178771	No	Single Fmly	2	\$17,431	-
Port Arthur	0041485	No	Single Fmly	3	\$11,872	-
Port Arthur	0025562	No	Single Fmly	2	\$22,315	-
Port Arthur	0240279	No	Single Fmly	2	\$37,789	-
Port Arthur	0044571	No	Single Fmly	3	\$8,639	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Port Arthur	0184077	No	Single Fmly	2	\$51,943	-
Port Arthur	0180264	No	Single Fmly	2	\$20,698	-
Port Arthur	0178770	No	Single Fmly	2	\$41,723	-
Port Arthur	0018482	No	Single Fmly	2	\$11,038	-
Port Arthur	0180162	Yes	Single Fmly	2	\$29,457	-
Port Arthur	0173145	No	Single Fmly	2	\$13,006	-
Port Arthur	0039584	No	Single Fmly	4	\$40,633	-
Port Arthur	0018444	No	Single Fmly	2	\$31,545	PU
Port Arthur	0158052	No	Single Fmly	2	\$16,852	-
Port Arthur	0073356	No	Single Fmly	3	\$9,619	-
Port Arthur	0070053	No	Other-Nonres	2	\$10,989	-
Port Arthur	0244554	No	Single Fmly	2	\$54,045	-
Port Arthur	0043795	No	Single Fmly	2	\$18,377	-
Port Arthur	0178998	No	Single Fmly	2	\$153,400	-
Port Arthur	0178897	Yes	Single Fmly	2	\$53,477	-
Port Arthur	0191890	No	Single Fmly	2	\$163,200	PU
Port Arthur	0181708	SDF	Single Fmly	2	\$390,511	V
Port Arthur	0162606	No	Single Fmly	2	\$25,753	-
Port Arthur	0069866	No	Single Fmly	2	\$6,385	-
Port Arthur	0179773	No	Single Fmly	3	\$40,961	-
Port Arthur	0183034	No	Single Fmly	2	\$16,780	-
Port Arthur	0037866	No	Single Fmly	3	\$28,910	-
Port Arthur	0148787	No	Single Fmly	2	\$14,174	-
Port Arthur	0158050	Yes	Single Fmly	2	\$47,988	-
Port Arthur	0025154	No	Single Fmly	2	\$31,859	-
Port Arthur	0026151	No	Single Fmly	2	\$33,739	-
Port Arthur	0076301	Yes	Single Fmly	4	\$51,543	-
Port Arthur	0154084	Yes	Single Fmly	2	\$24,729	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Port Arthur	0048784	No	Single Fmly	2	\$9,268	-
Port Arthur	0154136	No	Single Fmly	2	\$26,086	-
Port Arthur	0067839	No	Single Fmly	8	\$195,360	VU
Port Arthur	0122980	No	Single Fmly	3	\$97,516	-
Port Arthur	0026266	No	Single Fmly	2	\$4,040	-
Port Arthur	0157064	Yes	Single Fmly	3	\$59,999	-
Port Arthur	0158060	No	Other Resid	2	\$120,398	-
Port Arthur	0044729	Yes	Single Fmly	3	\$10,656	-
Port Arthur	0067846	No	Other-Nonres	3	\$28,577	-
Port Arthur	0153955	No	Single Fmly	3	\$45,599	-
Port Arthur	0184624	No	Single Fmly	2	\$75,251	-
Port Arthur	0015989	No	Single Fmly	2	\$29,135	-
Port Arthur	0026168	No	Single Fmly	2	\$18,394	-
Port Arthur	0239721	Yes	Single Fmly	2	\$95,833	-
Port Arthur	0183696	No	Single Fmly	2	\$48,619	-
Port Arthur	0015126	No	Single Fmly	3	\$80,405	-
Port Arthur	0122981	Yes	Single Fmly	4	\$182,330	-
Port Arthur	0157066	No	Single Fmly	2	\$43,629	-
Port Arthur	0039204	No	Single Fmly	3	\$25,238	-
Port Arthur	0026588	No	Single Fmly	2	\$8,636	-
Port Arthur	0026179	No	Single Fmly	2	\$3,759	-
Port Arthur	0070125	No	Single Fmly	2	\$6,361	-
Port Arthur	0148801	No	Single Fmly	2	\$36,473	-
Port Arthur	0040259	Yes	Single Fmly	3	\$37,080	-
Port Arthur	0018748	Yes	Single Fmly	2	\$19,248	-
Port Arthur	0178165	No	Single Fmly	2	\$15,848	-
Port Arthur	0016075	No	Single Fmly	2	\$12,746	-
Port Arthur	0182278	No	Single Fmly	2	\$149,353	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Port Arthur	0179741	Yes	Single Fmly	2	\$226,309	-
Port Arthur	0148778	No	Single Fmly	2	\$21,488	-
Port Arthur	0039990	No	Single Fmly	2	\$3,259	-
Port Arthur	0160605	Yes	Single Fmly	4	\$32,861	-
Port Arthur	0121521	No	Single Fmly	2	\$8,142	-
Port Arthur	0179067	No	Single Fmly	2	\$7,213	-
Port Arthur	0015179	No	Single Fmly	2	\$4,209	-
Port Arthur	0017256	Yes	Single Fmly	3	\$35,981	-
Port Arthur	0179089	Yes	Single Fmly	2	\$89,077	-
Port Arthur	0017339	Yes	Single Fmly	4	\$119,309	-
Port Arthur	0183160	Yes	Single Fmly	2	\$116,406	-
Port Arthur	0016044	Yes	Single Fmly	4	\$225,909	-
Port Arthur	0026284	Yes	Single Fmly	3	\$69,861	-
Port Arthur	0026714	Yes	Single Fmly	4	\$170,018	-
Port Arthur	0183695	No	Single Fmly	2	\$200,265	-
Port Arthur	0181388	No	Other-Nonres	2	\$98,457	PNU
Port Arthur	0185445	Yes	Single Fmly	2	\$19,106	-
Port Arthur	0244159	Yes	Single Fmly	2	\$6,001	-
Port Arthur	0080912	No	Single Fmly	2	\$4,863	-
Port Arthur	0197230	No	Single Fmly	3	\$69,257	-
Port Arthur	0042551	No	Single Fmly	2	\$16,857	-
Port Arthur	0026487	No	Single Fmly	2	\$14,434	-
Port Arthur	0182784	No	Single Fmly	2	\$40,483	-
Port Arthur	0036863	No	Single Fmly	5	\$21,564	-
Port Arthur	0035778	Yes	Single Fmly	4	\$12,502	-
Port Arthur	0067845	No	Single Fmly	2	\$4,644	-
Port Arthur	0158047	Yes	Single Fmly	2	\$16,268	-
Port Arthur	0240109	Yes	Single Fmly	2	\$19,504	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Port Arthur	0240284	No	Single Fmly	2	\$40,465	-
Port Arthur	0160057	Yes	Single Fmly	2	\$25,881	-
Port Arthur	0157065	Yes	Single Fmly	2	\$57,037	-
Port Arthur	0046161	No	Single Fmly	2	\$14,792	-
Port Arthur	0239942	Yes	Single Fmly	2	\$31,798	-
Port Arthur	0240100	Yes	Single Fmly	2	\$88,370	-
Port Arthur	0240676	Yes	Single Fmly	2	\$128,955	-
Port Arthur	0239722	No	Single Fmly	2	\$68,560	-
Port Arthur	0240111	No	Single Fmly	2	\$11,371	-
Port Arthur	0025821	No	Single Fmly	2	\$5,625	-
Port Arthur	0018631	No	Single Fmly	2	\$29,506	-
Port Arthur	0177210	Yes	Other-Nonres	2	\$210,072	-
Port Arthur	0043780	No	2-4 Family	2	\$2,573	-
Port Arthur	0025311	No	Single Fmly	3	\$16,117	-
Port Arthur	0015136	No	Single Fmly	2	\$11,454	-
Port Arthur	0033264	No	Single Fmly	2	\$12,349	-
Port Arthur	0178195	No	Single Fmly	2	\$82,305	-
Port Arthur	0023243	No	Single Fmly	3	\$8,001	-
Port Arthur	0068035	No	Single Fmly	2	\$27,235	-
Port Arthur	0048901	No	Other-Nonres	2	\$13,220	-
Port Arthur	0146219	No	Single Fmly	2	\$62,409	-
Port Arthur	0043272	No	Single Fmly	2	\$19,266	-
Port Arthur	0043036	No	Single Fmly	3	\$6,943	-
Port Arthur	0178901	No	Other-Nonres	2	\$399,300	-
Port Arthur	0183138	No	Other-Nonres	2	\$1,025,600	-
Port Arthur	0177058	No	2-4 Family	2	\$64,000	-
Port Arthur	0177123	No	Single Fmly	2	\$23,300	-
Port Arthur	0182274	No	Single Fmly	2	\$92,630	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Port Arthur	0183132	No	Single Fmly	2	\$201,764	-
Port Arthur	0043391	No	Single Fmly	3	\$8,365	-
Port Arthur	0083549	No	Single Fmly	3	\$40,436	-
Port Arthur	0044523	No	Single Fmly	2	\$20,439	-
Port Arthur	0026216	No	Single Fmly	2	\$22,813	-
Port Arthur	0167671	No	Single Fmly	2	\$92,681	-
Port Arthur	0181923	No	Single Fmly	2	\$488,277	MVU
Port Arthur	0144555	No	Single Fmly	2	\$116,031	-
Port Arthur	0179375	Yes	Single Fmly	2	\$244,316	-
Port Arthur	0180142	No	Single Fmly	2	\$69,500	-
Port Arthur	0181807	No	Single Fmly	2	\$155,109	-
Port Arthur	0238125	Yes	Single Fmly	2	\$337,696	-
Port Arthur	0181364	Yes	Single Fmly	2	\$700,000	MV
Port Arthur	0025694	Yes	Single Fmly	3	\$16,411	-
Port Neches	0108117	No	Single Fmly	2	\$12,755	-
Port Neches	0160055	No	Single Fmly	2	\$13,716	-
Port Neches	0160049	Yes	Single Fmly	2	\$61,248	-
Port Neches	0240283	Yes	Single Fmly	2	\$61,201	-
Port Neches	0160054	No	Single Fmly	2	\$18,904	-
Port Neches	0163795	Yes	Single Fmly	3	\$57,456	-
Port Neches	0125284	No	Single Fmly	3	\$10,074	-
Port Neches	0038458	No	Single Fmly	3	\$5,284	-
Port Neches	0158044	No	Single Fmly	2	\$6,641	-
Port Neches	0239723	Yes	Single Fmly	2	\$89,706	-
Port Neches	0117149	No	Single Fmly	3	\$42,584	-
Port Neches	0121919	No	Single Fmly	4	\$13,328	-
Port Neches	0181744	No	Single Fmly	2	\$9,687	-
Port Neches	0127223	Yes	Single Fmly	3	\$43,309	-

COMMUNITY NAME	PROPERTY	INSURED	BUILDING TYPE	LOSSES	TOTAL PAID	SRL INDICATOR
Port Neches	0080414	No	Other-Nonres	4	\$278,651	VNU
Port Neches	0182782	No	Single Fmly	2	\$10,935	-
Port Neches	0147441	No	Single Fmly	2	\$8,681	-
Port Neches	0038382	Yes	Single Fmly	2	\$20,193	-
Port Neches	0181378	No	Single Fmly	2	\$65,742	-
Port Neches	0179531	Yes	Single Fmly	2	\$269,884	-
Port Neches	0076703	Yes	Single Fmly	4	\$58,048	-
Port Neches	0108425	No	Single Fmly	6	\$116,641	VU
Port Neches	0113302	No	Single Fmly	2	\$20,894	-
Port Neches	0157062	No	Single Fmly	2	\$43,361	-
Port Neches	0179765	No	Single Fmly	2	\$19,209	-
Port Neches	0180884	Yes	Single Fmly	2	\$25,024	-
Port Neches	0073897	No	Single Fmly	3	\$49,693	-
Port Neches	0080415	No	Single Fmly	3	\$59,179	-
Port Neches	0038168	No	Single Fmly	2	\$3,941	-
Port Neches	0045286	No	Single Fmly	4	\$16,746	-

# **SECTION 6: LIGHTNING**

Hazard Description	1
Location	1
Extent	1
Historical Occurrences	2
Significant Past Events	3
Probability of Future Events	4
Vulnerability and Impact	4
Assessment of Impacts	6

# Hazard Description

Lightning is a discharge of electrical energy resulting from the buildup of positive and negative charges within a thunderstorm, creating a "bolt" when the buildup of charges becomes strong enough. This flash of light usually occurs within the clouds or between the clouds and the ground. A bolt of lightning can reach temperatures approaching 50,000 degrees Fahrenheit. Lightning rapidly heats the sky as it flashes but the surrounding air cools following the bolt. This rapid heating and cooling of the surrounding air causes the thunder, which often accompanies lightning strikes. While most often affiliated with severe thunderstorms, lightning often strikes outside of heavy rain and might occur as far as 10 miles away from any rainfall.

According to FEMA, an average of 300 people are injured and 80 people are killed in the United States each year by lightning. Direct lightning strikes also have the ability to cause significant damage to buildings, critical facilities, and infrastructure. Lightning is also responsible for igniting wildfires that can result in widespread damages to property before firefighters have the ability to contain and suppress the resultant fire.

# Location

Lightning can strike in any geographic location, and is considered a common occurrence in Texas. The Jefferson County planning area is located in a region of the country that is moderately susceptible to lightning strikes. Therefore lightning could occur at any location within the Jefferson County planning area. It is assumed that the Jefferson County planning area, including the SETRPC and all participating jurisdictions, is uniformly exposed to the threat of lightning.

# Extent

The planning area considers a flash density of less than two to be a minor severity and a flash density of three or greater to be a major severity. Any lightning strike that causes death or property damage is considered a major severity. Vaisala's U.S. National Lightning Detection Network lightning flash density

map (Figure 6-1) shows a range of 12 to 20 lightning flashes per square mile per year for the Jefferson County planning area, including the SETRPC and all participating jurisdictions.



#### Figure 6-1. Lightning Flash Density, 2005-2014<sup>1</sup>

## **Historical Occurrences**

Table 6-1 depicts historical occurrences of lightning for the Jefferson County planning area, including all participating jurisdictions, with associated damages according to the National Center for Environmental Information (NCEI) data. Since January 1996, 11 recorded lightning events are known to have impacted Jefferson County, based upon NCEI records.

The NCEI is a national data source organized under the National Oceanic and Atmospheric Administration and is the largest archive available for climate data. However, it is important to note that the incidents factored into this risk assessment only include incidents that were reported to the NCEI. SETRPC is located within the City of Beaumont. There may be some occurrences that have occurred for the SETRPC and may not have been recorded, but are included in the City of Beaumont occurrence data because of their

<sup>&</sup>lt;sup>1</sup> The black circle indicates the Jefferson County planning area.

location. Damage estimates provided in Table 6-1 and Table 6-2 have been modified to reflect the damage in 2016 dollars.

JURISDICTION	DATE	TIME	DEATHS	INJURIES	PROPERTY DAMAGE	CROP DAMAGE
Beaumont	8/12/1996	9:00 PM	0	1	\$15,249	\$0
Beaumont	7/12/1999	2:00 PM	0	1	\$0	\$0
Port Arthur	8/29/2007	8:00 PM	1	0	\$0	\$0
Port Arthur	7/23/2009	2:22 PM	0	8	\$5,576	\$0
Beaumont	7/5/2011	3:45 PM	0	0	\$31,910	\$0
Beaumont	7/6/2011	2:00 PM	0	0	\$106,367	\$0
Beaumont	7/6/2011	4:50 PM	0	0	\$265,918	\$0
Groves	7/19/2011	5:20 AM	0	0	\$74,457	\$0
Beaumont	8/19/2014	3:00 PM	0	0	\$5,053	\$0
Jefferson County	12/23/2014	12:55 PM	0	0	\$50,533	\$0
Port Neches	6/30/2015	10:39 AM	0	0	\$1,009	\$0

#### Table 6-1. Historical Lightning Events, 1996-2016

Table 6-2. Summary of Historical Lightning Events, 1996-2016

JURISDICTION	NUMBER OF EVENTS	DEATHS	INJURIES	PROPERTY DAMAGE	CROP DAMAGE
Jefferson County	1	0	0	\$50,533	\$0
Beaumont	6	0	2	\$424,497	\$0
Bevil Oaks	0	0	0	\$0	\$0
China	0	0	0	\$0	\$0
Groves	1	0	0	\$74,457	\$0
Nederland	0	0	0	\$0	\$0
Nome	0	0	0	\$0	\$0
Port Arthur	2	1	8	\$5,576	\$0
Port Neches	1	0	0	\$1,009	\$0
TOTAL LOSSES	11	1	10	\$556,0	)73

Based on the list of historical lightning events for the Jefferson County planning area (listed above), including the SETRPC and all participating jurisdictions, 7 of the events have occurred since the 2011 Plan.

#### Significant Past Events

#### August 12, 1996 – Beaumont

An unusual storm system produced extensive lightning in the area. As many as 9,000 lightning strikes that evening resulted in one man injured, one house fire, and several telephone poles damaged.

#### July 6, 2011 – Jefferson County

An upper level disturbance helped produce thunderstorms that had numerous lightning strikes in southeast Texas. A two story home on the west side of Beaumont was struck by lightning. The resulting fire was quickly put out, but not before the home was severely damaged. Excessive lightning also caused a fire at the Stone Hearth Apartments. One unit with eight apartments was on fire and the roof partially collapsed. Damage estimates exceeded \$350,000.

#### July 19, 2011 – Groves

An upper level low pressure area helped produce thunderstorms that had numerous lightning strikes in southeast Texas. A house that was struck by lightning caught fire in Groves. The fire started in the attic of the two story home, and caused considerable water damage on the first floor and fire damage on the second floor.

# Probability of Future Events

Based on historical records and input from the planning team, the probability of occurrence for future lightning events in the Jefferson County planning area, including the SETRPC and all participating jurisdictions, is considered highly likely, or an event probable in the next year. According to NOAA, the Jefferson County planning area is located in an area of the country that experiences 12-20 lightning flashes per square mile per year (approximately 13,332 to 22,220 flashes per year). Given this estimated frequency of occurrence, it can be expected that future lightning events will continue to threaten life and cause minor property damages throughout the planning area.

# Vulnerability and Impact

Vulnerability is difficult to evaluate since lightning events can occur at different strength levels, in random locations, and can create a broad range of damages depending on the strike location. Due to the randomness of these events, all existing and future structures, and facilities in the Jefferson County planning area could potentially be impacted and remain vulnerable to possible injury and property loss from lightning strikes.

The direct and indirect losses associated with these events include injury and loss of life, damage to structures and infrastructure, agricultural losses, utility failure (power outages), and stress on community resources. The entire population of Jefferson County is considered exposed to the lightning hazard. The peak lightning season in the State of Texas is from June to August; however, most fatalities occur in July. Fatalities occur most often when people are outdoors and/or participating in some form of recreation.

Populations located outdoors are considered at risk and more vulnerable to a lightning strike compared to populations inside a structure. Moving to a lower risk location will decrease a person's vulnerability.

The entire general building stock and all infrastructure of Jefferson County are considered exposed to the lightning hazard. Lightning can be responsible for damages to buildings, cause electrical, forest and/or wildfires, and damage infrastructure such as power transmission lines and communication towers. Agricultural losses can be extensive due to lightning and resulting fires.

The following critical facilities would be vulnerable to lightning events in each participating jurisdiction:

JURISDICTION	CRITICAL FACILITIES
Jefferson County	Port Authority Facility, Fire Station, Water District Facility, 1 School
Beaumont	2 Fire Stations, 2 Police Stations, 32 Schools, Port Authority Facility, 5 Water District Facilities, 3 Drainage District Facilities, 4 Hospitals
Bevil Oaks	Fire Station
China	Fire Station, 2 Schools
Groves	Fire Station, Police Station, 4 Schools
Nederland	Fire Station, Police Station, Water District Facility, 3 Hospitals, Airport, 8 Schools
Nome	Fire Station
Port Arthur	Fire Station, Police Station, Drainage District Facility, 2 Port Authority Facilities, 2 Hospitals, 14 Schools
Port Neches	Fire Station, Police Station, 5 Schools
SETRPC	SETRPC Facility

#### Table 6-2. Critical Facilities by Jurisdiction

The impact of lightning experienced in the Jefferson County planning area has resulted in ten injuries and one fatality. While property damage and shutdown of critical facilities would be limited, the risk posed to residents provides a "Substantial" impact of lightning events experienced in the Jefferson County planning area, including the SETRPC and all participating jurisdictions, with multiple potential injuries and fatalities. Overall, the average loss estimate for Jefferson County, including the SETRPC and all participating jurisdictions, (in 2016 dollars) is \$556,073 (Table 6-2), with an approximate annual loss estimate of \$26,480 (Table 6-3).

JURISDICTION	PROPERTY & CROP LOSS	ANNUAL LOSS ESTIMATES
Jefferson County	\$50,533	\$2,406
Beaumont	\$424,497	\$20,214
Bevil Oaks	\$0	\$0
China	\$0	\$0
Groves	\$74,457	\$3,546
Nederland	\$0	\$0
Nome	\$0	\$0
Port Arthur	\$5,576	\$266
Port Neches	\$1,009	\$48
SETRPC	\$0	\$0
Planning Area	\$556,073	\$26,480

#### Table 6-3. Potential Annualized Losses by Jurisdiction, 1996-2016<sup>2</sup>

#### Assessment of Impacts

Lightning events have the potential to pose a significant risk to people, and can create dangerous and difficult situations for public health and safety officials. Impacts to the planning area can include:

- Individuals exposed to the storm can be directly struck, posing significant health risks and potential death.
- Structures can be damaged or crushed by falling trees damaged by lightning, which can result in physical harm to the occupants.
- Lightning strikes can result in widespread power outages increasing the risk to more vulnerable portions of the population who rely on power for health and/or life safety.
- Extended power outage often results in an increase in structure fires and carbon monoxide poisoning, as individuals attempt to cook or heat their homes with alternate, unsafe cooking or heating devices, such as grills.
- Lightning strikes can be associated with structure fires and wildfires, creating additional risk to residents and first responders.
- The Lower Neches Wildlife Management Area may see an elevated risk of wildfire during lightning events.
- Residents and visitors engaged in outdoor recreational activities along Sabine River and Sabine Lake may be at greater risk during lightning events.
- Emergency operations and services may be significantly impacted due to power outages and/or loss of communications.

<sup>&</sup>lt;sup>2</sup> Values are in 2016 dollars.

- City or county departments may be damaged, delaying response and recovery efforts for the entire community.
- Economic disruption due to power outages and fires negatively impacts the programs and services provided by the community due to short and long term loss in revenue.
- Some businesses not directly damaged by lightning events may be negatively impacted while utilities are being restored, further slowing economic recovery.
- Businesses that are more reliant on utility infrastructure than others may suffer greater damages without a backup power source.

The economic and financial impacts of lightning on the area will depend entirely on the scale of the event, what is damaged, and how quickly repairs to critical components of the economy can be implemented. The level of preparedness and pre-event planning done by the county, communities, local businesses, and citizens will also contribute to the overall economic and financial conditions in the aftermath of any lightning event.

# **SECTION 7: HURRICANE**

Hazard Description	1
Location	2
Extent	4
Historical Occurrences	5
Significant Past Events	6
Probability of Future Events	7
Vulnerability and Impact	7
Assessment of Impacts	9

## Hazard Description

According to the National Oceanic and Atmospheric Administration (NOAA), a hurricane is an intense tropical weather system of strong thunderstorms with well-defined surface circulation and maximum sustained winds of 74 miles per hour (mph) or higher. In the Northern Hemisphere, circulation of winds near the Earth's surface is counterclockwise.

Hurricanes often begin as tropical depressions that intensify into tropical storms when maximum sustained winds increase to between 35 – 64 knots (39 – 73 mph). At these wind speeds, the storm becomes more organized and circular in shape and begins to resemble a hurricane. Tropical storms resulting in high winds and heavy rainfall can be equally problematic without ever becoming a hurricane and can be dangerous to people and property, resulting in high winds and heavy rainfall, as Tropical Storm Frances did for southeast Texas in September 1998. Once sustained winds reach or exceed



74 mph, the storm becomes a hurricane. The intensity of a land falling hurricane is expressed in categories relating wind speeds to potential damage. Tropical storm-force winds are strong enough to be dangerous to those caught in them. For this reason, emergency managers plan to have evacuations completed and personnel sheltered before winds of tropical storm-force arrive, which precedes the arrival of hurricane-force winds.

According to the National Hurricane Center (NHC), the greatest potential for loss of life related to a hurricane is from storm surge. This happens when low pressure and high circular winds "pile" the water into a dome shape that can be 50-100 miles wide. The surge travels with the storm and is most severe on the right side of the storm, relative to the direction the storm travels. The surge can be 15 feet deep, topped by waves, and make landfall ahead of the center, or "eye" of the hurricane. Wind-driven waves are superimposed on the storm tide. This rise in water level can cause severe flooding in coastal areas, particularly when the storm tide coincides with normal high tides.

Texas has some of the highest coastal erosion rates in the country, eroding at an average rate of 4.1 feet per year, according to the Texas General Land Office. Coastal erosion is caused by large storms, flooding, sea level rise, and human activities that wear away the beaches and bluffs along the ocean. Erosion can have long-term economic and social consequences. Coastal erosion is fully profiled in Section 15 of the plan.

## Location

As a coastal community, the Jefferson County planning area is vulnerable to threats directly and indirectly related to a hurricane event, such as high-force winds, storm surge, flooding, and coastal erosion (Section 15). Hurricanes and/or tropical storms can impact Jefferson County from June to November, the official Atlantic U.S. hurricane season. The Jefferson County planning area, including the SETRPC and all participating jurisdictions, is in a moderate to high risk area for hurricane wind speeds up to 155 mph as shown in Figure 7-1.



Figure 7-1. Location of Historic Hurricane Tracks

# Extent

As a hurricane develops, the barometric pressure (measured in millibars or inches) at its center falls and winds increase. If the atmospheric and oceanic conditions are favorable, it can intensify into a tropical depression. When maximum sustained winds reach or exceed 39 mph, the system is designated a tropical storm, given a name, and is closely monitored by the National Hurricane Center in Miami, Florida. When sustained winds reach or exceed 74 mph, the storm is deemed a hurricane.

Hurricanes are categorized according to the strength and intensity of their winds using the Saffir-Simpson Hurricane Scale (Table 7-1). A Category 1 storm has the lowest wind speeds, while a Category 5 hurricane has the highest. However, a lower category storm can inflict greater damage than higher category storms depending on where they strike, the amount of storm surge, other weather they interact with, and how slow they move.

CATEGORY	MAXIMUM SUSTAINED WIND SPEED (Mph)	MINIMUM SURFACE PRESSURE (Millibars)	STORM SURGE (Feet)
1	74–95	Greater than 980	3-5
2	96-110	979 – 965	6-8
3	111-130	964 - 945	9-12
4	131 – 155	944 - 920	13-18
5	155 +	Less than 920	19+

#### Table 7-1. Extent Scale for Hurricanes<sup>1</sup>

Based on the historical storm tracks for hurricanes and tropical storms, as well as the location of Jefferson County, the average extent to be mitigated is a Category 4 storm. The Jefferson County planning area, including the SETRPC and all participating jurisdictions, has experienced wind speeds up to 155 mph, therefore a Category 4 should be mitigated in the event of a hurricane. Figure 7-2 displays the location of hurricane risk by storm category along the Gulf Coast.

<sup>&</sup>lt;sup>1</sup> Source: National Hurricane Center



Figure 7-2. Location of Hurricane Risk along the Texas Coast

The planning area is located along the coast, and therefore all participating jurisdictions including the SETRPC have a greater risk, with all land and buildings being vulnerable to all storms, category 1 through 5.

# **Historical Occurrences**

Previous occurrences include storms that had a direct path through the Jefferson County study area. Table 7-2 below lists the storms that have impacted the Jefferson County planning area during the years of 1996-2016.

#### Table 7-2. Historic Hurricane/Tropical Storm Events, 1996-2016<sup>2</sup>

YEAR	STORM NAME	CATEGORY	PROPERTY DAMAGE	CROP DAMAGE
1998	Charlie	Tropical Storm	\$88,430	\$0

<sup>&</sup>lt;sup>2</sup> Only events resulting in injury, fatality, or damages are listed.
## Section 7: Hurricane

YEAR	STORM NAME	CATEGORY PROPERTY DAMAGE		CROP DAMAGE
1998	Earl	Category 2	\$14,738	\$0
1998	Frances	Tropical Storm	\$7,369,202	\$0
2001	Allison	Tropical Storm	\$0	\$0
2005	Rita	Category 3	\$1,537,608,808	\$0
2007	Humberto	Category 1	\$28,966,153	\$0
2008	Edouard	Tropical Storm	\$278,951	\$0
2008	Ike	Category 2	\$669,482,544	\$0
TOTALS			\$2,243,808,826	\$0

Based on the list of historical hurricane events for the Jefferson County planning area (listed above), including the SETRPC and all participating jurisdictions, none of the events occurred after the 2011 Plan.

## Significant Past Events

#### Tropical Storm Frances, September 9-11, 1998 – Jefferson County

Tropical Storm Frances was the third tropical system to impact southeast Texas in 3 weeks, and caused the worst damage. Wind gusts in excess of 50 mph occurred along the coast on September 11th, but most of the damage occurred from the high tides. At Sabine Pass, the tide reading reached 5.3 ft. Mean Sea Level (MSL), which was one of the highest tides in the last 30 years. On top of the high tides, heavy rain lasting several days dropped 8 to 10 inches of rain across the region. At Sea Rim State Park, water got to the top of the dunes, which is 8 to 9 feet higher than normal.

Jefferson County incurred millions of dollars in storm damages, primarily a result of road damages. Highway 87 between Sabine Pass and Port Arthur received major damage. Sabine Pass was totally isolated from road traffic for three days due to high water. Nearly every home and business in Sabine Pass had salt water flooding (over 70 places).

#### Hurricane Rita, September 18-26, 2005 – Jefferson County

Hurricane Rita made landfall just east of the Texas – Louisiana border. The hurricane moved northwest and across southeast Texas in the morning hours of September 24th as a dangerous Category 3 hurricane with sustained winds of 120 mph. Along the Jefferson County coast, storm surges near 10 feet occurred near Sabine Pass, where over 90 percent of the homes were severely damaged or destroyed. The storm surge backed up the Sabine River, and flooded a small section of neighboring Orange with around 4 to 5 feet of storm surge. Winds blew over 100 mph across the entire region, snapping and uprooting trees, and damaged over 125,000 homes and businesses. Some homes in neighboring Jasper and Newton counties did not have electricity restored for over six weeks. Six fatalities were indirectly attributed to the storm in the City of Beaumont where a family died of carbon monoxide poisoning after running a generator inside their apartment.

# Probability of Future Events

Based on historical occurrences of significant hurricane events, the probability of future events is highly likely, with a hurricane event probable in the next year for the Jefferson County planning area, including the SETRPC and all participating jurisdictions.

# Vulnerability and Impact

Hurricanes and Tropical storms can cause major damage to large areas; hence all existing buildings, facilities and populations are equally exposed and vulnerable to this hazard and could potentially be impacted. The Jefferson County planning area features multiple mobile or manufactured home parks throughout the planning area and all participating jurisdictions. These parks are typically more vulnerable to hurricane events than typical site built structures. In addition, manufactured homes are located sporadically throughout the planning area, including all jurisdictions. These homes would also be more vulnerable. The U.S. Census data indicates a total of 3,138 manufactured homes located in the Jefferson County planning area, including all participating jurisdictions (Table 7-3). In addition, 65.4% (approximately 69,478 structures) of the single family residential (SFR) structures in the Jefferson County planning area were built before 1980.<sup>3</sup> These structures would typically be built to lower or less stringent construction standards than newer construction, and may be more susceptible to damages during significant events.

JURISDICTION	MANUFACTURED HOMES	SFR STRUCTURES BUILT BEFORE 1980
Beaumont	718	33,386
Bevil Oaks	27	361
China	87	249
Groves	69	5,388
Nederland	219	5,473
Nome	45	107
Port Arthur	234	16,809
Port Neches	82	3,887
SETRPC	0	0
Jefferson County <sup>4</sup>	3,138	69,478

#### Table 7-3. Structures at Greater Risk by Jurisdiction

<sup>&</sup>lt;sup>3</sup> Source: U.S. Census Bureau data estimates for 2014.

<sup>&</sup>lt;sup>4</sup> County totals include all participating jurisdictions and unincorporated areas.

## Section 7: Hurricane

The following critical facilities would be vulnerable to hurricane events in each participating jurisdiction, respectively.

JURISDICTION	CRITICAL FACILITIES
Jefferson County	Port Authority Facility, Fire Station, Water District Facility, 1 School
Beaumont	2 Fire Stations, 2 Police Stations, 32 Schools, Port Authority Facility, 5 Water District Facilities, 3 Drainage District Facilities, 4 Hospitals
Bevil Oaks	Fire Station
China	Fire Station, 2 Schools
Groves	Fire Station, Police Station, 4 Schools
Nederland	Fire Station, Police Station, Water District Facility, 3 Hospitals, Airport, 8 Schools
Nome	Fire Station
Port Arthur	Fire Station, Police Station, Drainage District Facility, 2 Port Authority Facilities, 2 Hospitals, 14 Schools
Port Neches	Fire Station, Police Station, 5 Schools
SETRPC	SETRPC Facility

#### Table 7-4. Critical Facilities by Jurisdiction

Table 7-5 shows impact or loss estimation for storms impacting the county. Damages are reported on a countywide basis and are not available for each participating jurisdiction. Annual loss estimates were based on the 21 year reporting period for such damages (Table 7-2). The average annual loss estimate for Jefferson County, which includes the SETRPC and all participating jurisdictions, is approximately \$106.85 million.

#### Table 7-5. Summary of Hurricane Events and Potential Annualized Losses, 1996-2016<sup>5</sup>

JURISDICTION	NUMBER OF EVENTS	PROPERTY & CROP LOSS	ANNUAL LOSS ESTIMATES
Jefferson County	11	\$2,243,808,827	\$106,848,039

The potential severity of impact from a hurricane for the Jefferson County planning area, including the SETRPC and all participating jurisdictions, is classified as substantial; meaning multiple deaths, complete shutdown of critical facilities and services for 30 days or more, and more than 50 percent of property would be destroyed or have major damage.

<sup>&</sup>lt;sup>5</sup> Values are in 2016 dollars.

## Assessment of Impacts

Hurricane events have the potential to pose a significant risk to people, and can create dangerous and difficult situations for public health and safety officials. Impacts to the planning area can include:

- Individuals exposed to the storm can be struck by flying debris, falling limbs, or downed trees, causing serious injury or death.
- Structures can be damaged or crushed by falling trees, which can result in physical harm to the occupants.
- Coastal communities may suffer substantial damage, requiring immediate shelter and long term displacement assistance.
- Driving conditions in all jurisdictions may be dangerous during a hurricane event, especially over elevated bridges, heightening the risk of injury and accidents during evacuations if not timed properly.
- Additional resources may be required for emergency preparedness and response during the summer months due to increases in populations along the coast.
- Emergency evacuations may be necessary prior to a hurricane landfall, requiring emergency responders, evacuation routing, and temporary shelters.
- Significant debris and downed trees can result in emergency response vehicles being unable to access areas of the community.
- Downed power lines may result in roadways being unsafe for use, which may prevent first responders from answering calls for assistance or rescue.
- During hurricane landfall, first responders may be prevented from responding to calls, as the winds may reach a speed in which their vehicles and equipment are unsafe to operate.
- Hurricane events often result in widespread power outages, increasing the risk to more vulnerable portions of the population who rely on power for health and/or life safety.
- Extended power outage often results in an increase in structure fires and carbon monoxide poisoning, as individuals attempt to cook or heat their homes with alternate, unsafe cooking or heating devices, such as grills.
- Extreme hurricane events may rupture gas lines and down trees and power lines, increasing the risk of structure fires during and after a storm event.
- Extreme hurricane events may lead to prolonged evacuations during search and rescue, in addition to immediate recovery efforts requiring additional emergency personnel and resources to prevent entry and protect citizens and property.
- First responders are exposed to downed power lines, unstable and unusual debris, hazardous materials, and generally unsafe conditions.
- Emergency operations and services may be significantly impacted due to damaged facilities and/or loss of communications.
- Critical staff may be unable to report for duty, limiting response capabilities.
- City or county departments may be damaged, delaying response and recovery efforts for the entire community.
- Private sector entities that the city and its residents rely on, such as utility providers, financial institutions, and medical care providers may not be fully operational and may require assistance from neighboring communities until full services can be restored.

## Section 7: Hurricane

- Economic disruption negatively impacts the programs and services provided by the community due to short and long term loss in revenue.
- Some businesses not directly damaged by the hurricane may be negatively impacted while roads are cleared and utilities are being restored, further slowing economic recovery.
- Older structures built to less stringent building codes may suffer greater damage as they are typically more vulnerable to hurricane damage.
- Large scale hurricanes can have significant economic impact on the affected area, as it must now fund expenses such as infrastructure repair and restoration, temporary services and facilities, overtime pay for responders, as well as normal day-to-day operating expenses.
- Businesses that are more reliant on utility infrastructure than others may suffer greater damages without a backup power source.

The economic and financial impacts of a hurricane on the area will depend entirely on the scale of the event, what is damaged, and how quickly repairs to critical components of the economy can be implemented. The level of preparedness and pre-event planning done by the county, communities, local businesses, and citizens will also contribute to the overall economic and financial conditions in the aftermath of any hurricane event.

# **SECTION 8: EXTREME HEAT**

Location
Extent
Historical Occurrences
Probability of Future Events
Vulnerability and Impact
Assessment of Impacts8

# Hazard Description

Extreme heat is the condition whereby temperatures hover ten degrees or more above the average high temperature in a region for an extended period. Extreme heat during the summer months is a common occurrence throughout the State of Texas, and Jefferson County is no exception. Severe, excessive summer heat is characterized by a combination of exceptionally high temperatures and humidity. When these conditions persist over a period of time, it is defined as a heat wave.



Jefferson County and all participating jurisdictions typically experience extended heat waves.

Although heat can damage buildings and facilities, it presents a more significant threat to the safety and welfare of citizens. The major human risks associated with severe summer heat include: heat cramps; sunburn; dehydration; fatigue; heat exhaustion; and even heat stroke. The most vulnerable populations to heat casualties are children and the elderly or infirmed, who frequently live on low fixed incomes and cannot afford to run air-conditioning on a regular basis. This population is sometimes isolated, with no immediate family or friends to look out for their well-being.

# Location

Two heat related deaths have been reported in the Jefferson County area, including one in 2008 and one in 2011<sup>1</sup>. In addition, there have been heat related deaths reported in neighboring counties, including Liberty and Chambers County. There is no specific geographic scope to the extreme heat hazard. Extreme heat could occur anywhere within the Jefferson County planning area including the SETRPC and all participating jurisdictions.

<sup>&</sup>lt;sup>1</sup> Sources: Texas Department of State Health Services (2008) and Beaumont Enterprise (2011)

# Extent

The magnitude or intensity of an extreme heat event is measured according to temperature in relation to the percentage of humidity. According to the National Oceanic Atmospheric Administration (NOAA), this relationship is referred to as the "Heat Index," and is depicted in Figure 8-1. This index measures how hot it feels outside when humidity is combined with high temperatures.



#### Figure 8-1. Extent Scale for Extreme Summer Heat<sup>2</sup>

#### Likelihood of Heat Disorders with Prolonged Exposure or Strenuous Activity

The Extent Scale in Figure 8-1 displays varying categories of caution depending on the relative humidity combined with the temperature. For example, when the temperature is at 90 degrees Fahrenheit (°F) or lower, caution should be exercised if the humidity level is at or above 40 percent.

The shaded zones on the chart indicate varying symptoms or disorders that could occur depending on the magnitude or intensity of the event. "Caution" is the first category of intensity and it indicates when fatigue due to heat exposure is possible. "Extreme Caution" indicates that sunstroke, muscle cramps, or heat exhaustion are possible, and a "Danger" level means that these symptoms are likely. "Extreme Danger" indicates that heat stroke is likely. The National Weather Service (NWS) initiates alerts based on the Heat Index as shown in Table 8-1.

<sup>&</sup>lt;sup>2</sup> Source: NOAA

CATEGORY	HEAT INDEX	POSSIBLE HEAT DISORDERS	WARNING TYPE	
Extreme Danger	125°F and higher	Heat stroke or sun stroke likely.		
Danger	103 – 124°F	Sunstroke, muscle cramps, and/or heat exhaustion are likely. Heatstroke possible with prolonged exposure and/or physical activity.	A heat advisory will be issued to warn that the Heat Index may exceed 105°F.	
Extreme Caution	90 – 103°F	Sunstroke, muscle cramps, and/or heat exhaustion possible with prolonged exposure and/or physical activity.	An Excessive Heat Warning is issued if the Heat Index rises above 105°F at least 3 hours	
Caution	80 – 90°F	Fatigue is possible with prolonged exposure and/or physical activity.	during the day or above 80°F at night.	

#### Table 8-1. Heat Index & Warnings

Jefferson County's terrain is relatively level terrain, with limited elevation variations located in Southeast Texas. The county features saltwater marshes along the southern border of the county adjacent to the Gulf of Mexico, the northern portion of the county is heavily forested with Southern Yellow Pine, and the area along the middle of the county is primarily coastal prairie.

Due to its geography, and its warm, sunny, and humid subtropical climate, the Jefferson County planning area can expect an extreme heat event each summer. Citizens, especially children and the elderly, should exercise caution by staying out of the heat for prolonged periods when a heat advisory or excessive heat warning is issued. Also at risk are those working or remaining outdoors.

Figure 8-2 displays the daily maximum heat index as derived from NOAA and based on data compiled from 1838 to 2015. The black circle shows the Jefferson County area. The brown and pink colors indicate a daily maximum heat index of 100-110° F. The Jefferson County planning area, including the SETRPC and all participating jurisdictions, could experience extreme heat from 90°F to 110°F and should mitigate to the extent of "danger", which can include sunstroke, muscle cramps, heat exhaustion, and potential heatstroke with prolonged exposure.



Figure 8-2. Average Daily Maximum Heat Index Days<sup>3</sup>

# **Historical Occurrences**

Every summer, the hazard of heat related illness becomes a significant public health issue throughout much of the United States. Mortality from all causes increases during heat waves, and excessive heat is an important contributing factor to deaths from other causes, particularly among the elderly. Preliminary data suggest that by August 21, 2009, record high summer temperatures in Texas resulted in more than 120 heat related deaths statewide. Table 8-2 depicts historical occurrences of mortality from heat from 1994 to 2004, sourced from the Texas Department of State Health Services, and 2005 to 2016, sourced from the NCEI database.

YEAR	DEATHS
1994	1
1995	12
1996	10

Table 8-2.	Extreme	Heat	Related	Deaths i	n Texas

<sup>&</sup>lt;sup>3</sup> Source: NCEI; the black circle indicates the Jefferson County planning area.

### Section 8: Extreme Heat

YEAR	DEATHS
1997	2
1998	66
1999	22
2000	71
2001	20
2002	1
2003	0
2004	3
2005	49
2006	2
2007	2
2008	7
2009	6
2010	4
2011	20
2012	2
2013	1
2014	0
2015	5
2016	1

Because the Texas Department of State Health Services reports on total events statewide, previous occurrences for extreme heat are derived from the NCEI database. According to heat related incidents located solely within Jefferson County, there is two heat waves<sup>4</sup> on record for Jefferson County (Table 8-3). Historical extreme heat information, as provided by the NCEI, shows extreme heat activity across a multi-county forecast area for each event. The appropriate percentage of the total property and crop damage reported for the entire forecast area has been allocated to each county impacted by each event. All participating jurisdictions including the SETRPC are reported under Jefferson County events. Only

<sup>&</sup>lt;sup>4</sup> Even though Jefferson County experiences heat waves each summer, NCEI data only records events reported. Based on reports, only two events are on record.

extreme heat events that have been reported have been factored into this Risk Assessment. It is likely additional extreme heat occurrences have gone unreported before and during the recording period.

JURISDICTION	DATE	DEATHS	INJURIES	PROPERTY DAMAGE	CROP DAMAGE
Jefferson County	8/29/2000	0	0	\$0	\$0
Jefferson County	9/1/2000	0	0	\$0	\$0
TOTALS		0	0	\$0	\$0

Table 8-3. Historical Extreme Heat Events, 1996-2016

Based on the list of historical extreme heat events for the Jefferson County planning area (listed above), including the SETRPC and all participating jurisdictions, none of the events occurred after the 2011 Plan.

# Probability of Future Events

According to historical records, the Jefferson County planning area has experienced 2 events in a 21 year reporting period. This provides a frequency of occurrence of 1 event every five years. This frequency supports an occasional probability of future events for the entire planning area including the SETRPC and all participating jurisdictions.

# Vulnerability and Impact

There is no defined geographic boundary for extreme heat events. While all of Jefferson County is exposed to extreme temperatures, existing buildings, infrastructure, and critical facilities are not likely to sustain significant damage from extreme heat events. Therefore, any estimated property losses associated with the extreme heat hazard are anticipated to be minimal across the area.

However, extreme temperatures do present a significant threat to life and safety for the population of the county as a whole. For example, heat casualties are typically caused by a lack of adequate air-conditioning or heat exhaustion. The most vulnerable populations to heat casualties are the elderly or infirmed, who frequently live on low fixed incomes and cannot afford to run air-conditioning on a regular basis. This population is sometimes isolated, with no immediate family or friends to look out for their well-being.

Populations over 65 in the Jefferson County planning area are estimated at 13% of the total population and children under the age of 5 exceed 6% or an estimated total of 50,074<sup>5</sup> potentially vulnerable residents in the planning area based on age (Table 8-4).

<sup>&</sup>lt;sup>5</sup> US Census Bureau 2014 data for Jefferson County

JURISDICTION	POPULATION 65 AND OLDER	POPULATION UNDER 5
Beaumont	15,539	8,087
Bevil Oaks	290	42
China	183	25
Groves	2,685	971
Nederland	2,371	976
Nome	56	11
Port Arthur	6,344	5,073
Port Neches	1,939	904
Jefferson County <sup>6</sup>	32,774	17,300

#### Table 8-4. Populations at Greater Risk by Jurisdiction

Another segment of the population at risk are those whose jobs consist of strenuous labor outdoors. Livestock and crops can become stressed, decreasing in quality or in production, during times of extreme heat. Extreme high temperatures can have significant secondary impacts, leading to droughts, water shortages, increased fire danger, and prompt excessive demands for energy. The possibility of rolling blackouts increases with unseasonably high temperatures in what is a normally mild month with low power demands.

Typically more than 12 hours of warning time would be given before the onset of an extreme heat event. Only minor property damage would result. The potential impact of excessive summer heat is considered "Minor" as injuries and/or illnesses do not result in permanent disability.

In terms of vulnerability to structures, the impact from extreme heat would be negligible. It is possible that critical facilities and infrastructure could be shut down for 24 hours if cooling units are running constantly, leading to a temporary power outage. Less than ten percent of residential and commercial property could be damaged if extreme heat events lead to structure fires.

The potential impact of extreme heat for the Jefferson County planning area, including the SETRPC and all participating jurisdictions, can be considered "Minor", resulting in few injuries and minimal disruption to the quality of life. Based on historical records over a 21 year period, annualized losses for the entire Jefferson County planning area are negligible.

<sup>&</sup>lt;sup>6</sup> County totals include all participating jurisdictions and unincorporated areas.

## Assessment of Impacts

The greatest risk from extreme heat is to public health and safety. Potential impacts to the community may include:

- Vulnerable populations, particularly the elderly and infants, can face serious or life-threatening health problems from exposure to extreme heat including hyperthermia; heat cramps; heat exhaustion; and heat stroke (or sunstroke).
- Response personnel including utility workers, public works personnel, and any other professions where individuals are required to work outside, are more subject to extreme heat related illnesses since their exposure would typically be greater.
- High energy demand periods can outpace the supply of energy, potentially creating the need for rolling brownouts which would elevate the risk of illness to vulnerable residents.
- Highways and roads may be damaged by excessive heat causing asphalt roads to soften and concrete roads to shift or buckle.
- Vehicle engines and cooling systems typically run harder during extreme heat events, resulting in increases in mechanical failures.
- Extreme heat events during times of drought can exacerbate the environmental impacts associated with drought, decreasing water and air quality, and further degrading wildlife habitat.
- Extreme heat increases ground-level ozone (smog), increasing the risk of respiratory illnesses.
- Tourism and recreational activities predominant in the Sabine Lake area and Sea Rim State Park may be negatively impacted during extreme heat events, reducing seasonal revenue.
- Food suppliers can anticipate an increase in food costs due to increases in production costs and crop and livestock losses.
- Fisheries may be negatively impacted by extreme heat, suffering damage to fish habitats (either natural or man-made), and a loss of fish and/or other aquatic organisms due to decreased water flows or availability.
- Negatively impacted water suppliers may face increased costs resulting from the transport water or develop supplemental water resources.
- Outdoor activities may see an increase in injury or illness during extreme heat events.

The economic and financial impacts of extreme heat on the community will depend on the duration of the event, demand for energy, drought associated with extreme heat, and many other factors. The level of preparedness and the amount of planning done by the jurisdiction, local businesses, and citizens will impact the overall economic and financial conditions before, during, and after an extreme heat event.

# **SECTION 9: HAIL**

Hazard Description1	
Location1	
Extent1	
Historical Occurrences	
Significant Past Events6	
Probability of Future Events7	,
Vulnerability and Impact7	,
Assessment of Impacts9	

# Hazard Description



Hailstorm events are a potentially damaging outgrowth of severe thunderstorms. During the developmental stages of a hailstorm, ice crystals form within a low pressure front due to the rapid rising of warm air into the upper atmosphere, and the subsequent cooling of the air mass. Frozen droplets gradually accumulate into ice crystals until they fall as round or irregularly shaped masses of ice typically greater than 0.75 inches in diameter. The size of hailstones is a direct result of the size and severity of the storm. High velocity updraft winds are required to keep hail in suspension in thunderclouds. The strength of the updraft is a by-product of heating on the Earth's surface. Higher temperature gradients above Earth's surface result in increased suspension time and hailstone size.

# Location

Hailstorms are an extension of severe thunderstorms that could potentially cause severe damage. As a result, they are not confined to any specific geographic location, and can vary greatly in size, location, intensity, and duration. Therefore, the Jefferson County planning area, including the SETRPC and all participating jurisdictions, is equally at risk to the hazard of hail.

# Extent

The National Weather Service (NWS) classifies a storm as "severe" if there is hail 0.75 inches in diameter (approximately the size of a penny) or greater, based on radar intensity or as seen by observers. The intensity category of a hailstorm depends on hail size and the potential damage it could cause, as depicted in the National Center for Environmental Information (NCEI) Intensity Scale in Table 9-1.

SIZE CODE	INTENSITY CATEGORY	SIZE (Diameter Inches)	DESCRIPTIVE TERM	TYPICAL DAMAGE
НО	Hard Hail	Up to 0.33	Реа	No damage
H1	Potentially Damaging	0.33 - 0.60	Marble	Slight damage to plants and crops
H2	Potentially Damaging	0.60 - 0.80	Dime	Significant damage to plants and crops
H3	Severe	0.80 - 1.20	Nickel	Severe damage to plants and crops
Н4	Severe	1.2 – 1.6	Quarter	Widespread glass and auto damage
H5	Destructive	1.6 - 2.0	Half Dollar	Widespread destruction of glass, roofs, and risk of injuries
Н6	Destructive	2.0 - 2.4	Ping Pong Ball	Aircraft bodywork dented and brick walls pitted
H7	Very Destructive	2.4 - 3.0	Golf Ball	Severe roof damage and risk of serious injuries
H8	Very Destructive	3.0 - 3.5	Hen Egg	Severe damage to all structures
Н9	Super Hailstorms	3.5 - 4.0	Tennis Ball	Extensive structural damage, could cause fatal injuries
H10	Super Hailstorms	4.0 +	Baseball	Extensive structural damage, could cause fatal injuries

Table 9-1.	Hail Intensity	v and Magnitude <sup>1</sup>
10010 0 21		

The intensity scale in Table 9-1 ranges from H0 to H10, with increments of intensity or damage potential in relation to hail size (distribution and maximum), texture, fall speed, speed of storm translation, and strength of the accompanying wind. Based on available data regarding the previous occurrences for the area, the Jefferson County planning area, including the SETRPC and all participating jurisdictions, may experience hailstorms ranging from an H0 to an H7. Jefferson County can mitigate a storm from low risk (hard hail) to a serious hailstorm with golf ball sized hail that leads to severe roof damage and could cause serious injuries.

# Historical Occurrences

Historical evidence shown in Figure 9-1 demonstrates that the planning area is vulnerable to hail events overall, which typically result from severe thunderstorm activity. Only those events for Jefferson County, including the SETRPC and all participating jurisdictions, with latitude and longitude available were plotted (Figure 9-1). Historical events with reported damages, injuries, or fatalities are shown in Table 9-2. A total of 58 reported historical hail events impacted Jefferson County between 1996 and August 2016 (Table 9-3). These events were reported to NCEI and NOAA databases, and may not represent all hail events to have occurred during the past 21 years. It is important to note that the SETRPC is located within the City

<sup>&</sup>lt;sup>1</sup> Source: NCEI Intensity Scale, based on the TORRO Hailstorm Intensity Scale.

of Beaumont. There may be some occurrences that have occurred for the SETRPC and may not have been recorded, but are included in the City of Beaumont occurrence data because of their location. Only hail events that have been reported have been factored into this Risk Assessment. It is likely that additional hail occurrences have gone unreported before and during the recording period.



Figure 9-1. Spatial Historical Hail Events, 1996–2016

Jefferson County | Hazard Mitigation Plan Update 2017 | Page 3

JURISDICTION	Date	MAGNITUDE	INJURIES	FATALITIES	PROPERTY DAMAGE	CROP DAMAGE
Beaumont	4/12/1996	1.75	0	0	\$0	\$0
Beaumont	9/9/1997	0.75	0	0	\$0	\$0
Jefferson County	1/21/1998	0.75	0	0	\$0	\$0
Jefferson County	1/21/1998	1	0	0	\$0	\$0
Port Arthur	1/21/1998	0.75	0	0	\$0	\$0
Port Neches	3/7/1998	0.75	0	0	\$0	\$0
Nederland	7/17/1998	0.88	0	0	\$0	\$0
Port Arthur	5/11/1999	1.75	0	0	\$0	\$0
Jefferson County	8/3/1999	0.75	0	0	\$0	\$0
Beaumont	8/31/1999	0.75	0	0	\$0	\$0
Jefferson County	4/2/2000	1.75	0	0	\$0	\$0
Beaumont	4/2/2000	0.75	0	0	\$0	\$0
Port Arthur	4/2/2000	1.75	0	0	\$0	\$0
Jefferson County	4/3/2000	0.88	0	0	\$0	\$0
Bevil Oaks	9/1/2000	1.75	0	0	\$0	\$0
Jefferson County	9/2/2000	0.75	0	0	\$0	\$0
Nederland	7/1/2001	0.75	0	0	\$0	\$0
Nederland	3/13/2003	0.75	0	0	\$0	\$0
Jefferson County	4/7/2003	0.75	0	0	\$0	\$0
Port Arthur	4/7/2003	2.75	0	0	\$0	\$0
Port Arthur	4/8/2003	0.88	0	0	\$0	\$0
Beaumont	4/20/2003	1.5	0	0	\$0	\$0
China	4/20/2003	0.75	0	0	\$0	\$0
Port Arthur	8/21/2003	0.88	0	0	\$0	\$0
Port Arthur	4/30/2004	1	0	0	\$0	\$0

### Table 9-2. Historical Hail Events, 1996-2016<sup>2</sup>

<sup>2</sup> Damages reported in 2016 dollars.

## Section 9: Hail

JURISDICTION	Date	MAGNITUDE	INJURIES	FATALITIES	PROPERTY DAMAGE	CROP DAMAGE
Groves	5/13/2004	0.75	0	0	\$0	\$0
Nederland	5/13/2004	0.88	0	0	\$0	\$0
Port Arthur	5/13/2004	0.75	0	0	\$0	\$0
Beaumont	9/18/2004	0.75	0	0	\$0	\$0
Beaumont	6/15/2005	0.75	0	0	\$0	\$0
Bevil Oaks	7/11/2005	0.88	0	0	\$0	\$0
Beaumont	12/4/2005	0.75	0	0	\$0	\$0
Groves	6/14/2007	0.75	0	0	\$0	\$0
Jefferson County	12/20/2007	0.75	0	0	\$0	\$0
Jefferson County	2/12/2008	0.75	0	0	\$0	\$0
Groves	5/22/2008	0.88	0	0	\$0	\$0
Beaumont	3/25/2009	0.75	0	0	\$0	\$0
Bevil Oaks	3/27/2009	0.88	0	0	\$0	\$0
Bevil Oaks	3/27/2009	1.5	0	0	\$0	\$0
Beaumont	2/26/2010	0.75	0	0	\$0	\$0
Jefferson County	3/29/2011	0.75	0	0	\$0	\$0
Jefferson County	6/6/2011	0.75	0	0	\$0	\$0
Beaumont	6/6/2011	1	0	0	\$0	\$0
Groves	6/6/2011	0.75	0	0	\$0	\$0
Port Arthur	6/6/2011	0.75	0	0	\$0	\$0
Port Arthur	6/6/2011	0.75	0	0	\$0	\$0
Port Arthur	9/29/2011	1	0	0	\$0	\$0
Groves	4/2/2012	0.75	0	0	\$0	\$0
Beaumont	6/8/2012	1	0	0	\$0	\$0
China	5/10/2013	0.75	0	0	\$0	\$0
Jefferson County	5/22/2013	1	0	0	\$0	\$0
Bevil Oaks	7/12/2013	0.88	0	0	\$0	\$0
Beaumont	7/3/2014	0.88	0	0	\$0	\$0

Jefferson County | Hazard Mitigation Plan Update 2017 | Page 5

## Section 9: Hail

JURISDICTION	Date	MAGNITUDE	INJURIES	FATALITIES	PROPERTY DAMAGE	CROP DAMAGE
Jefferson County	4/19/2015	1	0	0	\$0	\$0
Jefferson County	4/27/2015	1	0	0	\$0	\$0
Beaumont	4/27/2015	1	0	0	\$0	\$0
Port Arthur	4/27/2015	1.75	0	0	\$0	\$0
Port Arthur	1/8/2016	1.25	0	0	\$0	\$0

#### Table 9-3. Historical Hail Events Summary, 1996-2016<sup>3</sup>

JURISDICTION	Number of Events	MAGNITUDE	INJURIES	FATALITIES	PROPERTY DAMAGE	CROP DAMAGE
Jefferson County	20	2.75 inches	0	0	\$0	\$0
Beaumont	10	1.75 inches	0	0	\$0	\$0
Bevil Oaks	5	1.75 inches	0	0	\$0	\$0
China	2	0.75 inches	0	0	\$0	\$0
Groves	5	0.88 inches	0	0	\$0	\$0
Nederland	4	0.88 inches	0	0	\$0	\$0
Nome	0	N/A	0	0	\$0	\$0
Port Arthur	11	1.75 inches	0	0	\$0	\$0
Port Neches	1	0.75 inches	0	0	\$0	\$0
SETRPC	0	N/A	0	0	\$0	\$0
TOTAL LOSSES	52	(Max Extent)	0	0	\$0	)

Based on the list of historical hail events for the Jefferson County planning area (listed above), including the SETRPC and all participating jurisdictions, 18 events have occurred since the 2011 Plan.

## Significant Past Events

#### May 11, 1999 – Port Arthur

On May 11, 1999 a hail storm brought half dollar size hail to the City of Port Arthur. Hail of sizes up to 1.75 inches damaged several cars in the area.

#### March 25-27, 2009 – Jefferson County/Bevil Oaks

A 3-day series of severe weather across southeast Texas began when a squall line of severe thunderstorms developed across central Texas during the evening hours on March 25,2009 and moved rapidly eastward

<sup>&</sup>lt;sup>3</sup> Values are in 2016 dollars.

across Southeast Texas. A few reports of wind damage and large hail were received. An industrial plant along the Neches River ESE of Beaumont reported penny size hail. A Beaumont TV station reported nickel size hail in Bevil Oaks.

# Probability of Future Events

Based on available records of historic events, 58 events in a 21 year reporting period for the Jefferson County planning area provides a frequency of occurrence of 2 to 3 events every year. This frequency supports a "highly likely" probability of future events for the entire planning area including the SETRPC and all participating jurisdictions. The numbers listed for the jurisdictions within the County are historical events that are known to have specifically impacted those jurisdictions.

# Vulnerability and Impact

Damage from hail approaches \$1 billion in the U.S. each year. Much of the damage inflicted by hail impacts crops. Even relatively small hail can shred plants to ribbons in a matter of minutes. Vehicles, roofs of buildings and homes, and landscaping are also most commonly damaged by hail.

Utility systems on roofs at school districts and critical facilities would be vulnerable and could be damaged. Hail could cause a significant threat to people as they could be struck by hail and falling trees and branches. Outdoor activities and events may elevate the risk to residents and visitors in the planning area when a hailstorm strikes with little warning. Older structures not built to current codes may be more vulnerable to damages than newer structures.

The Jefferson County planning area features multiple mobile or manufactured home parks throughout the planning area and all participating jurisdictions. These parks are typically more vulnerable to hail events than typical site built structures. In addition, manufactured homes are located sporadically throughout the planning area, including all jurisdictions. These homes would also be more vulnerable. The U.S. Census data indicates a total of 3,138 manufactured homes located in the Jefferson County planning area, including all participating jurisdictions (Table 9-4). In addition, 65.4% (approximately 69,478 structures) of the single family residential (SFR) structures in the Jefferson County planning area were built before 1980.<sup>4</sup> These structures would typically be built to lower or less stringent construction standards than newer construction and may be more susceptible to damages during significant hail events.

JURISDICTION	MANUFACTURED HOMES	SFR STRUCTURES BUILT BEFORE 1980
Beaumont	718	33,386
Bevil Oaks	27	361
China	87	249

#### Table 9-4. Structures at Greater Risk by Jurisdiction

<sup>&</sup>lt;sup>4</sup> Source: U.S. Census Bureau data estimates for 2014.

## Section 9: Hail

JURISDICTION	MANUFACTURED HOMES	SFR STRUCTURES BUILT BEFORE 1980
Groves	69	5,388
Nederland	219	5,473
Nome	45	107
Port Arthur	234	16,809
Port Neches	82	3,887
SETPRC	0	0
Jefferson County⁵	3,138	69,478

The following critical facilities would be vulnerable to hail events in each participating jurisdiction:

#### Table 9-5. Critical Facilities by Jurisdiction

JURISDICTION	CRITICAL FACILITIES
Jefferson County	Port Authority Facility, Fire Station, Water District Facility, 1 School
Beaumont	2 Fire Stations, 2 Police Stations, 32 Schools, Port Authority Facility, 5 Water District Facilities, 3 Drainage District Facilities, 4 Hospitals
Bevil Oaks	Fire Station
China	Fire Station, 2 Schools
Groves	Fire Station, Police Station, 4 Schools
Nederland	Fire Station, Police Station, Water District Facility, 3 Hospitals, Airport, 8 Schools
Nome	Fire Station
Port Arthur	Fire Station, Police Station, Drainage District Facility, 2 Port Authority Facilities, 2 Hospitals, 14 Schools
Port Neches	Fire Station, Police Station, 5 Schools
SETRPC	SETRPC Facility

First responders could not be able to respond to calls due to blocked roads. Also, hail could cause power outages which could cause health and safety risks to more vulnerable populations in the planning area.

Hail has been known to cause injury to humans, and occasionally has been fatal. There are no reported damages to crops or property in the planning area, including the SETRPC and all participating jurisdictions. Based on historic loss and damages, the impact of hail damages on the Jefferson County planning area,

<sup>&</sup>lt;sup>5</sup> County totals includes all participating jurisdictions and unincorporated areas.

## Section 9: Hail

including the SETRPC and all participating jurisdictions, can be considered "Limited". This severity of impact indicates minor injuries that are treatable with first aid, Jefferson County area facilities shut down for 24 hours or less, and less than 10% of property destroyed or with major damage.

JURISDICTION	PROPERTY & CROP DAMAGE	ANNUAL LOSS ESTIMATE
Jefferson County	\$0	\$0
Beaumont	\$0	\$0
Bevil Oaks	\$0	\$0
China	\$0	\$0
Groves	\$0	\$0
Nederland	\$0	\$0
Nome	\$0	\$0
Port Arthur	\$0	\$0
Port Neches	\$0	\$0
SETRPC	\$0	\$0
Planning Area	<b>\$0</b>	\$0

#### Table 9-4. Potential Annualized Losses by Jurisdiction, 1996-2016

## **Assessment of Impacts**

Hail events have the potential to pose a significant risk to people and can create dangerous situations. Impacts to the planning area can include:

- Hail may create hazardous road conditions during and immediately following an event, delaying first responders from preserving or providing for public health and safety.
- Individuals and first responders who are exposed to the storm may be struck by hail, falling branches, or downed trees resulting in injuries or possible fatalities.
- Residential structures can be damaged by falling trees, which can result in physical harm to occupants.
- Large hail events will likely cause extensive roof damage to residential structures, along with siding damage and broken windows, creating a spike in insurance claims and a rise in premiums.
- Automobile damage may be extensive depending on the size of the hail and length of the storm.
- Hail events can result in power outages over widespread areas, increasing the risk to more vulnerable portions of the population who rely on power for health and/or life safety.
- Extended power outage can result in an increase in structure fires and/or carbon monoxide poisoning, as individuals attempt to cook or heat their home with alternate, unsafe cooking or heating devices, such as grills.

- First responders are exposed to downed power lines, damaged structures, hazardous spills, and debris that often accompany hail events, elevating the risk of injury to first responders and potentially diminishing emergency response capabilities.
- Downed power lines and large debris, such as downed trees, can result in the inability of emergency response vehicles to access areas of the community.
- Hazardous road conditions may prevent critical staff from reporting for duty, limiting response capabilities.
- Economic disruption negatively impacts the programs and services provided by the community due to short and long term loss in revenue.
- Some businesses not directly damaged by the hail event may be negatively impacted while roads are cleared and utilities are being restored, further slowing economic recovery.
- Businesses that are more reliant on utility infrastructure than others may suffer greater damages without a backup power source.
- Hazardous road conditions will likely lead to increases in automobile accidents, further straining emergency response capabilities.
- Depending on the severity and scale of damage caused by large hail events, damage to power transmission and distribution infrastructure can require days or weeks to repair.
- A significant hail event could significantly damage agricultural crops, resulting in extensive economic losses for the community and surrounding area.
- Hail events may injure or kill livestock and wildlife.
- A large hail event could impact the accessibility of recreational areas and parks due to extended power outages or debris clogged access roads.

The economic and financial impacts of hail will depend entirely on the scale of the event, what is damaged, and how quickly repairs to critical components of the economy can be implemented. The level of preparedness and pre-event planning conducted by the community, local businesses, and citizens will contribute to the overall economic and financial conditions in the aftermath of any hail event.

# **SECTION 10: THUNDERSTORM WIND**

Hazard Description	1
Location	1
Extent	2
Historical Occurrences	3
Significant Past Events	7
Probability of Future Events	8
Vulnerability and Impact	8
Assessment of Impacts	. 10

# Hazard Description

Thunderstorms create extreme wind events which includes straight line winds. Wind is the horizontal motion of the air past a given point, beginning with differences in air pressures. Pressure that is higher at one place than another sets up a force pushing from the high pressure toward the low pressure; the greater the difference in pressures, the stronger the force. The distance between the area of high pressure and the area of low pressure also determines how fast the moving air is accelerated.

Thunderstorms are created when heat and moisture near the Earth's surface are transported to the upper levels of the atmosphere. By-products of this process are the clouds, precipitation, and wind that become the thunderstorm.

According to the National Weather Service (NWS), a thunderstorm occurs when thunder accompanies rainfall. Radar observers use the intensity of radar echoes to distinguish between rain showers and thunderstorms.



Straight line winds can have gusts of 100 miles per hour (mph) or more. Unlike tornadoes, windstorms have a broader path that is several miles wide and can cover several counties. Straight line wind may down trees and power lines, overturn mobile homes, and cause damage to well-built structures.

Straight line winds are responsible for most thunderstorm wind damages. One type of straight line wind, the downburst, is a small area of rapidly descending air beneath a thunderstorm. A downburst can cause damage equivalent to a strong tornado and make air travel extremely hazardous.

# Location

Thunderstorm wind events can develop in any geographic location and are considered a common occurrence in Texas. Therefore, a thunderstorm wind event could occur at any location within Jefferson County's planning area, as these storms develop randomly and are not confined to any geographic area

within the County. It is assumed that the Jefferson County planning area, including the SETRPC and all participating jurisdictions, is uniformly exposed to the threat of thunderstorm winds.

## Extent

The extent or magnitude of a thunderstorm wind event is measured by the Beaufort Wind Scale. Table 10-1 describes the different intensities of wind in terms of speed and effects, from calm to violent and destructive.

FORCE	WIND (KNOTS)	WMO CLASSIFICATION	APPEARANCE OF WIND EFFECTS
0	Less than 1	Calm	Calm, smoke rises vertically
1	1-3	Light Air	Smoke drift indicates wind direction, still wind vanes
2	4-7	Light Breeze	Wind felt on face, leaves rustle, vanes begin to move
3	8-12	Gentle Breeze	Leaves and small twigs constantly moving, light flags extended
4	13-18	Moderate Breeze	Dust, leaves and loose paper lifted, small tree branches move
5	19-24	Fresh Breeze	Small trees in leaf begin to sway
6	25-31	Strong Breeze	Larger tree branches moving, whistling in wires
7	32-38	Near Gale	Whole trees moving, resistance felt walking against wind
8	39-46	Gale	Whole trees in motion, resistance felt walking against wind
9	47-54	Strong Gale	Slight structural damage occurs, slate blows off roofs
10	55-63	Storm	Seldom experienced on land, trees broken or uprooted, considerable structural damage
11	64-72	Violent Storm	If experienced on land, widespread damage
12	73+	Hurricane	Violence and destruction

#### Table 10-1. Beaufort Wind Scale<sup>1</sup>

Figure 10-1 displays the wind zones as derived from the National Oceanic and Atmospheric Administration (NOAA).

<sup>&</sup>lt;sup>1</sup> Source: World Meteorological Organization



Figure 10-1. Wind Zones in the United States<sup>2</sup>

On average, the planning area experiences 3 to 4 thunderstorm wind events every year. The County is located within the Zone III, meaning the entire planning area including the SETRPC and all participating jurisdictions can experience winds up to 200 mph. Jefferson County has experienced a significant wind event – an event with winds above 64 knots in the range of "Force 11" on the Beaufort Wind Scale.

# **Historical Occurrences**

Tables 10-2, 10-3, and 10-4 depict historical occurrences of thunderstorm wind events for the Jefferson County planning area according to the National Center for Environmental Information (NCEI) data. Since January 1996, 77 thunderstorm wind events are known to have impacted Jefferson County, based upon NCEI records. Table 10-3 presents information on known historical events impacting the Jefferson County planning area, with resulting damages. It is important to note that high wind events associated with other hazards, such as tornadoes, are not accounted for in this section.

The NCEI is a national data source organized under NOAA and is the largest archive available for climate data. Only NCEI reported incidents were factored into this risk assessment. It is important to note that the

<sup>&</sup>lt;sup>2</sup> Source: NOAA; the black circle indicates the Jefferson County planning area.

SETRPC is located within the City of Beaumont. There may be some occurrences that have occurred for the SETRPC and may not have been recorded, but are included in the City of Beaumont occurrence data because of their location. In the tables that follow throughout this section, some occurrences seem to appear multiple times in one table. This is due to reports from various locations throughout the County. In addition, property damage estimates are not always reported. When this occurs, estimates are provided when reasonable. Where an estimate has been provided in a table for losses, the dollar amounts have been altered to indicate the damage in 2016 dollars.

MAXIMUM WIND SPEED RECORDED (KNOTS)	NUMBER OF REPORTED EVENTS
0-30	0
31-40	0
41-50	19
51-60	20
61-70	4
71-80	0
81-90	0
91-100	0
Unknown	34

#### Table 10-2. Historical Thunderstorm Wind Events, With Reported Damages, 1996-2016

#### Table 10-3. Historical Thunderstorm Wind Events, 1996-2016<sup>3</sup>

JURISDICTION	DATE	TIME	MAGNITUDE	DEATHS	INJURIES	PROPERTY DAMAGE	CROP DAMAGE
Beaumont	5/29/1996	9:15 PM	Unknown	0	0	\$15,249	\$0
Beaumont	8/12/1996	10:40 PM	Unknown	0	0	\$7,625	\$0
China	4/5/1997	2:39 AM	Unknown	0	0	\$7,454	\$0
Beaumont	8/21/1997	5:30 PM	Unknown	0	0	\$29,814	\$0
Groves	8/22/1997	4:25 PM	Unknown	0	0	\$14,907	\$0
Nederland	12/3/1997	5:12 AM	Unknown	0	1	\$74,536	\$0
Port Arthur	2/10/1998	2:30 PM	Unknown	0	0	\$29,357	\$0
Port Arthur	2/10/1998	2:40 PM	57 knots	0	0	\$14,679	\$0

<sup>3</sup> Only recorded events with fatalities, injuries, or damages are listed. Magnitude is listed when available. Damage values are in 2016 dollars.

JURISDICTION	DATE	TIME	MAGNITUDE	DEATHS	INJURIES	PROPERTY DAMAGE	CROP DAMAGE
Nederland	2/22/1998	12:30 AM	Unknown	0	0	\$14,679	\$0
Port Arthur	3/16/1998	1:00 PM	Unknown	0	0	\$146,786	\$0
China	7/14/1998	1:50 PM	Unknown	0	0	\$220,179	\$0
Beaumont	8/13/1998	3:52 PM	Unknown	0	0	\$22,018	\$0
Beaumont	8/14/1998	12:00 PM	Unknown	1	1	\$110,089	\$0
Beaumont	8/29/1998	8:00 PM	Unknown	0	0	\$73,393	\$0
Beaumont	5/10/1999	7:00 AM	Unknown	0	0	\$143,614	\$0
Beaumont	5/29/1999	11:10 AM	Unknown	0	0	\$7,181	\$0
Beaumont	7/12/1999	1:50 PM	Unknown	0	0	\$7,181	\$0
Groves	8/3/1999	4:50 PM	Unknown	0	0	\$143,614	\$0
Beaumont	8/20/1999	6:30 PM	Unknown	0	0	\$71,807	\$0
Beaumont	8/31/1999	1:15 PM	Unknown	0	0	\$35,904	\$0
Port Arthur	4/3/2000	3:30 AM	52 knots	0	0	\$27,789	\$0
Beaumont	7/23/2000	2:05 PM	Unknown	0	0	\$2,779	\$0
Jefferson County	8/11/2000	3:30 PM	Unknown	0	0	\$2,779	\$0
Bevil Oaks	9/2/2000	6:55 PM	Unknown	0	0	\$2,779	\$0
Nederland	9/2/2000	5:50 PM	Unknown	0	0	\$2,779	\$0
Nome	2/28/2001	6:30 AM	Unknown	0	0	\$33,775	\$0
Port Arthur	3/14/2001	6:15 PM	Unknown	0	0	\$6,755	\$0
Port Arthur	3/14/2001	5:45 PM	63 knots	0	0	\$6,755	\$0
Beaumont	5/26/2001	3:00 PM	Unknown	0	0	\$13,510	\$0
China	10/11/2001	11:30 AM	Unknown	0	0	\$13,510	\$0
Jefferson County	5/17/2002	8:17 AM	Unknown	0	0	\$33,249	\$0
Groves	7/16/2002	8:00 AM	65 knots	0	0	\$1,994,950	\$0
Beaumont	8/26/2002	6:10 PM	Unknown	0	0	\$33,249	\$0
Beaumont	8/27/2002	12:00 PM	Unknown	0	0	\$6,650	\$0
Beaumont	10/28/2002	11:25 PM	Unknown	0	0	\$6,650	\$0
Nederland	10/29/2002	1:07 AM	Unknown	0	0	\$6,650	\$0

Jefferson County | Hazard Mitigation Plan Update 2017 | Page 5

JURISDICTION	DATE	TIME	MAGNITUDE	DEATHS	INJURIES	PROPERTY DAMAGE	CROP DAMAGE
Port Arthur	12/23/2002	11:15 PM	Unknown	0	0	\$6,650	\$0
Port Arthur	12/30/2002	10:45 PM	Unknown	0	0	\$6,650	\$0
Beaumont	5/11/2004	5:00 PM	50 knots	0	0	\$31,665	\$0
Groves	11/27/2004	12:25 AM	50 knots	0	0	\$25,332	\$0
Beaumont	5/29/2005	7:45 PM	50 knots	0	0	\$30,627	\$0
Beaumont	6/15/2005	5:20 PM	50 knots	0	0	\$2,450	\$0
China	8/7/2005	5:15 PM	50 knots	0	0	\$2,450	\$0
Jefferson County	4/29/2006	11:39 AM	50 knots	0	0	\$59,341	\$0
Jefferson County	7/17/2006	5:00 PM	50 knots	0	0	\$5,934	\$0
Beaumont	8/27/2007	4:50 PM	50 knots	0	0	\$23,079	\$0
Beaumont	8/27/2007	4:10 PM	50 knots	0	0	\$2,308	\$0
Jefferson County	6/25/2008	11:55 AM	50 knots	0	0	\$5,556	\$0
Jefferson County	8/3/2008	5:00 PM	57 knots	0	0	\$11,113	\$0
Beaumont	2/1/2009	1:50 PM	50 knots	0	0	\$5,576	\$0
Beaumont	7/18/2009	4:25 PM	52 knots	0	0	\$2,230	\$0
Beaumont	8/16/2010	12:45 PM	52 knots	0	0	\$1,097	\$0
Beaumont	8/16/2010	12:40 PM	52 knots	0	0	\$5,486	\$0
Beaumont	8/16/2010	12:40 PM	52 knots	0	1	\$27,431	\$0
Port Arthur	8/16/2010	1:30 PM	52 knots	0	0	\$1,097	\$0
Jefferson County	8/23/2010	6:30 PM	52 knots	0	0	\$3,292	\$0
Beaumont	3/30/2011	3:00 AM	48 knots	0	0	\$1,064	\$0
Groves	10/31/2013	8:33 AM	61 knots	0	0	\$5,135	\$0
Groves	4/27/2016	5:44 AM	50 knots	0	0	\$5,000	\$0

JURISDICTION	NUMBER OF EVENTS	MAGNITUDE	DEATHS	INJURIES	PROPERTY DAMAGE	CROP DAMAGE
Jefferson County	8	57 knots	0	0	\$121,263	\$0
Beaumont	35	63 knots	1	2	\$719,727	\$0
Bevil Oaks	1	Unknown	0	0	\$2,779	\$0
China	4	50 knots	0	0	\$243,593	\$0
Groves	6	65 knots	0	0	\$2,188,939	\$0
Nederland	4	Unknown	0	1	\$98,643	\$0
Nome	1	Unknown	0	0	\$33,775	\$0
Port Arthur	17	63 knots	0	0	\$246,517	\$0
Port Neches	1	50 knots	0	0	\$0	\$0
SETRPC	0	N/A	0	0	\$0	\$0
TOTAL LOSSES	77	50-65 knots	1	3	\$3,655,	,236

Table 10-4. Summary of Historical Thunderstorm Wind Events, 1996-2016<sup>4</sup>

Based on the list of historical thunderstorm wind events for the Jefferson County planning area (listed above), including the SETRPC and all participating jurisdictions, 7 events have occurred since the 2011 Plan.

## Significant Past Events

### July 14, 1998 – China

Severe thunderstorm winds blew down trees across China in western Jefferson County. A home under construction was leveled due to the estimated 60 mph winds. Houses and barns next door were not damaged. The China Elementary School had parts of its roof torn off, and debris from the roof went through the windows of a classroom. No injuries were reported.

#### August 14, 1998 – Beaumont

Two men were in a house under construction when a combination of wind and rain caused the collapse of the building. One man was killed, and the other was slightly injured.

### July 16, 2002 – Groves

High winds associated with a severe thunderstorm blew down numerous trees and power lines. The most significant damage was to the Groves Middle School, where the gymnasium lost part of its roof and one wall.

<sup>&</sup>lt;sup>4</sup> Values are in 2016 dollars.

#### April 29, 2006 – Jefferson County

Severe thunderstorm winds produced damages in the county estimated at \$50,000. An old rice dryer was wrapped around a utility pole. A small building was moved 2 feet off its blocks.

# Probability of Future Events

Most thunderstorm winds occur during the spring, in the months of March, April, and May, and in the fall, during the month of September. Based on available records of historic events, 77 events in a 21 year reporting period provides a frequency of occurrence of 3 to 4 events every year. Even though the intensity of thunderstorm wind events is not always damaging for the Jefferson County planning area, the frequency of occurrence for a thunderstorm wind event is highly likely, meaning that an event is probable within the next year for the Jefferson County planning area, including the SETRPC and all participating jurisdictions.

# Vulnerability and Impact

Vulnerability is difficult to evaluate since thunderstorm wind events can occur at different strength levels, in random locations, and can create relatively narrow paths of destruction. Due to the randomness of these events, all existing and future structures, and facilities in Jefferson County could potentially be impacted and remain vulnerable to possible injury and property loss from strong winds.

Trees, power lines and poles, signage, manufactured housing, radio towers, concrete block walls, storage barns, windows, garbage receptacles, brick facades, and vehicles, unless reinforced, are vulnerable to thunderstorm wind events. The Jefferson County planning area features multiple mobile or manufactured home parks throughout the planning area and all participating jurisdictions. These parks are typically more vulnerable to thunderstorm wind events than typical site built structures. In addition, manufactured homes are located sporadically throughout the planning area, including all jurisdictions. These homes would also be more vulnerable. The U.S. Census data indicates a total of 3,138 manufactured homes located in the Jefferson County planning area, including all participating jurisdictions (Table 10-5). In addition, 65.4% (approximately 69,478 structures) of the single family residential (SFR) structures in the Jefferson County planning area were built before 1980.<sup>5</sup> These structures would typically be built to lower or less stringent construction standards than newer construction and may be more susceptible to damages during significant thunderstorm wind events.

JURISDICTION	MANUFACTURED HOMES	SFR STRUCTURES BUILT BEFORE 1980
Beaumont	718	33,386
Bevil Oaks	27	361
China	87	249

#### Table 10-5. Structures at Greater Risk by Jurisdiction

<sup>5</sup> Source: U.S. Census Bureau data estimates for 2014.

JURISDICTION	MANUFACTURED HOMES	SFR STRUCTURES BUILT BEFORE 1980
Groves	69	5,388
Nederland	219	5,473
Nome	45	107
Port Arthur	234	16,809
Port Neches	82	3,887
SETRPC	0	0
Jefferson County <sup>6</sup>	3,138	69,478

More severe damage involves windborne debris; in some instances, patio furniture and other lawn items have been reported to have been blown around by wind and, very commonly, debris from damaged structures in turn have caused damage to other buildings not directly impacted by the event.

The following critical facilities would be vulnerable to thunderstorm wind events in each participating jurisdiction:

JURISDICTION	CRITICAL FACILITIES
Jefferson County	Port Authority Facility, Fire Station, Water District Facility, 1 School
Beaumont	2 Fire Stations, 2 Police Stations, 32 Schools, Port Authority Facility, 5 Water District Facilities, 3 Drainage District Facilities, 4 Hospitals
Bevil Oaks	Fire Station
China	Fire Station, 2 Schools
Groves	Fire Station, Police Station, 4 Schools
Nederland	Fire Station, Police Station, Water District Facility, 3 Hospitals, Airport, 8 Schools
Nome	Fire Station
Port Arthur	Fire Station, Police Station, Drainage District Facility, 2 Port Authority Facilities, 2 Hospitals, 14 Schools
Port Neches	Fire Station, Police Station, 5 Schools
SETRPC	SETRPC Facility

#### Table 10-6. Critical Facilities by Jurisdiction

<sup>&</sup>lt;sup>6</sup> County totals include all participating jurisdictions and unincorporated areas.

A thunderstorm wind event can also result in traffic disruptions, injuries, and in rare cases, fatalities. The impact of extreme winds experienced in the Jefferson County planning area has resulted in three injuries and one fatality. While damages and shutdown of critical facilities would have a minor impact on the planning area, historic injuries and fatalities indicate an impact of "Substantial" with multiple potential deaths and injuries. Overall, the average loss estimate (in 2016 dollars) is \$3,655,236, having an approximate annual loss estimate of \$174,058 (Table 10-7).

JURISDICTION	PROPERTY & CROP LOSS	AVERAGE ANNUALIZED LOSSES
Jefferson County	\$121,263	\$5,774
Beaumont	\$719,727	\$34,273
Bevil Oaks	\$2,779	\$132
China	\$243,593	\$11,600
Groves	\$2,188,939	\$104,235
Nederland	\$98,643	\$4,697
Nome	\$33,775	\$1,608
Port Arthur	\$246,517	\$11,739
Port Neches	\$0	\$0
SETRPC	\$0	\$0
Planning Area	\$3,655,236	\$174,058

#### Table 10-7. Potential Annualized Losses by Jurisdiction, 1996-2016

## Assessment of Impacts

Thunderstorm wind events have the potential to pose a significant risk to people, and can create dangerous and difficult situations for public health and safety officials. Impacts to the planning area can include:

- Individuals exposed to the storm can be struck by flying debris, falling limbs, or downed trees, causing serious injury or death.
- Structures can be damaged or crushed by falling trees, which can result in physical harm to the occupants.
- Significant debris and downed trees can result in emergency response vehicles being unable to access areas of the community.
- Downed power lines may result in roadways being unsafe for use, which may prevent first responders from answering calls for assistance or rescue.
- During exceptionally heavy wind events, first responders may be prevented from responding to calls, as the winds may reach a speed in which their vehicles and equipment are unsafe to operate.

- Thunderstorm wind events often result in widespread power outages, increasing the risk to more vulnerable portions of the population who rely on power for health and/or life safety.
- Extended power outage often results in an increase in structure fires and carbon monoxide poisoning, as individuals attempt to cook or heat their homes with alternate, unsafe cooking or heating devices, such as grills.
- First responders are exposed to downed power lines, unstable and unusual debris, hazardous materials, and generally unsafe conditions.
- Emergency operations and services may be significantly impacted due to damaged facilities and/or loss of communications.
- Critical staff may be unable to report for duty, limiting response capabilities.
- County or City departments may be damaged, delaying response and recovery efforts for the entire community.
- Private sector entities that the County or City and its residents rely on, such as utility providers, financial institutions, and medical care providers may not be fully operational and may require assistance from neighboring communities until full services can be restored.
- Economic disruption negatively impacts the programs and services provided by the community due to short and long term loss in revenue.
- Some businesses not directly damaged by extreme wind events may be negatively impacted while roads are cleared and utilities are being restored, further slowing economic recovery.
- Older structures built to less stringent building codes may suffer greater damage as they are typically more vulnerable to extreme winds.
- Large scale wind events can have significant economic impact on the affected area, as it must now fund expenses such as infrastructure repair and restoration, temporary services and facilities, overtime pay for responders, as well as normal day-to-day operating expenses.
- Businesses that are more reliant on utility infrastructure than others may suffer greater damages without a backup power source.
- Sabine Lake is a large recreational lake that attracts fishing and boating activities throughout the year. A large thunderstorm wind event could impact recreational water activities, placing boaters and campers in imminent danger, potentially requiring emergency services or lake evacuation.
- Recreational areas and parks may be damaged or inaccessible due to downed trees or debris, causing temporary impacts to area businesses.

The economic and financial impacts of thunderstorm winds on the area will depend entirely on the scale of the event, what is damaged, and how quickly repairs to critical components of the economy can be implemented. The level of preparedness and pre-event planning done by the community, local businesses, and citizens will also contribute to the overall economic and financial conditions in the aftermath of any thunderstorm wind event.

# **SECTION 11: TORNADO**

Hazard Description	1
Location	2
Extent	2
Historical Occurrences	5
Significant Past Events	8
Probability of Future Events	8
Vulnerability and Impact	8
Assessment of Impacts	. 11

# Hazard Description



Tornadoes are among the most violent storms on the planet. A tornado is a rapidly rotating column of air extending between, and in contact with, a cloud and the surface of the earth. The most violent tornadoes are capable of tremendous destruction, with wind speeds of 250 miles per hour (mph) or more. In extreme cases, winds may approach 300 mph. Damage paths can be in excess of 1 mile wide and 50 miles long.

The most powerful tornadoes are produced by "Supercell Thunderstorms." Supercell Thunderstorms are created when horizontal wind shears (winds moving in different directions at different altitudes) begin to rotate the storm. This horizontal rotation can be tilted vertically by violent updrafts, and the rotation radius can shrink, forming a vertical column of very quickly swirling air. This rotating air can eventually reach the ground, forming a tornado.

#### Table 11-1. Tornado Variations

WEAK TORNADOES	STRONG TORNADOES	VIOLENT TORNADOES
• 69% of all tornadoes	• 29% of all tornadoes	• 2% of all tornadoes
<ul> <li>Less than 5% of tornado deaths</li> </ul>	<ul> <li>Nearly 30% of all tornado deaths</li> </ul>	<ul> <li>70% of all tornado deaths</li> <li>Lifetime can exceed 1 hour</li> </ul>
<ul> <li>Lifetime 1-10+ minutes</li> <li>Winds less than 110 mph</li> </ul>	May last 20 minutes or longer	• Winds greater than 205 mph
	• Winds 110 – 205 mph	

## Section 11: Tornado

# Location

As with thunderstorms, tornadoes do not have any specific geographic boundary and can occur throughout the Jefferson County planning area, including the SETRPC and all participating jurisdictions. It is assumed that the Jefferson County planning area is equally exposed to tornado activity. Jefferson County is located in Wind Zone III (Figure 11-1), where tornado winds can be as high as 200 mph.





# Extent

The destruction caused by tornadoes ranges from light to inconceivable depending on the intensity, size, and duration of the storm. Typically, tornadoes cause the greatest damage to structures of light construction, such as residential homes (particularly mobile homes).

Tornado magnitudes prior to 2005 were determined using the traditional version of the Fujita Scale (Table 11-2). Since February 2007, the Fujita Scale has been replaced by the Enhanced Fujita Scale (Table 11-3), which retains the same basic design and 6 strength categories as the previous scale. The newer scale

<sup>&</sup>lt;sup>1</sup> Jefferson County is indicated by the star.
reflects more refined assessments of tornado damage surveys, standardization, and damage consideration to a wider range of structures.

F-SCALE NUMBER	INTENSITY	WIND SPEED (MPH)	TYPE OF DAMAGE DONE	PERCENT OF APPRAISED STRUCTURE VALUE LOST DUE TO DAMAGE
FO	Gale Tornado	40 - 72	Some damage to chimneys; breaks branches off trees; pushes over shallow-rooted trees; damages sign boards.	None Estimated
F1	Moderate Tornado	73 – 112	The lower limit is the beginning of hurricane wind speed; peels surface off roofs; mobile homes pushed off foundations or overturned; moving autos pushed off roads; attached garages may be destroyed.	0% – 20%
F2	Significant Tornado	113 - 157	Considerable damage. Roofs torn off frame houses; mobile homes demolished; boxcars pushed over; large trees snapped or uprooted; light object missiles generated.	50% – 100%
F3	Severe Tornado	158 – 206	Roofs and some walls torn off well- constructed houses; trains overturned; most trees in forest uprooted.	100%
F4	Devastating Tornado	207 – 260	Well-constructed homes leveled; structures with weak foundations blown off some distance; cars thrown and large missiles generated.	100%
F5	Incredible Tornado	261 - 318	Strong frame houses lifted off foundations and carried considerable distances to disintegrate; automobile sized missiles flying through the air in excess of 330 yards; trees debarked; steel reinforced concrete badly damaged.	100%

### Table 11-2. The Fujita Tornado Scale<sup>2</sup>

<sup>&</sup>lt;sup>2</sup> Source: http://www.tornadoproject.com/fscale/fscale.htm

STORM CATEGORY	DAMAGE LEVEL	3 SECOND GUST (MPH)	DESCRIPTION OF DAMAGES	PHOTO EXAMPLE
EFO	Gale	65 - 85	Some damage to chimneys; breaks branches off trees; pushes over shallow-rooted trees; damages sign boards.	
EF1	Weak	86-110	The lower limit is the beginning of hurricane wind speed; peels surface off roofs; mobile homes pushed off foundations or overturned; moving autos pushed off roads; attached garages may be destroyed.	
EF2	Strong	111–135	Considerable damage; roofs torn off frame houses; mobile homes demolished; boxcars pushed over; large trees snapped or uprooted; light object missiles generated.	
EF3	Severe	136-165	Roof and some walls torn off well-constructed houses; trains overturned; most trees in forest uprooted.	
EF4	Devastating	166-200	Well-constructed homes leveled; structures with weak foundations blown off some distance; cars thrown and large missiles generated.	
EF5	Incredible	200+	Strong frame houses lifted off foundations and carried considerable distances to disintegrate; automobile sized missiles flying through the air in excess of 330 yards; trees debarked; steel reinforced concrete badly damaged.	

### Table 11-3. Enhanced Fujita Scale for Tornados

Both the Fujita Scale and Enhanced Fujita Scale should be referenced in reviewing previous occurrences since tornado events prior to 2007 will follow the original Fujita Scale. The largest magnitude reported within the planning area is F3 on the Fujita Scale, a "Severe Tornado." Based on the planning areas location in Wind Zone III, the planning area could experience anywhere from an EF0 to an EF5 depending on the wind speed.

The events in Jefferson County have been between F0 and F3 (Table 11-4). Therefore, the range of intensity that the Jefferson County planning area, including the SETRPC and all participating jurisdictions, would be expected to mitigate is a tornado event that would be a low to severe risk, an EF0 to EF5.

# **Historical Occurrences**

The NCEI is a national data source organized under NOAA and is the largest archive available for climate data. Only NCEI reported incidents were factored into this risk assessment. It is likely that a number of occurrences have gone unreported over the past 21 years. It is important to note that the SETRPC is located within the City of Beaumont. There may be some occurrences that have occurred for the SETRPC and may not have been recorded, but are included in the City of Beaumont occurrence data because of their location.

Figure 11-2 identifies the locations of previous occurrences in the Jefferson County planning area from 1996 to 2016. A total of 10 events have been recorded by the Storm Prediction Center (National Oceanic and Atmospheric Administration – NOAA) and the National Center for Environmental Information (NCEI) databases for the Jefferson County planning area, including the SETRPC and all participating jurisdictions. The most significant event reported occurred in Jefferson County near the Parkdale Mall area on August 18, 2009. The EF1 tornado was 100 yards wide and stayed on the ground in Jefferson County for just under 1 mile. The area impacted by the tornado was densely populated with commercial development and damages exceeded 22 million dollars (2016 dollar value).





<sup>3</sup> Source: NOAA Records

JURISDICTION	DATE	TIME	MAGNITUDE	DEATHS	INJURIES	PROPERTY DAMAGE	CROP DAMAGE
Port Arthur	7/14/1997	3:05 PM	FO	0	3	\$74,536	\$0
Nome	1/1/1999	11:45 PM	F3	0	5	\$718,070	\$0
Jefferson County	4/3/2000	3:08 AM	F1	0	1	\$138,944	\$0
Groves	4/3/2000	3:20 AM	F1	0	0	\$4,168,310	\$0
Bevil Oaks	10/12/2001	2:04 PM	FO	0	0	\$0	\$0
Beaumont	10/13/2001	3:55 AM	F1	0	0	\$1,350,994	\$0
Beaumont	11/18/2003	1:00 AM	FO	0	0	\$130,033	\$0
China	10/16/2006	5:00 AM	F1	0	0	\$356,043	\$0
Beaumont	8/18/2009	12:57 PM	F1	0	10	\$22,304,870	\$0
Nome	6/9/2010	4:32 AM	F1	0	0	\$109,725	\$0

### Table 11-4. Historical Tornado Events, 1996-2016<sup>4</sup>

#### Table 11-5. Summary of Historical Tornado Events, 1996-2016<sup>5</sup>

JURISDICTION	NUMBER OF EVENTS	MAGNITUDE	FATALITIES	INJURIES	PROPERTY DAMAGE	CROP DAMAGE
Jefferson County	1	F1	0	1	\$138,944	\$0
Beaumont	3	F1	0	10	\$23,785,897	\$0
Bevil Oaks	1	FO	0	0	\$0	\$0
China	1	F1	0	0	\$356,043	\$0
Groves	1	F1	0	0	\$4,168,310	\$0
Nederland	0	N/A	0	0	\$0	\$0
Nome	2	F3	0	5	\$827,795	\$0
Port Arthur	1	FO	0	3	\$74,536	\$0
Port Neches	0	N/A	0	0	\$0	\$0
SETRPC	0	N/A	0	0	\$0	\$0
TOTAL LOSSES	10	(Max Extent)	0	19	\$29,351,525	

<sup>5</sup> Values are in 2016 dollars.

<sup>&</sup>lt;sup>4</sup> Values are in 2016 dollars.

Based on the list of historical tornado events for the Jefferson County planning area (listed above), including the SETRPC and all participating jurisdictions, no events have occurred since the 2011 Plan.

### Significant Past Events

#### January 1, 1999 – Nome

A strong tornado that developed in Liberty County moved into western Jefferson County, destroying several rice dryers, 2 mobile homes, and a brick house. 5 people were injured in the mobile homes and brick house. 2 people received broken necks and other injuries after they were ejected from a mobile home.

A fiberglass tub from the destroyed brick home was found 1 mile away. Cancelled checks and other paperwork were found 7 miles away in southern Hardin County. This tornado was strongest southwest of Nome, and was weakening as it passed through the small community of Nome. At least 20 homes received minor damage, such as blown off shingles, and many trees were blown down.

#### October 16, 2006 – China/Jefferson County

An abundance of moisture and high wind shear resulted in several tornadoes and flash floods across southeast Texas. A tornado destroyed 5 mobile homes and damaged an additional 20 homes just east of China. Trees and power lines were blown down.

#### August 18, 2009 – Jefferson County

An EF1 Tornado touched down just west of the Kohl's Department Store in the Parkdale Mall area. The tornado struck the Kohl's, tearing off part of its roof and collapsing the front entrance. Damage to the store forced it to close for several weeks. The tornado continued to the northeast and crossed a Walmart parking lot, flipping over 4 vehicles and damaging dozens of others. Further to the northeast, the tornado struck Parkdale Mall, damaging the roofs of several department stores. The tornado lifted in the east parking lot of Parkdale Mall before reaching Highway 69. Numerous photographs and videos of the tornado were taken.

# Probability of Future Events

Tornadic storms can occur at any time of year and at any time of day, but they are typically more common in the spring months during the late afternoon and evening hours. A smaller, high frequency period can emerge in the fall during the brief transition between the warm and cold seasons. According to historical records, Jefferson County experiences a tornado touchdown approximately every year. This frequency supports a highly likely probability of future events for the Jefferson County planning area, including the SETRPC and all participating jurisdictions.

# Vulnerability and Impact

Because tornadoes often cross jurisdictional boundaries, all existing and future buildings, facilities, and populations in Jefferson County are considered to be exposed to this hazard and could potentially be impacted. The damage caused by a tornado is typically a result of high wind velocity, wind-blown debris, lightning, and large hail.

The average tornado moves from southwest to northeast, but tornadoes have been known to move in any direction. Consequently, the vulnerability of humans and property is difficult to evaluate since tornadoes form at different strengths, in random locations, and create relatively narrow paths of destruction. Although tornadoes strike at random, making all buildings vulnerable, three types of structures are more likely to suffer damage:

- Manufactured Homes;
- Homes on crawlspaces (more susceptible to lift); and
- Buildings with large spans, such as shopping malls, gymnasiums, and factories.

Tornadoes can possibly cause a significant threat to people as they could be struck by flying debris, falling trees/branches, utility lines, and poles. First responders could also not be able to respond to calls due to blocked roads. Tornadoes commonly cause power outages, which could cause health and safety risks to patients in hospitals or other vulnerable populations that rely on power for medical necessities.

The Jefferson County planning area features multiple mobile or manufactured home parks throughout the planning area and all participating jurisdictions. These parks are typically more vulnerable to tornado events than typical site built structures. In addition, manufactured homes are located sporadically throughout the planning area, including all jurisdictions. These homes would also be more vulnerable. The U.S. Census data indicates a total of 3,138 manufactured homes located in the Jefferson County planning area, including all participating jurisdictions (Table 11-6). In addition, 65.4% (approximately 69,478 structures) of the single family residential (SFR) structures in the Jefferson County planning area were built before 1980.<sup>6</sup> These structures would typically be built to lower or less stringent construction standards than newer construction and may be more susceptible to damages during significant tornado events.

	MANUFACTURED	SFR STRUCTURES BUILT BEFORE
Johisbierren	HOMES	1980
Beaumont	718	33,386
Bevil Oaks	27	361
China	87	249
Groves	69	5,388
Nederland	219	5,473
Nome	45	107
Port Arthur	234	16,809
Port Neches	82	3,887
SETRPC	0	0
Jefferson County <sup>7</sup>	3,138	69,478

### Table 11-6. Structures at Greater Risk by Jurisdiction

<sup>&</sup>lt;sup>6</sup> Source: U.S. Census Bureau data estimates for 2014.

<sup>&</sup>lt;sup>7</sup> County totals includes SETRPC and all participating jurisdictions and unincorporated areas.

The following critical facilities would be vulnerable to tornado events in each participating jurisdiction:

JURISDICTION	CRITICAL FACILITIES
Jefferson County	Port Authority Facility, Fire Station, Water District Facility, 1 School
Beaumont	2 Fire Stations, 2 Police Stations, 32 Schools, Port Authority Facility, 5 Water District Facilities, 3 Drainage District Facilities, 4 Hospitals
Bevil Oaks	Fire Station
China	Fire Station, 2 Schools
Groves	Fire Station, Police Station, 4 Schools
Nederland	Fire Station, Police Station, Water District Facility, 3 Hospitals, Airport, 8 Schools
Nome	Fire Station
Port Arthur	Fire Station, Police Station, Drainage District Facility, 2 Port Authority Facilities, 2 Hospitals, 14 Schools
Port Neches	Fire Station, Police Station, 5 Schools
SETRPC	SETRPC Facility

#### Table 11-7. Critical Facilities by Jurisdiction

The average loss estimate of crops and property is \$29,351,525 (in 2016 dollars), having an approximate annual loss estimate of \$1,397,692. Based on historic loss and damages, the impact of tornadoes on the Jefferson County planning area, including the SETRPC and all participating jurisdictions, can be considered "Major", with more than 25 percent of property expected to be destroyed or with major damage, injuries and/or illness that result in permanent disability, and critical facilities shut down for at least 2 weeks.

Annualized losses are not included for the SETRPC as there have not been events or losses to effect the facilities separate and apart from a historical occurrence for the City of Beaumont.

#### Table 11-8. Potential Annualized Losses by Jurisdiction, 1996-2016

JURISDICTION	PROPERTY AND CROP DAMAGE	ANNUAL LOSS ESTIMATE
Jefferson County	\$138,944	\$6,616
Beaumont	\$23,785,897	\$1,132,662
Bevil Oaks	\$0	\$0
China	\$356,043	\$16,954
Groves	\$4,168,310	\$198,491
Nederland	\$0	\$0
Nome	\$827,795	\$39,419

JURISDICTION	PROPERTY AND CROP DAMAGE	ANNUAL LOSS ESTIMATE
Port Arthur	\$74,536	\$3,549
Port Neches	\$0	\$0
SETRPC	\$0	\$0

# Assessment of Impacts

Tornadoes have the potential to pose a significant risk to the population and can create dangerous situations. Providing and preserving public health and safety is often difficult. Impacts to the planning area can include:

- Individuals exposed to the storm can be struck by flying debris, falling limbs, or downed trees, causing serious injury or death.
- Structures can be damaged or crushed by falling trees, which can result in physical harm to the occupants.
- Manufactured homes may suffer substantial damage as they would be more vulnerable than typical site built structures.
- Sub-standard construction may suffer substantial damage as they are not built to code and would be more vulnerable to tornado events than code compliant structures.
- Significant debris and downed trees can result in emergency response vehicles being unable to access areas of the community.
- Downed power lines may result in roadways being unsafe for use, which may prevent first responders from answering calls for assistance or rescue.
- Tornadoes often result in widespread power outages, increasing the risk to more vulnerable portions of the population who rely on power for health and/or life safety.
- Extended power outage can result in an increase in structure fires and/or carbon monoxide poisoning, as individuals attempt to cook or heat their home with alternate, unsafe cooking or heating devices, such as grills.
- Tornadoes can destroy or make residential structures uninhabitable, requiring shelter or relocation of residents in the aftermath of the event.
- First responders must enter the damage area shortly after the tornado passes to begin rescue operations and to organize cleanup and assessments efforts. Therefore, they are exposed to downed power lines, unstable and unusual debris, hazardous materials, and generally unsafe conditions, elevating the risk of injury to first responders and potentially diminishing emergency response capabilities.
- Emergency operations and services may be significantly impacted due to damaged facilities, loss of communications, and damaged emergency vehicles and equipment.
- County or City departments may be damaged or destroyed, delaying response and recovery efforts for the entire community.
- Private sector entities that the County or City and its residents rely on, such as utility providers, financial institutions, and medical care providers may not be fully operational and may require assistance from neighboring communities until full services can be restored.

- Economic disruption negatively impacts the programs and services provided by the community due to short and long term loss in revenue.
- Damage to infrastructure may slow economic recovery since repairs may be extensive and lengthy.
- Some businesses not directly damaged by the tornado may be negatively impacted while roads and utilities are being restored, further slowing economic recovery.
- When the community is affected by significant property damage it is anticipated that funding would be required for infrastructure repair and restoration, temporary services and facilities, overtime pay for responders, as well as normal day-to-day operating expenses.
- Displaced residents may not be able to immediately return to work, further slowing economic recovery.
- Residential structures destroyed by a tornado may not be rebuilt for years, reducing the tax base for the community.
- Large or intense tornadoes may result in a dramatic population fluctuation, as people are unable to return to their homes or jobs and must seek shelter and/or work outside of the affected area.
- Businesses that are uninsured or underinsured may have difficulty reopening, which results in a net loss of jobs for the community and a potential increase in the unemployment rate.
- Recreation activities may be unavailable and tourism can be unappealing for years following a large tornado, devastating directly related local businesses.

The economic and financial impacts of a tornado event on the community will depend on the scale of the event, what is damaged, costs of repair or replacement, lost business days in impacted areas, and how quickly repairs to critical components of the economy can be implemented. The level of preparedness and pre-event planning done by government, businesses, and citizens will contribute to the overall economic and financial conditions in the aftermath of a tornado event.

# **SECTION 12: DROUGHT**

Hazard Description	1
Location	2
Extent	3
Historical Occurrences	5
Significant Past Events	5
Probability of Future Events	6
Vulnerability and Impact	6
Assessment of Impacts	8

# Hazard Description

Drought is a period of time without substantial rainfall that persists from one year to the next. Drought is a normal part of virtually all climatic regions, including areas with high and low average rainfall. Drought is the consequence of anticipated natural precipitation reduction over an extended period of time, usually a season or more in length. Droughts can be classified as meteorological, hydrologic, agricultural, and socioeconomic. Table 12-1 presents definitions for these different types of drought.



METEOROLOGICAL DROUGHT	The degree of dryness or departure of actual precipitation from an expected average or normal amount based on monthly, seasonal, or annual time scales.
HYDROLOGIC DROUGHT	The effects of precipitation shortfalls on stream flows and reservoir, lake, and groundwater levels.
AGRICULTURAL DROUGHT	Soil moisture deficiencies relative to water demands of plant life, usually crops.
SOCIOECONOMIC DROUGHT	The effect of demands for water exceeding the supply as a result of a weather- related supply shortfall.

#### Table 12-1. Drought Classification Definitions<sup>1</sup>

Droughts are one of the most complex of all natural hazards as it is difficult to determine their precise beginning or end. In addition, droughts can lead to other hazards such as extreme heat and wildfires. Their impact on wildlife and area farming is enormous, often killing crops, grazing land, edible plants, and

<sup>&</sup>lt;sup>1</sup> Source: Multi-Hazard Identification and Risk Assessment: A Cornerstone of the National Mitigation Strategy, FEMA

even in severe cases, trees. A secondary hazard to drought is wildfire because dying vegetation serves as a prime ignition source. Therefore, a heat wave combined with a drought is a very dangerous situation.

# Location

Droughts occur regularly throughout Texas and Jefferson County, and are a frequent condition. However, they can vary greatly in their intensity and duration. The Drought Monitor (Figure 12-1) shows the study region is currently experiencing normal conditions. The planning area has experienced abnormally dry to exceptional drought conditions over the last ten years (Figure 12-2). There is no distinct geographic boundary to drought; therefore, it can occur throughout the Jefferson County planning area, including the SETRPC and all participating jurisdictions, equally.









# Extent

The Palmer Drought Index is used to measure the extent of drought by measuring the duration and intensity of long-term drought-inducing circulation patterns. Long-term drought is cumulative, with the intensity of drought during the current month dependent upon the current weather patterns plus the cumulative patterns of previous months. The hydrological impacts of drought (e.g., reservoir levels, groundwater levels, etc.) take longer to develop. Table 12-2 depicts magnitude of drought, while Table 12-3 describes the classification descriptions.

	DROUGHT CONDITION CLASSIFICATIONS							
DROUGHT INDEX	Extreme	Severe	Moderate	Normal	Moderately Moist	Very Moist	Extremely Moist	
Z Index	-2.75 and below	-2.00 to -2.74	-1.25 to -1.99	-1.24 to +.99	+1.00 to +2.49	+2.50 to +3.49	n/a	
Meteorological	-4.00 and below	-3.00 to -3.99	-2.00 to -2.99	-1.99 to +1.99	+2.00 to +2.99	+3.00 to +3.99	+4.00 and above	
Hydrological	-4.00 and below	-3.00 to -3.99	-2.00 to -2.99	-1.99 to +1.99	+2.00 to +2.99	+3.00 to +3.99	+4.00 and above	

#### Table 12-2. Palmer Drought Index

#### Table 12-3. Palmer Drought Category Descriptions<sup>2</sup>

CATEGORY	DESCRIPTION	POSSIBLE IMPACTS	PALMER DROUGHT INDEX
D0	Abnormally Dry	Going into drought: short-term dryness slowing planting, growth of crops or pastures; fire risk above average. Coming out of drought: some lingering water deficits; pastures or crops not fully recovered.	-1.0 to -1.9
D1	Moderate Drought	Some damage to crops, pastures; fire risk high; streams, reservoirs, or wells low, some water shortages developing or imminent, voluntary water use restrictions requested.	-2.0 to -2.9
D2	Severe Drought	Crop or pasture losses likely; fire risk very high; water shortages common; water restrictions imposed.	-3.0 to -3.9
D3	Extreme Drought	Major crop/pasture losses; extreme fire danger; widespread water shortages or restrictions.	-4.0 to -4.9
D4	Exceptional Drought	Exceptional and widespread crop/pasture losses; exceptional fire risk; shortages of water in reservoirs, streams, and wells, creating water emergencies.	-5.0 or less

Drought is monitored nationwide by the National Drought Mitigation Center (NDMC). Indicators are used to describe broad scale drought conditions across the United States. Indicators correspond to the intensity of drought.

<sup>&</sup>lt;sup>2</sup> Source: National Drought Mitigation Center

Based on the historical occurrences for drought and the location of Jefferson County, the entire planning area, including the SEPRTC and all participating jurisdictions, can anticipate a range of drought from abnormally dry to exceptional, or D0 to D4 based on the Palmer Drought Category.

# **Historical Occurrences**

Jefferson County may typically experience a severe drought. Table 12-4 and 12-5 list historical events that have occurred in Jefferson County as reported in the National Center for Environmental Information (NCEI). Historical drought information, as provided by the NCEI, shows drought activity across a multicounty forecast area for each event. The appropriate percentage of the total property and crop damage reported for the entire forecast area has been allocated to each county impacted by the event. Historical drought data for the participating jurisdictions in the Jefferson County planning area, including the SETRPC and all participating jurisdictions, is provided on a County-wide basis per the NCEI database.

Table 12-4. Historical Drought Years, 1996-2016



Table 12-5. Historical Drought Events, 1996-2016<sup>3</sup>

JURISDICTION	DATE	DEATHS	INJURIES	PROPERTY DAMAGE	CROP DAMAGE
Jefferson County	5/1/1996	0	0	\$0	\$0
Jefferson County	5/20/1998	0	0	\$0	\$0
Jefferson County	6/1/1998	0	0	\$0	\$0
Jefferson County	7/1/1998	0	0	\$0	\$0
Jefferson County	2/1/2000	0	0	\$0	\$0
TOTALS		0	0	\$0	

Based on the list of historical drought events for the Jefferson County planning area (listed above), including the SETRPC and all participating jurisdictions, no events have occurred since the 2011 Plan.

<sup>&</sup>lt;sup>3</sup> Values are in 2016 dollars.

### Significant Past Events

### January - May, 1996 – Jefferson County

Rainfall totals from January through May averaged 10 to 15 inches below normal. The main areas affected include farming and fire protection. Crop damage in neighboring counties exceeded 1 million dollars. Drought conditions continue across southeast Texas through May.

#### May - July, 1998 – Jefferson County

Drought conditions began by mid-May, as southeast Texans had gone over 6 weeks without any significant rainfall. By the end of May, many locations had seen less than 0.10 inches of rain for the month. This was the start of a significant impact on agriculture and forestry resources. A mild to moderate drought continued across southeast Texas in the month of June. Only 2 days provided any relief from the dry weather, June 5 and June 26, 1998. Many places recorded less than 2 inches of rain for the entire month of June. Crop losses continued to mount, primarily in the rice business, as well as forestry resources.

#### February, 2000 – Jefferson County

The month of February was one of the 5 driest Februarys on record across southeast Texas. Less than 1 inch of rain fell across the entire region. The 2 month total for January and February 2000 was the second driest on record for the Beaumont/Port Arthur area, with less than 2.5 inches of rainfall.

# Probability of Future Events

Based on available records of historic events, there have been 3 extended time periods of drought within a 21 year reporting period, which provides a frequency of occurrence of 1 event probable in the next 5 years. This frequency supports an occasional probability of future events. All participating jurisdictions including the SETRPC are included under the County.

# Vulnerability and Impact

Loss estimates were based on 21 years of statistical data from the NCEI. A drought event frequencyimpact was then developed to determine an impact profile on agriculture products and estimate potential losses due to drought in the area. Table 12-6 shows annualized exposure.

Table 12-6. [	<b>Drought Event</b>	Damage Totals,	, 1996-2016
---------------	----------------------	----------------	-------------

JURISDICTION	PROPERTY & CROP LOSS	ANNUALIZED LOSS ESTIMATES
Jefferson County	\$0	\$0

Drought impacts large areas and crosses jurisdictional boundaries. All existing and future buildings, facilities, and populations are exposed to this hazard and could potentially be impacted. However, drought impacts are mostly experienced in water shortages and crop/livestock losses on agricultural lands and typically have no impact on buildings.

In terms of vulnerability, population, agriculture, property, and environment are all vulnerable to drought. The average person will survive only a few days without water, and this timeframe can be drastically

shortened for those people with more fragile health – typically children, the elderly, and the ill. Populations over 65 in the Jefferson County planning area are estimated at 13% of the total population and children under the age of 5 exceed 6% – an estimated total of  $50,074^4$  potentially vulnerable residents in the planning area based on age (Table 12-7).

JURISDICTION	POPULATION 65 AND OLDER	POPULATION UNDER 5
Beaumont	15,539	8,087
Bevil Oaks	290	42
China	183	25
Groves	2,685	971
Nederland	2,371	976
Nome	56	11
Port Arthur	6,344	5,073
Port Neches	1,939	904
Jefferson County⁵	32,774	17,300

#### Table 12-7. Populations at Greater Risk by Jurisdiction

The population is also vulnerable to food shortages when drought conditions exist and potable water is in short supply. Potable water is used for drinking, sanitation, patient care, sterilization, equipment, heating and cooling systems, and many other essential functions in medical facilities. All residents in the Jefferson County planning area could be adversely affected by drought conditions, which could limit water supplies and present health threats. However, during summer drought, or hot and dry conditions, elderly persons, small children, infants, and the chronically ill who do not have adequate cooling units in their homes may become more vulnerable to injury and/or death.

The economic impact of droughts can be significant as it produces a complex web of effects that span many sectors of the economy and reach well beyond the area experiencing physical drought. This complexity exists because water is integral to our ability to produce goods and provide services. If droughts extend over a number of years, the direct and indirect economic impact can be significant.

Habitat damage is a vulnerability of the environment during periods of drought, for both aquatic and terrestrial species. The environment also becomes vulnerable during periods of extreme or prolonged drought due to severe erosion and land degradation.

The impact of droughts experienced in the Jefferson County planning area, including the SETRPC and all participating jurisdictions, has resulted in 0 injuries and fatalities. This supports a "limited" severity of

<sup>&</sup>lt;sup>4</sup> U.S. Census Bureau 2014 data for Jefferson County

<sup>&</sup>lt;sup>5</sup> County totals includes all participating jurisdictions and unincorporated areas.

impact, meaning injuries and/or illnesses are treatable with first aid, shutdown of facilities and services for 24 hours or less, and less than 10% of property is destroyed or sustains major damage. Annualized loss over the 21-year reporting period in Jefferson County is negligible.

# Assessment of Impacts

The Drought Impact Reporter was developed in 2005 by the University of Nebraska-Lincoln to provide a national database of drought impacts. Droughts can have an impact on: agriculture; business and industry; energy; fire; plants and wildlife; relief, response, and restrictions; society and public health; tourism and recreation; and water supply and quality. Table 12-8 lists the drought impacts for Jefferson County from 2005 to 2016, based on reports received by the Drought Impact Reporter.

DROUGHT IMPACTS		
Agriculture	29	
Business & Industry	3	
Energy	1	
Fire	9	
Plants & Wildlife	21	
Relief, Response, & Restrictions	12	
Society & Public Health	5	
Tourism & Recreation	1	
Water Supply & Quality	10	

### Table 12-8. Drought Impacts, 2005-2016

Drought has the potential to impact people in the Jefferson County planning area. While it is rare that drought, in and of itself, leads to a direct risk to the health and safety of people in the U.S., severe water shortages could result in inadequate supply for human needs. Drought is also frequently associated with a variety of impacts, including:

- Recreational activities at Sabine Lake that rely on water may be curtailed, such as hunting and fishing, resulting in fewer tourists and lower revenue.
- The McFaddin National Wildlife Refuge area may be especially vulnerable as severe and prolonged drought can result in the reduction of a species, or cause the extinction of a species altogether.
- Plant life will suffer from long-term drought. Wind and erosion will also pose a threat to plant life as soil quality will decline.
- The number of health-related low-flow issues (e.g., diminished sewage flows, increased pollution concentrations, reduced firefighting capacity, and cross-connection contamination) will increase as the drought intensifies.

- Public safety from forest/range/wildfires will increase as water availability and/or pressure decreases.
- Respiratory ailments may increase as the air quality decreases.
- There may be an increase in disease due to wildlife concentrations (e.g., rabies, Rocky Mountain spotted fever, Lyme disease).
- Jurisdictions and residents may disagree over water use/water rights, creating conflict.
- Political conflicts may increase between municipalities, counties, states, and regions.
- Water management conflicts may arise between competing interests.
- Increased law enforcement activities may be required to enforce water restrictions.
- Severe water shortages could result in inadequate supply for human needs as well as lower quality of water for consumption.
- Firefighters may have limited water resources to aid in firefighting and suppression activities, increasing risk to lives and property.
- During drought there is an increased risk for wildfires and dust storms.
- The community may need increased operational costs to enforce water restriction or rationing.
- Prolonged drought can lead to increases in illness and disease related to drought.
- Utility providers can see decreases in revenue as water supplies diminish.
- Utilities providers may cut back energy generation and service to their customers in order to prioritize critical service needs.
- Hydroelectric power generation facilities and infrastructure would have significantly diminished generation capability. Dams simply cannot produce as much electricity from low water levels as they can from high water levels.
- Fish and wildlife food and habitat will be reduced or degraded over time during a drought and disease will increase, especially for aquatic life.
- Wildlife will move to more sustainable locations, creating higher concentrations of wildlife in smaller areas, increasing vulnerability and further depleting limited natural resources.
- Dry and dead vegetation will increase the risk of wildfire.
- Land subsidence threat increases as groundwater is depleted.
- Drought poses a significant risk to annual and perennial crop production and overall crop quality, leading to higher food costs.
- Drought related declines in production may lead to an increase in unemployment.
- Drought may limit livestock grazing resulting in decreased livestock weight, potential increased livestock mortality, and increased cost for feed.
- Negatively impacted water suppliers may face increased costs resulting from the transport water or develop supplemental water resources.
- Long term drought may negatively impact future economic development.

The overall extent of damages caused by periods of drought is dependent on its extent and duration. The level of preparedness and pre-event planning done by government, businesses, and citizens will contribute to the overall economic and financial conditions in the aftermath of a drought event.

# **SECTION 13: WILDFIRE**

Hazard Description	1
location	1
Evtent	11
Historical Occurrences	23
Probability of Future Events	25
Vulnerability and Impact	26
Assessment of Impacts	37

# Hazard Description

A wildfire event can rapidly spread out of control and occurs most often in the summer, when the brush is dry and flames can move unchecked through a highly vegetative area. Wildfires can start as a slow burning fire along the forest floor, killing and damaging trees. The fires often spread more rapidly as they reach the tops of trees, with wind carrying the flames from tree to tree. Usually, dense smoke is the first indication of a wildfire.

A wildfire event often begins unnoticed and spreads quickly, lighting brush, trees, and homes on fire. For example, a wildfire may be started by a campfire that was not doused properly, tossed cigarette, burning debris, or arson.

Texas has seen a significant increase in the number of wildfires in the past 30 years, which included wildland, interface, or intermix fires. Wildland Urban Interface or Intermix (WUI) fires occur in areas where structures and other human improvements meet or intermingle with undeveloped wildland or vegetative fuels. Wildland fires are fueled almost exclusively by natural vegetation, while interface or intermix fires are urban/wildland fires in which vegetation and the built-environment provide the fuel.

# Location

A wildfire event can be a potentially damaging consequence of drought. Wildfires can vary greatly in terms of size, location, intensity, and duration. While wildfires are not confined to any specific geographic location, they are most likely to occur in open grasslands. The threat to people and property from a wildfire event is greater in the fringe areas where developed areas meet open grass lands, such as the WUI. (Figures 13-1 through 13-9). It is estimated that 30 percent of the total population in Jefferson County live within the WUI. However, the entire Jefferson County planning area is at risk for wildfires.







Figure 13-2. Wildland Urban Interface Map – City of Beaumont, SETRPC

It is estimated that 23 percent of the total population in Beaumont live within the WUI. However, the entire City of Beaumont, including the SETRPC, is at risk for wildfires.



Figure 13-3. Wildland Urban Interface Map – City of Bevil Oaks

It is estimated that 83 percent of the total population in Bevil Oaks live within the WUI. However, the entire City of Bevil Oaks is at risk for wildfires.



Figure 13-4. Wildland Urban Interface Map – City of China

It is estimated that 96 percent of the total population in China live within the WUI. However, the entire City of China is at risk for wildfires.



Figure 13-5. Wildland Urban Interface Map – City of Groves

It is estimated that 7 percent of the total population in Groves live within the WUI. However, the entire City of Groves is at risk for wildfires.



Figure 13-6. Wildland Urban Interface Map – City of Nederland

It is estimated that 4 percent of the total population in Nederland live within the WUI. However, the entire City of Nederland is at risk for wildfires.



Figure 13-7. Wildland Urban Interface Map – City of Nome

It is estimated that 86 percent of the total population in Nome live within the WUI. However, the entire City of Nome is at risk for wildfires.



Figure 13-8. Wildland Urban Interface Map – City of Port Arthur

It is estimated that 16 percent of the total population in Port Arthur live within the WUI. However, the entire City of Port Arthur is at risk for wildfires.



Figure 13-9. Wildland Urban Interface Map – City of Port Neches

It is estimated that 19 percent of the total population in Port Neches live within the WUI. However, the entire City of Port Neches is at risk for wildfires.

### Section 13: Wildfire

# Extent



Risk for a wildfire event is measured in terms of magnitude and intensity using the Keetch Byram Drought Index (KBDI), a mathematical system for relating current and recent weather conditions to potential or expected fire behavior. The KBDI determines forest fire potential based on a daily water balance, derived by balancing a drought factor with precipitation and soil moisture (assumed to have a maximum storage capacity of 8 inches), and is expressed in hundredths of an inch of soil moisture depletion.

Each color in Figure 13-10 represents the drought index at that location. The drought index ranges from 0 to 800. A drought index of 0 represents no moisture depletion, and a drought index of 800 represents absolutely dry conditions.





Fire behavior can be categorized at four distinct levels on the KBDI:

• **0-200:** Soil and fuel moisture are high. Most fuels will not readily ignite or burn. However, with sufficient sunlight and wind, cured grasses and some light surface fuels will burn in spots and patches.

<sup>&</sup>lt;sup>1</sup> The black circle indicates the Jefferson County planning area.

### Section 13: Wildfire

- **200 -400:** Fires more readily burn and will carry across an area with no gaps. Heavier fuels will not readily ignite and burn. Expect smoldering and the resulting smoke to carry into and possibly through the night.
- **400 -600:** Fires intensity begins to significantly increase. Fires will readily burn in all directions exposing mineral soils in some locations. Larger fuels may burn or smolder for several days creating possible smoke and control problems.
- **600 -800:** Fires will burn to mineral soil. Stumps will burn to the end of underground roots and spotting will be a major problem. Fires will burn through the night and heavier fuels will actively burn and contribute to fire intensity.

The KBDI is a good measure of the readiness of fuels for a wildfire event. The KBDI should be referenced as the area experiences changes in precipitation and soil moisture, and caution exercised in dryer, hotter conditions.

The current range of intensity for Jefferson County in a wildfire event is within 0 to 200. The average extent to be mitigated for the Jefferson County planning area, including the SETRPC and all participating jurisdictions, is a KBDI of 566. At this level the intensity of fires begins to significantly increase and fires readily burn in all directions, exposing mineral soils in some locations.

The Texas Forest Service's Fire Intensity Scale identifies areas where significant fuel hazards and associated dangerous fire behavior potential exist based on the weighted average of 4 percentile weather categories. Jefferson County, including the SETRPC and all participating jurisdictions, is between a potential low to moderate wildfire intensities. Figures 13-11 through 13-19 identify the wildfire intensity for the Jefferson County planning area.











Figure 13-13. Fire Intensity Scale Map – Bevil Oaks



Figure 13-14. Fire Intensity Scale Map – China


Figure 13-15. Fire Intensity Scale Map – Groves



Figure 13-16. Fire Intensity Scale Map – Nederland



Figure 13-17. Fire Intensity Scale Map – Nome







Figure 13-19. Fire Intensity Scale Map – Port Neches

# **Historical Occurrences**

The Texas Forest Service reported 227 wildfire events between 2005 and 2015. The National Center for Environmental Center (NCEI) reported 1 event from 1996 through June 2016. The Texas Forest Service (TFS) started collecting wildfire data in 1985 and volunteer fire departments started reporting events after 2005. Due to a lack of recorded data for wildfire events prior to 2005, frequency calculations are based on a twelve-year period, using only data from recorded years. The map below shows approximate locations of wildfires, which can be grass or brushfires of any size (Figure 13-20). Table 13-1 identifies the number of wildfires by jurisdiction, and total acreage burned. It is important to note that the SETRPC is located within the City of Beaumont, however, none of the reported wildfire events have impacted the SETRPC.



Figure 13-20. Location and Historic Wildfire Events for Jefferson County

### Section 13: Wildfire

JURISDICTION	NUMBER OF EVENTS	ACRES BURNED
Jefferson County	228	4,207
Beaumont	13	345
Bevil Oaks	22	24
China	4	6
Groves	0	0
Nederland	4	4
Nome	0	0
Port Arthur	3	5
Port Neches	0	0

#### Table 13-1. Historical Wildfire Events Summary

#### Table 13-2. Acreage of Suppressed Wildfire by Year

JURISDICTION	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015
Jefferson County	157	235	0	929	766	339	1,308	331	139	2	1
Beaumont	0	0	0	4	6	18	317	0	0	0	0
Bevil Oaks	0	0	0	0	11	2	11	0	0	0	0
China	0	0	0	0	0	4	2	0	0	0	0
Groves	0	0	0	0	0	0	0	0	0	0	0
Nederland	0	0	0	0	0	0	2	1	1	0	0
Nome	0	0	0	0	0	0	0	0	0	0	0
Port Arthur	0	0	0	0	3	0	2	0	0	0	0
Port Neches	0	0	0	0	0	0	0	0	0	0	0

# **Probability of Future Events**

Wildfires can occur at any time of the year. As the jurisdictions within the County move into wildland, the potential area of occurrence of wildfire increases. With 228 events in a 12 year period, an event within Jefferson County, including the SETRPC and all participating jurisdictions, is highly likely, meaning an event is probable within the next year.

# Vulnerability and Impact

Periods of drought, dry conditions, high temperatures, and low humidity are factors that contribute to the occurrence of a wildfire event. Areas along railroads and people whose homes are in woodland settings have an increased risk of being affected by wildfire.

The heavily populated urban areas of Jefferson County are not likely to experience large and sweeping fires. Areas outside of city limits and in the unincorporated areas of Jefferson County are vulnerable. Unoccupied buildings and open spaces that have not been maintained have the greatest vulnerability to wildfire. The overall level of concern for wildfires is located mostly along the perimeter of the study area where wildland and urban areas interface. Figures 13-1 through 13-9 (above) illustrate the areas that are the most vulnerable to wildfire throughout the County.

The sparsely populated unincorporated areas of Dowling and Viterbo are capable of experiencing large sweeping fires, especially where areas of vegetation are not maintained. Areas along major highways in Cheek and China, as well as Jefferson County, have an increased vulnerability where empty lots and unoccupied areas are located.

The following critical facilities (Table 13-3) are located in the WUI and are more susceptible to wildfire in each participating jurisdiction:

JURISDICTION	CRITICAL FACILITIES
Jefferson County	Port Authority Facility, Fire Station, Water District Facility, 1 School
Beaumont	1 Fire Station, 7 Schools, 3 Water District Facilities, 1 Drainage District Facility, 1 Hospital
Bevil Oaks	Fire Station
China	Fire Station, 2 Schools
Groves	None
Nederland	Water District Facility
Nome	Fire Station
Port Arthur	None
Port Neches	1 School
SETRPC	None

#### Table 13-3. Critical Facilities Located in WUI by Jurisdiction

Within Jefferson County, a total of 228 fire events were reported from 2005 to 2016. All of these events were suspected wildfires. Historic loss and annualized estimates due to wildfires are presented in Table 13-4 below. The frequency is approximately 19 events every year.

JURISDICTION	NUMBER OF EVENTS	ACRES BURNED	ANNUAL ACRE LOSSES
Jefferson County	228	4,207	350.58
Beaumont	13	345	28.75
Bevil Oaks	22	24	2
China	4	6	0.5
Groves	0	0	0
Nederland	4	4	0.33
Nome	0	0	0
Port Arthur	3	5	0.42
Port Neches	0	0	0
SETRPC	0	0	0

#### Table 13-4. Historic Loss Estimates Due to Wildfire<sup>2</sup>

Figures 13-21 through 13-29 show Jefferson County and the threat of wildfire to the County and participating jurisdictions.

<sup>&</sup>lt;sup>2</sup> Events divided by 12 years of data.



Figure 13-21. Wildfire Ignition Density – Jefferson County

#### Section 13: Wildfire



Figure 13-22. Wildfire Ignition Density – Beaumont, SETRPC



Figure 13-23. Wildfire Ignition Density – Bevil Oaks







Figure 13-25. Wildfire Ignition Density – Groves



Figure 13-26. Wildfire Ignition Density – Nederland

### Section 13: Wildfire







Figure 13-28. Wildfire Ignition Density – Port Arthur



Figure 13-29. Wildfire Ignition Density – Port Neches

### Section 13: Wildfire

Diminished air quality is an environmental impact that can result from a wildfire event and pose a potential health risk. The smoke plumes from wildfires can contain potentially inhalable carcinogenic matter. Fine particles of invisible soot and ash that are too microscopic for the respiratory system to filter can cause immediate and possibly long term health effects. The elderly or those individuals with compromised respiratory systems may be more vulnerable to the effects of diminished air quality after a wildfire event.

Climatic conditions such as severe freezes and drought can significantly increase the intensity of wildfires since these conditions kill vegetation, creating a prime fuel source for wildfires. The intensity and rate at which wildfires spread are directly related to wind speed, temperature, and relative humidity.

The severity of impact from major wildfire events can be substantial. Such events can cause multiple deaths, shut down facilities for 30 days or more, and cause more than 50 percent of affected properties to be destroyed or suffer major damage. Severity of impact is gauged by acreage burned, homes and structures lost, and the number of resulting injuries and fatalities. For the Jefferson County planning area, the impact from a wildfire event can be considered "Minor", meaning injuries and/or illnesses are treatable with first aid, shutdown of facilities and services for 24 hours or less, and less than 10% of property is destroyed or sustains major damage.

### Assessment of Impacts

A wildfire event poses a potentially significant risk to public health and safety, particularly if the wildfire is initially unnoticed and spreads quickly. The impacts associated with a wildfire are not limited to the direct damages. Potential impacts for the planning area include:

- Persons in the area at the time of the fire are at risk for injury or death from burns and/or smoke inhalation.
- First responders are at greater risk of physical injury since they are in close proximity to the hazard while extinguishing flames, protecting property, or evacuating residents in the area.
- First responders can experience heart disease, respiratory problems, and other long term related illnesses from prolonged exposure to smoke, chemicals, and heat.
- Emergency services may be disrupted during a wildfire if facilities are impacted, roadways are inaccessible, or personnel are unable to report for duty.
- Critical City and/or County departments may not be able to function and provide necessary services depending on the location of the fire, and the structures or personnel impacted.
- Non-critical businesses may be directly damaged, suffer loss of utility services, or be otherwise inaccessible, delaying normal operations and slowing the recovery process.
- Displaced residents may not be able to immediately return to work, further slowing economic recovery.
- Roadways in or near the WUI could be damaged or closed due to smoke and limited visibility.
- Older homes are generally exempt from modern building code requirements, which may require fire suppression equipment in the structure.
- Some high density neighborhoods feature small lots with structures close together, increasing the potential for fire to spread rapidly.
- Air pollution from smoke may exacerbate respiratory problems of vulnerable residents.

### Section 13: Wildfire

- Charred ground after a wildfire cannot easily absorb rainwater, increasing the risk of flooding and potential mudflows.
- Wildfires can cause erosion, degrading stream water quality.
- Wildlife may be displaced or destroyed.
- Historical or cultural resources may be damaged or destroyed.
- Tourism can be significantly disrupted, further delaying economic recovery for the area.
- Vegetated dunes can be stripped, significantly damaging the function of the dunes to protect inland areas from the destructive forces of wind and waves.
- Economic disruption negatively impacts the programs and services provided by the community due to short and long term loss in revenue.
- Fire suppression costs can be substantial, exhausting the financial resources of the community.
- Residential structures lost in a wildfire may not be rebuilt for years, reducing the tax base for the community.
- Sabine Lake recreation and tourism can be unappealing for years following a large wildfire, devastating directly related businesses.
- Direct impacts to municipal water supply may occur through contamination of ash and debris during the fire, destruction of aboveground delivery lines, and soil erosion or debris deposits into waterways after the fire.

The economic and financial impacts of a wildfire event on local government will depend on the scale of the event, what is damaged, costs of repair or replacement, lost business days in impacted areas, and how quickly repairs to critical components of the economy can be implemented. The level of preparedness and pre-event planning done by government, businesses, and citizens will contribute to the overall economic and financial conditions in the aftermath of a wildfire event.

# **SECTION 14: WINTER STORM**

Hazard Description	1
Location	3
Extent	3
Historical Occurrences	4
Significant Past Events	5
Probability of Future Events	6
Vulnerability and Impact	6
Assessment of Impacts	8

# Hazard Description



A severe winter storm event is identified as a storm with snow, ice, or freezing rain. This type of storm can cause significant problems for area residents. Winter storms are associated with freezing or frozen precipitation such as freezing rain, sleet, snow and the combined effects of winter precipitation and strong winds. Wind chill is a function of temperature and wind. Low wind chill is a product of high winds and freezing temperatures.

Winter storms that threaten Jefferson County usually begin as powerful cold fronts that push south from central Canada. The County is at risk to ice hazards, extremely cold temperatures, and snow. However, the effects and frequencies of winter storm events are generally mild and short-lived. As indicated in Figure 14-1, on average, the area experiences 1-10 cold days a year, meaning 1-10 days per year are at or around freezing temperatures. During these times of ice and snow accumulation, response times will increase until public works road crews are able to assist in making the major roads passable. Table 14-1 describes the types of winter storms possible to occur in Jefferson County.



#### Figure 14-1. Extreme Cold Days 1960-2003<sup>1</sup>

#### Table 14-1. Types of Winter Storms

TYPE OF WINTER STORM	DESCRIPTION
Winter Weather Advisory	This alert may be issued for a variety of severe conditions. Weather advisories may be announced for snow, blowing or drifting snow, freezing drizzle, freezing rain, or a combination of weather events.
Winter Storm Watch	Severe winter weather conditions may affect your area (freezing rain, sleet, or heavy snow may occur separately or in combination).
Winter Storm Warning	Severe winter weather conditions are imminent.
Freezing Rain or Freezing Drizzle	Rain or drizzle is likely to freeze upon impact, resulting in a coating of ice glaze on roads and all other exposed objects.
Sleet	Small particles of ice usually mixed with rain. If enough sleet accumulates on the ground, it makes travel hazardous.

<sup>&</sup>lt;sup>1</sup> Source: National Weather Service. The black circle indicates the Jefferson County planning area.

TYPE OF WINTER STORM	DESCRIPTION
Blizzard Warning	Sustained wind speeds of at least 35 miles per hour (mph) are accompanied by considerable falling or blowing snow. This alert is the most perilous winter storm, with visibility dangerously restricted.
Frost/Freeze	Below freezing temperatures are expected and may cause significant damage
Warning	to plants, crops, and fruit trees.
Wind Chill	A strong wind combined with a temperature slightly below freezing can have the same chilling effect as a temperature nearly 50 degrees lower in a calm atmosphere. The combined cooling power of the wind and temperature on exposed flesh is called the wind chill factor.

## Location

Winter storm events are not confined to specific geographic boundaries. Therefore, all existing and future buildings, facilities, and populations in the Jefferson County planning area, including the SETRPC and all participating jurisdictions, are considered to be exposed to a winter storm hazard and could potentially be impacted.

# Extent

The extent or magnitude of a severe winter storm is measured in intensity based on the temperature and level of accumulations as shown in Table 14-2. To determine the intensity of a winter storm, Table 14-2 should be read in conjunction with the wind-chill factor chart described in Figure 14-2. The chart is an index developed by the National Weather Service (NWS) and is not applicable when temperatures are over 50°F or winds are calm.

INTENSITY	TEMPERATURE RANGE (Fahrenheit)	EXTENT DESCRIPTION					
Mild	$40^{\circ} - 50^{\circ}$	Winds less than 10 mph and freezing rain or light snow falling for short durations with little or no accumulations.					
Moderate	$30^{\circ} - 40^{\circ}$	Winds 10 to 15 mph and sleet and/or snow up to 4 inches.					
Significant	25° – 30°	Intense snow showers accompanied with strong gusty winds between 15 to 20 mph, with significant accumulation.					
Extreme	20° – 25°	Wind driven snow that reduces visibility, heavy winds (between 20 to 30 mph), and sleet or ice up to 5 millimeters in diameter.					
Severe	Below 20°	Winds of 35 mph or more and snow and sleet greater than 4 inches.					

#### Table 14-2. Magnitude of Severe Winter Storms

NEATHER

				MOILAN	SEA SEA	V	Vir	ld	Cł	hill	C	ha	rt						
									Tem	pera	ture	(°F)	. –						
		40	35	30	25	20	15	10	5	0	-5	-10	-15	-20	-25	-30	-35	-40	-45
	5	36	31	25	19	13	7	1	-5	-11	-16	-22	-28	-34	-40	-46	-52	-57	-63
	10	34	27	21	15	9	3	-4	-10	-16	-22	-28	-35	-41	-47	-53	-59	-66	-72
	15	32	25	19	13	6	0	-7	-13	-19	-26	-32	-39	-45	-51	-58	-64	-71	-77
	20	30	24	17	11	4	-2	-9	-15	-22	-29	-35	-42	-48	-55	-61	-68	-74	-81
(Y	25	29	23	16	9	3	-4	-11	-17	-24	-31	-37	-44	-51	-58	-64	-71	-78	-84
Ë	30	28	22	15	8	1	-5	-12	-19	-26	-33	-39	-46	-53	-60	-67	-73	-80	-87
ě	35	28	21	14	7	0	-7	-14	-21	-27	-34	-41	-48	-55	-62	-69	-76	-82	-89
M	40	27	20	13	6	-1	-8	-15	-22	-29	-36	-43	-50	-57	-64	-71	-78	-84	-91
	45	26	29	12	5	-2	-9	-16	-23	-30	-37	-44	-51	-58	-65	-72	-79	-86	-93
	50	26	19	12	4	-3	-10	-17	-24	-31	-38	-45	-52	-60	-67	-74	-81	-88	-95
	55	25	18	11	4	-3	-11	-18	-25	-32	-39	-46	-54	-61	-68	-75	-82	-89	-97
	60	25	17	10	3	-4	-11	-19	-26	-33	-40	-48	-55	-62	-69	-76	-84	-91	-98
					Frosth	ite Tir	nes	30	minut	es	10	minut		5 m	inutes				
									-										
			W	ind (	Chill	(°F) =	= 35.	74 +	0.62	15T ·	35.	75(V	<sup>).16</sup> ) -	+ 0.4	2751	Γ(V <sup>0.1</sup>	<sup>16</sup> )		
						Whe	ere, T=	Air Ter	nperat	ture (°	F) V=	Wind S	peed	(mph)			Effe	ctive 1	/01/01

Figure 14-2. Wind Chill Chart

Wind chill temperature is a measure of how cold the wind makes real air temperature feel to the human body. Since wind can dramatically accelerate heat loss from the body, a blustery 30°F day would feel just as cold as a calm day with 0°F temperatures. Jefferson County has never experienced a blizzard, but based on 10 previous occurrences recorded from 1996 through August 2016, it has been subject to winter storm watches, warnings, freezing rain, sleet, snow, and wind chill.

The average number of cold days is similar for the entire County planning area including the SETRPC and all participating jurisdictions. Therefore, the intensity or extent of a winter storm event to be mitigated for the area ranges from mild to significant according to the definitions from Table 14-2. During a winter storm event, the Jefferson County planning area, including the SETRPC and all participating jurisdictions, can expect anywhere between 0.1 to 3.0 inches of ice and snow, temperatures between 25 and 50 degrees, with winds ranging from 0 to 20 mph.

## **Historical Occurrences**

Table 14-3 shows historical occurrences for Jefferson County from 1996 through August 2016 provided by the National Center for Environmental Information (NCEI) database. There have been 10 recorded winter storm events in Jefferson County. Historical winter storm information, as provided by the NCEI, identifies winter storm activity across a multi-county forecast area for each event. The appropriate percentage of the total property and crop damage reported for the entire forecast area has been allocated to each county impacted by the event. Historical winter storm data for the SETRPC and all participating jurisdictions are provided on a county-wide basis per the NCEI database. Table 14-3 shows historical incident information which resulted in property or crop damage for the Jefferson County planning area.

JURISDICTION	DATE	DEATHS	INJURIES	PROPERTY DAMAGE	CROP DAMAGE
Jefferson County	2/4/1996	0	0	\$76,557	\$0
Jefferson County	1/12/1997	0	10	\$14,967,975	\$0
Jefferson County	12/11/2008	0	0	\$0	\$0
Jefferson County	12/4/2009	0	0	\$0	\$0
Jefferson County	1/8/2010	0	0	\$550,859	\$0
Jefferson County	2/23/2010	0	0	\$0	\$0
Jefferson County	2/3/2011	0	0	\$10,680	\$0
Jefferson County	1/23/2014	0	0	\$0	\$0
Jefferson County	1/28/2014	0	0	\$0	\$0
Jefferson County	3/4/2014	0	0	\$0	\$0
TOTAL	TOTALS		10	\$15,60	6,071

Table 14-3. Historical Winter Storm Events, 1996-2016<sup>2</sup>

Based on the list of historical winter storm events for the Jefferson County planning area (listed above), including the SETRPC and all participating jurisdictions, 4 events have occurred since the 2011 Plan.

### Significant Past Events

#### January 12 – 14, 1997 – Jefferson County

A record ice storm paralyzed southeast Texas and southwest Louisiana. Around 90,000 electric customers across southeast Texas were without power for up to 6 days. Emergency shelters were opened for several nights due to the cold weather following the ice storm. Hundreds of homes received minor damage due to trees or tree limbs falling on roofs. Several house fires were directly or indirectly related to the ice storm. Numerous traffic accidents attributed to icy roads led to several minor injuries. 1 death was indirectly attributed to the ice storm. 2 men were electrocuted on Tuesday, January 21, 1997 while doing cleanup work for a local electric company. A 48 year old man died, and a 19 year old man was seriously injured in the accident.

#### January 8 – 11, 2010 – Jefferson County

A deep upper level trough moving eastward across the United States forced a bitterly cold Arctic air mass southward from Canada into the Gulf Coast states on Thursday, January 7, 2010. This air mass remained in place for several days across southeast Texas, leading to the coldest temperatures seen across this region since February 1996. A few record low temperatures and record low maximum temperatures were set. Many locations in the Lakes Region of southeast Texas remained below freezing for over 36 hours

<sup>&</sup>lt;sup>2</sup> Values are in 2016 dollars.

from around midnight early on Friday, January 8, 2010 through the afternoon on Saturday, January 9, 2010.

The cold temperatures led to several school closures, numerous weather-related fires, and widespread plumbing ruptures throughout southeast Texas. The Insurance Council of Texas estimated losses across southeast Texas from the cold weather at around \$1 million. KFDM-TV and the Beaumont Enterprise reported widespread damaged plumbing across Jefferson County due to the record cold temperatures. 10 petrochemical plants and refineries across the county experienced excessive emissions due to cracked pipes, instrument failures, and equipment malfunctions. Entergy reported sporadic power outages through the county. CenterPoint Energy used trucks to pump natural gas directly into a northwest Beaumont neighborhood of 100 homes after natural gas lines failed due to the cold weather.

# **Probability of Future Events**

According to historical records, Jefferson County experiences approximately 1 winter storm event per year. Hence, the probability of a future winter storm event affecting the Jefferson County planning area is highly likely, with a winter storm likely to occur within the next year. All participating jurisdiction events including the SETRPC are included under the County.

## Vulnerability and Impact

During periods of extreme cold and freezing temperatures, water pipes can freeze and crack and ice can build up on power lines, causing them to break under the weight or causing tree limbs to fall on the lines. These events can disrupt electric service for long periods.

An economic impact may occur due to increased consumption of heating fuel, which can lead to energy shortages and higher prices. House fires and resulting deaths tend to occur more frequently from increased and improper use of alternate heating sources. Fires during winter storms also present a greater danger because water supplies may freeze and impede firefighting efforts.

All populations, buildings, critical facilities, and infrastructure in the entire Jefferson County planning area, including all participating jurisdictions, are vulnerable to severe winter events.

People and animals are subject to health risks from extended exposure to cold air. Elderly people are at greater risk of death from hypothermia during these events, especially in the rural areas of the county where populations are sparse, icy roads may impede travel, and there are fewer neighbors to check in on the elderly. According to the U.S. Center for Disease Control, every year hypothermia kills about 600 Americans, half of whom are 65 years of age or older.

Populations over 65 in the Jefferson County planning area are approximately 13% of the total population; there is an estimated total of 32,774<sup>3</sup> potentially vulnerable residents in the planning area based on age (Table 14-4).

<sup>&</sup>lt;sup>3</sup> Source: U.S. Census Bureau 2014 data for Jefferson County.

JURISDICTION	POPULATION 65 AND OLDER				
Beaumont	15,539				
Bevil Oaks	290				
China	183				
Groves	2,685				
Nederland	2,371				
Nome	56				
Port Arthur	6,344				
Port Neches	1,939				
Jefferson County⁴	32,774				

#### Table 14-4. Populations at Greater Risk by Jurisdiction

The following critical facilities would be vulnerable to winter storm events in each participating jurisdiction:

Table 14-5.	Critical	Facilities	by	Jurisdiction
-------------	----------	------------	----	--------------

JURISDICTION	CRITICAL FACILITIES
Jefferson County	Port Authority Facility, Fire Station, Water District Facility, 1 School
Beaumont	2 Fire Stations, 2 Police Stations, 32 Schools, Port Authority Facility, 5 Water District Facilities, 3 Drainage District Facilities, 4 Hospitals
Bevil Oaks	Fire Station
China	Fire Station, 2 Schools
Groves	Fire Station, Police Station, 4 Schools
Nederland	Fire Station, Police Station, Water District Facility, 3 Hospitals, Airport, 8 Schools
Nome	Fire Station
Port Arthur	Fire Station, Police Station, Drainage District Facility, 2 Port Authority Facilities, 2 Hospitals, 14 Schools
Port Neches	Fire Station, Police Station, 5 Schools
SETRPC	SETRPC Facility

<sup>&</sup>lt;sup>4</sup> County totals includes all participating jurisdictions and unincorporated areas.

Historic loss, in 2016 dollars, is estimated at \$15,606,071 in damages over the 21-year recording period, giving an approximate loss of \$743,146 in damages annually (Table 14-6). The potential severity of impact for the planning area, including the SETRPC and all participating jurisdictions, is limited, meaning injuries are treatable with first aid, shutdown of facilities and services for 24 hours or less, and less than 10% of property destroyed or sustains major damage.

Table 14-6. Potential Annualized Losses for Jefferson County, 1996-201	. <b>6</b> <sup>5</sup>
--	-------------------------

JURISDICTION	PROPERTY & CROP LOSS	ANNUALIZED LOSS ESTIMATES
Jefferson County	\$15,606,071	\$743,146

### Assessment of Impacts

The greatest risk from a winter storm hazard is to public health and safety. Potential impacts for the planning area may include:

- Vulnerable populations, particularly the elderly and infants, can face serious or life-threatening health problems from exposure to extreme cold including hypothermia and frostbite.
- Loss of electric power or other heat sources can result in increased potential for fire injuries or hazardous gas inhalation because residents burn candles for light or use fires or generators to stay warm.
- Response personnel, including utility workers, public works personnel, debris removal staff, tow truck operators, and other first responders are subject to injury or illness resulting from exposure to extreme cold temperatures.
- Response personnel would be required to travel in potentially hazardous conditions, elevating the safety risk due to accidents and potential contact with downed power lines.
- Operations or service delivery may experience impacts from electricity blackouts due to winter storms.
- Power outages are possible throughout the planning area due to downed trees and power lines and/or rolling blackouts.
- Critical facilities without emergency backup power may not be operational during power outages.
- Emergency response and service operations may be impacted by limitations on access and mobility if roadways are closed, unsafe, or obstructed.
- Hazardous road conditions will likely lead to increases in automobile accidents, further straining emergency response capabilities.
- Depending on the severity and scale of damage caused by ice and snow events, damage to power transmission and distribution infrastructure can require days or weeks to repair.
- A winter storm event could lead to tree, shrub, and plant damage or death.
- Severe cold and ice could significantly damage agricultural crops.
- Schools may be forced to shut early due to treacherous driving conditions.

<sup>&</sup>lt;sup>5</sup> Values are in 2016 dollars.

• Exposed water pipes may be damaged by severe or late season winter storms at both residential and commercial structures, causing significant damages.

The economic and financial impacts of winter weather on the community will depend on the scale of the event, what is damaged, and how quickly repairs to critical components of the economy can be implemented. The level of preparedness and pre-event planning done by businesses and citizens will also contribute to the overall economic and financial conditions in the aftermath of a winter storm event.

# **SECTION 15: COASTAL EROSION**

Hazard Description	1
Location	1
Extent	2
Historical Occurrences	4
Probability of Future Events	4
Vulnerability and Impact	4
Assessment of Impacts	5

## Hazard Description

Coastal erosion is the wearing away of land and loss of beach, shoreline, or dune material because of natural coastal processes or manmade influences. Erosion is the process by which large storms, flooding, strong wave action, sea level rise, and human activities wear away beaches and bluffs along coastlines. All beaches are affected by storms and other natural events that cause erosion; however, the extent and severity of the problem differs in different parts of the country. The two major erosion mechanisms are wind and water. Wind that blows across sparsely vegetated or disturbed lands can cause erosion by picking up soil, carrying it through the air, and displacing it in another place. Water erosion occurs over land, and in streams and channels. Major storms can cause coastal erosion from the combination of high winds and heavy surf and storm surge. Human interactions, such as construction and development in coastal and riparian regions, can also exacerbate erosion.

While coastal erosion affects all regions of the United States, erosion rates and potential impacts are highly localized. Average coastline recession rates of 25 feet per year are not uncommon on some barrier islands in the Southeast. Texas has one of the longest coastlines in America coupled with some of the highest rates of coastal erosion in the nation. Sixty-four percent of the Texas coast is eroding at an average of 6 feet per year, with an overall average rate of 4.1 feet per year for the 367 miles of Texas coast, according to the Texas General Land Office. However, some locations are losing more than 30 feet per year. Coastal erosion can have long-term economic and social consequences.

### Location

While the Jefferson County planning area is considered a coastal community, only one of the participation jurisdictions is located directly on the coast and is subject to coastal erosion. The McFaddin National Wildlife Refuge is located in unincorporated Jefferson County and expands the entire coast line of the county. The wildlife refuge is vulnerable to threats directly related to coastal erosion resulting from extreme hazards such as hurricane and tropical storm events. The most common time for such extreme storm events to impact the planning area is from June to November, the official Atlantic U.S. hurricane season. The water front community of Port Arthur is located on the mainland and protected by the barrier island system along the gulf. As such, this community is not subject to coastal erosion.



Figure 15-1. Critical Eroding Areas of the Texas Gulf Shoreline

### Extent

Some of the highest rates of Gulf shoreline erosion in Texas occur in Jefferson County.<sup>1</sup> The McFaddin National Wildlife Refuge is vulnerable to the effects of coastal erosion from the Gulf of Mexico. The island has no stable (vegetated) dunes in the area located as close to the mean low water (MLW) line. Through experience it has proven that barrier island development imposes risks on private property owners, investors, and to taxpayers statewide. The average rate of retreat or extent of coastal erosion is estimated between 4.9 and more than 14.8 feet per year for Jefferson County. The highest erosion rate occurs at the eastern most portion of the wildlife refuge near the Sabine Pass.

<sup>&</sup>lt;sup>1</sup> Texas Coastal Resiliency Master Plan, March 2017, Texas General Land Office





<sup>&</sup>lt;sup>2</sup> Coastal Erosion Planning and Response Act, A Report to the 84<sup>th</sup> Texas Legislature, Texas General Land Office, 2015

# Historical Occurrences

Previous occurrences for coastal erosion are not reported by the NCEI. In addition, local governments do not typically have the capabilities to monitor or report statistical data for coastal erosion for a specific event. Coastal erosion is typically measured as an average annual shoreline change rate in linear feet. While the Jefferson County Planning area does not record historical coastal erosion rates per event, the 2013 Texas Hazard Mitigation Plan depicts coastal erosion occurrences for the Jefferson County Planning Area, including unincorporated areas along the coast (Table 15-1).

JURISDICTION	GULF SHORELINE	BAY SHORELINE	CRITICAL EROSION	EROSION RATES
Jefferson County	168,960 ft.	327,360 ft.	142,560 ft.	-2 to -50 ft./yr.

# Probability of Future Events

Due to data limitations, the planning team relied on available studies and research as well as the Texas State Hazard Mitigation Plan to determine coastal erosion probability. According to Texas General Land Office (GLO) the average coastal erosion rate for unincorporated Jefferson County is from 4 to more than 15 feet per year with an average of approximately 13 feet per year. This rate supports a highly likely probability of future events, with an event probable in the next year.

# Vulnerability and Impact

The unincorporated area of Jefferson County along the coast is continuously subject to coastal erosion, as all barrier islands are. While usually a slow-evolving hazard, coastal erosion presents a serious threat to the McFaddin National Wildlife Refuge. Any loss of land equates to an increase in the planning areas vulnerability to hurricanes, coastal storms and above-average tidal events. When the land lost is beach that provides valuable protections from these coastal storm events, that loss results in greater vulnerability.

The wildlife refuge has a total area of 58,861 acres and is primarily tidally influenced, creating estuarine environments important to a variety of fish, shrimp and crabs, as well as other life forms higher on the food chain that feed on such organisms. These estuaries are productive communities and are vital to the life cycle of many marine species. Land in the refuge is vulnerable to coastal erosion, particularly on the eastern portions near Sabine Pass) which in turn threatens fish and wildlife in the area due to loss of habitat. There are no critical structures or infrastructure vulnerable to coastal erosion in the planning area.

The potential severity of impact from coastal erosion for the Jefferson County planning area is classified as limited, meaning minor quality of life is lost and shutdown of critical facilities; services are loss less than 24 hours; and less than 10 percent of property would be destroyed or have major damage.

<sup>&</sup>lt;sup>3</sup> State of Texas Mitigation Plan Update 2013 Page 126 as reported by the Texas General Land Office

### Section 15: Coastal Erosion

### Assessment of Impacts

Coastal erosion events have the potential to pose a significant risk to structures, infrastructure and the local economy. Impacts to the planning area can include:

- Fish and wildlife habitat can be damaged or destroyed. Extreme erosion, typically resulting from a significant storm event, may result in significant marine and wildlife losses.
- Damaged bridges in and out of the wildlife refuge areas could prevent or delay emergency response, strand or prevent entry of tourists, commuters, supply delivery, or goods and services for extended periods.
- Coastal erosion may dramatically reduce tourism negatively impacting the economy.
- Economic disruption negatively impacts the programs and services provided by the community due to short and long term loss in revenue.
- Some businesses not directly damaged by the coastal erosion may be negatively impacted while access roads or camping areas are repaired.

The economic and financial impacts of coastal erosion on the area will depend entirely on the scale of the event, what is damaged, and how quickly repairs to critical components of the economy can be implemented. The level of preparedness and pre-event planning done by the community, local businesses and citizens will also contribute to the overall reduction of coastal erosion impacts.

# **SECTION 16: MITIGATION STRATEGY**

litigation Goals	. 1
Goal 1	. 1
Goal 2	. 1
Goal 3	.2
Goal 4	.2
Goal 5	.2
Goal 6	.3

### **Mitigation Goals**

Based on the results of the risk and capability assessments, the Planning Team developed and prioritized the mitigation strategy. This involved utilizing the results of both assessments and reviewing the goals and objectives that were included in the previous 2011 Plan.

At the Mitigation Workshop in August 2016, Planning Team members reviewed the mitigation strategy from the previous 2011 Plan. The consensus among all members present was that the strategy developed for the 2011 Plan did not require changes, as it identified overall improvements to be sought in the Plan Update. However, the order and priority of the goals and objectives were reorganized.

### Goal 1

Protect public health and safety.

#### **Objective 1.1**

Advise the public about health and safety precautions to guard against injury and loss of life from hazards.

#### **Objective 1.2**

Maximize utilization of the latest technology to provide adequate warning, communication, and mitigation of hazard events.

#### **Objective 1.3**

Reduce the danger to, and enhance protection of, high risk areas during hazard events.

#### Objective 1.4

Protect critical facilities and services.

### Goal 2

Build and support local capacity and commitment to continuously become less vulnerable to hazards.
### Section 16: Mitigation Strategy

### Objective 2.1

Build and support local partnerships to continuously become less vulnerable to hazards.

### **Objective 2.2**

Build a cadre of committed volunteers to safeguard the community before, during, and after a disaster.

### **Objective 2.3**

Build hazard mitigation concerns into county planning and budgeting processes.

### Goal 3

Increase public understanding, support, and demand for hazard mitigation.

### **Objective 3.1**

Heighten public awareness regarding the full range of natural and man-made hazards the public may face.

### **Objective 3.2**

Educate the public on actions they can take to prevent or reduce the loss of life or property from all hazards and increase individual efforts to respond to potential hazards.

### **Objective 3.3**

Publicize and encourage the adoption of appropriate hazard mitigation measures.

### Goal 4

Protect new and existing properties.

#### *Objective 4.1*

Reduce repetitive losses to the National Flood Insurance Program (NFIP).

#### **Objective 4.2**

Use the most cost-effective approach to protect existing buildings and public infrastructure from hazards.

#### **Objective 4.3**

Enact and enforce regulatory measures to ensure that future development will not put people in harm's way or increase threats to existing properties.

### Goal 5

Maximize the resources for investment in hazard mitigation.



### Section 16: Mitigation Strategy



### **Objective 5.1**

Maximize the use of outside sources of funding.

### **Objective 5.2**

Maximize participation of property owners in protecting their properties.

### *Objective 5.3*

Maximize insurance coverage to provide financial protection against hazard events.

### **Objective 5.4**

Prioritize mitigation projects, based on cost-effectiveness and sites facing the greatest threat to life, health and property.

### Goal 6

Promote growth in a sustainable manner.

### Objective 6.1

Incorporate hazard mitigation activities into long-range planning and development activities.

### **Objective 6.2**

Promote beneficial uses of hazardous areas while expanding open space and recreational opportunities.

### **Objective 6.3**

Utilize regulatory approaches to prevent creation of future hazards to life and property.

# **SECTION 17: PREVIOUS ACTIONS**

Summary	1
Jefferson County	2
Beaumont	19
Bevil Oaks	47
China	60
Groves	71
Nederland	86
Nome	101
Port Arthur	112
Port Neches	132
South East Texas Regional Planning Commission (SETRPC)	144

## Summary

Planning Team members were given copies of the previous mitigation actions submitted in the 2011 Plan at the mitigation workshop. Jefferson County reviewed the previous actions and provided an analysis as to whether the action had been completed, should be deferred as an ongoing activity, or be deleted from the Plan. The actions from the 2011 Plan are included in this section as they were written in 2011, with the exception of the "2017 Analysis" section.

# Jefferson County

	Jefferson County (Past Action) – 1
Proposed Action:	To build a structure or structures (including a dome or domes) in coastal (or near coastal) jurisdictions that can withstand 200 mile per hour
	winds and act as shelters. This is in conjunction with the Texas Safe Shelter Initiative.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal	
Priority:	High	
Estimated Cost:	\$1.6 Million	
Implementation Schedule:	1-3 years	
Coordinating Agency:	Jefferson County Emergency Management, TDEM	
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM	

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Jefferson County (Past Action) – 2
Proposed Action:	Retrofit existing structures and/or construct new structures to act as shelters during and after Hurricanes/Tropical Storms and other severe weather events.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal	
Priority:	High	
Estimated Cost:	\$1 - \$2 Million	
Implementation Schedule:	1-3 years	
Coordinating Agency:	Jefferson County, SETRPC, TDEM	
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM	

### 2017 Analysis:

	Jefferson County (Past Action) – 3
Proposed Action:	Storm harden/retrofit critical facilities throughout Jefferson County.
	Actions can include but are not limited to window shutters, roof straps,
	flood proofing, roll-up door reinforcement (i.e. for fire stations, Ford
	Park).

MITIGATION ACTION DETAILS		
Hazard Addressed:	Dam Failure, Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado, Tsunami	
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic	
Priority:	High	
Estimated Cost:	\$10,000 - \$500,000	
Implementation Schedule:	1-2 years	
Coordinating Agency:	Jefferson County, municipalities	
Potential Funding Sources:	HGMP, PDM, FMA, local operating budgets	

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

			J	efferson Co	ounty (Pa	st Action) – 4
Proposed Action:	Public structure S.O./Jail.	strengthening	for	Jefferson	County	Courthouse/

MITIGATION ACTION DETAILS		
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic	
Priority:	High	
Estimated Cost:	\$1,000 - \$300,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	Jefferson County	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update. Revise Estimated Cost to \$200,000 - \$1 Million.

	Jefferson County (Past Action) – 5
Proposed Action:	Acquire flood prone properties (including Repetitive Loss and Severe
	Repetitive Loss Properties).

MITIGATION ACTION DETAILS		
Hazard Addressed:	Dam Failure, Flooding, Hurricane/Tropical Storm, Thunderstorm	
STAPLE-E Summary:	Technical, Administrative, Economic, Environmental	
Priority:	High	
Estimated Cost:	\$50,000 - \$140,000	
Implementation Schedule:	12 months	
Coordinating Agency:	Jefferson County, local municipalities	
Potential Funding Sources:	HMGP, FMA, PDM, RFC, SRL	

#### 2017 Analysis:

**Defer Action** – Will include in the 2017 Plan Update. Revise Estimated Cost to \$500,000 - \$1 Million; revise Implementation Schedule to 2 years.

	Jefferson County (Past Action) – 6
Proposed Action:	Storm harden/retrofit regional communication sites and infrastructure throughout the County.

MITIGATION ACTION DETAILS						
Hazard Addressed:	Dam Failure, Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado					
STAPLE-E Summary:	Technical, Administrative, Legal					
Priority:	High					
Estimated Cost:	\$300,000					
Implementation Schedule:	1-3 years					
Coordinating Agency:	Jefferson County Emergency Management					
Potential Funding Sources:	HMGP, PDM, EMPG					

#### 2017 Analysis:

	Jefferson County (Past Action) – 7							
Proposed Action:	Elevate new and existing flood prone structures and infrastructure throughout Jefferson County.							

MITIGATION ACTION DETAILS	
Hazard Addrossod:	Dam Failure, Flooding, Hurricane/Tropical Storm, Thunderstorm,
Hazaru Aduresseu:	Tsunami
STAPLE-E Summary:	Technical, Administrative, Economic
Priority:	High
Estimated Cost:	\$20,000 - \$100,000
Implementation Schedule:	1-3 years
Coordinating Agency:	Jefferson County, local municipalities
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL

#### 2017 Analysis:

Defer Action - Will include in the 2017 Plan Update. Revise Estimated Cost to \$200,000 - \$1 Million

Jefferson County (Past Action) – 8								
Proposed Action:	Replace througho	and/or out Jeffei	upgrade rson Count	bridges, :y.	culverts,	and	other	crossings

MITIGATION ACTION DETAILS					
Hazard Addressed:	Dam Failure, Flooding, Geologic Hazards, Hurricane/Tropical Storm,				
	Terrorism, Thunderstorm, Tornado, Tsunami				
STAPLE-E Summary:	Technical, Administrative, Legal				
Priority:	High				
Estimated Cost:	\$200,000 - \$1 Million				
Implementation Schedule:	2-5 years				
Coordinating Agency:	Jefferson County, TXDOT				
Potential Funding Sources:	HMGP, PDM, local operating budgets				

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update. Revise Estimated Cost to \$2 -\$5 Million.

	Jefferson County (Past Action) – 9
Proposed Action:	Pursue drainage improvements through Jefferson County. Actions can
	include but are not limited to installing/upgrading culverts and
	headwalls as well as enlarging storm water ditches and canals.

MITIGATION ACTION DETAILS						
Hazard Addressed:	Dam Failure, Flooding, Hurricanes/Tropical Storms, Thunderstorms					
STAPLE-E Summary:         Technical, Administrative, Political, Legal, Economic						
Priority:	High					
Estimated Cost:	\$50,000 - \$500,000					
Implementation Schedule:	1-3 years					
Coordinating Agency:	Jefferson County, local municipalities					
Potential Funding Sources:	HMGP, PDM, FMA local operating budgets					

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update. Revise Estimated Cost to \$1 - \$5 Million.

	Jefferson County (Past Action) – 10
Proposed Action:	Coordinate with federal, state, and local partners to provide training opportunities for first responders, including but not limited to HAZMAT terrorism all bazard and other training

MITIGATION ACTION DETAILS							
Hazard Addressed:	Dam Failure, Drought, Earthquake, Extreme Heat, Flood, Geologic Hazard, Hazardous Materials, Hurricane/Tropical Storm, Severe Winter Weather, Terrorism, Thunderstorm, Tornado, Tsunami, Water Contamination, Wildfire						
STAPLE-E Summary:	Technical, Administrative, Political, Legal, Economic						
Priority:	High						
Estimated Cost:	\$5,000 - \$10,000						
Implementation Schedule:	6-12 months						
Coordinating Agency:	Jefferson County Emergency Management						
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM						

#### 2017 Analysis:

	Jefferson County (Past Action) – 11
Proposed Action:	Bury underground, secure or otherwise harden exposed or vulnerable pipelines.

MITIGATION ACTION DETAILS								
Hazard Addressed:	Flood, Hazardous Material, Hurricane/Tropical Storm, Thunderstorm, Tornado							
STAPLE-E Summary:	Technical							
Priority:	High							
Estimated Cost:	\$100,000 - \$500,000							
Implementation Schedule:	3-5 years							
Coordinating Agency:	Jefferson County, local municipalities, Entergy							
Potential Funding Sources:	HMGP, PDM, local operating budgets							

#### 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update. Revise action to include Water/Sewer.

			Jeff	erson Cou	unty	(Past Act	tion) – 12
Proposed Action:	Provide	generators/back-up	power	systems	for	critical	facilities
	(including and othe	g but not limited to lif r first responder facil	t station ities) thr	s, water p oughout t	lants he Je	, police, efferson	EMS, Fire County.

MITIGATION ACTION DETAILS	
Hazard Addressed: Earthquake, Hurricane/Tropical Storm, Thunderstorm, Torn	
STAPLE-E Summary:	Technical, Administrative, Political, Legal
Priority:	High
Estimated Cost:	\$50,000 - \$200,000
Implementation Schedule:	1-3 years
Coordinating Agency:	Jefferson County, local municipalities
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM

#### 2017 Analysis:

**Defer Action** – Will include in the 2017 Plan Update. Progress has been made, but not fully completed. Revise Estimated Cost to \$1 - \$2 Million.

	Jefferson County (Past Action) – 13
Proposed Action:	Pursue flood protection measures to protect from surge from
	Hurricanes/Tropical Storms. Actions can include but are not limited to
	constructing and/or upgrading sea walls, flood barriers, berms and
	various wet and dry flood proofing measures.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tsunami, Water	
	Contamination	
STAPLE-E Summary:	Technical, Administrative, Political	
Priority:	High	
Estimated Cost:	\$50,000 - \$3 Million	
Implementation Schedule:	1-5 years	
Coordinating Agency:	Jefferson County, USACE, TDEM	
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL	

2017 Analysis:
Defer Action – Will include in the 2017 Plan Update. Baffles at Keith Lake (\$2.5 Million) completed in June
2015. Will pursue additional locations. Revise Estimated Cost to \$50,000 - \$60 Million.

_		Jefferson County (Past Action) – 14
	Proposed Action:	Pursue the coordination, construction, expansion and maintenance of flood control structures/barriers for the purpose of mitigating damage and protect freshwater resources from storm surge, sea level rise and other sources of salt water intrusion.
		flood control structures/barriers for the purpose of mitigating dam and protect freshwater resources from storm surge, sea level rise a other sources of salt water intrusion.

MITIGATION ACTION DETAILS	
Hazard Addressed: Hurricane/Tropical Storm	
STAPLE-E Summary:	Technical, Administrative, Economic
Priority:	High
Estimated Cost:	\$1 - \$2 Million
Implementation Schedule:	2-5 years
Coordinating Agency:	Jefferson County, USACE, SETRPC, FEMA, local municipalities
Potential Funding Sources:	HGMP, PDM, USACE, Jefferson County, SETRPC

	Jefferson County (Past Action) – 15
Proposed Action:	Restore sand dunes to protect inland resources during storm surge
	events.

MITIGATION ACTION DETAILS		
Hazard Addressed: Hurricanes/Tropical Storms		
STAPLE-E Summary:	Technical, Administrative, Legal, Environmental, Economic	
Priority:	High	
Estimated Cost:	\$60,000,000	
Implementation Schedule:	1-2 years	
Coordinating Agency:	Jefferson County	
Potential Funding Sources:	PDM, HMGP, FMA, local operating budgets	

#### 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update. Revise Estimated Cost to \$60,000,000 - \$100,000,000.

					Jefferson County (Pa	ast Actio	on) – 16
Proposed Action:	Construct	or	improve	existing	detention/retention	ponds	where
	appropriat	e to	collect st	orm wate	r to reduce flooding.		

MITIGATION ACTION DETAILS		
Hazard Addressed:	Dam Failure, Flooding, Hurricanes/Tropical Storm, Thunderstorms, Tsunami	
STAPLE-E Summary:	Social, Technical, Administrative, Economic, Environmental	
Priority:	High	
Estimated Cost:	\$50,000 - \$100,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	Jefferson County, local municipalities	
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL, local operating budgets	

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update. Revise Estimated Cost to \$50,000 - \$1 Million.

	Jefferson County (Past Action) – 17
Proposed Action:	Retrofit existing structures or construct new structures to act as safe rooms during tornados or other severe weather events.

MITIGATION ACTION DETAILS		
Hazard Addressed: Tornados, Thunderstorms		
STAPLE-E Summary:	Social, Technical, Administrative, Economic	
Priority:	High	
Estimated Cost:	\$5,000 - \$100,000	
Implementation Schedule:	1-5 years	
Coordinating Agency:	Jefferson County	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update. Revise Estimated Cost to \$50,000 - \$1 Million.

	Jefferson County (Past Action) – 18
Proposed Action:	Identify and pursue any mitigation activities that would aid/ enhance evacuations throughout the Jefferson County.

MITIGATION ACTION DETAILS					
Hazard Addressed:	Dam Failure, Flood, Hazardous Materials, Hurricane/Tropical Stor Thunderstorm, Tornado, Wildfire				
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic				
Priority:	High				
Estimated Cost:	\$10,000 - \$500,000				
Implementation Schedule:	1-5 years				
Coordinating Agency:	Jefferson County, local municipalities				
Potential Funding Sources:	HMGP, PDM, local operating budgets				

#### 2017 Analysis:

	Jefferson County (Past Action) – 19
Proposed Action:	Construct water retention ponds to collect storm water run-off and use
	as an alternate water source throughout Jefferson County.

MITIGATION ACTION DETAILS				
Hazard Addressed:	Drought, Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado, Water Contamination			
STAPLE-E Summary:	Social, Technical, Administrative, Environmental			
Priority:	High			
Estimated Cost:	\$75,000 - \$200,000			
Implementation Schedule:	1-3 years			
Coordinating Agency:	Jefferson County, local municipalities			
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets			

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update. Revise Priority to Low.

	Jefferson County (Past Action) – 20
Proposed Action:	Coordinate and work with the Lower Neches Valley Authority in order to use an LNVA sand pit as a potential freshwater reservoir for all areas south of I-10.

MITIGATION ACTION DETAILS							
Hazard Addressed:	Drought, Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado, Water Contamination						
STAPLE-E Summary:	Social, Technical, Administrative, Environmental						
Priority:	High						
Estimated Cost:	\$75,000 - \$200,000						
Implementation Schedule:	1-3 years						
Coordinating Agency:	Jefferson County, local municipalities, Lower Neches Valley Authority						
Potential Funding Sources:	HMGP, PDM, FMA, LNVA, local operating budget						

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update. Revise Priority to Low.

	Jefferson County (Past Action) – 21				
Proposed Action:	Mitigate damage to utilities in order to maintain function during and after a hazard event. Actions can include but are not limited to:				
	<ul> <li>Burying utility lines underground</li> </ul>				
	<ul> <li>Provide frangible links/break away connections on utility poles</li> </ul>				
	<ul> <li>Harden utility poles by converting from wood to concrete or metal utility poles</li> </ul>				
	<ul> <li>Increasing the easement area/clearance of utility lines/ poles</li> </ul>				
	from tree lines				

MITIGATION ACTION DETAILS				
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado			
STAPLE-E Summary:	Social, Technical, Administrative, Legal, Economic			
Priority:	High			
Estimated Cost:	\$100,000 - \$200,000			
Implementation Schedule:	1-5 years			
Coordinating Agency:	Jefferson County, local municipalities, Entergy			
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM			

**Defer Action –** Will include in the 2017 Plan Update.

		Jefferson County (	(Past Action)	- 22
Proposed Action:	Joint, multi-jurisdictional EOC.			

MITIGATION ACTION DETAILS					
Hazard Addressed:	Dam Failure, Drought, Extreme Heat, Flood, Hazardous Materials Hurricane/Tropical Storm, Thunderstorm, Tornado, Tsunami, Wate Contamination, Wildfire				
STAPLE-E Summary:	Technical, Administrative, Political				
Priority:	Medium				
Estimated Cost:	\$1-2 Million				
Implementation Schedule:	1-5 years				
Coordinating Agency:	Jefferson County, TDEM, FEMA				
Potential Funding Sources:	HMGP, PDM, EMGP, Homeland Security Grants				

#### 2017 Analysis:

 Jefferson County (Past Action) – 23									
Proposed Action:	Provide	the	public	with	educational	brochures	for	the	hazards
	identifie	d as p	oart of t	he 201	L1 Plan Updat	e.			

MITIGATION ACTION DETAILS						
Hazard Addressed:	Drought, Earthquake, Extreme Heat, Flood, Geologic Hazar Hazardous Materials, Hurricane/Tropical Storm, Severe Wint Weather, Terrorism, Thunderstorm, Tornado, Tsunami, Wat Contamination, Wildfire					
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic					
Priority:	Medium					
Estimated Cost:	\$5,000 - \$50,000					
Implementation Schedule:	6-12 months					
Coordinating Agency:	Jefferson County, local municipalities, TDEM, FEMA					
Potential Funding Sources:	HMGP, PDM, local operating budgets, EMPG					

**Defer Action –** Will include in the 2017 Plan Update.

	Jefferson County (Past Action) – 24
Proposed Action:	Retrofit existing structures to act as cooling stations in times of extreme heat.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Extreme Heat
STAPLE-E Summary:	Social, Technical, Administrative
Priority:	Medium
Estimated Cost:	\$100,000
Implementation Schedule:	1-3 years
Coordinating Agency:	Jefferson County, local municipalities
Potential Funding Sources:	HMGP, PDM, local operating budget

#### 2017 Analysis:

	Jefferson County (Past Action) – 25
Proposed Action:	Develop areas of defensible space to prevent damage due to wildfires.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Wildfire	
STAPLE-E Summary:	Technical, Administrative, Economic	
Priority:	Medium	
Estimated Cost:	\$50,000 - \$100,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	Jefferson County Emergency Management, local fire departments	
Potential Funding Sources:	HMGP, EMPG, PDM, local operating budgets	

### 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update. Revise Priority to Low.

	Jefferson County (Past Action) – 26
Proposed Action:	Develop and enact water conservation or drought management plans, ordinances or strategies to be used during times of drought.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Drought
STAPLE-E Summary:	Social, Technical, Administrative
Priority:	Medium
Estimated Cost:	\$25,000 - \$50,000
Implementation Schedule:	6-12 months
Coordinating Agency:	Jefferson County
Potential Funding Sources:	HMGP, PDM, local operating budgets

#### 2017 Analysis:

	Jefferson County (Past Action) – 27	
Proposed Action:	Minimize damage to structures and infrastructure from falling trees. Actions include but are not limited to the following:	
	<ul> <li>Pursue and coordinate a dangerous tree and limb removal program to protect infrastructure and critical facilities from damage. This includes working with private homeowners for voluntary removal of hazardous trees and limbs on private property.</li> <li>Coordinate contracting to remove and/or trim trees that</li> </ul>	
	endanger structures, infrastructure, and vital roadways.	

MITIGATION ACTION DETAILS		
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Social, Technical, Administrative, Economic	
Priority:	Medium	
Estimated Cost:	\$50,000 - \$100,000	
Implementation Schedule:	6-12 months	
Coordinating Agency:	Jefferson County, local municipalities	
Potential Funding Sources:	FEMA PA, HMGP, local operating budgets	

2017	Ana	lycice
2017	Alla	12313.

Defer Action – Will include in the 2017 Plan Update. Revise Priority to Low.

	Jefferson County (Past Action) – 28
Proposed Action:	Work with State Fire Marshall and County and local fire departments
	to enforce burn ban ordinances during times of drought to prevent wildfire.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Wildfire
STAPLE-E Summary:	Technical
Priority:	Low
Estimated Cost:	\$1,000 - \$10,000
Implementation Schedule:	6-12 months
Coordinating Agency:	Jefferson County, State Fire Marshall, local fire districts
Potential Funding Sources:	HMGP, local operating budgets

### 2017 Analysis:

Completed.

	Jefferson County (Past Action) – 29
Proposed Action:	Flood proof the Jefferson County courthouse elevators by installing pump system.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Dam Failure, Flooding, Hurricane/Tropical Storm, Thunderstorm,
	Tsunami
STAPLE-E Summary:	Technical, Administrative, Economic
Priority:	Low
Estimated Cost:	\$100,000
Implementation Schedule:	1-3 years
Coordinating Agency:	Jefferson County, local municipalities
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets

### 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update.

		Jefferson County (Past Action) – 30
Pr	oposed Action:	Retrofit the LNVA pumping system, which includes a number of pump
		stations, in order to increase capacity and allow stand-alone service when the Neches River is contaminated or a failure of the primary system.

MITIGATION ACTION DETAILS	
Hazard Addressed: STAPLE-E Summary:	Flood, Hazardous Materials, Hurricanes/Tropical Storm, Terrorism, Tsunami, Water Contamination
	Technical. Administrative
Priority:	High
Estimated Cost:	\$3,680,000
Implementation Schedule:	1-3 years
Coordinating Agency:	LNVA, Jefferson County, TDEM, local municipalities
Potential Funding Sources:	LNVA operating budget, HMGP, PDM

### 2017 Analysis:

tion) – 31
order for
ce of the
ither the
ır.
r e

MITIGATION ACTION DETAILS	
Hazard Addressed:	Flood, Hazardous Materials, Hurricane/Tropical Storm, Terrorism,
	Isunami, Water Contamination
STAPLE-E Summary:	Technical, Administrative
Priority:	High
Estimated Cost:	\$562,000
Implementation Schedule:	1-3 years
Coordinating Agency:	LNVA, Jefferson County, TDEM, local municipalities
Potential Funding Sources:	LNVA operating budget, HMGP, PDM

### 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update.

	Jefferson County (Past Action) – 32
Proposed Action:	Convert an existing 56 acre excavated dirt pit into a localized
	freshwater storage impoundment which would be used to supply water
	to municipal water plants and industrial users in the Nederland, Port
	Neches, and Groves area of Jefferson County in the event of an
	interruption of canal service to the region.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Flood, Hazardous Materials, Hurricanes/Tropical Storm, Terrorism, Tsunami, Water Contamination
STAPLE-E Summary:	Technical, Administrative
Priority:	High
Estimated Cost:	\$5,468,000
Implementation Schedule:	1-3 years
Coordinating Agency:	LNVA, Jefferson County, TDEM, local municipalities
Potential Funding Sources:	LNVA operating budget, HMGP, PDM

#### 2017 Analysis:

	Jefferson County (Past Action) – 33
Proposed Action:	Enhance existing structures and construct additional water control
	features to allow the LNVA to isolate segments of canals in the event of
	contamination or localized bank failures in order to conserve and
	protect unaffected waters and continue deliveries to as many
	customers as possible while a clean-up or repair is addressed rather
	than having a single event affect all customers of the entire system.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Flood, Hazardous Materials, Hurricanes/Tropical Storm, Terrorism,
	Tsunami, Water Contamination
STAPLE-E Summary:	Technical, Administrative
Priority:	High
Estimated Cost:	\$375,000
Implementation Schedule:	1-3 years
Coordinating Agency:	LNVA, Jefferson County, TDEM, local municipalities
Potential Funding Sources:	LNVA operating budget, HMGP, PDM

### 2017 Analysis:

# Beaumont

Proposed Action:To build a structure or structures (including a dome or domes) in (or near coastal) jurisdictions that can withstand 200 mile powinds and act as shelters. This is in conjunction with the Tex Shelter Initiative.	coastal er hour as Safe

MITIGATION ACTION DETAILS	
Hazard Addressed:	Hurricane/Tropical Storm
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal
Priority:	High
Estimated Cost:	\$1 - \$2 Million
Implementation Schedule:	1-5 years
Coordinating Agency:	City of Beaumont, Jefferson County, TDEM
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, TDEM, local funding sources

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update. Revise Estimated Cost to \$2,000,000.

	Beaumont (Past Action) – 2
Proposed Action:	Retrofit existing or construct new structures to act as shelters during hurricanes and tropical storms and other severe weather events.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal	
Priority:	High	
Estimated Cost:	\$1 - \$2 million	
Implementation Schedule:	1-5 years	
Coordinating Agency:	City of Beaumont, Jefferson County	
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM, local funding sources	

#### 2017 Analysis:

**Defer Action** – Will include in the 2017 Plan Update. This action was completed on some City structures, but still needs to be done on existing Fire Stations 3, 5, 6, 7, 8, and 10.

	Beaumont (Past Action) – 3
Proposed Action:	Retrofit existing or construct new structures to act as safe rooms during
	tornados.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Tornados	
STAPLE-E Summary:	Social, Technical, Administrative	
Priority:	High	
Estimated Cost:	\$100,000 - \$500,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	City of Beaumont	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

#### 2017 Analysis:

**Delete Action.** This action should be combined with Beaumont (Past Action) – 2.

					Beaumont (Pa	ast Ac	tion)	-4
Proposed Action:	Storm	harden/retrofit	critical	facilities	throughout	the	City	of
	Beaum	ont. Actions can	include b	ut are not	limited to wir	ndow	shutte	ers,
	roof, flo	ood proofing, rol	-up door	reinforcen	nent (i.e. fire s	tatio	ns, pol	lice
	headqu	arters, Health	Departme	ent, EMS	stations, and	othe	r criti	ical
	infrastr	ucture facilities)						

MITIGATION ACTION DETAILS		
Hazard Addressed:	Dam Failure, Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado, Tsunami	
STAPLE-E Summary:	Technical, Administrative, Legal, Economic	
Priority:	High	
Estimated Cost:	\$7.4 Million	
Implementation Schedule:	1-5 years	
Coordinating Agency:	City of Beaumont, Jefferson County	
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets	

#### 2017 Analysis:

**Defer Action** – Will include in the 2017 Plan Update. A new health department building was built to standards; include EMS stations 1 and 2 in proposed action.

	Beaumont (Past Action) – 5
Proposed Action:	Harden the City of Beaumont Emergency Operations Center at 700 Orleans and Police beadquarters at 255 College Actions include but
	are not limited to installing a generator and storm shutters.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Technical, Administrative, Legal, Economic	
Priority:	High	
Estimated Cost:	\$1 Million	
Implementation Schedule:	1-3 years	
Coordinating Agency:	City of Beaumont, Jefferson County	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

**Defer Action** – Will include in the 2017 Plan Update. The EOC facility was hardened, but a generator still needs to be installed there. The Police headquarters received a generator, but still needs to be hardened.

	Beaumont (Past Action) – 6
Proposed Action:	Wind harden the Radio SHOP at 620 Marina Dr., the South Radio Tower
	at 1550 Pine, Fire Headquarters at 400 Walnut, and EMS Headquarters
	at 2870 Laurel. Actions include but are not limited to roof retrofits,
	installing storm shutters/screens, installing generators, and hardening
	of bay doors (specifically fire stations).

MITIGATION ACTION DETAILS		
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Technical, Administrative, Legal	
Priority:	High	
Estimated Cost:	\$3,750,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	City of Beaumont, Jefferson County	
Potential Funding Sources:	HMGP, PDM, EMPG, local operating budgets	

#### 2017 Analysis:

**Defer Action** – Will include in the 2017 Plan Update. The radio shop on Marina Drive was rebuilt to include wind hardening measures, and EMS headquarter was merged into the new Public Health facility. Other listed locations still need to be completed.

	Beaumont (Past Action) – 7
Proposed Action:	Wind harden Baptist Hospital, at 3080 College, and Christus St.
	Elizabeth Hospital at 2830 Calder. Actions include but are not limited to
	roof retrofits, installing storm shutters/screens, installing generators,
	and hardening of bay doors (specifically maintenance and facility
	areas).

MITIGATION ACTION DETAILS		
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Technical, Administrative, Legal	
Priority:	High	
Estimated Cost:	\$8 Million	
Implementation Schedule:	1-3 years	
Coordinating Agency:	City of Beaumont, Jefferson County	
Potential Funding Sources:	HMGP, PDM, EMPG, local operating budgets	

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Beaumont (Past Action) – 8
Proposed Action:	Retrofit main facility at Baptist Hospital with Built-in Decontamination
	System to enable decontamination of patients from a hazardous material incident.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Hazardous Materials
STAPLE-E Summary:	Technical, Administrative, Economic, Environmental
Priority:	High
Estimated Cost:	\$150,000
Implementation Schedule:	6-12 months
Coordinating Agency:	City of Beaumont, Baptist Hospital
Potential Funding Sources:	Local operating budgets

#### 2017 Analysis:

	Beaumont (Past Action) – 9
Proposed Action:	At Christus St. Elizabeth Hospital install backup generators and elevate key electrical equipment (such as Switchgear and ATS).

MITIGATION ACTION DETAILS	
Hazard Addrossod:	Dam Failure, Flood, Hurricane/Tropical Storm, Thunderstorm,
	Tsunami
STAPLE-E Summary:	Technical, Administrative
Priority:	High
Estimated Cost:	\$3.1 Million
Implementation Schedule:	6-12 months
Coordinating Agency:	City of Beaumont, St. Elizabeth Hospital
Potential Funding Sources:	HMGP, PDM, EMPG, local operating budgets

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Beaumont (Past Act	ion) – 10
Proposed Action:	Upgrade Christus St. Elizabeth Hospitals and Port of Bo	eaumont
	emergency communication systems to ensure co	ontinued
	communication with outside sources and first responders.	

MITIGATION ACTION DETAILS	
Hazard Addressed:	Earthquake, Hurricane/Tropical Storm, Severe Winter Weather, Terrorism, Thunderstorm, Tornado
STAPLE-E Summary:	Technical, Administrative, Legal
Priority:	High
Estimated Cost:	\$62,000
Implementation Schedule:	1-3 years
Coordinating Agency:	City of Beaumont, St. Elizabeth Hospital, Port of Beaumont
Potential Funding Sources:	HMGP, PDM, EMPG, local operating budgets

### 2017 Analysis:

	Beaumont (Past Action) – 11
Proposed Action:	Install generators for the Beaumont Independent School District at
	sites including but not limited to Westbrook, Police Building,
	Administrative Building, and the Thomas Education Support Center
	which are used for sheltering and emergency operation coordination
	centers.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Earthquake, Hurricane/Tropical Storm, Severe Winter Weather, Thunderstorm, Tornado
STAPLE-E Summary:	Technical, Administrative, Legal
Priority:	High
Estimated Cost:	\$650,000
Implementation Schedule:	1-3 years
Coordinating Agency:	City of Beaumont, Beaumont Independent School District
Potential Funding Sources:	HMGP, PDM, local operating budgets

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Beaumont (Past Action) – 12
Proposed Action:	Procure mobile backup generators for the Port of Beaumont.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Earthquake, Hurricane/Tropical Storm, Severe Winter Weather, Thunderstorm, Tornado
STAPLE-E Summary:	Technical, Administrative, Legal
Priority:	High
Estimated Cost:	\$60,000
Implementation Schedule:	1-3 years
Coordinating Agency:	City of Beaumont, Port of Beaumont
Potential Funding Sources:	HMGP, PDM, local operating budgets

### 2017 Analysis:

	Beaumont (Past Action) – 13
Proposed Action:	Install backup generators for the 88 lift stations throughout the City of
	Beaumont.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Earthquake, Hurricane/Tropical Storm, Severe Winter Weather, Thunderstorm, Tornado
STAPLE-E Summary:	Technical, Administrative, Legal
Priority:	High
Estimated Cost:	\$3,400,000
Implementation Schedule:	1-3 years
Coordinating Agency:	City of Beaumont, Jefferson County
Potential Funding Sources:	HMGP, PDM, local operating budgets

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

		Beaumont (Past Action) – 14
F	Proposed Action:	Mitigate damage to utilities in order to maintain function during and after a hazard event. Actions can include but are not limited to:
		<ul> <li>Burying utility lines underground</li> </ul>
		<ul> <li>Provide frangible links/break away connections on utility poles</li> </ul>
		<ul> <li>Harden utility poles by converting from wood to concrete or metal utility poles</li> </ul>
		<ul> <li>Increasing the easement area/clearance of utility lines/ poles</li> </ul>

VITIGATION ACTION DETAILS	
Hazard Addressed:	Hurricane/Tropical Storm, Severe Winter Weather, Thunderstorm, Tornado
STAPLE-E Summary:	Social, Technical, Administrative
Priority:	High
Estimated Cost:	\$100,000 - \$300,000 per line
Implementation Schedule:	1-5 years
Coordinating Agency:	City of Beaumont, Jefferson County, Entergy
Potential Funding Sources:	HMGP, PDM, local operating budgets

#### 2017 Analysis:

**Defer Action** – Will include in the 2017 Plan Update. Revise Estimated Cost to \$5.5 Million. Modifications have been made to the Proposed Action.

	Beaumont (Past Action) – 15
Proposed Action:	Hardening Electric Grid and Communications to prevent damage to
	electric, phone and cable infrastructure for major roadways/
	thoroughfares or access routes to critical infrastructure.
	thoroughfares or access routes to critical infrastructure.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Hurricane/Tropical Storm, Severe Winter Weather, Thunderstorm, Tornado
STAPLE-E Summary:	Social, Technical, Administrative
Priority:	High
Estimated Cost:	\$100,000 - \$300,000 per line
Implementation Schedule:	1-5 years
Coordinating Agency:	City of Beaumont, Jefferson County, Entergy
Potential Funding Sources:	HMGP, PDM, local operating budgets

2017 Analysis:	
Defer Action – Will include in the 2017 Plan Update. Revise Estimated Cost to \$1 Million.	

	Beaumont (Past Action) – 16
Proposed Action:	Pursue drainage improvements in the City of Beaumont. Projects
	include but are not limited to the South Park Relief Project (which
	includes Moore Street project, Avenue A Project, Washington
	Boulevard Phase I and II Projects), Madison Street Project, Tyrell Park
	Project, Caldwood Outfall, Phelan Boulevard Drainage Project, and the
	Cartwright/Corley Project.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Dam Failure, Flood, Hurricane/Tropical Storm, Thunderstorm, Tsunami
STAPLE-E Summary:	Social, Technical, Administrative, Economic
Priority:	High
Estimated Cost:	\$60,302,900
Implementation Schedule:	1-3 years
Coordinating Agency:	City of Beaumont, Jefferson County
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets

**Defer Action** – Will include in the 2017 Plan Update. There were several drainage projects completed that were associated with this action, however, not all projects were completed, thus this action will be included in the Plan Update with modifications to the Proposed Action.

	Beaumont (Past Action) – 17
Proposed Action:	Maintain, pursue, and complete drainage improvements in the City of
	Beaumont. Projects include, but are not limited to join Drainage
	District 6 and Beaumont projects and Beaumont individual projects
	such as the Calder Street Mitigation Project, Steve's Drive project,
	Concord Street project, High School Ditch Project (which include
	Seventh Street, North Street, Broadway Box projects).

MITIGATION ACTION DETAILS	
Hazard Addressed:	Dam Failure, Flood, Hurricane/Tropical Storm, Thunderstorm, Tsunami
STAPLE-F Summary:	Social Technical Administrative Economic
Detector	
Priority:	High
Estimated Cost:	\$62,255,000
Implementation Schedule:	1-3 years
Coordinating Agency:	City of Beaumont, Jefferson County
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets

**Defer Action** – Will include in the 2017 Plan Update. The Concord project is expected to be completed by the completion of this plan. The Calder Street project was completed. Revise Estimated Cost to \$50 Million.

	Beaumont (Past Action) – 18
Proposed Action:	Pursue drainage improvements throughout the City of Beaumont.
	Actions can include but are not limited to installing/ upgrading culverts
	and headwalls as well as enlarging storm water ditches and canals.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Dam Failure, Flood, Hurricane/Tropical Storm, Thunderstorm, Tsunami
STAPLE-E Summary:	Social, Technical, Administrative, Economic
Priority:	High
Estimated Cost:	\$10,000,000
Implementation Schedule:	1-3 years
Coordinating Agency:	City of Beaumont, Jefferson County
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets

#### 2017 Analysis:

	Beaumont (Past Action) – 19
Proposed Action:	Construct water retention ponds to collect storm water run-off and use
	as an alternate water source for agricultural resources throughout the
	City of Beaumont.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Dam Failure, Flooding, Hurricane/Tropical Storm, Thunderstorm,
	Tsunami
STAPLE-E Summary:	Technical, Administrative, Environmental
Priority:	High
Estimated Cost:	\$1,000,000
Implementation Schedule:	1-5 years
Coordinating Agency:	City of Beaumont, Jefferson County
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets

#### 2017 Analysis:

**Defer Action** – Will include in the 2017 Plan Update. Modify Proposed Action to read "storm control" rather than "agricultural resources".

	Beaumont (Past Action) – 20
Proposed Action:	Complete bank stabilization project at Riverfront Park.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Flooding, Geologic Hazard
STAPLE-E Summary:	Technical, Administrative
Priority:	High
Estimated Cost:	\$3,000,000
Implementation Schedule:	1-3 years
Coordinating Agency:	City of Beaumont, Jefferson County, SETRPC
Potential Funding Sources:	HMGP, PDM, local operating budgets

#### 2017 Analysis:

**Completed.** Along COB Property Line.

**Defer Action** – Will include in the 2017 Plan Update. A second phase that extends beyond City Hall area up to the train tracks will be included.

	Beaumont (Past Action) – 21
Proposed Action:	Pursue building a 50 million gallon holding lagoon to store wastewater
	in case of power outages and plant failures, and removal of sludge
	build-up in the two lagoons to increase the storage capacity at the
	Wastewater Treatment Plan in case of power outages and plant failure.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Geologic Hazard, Hazardous Material, Hurricane/Tropical Storm, Severe Winter Weather, Terrorism, Thunderstorm, Tornado
STAPLE-E Summary:	Technical, Administrative
Priority:	High
Estimated Cost:	\$3,500,000
Implementation Schedule:	1-5 years
Coordinating Agency:	City of Beaumont, Jefferson County
Potential Funding Sources:	HMGP, PDM, local operating budgets

2017 Analysis:	
Defer Action – Will include in the 2017 Plan Update.	

	Beaumont (Past Action) – 22
Proposed Action:	Pursue and construct alternate fresh water sources throughout the City
	of Beaumont. This includes but is not limited to increased capacity to
	maintain water pressure in case of system failures at the water
	treatment plant; installation of a 36" water transmission line to provide
	an alternate water transport method; installation of a raw water
	pipeline to replace the existing canal to prevent intention or natural
	pollution of the City's water supply; installation of new chemical feed
	facilities and building to safely store and use chlorine, etc.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Drought
STAPLE-E Summary:	Technical, Administrative, Environmental
Priority:	High
Estimated Cost:	\$500,000
Implementation Schedule:	1-5 years
Coordinating Agency:	City of Beaumont, Jefferson County
Potential Funding Sources:	HMGP, PDM, FMA

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update. Revise Estimated Cost to \$5 Million.

	Beaumont (Past Action) – 23
Proposed Action:	At Hermann Memorial Hospital, install 1) an on-site filtering system for
	water well and new pressurized water supply system, and 2) install a
	Built-in Decontamination System (includes shower, curtain system,
	hazardous water tank, and drain) in the main facility.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Hazardous Materials
STAPLE-E Summary:	Technical, Administrative, Economic
Priority:	High
Estimated Cost:	\$210,000
Implementation Schedule:	12 months
Coordinating Agency:	City of Beaumont, Hermann Memorial Baptist Hospital
Potential Funding Sources:	PDM, EMPG, local operating budgets

### 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update.

	Beaumont (Past Action) – 24
Proposed Action:	Retrofit the LNVA pumping system at Highway 105 to upgrade pumping capacity. This will allow the system to operate at full stand-alone service in times of contamination of water in the Neches River due to a hazard event.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Water Contamination, Terrorism
STAPLE-E Summary:	Social, Technical, Administrative, Legal
Priority:	High
Estimated Cost:	\$3,680,000
Implementation Schedule:	1-3 years
Coordinating Agency:	City of Beaumont, LNVA, Jefferson County
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets

#### 2017 Analysis:

	Beaumont (Past Action) – 25
Proposed Action:	Retrofit the primary LNVA diversion point and pumping system at
	10550 Helbig Rd. To allow the LNVA to fully control, isolate, and section
	off the source of the water that flows into the pumping station. This
	would allow the LNVA to isolate either the Neches River or Pine Island
	Bayou should one suffer any form of contamination.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Water Contamination, Terrorism	
STAPLE-E Summary:	Social, Technical, Administrative, Legal	
Priority:	High	
Estimated Cost:	\$562,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	City of Beaumont, LNVA, Jefferson County	
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets	

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Beaumont (Past Action) – 26
Proposed Action:	Develop a database of contact information for first responders,
	volunteers, and vulnerable populations. This also includes a database
	of assisted living/nursing homes throughout the City of Beaumont.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Dam Failure, Drought, Earthquake, Extreme Heat, Flood, Geologic Hazard, Hazardous Materials, Hurricane/Tropical Storm, Severe Winter Weather, Terrorism, Thunderstorm, Tornado, Tsunami, Water Contamination, Wildfire	
STAPLE-E Summary:	Social, Technical, Administrative, Economic	
Priority:	High	
Estimated Cost:	\$15,000	
Implementation Schedule:	6-12 months	
Coordinating Agency:	City of Beaumont	
Potential Funding Sources:	EMPG, local operating budgets	

### 2017 Analysis:

	Beaumont (Past Action) – 27
Proposed Action:	Facilitate use of all mass notifications systems including but not limited
	to the Southeast Texas Alerting Network (STAN), to notify and educate
	the public of impending hazardous events.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Dam Failure, Drought, Earthquake, Extreme Heat, Flood, Geologic Hazard, Hazardous Material, Hurricane/Tropical Storm, Severe Winter Weather, Terrorism, Thunderstorm, Terrorism, Tornado, Tsunami, Water Contamination, Wildfire
STAPLE-E Summary:	Social, Technical, Administrative, Political
Priority:	High
Estimated Cost:	\$25,000
Implementation Schedule:	6-12 months
Coordinating Agency:	City of Beaumont Emergency Management, Jefferson County Emergency Management, SETRPC
Potential Funding Sources:	EMPG, local operating budgets

**Defer Action –** Will include in the 2017 Plan Update. This is an ongoing mitigation item.

_		Beaumont (Past Action) – 28
	Proposed Action:	Identify special needs populations in the city by coordinating with
		home health agencies, medical equipment companies, local churches,
		and neighborhood associations. Organize strategies for evacuating
		special needs populations during a coastal storm, hurricane, or other
		such hazard.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Flooding, Hurricane/Tropical Storm, Thunderstorm, Tsunami
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal
Priority:	High
Estimated Cost:	\$5,000 - \$10,000
Implementation Schedule:	6-12 months
Coordinating Agency:	City of Beaumont, local health care agencies/facilities, Jefferson
	County
Potential Funding Sources:	Local operating budgets

### 2017 Analysis:

	Beaumont (Past Action) – 29
Proposed Action:	Coordinate public/private partnerships to ensure special needs
	populations are protected from health risks due to extreme weather
	conditions. Actions will be targeted toward citizens with physical
	limitations and may be unable to reach safety in times of severe
	weather. Volunteer groups may be available to assist by visiting special
	needs groups to ensure their safety and comfort during severe
	temperature extremes.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Dam Failure, Drought, Earthquake, Extreme Heat, Flood, Geologic Hazard, Hazardous Materials, Hurricane/Tropical Storm, Severe Winter Weather, Terrorism, Thunderstorm, Tornado, Tsunami, Water Contamination, Wildfire
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal
Priority:	High
Estimated Cost:	\$50,000
Implementation Schedule:	6-12 months
Coordinating Agency:	City of Beaumont Emergency Management, local health and special needs agencies, Jefferson County Emergency Management
Potential Funding Sources:	EMPG, local funding sources

### 2017 Analysis:

	Beaumont (Past Action) – 30
Proposed Action:	Coordinate a natural hazards public awareness campaign among
	agencies and the community. Efforts may include tropical
	storm/hurricane awareness presentations, shelter-in-place
	presentations, evacuation maps, floodplain maps, flood control
	projects, storm tracking maps, safety tips flyers, preparedness articles
	in local newspapers, and other such information as it relates to natural
	hazards. Target audiences will include schools, neighborhood watch
	groups, various civic groups, neighborhood associations, community
	groups, and industry groups. FEMA publication will also be made
	available in city hall libraries, municipal courts, police and fire
	departments, public works departments, public access TV channels,
	city libraries, and on the SETRPC and jurisdictional websites.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Dam Failure, Drought, Earthquake, Extreme Heat, Flood, Geologic Hazard, Hazardous Materials, Hurricane/Tropical Storm, Severe Winter Weather, Terrorism, Thunderstorm, Tornado, Tsunami, Water Contamination, Wildfire
STAPLE-E Summary:	Social, Technical, Administrative, Political
Priority:	High
Estimated Cost:	\$25,000 - \$50,000
Implementation Schedule:	6-12 months
Coordinating Agency:	City of Beaumont Emergency Management, Jefferson County Emergency Management
Potential Funding Sources:	HMGP, PDM, EMPG, local funding sources
	Beaumont (Past Action) – 31
------------------	--
Proposed Action:	Coordinate Emergency Management Plans for coastal storms/
	hurricane events. Specific efforts will include encouraging agencies to
	install and maintain back-up power at identified facilities, construct and
	designate emergency operations centers for disaster/emergency
	operations, and solicit participation in Community Emergency
	Response Training.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Hurricane/Tropical Storm, Tsunami	
STAPLE-E Summary:	Technical, Administrative, Political	
Priority:	High	
Estimated Cost:	\$50,000	
Implementation Schedule:	1-5 years	
Coordinating Agency:	City of Beaumont, Jefferson County, SETRPC	
Potential Funding Sources:	Local operating budgets	

#### 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update. This is an ongoing mitigation item.

	Beaumont (Past Action) – 32
Proposed Action:	Maintain the floodplain mapping, planning, and databases project to identify man and maintain systematic accountability for flood prope
	areas.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Dam Failure, Flooding, Hurricane/Tropical Storm, Thunderstorm, Tsunami	
STAPLE-E Summary:	Technical, Administrative, Legal, Economic	
Priority:	High	
Estimated Cost:	\$50,000	
Implementation Schedule:	6-12 months	
Coordinating Agency:	City of Beaumont Planning, Jefferson County, TDEM, FEMA	
Potential Funding Sources:	Local operating budgets	

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update. This is an ongoing mitigation item.

	Beaumont (Past Action) – 33
Proposed Action:	Conduct flood insurance educational seminars for area realtors to increase their knowledge of the National Flood Insurance Program (NFIP) and the benefits to homeowners of securing flood insurance.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Dam Failure, Flood, Hurricane/Tropical Storm, Thunderstorm, Tsunami	
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal	
Priority:	High	
Estimated Cost:		
Implementation Schedule:	6-12 months	
Coordinating Agency:	City of Beaumont, Jefferson County	
Potential Funding Sources:	HMGP, PDM, EMPG, local operating budgets	

## 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update. This is an ongoing mitigation item.

	Beaumont (Past Action) – 34
Proposed Action:	Coordinate with federal, state, and local partners to provide training opportunities for first responders.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Dam Failure, Drought, Earthquake, Extreme Heat, Flood, Geologic Hazard, Hazardous Material, Hurricane/Tropical Storm, Severe Winter Weather, Terrorism, Thunderstorm, Tornado, Tsunami, Water Contamination, Wildfire	
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal	
Priority:	High	
Estimated Cost:	\$15,000	
Implementation Schedule:	6-12 months	
Coordinating Agency:	City of Beaumont Emergency Management	
Potential Funding Sources:	EMPG, PDM, HMGP, local operating budgets	

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update. This is an ongoing mitigation item.

	Beaumont (Past Action) – 35
Proposed Action:	Relocations of Fire Headquarters, Fire stations 1, 2, 7, and 11, and
	Health Department (950 Washington) to improve neighborhood coverage in accordance with the 2005 Pietsch (ISO) study.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Dam Failure, Earthquake, Flood, Geologic Hazard, Hazardous Material, Hurricane/Tropical Storm, Severe Winter Weather, Terrorism, Thunderstorm, Tornado, Tsunami, Water Contamination, Wildfire	
STAPLE-E Summary:	Social, Technical, Administrative, Legal	
Priority:	Medium	
Estimated Cost:	\$23,500,000	
Implementation Schedule:	1-5 years	
Coordinating Agency:	City of Beaumont Emergency Management, Local Fire Departments, Jefferson County Emergency Management	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

# 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update. Revise Estimated Cost to \$21 Million.

	Beaumont (Past Action) – 36
Proposed Action:	Install on-site well and new pressurized water supply system to support Christus St. Elizabeth Hospital during loss of potable water.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Water Contamination
STAPLE-E Summary:	Social, Technical, Administrative, Legal
Priority:	Medium
Estimated Cost:	\$162,000
Implementation Schedule:	1-3 years
Coordinating Agency:	City of Beaumont, St. Elizabeth Hospital, Jefferson County
Potential Funding Sources:	PDM, EMPG, local operating budgets

#### 2017 Analysis:

	Beaumont (Past Action) – 37
Proposed Action:	Develop a Hazard Mitigation Action specific to the City of Beaumont as
	well as maintain interaction with the Jefferson County and Southeast
	Texas Regional Planning Commission Hazard Mitigation Action Plans.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Dam Failure, Drought, Earthquake, Extreme Heat, Flood, Geologic Hazard, Hazardous Materials, Hurricane/Tropical Storm, Severe Winter Weather, Terrorism, Thunderstorm, Tornado, Tsunami, Water Contamination, Wildfire	
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal	
Priority:	Medium	
Estimated Cost:	\$50,000	
Implementation Schedule:	12 months	
Coordinating Agency:	City of Beaumont Emergency Management, TDEM, FEMA	
Potential Funding Sources:	HMGP, PDM, EMPG, local funding sources	

2017 Analysis:	
Defer Action – Will include in the 2017 Plan Update. This is an ongoing mitigation item.	

	Beaumont (Past Action) – 38
Proposed Action:	Acquire flood prone structures throughout the City of Beaumont. This
	includes the 481 properties on the current Repetitive Loss and Severe
	Repetitive Loss lists.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Dam Failure, Flood, Hurricane/Tropical Storm, Thunderstorm, Tsunami
STAPLE-E Summary:	Social, Technical, Administrative, Economic, Environmental
Priority:	Medium
Estimated Cost:	\$55,800,000
Implementation Schedule:	1-3 years
Coordinating Agency:	City of Beaumont, Jefferson County
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL, local operating budgets

#### 2017 Analysis:

**Defer Action** – Will include in the 2017 Plan Update. Some citizens were not willing to sell their property. Modify Proposed Action to include "elevate" flood prone structures.

	Beaumont (Past Action) – 39
Proposed Action:	Coordinate a consolidated security checkpoint on entry to Plant Road to access industrial and chemical production and storage complexes.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Terrorism
STAPLE-E Summary:	Technical, Administrative, Legal
Priority:	Medium
Estimated Cost:	\$175,000
Implementation Schedule:	6-12 months
Coordinating Agency:	City of Beaumont Emergency Management, Port of Beaumont, Jefferson County Emergency Management
Potential Funding Sources:	EMPG, Homeland Security Grants, local operating budgets

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Beaumont (Past Action) – 40
Proposed Action:	Upgrade security system at the Port of Beaumont for access control on
	all exterior doors for all buildings as well as installing cameras and
	increased perimeter surveillance capabilities.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Terrorism
STAPLE-E Summary:	Technical, Administrative, Legal
Priority:	Medium
Estimated Cost:	\$175,000
Implementation Schedule:	6-12 months
Coordinating Agency:	City of Beaumont Emergency Management, Port of Beaumont, Jefferson County Emergency Management
Potential Funding Sources:	EMPG, Homeland Security Grants, local operating budgets

#### 2017 Analysis:

	Beaumont (Past Action) – 41
Proposed Action:	Install security systems at the City of Beaumont water utility sites, to
	include but not limited to, security walls around chemical tanks, water
	treatment plant and a river pump station; monitoring stations for purity
	testing at various sites throughout the city; surveillance at six elevated
	storage tanks and wastewater plant, etc. for access control on all
	exterior doors for all buildings as well as installing cameras and
	increased perimeter surveillance capabilities.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Terrorism
STAPLE-E Summary:	Technical, Administrative, Legal
Priority:	Medium
Estimated Cost:	\$4,500,000
Implementation Schedule:	6-12 months
Coordinating Agency:	City of Beaumont Emergency Management, Jefferson County Emergency Management
Potential Funding Sources:	EMPG, Homeland Security Grants, local operating budgets

# 2017 Analysis: Defer Action – Will include in the 2017 Plan Update.

	Beaumont (Past Action) – 42
Proposed Action:	Upgrade surveillance capabilities at critical infrastructure sites around
	the City of Beaumont, to include, fire stations, police, fire and EMS
	headquarters, lift stations, communication towers and headquarters,
	etc.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Terrorism
STAPLE-E Summary:	Technical, Administrative, Legal
Priority:	Medium
Estimated Cost:	\$200,000
Implementation Schedule:	6-12 months
Coordinating Agency:	City of Beaumont Emergency Management, Jefferson County Emergency Management
Potential Funding Sources:	EMPG, Homeland Security Grants, local operating budgets

# 2017 Analysis:

	Beaumont (Past Action) – 43
Proposed Action:	Install generators and harden roofs at sites at Lamar University. Sites
	Field House which serves as an emergency operations center.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Hurricanes/Tropical Storm, Thunderstorm, Tornado
STAPLE-E Summary:	Social, Technical, Administrative
Priority:	Medium
Estimated Cost:	\$1,000,000
Implementation Schedule:	1-3 years
Coordinating Agency:	City of Beaumont, Lamar University, Jefferson County
Potential Funding Sources:	HMGP, PDM, local operating budgets

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

				Веа	numont (	Past Action) – 44
Proposed Action:	At	Lamar	University,	replace/upgrade	radio	repeater/tower,
	repl	lace/upgr	ade emerger	ncy notification sire	en towei	r equipment and
	tow	er, cleani	ing and resto	ration of tunnel net	work, an	d upgrade safety
	and	security	lighting throu	ughout campus.		

MITIGATION ACTION DETAILS	
Hazard Addressed:	Dam Failure, Earthquake, Flood, Hazardous Materials, Hurricanes/Tropical Storm, Severe Winter Weather, Terrorism, Thunderstorm, Tornado, Tsunami, Water Contamination, Wildfire
STAPLE-E Summary:	Technical, Administrative
Priority:	Medium
Estimated Cost:	\$2,357,000
Implementation Schedule:	1-3 years
Coordinating Agency:	City of Beaumont, Lamar University, Jefferson County
Potential Funding Sources:	HMGP, PDM, EMPG, local operating budgets

#### 2017 Analysis:

	Beaumont (Past Action) – 45
Proposed Action:	Elevate flood prone properties/structures and key infrastructure and
	electrical equipment throughout the City of Beaumont.

MITIGATION ACTION DETAILS	
Hazard Addressed:         Dam Failure, Flooding, Hurricane/Tropical Storm, Thunder           Tsunami         Tsunami	
STAPLE-E Summary:	Social, Technical, Administrative, Economic
Priority:	Medium
Estimated Cost:	\$500,000 - \$1,000,000
Implementation Schedule:	1-3 years
Coordinating Agency:	City of Beaumont, Jefferson County
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL, local operating budgets

#### 2017 Analysis:

	Beaumont (Past Action) – 46
Proposed Action:	Minimize damage to structures and infrastructures due to falling trees.
	Actions include but are not limited to the following:
	Pursue and coordinate a dangerous tree and limb removal
	program to protect infrastructure and critical facilities from
	damage. This includes working with private homeowners for
	voluntary removal of hazardous trees and limbs on private
	property.
	<ul> <li>Coordinate to remove and/or trim trees that endanger</li> </ul>
	structures, infrastructure, and vital roadways.
	<ul> <li>Removal of dangerous trees and limbs (dead, leaners, and</li> </ul>
	hangers). Prevent blockage or damage to infrastructure
	and/or major roadways/thoroughfares.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado
STAPLE-E Summary:	Social, Technical, Administrative, Political
Priority:	Medium
Estimated Cost:	\$1 Million
Implementation Schedule:	6-12 months
Coordinating Agency:	City of Beaumont (Public Works), Jefferson County
Potential Funding Sources:	PDM, local operating budgets

## 2017 Analysis:

					Be	aumont (	Pas	t Actio	on) – 47
Proposed Action:	Secure	and	maintain	backup	information	systems	to	store	critical
	informa	tion	at off-site	location	S.				

MITIGATION ACTION DETAILS	
Hazard Addressed:	Dam Failure, Drought, Earthquake, Extreme Heat, Flood, Geologic Hazard, Hazardous Materials, Hurricane/Tropical Storm, Severe Winter Weather, Terrorism, Thunderstorm, Tornado, Tsunami, Water Contamination, Wildfire
STAPLE-E Summary:	Technical, Administrative, Political
Priority:	Medium
Estimated Cost:	\$50,000
Implementation Schedule:	6-12 months
Coordinating Agency:	City of Beaumont, Jefferson County, TDEM
Potential Funding Sources:	HMGP, PDM, EMPG, local operating budgets

2017 Analysis:
Defer Action – Will include in the 2017 Plan Update.

	Beaumont (Past Action) – 48
Proposed Action:	Improve quality of local information on vulnerable items (assets and
	populations) for the purpose of more accurate risk and damage
	assessments. Work with other agencies in city to get data as up to date
	and complete as possible.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Dam Failure, Drought, Earthquake, Extreme Heat, Flood, Geologic Hazard, Hazardous Materials, Hurricane/Tropical Storm, Severe Winter Weather, Terrorism, Thunderstorm, Tornado, Tsunami, Water Contamination, Wildfire
STAPLE-E Summary:	Technical, Administrative, Political
Priority:	Medium
Estimated Cost:	\$1,000 - \$5,000
Implementation Schedule:	6-12 months
Coordinating Agency:	City of Beaumont, Jefferson County, TDEM
Potential Funding Sources:	HMGP, PDM, EMPG, local operating budgets

#### 2017 Analysis:

	Beaumont (Past Action) – 49
Proposed Action:	Elevate and/or upgrade Marina Drive in the City of Beaumont.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Dam Failure, Flooding, Geologic Hazard, Hurricane/Tropical Storm,	
nazaru Addressed.	Thunderstorm, Tornado, Tsunami	
STAPLE-E Summary:	Technical, Administrative	
Priority:	Low	
Estimated Cost:	\$6 Million	
Implementation Schedule:	1-3 years	
Coordinating Agency:	City of Beaumont, Jefferson County, TXDOT	
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets	

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

		Beaumont (Past Action) – 50
	Proposed Action:	Provide educational seminars and brochures regarding the voluntary Community Rating System (CRS).

MITIGATION ACTION DETAILS	
Hazard Addressed:	Dam Failure, Flooding, Hurricane/Tropical Storm, Thunderstorm,
	Tsunami
STAPLE-E Summary:	Social, Technical, Administrative, Economic
Priority:	Low
Estimated Cost:	\$15,000
Implementation Schedule:	6-12 months
Coordinating Agency:	City of Beaumont, Jefferson County, TDEM
Potential Funding Sources:	HMGP, PDM, local operating budgets

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update. This is an ongoing mitigation item.

	Beaumont (Past Action) – 51
Proposed Action:	Expand and upgrade security systems at St. Elizabeth Hospital for
	access control on all exterior doors for all buildings as well as installing cameras and increased perimeter surveillance and safety capabilities.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Terrorism
STAPLE-E Summary:	Social, Technical, Administrative
Priority:	Low
Estimated Cost:	\$350,000
Implementation Schedule:	1-3 years
Coordinating Agency:	City of Beaumont, St. Elizabeth Hospital
Potential Funding Sources:	EMPG, Homeland Security Grants, local operating budgets

#### 2017 Analysis:

# Bevil Oaks

	Bevil Oaks (Past Action) – 1
Proposed Action:	To build a structure or structures (including a dome or domes) in coastal (or near coastal) jurisdictions that can withstand 200 mile per hour winds and act as shelters. This is in conjunction with the Texas Safe Shelter Initiative.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Hurricanes/Tropical Storms, Thunderstorms, Tornado
STAPLE-E Summary:	Social, Technical, Administrative, Political
Priority:	Low
Estimated Cost:	\$1.6 Million
Implementation Schedule:	1-3 years
Coordinating Agency:	Bevil Oaks, Jefferson County, TDEM, SETRPC
Potential Funding Sources:	Texas Safe Shelter funding, HMGP, PDM

# 2017 Analysis: Delete Action.

	Bevil Oaks (Past Action) – 2
Proposed Action:	Retrofit existing structures and/or construct new structures to act as shelters during and after Hurricanes/Tropical Storms and other severe weather events.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Hurricane/Tropical Storms, Thunderstorms, Tornado
STAPLE-E Summary:	Social, Technical, Administrative, Political
Priority:	Low
Estimated Cost:	\$1 - \$3 Million
Implementation Schedule:	1-5 years
Coordinating Agency:	Bevil Oaks, Jefferson County, TDEM, SETRPC
Potential Funding Sources:	HMGP, PDM

## 2017 Analysis:

	Bevil Oaks (Past Action) – 3
Proposed Action:	Storm harden/retrofit critical facilities throughout Bevil Oaks. Actions can include but are not limited to window shutters, roof straps, flood proofing, and roll-up door reinforcement (i.e. for fire stations).

MITIGATION ACTION DETAILS			
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado		
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic		
Priority:	High		
Estimated Cost:	\$10,000 - \$500,000		
Implementation Schedule:	1-2 years		
Coordinating Agency:	Bevil Oaks, Jefferson County		
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets		

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update. Revise Priority to Low.

		Bevil Oaks (Past Action) – 4
	Proposed Action:	Acquire flood prone properties (including Repetitive Loss and Severe
		Repetitive Loss Properties).
I		

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flooding, Hurricane/Tropical Storm, Thunderstorm	
STAPLE-E Summary:	Technical, Administrative, Economic, Environmental	
Priority:	High	
Estimated Cost:	\$50,000 - \$140,000	
Implementation Schedule:	12 months	
Coordinating Agency:	Bevil Oaks, Jefferson County	
Potential Funding Sources:	HMGP, FMA, PDM, RFC, SRL	

#### 2017 Analysis:

**Defer Action** – Will include in the 2017 Plan Update. Note: Progress has been made on this action – HMGP funding submitted to purchase 4 Repetitive Loss Properties: 2-100%, 1-90%, 1-75%.

	Bevil Oaks (Past Action) – 5
Proposed Action:	Elevate new and existing flood prone structures and infrastructure
	throughout Bevil Oaks.

MITIGATION ACTION DETAILS			
Hazard Addressed:	Flooding, Hurricane/Tropical Storm, Thunderstorm		
STAPLE-E Summary:	Technical, Administrative, Economic		
Priority:	High		
Estimated Cost:	\$20,000 - \$100,000		
Implementation Schedule:	1-3 years		
Coordinating Agency:	Bevil Oaks, Jefferson County		
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL		

#### 2017 Analysis:

**Defer Action** – Will include in the 2017 Plan Update. Note: Progress Made – City ordinance now requires new and substantially improved structures to use 2 foot free board.

					Bevi	l Oaks	s (Past A	Action) – 6
Proposed Action:	Replace througho	and/or out Bevil	upgrade Oaks.	bridges,	culverts	and	other	crossings

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flooding, Geologic Hazards	
STAPLE-E Summary:	Technical, Administrative, Legal	
Priority:	High	
Estimated Cost:	\$200,000 - \$1 Million	
Implementation Schedule:	2-5 years	
Coordinating Agency:	Bevil Oaks, Jefferson County, TXDOT	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

#### 2017 Analysis:

**Defer Action** – Will include in the 2017 Plan Update. Note: Progress made – Culvert on River Road 100% complete. Major ditch project underway, which should be completed in 2017.

	Bevil Oaks (Past Action) – 7
Proposed Action:	Pursue drainage improvements throughout Bevil Oaks. Actions can
	include but are not limited to installing/upgrading culverts and headwalls as well as enlarging storm water ditches and canals.

MITIGATION ACTION DETAILS			
Hazard Addressed:	Flooding, Hurricanes/Tropical Storms, Thunderstorms		
STAPLE-E Summary:	Technical, Administrative, Political, Legal, Economic		
Priority:	High		
Estimated Cost:	\$50,000 - \$500,000		
Implementation Schedule:	1-3 years		
Coordinating Agency:	Bevil Oaks, Jefferson County		
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets		

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

		Bevil Oaks (Past Action) – 8
	Proposed Action:	Bury underground, secure, or otherwise harden exposed or vulnerable pipelines.

MITIGATION ACTION DETAILS						
Hazard Addressed:	Flood, Hazardous Material, Hurricane/Tropical Storm, Thunderstorm, Tornado					
STAPLE-E Summary:	Technical, Administrative, Economic					
Priority: High						
Estimated Cost:	\$100,000 - \$500,000					
Implementation Schedule:	3-5 years					
Coordinating Agency:	Bevil Oaks, Jefferson County, Entergy					
Potential Funding Sources:	HMGP, PDM, local operating budgets					

#### 2017 Analysis:

				Bevil	Oaks	s (Past A	ction) – 9
Proposed Action:	Provide	generators/back-up	power	systems	for	critical	facilities
	(includin and othe	ng but not limited to lif er first responder facili	t station ities) thr	is, water p oughout I	lants 3evil (	, police, Oaks.	EMS, Fire

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Technical, Administrative, Political, Legal	
Priority:	Medium	
Estimated Cost:	\$50,000 - \$200,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	Bevil Oaks, Jefferson County	
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM	

#### 2017 Analysis:

**Defer Action** – Will include in the 2017 Plan Update. Revise Priority to Low; revise Implementation Schedule to 3-5 years.

	Bevil Oaks (Past Action) – 10
Proposed Action:	Pursue flood protection measures to protect from surges from
	Hurricanes/Tropical Storms. Actions can include but are not limited to
	constructing and/or upgrading sea walls, flood barriers, berms and
	various wet and dry flood proofing measures.

MITIGATION ACTION DETAILS						
Hazard Addressed:	Flood,	Hurricane/Tropical	Storm,	Thunderstorm,	Water	
	Contami	nation				
STAPLE-E Summary:	Technica	al, Administrative, Poli	tical			
Priority:	High					
Estimated Cost:	\$50,000	\$50,000 - \$3 Million				
Implementation Schedule:	1-5 years					
Coordinating Agency:	Bevil Oaks, Jefferson County, USACE, TDEM					
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL					

#### 2017 Analysis:

**Defer Action** – Will include in the 2017 Plan Update. Note: Progress made – permanent generator fir Fire Station, EMS, water/sewer/City Hall and Civic Center. 3 lift stations have portable generators with plans to install permanent ones with city funds.

	Bevil Oaks (Past Action) – 11
Proposed Action:	Pursue the coordination, construction, expansion and maintenance of
	flood control structures/barriers for the purpose of mitigating damages
	and protect fresh water resources from storm surge, sea level rise, and
	other sources of salt water intrusion.
4	

MITIGATION ACTION DETAILS		
Hazard Addressed:	Hurricane/Tropical Storm	
STAPLE-E Summary:	Technical, Administrative, Economic	
Priority:	High	
Estimated Cost:	\$1 - \$2 Million	
Implementation Schedule:	2-5 years	
Coordinating Agency:	Bevil Oaks, Jefferson County, USACE, SETRPC, FEMA	
Potential Funding Sources:	HMGP, PDM< USACE, Jefferson County, SETRPC	

## 2017 Analysis:

**Delete Action.** 

					Bevil Oaks (Pa	ast Actic	on) – 12
Proposed Action:	Construct appropriat	or e to	improve collect st	existing orm wate	detention/retention r to reduce flooding.	ponds	where

MITIGATION ACTION DETAILS			
Hazard Addressed:	Flooding, Hurricanes/Tropical Storm, Thunderstorms		
STAPLE-E Summary:	Social, Technical, Administrative, Economic, Environmental		
Priority:	Low		
Estimated Cost:	\$50,000 - \$100,000		
Implementation Schedule:	1-3 years		
Coordinating Agency:	Bevil Oaks, Jefferson County		
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL, local operating budgets		

#### 2017 Analysis:

**Delete Action.** 

	Bevil Oaks (Past Action) – 13
Proposed Action:	Retrofit existing structures or construct new structures to act as safe
	rooms during tornado or other severe weather events.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Tornados, Thunderstorms	
STAPLE-E Summary:	Social, Technical, Administrative, Economic	
Priority:	High	
Estimated Cost:	\$5,000 - \$100,000	
Implementation Schedule:	1-5 years	
Coordinating Agency:	Bevil Oaks, Jefferson County	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Bevil Oaks (Past Action) – 14
Proposed Action:	Identify and pursue any mitigation activities that would aid/ enhance evacuations throughout Bevil Oaks.

MITIGATION ACTION DETAILS				
Hazard Addressed:	Flood, Hazardous Materials, Hurricane/Tropical Storm, Thunderstorm, Tornado, Wildfire			
<b>STAPLE-E Summary:</b> Social, Technical, Administrative, Political, Legal, Economic				
Priority:	High			
Estimated Cost:	\$10,000 - \$500,000			
Implementation Schedule:	1-5 years			
Coordinating Agency:	Bevil Oaks, Jefferson County			
Potential Funding Sources:	HMGP, PDM, local operating budgets			

#### 2017 Analysis:

	Bevil Oaks (Past Action) – 15
Proposed Action:	Construct water retention ponds to collect storm water run-off and use
	as an alternate water source throughout Bevil Oaks.

MITIGATION ACTION DETAILS				
Hazard Addressed:	Drought, Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado, Water Contamination			
STAPLE-E Summary:	Social, Technical, Administrative, Environmental			
Priority:	Low			
Estimated Cost:	\$75,000 - \$200,000			
Implementation Schedule:	1-3 years			
Coordinating Agency:	Bevil Oaks, Jefferson County			
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets			

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Bevil Oaks (Past Action) – 16
Proposed Action:	Mitigate damage to utilities in order to maintain function during and after a hazard event. Actions can include but are not limited to:
	<ul> <li>Burying utility lines underground</li> </ul>
	<ul> <li>Provide frangible links/break away connections on utility poles</li> </ul>
	<ul> <li>Harden utility poles by converting from wood to concrete or metal utility poles</li> </ul>
	<ul> <li>Increasing the easement area/clearance of utility lines/ poles</li> </ul>
	from tree lines

MITIGATION ACTION DETAILS				
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado			
STAPLE-E Summary:	Social, Technical, Administrative, Legal, Economic			
Priority:	High			
Estimated Cost:	\$100,000 - \$200,000			
Implementation Schedule:	1-5 years			
Coordinating Agency:	Bevil Oaks, Jefferson County, Entergy			
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM			

#### 2017 Analysis:

_							Bevil Oaks	(Pas	t Act	ion) – 17
	Proposed Action:	Provide	the	public	with	educational	brochures	for	the	hazards
		identifie	d as p	part of t	he 201	L1 Plan Updat	e.			

MITIGATION ACTION DETAILS					
Hazard Addressed:	Drought, Earthquake, Extreme Heat, Flood, Geologic Hazard, Hazardous Materials, Hurricane/Tropical Storm, Severe Winter Weather, Terrorism, Thunderstorm, Tornado, Tsunami, Water Contamination, Wildfire				
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic				
Priority:	High				
Estimated Cost:	\$5,000 - \$50,000				
Implementation Schedule:	6-12 months				
Coordinating Agency:	Bevil Oaks, Jefferson County, TDEM, FEMA				
Potential Funding Sources:	HMGP, PDM, local operating budgets, EMPG				

#### 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update.

	Bevil Oaks (Past Action) – 18
Proposed Action:	Retrofit existing structures to act as cooling stations in times of extreme heat.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Extreme Heat	
STAPLE-E Summary:	Social, Technical, Administrative	
Priority:	High	
Estimated Cost:	\$100,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	Bevil Oaks, Jefferson County	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

#### 2017 Analysis:

	Bevil Oaks (Past Action) – 19
Proposed Action:	Develop areas of defensible space to prevent damage due to wildfires.

MITIGATION ACTION DETAILS				
Hazard Addressed:	Wildfire			
STAPLE-E Summary:	Technical, Administrative, Economic			
Priority:	Low			
Estimated Cost:	\$50,000 - \$100,000			
Implementation Schedule:	1-3 years			
Coordinating Agency:	Bevil Oaks, Jefferson County Emergency Management, Texas Forest			
coordinating Agency.	Service			
Potential Funding Sources:	HMGP, EMPG, PDM, local operating budgets			

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Bevil Oaks (Past Action) – 20
Proposed Action:	Develop and enact water conservation or drought management plans, ordinances or strategies to be used during times of drought.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Drought	
STAPLE-E Summary:	Social, Technical, Administrative	
Priority:	High	
Estimated Cost:	\$25,000 - \$50,000	
Implementation Schedule:	6-12 months	
Coordinating Agency:	Bevil Oaks, Jefferson County	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

#### 2017 Analysis:

	Bevil Oaks (Past Action) – 21	
Proposed Action:	Minimize damage to structures and infrastructure from falling trees. Actions include but are not limited to the following:	
	<ul> <li>Pursue and coordinate a dangerous tree and limb removal program to protect infrastructure and critical facilities from damage. This includes working with private homeowners for voluntary removal of hazardous trees and limbs on property.</li> <li>Coordinate contracting to remove and/or trim trees that endanger structures, infrastructure, and vital roadways.</li> </ul>	

MITIGATION ACTION DETAILS		
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Social, Technical, Administrative, Economic	
Priority:	High	
Estimated Cost:	\$50,000 - \$100,000	
Implementation Schedule:	6-12 months	
Coordinating Agency:	Bevil Oaks, Jefferson County	
Potential Funding Sources:	FEMA PA, HMGP, local operating budgets	

2017 Analysis:	
<b>Defer Action</b> – Will include in the 2017 Plan Update. Note: this is ongoing.	

	Bevil Oaks (Past Action) – 22
Proposed Action:	Public structure strengthening for City Hall and Fire Department.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic	
Priority:	High	
Estimated Cost:	\$100,000 - \$200,000	
Implementation Schedule:	1-2 years	
Coordinating Agency:	Bevil Oaks, Jefferson County	
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets	

2017 Analysis:		
Completed.		

	Bevil Oaks (Past Action) – 23
Proposed Action:	Harden and provide generators/alternate power sources for the
	following lift stations:
	<ul> <li>Lift Station #1 at the end of Shipley at the Sewer Plant</li> </ul>
	<ul> <li>Lift Station #2 located at the end of Rolling Hills</li> </ul>
	<ul> <li>Lift Station #3 located on Riverbend Road</li> </ul>
	<ul> <li>Lift Station #4 located on River Oaks Blvd. at the east end</li> </ul>

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Technical, Administrative, Political, Legal	
Priority:	High	
Estimated Cost:	\$50,000 - \$200,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	Bevil Oaks, Jefferson County	
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM	

2017 Analysis:	
Completed.	

	Bevil Oaks (Past Action) – 24
Proposed Action:	Harden and install generators/alternate power sources at Bevil Oaks
	Emergency Operations Center located at 7390 Sweetgum, Beaumont,
	Texas.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Technical, Administrative, Political, Legal	
Priority:	High	
Estimated Cost:	\$50,000 - \$200,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	Bevil Oaks, Jefferson County	
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM	

Completed.

	Bevil Oaks (Past Action) – 25
Proposed Action:	Provide generators/alternate power sources for the Bevil Oaks Fire
	Station.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Technical, Administrative, Political, Legal	
Priority:	High	
Estimated Cost:	\$50,000 - \$200,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	Bevil Oaks, Jefferson County	
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM	

#### 2017 Analysis:

Completed.

# China

Proposed Action:To build a structure or structures (including a done or domes) in coat (or near coastal) jurisdictions that can withstand 200 mile per h winds and act as shelters. This is in conjunction with the Texas Shelter Initiative.	

MITIGATION ACTION DETAILS		
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal	
Priority:	High	
Estimated Cost:	\$1.6 Million	
Implementation Schedule:	1-3 years	
Coordinating Agency:	City of China, Jefferson County Emergency Management, TDEM	
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM	

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	China (Past Action) – 2
Proposed Action:	Retrofit existing structures and/or construct new structures to act as shelters during and after Hurricanes/Tropical Storms and other severe weather events.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal
Priority:	High
Estimated Cost:	\$1 - \$2 Million
Implementation Schedule:	1-3 years
Coordinating Agency:	City of China, Jefferson County, SETRPC, TDEM
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM

#### 2017 Analysis:

	China (Past Action) – 3
Proposed Action:	Develop and enact water conservation or drought management plans,
	ordinances or strategies to be used during times of drought.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Drought
STAPLE-E Summary:	Social, Technical, Administrative
Priority:	High
Estimated Cost:	\$25,000 - \$50,000
Implementation Schedule:	6-12 months
Coordinating Agency:	City of China, Jefferson County
Potential Funding Sources:	HMGP, PDM, local operating budgets

#### 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update.

	China (Past Action) – 4
Proposed Action:	<ul> <li>Minimize damage to structures and infrastructure from falling trees.</li> <li>Actions include but are not limited to the following: <ul> <li>Pursue and coordinate a dangerous tree and limb removal program to protect infrastructure and critical facilities from damage. This includes working with private homeowners for voluntary removal of hazardous trees and limbs on private property.</li> <li>Coordinate contracting to remove and/or trim trees that</li> </ul> </li> </ul>
	endanger structures, infrastructure, and vital roadways.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado
STAPLE-E Summary:	Social, Technical, Administrative, Economic
Priority:	High
Estimated Cost:	\$50,000 - \$100,000
Implementation Schedule:	6-12 months
Coordinating Agency:	City of China, Jefferson County
Potential Funding Sources:	FEMA PA, HMGP, local operating budgets

#### 2017 Analysis:

	China (Past Action) – 5
Proposed Action:	Public structure strengthening for City Hall.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic	
Priority:	High	
Estimated Cost:	\$10,000 - \$500,000	
Implementation Schedule:	1-2 years	
Coordinating Agency:	City of China, Jefferson County	
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets	

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

						China	a (Past A	ction) – 6
Proposed Action:	Provide	generator	rs/back-up	power	systems	for	critical	facilities
	(includin	g but not li	imited to lif	ft station	ıs, water p	วlants	, police,	EMS, Fire
	and othe	er first resp	onder facil	ities) thr	oughout	China	J.	

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado				
STAPLE-E Summary:	Technical, Administrative, Political, Legal				
Priority:	High				
Estimated Cost:	\$50,000 - \$200,000				
Implementation Schedule:	1-3 years				
Coordinating Agency:	City of China, Jefferson County				
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM				

#### 2017 Analysis:

	China (Past Action) – 7
Proposed Action:	Retrofit existing structures or construct new structures to act as safe
	rooms during tornados or other severe weather events.

MITIGATION ACTION DETAILS					
Hazard Addressed:	Tornados, Thunderstorms				
STAPLE-E Summary:	Social, Technical, Administrative, Economic				
Priority:	High				
Estimated Cost:	\$5,000 - \$100,000				
Implementation Schedule:	1-5 years				
Coordinating Agency:	City of China, Jefferson County				
Potential Funding Sources:	HMGP, PDM, local operating budgets				

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	China (Past Action) – 8
Proposed Action:	Identify and pursue any mitigation activities that would aid/ enhance evacuations throughout China.

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flood, Hazardous Materials, Hurricane/Tropical Storm, Thunderstorm, Tornado, Wildfire				
Social, Technical, Administrative, Political, Legal, Economic					
Priority:	High				
Estimated Cost:	\$10,000 - \$500,000				
Implementation Schedule:	1-5 years				
Coordinating Agency:	City of China, Jefferson County				
Potential Funding Sources:	HMGP, PDM, local operating budgets				

#### 2017 Analysis:

	China (Past Action) – 9
Proposed Action:	Construct water retention ponds to collect storm water run-off and use
	as an alternate water source throughout China.

MITIGATION ACTION DETAILS					
Hazard Addressed:	Drought, Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado, Water Contamination				
STAPLE-E Summary:	Social, Technical, Administrative, Environmental				
Priority:	High				
Estimated Cost:	\$75,000 - \$200,000				
Implementation Schedule:	1-3 years				
Coordinating Agency:	City of China, Jefferson County				
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets				

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	China (Past Action) – 10
Proposed Action:	Mitigate damage to utilities in order to maintain function during and after a hazard event. Actions can include but are not limited to:
	<ul> <li>Burying utility lines underground</li> </ul>
	<ul> <li>Provide frangible links/break away connections on utility poles</li> </ul>
	<ul> <li>Harden utility poles by converting from wood to concrete or metal utility poles</li> </ul>
	<ul> <li>Increasing the easement area/clearance of utility lines/ poles</li> </ul>
	from trees lines

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado				
STAPLE-E Summary:	Social, Technical, Administrative, Legal, Economic				
Priority:	High				
Estimated Cost:	\$100,000 - \$200,000				
Implementation Schedule:	1-5 years				
Coordinating Agency:	City of China, Jefferson County, Entergy				
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM				

#### 2017 Analysis:

						China	(Pas	t Act	ion) – 11
Proposed Action:	Provide	the	public	with	educational	brochures	for	the	hazards
	identifie	d as p	oart of t	he 201	L1 Plan Updat	e.			

MITIGATION ACTION DETAILS						
Hazard Addressed:	Drought, Earthquake, Extreme Heat, Flood, Geologic Hazard, Hazardous Materials, Hurricane/Tropical Storm, Severe Winter Weather, Terrorism, Thunderstorm, Tornado, Tsunami, Water Contamination, Wildfire					
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic					
Priority:	High					
Estimated Cost:	\$5,000 - \$50,000					
Implementation Schedule:	6-12 months					
Coordinating Agency:	City of China, Jefferson County, TDEM, FEMA					
Potential Funding Sources:	HMGP, PDM, local operating budgets, EMPG					

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

		China (Past Action) – 12
Proj	posed Action:	Storm harden/retrofit critical facilities throughout China. Actions can include but are not limited to window shutters, roof straps, flood proofing, and roll-up door reinforcement (i.e. for fire stations).

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado				
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic				
Priority:	Medium				
Estimated Cost:	\$10,000 - \$500,000				
Implementation Schedule:	1-2 years				
Coordinating Agency:	City of China, Jefferson County				
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets				

#### 2017 Analysis:

					C	hina	(Past Ad	ction) – 13
Proposed Action:	Replace througho	and/or out China	upgrade	bridges,	culverts	and	other	crossings

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flooding, Geologic Hazards				
STAPLE-E Summary:	Technical, Administrative, Legal				
Priority:	Medium				
Estimated Cost:	\$200,000 - \$1 Million				
Implementation Schedule:	2-5 years				
Coordinating Agency:	City of China, Jefferson County, TXDOT				
Potential Funding Sources:	HMGP, PDM, local operating budgets				

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	China (Past Action) – 14
Proposed Action:	Pursue drainage improvements throughout China. Actions can include
	but are not limited to installing/upgrading culverts and headwalls as
	well as enlarging storm water ditches and canals.

MITIGATION ACTION DETAILS				
Hazard Addressed:	Flooding, Hurricanes/Tropical Storms, Thunderstorms			
STAPLE-E Summary:	Technical, Administrative, Political, Legal, Economic			
Priority:	Medium			
Estimated Cost:	\$50,000 - \$500,000			
Implementation Schedule:	1-3 years			
Coordinating Agency:	City of China, Jefferson County			
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets			

#### 2017 Analysis:

	China (Past Action) – 15
Proposed Action:	Pursue flood protection measures to protect from surges from
	Hurricanes/Tropical Storms. Actions can include but are not limited to
	constructing and/or upgrading sea walls, flood barriers, berms and
	various wet and dry flood proofing measures.

MITIGATION ACTION DETAILS							
Hazard Addressed:	Flood, Hurricane/Tropical storm, Thunderstorm, Water						
	Contamination						
STAPLE-E Summary: Technical, Administrative, Political							
Priority:	Medium						
Estimated Cost:	\$50,000 - \$3 Million						
Implementation Schedule:	1-5 years						
Coordinating Agency:	City of China, Jefferson County, USACE, TDEM						
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL						

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

					China (Pa	ist Actio	on) – 16
Proposed Action:	Construct appropriat	or e to	improve collect st	existing orm wate	detention/retention r to reduce flooding.	ponds	where

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flooding, Hurricanes/Tropical Storm, Thunderstorms				
STAPLE-E Summary:	Social, Technical, Administrative, Economic, Environmental				
Priority:	Medium				
Estimated Cost:	\$50,000 - \$100,000				
Implementation Schedule:	1-3 years				
Coordinating Agency:	City of China, Jefferson County				
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL, local operating budgets				

#### 2017 Analysis:

	China (Past Action) – 17
Proposed Action:	Retrofit existing structures to act as cooling stations in times of extreme
	heat.

MITIGATION ACTION DETAILS				
Hazard Addressed:	Extreme Heat			
STAPLE-E Summary:	Social, Technical, Administrative			
Priority:	Medium			
Estimated Cost:	\$100,000			
Implementation Schedule:	1-3 years			
Coordinating Agency:	City of China, Jefferson County			
Potential Funding Sources:	HMGP, PDM, local operating budgets			

#### 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update.

	China (Past Action) – 18
Proposed Action:	Acquire flood prone properties (including Repetitive Loss and Severe Repetitive Loss Properties).

MITIGATION ACTION DETAILS	
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm
STAPLE-E Summary:	Social, Technical, Administrative, Environmental, Economic
Priority:	Low
Estimated Cost:	\$50,000 - \$140,000
Implementation Schedule:	12 months
Coordinating Agency:	City of China, Jefferson County
Potential Funding Sources:	HMGP, FMA, PDM, RFC, SRL

#### 2017 Analysis:

	China (Past Action) – 19
Proposed Action:	Elevate new and existing flood prone structures and infrastructure
	throughout China.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Flooding, Hurricane/Tropical Storm
STAPLE-E Summary:	Technical, Administrative, Economic
Priority:	Low
Estimated Cost:	\$20,000 - \$100,000
Implementation Schedule:	1-3 years
Coordinating Agency:	City of China, Jefferson County
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL

#### 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update.

	China (Past Action) – 20
Proposed Action:	Bury underground, secure or otherwise harden exposed or vulnerable pipelines.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flood, Hazardous Material, Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Technical, Administrative, Economic	
Priority:	Low	
Estimated Cost:	\$100,000 - \$500,000	
Implementation Schedule:	3-5 years	
Coordinating Agency:	City of China, Jefferson County, Entergy	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

#### 2017 Analysis:

	China (Past Action) – 21
Proposed Action:	Pursue the coordination, construction, expansion and maintenance of
	flood control structures/barriers for the purpose of mitigating damage
	and protect fresh water resources from storm surge, sea level rise, and
	other sources of salt water intrusion.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Hurricane/Tropical Storm
STAPLE-E Summary:	Technical, Administrative, Economic
Priority:	Low
Estimated Cost:	\$1 - \$2 Million
Implementation Schedule:	2-5 years
Coordinating Agency:	City of China, Jefferson County, USACE, SETRPC, FEMA
Potential Funding Sources:	HMGP, PDM, USACE, Jefferson County, SETRPC

#### 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update.

	China (Past Action) – 22
Proposed Action:	Develop areas of defensible space to prevent damage due to wildfires.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Wildfire
STAPLE-E Summary:	Technical, Administrative, Economic
Priority:	Low
Estimated Cost:	\$50,000 - \$100,000
Implementation Schedule:	1-3 years
Coordinating Agency:	City of China, Jefferson County Emergency Management
Potential Funding Sources:	HMGP, EMPG, PDM, local operating budgets

#### 2017 Analysis:
# Groves

	Groves (Past Action) – 1
Proposed Action:	To build a structure or structures (including a dome or domes) in coastal (or near coastal) jurisdictions that can withstand 200 mile per hour winds and act as shelters. This is in conjunction with the Texas Safe Shelter Initiative.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal	
Priority:	High	
Estimated Cost:	\$1.6 Million	
Implementation Schedule:	1-3 years	
Coordinating Agency:	City of Groves, Jefferson County Emergency Management, TDEM	
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM	

017 Analysis:	
Completed.	

	Groves (Past Action) – 2
Proposed Action:	Retrofit existing structures and/or construct new structures to act as shelters during and after Hurricanes/Tropical Storms and other severe weather events.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal	
Priority:	High	
Estimated Cost:	\$1 - \$2 Million	
Implementation Schedule:	1-3 years	
Coordinating Agency:	City of Groves, Jefferson County, SETRPC, TDEM	
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM	

# 2017 Analysis:

	Groves (Past Action) – 3
Proposed Action:	Storm harden/retrofit critical facilities throughout Groves. Actions can include but are not limited to window shutters, roof straps, flood proofing, and roll-up door reinforcement (i.e. for fire stations).

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic	
Priority:	High	
Estimated Cost:	\$10,000 - \$500,000	
Implementation Schedule:	1-2 years	
Coordinating Agency:	City of Groves, Jefferson County	
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets	

### 2017 Analysis:

	Groves (Past Action) – 4
Proposed Action:	Installation of a generator at the South Lift Station.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Technical, Administrative, Political, Legal	
Priority:	High	
Estimated Cost:	\$50,000 - \$200,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	City of Groves, Jefferson County	
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM	

2017 Analysis:	
Completed.	

	Groves (Past Action) – 5
Proposed Action:	Installation of a generator at the North Lift Station.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Technical, Administrative, Political, Legal	
Priority:	High	
Estimated Cost:	\$50,000 - \$200,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	City of Groves, Jefferson County	
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM	

## 2017 Analysis:

Completed.

	Groves (Past Action	n) — 6
Proposed Action:	Provide generators/back-up power systems for critical faci	ilities
	(including but not limited to lift stations, water plants, police, EMS, and other first responder facilities) throughout Groves.	, Fire

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Technical, Administrative, Political, Legal	
Priority:	High	
Estimated Cost:	\$50,000 - \$200,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	City of Groves, Jefferson County	
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM	

2017 Analysis:	
Completed.	

	Groves (Past Action) – 7
Proposed Action:	Identify and pursue any mitigation activities that would aid/ enhance evacuations throughout Groves.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flood, Hazardous Materials, Hurricane/Tropical Storm, Thunderstorm, Tornado, Wildfire	
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic	
Priority: High		
Estimated Cost:	\$10,000 - \$500,000	
Implementation Schedule:	1-5 years	
Coordinating Agency:	City of Groves, Jefferson County	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Groves (Past Action) – 8
Proposed Action:	Public structure strengthening for City Hall, Police Station and Activity Center Complex.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic	
Priority:	High	
Estimated Cost:	\$10,000 - \$500,000	
Implementation Schedule:	1-2 years	
Coordinating Agency:	City of Groves, Jefferson County	
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets	

#### 2017 Analysis:

	Groves (Past Action) – 9
Proposed Action:	Public structure strengthening for Groves Fire Station.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic	
Priority:	High	
Estimated Cost:	\$10,000 - \$500,000	
Implementation Schedule:	1-2 years	
Coordinating Agency:	City of Groves, Jefferson County	
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets	

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Groves (Past Action) – 10
Proposed Action:	Public structure strengthening for Public Works Complex.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic
Priority:	High
Estimated Cost:	\$10,000 - \$500,000
Implementation Schedule:	1-2 years
Coordinating Agency:	City of Groves, Jefferson County
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets

### 2017 Analysis:

	Groves (Past Action) – 11
Proposed Action:	Public structure strengthening for Wastewater Treatment Plan.

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado				
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic				
Priority:	High				
Estimated Cost:	\$10,000 - \$500,000				
Implementation Schedule:	1-2 years				
Coordinating Agency:	City of Groves, Jefferson County				
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets				

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

							Groves	(Pas	t Acti	ion) – 12
1	Proposed Action:	Provide	the	public	with	educational	brochures	for	the	hazards
		identifie	d as p	part of t	he 201	L1 Plan Updat	e.			

MITIGATION ACTION DETAILS	
Hazard Addressed:	Drought, Earthquake, Extreme Heat, Flood, Geologic Hazard, Hazardous Materials, Hurricane/Tropical Storm, Severe Winter Weather, Terrorism, Thunderstorm, Tornado, Tsunami, Water Contamination, Wildfire
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic
Priority:	Medium
Estimated Cost:	\$5,000 - \$50,000
Implementation Schedule:	6-12 months
Coordinating Agency:	City of Groves, Jefferson County, TDEM, FEMA
Potential Funding Sources:	HMGP, PDM, local operating budgets, EMPG

#### 2017 Analysis:

					Gr	oves	(Past Ac	ction) – 13
Proposed Action:	Replace	and/or	upgrade	bridges,	culverts	and	other	crossings
	throught	out Grove	25.					

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flooding, Geologic Hazards				
STAPLE-E Summary:	Technical, Administrative, Legal				
Priority:	Medium				
Estimated Cost:	\$200,000 - \$1 Million				
Implementation Schedule:	2-5 years				
Coordinating Agency:	City of Groves, Jefferson County, TXDOT				
Potential Funding Sources:	HMGP, PDM, local operating budgets				

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Groves (Past Action) – 14
Proposed Action:	Pursue drainage improvements throughout Groves. Actions can include
	but are not limited to installing/upgrading culverts and headwalls as
	well as enlarging storm water ditches, drains and canals.

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flooding, Hurricanes/Tropical Storms, Thunderstorms				
STAPLE-E Summary:	Technical, Administrative, Political, Legal, Economic				
Priority:	Medium				
Estimated Cost:	\$50,000 - \$500,000				
Implementation Schedule:	1-3 years				
Coordinating Agency:	City of Groves, Jefferson County				
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets				

#### 2017 Analysis:

	Groves (Past Action) – 15
Proposed Action:	Pursue flood protection measures to protect from surge from
	Hurricanes/Tropical Storms. Actions can include but are not limited to
	constructing and/or upgrading sea walls, flood barriers, berms and
	various wet and dry flood proofing measures.

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flood,	Hurricane/Tropical	Storm,	Thunderstorm,	Water
	Contamir	nation			
STAPLE-E Summary:	Technical, Administrative, Political				
Priority:	Medium				
Estimated Cost:	\$50,000 -	- \$3 Million			
Implementation Schedule:	1-5 years				
Coordinating Agency:	City of Gr	roves, Jefferson Count	ty, USACE,	TDEM	
Potential Funding Sources:	HMGP, P	DM, FMA, RFC, SRL			

## 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update.

	Groves (Past Action) – 16
Proposed Action:	Pursue the coordination, construction, expansion and maintenance of flood control structures/barriers for the purpose of mitigating damage and protect fresh water resources from storm surge, seal level rise and other sources of salt water intrusion.

MITIGATION ACTION DETAILS					
Hazard Addressed:	Hurricane/Tropical Storm				
STAPLE-E Summary:	Technical, Administrative, Economic				
Priority:	Medium				
Estimated Cost:	\$1 - \$2 Million				
Implementation Schedule:	2-5 years				
Coordinating Agency:	City of Groves, Jefferson County, USACE, SETRPC, FEMA				
Potential Funding Sources:	HMGP, PDM, USACE, Jefferson County, SETRPC				

#### 2017 Analysis:

	Groves (Past Action) – 17
Proposed Action:	Develop and enact water conservation or drought management plans, ordinances, or strategies to be used during times of drought.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Drought
STAPLE-E Summary:	Social, Technical, Administrative
Priority:	Medium
Estimated Cost:	\$25,000 - \$50,000
Implementation Schedule:	6-12 months
Coordinating Agency:	City of Groves, Jefferson County
Potential Funding Sources:	HMGP, PDM, local operating budgets

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Groves (Past Action) – 18
Proposed Action:	Construct water retention ponds to collect storm water run-off and use as an alternate water source throughout Groves.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Drought, Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado, Water Contamination
STAPLE-E Summary:	Social, Technical, Administrative, Environmental
Priority:	Medium
Estimated Cost:	\$75,000 - \$200,000
Implementation Schedule:	1-3 years
Coordinating Agency:	City of Groves, Jefferson County
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets

### 2017 Analysis:

	Groves (Past Action) – 19
Proposed Action:	Mitigate damage to utilities in order to maintain function during and after a hazard event. Actions can include but are not limited to:
	<ul> <li>Burying utility lines underground</li> </ul>
	<ul> <li>Provide frangible links/break away connections on utility poles</li> </ul>
	<ul> <li>Harden utility poles by converting from wood to concrete or metal utility poles</li> </ul>
	<ul> <li>Increasing the easement area/clearance of utility lines/ poles</li> </ul>
	from tree lines

MITIGATION ACTION DETAILS	
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado
STAPLE-E Summary:	Social, Technical, Administrative, Legal, Economic
Priority:	Low
Estimated Cost:	\$100,000 - \$200,000
Implementation Schedule:	1-5 years
Coordinating Agency:	City of Groves, Jefferson County, Entergy
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM

2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Groves (Past Action) – 20
Proposed Action:	Retrofit existing structures to act as cooling stations in times of extreme heat.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Extreme Heat
STAPLE-E Summary:	Social, Technical, Administrative
Priority:	Low
Estimated Cost:	\$100,000
Implementation Schedule:	1-3 years
Coordinating Agency:	City of Groves, Jefferson County
Potential Funding Sources:	HMGP, PDM, local operating budgets

### 2017 Analysis:

	Groves (Past Action) – 21
Proposed Action:	Develop areas of defensible space to prevent damage due to wildfires.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Wildfire
STAPLE-E Summary:	Technical, Administrative, Economic
Priority:	Low
Estimated Cost:	\$50,000 - \$100,000
Implementation Schedule:	1-3 years
Coordinating Agency:	City of Groves, Jefferson County Emergency Management
Potential Funding Sources:	HMGP, EMPG, PDM, local operating budgets

#### 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update.

	Groves (Past Action) – 22
Proposed Action:	<ul> <li>Minimize damage to structures and infrastructure from falling trees.</li> <li>Actions include but are not limited to the following:         <ul> <li>Pursue and coordinate dangerous tree and limb removal program to protect infrastructure and critical facilities from damage. This includes working with private homeowners for voluntary removal of hazardous trees and limbs on private property.</li> </ul> </li> </ul>
	<ul> <li>Coordinate contracting to remove and/or trim trees that endanger structures, infrastructure and vital roadways.</li> </ul>

MITIGATION ACTION DETAILS	
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado
STAPLE-E Summary:	Social, Technical, Administrative, Economic
Priority:	Low
Estimated Cost:	\$50,000 - \$100,000
Implementation Schedule:	6-12 months
Coordinating Agency:	City of Groves, Jefferson County
Potential Funding Sources:	FEMA PA, HMGP, local operating budgets

# 2017 Analysis:

	Groves (Past Action) – 23
Proposed Action:	Reduce flooding on Van Buren from Wilson to Grant.

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flooding, Hurricanes/Tropical Storms, Thunderstorms				
STAPLE-E Summary:	Technical, Administrative, Political, Legal, Economic				
Priority:	Low				
Estimated Cost:	\$50,000 - \$500,000				
Implementation Schedule:	1-3 years				
Coordinating Agency:	City of Groves, Jefferson County				
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets				

#### 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update.

Proposed Action: Address flooding issues on 34 <sup>th</sup> Street and the south end of Franklin Street.		Groves (Past Action) – 24
	Proposed Action:	Address flooding issues on 34 <sup>th</sup> Street and the south end of Franklin Street.

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flooding, Hurricanes/Tropical Storms, Thunderstorms				
STAPLE-E Summary:	Technical, Administrative, Political, Legal, Economic				
Priority:	Low				
Estimated Cost:	\$50,000 - \$500,000				
Implementation Schedule:	1-3 years				
Coordinating Agency:	City of Groves, Jefferson County				
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets				

#### 2017 Analysis:

	Groves (Past Action) – 25
Proposed Action:	Acquire flood prone properties (including Repetitive Loss and Severe
	Repetitive Loss Properties).

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm				
STAPLE-E Summary:	Social, Technical, Administrative, Environmental, Economic				
Priority:	Low				
Estimated Cost:	\$50,000 - \$140,000				
Implementation Schedule:	12 months				
Coordinating Agency:	City of Groves, Jefferson County				
Potential Funding Sources:	HMGP, FMA, PDM, RFC, SRL				

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Groves (Past Action) – 26
Proposed Action:	Elevate new and existing flood prone structures and infrastructure throughout Groves.

MITIGATION ACTION DETAILS				
Hazard Addressed:	Flooding, Hurricane/Tropical Storm			
STAPLE-E Summary:	Technical, Administrative, Economic			
Priority:	Low			
Estimated Cost:	\$20,000 - \$100,000			
Implementation Schedule:	1-3 years			
Coordinating Agency:	City of Groves, Jefferson County			
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL			

#### 2017 Analysis:

	Groves (Past Action) – 27
Proposed Action:	Bury underground, secure or otherwise harden exposed or vulnerable
	pipeline.

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flood,	Hazardous	Material,	Hurricane/Tropical	Storm,
	Thunderst	torm, Tornado	כ		
STAPLE-E Summary:	Technical, Administrative, Economic				
Priority:	Low				
Estimated Cost:	\$100,000 - \$500,000				
Implementation Schedule:	3-5 years				
Coordinating Agency:	City of Gro	oves, Jefferso	n County, Ent	tergy	
Potential Funding Sources:	HMGP, PD	DM, local oper	rating budget	S	

#### 2017 Analysis:

**Defer Action** – Will include in the 2017 Plan Update. Revise action to include water, sewer, liquid petroleum, and natural gas.

					Groves (Pa	ist Actio	n) – 28
Proposed Action:	Construct appropriat	or :e to	improve collect st	existing orm wate	detention/retention r to reduce flooding.	ponds	where

MITIGATION ACTION DETAILS				
Hazard Addressed:	Flooding, Hurricanes/Tropical Storm, Thunderstorms			
STAPLE-E Summary:	Social, Technical, Administrative, Economic, Environmental			
Priority:	Low			
Estimated Cost:	\$50,000 - \$100,000			
Implementation Schedule:	1-3 years			
Coordinating Agency:	City of Groves, Jefferson County			
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL, local operating budgets			

## 2017 Analysis:

**Delete Action.** 

	Groves (Past Action) – 29
Proposed Action:	Retrofit existing structures or construct new structures to act as safe
	rooms during tornados or other severe weather events.

MITIGATION ACTION DETAILS					
Hazard Addressed:	Tornados, Thunderstorms				
STAPLE-E Summary:	Social, Technical, Administrative, Economic				
Priority:	Low				
Estimated Cost:	\$10,000 - \$500,000				
Implementation Schedule:	1-5 years				
Coordinating Agency:	City of Groves, Jefferson County				
Potential Funding Sources:	HMGP, PDM, local operating budgets				

# 2017 Analysis:

**Delete Action.** 

# Nederland

	Nederland (Past Action) – 1
Proposed Action:	To build a structure or structures (including a done or domes) in coastal
	(or near coastal) jurisdictions that can withstand 200 mile per hour
	winds and act as shelters. This is in conjunction with the Texas Safe
	Shelter Initiative.

MITIGATION ACTION DETAILS							
Iazard Addressed:         Hurricane/Tropical Storm, Thunderstorm, Tornado							
STAPLE-E Summary:	TAPLE-E Summary: Social, Technical, Administrative, Political, Legal						
Priority: High							
Estimated Cost: \$1.6 Million							
Implementation Schedule: 1-3 years							
Coordinating Agency:	City of Nederland, Jefferson County Emergency Management, TDEM						
Potential Funding Sources: Texas Safe Shelter Initiative, HMGP, PDM							

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Nederland (Past Action) – 2
Proposed Action:	Retrofit existing structures and/or construct new structures to act as shelters during and after Hurricanes/Tropical Storms and other severe weather events
	weather events.

MITIGATION ACTION DETAILS				
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado			
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal			
Priority:	rity: High			
Estimated Cost:	\$1 - \$2 Million			
Implementation Schedule:	1-3 years			
Coordinating Agency: City of Nederland, Jefferson County, SETRPC, TDEM				
Potential Funding Sources:         Texas Safe Shelter Initiative, HMGP, PDM				

## 2017 Analysis:

	Nederland (Past Action) – 3
Proposed Action:	Storm harden/retrofit critical facilities throughout Nederland. Actions can include but are not limited to window shutters, roof straps, flood proofing, and roll-up door reinforcement (i.e. for fire stations).

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado				
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic				
Priority:	High				
Estimated Cost:	\$10,000 - \$500,000				
Implementation Schedule:	1-2 years				
Coordinating Agency: City of Nederland, Jefferson County					
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets				

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Nederland (Past Action) – 4
Proposed Action:	Acquire flood prone properties (including Repetitive Loss and Severe Repetitive Loss Properties).

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm				
STAPLE-E Summary:	Social, Technical, Administrative, Environmental, Economic				
Priority:	High				
Estimated Cost:	\$50,000 - \$140,000				
Implementation Schedule:	12 months				
Coordinating Agency:	City of Nederland, Jefferson County				
Potential Funding Sources:	HMGP, FMA, PDM, RFC, SRL				

#### 2017 Analysis:

**Defer Action** – Will include in the 2017 Plan Update. Revise action to include relocation of flood-prone properties.

	Nederland (Past Action) – 5
Proposed Action:	Elevate new and existing flood prone structures and infrastructure
	throughout Nederland.

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flooding, Hurricane/Tropical Storm				
STAPLE-E Summary:	Technical, Administrative, Economic				
Priority:	High				
Estimated Cost:	\$20,000 - \$100,000				
Implementation Schedule:	1-3 years				
Coordinating Agency:	City of Nederland, Jefferson County				
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL				

### 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update.

	Nederland (Past Action) – e							Action) – 6	
	Proposed Action:	Replace througho	and/or out Nede	upgrade rland.	bridges,	culverts,	and	other	crossings

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flooding, Geologic Hazards				
STAPLE-E Summary:	Technical, Administrative, Legal				
Priority:	High				
Estimated Cost:	\$200,000 - \$1 Million				
Implementation Schedule:	2-5 years				
Coordinating Agency:	City of Nederland, Jefferson County, TXDOT				
Potential Funding Sources:	HMGP, PDM, local operating budgets				

# 2017 Analysis:

	Nederland (Past Action) – 7
Proposed Action:	Pursue drainage improvements throughout Nederland. Actions can
	include but are not limited to installing/upgrading culverts and headwalls as well as enlarging storm water ditches and canals.

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flooding, Hurricanes/Tropical Storms, Thunderstorms				
STAPLE-E Summary:	Technical, Administrative, Political, Legal, Economic				
Priority:	High				
Estimated Cost:	\$50,000 - \$500,000				
Implementation Schedule:	1-3 years				
Coordinating Agency:	City of Nederland, Jefferson County				
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets				

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

-		
		Nederland (Past Action) – 8
	Proposed Action:	Bury underground, secure or otherwise harden exposed or vulnerable pipelines.

MITIGATION ACTION DETAILS							
Hazard Addressed:	Flood, Hazardous Material, Hurricane/Tropical Storm, Thunderstorm, Tornado						
STAPLE-E Summary:	Technical, Administrative, Economic						
Priority:	High						
Estimated Cost:	\$100,000 - \$500,000						
Implementation Schedule:	3-5 years						
Coordinating Agency:	City of Nederland, Jefferson County, Entergy						
Potential Funding Sources:	HMGP, PDM, local operating budgets						

#### 2017 Analysis:

**Defer Action** – Will include in the 2017 Plan Update. Revise action to include water, sewer, liquid petroleum, and natural gas.

	Nederland (Past Action) – 9							
Proposed Action:	Provide	generators/back-up	power	systems	for	critical	facilities	
	(includin and othe	ng but not limited to lif er first responder facil	ft station ities) thr	s, water p oughout I	lants Nede	s, police, rland.	EMS, Fire	

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado				
STAPLE-E Summary:	Technical, Administrative, Political, Legal				
Priority:	High				
Estimated Cost:	\$50,000 - \$200,000				
Implementation Schedule:	1-3 years				
Coordinating Agency:	City of Nederland, Jefferson County				
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM				

#### 2017 Analysis:

**Defer Action** – Will include in the 2017 Plan Update. This will be combined with the action to storm harden/ retrofit critical facilities.

	Nederland (Past Action) – 10					
Proposed Action:	Pursue flood protection measure to protect from surge from					
	Hurricanes/Tropical Storms. Actions can include but are not limited to					
	constructing and/or upgrading sea walls, flood barriers, berms, and					
	various wet and dry flood proofing measures.					

MITIGATION ACTION DETAILS						
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Water Contamination					
STAPLE-E Summary:	Summary: Technical, Administrative, Political					
Priority:	High					
Estimated Cost:	\$50,000 - \$3 Million					
Implementation Schedule:	1-5 years					
Coordinating Agency:	City of Nederland, Jefferson County, USACE, TDEM					
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL					

#### 2017 Analysis:

	Nederland (Past Action) – 11
Proposed Action:	Pursue the coordination, construction, expansion and maintenance of
	flood control structures/barriers for the purpose of mitigating damage
	and protect fresh water resources from storm surge, sea level rise and
	other sources of salt water intrusion.

MITIGATION ACTION DETAILS				
Hazard Addressed:	Hurricane/Tropical Storm			
STAPLE-E Summary:	Technical, Administrative, Economic			
Priority:	High			
Estimated Cost:	\$1 - \$2 Million			
Implementation Schedule:	2-5 years			
Coordinating Agency:	City of Nederland, Jefferson County, USACE, SETRPC, FEMA			
Potential Funding Sources:	HMGP, PDM, USACE, Jefferson County, SETRPC			

### 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update.

	Nederland (Past Action) – 12						
Proposed Action:	Construct appropriat	or e to	improve collect st	existing orm wate	detention/retention r to reduce flooding.	ponds	where

MITIGATION ACTION DETAILS				
Hazard Addressed: Flooding, Hurricanes/Tropical Storm, Thunderstorms				
STAPLE-E Summary:	Social, Technical, Administrative, Economic, Environmental			
Priority:	High			
Estimated Cost:	\$50,000 - \$100,000			
Implementation Schedule:	1-3 years			
Coordinating Agency:	City of Nederland, Jefferson County			
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL< local operating budgets			

#### 2017 Analysis:

	Nederland (Past Action) – 13
Proposed Action:	Retrofit existing structures or construct new structures to act as safe
	rooms during tornados or other severe weather events.

MITIGATION ACTION DETAILS				
Hazard Addressed: Tornados, Thunderstorms				
STAPLE-E Summary:	Social, Technical, Administrative, Economic			
Priority:	High			
Estimated Cost:	\$5,000 - \$100,000			
Implementation Schedule:	1-5 years			
Coordinating Agency:	City of Nederland, Jefferson County			
Potential Funding Sources:	HMGP, PDM, local operating budgets			

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Nederland (Past Action) – 14
Proposed Action:	Identify and pursue any mitigation activities that would aid/ enhance evacuations throughout Nederland.

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flood, Hazardous Materials, Hurricane/Tropical Storm, Thunderstorm, Tornado, Wildfire				
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic				
Priority:	High				
Estimated Cost:	\$10,000 - \$500,000				
Implementation Schedule:	1-5 years				
Coordinating Agency:	City of Nederland, Jefferson County				
Potential Funding Sources:	HMGP, PDM, local operating budgets				

#### 2017 Analysis:

	Nederland (Past Action) – 15
Proposed Action:	Construct water retention ponds to collect storm water run-off and use
	as an alternate water source throughout Nederland.

MITIGATION ACTION DETAILS				
Hazard Addressed:	Drought, Flood, Hurricane/Tropical Storm, Thunderstorm, Tornad Water Contamination			
STAPLE-E Summary: Social, Technical, Administrative, Environmental				
Priority:	High			
Estimated Cost: \$75,000 - \$200,000				
Implementation Schedule:	1-3 years			
Coordinating Agency:	City of Nederland, Jefferson County			
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets			

2017	Anal	lvsis:
201/	/	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,

**Defer Action –** Will include in the 2017 Plan Update.

	Nederland (Past Action) – 16
Proposed Action:	<ul> <li>Mitigate damage to utilities in order to maintain function during and after a hazard event. Actions can include but are not limited to: <ul> <li>Burying utility lines underground</li> <li>Provide frangible links/breakaway connections on utility poles</li> <li>Harden utility poles by converting from wood to concrete or metal utility poles</li> <li>Increasing the easement area/clearance of utility lines/ poles from trees lines</li> </ul></li></ul>

MITIGATION ACTION DETAILS					
Hazard Addressed: Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado					
STAPLE-E Summary:	Social, Technical, Administrative, Legal, Economic				
Priority:	High				
Estimated Cost:	\$100,000 - \$200,000				
Implementation Schedule:	1-5 years				
Coordinating Agency:	City of Nederland, Jefferson County, Entergy				
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM				

### 2017 Analysis:

					Ν	leder	land	Past Ac	tion)	- 17
Proposed Action:	Public	structure	strengthening	for	City	Hall	and	Police	and	Fire
	Compl	ex.								

MITIGATION ACTION DETAILS				
Hazard Addressed: Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado				
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic			
Priority:	High			
Estimated Cost:	\$10,000 - \$500,000			
Implementation Schedule:	1-2 years			
Coordinating Agency:	City of Nederland, Jefferson County			
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets			

## 2017 Analysis:

Completed.

	Nederland (Past Action) – 18
Proposed Action:	Public structure strengthening for Water Treatment Plant.

MITIGATION ACTION DETAILS			
Hazard Addressed:	Flood, Hurricane/ Tropical Storm, Thunderstorm, Tornado		
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic		
Priority:	High		
Estimated Cost:	\$10,000 - \$500,000		
Implementation Schedule:	1-2 years		
Coordinating Agency:	City of Nederland, Jefferson County		
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets		

## 2017 Analysis:

	Nederland (Past Action) – 19
Proposed Action:	Public structure strengthening for Service Center.

MITIGATION ACTION DETAILS				
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado			
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic			
Priority:	High			
Estimated Cost:	\$10,000 - \$500,000			
Implementation Schedule:	1-2 years			
Coordinating Agency:	City of Nederland, Jefferson County			
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets			

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Nederland (Past Action) – 20
Proposed Action:	Public structure strengthening for Wastewater Treatment Plant located at 515 Hardy Avenue.

MITIGATION ACTION DETAILS				
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado			
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic			
Priority:	High			
Estimated Cost:	\$10,000 - \$500,000			
Implementation Schedule:	1-2 years			
Coordinating Agency:	City of Nederland, Jefferson County			
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets			

#### 2017 Analysis:

	Nederland (Past Action) – 21
Proposed Action:	Public structure strengthening for Hughes Library.

MITIGATION ACTION DETAILS				
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado			
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic			
Priority:	High			
Estimated Cost:	\$10,000 - \$500,000			
Implementation Schedule:	1-2 years			
Coordinating Agency:	City of Nederland, Jefferson County			
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets			

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

						Nederland	(Pas	t Acti	ion) – 22
Proposed Action:	Provide	the	public	with	educational	brochures	for	the	hazards
	identifie	d as p	part of t	he 201	L1 Plan Updat	e.			

MITIGATION ACTION DETAILS					
Hazard Addressed:	Drought, Earthquake, Extreme Heat, Flood, Geologic Hazard, Hazardous Materials, Hurricane/Tropical Storm, Severe Winter Weather, Terrorism, Thunderstorm, Tornado, Tsunami, Water Contamination, Wildfire				
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic				
Priority:	Medium				
Estimated Cost:	\$5,000 - \$50,000				
Implementation Schedule:	6-12 months				
Coordinating Agency:	City of Nederland, Jefferson County, TDEM, FEMA				
Potential Funding Sources:	HMGP, PDM, local operating budgets, EMPG				

#### 2017 Analysis:

	Nederland (Past Action) – 23
Proposed Action:	Retrofit existing structures to act as cooling stations in times of extreme
	heat.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Extreme Heat	
STAPLE-E Summary:	Social, Technical, Administrative	
Priority:	Medium	
Estimated Cost:	\$100,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	City of Nederland, Jefferson County	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Nederland (Past Action) – 24
Proposed Action:	Develop areas of defensible space to prevent damage due to wildfires.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Wildfire	
STAPLE-E Summary:	Technical, Administrative, Economic	
Priority:	Medium	
Estimated Cost:	\$50,000 - \$100,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	City of Nederland, Jefferson County Emergency Management	
Potential Funding Sources:	HMGP, EMPG, PDM, local operating budgets	

#### 2017 Analysis:

	Nederland (Past Action) – 25
Proposed Action:	Develop and enact water conservation or drought management plans, ordinances or strategies to be used during times of drought.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Drought	
STAPLE-E Summary:	Social, Technical, Administrative	
Priority:	Medium	
Estimated Cost:	\$25,000 - \$50,000	
Implementation Schedule:	6-12 months	
Coordinating Agency:	City of Nederland, Jefferson County	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Nederland (Past Action) – 26
Proposed Action:	<ul> <li>Minimize damage to structures and infrastructure from falling trees.</li> <li>Actions include but are not limited to the following:         <ul> <li>Pursue and coordinate a dangerous tree and limb removal program to protect infrastructure and critical facilities from danger. This includes working with private homeowners for voluntary removal of hazardous trees and limbs on private property.</li> </ul> </li> </ul>
	<ul> <li>Coordinate contracting to remove and/or trim trees that endanger structures, infrastructure, and vital roadways.</li> </ul>

MITIGATION ACTION DETAILS		
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Social, Technical, Administrative, Economic	
Priority:	Medium	
Estimated Cost:	\$50,000 - \$100,000	
Implementation Schedule:	6-12 months	
Coordinating Agency:	City of Nederland, Jefferson County	
Potential Funding Sources:	FEMA PA, HMGP, local operating budgets	

# 2017 Analysis:

	Nederland (Past Action) – 27					
Proposed Action:	Public structure feeders.	strengthening	by replacing	drainage	tile main	and

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic	
Priority:	Medium	
Estimated Cost:	\$10,000 - \$500,000	
Implementation Schedule:	1-2 years	
Coordinating Agency:	City of Nederland, Jefferson County	
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets	

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Nederland (Past Action) – 28
Proposed Action:	Public structure strengthening by installing transfer switches at named lift stations.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic	
Priority:	Medium	
Estimated Cost:	\$10,000 - \$500,000	
Implementation Schedule:	1-2 years	
Coordinating Agency:	City of Nederland, Jefferson County	
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets	

2017 Analysis:	
----------------	--

Completed.

	Nederland (Past Action) – 29
Proposed Action:	Public structure strengthening for D. Bob Henson Building.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic	
Priority:	Medium	
Estimated Cost:	\$10,000 - \$500,000	
Implementation Schedule:	1-2 years	
Coordinating Agency:	City of Nederland, Jefferson County	
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets	

### 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update.

	Nederland (Past Action) – 30
Proposed Action:	Improve underground storm sewer culvert size on Detroit Avenue.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Water Contamination
STAPLE-F Summary:	Technical Administrative Political
Drianitan	
Priority:	LOW
Estimated Cost:	\$50,000 - \$3 Million
Implementation Schedule:	1-5 years
Coordinating Agency:	City of Nederland, Jefferson County, USACE, TDEM
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL

### 2017 Analysis:

# Nome

	Nome (Past Action) – 1
Proposed Action:	To build a structure or structures (including a dome or domes) in coastal (or near coastal) jurisdictions that can withstand 200 mile per hour winds and act as shelters. This is in conjunction with the Texas Safe Shelter Initiative.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal
Priority:	High
Estimated Cost:	\$1.6 Million
Implementation Schedule:	1-3 years
Coordinating Agency:	City of Nome, Jefferson County Emergency Management, TDEM
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Nome (Past Action) – 2
Proposed Action:	Retrofit existing structures and/or construct new structures to act as shelters during and after Hurricanes/Tropical Storms and other severe weather events.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal
Priority:	High
Estimated Cost:	\$1- \$2 Million
Implementation Schedule:	1-3 years
Coordinating Agency:	City of Nome, Jefferson County, SETRPC, TDEM
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM

## 2017 Analysis:

	Nome (Past Action) – 3
Proposed Action:	Storm harden/retrofit critical facilities throughout Nome. Actions can include but are not limited to window shutters, roof straps, flood proofing, and roll-up door reinforcement (i.e. for fire stations).

MITIGATION ACTION DETAILS	
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic
Priority:	High
Estimated Cost:	\$10,000 - \$500,000
Implementation Schedule:	1-2 years
Coordinating Agency:	City of Nome, Jefferson County
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Nome (Past Action) – 4
Proposed Action:	Acquire flood prone properties (including Repetitive Loss and Severe Repetitive Loss Properties).

MITIGATION ACTION DETAILS	
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm
STAPLE-E Summary:	Social, Technical, Administrative, Environmental, Economic
Priority:	High
Estimated Cost:	\$50,000 - \$140,000
Implementation Schedule:	12 months
Coordinating Agency:	City of Nome, Jefferson County
Potential Funding Sources:	HMGP, FMA, PDM, RFC, SRL

### 2017 Analysis:

	Nome (Past Action) – 5
Proposed Action:	Elevate new and existing flood prone structures and infrastructure
	throughout Nome.

MITIGATION ACTION DETAILS				
Hazard Addressed:	Flooding, Hurricane/Tropical Storm			
STAPLE-E Summary:	Technical, Administrative, Economic			
Priority:	High			
Estimated Cost:	\$20,000 - \$100,000			
Implementation Schedule:	1-3 years			
Coordinating Agency:	City of Nome, Jefferson County			
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL			

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

						Nome	e (Past A	Action) – 6
Proposed Action:	Replace througho	and/or out Nome	upgrade e.	bridges,	culverts	and	other	crossings

MITIGATION ACTION DETAILS				
Hazard Addressed:	Flooding, Geologic Hazards			
STAPLE-E Summary:	Technical, Administrative, Legal			
Priority:	High			
Estimated Cost:	\$200,000 - \$1 Million			
Implementation Schedule:	2-5 years			
Coordinating Agency:	City of Nome, Jefferson County, TXDOT			
Potential Funding Sources:	HMGP, PDM, local operating budgets			

### 2017 Analysis:

	Nome (Past Action) – 7
Proposed Action:	Pursue drainage improvements throughout Nome. Actions can include but are not limited to installing/upgrading culverts and headwalls as well as enlarging storm water ditches and canals.

MITIGATION ACTION DETAILS				
Hazard Addressed:	Flooding, Hurricanes/Tropical Storms, Thunderstorms			
STAPLE-E Summary:	Technical, Administrative, Political, Legal, Economic			
Priority:	High			
Estimated Cost:	\$50,000 - \$500,000			
Implementation Schedule:	1-3 years			
Coordinating Agency:	City of Nome, Jefferson County			
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets			

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Nome (Past Action) – 8	
Proposed Action:         Bury underground, secure or otherwise harden exposed or vulnera pipelines.	ound, secure or otherwise harden exposed or vulnerable	Proposed Action:

MITIGATION ACTION DETAILS			
Hazard Addressed:	Flood, Hazardous Material, Hurricane/Tropical Storm, Thunderstorm, Tornado		
STAPLE-E Summary:	Technical, Administrative, Economic		
Priority:	High		
Estimated Cost:	\$100,000 - \$500,000		
Implementation Schedule:	3-5 years		
Coordinating Agency:	City of Nome, Jefferson County, Entergy		
Potential Funding Sources:	HMGP, PDM, local operating budgets		

#### 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update. Revise action to include water and sewer.

	Nome (Past Action) ·	- 9
Proposed Action:	Provide generators/back-up power systems for critical facilit	ies
	(including but not limited to lift stations, water plants, police, EMS, F and other first responder facilities) throughout Nome.	ire

MITIGATION ACTION DETAILS				
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado			
STAPLE-E Summary:	Technical, Administrative, Political, Legal			
Priority:	High			
Estimated Cost:	\$50,000 - \$200,000			
Implementation Schedule:	1-3 years			
Coordinating Agency:	City of Nome, Jefferson County			
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM			

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update. Revise Estimated Cost to \$1-\$2 Million.

	Nome (Past Action) – 10
Proposed Action:	Pursue flood protection measures to protect from surge from Hurricanes/Tropical Storms. Actions can include but are not limited to constructing and/or upgrading sea walls, flood barriers, berms and various wet and dry flood proofing measures.

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flood,	Hurricane/Tropical	Storm,	Thunderstorm,	Water
Hazaru Audresseu.	Contami	nation			
STAPLE-E Summary:	Technical, Administrative, Political				
Priority:	High				
Estimated Cost:	\$50,000	- \$3 Million			
Implementation Schedule:	1-5 year	S			
Coordinating Agency:	City of N	lome, Jefferson Count	y, USACE, <sup>-</sup>	TDEM	
Potential Funding Sources:	HMGP, F	PDM, FMA, RFC, SRL			

### 2017 Analysis:

	Nome (Past Action) – 11
Proposed Action:	Pursue the coordination, construction, expansion and maintenance of
	flood control structures/barriers for the purpose of mitigating damage
	and protect fresh water resources from storm surge, seal level rise and
	other sources of salt water intrusion.

MITIGATION ACTION DETAILS					
Hazard Addressed:	Hurricane/Tropical Storm				
STAPLE-E Summary:	Technical, Administrative, Economic				
Priority:	High				
Estimated Cost:	\$1- \$2 Million				
Implementation Schedule:	2-5 years				
Coordinating Agency:	City of Nome, Jefferson County, USACE, SETRPC, FEMA				
Potential Funding Sources:	HMGP, PDM, USACE, Jefferson County, SETRPC				

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

					Nome (Pa	ast Actic	on) – 12
Proposed Action:	Construct appropriat	or e to	improve collect st	existing orm wate	detention/retention r to reduce flooding.	ponds	where

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flooding, Hurricanes/Tropical Storm, Thunderstorms				
STAPLE-E Summary:	Social, Technical, Administrative, Economic, Environmental				
Priority:	High				
Estimated Cost:	\$50,000 - \$100,000				
Implementation Schedule:	1-3 years				
Coordinating Agency:	City of Nome, Jefferson County				
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL, local operating budgets				

#### 2017 Analysis:
	Nome (Past Action) – 13
Proposed Action:	Retrofit existing structures or construct new structures to act as safe
	rooms during tornados or other severe weather events.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Tornados, Thunderstorms	
STAPLE-E Summary:	Social, Technical, Administrative, Economic	
Priority:	High	
Estimated Cost:	\$5,000 - \$100,000	
Implementation Schedule:	1-5 years	
Coordinating Agency:	City of Nome, Jefferson County	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

_		
		Nome (Past Action) – 14
	Proposed Action:	Identify and pursue any mitigation activities that would aid/ enhance evacuations throughout Nome.

MITIGATION ACTION DETAILS				
Hazard Addressed:	Flood, Hazardous Materials, Hurricane/Tropical Storm, Thunderstorm Tornado Wildfire			
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic			
Priority:	High			
Estimated Cost:	\$10,000 - \$500,000			
Implementation Schedule:	1-5 years			
Coordinating Agency:	City of Nome, Jefferson County			
Potential Funding Sources:	HMGP, PDM, local operating budgets			

#### 2017 Analysis:

	Nome (Past Action) – 15
Proposed Action:	Construct water retention ponds to collect storm water run-off and use
	as an alternate water source throughout Nome.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Social, Technical, Administrative, Legal, Economic	
Priority:	High	
Estimated Cost:	\$100,000 - \$200,000	
Implementation Schedule:	1-5 years	
Coordinating Agency:	City of Nome, Jefferson County, Entergy	
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM	

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Nome (Past Action) – 16			
Proposed Action:	Mitigate damage to utilities in order to maintain function during and after a hazard event. Actions can include but are not limited to:			
	<ul> <li>Burying utility lines underground</li> </ul>			
	<ul> <li>Provide frangible links/break away connections on utility poles</li> </ul>			
	<ul> <li>Harden utility poles by converting from wood to concrete or metal utility poles</li> </ul>			
	<ul> <li>Increasing the easement area/clearance of utility lines/ poles</li> </ul>			
	from trees lines			

MITIGATION ACTION DETAILS			
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado		
STAPLE-E Summary:	Social, Technical, Administrative, Legal, Economic		
Priority:	High		
Estimated Cost:	\$100,000 - \$200,000		
Implementation Schedule:	1-5 years		
Coordinating Agency:	City of Nome, Jefferson County, Entergy		
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM		

## 2017 Analysis:

						Nome	(Pas	t Acti	ion) – 17
Proposed Action:	Provide	the	public	with	educational	brochures	for	the	hazards
	identifie	d as p	part of t	he 201	L1 Plan Updat	e.			

MITIGATION ACTION DETAILS				
Hazard Addressed:	Drought, Earthquake, Extreme Heat, Flood, Geologic Hazard, Hazardous Materials, Hurricane/Tropical Storm, Severe Winter Weather, Terrorism, Thunderstorm, Tornado, Tsunami, Water Contamination, Wildfire			
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic			
Priority:	Medium			
Estimated Cost:	\$5,000 - \$50,000			
Implementation Schedule:	6-12 months			
Coordinating Agency:	City of Nome, Jefferson County, TDEM, FEMA			
Potential Funding Sources:	HMGP, PDM, local operating budgets, EMPG			

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Nome (Past Action) – 18
Proposed Action:	Retrofit existing structures to act as cooling stations in times of extreme heat.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Extreme Heat	
STAPLE-E Summary:	Social, Technical, Administrative	
Priority:	Medium	
Estimated Cost:	\$100,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	City of Nome, Jefferson County	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

#### 2017 Analysis:

	Nome (Past Action) – 19
Proposed Action:	Develop areas of defensible space to prevent damage due to wildfires.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Wildfire	
STAPLE-E Summary:	Technical, Administrative, Economic	
Priority:	Medium	
Estimated Cost:	\$50,000 - \$100,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	City of Nome, Jefferson County Emergency Management	
Potential Funding Sources:	HMGP, EMPG, PDM, local operating budgets	

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update. Revise Priority to Low.

	Nome (Past Action) – 20
Proposed Action:	Develop and enact water conservation or drought management plans, ordinances or strategies to be used during times of drought.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Drought	
STAPLE-E Summary:	Social, Technical, Administrative	
Priority:	Medium	
Estimated Cost:	\$25,000 - \$50,000	
Implementation Schedule:	6-12 months	
Coordinating Agency:	City of Nome, Jefferson County	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

### 2017 Analysis:

	Nome (Past Action) – 21
Proposed Action:	Minimize damage to structures and infrastructure from falling trees. Actions include but are not limited to the following:
	<ul> <li>Pursue and coordinate a dangerous tree and limb removal program to protect infrastructure and critical facilities from damage. This includes working with private homeowners for voluntary removal of hazardous trees and limbs on private property.</li> <li>Coordinate contracting to remove and/or trim trees that endanger structures, infrastructure, and vital roadways.</li> </ul>

MITIGATION ACTION DETAILS	
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado
STAPLE-E Summary:	Social, Technical, Administrative, Economic
Priority:	Medium
Estimated Cost:	\$50,000 - \$100,000
Implementation Schedule:	6-12 months
Coordinating Agency:	City of Nome, Jefferson County
Potential Funding Sources:	FEMA PA, HMGP, local operating budgets

#### 2017 Analysis:

# Port Arthur

	Port Arthur (Past Action) – 1
Proposed Action:	To build a structure or structures (including a dome or domes) in coastal (or near coastal) jurisdictions that can withstand 200 mile per hour winds and act as shelters. This is in conjunction with the Texas Safe Shelter Initiative.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal
Priority:	High
Estimated Cost:	\$1.6 Million
Implementation Schedule:	1-3 years
Coordinating Agency:	Jefferson County Emergency Management, TDEM
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Port Arthur (Past Action) – 2
Proposed Action:	Retrofit existing structures and/or construct new structures to act as shelters during and after Hurricanes/Tropical Storms and other severe weather events.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal
Priority:	High
Estimated Cost:	\$1 - \$2 Million
Implementation Schedule:	1-3 years
Coordinating Agency:	Jefferson County, SETRPC, TDEM
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM

#### 2017 Analysis:

	Port Arthur (Past Action) – 3
Proposed Action:	Storm harden/retrofit critical facilities throughout Port Arthur. Actions can include but are not limited to window shutters, roof straps, flood proofing, and roll-up door reinforcement (i.e. for fire stations).

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic	
Priority:	High	
Estimated Cost:	\$10,000 - \$500,000	
Implementation Schedule:	1-2 years	
Coordinating Agency:	City of Port Arthur, Jefferson County	
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets	

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Port Arthur (Past Action) – 4
Proposed Action:	Acquire flood prone properties (including Repetitive Loss and Severe Repetitive Loss Properties).

MITIGATION ACTION DETAILS				
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm			
STAPLE-E Summary:	Social, Technical, Administrative, Environmental, Economic			
Priority:	High			
Estimated Cost:	\$50,000 - \$140,000			
Implementation Schedule:	12 months			
Coordinating Agency:	City of Port Arthur, Jefferson County			
Potential Funding Sources:	HMGP, FMA, PDM, RFC, SRL			

### 2017 Analysis:

	Port Arthur (Past Action) – 5
Proposed Action:	Elevate new and existing flood prone structures and infrastructure throughout Port Arthur.

MITIGATION ACTION DETAILS				
Hazard Addressed:	Flooding, Hurricane/Tropical Storm			
STAPLE-E Summary:	Technical, Administrative, Economic			
Priority:	High			
Estimated Cost:	\$20,000 - \$100,000			
Implementation Schedule:	1-3 years			
Coordinating Agency:	City of Port Arthur, Jefferson County			
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL			

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

					Port A	Arthu	r (Past A	Action) – 6
Proposed Action:	Replace througho	and/or out Port A	upgrade Arthur.	bridges,	culverts	and	other	crossings

MITIGATION ACTION DETAILS				
Hazard Addressed:	Flooding, Geologic Hazards			
STAPLE-E Summary:	Technical, Administrative, Legal			
Priority:	High			
Estimated Cost:	\$200,000 - \$1 Million			
Implementation Schedule:	2-5 years			
Coordinating Agency:	City of Port Arthur, Jefferson County, TXDOT			
Potential Funding Sources:	HMGP, PDM, local operating budgets			

### 2017 Analysis:

	Port Arthur (Past Action) – 7
Proposed Action:	Pursue drainage improvements throughout Port Arthur. Actions can
	include but are not limited to installing/upgrading culverts and headwalls as well as enlarging storm water ditches and canals.

MITIGATION ACTION DETAILS				
Hazard Addressed:	Flooding, Hurricanes/Tropical Storms, Thunderstorms			
STAPLE-E Summary:	Technical, Administrative, Political, Legal, Economic			
Priority:	High			
Estimated Cost:	\$50,000 - \$500,000			
Implementation Schedule:	1-3 years			
Coordinating Agency:	City of Port Arthur, Jefferson County			
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets			

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Port Arthur (Past Action) – 8
Proposed Action:	Bury underground, secure or otherwise harden exposed or vulnerable pipelines.

MITIGATION ACTION DETAILS						
Hazard Addressed:	Flood,	Hazardous	Material,	Hurricane/Tropical	Storm,	
nazaru Audresseu.	Thunderst	torm, Tornado	)			
STAPLE-E Summary:	Technical, Administrative, Economic					
Priority:	High					
Estimated Cost:	\$100,000 - \$500,000					
Implementation Schedule:	3-5 years					
Coordinating Agency:	City of Po	rt Arthur, Jeff	erson County	, Entergy		
Potential Funding Sources: HMGP, PDM, loca			ating budget	S		

#### 2017 Analysis:

				Port A	rthu	r (Past A	ction) – 9
Proposed Action:	Provide	generators/back-up	power	systems	for	critical	facilities
	(includin	ng but not limited to l	ift statior	ns, water p	lants	, police,	EMS, Fire
	and othe	er first responder fac	ilities) thr	oughout t	the Je	efferson	County.

MITIGATION ACTION DETAILS				
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado			
STAPLE-E Summary:	Technical, Administrative, Political, Legal			
Priority:	High			
Estimated Cost:	\$50,000 - \$200,000			
Implementation Schedule:	1-3 years			
Coordinating Agency:	City of Port Arthur, Jefferson County			
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM			

# 2017 Analysis:

Completed.

					Po	ort Arthur	r (Past	Action	) – 10
Proposed Action:	Pursue	flood	protection	measures	to	protect	from	sure	from
	Hurricar	nes/Tro	pical Storms	. Actions ca	an in	clude but	t are no	ot limi	ted to
	constru	cting a	nd/or upgra	ding sea w	alls,	flood ba	arriers,	berm	s and
	various	wet an	d dry flood p	proofing me	asur	es.			

MITIGATION ACTION DETAILS						
Hazard Addressed:	Flood,	Hurricane/Tropical	Storm,	Thunderstorm,	Water	
	Contami	ination				
STAPLE-E Summary:	Technica	al, Administrative, Poli	tical			
Priority:	High					
Estimated Cost:	\$50,000 - \$3 Million					
Implementation Schedule:	1-5 years					
Coordinating Agency:	City of Port Arthur, Jefferson County, USACE, TDEM					
Potential Funding Sources: HMGP, PDM, FMA, RFC, SRL						

# 2017 Analysis:

	Port Arthur (Past Action) – 11
Proposed Action:	Pursue the coordination, construction, expansion and maintenance of
	flood control structures/barriers for the purpose of mitigating damage
	and protect fresh water resources from storm surge, seal level rise, and
	other sources of salt water intrusion.

MITIGATION ACTION DETAILS			
Hazard Addressed:	Hurricane/Tropical Storm		
STAPLE-E Summary:	Technical, Administrative, Economic		
Priority:	High		
Estimated Cost:	\$1 - \$2 Million		
Implementation Schedule:	2-5 years		
Coordinating Agency:	City of Port Arthur, Jefferson County, USACE, SETRPC, FEMA		
Potential Funding Sources:	HMGP, PDM, USACE, Jefferson County, SETRPC		

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

					Port Arthur (Pa	ist Actic	on) — 12
Proposed Action:	Construct appropriat	or e to	improve collect st	existing orm wate	detention/retention r to reduce flooding.	ponds	where

MITIGATION ACTION DETAILS			
Hazard Addressed:	Flooding, Hurricanes/Tropical Storm, Thunderstorms		
STAPLE-E Summary:	Social, Technical, Administrative, Economic, Environmental		
Priority:	High		
Estimated Cost:	\$50,000 - \$100,000		
Implementation Schedule:	1-3 years		
Coordinating Agency:	City of Port Arthur, Jefferson County		
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL, local operating budgets		

# 2017 Analysis:

	Port Arthur (Past Action) – 13
Proposed Action:	Retrofit existing structures or construct new structures to act as safe rooms during tornados or other severe weather events.

MITIGATION ACTION DETAILS			
Hazard Addressed:	Tornados, Thunderstorms		
STAPLE-E Summary:	Social, Technical, Administrative, Economic		
Priority:	High		
Estimated Cost:	\$5,000 - \$100,000		
Implementation Schedule:	1-5 years		
Coordinating Agency:	City of Port Arthur, Jefferson County		
Potential Funding Sources:	HMGP, PDM, local operating budgets		

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Port Arthur (Past Action) – 14
Proposed Action:	Identify and pursue any mitigation activities that would aid/ enhance evacuations throughout Port Arthur.

MITIGATION ACTION DETAILS						
Hazard Addressed:	Flood, Hazardous Materials, Hurricane/Tropical Storm, Thunderstorm, Tornado, Wildfire					
STAPLE-E Summary: Social, Technical, Administrative, Political, Legal, Economic						
Priority:	High					
Estimated Cost:	\$10,000 - \$500,000					
Implementation Schedule:	1-5 years					
Coordinating Agency:	City of Port Arthur, Jefferson County					
Potential Funding Sources:	HMGP, PDM, local operating budgets					

### 2017 Analysis:

	Port Arthur (Past Action) – 15
Proposed Action:	Construct water retention ponds to collect storm water run-off and use
	as an alternate water source throughout Port Arthur where possible.

MITIGATION ACTION DETAILS				
Hazard Addressed:	Drought, Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado, Water Contamination			
STAPLE-E Summary:	Social, Technical, Administrative, Environmental			
Priority:	High			
Estimated Cost:	\$75,000 - \$200,000			
Implementation Schedule:	1-3 years			
Coordinating Agency:	City of Port Arthur, Jefferson County			
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets			

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Port Arthur (Past Action) – 16		
Proposed Action:	Mitigate damage to utilities in order to maintain function during and after a hazard event. Actions can include but are not limited to:		
	<ul> <li>Burying utility lines underground</li> </ul>		
	<ul> <li>Provide frangible links/break away connections on utility poles</li> </ul>		
	<ul> <li>Harden utility poles by converting from wood to concrete or metal utility poles</li> </ul>		
	<ul> <li>Increasing the easement area/clearance of utility lines/ poles from trees lines</li> </ul>		

MITIGATION ACTION DETAILS			
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado		
STAPLE-E Summary:	Social, Technical, Administrative, Legal, Economic		
Priority:	High		
Estimated Cost:	\$100,000 - \$200,000		
Implementation Schedule:	1-5 years		
Coordinating Agency:	City of Port Arthur, Jefferson County, Entergy		
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM		

# 2017 Analysis:

						Port Arthur	(Pas	t Acti	ion) – 17
Proposed Action:	Provide	the	public	with	educational	brochures	for	the	hazards
	identified as part of the 2011 Plan Update.								

MITIGATION ACTION DETAILS				
Hazard Addressed:	Drought, Earthquake, Extreme Heat, Flood, Geologic Hazard, Hazardous Materials, Hurricane/Tropical Storm, Severe Winter Weather, Terrorism, Thunderstorm, Tornado, Tsunami, Water Contamination, Wildfire			
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic			
Priority:	High			
Estimated Cost:	\$5,000 - \$50,000			
Implementation Schedule:	6-12 months			
Coordinating Agency:	City of Port Arthur, Jefferson County, TDEM, FEMA			
Potential Funding Sources:	HMGP, PDM, local operating budgets, EMPG			

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Port Arthur (Past Action) – 18
Proposed Action:	Retrofit existing structures to act as cooling stations in times of extreme heat.

MITIGATION ACTION DETAILS			
Hazard Addressed:	Extreme Heat		
STAPLE-E Summary:	Social, Technical, Administrative		
Priority:	High		
Estimated Cost:	\$100,000		
Implementation Schedule:	1-3 years		
Coordinating Agency:	City of Port Arthur, Jefferson County		
Potential Funding Sources:	HMGP, PDM, local operating budgets		

# 2017 Analysis:

	Port Arthur (Past Action) – 19
Proposed Action:	Develop areas of defensible space to prevent damage due to wildfires.

MITIGATION ACTION DETAILS				
Hazard Addressed:	Wildfire			
STAPLE-E Summary:	Technical, Administrative, Economic			
Priority:	High			
Estimated Cost:	\$50,000 - \$100,000			
Implementation Schedule:	1-3 years			
Coordinating Agency:	City of Port Arthur, Jefferson County Emergency Management			
Potential Funding Sources:	HMGP, EMPG, PDM, local operating budgets			

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Port Arthur (Past Action) – 20
Proposed Action:	Develop and enact water conservation or drought management plans, ordinance or strategies to be used during times of drought.

MITIGATION ACTION DETAILS			
Hazard Addressed:	Drought		
STAPLE-E Summary:	Social, Technical, Administrative		
Priority:	High		
Estimated Cost:	\$25,000 - \$50,000		
Implementation Schedule:	6-12 months		
Coordinating Agency:	City of Port Arthur, Jefferson County		
Potential Funding Sources:	HMGP, PDM, local operating budgets		

#### 2017 Analysis:

		Port Arthur (Past Action) – 21			
Proposed A	ction:	Minimize damage to structures and infrastructures from falling trees. Actions include but are not limited to the following:			
		<ul> <li>Pursue and coordinate a dangerous tree and limb removal program to protect infrastructure and critical facilities from damage. This includes working with private homeowners for voluntary removal of hazardous trees and limbs on private property.</li> <li>Coordinate contracting to remove and/or trim trees that endanger structures infrastructure, and vial roadways.</li> </ul>			

MITIGATION ACTION DETAILS		
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Social, Technical, Administrative, Economic	
Priority:	High	
Estimated Cost:	\$50,000 - \$100,000	
Implementation Schedule:	6-12 months	
Coordinating Agency:	City of Port Arthur, Jefferson County	
Potential Funding Sources:	FEMA PA, HMGP, local operating budgets	

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Port Arthur (Past Action) – 22
Proposed Action:	Increase channel capacity and improve multiple culvert crossings of
	Drainage Channel Main B.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Flooding, Hurricanes/Tropical Storms, Thunderstorms
STAPLE-E Summary:	Technical, Administrative, Political, Legal, Economic
Priority:	High
Estimated Cost:	\$50,000 - \$500,000
Implementation Schedule:	1-3 years
Coordinating Agency:	City of Port Arthur, Jefferson County
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets

### 2017 Analysis:

	Port Arthur (Past Action) – 23
Proposed Action:	In Drainage Channel Main C, improve culvert crossing at Hwy. 69.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Flooding, Hurricanes/Tropical Storms, Thunderstorms
STAPLE-E Summary:	Technical, Administrative, Political, Legal, Economic
Priority:	High
Estimated Cost:	\$50,000 - \$500,000
Implementation Schedule:	1-3 years
Coordinating Agency:	City of Port Arthur, Jefferson County
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets

2017 Analysis:	
Completed.	

	Port Arthur (Past Action) – 24
Proposed Action:	Improve channel capacity and crossings in Lateral 3 of the Drainage Channel Main A system.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Flooding, Hurricanes/Tropical Storms, Thunderstorms
STAPLE-E Summary:	Technical, Administrative, Political, Legal, Economic
Priority:	High
Estimated Cost:	\$50,000 - \$500,000
Implementation Schedule:	1-3 years
Coordinating Agency:	City of Port Arthur, Jefferson County
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets

#### 2017 Analysis:

_		Port Arthur (Past Action) – 25
	Proposed Action:	Improve culvert crossings in the Lakeview Drainage system.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flooding, Hurricanes/Tropical Storms, Thunderstorms	
STAPLE-E Summary:	Technical, Administrative, Political, Legal, Economic	
Priority:	High	
Estimated Cost:	\$50,000 - \$500,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	City of Port Arthur, Jefferson County	
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets	

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Port Arthur (Past Action) – 26
Proposed Action:	Increase drainage capacity to reduce flooding on Westside.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flooding, Hurricanes/Tropical Storms, Thunderstorms	
STAPLE-E Summary:	Technical, Administrative, Political, Legal, Economic	
Priority:	High	
Estimated Cost:	\$50,000 - \$500,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	City of Port Arthur, Jefferson County	
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets	

### 2017 Analysis:

	Port Arthur (Past Action) – 27
Proposed Action:	Improve concrete lining of the El Vista Pump Station.

MITIGATION ACTION DETAILS				
Hazard Addressed:	Flooding, Hurricanes/Tropical Storms, Thunderstorms			
STAPLE-E Summary:	Technical, Administrative, Political, Legal, Economic			
Priority:	High			
Estimated Cost:	\$50,000 - \$500,000			
Implementation Schedule:	1-3 years			
Coordinating Agency:	City of Port Arthur, Jefferson County			
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets			

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Port Arthur (Past Action) – 28
Proposed Action:	Upgrade existing storm sewer in the Port Acres area and along Procter Street.

MITIGATION ACTION DETAILS			
Hazard Addressed:	Flooding, Hurricanes/Tropical Storms, Thunderstorms		
STAPLE-E Summary:	Technical, Administrative, Political, Legal, Economic		
Priority:	High		
Estimated Cost:	\$50,000 - \$500,000		
Implementation Schedule:	1-3 years		
Coordinating Agency:	City of Port Arthur, Jefferson County		
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets		

#### 2017 Analysis:

	Port Arthur (Past Action) – 29
Proposed Action:	Develop/implement shelter-in-place presentations.

MITIGATION ACTION DETAILS			
Hazard Addressed:	Hurricanes/Tropical Storms		
STAPLE-E Summary:	Social, Technical, Administrative, Political		
Priority:	High		
Estimated Cost:	\$1,000 - \$5,000		
Implementation Schedule:	6-12 months		
Coordinating Agency:	City of Port Arthur, Jefferson County		
Potential Funding Sources:	HMGP, PDM, local operating budgets		

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

				Port Arth	ur (Past	Actio	on) – 30
Proposed Action:	Develop/implement	emergency	first	responder	teams	with	Sabine
	Neches Chief's Assoc	iation.					

MITIGATION ACTION DETAILS				
Hazard Addressed:	Dam Failure, Drought, Earthquake, Extreme Heat, Flood, Geologic Hazard, Hazardous Materials, Hurricane/Tropical Storm, Severe Winter Weather, Terrorism, Thunderstorm, Tornado, Tsunami, Water Contamination, Wildfire			
STAPLE-E Summary:	Technical, Administrative, Political			
Priority:	High			
Estimated Cost:	\$1,000 - \$10,000			
Implementation Schedule:	1-5 years			
Coordinating Agency:	Port Arthur Emergency Management, Jefferson County			
Potential Funding Sources:	Local operating budgets			

#### 2017 Analysis:

	Port Arthur (Past Action) – 31
Proposed Action:	Develop/implement coastal storm presentations to groups, schools,
	etc.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Hurricanes/Tropical Storms	
STAPLE-E Summary:	Social, Technical, Administrative, Political	
Priority:	High	
Estimated Cost:	\$1,000 - \$5,000	
Implementation Schedule:	6-12 months	
Coordinating Agency:	City of Port Arthur, Jefferson County	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Port Arthur (Past Action) – 32
Proposed Action:	Require all new construction to meet/exceed minimum established flood elevations.

MITIGATION ACTION DETAILS				
Hazard Addrossod:	Dam Failure, Flooding, Hurricanes/Tropical Storms, Thunderstorms,			
Hazaru Aduresseu.	Tsunami			
STAPLE-E Summary:	Technical, Administrative			
Priority:	High			
Estimated Cost:	\$1,000 - \$30,000			
Implementation Schedule:	1-3 years			
Coordinating Agency:	Port Arthur Floodplain Management			
Potential Funding Sources:	Local operating budgets			

#### 2017 Analysis:

	Port Arthur (Past Action) – 33
Proposed Action:	Receive maximum credit for the NFIP CRS.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Dam Failure, Flooding, Hurricanes/Tropical Storms, Thunderstorms,	
	Tsunami	
STAPLE-E Summary:	Technical, Administrative	
Priority:	High	
Estimated Cost:	\$1,000 - \$30,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	Port Arthur Floodplain Management	
Potential Funding Sources:	Local operating budgets	

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Port Arthur (Past Action) – 34
Proposed Action:	Develop/implement bus transportation for hurricane evacuation.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Hurricanes/Tropical Storms
STAPLE-E Summary:	Social, Technical, Administrative, Political
Priority:	High
Estimated Cost:	\$100,000 - \$300,000
Implementation Schedule:	6-12 months
Coordinating Agancy:	Port Arthur Emergency Management, Jefferson County Emergency
coordinating Agency.	Management
Potential Funding Sources:	Local operating budgets

### 2017 Analysis:

	Port Arthur (Past Action) – 35
Proposed Action:	Develop/implement transportation plan for special needs populations.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Dam Failure, Drought, Earthquake, Extreme Heat, Flood, Geologic Hazard, Hazardous Materials, Hurricane/Tropical Storm, Severe Winter Weather, Terrorism, Thunderstorm, Tornado, Tsunami, Water Contamination, Wildfire
STAPLE-E Summary:	Social, Technical, Administrative, Political
Priority:	High
Estimated Cost:	\$100,000 - \$300,000
Implementation Schedule:	6-12 months
Coordinating Agency:	City of Port Arthur, Jefferson County Emergency Management
Potential Funding Sources:	Local operating budgets

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Port Arthur (Past Action) – 36
Proposed Action:	Assist in implementation of 211 TX Linkage Access Service.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Flooding, Hurricanes/Tropical Storms, Thunderstorms
STAPLE-E Summary:	Technical, Administrative, Political, Legal, Economic
Priority:	High
Estimated Cost:	\$50,000 - \$500,000
Implementation Schedule:	1-3 years
Coordinating Agency:	City of Port Arthur, Jefferson County
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets

# 2017 Analysis:

	Port Arthur (Past Action) – 37
Proposed Action:	Add 8-foot by 7-foot concrete box in Tiger Bayou.

MITIGATION ACTION DETAILS		
Hazard Addrossod:	Dam Failure, Flooding, Hurricanes/Tropical Storms, Thunderstorms,	
	Tsunami	
STAPLE-E Summary:	Technical, Administrative, Political, Legal, Economic	
Priority:	High	
Estimated Cost:	\$50,000 - \$500,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	City of Port Arthur, Jefferson County	
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets	

# 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Port Arthur (Past Action) – 38
Proposed Action:	Improve Hwy. 365 from Hwy. 69 to Rhodair Gully.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Dam Failure, Flooding, Hurricanes/Tropical Storms, Thunderstorms, Tsunami
STAPLE-E Summary:	Technical, Administrative, Political, Legal, Economic
Priority:	High
Estimated Cost:	\$50,000 - \$500,000
Implementation Schedule:	1-3 years
Coordinating Agency:	City of Port Arthur, Jefferson County
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets

#### 2017 Analysis:

	Port Arthur (Past Action) – 39
Proposed Action:	Decrease floodplain width in North Port Acres Ditch.

MITIGATION ACTION DETAILS					
Hazard Addrossod:	Dam Failure, Flooding, Hurricanes/Tropical Storms, Thunderstorms,				
	Tsunami				
STAPLE-E Summary:	Technical, Administrative				
Priority:	High				
Estimated Cost:	\$100,000 - \$300,000				
Implementation Schedule:	1-3 years				
Coordinating Agency:	Port Arthur Floodplain Management				
Potential Funding Sources:	Local operating budgets				

### 2017 Analysis:

# Port Neches

	Port Neches (Past Action) – 1
Proposed Action:	To build a structure or structures (including a dome or domes) in coastal (or near coastal) jurisdictions that can withstand 200 mile per hour winds and act as shelters. This is in conjunction with the Texas Safe Shelter Initiative.

MITIGATION ACTION DETAILS					
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado				
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal				
Priority:	High				
Estimated Cost:	\$1.6 Million				
Implementation Schedule:	1-3 years				
Coordinating Agency:	Jefferson County Emergency Management, TDEM				
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM				

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Port Neches (Past Action) – 2
Proposed Action:	Retrofit existing structures and/or construct new structures to act as shelters during and after Hurricanes/Tropical Storms and other severe weather events.

MITIGATION ACTION DETAILS					
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado				
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal				
Priority:	High				
Estimated Cost:	\$1 - \$2 Million				
Implementation Schedule:	1-3 years				
Coordinating Agency:	Jefferson County, SETRPC, TDEM				
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM				

#### 2017 Analysis:

	Port Neches (Past Action) – 3
Proposed Action:	Storm harden/retrofit critical and non-critical facilities throughout Port
	Neches. Actions can include but are not limited to window shutters,
	roof straps, flood proofing, and roll-up door reinforcement (i.e. for fire
	stations).

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado				
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic				
Priority:	High				
Estimated Cost:	\$10,000 - \$500,000				
Implementation Schedule:	1-2 years				
Coordinating Agency:	City of Port Neches, Jefferson County				
Potential Funding Sources:	HMPG, PDM, FMA, local operating budgets				

### 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update.

						Port Nech	es (P	ast Actio	n) – 4
Proposed Action:	Elevate new a throughout Po	and ex ort Nec	tisting hes.	flood	prone	structures	and	infrastru	icture

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flooding, Hurricane/Tropical Storm				
STAPLE-E Summary:	Technical, Administrative, Economic				
Priority:	High				
Estimated Cost:	\$20,000 - \$100,000				
Implementation Schedule:	1-3 years				
Coordinating Agency:	City of Port Neches, Jefferson County				
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL				

### 2017 Analysis:

					Port N	leches	s (Past A	Action) – 5
Proposed Action:	Replace througho	and/or out Port I	upgrade Neches.	bridges,	culverts	and	other	crossings

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flooding, Geologic Hazards				
STAPLE-E Summary:	Technical, Administrative, Legal				
Priority:	High				
Estimated Cost:	\$200,000 - \$1 Million				
Implementation Schedule:	2-5 years				
Coordinating Agency:	City of Port Neches, Jefferson County, TXDOT				
Potential Funding Sources:	HMGP, PDM, local operating budgets				

### 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update.

	Port Neches (Past Action) – 6
Proposed Action:	Pursue drainage improvements throughout Port Neches. Actions can
	include but are not limited to installing/upgrading culverts and headwalls as well as enlarging storm water ditches and canals.

MITIGATION ACTION DETAILS				
Hazard Addressed:	Flooding, Hurricanes/Tropical Storms, Thunderstorms			
STAPLE-E Summary:	Technical, Administrative, Political, Legal, Economic			
Priority:	High			
Estimated Cost:	\$50,000 - \$500,000			
Implementation Schedule:	1-3 years			
Coordinating Agency:	City of Port Neches, Jefferson County			
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets			

### 2017 Analysis:

		Port Neches (Past Action) – 7
Propos	sed Action:	Bury underground, secure or otherwise harden exposed or vulnerable pipelines.

MITIGATION ACTION DETAILS							
Hazard Addressed:	Flood,	Hazardous	Material,	Hurricane/Tropical	Storm,		
	Thunderstorm, Tornado						
STAPLE-E Summary:	Technical, Administrative, Economic						
Priority:	High						
Estimated Cost:	\$100,000 - \$500,000						
Implementation Schedule:	3-5 years						
Coordinating Agency:	City of Port Neches, Jefferson County, Entergy						
Potential Funding Sources:	HMGP, PDM, local operating budgets						

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Port Neches (Past Action) – 8
Proposed Action:	Provide generators/back-up power systems for critical facilities (including but not limited to lift stations, water plants, police, EMS, Fire and other first responder facilities) throughout Port Neches.

MITIGATION ACTION DETAILS				
Hazard Addressed: Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado				
STAPLE-E Summary:	Technical, Administrative, Political, Legal			
Priority:	High			
Estimated Cost:	\$50,000 - \$200,000			
Implementation Schedule:	1-3 years			
Coordinating Agency:	City of Port Neches, Jefferson County			
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM			

### 2017 Analysis:

	Port Neches (Past Action) – 9			
Proposed Action:	Mitigate damage to utilities in order to maintain function during and after a hazard event. Actions can include but are not limited to:			
	<ul> <li>Burying utility lines underground</li> </ul>			
	<ul> <li>Provide frangible links/break away connections on utility poles</li> </ul>			
	<ul> <li>Harden utility poles by converting from wood to concrete or metal utility poles</li> </ul>			
	<ul> <li>Increasing the easement area/clearance of utility lines/ poles</li> </ul>			
	from trees lines			

MITIGATION ACTION DETAILS				
Hazard Addressed: Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado				
STAPLE-E Summary:	Social, Technical, Administrative, Legal, Economic			
Priority:	High			
Estimated Cost: \$100,000 - \$200,000				
Implementation Schedule:	1-5 years			
Coordinating Agency:	City of Port Neches, Jefferson County, Entergy			
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM			

### 2017 Analysis:

	Port Neches (Past Action) – 10			
Proposed Action:	Minimize damage to structures and infrastructure from falling trees. Actions include but are not limited to the following:			
	<ul> <li>Pursue and coordinate a dangerous tree and limb removal program to protect infrastructure and critical facilities from damage. This includes working with private homeowners for voluntary removal of hazardous trees and limbs on private property.</li> <li>Coordinate contracting to remove and/or trim trees that</li> </ul>			
	endanger structures, infrastructure, and vital roadways.			

MITIGATION ACTION DETAILS				
Hazard Addressed: Hurricane/Tropical Storm, Thunderstorm, Tornado				
STAPLE-E Summary:	Social, Technical, Administrative, Economic			
Priority:	High			
Estimated Cost:	\$50,000 - \$100,000			
Implementation Schedule:	6-12 months			
Coordinating Agency:	City of Port Neches, Jefferson County			
Potential Funding Sources:	FEMA PA, HMGP, local operating budgets			

2017	Analy	/cic·
2017	Anar	ysis:

**Defer Action –** Will include in the 2017 Plan Update.

			Po	ort N	eches (P	ast Actio	on) – 11
Proposed Action:	Coordinate	public-private	partnerships	to	ensure	special	needs
	population p	protected from	winter weathe	r.			

MITIGATION ACTION DETAILS				
Hazard Addressed: Severe Winter Weather				
Social, Technical, Administrative, Political				
riority: High				
Estimated Cost:	\$50,000			
Implementation Schedule:	1-2 years			
Coordinating Agency: Port Neches EMC				
Potential Funding Sources:	HMGP, PDM, local operating budgets			

# 2017 Analysis:

Proposed	Action:	

Conduct coastal storm presentations.

Port Neches (Past Action) – 12

MITIGATION ACTION DETAILS		
Hazard Addressed:	Hurricane/Tropical Storms	
STAPLE-E Summary:	Social, Technical, Administrative, Political, Economic	
Priority:	High	
Estimated Cost:	\$100 - \$5,000	
Implementation Schedule:	Every 12 months	
Coordinating Agency:	Port Neches EMC	
Potential Funding Sources:	Local operating budgets, HMGP, PDM	

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

					F	Port Neches	(Pas	t Acti	ion) – 13
Proposed Action:	Provide	the	public	with	educational	brochures	for	the	hazards
	identified as part of the 2011 Plan Update.								

MITIGATION ACTION DETAILS			
Hazard Addressed:	Drought, Earthquake, Extreme Heat, Flood, Geologic Hazard, Hazardous Materials, Hurricane/Tropical Storm, Severe Winter Weather, Terrorism, Thunderstorm, Tornado, Tsunami, Water Contamination, Wildfire		
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal, Economic		
Priority:	High		
Estimated Cost:	\$5,000 - \$50,000		
Implementation Schedule:	6-12 months		
Coordinating Agency:	City of Port Neches, Jefferson County, TDEM, FEMA		
Potential Funding Sources:	tial Funding Sources: HMGP, PDM, local operating budgets, EMPG		

#### 2017 Analysis:

	Port Neches (Past Action) – 14
Proposed Action:	Pursue the coordination, construction, expansion and maintenance of
	flood control structures/barriers for the purpose of mitigating damage
	and protect fresh water resources from storm surge, sea level rise, and
	other sources of salt water intrusion.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Hurricane/Tropical Storm	
STAPLE-E Summary:	Technical, Administrative, Economic	
Priority:	Medium	
Estimated Cost:	\$1 - \$2 Million	
Implementation Schedule:	2-5 years	
Coordinating Agency:	City of Port Neches, Jefferson County, USACE, SETRPC, FEMA	
Potential Funding Sources:	HMGP, PDM, USACE, Jefferson County, SETRPC	

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

					Port Neches (Pa	ist Actio	on) – 15
Proposed Action:	Construct appropriat	or e to	improve collect st	existing orm wate	detention/retention r to reduce flooding.	ponds	where

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flooding, Hurricanes/Tropical Storm, Thunderstorms	
STAPLE-E Summary:	Social, Technical, Administrative, Economic, Environmental	
Priority:	Medium	
Estimated Cost:	\$50,000 - \$100,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	City of Port Neches, Jefferson County	
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL, local operating budgets	

#### 2017 Analysis:

	Port Neches (Past Action) – 16
Proposed Action:	Retrofit existing structures or construct new structures to act as safe
	rooms during tornados or other severe weather events.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Tornados, Thunderstorms	
STAPLE-E Summary:	Social, Technical, Administrative, Economic	
Priority:	Medium	
Estimated Cost:	\$5,000 - \$100,000	
Implementation Schedule:	1-5 years	
Coordinating Agency:	City of Port Neches, Jefferson County	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

### 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update.

	Port Neches (Past Action) – 17
Proposed Action:	Identify and pursue any mitigation activities that would aid/ enhance evacuations throughout the Port Neches.

MITIGATION ACTION DETAILS			
Hazard Addressed:	Flood, Hazardous Materials, Hurricane/Tropical Storm, Thunderstorm, Tornado, Wildfire		
STAPLE-E Summary: Social, Technical, Administrative, Political, Legal, Economic			
Priority:	Medium		
Estimated Cost:	\$10,000 - \$500,000		
Implementation Schedule:	1-5 years		
Coordinating Agency:	City of Port Neches, Jefferson County		
Potential Funding Sources:	HMGP, PDM, local operating budgets		

### 2017 Analysis:

	Port Neches (Past Action) – 18
Proposed Action:	Acquire flood prone properties (including Repetitive Loss and Severe
	Repetitive Loss Properties).

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flooding, Hurricane/Tropical Storm, Thunderstorm	
STAPLE-E Summary:	Technical, Administrative, Economic, Environmental	
Priority:	Low	
Estimated Cost:	\$50,000 - \$140,000	
Implementation Schedule:	12 months	
Coordinating Agency:	City of Port Neches, Jefferson County	
Potential Funding Sources:	HMGP, FMA, PDM, RFC, SRL	

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Port Neches (Past Action) – 19
Proposed Action:	Pursue flood protection measures to protect from surge from Hurricanes/Tropical Storms. Actions can include but are not limited to constructing and/or upgrading sea walls, flood barriers, berms and various wet and dry flood proofing measures.

MITIGATION ACTION DETAILS					
Hazard Addressed:	Flood,	Hurricane/Tropical	Storm,	Thunderstorm,	Water
	Contam	ination			
STAPLE-E Summary:	Technica	al, Administrative, Poli	tical		
Priority:	Low				
Estimated Cost:	\$50,000	- \$3 Million			
Implementation Schedule:	1-5 years				
Coordinating Agency:	City of Port Neches, Jefferson County, USACE, TDEM				
Potential Funding Sources:	HMGP, I	PDM, FMA, RFC, SRL			

### 2017 Analysis:

	Port Neches (Past Action) – 20
Proposed Action:	Retrofit existing structures to act as cooling stations in times of extreme
	heat.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Extreme Heat	
STAPLE-E Summary:	Social, Technical, Administrative	
Priority:	Low	
Estimated Cost:	\$100,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	City of Port Neches, Jefferson County	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	Port Neches (Past Action) – 21
Proposed Action:	Develop areas of defensible space to prevent damage due to wildfires.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Wildfire	
STAPLE-E Summary:	Technical, Administrative, Economic	
Priority:	Low	
Estimated Cost:	\$50,000 - \$100,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	City of Port Neches, Jefferson County Emergency Management	
Potential Funding Sources:	HMGP, EMPG, PDM, local operating budgets	

#### 2017 Analysis:
	Port Neches (Past Action) – 22
Proposed Action:	Develop and enact water conservation or drought management plan,
	ordinance or strategies to be used during times of drought.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Drought	
STAPLE-E Summary:	Social, Technical, Administrative	
Priority:	Low	
Estimated Cost:	\$25,000 - \$50,000	
Implementation Schedule:	6-12 months	
Coordinating Agency:	City of Port Neches, Jefferson County	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

#### 2017 Analysis:

## South East Texas Regional Planning Commission (SETRPC)

	SETRPC (Past Action) – 1
Proposed Action:	To build a structure or structures (including a dome or domes) in coastal
	(or near coastal) jurisdictions that can withstand 200 mile per hour
	winds and act as shelters. This is in conjunction with the Texas Safe
	Shelter Initiative.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado, Tsunami	
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal	
Priority:	High	
Estimated Cost:	\$1 - \$2 Million	
Implementation Schedule:	1-3 years	
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, TDEM	
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM	

## 2017 Analysis:

Delete Action.

_		SETRPC (Past Action) – 2
	Proposed Action:	Retrofit existing structures and/or construct new structures to act as shelters during and after Hurricanes/Tropical Storms and other severe weather events.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Hurricane/Tropical Storm, Thunderstorm, Tornado, Tsunami
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal
Priority:	High
Estimated Cost:	\$1 - \$2 Million
Implementation Schedule:	1-3 years
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, TDEM
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM

#### 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update. Revise Estimate Cost to \$1.5 - \$5 Million.

	SETRPC (Past Action) – 3
Proposed Action:	Develop a database of contact information for 1 <sup>st</sup> responders,
	volunteers, and vulnerable populations. This also includes a database
	of assisted living/nursing homes throughout the region.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Dam Failure, Drought, Earthquake, Extreme Heat, Flood, Geologic Hazards, Hazardous Materials Incidents, Hurricanes/Tropical Storms, Severe Winter Weather, Terrorism, Thunderstorms, Tornados, Tsunami, Water Contamination, Wildfire	
STAPLE-E Summary:	Social, Technical, Political, Legal	
Priority:	High	
Estimated Cost:	\$1,000 - \$5,000	
Implementation Schedule:	6-12 months	
Coordinating Agency:	SETRPC, Harding County, Jefferson County, Orange County, local municipalities	
Potential Funding Sources:	HMGP, PDM, EMPG, local operating budgets	

**Defer Action –** Will include in the 2017 Plan Update.

	SETRPC (Past Action) – 4
Proposed Action:	Facilitate use of all mass notification systems including but not limited
	to the Southeast Texas Alerting Network (STAN), to notify and educate
	the public of impending hazardous events.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Dam Failure, Drought, Earthquake, Extreme Heat, Flood, Geologic Hazards, Hazardous Materials Incidents, Hurricanes/Tropical Storms, Severe Winter Weather, Terrorism, Thunderstorms, Tornados, Tsunami, Water Contamination, Wildfire	
STAPLE-E Summary:	Social, Technical, Political, Legal	
Priority:	High	
Estimated Cost:	\$1,000 - \$5,000	
Implementation Schedule:	6-12 months	
Coordinating Agency:	SETRPC, Harding County, Jefferson County, Orange County, local municipalities	
Potential Funding Sources:	HMGP, PDM, EMPG, local operating budgets	

#### 2017 Analysis:

	SETRPC (Past Action) – 5
Proposed Action:	Coordinate public/private partnerships to ensure special need
	populations are protected from health risks due to extreme weather
	conditions. Actions will be targeted toward citizens with physical
	limitations and may be unable to reach safety in times of severe
	weather. Volunteer groups may be available to assist by visiting special
	needs groups to ensure their safety and comfort during severe
	temperature extremes.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Dam Failure, Drought, Earthquake, Extreme Heat, Flood, Geologic Hazards, Hazardous Materials Incidents, Hurricanes/Tropical Storms, Severe Winter Weather, Terrorism, Thunderstorms, Tornados, Tsunami, Water Contamination, Wildfire	
STAPLE-E Summary:	Social, Technical, Administrative, Political	
Priority:	High	
Estimated Cost:	\$1,000 - \$5,000	
Implementation Schedule:	6-12 months	
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, local municipalities, hospitals, Councils of Aging	
Potential Funding Sources:	HMGP, PDM	

Defer Action – Will include in the 2017 Plan Update. Revise Estimated Cost to \$50,000 - \$100,000.

	SETRPC (Past Action) – 6
Proposed Action:	Coordinate a natural hazard public awareness campaign among the
	jurisdictions. Efforts may include tropical storm/hurricane awareness
	presentations, shelter-in-place presentations, evacuation maps,
	floodplain maps, flood control projects, storm tracking maps, safety
	tips flyers, preparedness articles in local newspapers, and other such
	information as it relates to natural hazards. Target audiences will
	include schools, neighborhood watch groups, various civic groups,
	neighborhood associations, community groups, and industry groups.
	FEMA publications will also be made available in city hall libraries,
	municipal courts, police and fire departments, public works
	departments, public access TV channels, city libraries, and on the
	SETRPC and jurisdictional websites.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Dam Failure, Drought, Earthquake, Extreme Heat, Flood, Geologic Hazards, Hazardous Materials Incidents, Hurricanes/Tropical Storms, Severe Winter Weather, Terrorism, Thunderstorms, Tornados, Tsunami, Water Contamination, Wildfire	
STAPLE-E Summary:	Social, Technical, Administrative, Legal	
Priority:	High	
Estimated Cost:	\$1,000 - \$50,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, local municipalities Emergency Management	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

**Defer Action –** Will include in the 2017 Plan Update. Revise Estimated Cost to \$25,000 to \$75,000.

				SETRPC (	Past A	ction) – 7
Proposed Action:	Coordinate	Emergency	Management	plans	for	coastal
	storms/hurric	cane events.	Specific efforts	will inclue	de en	couraging
	jurisdictions	to install an	d maintain back	k-up powe	er at	identified
	facilities, con	struct and de	esignate emergen	cy operati	ons ce	enters for
	disaster/eme	rgency operat	ions, and solicit p	articipatio	n in Co	ommunity
	Emergency Re	esponse Train	ing.			

MITIGATION ACTION DETAILS		
Hazard Addressed:	Hurricanes/Tropical Storms	
STAPLE-E Summary:	Technical, Administrative	
Priority:	High	
Estimated Cost:	\$1,000 - \$50,000	
Implementation Schedule:	1-5 years	
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, local municipalities	
Potential Funding Sources:	HMGP, EMPG, local operating budgets	

#### 2017 Analysis:

	SETRPC (Past Action) – 8
Proposed Action:	Identify special needs populations in the region by coordinating with
	home health agencies, medical equipment companies, local churches,
	and neighborhood associations. Organize strategies for evacuating
	special needs populations during a coastal storm, hurricane, or other
	such hazard.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Dam Failure, Drought, Earthquake, Extreme Heat, Flood, Geologic Hazards, Hazardous Materials Incidents, Hurricanes/Tropical Storms, Severe Winter Weather, Terrorism, Thunderstorms, Tornados, Tsunami, Water Contamination, Wildfire	
STAPLE-E Summary:	Social, Technical, Administrative, Political	
Priority:	High	
Estimated Cost:	\$1,000 - \$5,000	
Implementation Schedule:	6-12 months	
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, local municipalities, hospitals, Councils of Aging	
Potential Funding Sources:	HMGP, PDM	

#### 2017 Analysis:

	SETRPC (Past Action) – 9
Proposed Action:	Conduct flood insurance educational seminars for area realtors to
	increase their knowledge of the National Flood Insurance Program
	(NFIP) and the benefits to homeowners of securing flood insurance.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Dam Failure, Flooding, Hurricanes/Tropical Storms, Thunderstorms,	
STAPLE-E Summary:	Social, Technical, Administrative, Political	
Priority:	High	
Estimated Cost:	\$1,000 - \$5,000	
Implementation Schedule:	6-12 months	
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, local	
coordinating Agency.	municipalities, hospitals, Councils of Aging	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update. Revise Estimated Cost to \$20,000 - \$50,000.

	SETRPC (Past Action) – 10
Proposed Action:	Storm harden/retrofit existing and future critical facilities throughout
	the Southeast Texas Region. Actions can include but are not limited to
	window shutters, roof, flood proofing, and roll-up door reinforcement
	(i.e. fire stations).

MITIGATION ACTION DETAILS		
Hazard Addressed:	Dam Failure, Flood, Hurricane/Tropical Storm, Thunderstorm, Tornado, Tsunami	
STAPLE-E Summary:	Technical, Administrative, Political, Economic	
Priority:	High	
Estimated Cost:	\$10,000 - \$150,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, local municipalities	
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets	

#### 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update. Revise Estimated Cost to \$2 Million.

	SETRPC (Past Action) – 11
Proposed Action:	Elevate and/or upgrade existing flood prone roadways throughout the
	Southeast Texas region.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Dam Failure, Flood, Hurricane/Tropical Storm, Thunderstorm, Tsunami	
STAPLE-E Summary:	Social, Technical, Administrative, Political, Economic	
Priority:	High	
Estimated Cost:	\$70,000 - \$200,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, local municipalities	
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL, local operating budgets	

### 2017 Analysis: Delete Action.

	SETRPC (Past Action) – 12
Proposed Action:	Replace and/or upgrade existing bridges throughout the Southeast
	Texas region.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Dam Failure, Flooding, Geologic Hazard, Hurricane/Tropical Storm, Thunderstorm, Tsunami	
STAPLE-E Summary:	Technical, Administrative	
Priority:	High	
Estimated Cost:	\$100,000 - \$1 Million	
Implementation Schedule:	1-5 years	
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, TXDOT	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

### 2017 Analysis:

**Delete Action.** 

	SETRPC (Past Action) – 13
Proposed Action:	Pursue drainage improvements throughout the region. Actions can
	include but are not limited to installing/upgrading culverts and
	headwalls as well as enlarging storm water ditches and canals.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Dam Failure, Flooding, Hurricane/Tropical Storm, Thunderstorm,	
STAPLE-E Summary:	Social, Technical, Administrative, Legal, Economic	
Priority:	High	
Estimated Cost:	\$50,000 - \$500,000	
Implementation Schedule:	1-3 years	
Coordinating Agancy	SETRPC, Hardin County, Jefferson County, Orange County, local	
Coordinating Agency.	municipalities	
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets	

**Defer Action –** Will include in the 2017 Plan Update.

	SETRPC (Past Action) – 14
Proposed Action:	Coordinate with federal, state, and local partners to provide training
	opportunities for first responders.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Dam Failure, Drought, Earthquake, Extreme Heat, Flood, Geologic Hazards, Hazardous Materials Incidents, Hurricanes/Tropical Storms, Severe Winter Weather, Terrorism, Thunderstorms, Tornados, Tsunami, Water Contamination, Wildfire	
STAPLE-E Summary:	Social, Technical, Administrative, Legal, Economic	
Priority:	High	
Estimated Cost:	\$1,000 - \$50,000	
Implementation Schedule:	6-12 months	
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, local municipalities, TDEM, DOJ, DHS	
Potential Funding Sources:	HMGP, EMPG, Homeland Security grants, local operating budgets	

### 2017 Analysis:

	SETRPC (Past Action) – 15
Proposed Action:	Mitigate damage to utilities in order to maintain function during and after a hazard event. Actions can include but are not limited to:
	<ul> <li>Burying utility lines underground</li> </ul>
	<ul> <li>Provide frangible links/break away connections on utility poles</li> </ul>
	<ul> <li>Harden utility poles by converting from wood to concrete or metal utility poles</li> </ul>
	<ul> <li>Increasing the easement area/clearance of utility lines/ poles</li> </ul>
	from trees lines

MITIGATION ACTION DETAILS		
Hazard Addressed:	Dam Failure, Flooding, Hurricane/Tropical Storm, Severe Winter Storm, Thunderstorm, Tornado	
STAPLE-E Summary:	Social, Technical, Administrative, Political, Economic	
Priority:	High	
Estimated Cost:	\$100,000 - \$500,000	
Implementation Schedule:	1-5 years	
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, local municipalities, TXDOT, Entergy	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update. Revise Estimated Cost to \$150,000 - \$600,000.

	SETRPC (Past Action) – 16
Proposed Action:	Identify and pursue any mitigation activities that would aid/ enhance
	evacuations throughout the region.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Dam Failure, Flood, Hazardous Materials, Hurricane/Tropical Storm, Severe Winter Storm, Thunderstorm, Tornado, Tsunami, Wildfire	
STAPLE-E Summary:	Social, Technical, Administrative	
Priority:	High	
Estimated Cost:	\$50,000 - \$500,000	
Implementation Schedule:	1-5 years	
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, local municipalities, TXDOT	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update.

	SETRPC (Past Action) – 17
Proposed Action:	Coordinate local match/cost-share agreements between the SETRPC,
	county and municipal governments.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Dam Failure, Drought, Earthquake, Extreme Heat, Flood, Geologic Hazards, Hazardous Materials Incidents, Hurricanes/Tropical Storms, Severe Winter Weather, Terrorism, Thunderstorms, Tornados, Tsunami, Water Contamination, Wildfire
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal
Priority:	High
Estimated Cost:	\$1,000
Implementation Schedule:	6-12 months
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, local municipalities
Potential Funding Sources:	HMGP, PDM, local operating budgets

#### 2017 Analysis:

Delete Action.

					SETRP	PC (Past Action) – 18
	Proposed Action:	Coordinate	project	application/funding	for	cross-jurisdictional
I		mitigation ne	eas.			

MITIGATION ACTION DETAILS		
Hazard Addressed:	Dam Failure, Drought, Earthquake, Extreme Heat, Flood, Geologic Hazards, Hazardous Materials Incidents, Hurricanes/Tropical Storms, Severe Winter Weather, Terrorism, Thunderstorms, Tornados, Tsunami, Water Contamination, Wildfire	
STAPLE-E Summary:	Social, Technical, Administrative, Political, Legal	
Priority:	High	
Estimated Cost:	\$1,000	
Implementation Schedule:	6-12 months	
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, local municipalities	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

2017 Analysis:		
Delete Action.		

	SETRPC (I	Past Action) – 19
Proposed Actio	Continually review, revise, update, and systema floodplain data and maps of flood prone areas Southeast Texas Region.	tically maintain throughout the

MITIGATION ACTION DETAILS		
Hazard Addressed:	Flooding, Hurricane/Tropical Storm, Thunderstorm	
STAPLE-E Summary:	Social, Technical, Administrative, Legal	
Priority:	High	
Estimated Cost:	\$1,000 - \$5,000	
Implementation Schedule:	1-5 years	
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, local municipalities	
Potential Funding Sources:	HMGP, local operating budgets	

	SETRPC (Past Action) – 20
Proposed Action:	Pursue the coordination, construction, expansion and maintenance of
	flood control structures/barriers for the purpose of mitigating damage
	from storm surge and seal level rise and other sources of salt water
	intrusion.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Hurricane/Tropical Storm, Tsunami, Water Contamination	
STAPLE-E Summary:	Technical, Administrative	
Priority:	High	
Estimated Cost:	\$10 Million	
Implementation Schedule:	5 years	
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, local municipalities, TXDOT, USACE	
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets	

#### 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update.

	SETRPC (Past Action) – 21
Proposed Action:	Pursue the coordination, construction, expansion and maintenance of
	flood control structures/barriers for the purpose of protecting potable
	water sources and agricultural resources from water contamination
	and salt water intrusion.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Hurricane/Tropical Storm, Tsunami, Water Contamination	
STAPLE-E Summary:	Technical, Administrative	
Priority:	High	
Estimated Cost:	\$10 Million	
Implementation Schedule:	5 years	
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, local municipalities, TXDOT, USACE	
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets	

#### 2017 Analysis:

	SETRPC (Past Action) – 22
Proposed Action:	Pursue the identification and construction of alternate fresh water
	sources.

MITIGATION ACTION DETAILS		
Hazard Addressed:	Drought, Water Contamination	
STAPLE-E Summary:	Technical, Administrative, Economic	
Priority:	High	
Estimated Cost:	\$75,000 - \$100,000	
Implementation Schedule:	1-3 years	
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, local municipalities	
Potential Funding Sources:	HMGP, PDM, local operating budgets	

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update. Revise Estimated Cost to \$100,000 - \$200,000.

	SETRPC (Past Action) – 23
Proposed Action:	Construct water retention ponds to collect storm water run-off and use
	as an alternate water source for agricultural resources throughout the
	Southeast Texas Region.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Drought, Flooding, Hurricane/Tropical Storm, Thunderstorm,
	Isunami
STAPLE-E Summary:	Technical, Administrative, Economic
Priority:	High
Estimated Cost:	\$75,000 - \$100,000
Implementation Schedule:	1-3 years
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, local
	municipalities
Potential Funding Sources:	HMGP, PDM, local operating budgets

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update. Revise Estimated Cost to \$100,000 - \$250,000.

	SETRPC (Past Action) – 24
Proposed Action:	<ul> <li>Minimize damage to existing and future structures and infrastructure from falling trees. Actions include but are not limited to the following:</li> <li>Pursue and coordinate a dangerous tree and limb removal program to protect infrastructure and critical facilities from damage. This includes working with private homeowners for voluntary removal of hazardous trees and limbs on private property.</li> <li>Coordinate contracting to remove and/or trim trees that endanger structures, infrastructure, and vital roadways.</li> </ul>

MITIGATION ACTION DETAILS				
Hazard Addressed:	Hurricane/Tropical Storm, Severe Winter Weather, Thunderstorm, Tornado			
STAPLE-E Summary:	Social, Technical, Administrative, Economic			
Priority:	Medium			
Estimated Cost:	\$50,000 - \$100,000			
Implementation Schedule:	6 months – 5 years			
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, local municipalities, TXDOT, Entergy			
Potential Funding Sources:	HMGP, PDM, local operating budgets			

### 2017 Analysis:

						SETRPC (	Pas	t Actio	on) – 25
Proposed Action:	Secure	and	maintain	backup	information	systems	to	store	critical
	informa	ition	at off-site	location	s.				

MITIGATION ACTION DETAILS	
Hazard Addressed:	Dam Failure, Drought, Earthquake, Extreme Heat, Flood, Geologic Hazards, Hazardous Materials Incidents, Hurricanes/Tropical Storms, Severe Winter Weather, Terrorism, Thunderstorms, Tornados, Tsunami, Water Contamination, Wildfire
STAPLE-E Summary:	Technical, Administrative
Priority:	Medium
Estimated Cost:	\$50,000
Implementation Schedule:	6-12 months
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, local municipalities
Potential Funding Sources:	HMGP, PDM, EMPG, local operating budgets

**Defer Action –** Will include in the 2017 Plan Update. Revise Estimated Cost to \$50,000 - \$350,000.

	SETRPC (Past Action) – 26
Proposed Action:	Coordinate with county and municipal governments to allow the
	SETRPC to maintain a copy of all local ordinances relevant to mitigation
	activities.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Dam Failure, Drought, Earthquake, Extreme Heat, Flood, Geologic Hazards, Hazardous Materials Incidents, Hurricanes/Tropical Storms, Severe Winter Weather, Terrorism, Thunderstorms, Tornados, Tsunami, Water Contamination, Wildfire
STAPLE-E Summary:	Technical, Administrative
Priority:	Medium
Estimated Cost:	\$1,000 - \$5,000
Implementation Schedule:	6-12 months
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, local municipalities
Potential Funding Sources:	HMGP, PDM, local operating budgets

### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update. Revise Estimated Cost to \$1,000 - \$15,000.

Proposed Action:
------------------

Acquire flood prone properties throughout the region.

SETRPC (Past Action) – 27

MITIGATION ACTION DETAILS	
Hazard Addressed:	Dam Failure, Flooding, Hurricane/Tropical Storm, Thunderstorm, Tsunami
STAPLE-E Summary:	Social, Technical, Administrative, Environmental, Economic
Priority:	Medium
Estimated Cost:	\$90,000 - \$200,000
Implementation Schedule:	1-3 years
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, local municipalities
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL, local operating budgets

#### 2017 Analysis:

Defer Action – Will include in the 2017 Plan Update. Revise Estimated Cost to \$90,000 - \$10 Million.

	SETRPC (Past Action) – 28
Proposed Action:	Elevate existing flood prone properties throughout the region.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Dam Failure, Flood, Hurricane/Tropical Storm, Thunderstorm, Tsunami
STAPLE-E Summary:	Social, Technical, Administrative, Environmental, Economic
Priority:	Medium
Estimated Cost:	\$90,000 - \$200,000
Implementation Schedule:	1-3 years
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, local municipalities
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL, local operating budgets

#### 2017 Analysis:

**Defer Action –** Will include in the 2017 Plan Update. Revise Estimated Cost to \$90,000 - \$2 Million.

	SETRPC (Past Action) – 29
Proposed Action:	Improve quality of local information on vulnerable items (assets and
	populations) for the purpose of more accurate risk and damage
	assessments. Work with other jurisdictions in region to get data as up
	to date and complete as possible.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Dam Failure, Drought, Earthquake, Extreme Heat, Flood, Geologic Hazards, Hazardous Materials Incidents, Hurricanes/Tropical Storms, Severe Winter Weather, Terrorism, Thunderstorms, Tornados, Tsunami, Water Contamination, Wildfire
STAPLE-E Summary:	Technical, Administrative
Priority:	Medium
Estimated Cost:	\$1,000 - \$10,000
Implementation Schedule:	6-12 months
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, local municipalities
Potential Funding Sources:	HMGP, PDM, local operating budgets

2017 Analysis:	
Delete Action.	

	SETRPC (Past Action) – 30
Proposed Action:	Provide educational seminars and brochures regarding the voluntary
	Community Rating System (CRS).

MITIGATION ACTION DETAILS	
Hazard Addressed:	Dam Failure, Flooding, Hurricane/Tropical Storm, Thunderstorm, Tsunami
STAPLE-E Summary:	Technical, Administrative
Priority:	Low
Estimated Cost:	\$1,000 - \$10,000
Implementation Schedule:	6-12 months
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, local municipalities
Potential Funding Sources:	HMGP, PDM, local operating budgets

Delete Action.

	SETRPC (Past Action) – 31
Proposed Action:	Work with local municipalities, county governments, local universities
	and other related entities to gather information on previous
	occurrences and the extent of Landslide and Riverine Erosion
	throughout the region.

MITIGATION ACTION DETAILS	
Hazard Addressed:	Geologic Hazard
STAPLE-E Summary:	Technical, Administrative
Priority:	Low
Estimated Cost:	\$1,000 - \$10,000
Implementation Schedule:	6-12 months
Coordinating Agency:	SETRPC, Hardin County, Jefferson County, Orange County, local municipalities, and local universities
Potential Funding Sources:	HMGP, PDM, local operating budgets

#### 2017 Analysis:

Delete Action.

# **SECTION 18: MITIGATION ACTIONS**

Summary	1
Jefferson County	7
Beaumont	42
Bevil Oaks	93
China	
Groves	131
Nederland	154
Nome	
Port Arthur	202
Port Neches	239
SETRPC	

## Summary

As discussed in Section 2, at the mitigation workshop the planning team and stakeholders met to develop mitigation actions for each of the natural and human-caused hazards included in the Plan. Each of the actions in this section were prioritized based on FEMA's Social, Technical, Administrative, Political, Legal, Economic and Environmental (STAPLEE) criteria necessary for the implementation of each action. As a result of this exercise, an overall priority was assigned to each mitigation action.

As part of the economic evaluation of the STAPLEE analysis, jurisdictions analyzed each action in terms of the overall costs, measuring whether the potential benefit to be gained from the action outweighed costs associated with it. As a result of this exercise, priority was assigned to each mitigation action by marking them as High (H), Moderate (M), or Low (L). An action that is ranked as "High" indicates that the action will be implemented as soon as funding is received. A "Moderate" action is one that may not be implemented right away depending on the cost and number of citizens served by the action. Actions ranked as "Low" indicate that they will not be implemented without first seeking grant funding and after "High" and "Moderate" actions have been completed.

Planning Team Members developed the actions below while also considering the risk reduction benefits and the effects the proposed action would have on new and existing buildings and infrastructure.

All mitigation actions created by Planning Team members are presented in this section in the form of Mitigation Action Worksheets. More than one hazard is sometimes listed for an action, if appropriate. Actions presented in this section represent a comprehensive range of mitigation actions per current State and FEMA Guidelines, including two actions, per hazard, and of two different types.

#### Table 18-1. Jefferson County and Participating Jurisdictions Mitigation Action Matrix

#### MITIGATION ACTION MATRIX

Actions presented in this matrix represent a comprehensive range and minimum number of required mitigation actions per current State and FEMA Guidelines, including two actions per hazard, and of two different types.

JEFFERSON COUNTY: MITIGATION ACTION MATRIX				
		Types of A	ction:	
HAZARDS	LOCAL PLANS/ REGULATIONS	STRUCTURAL/ INFRASTRUCTURE	NATURAL SYSTEM PROTECTION	EDUCATION & AWARENESS
Flood		XXXXXXXXXXXXXXXXXXXX		XX
Lightning	Х	XXXXX		Х
Hurricane	Х	XXXXXXXXXXXXXXXXXX	Х	XX
Extreme Heat		XXXXX		Х
Hail	Х	XXXX		Х
Thunderstorm Wind	Х	XXXXXXXXXX		XX
Tornado	Х	XXXXXXX		XX
Drought	Х	XXXX		Х
Wildfire		XXXXX		XX
Winter Storm	Х	XXXXX		XX
Coastal Erosion	х	XX	Х	

#### **BEAUMONT: MITIGATION ACTION MATRIX**

	Types of Action:			
HAZARDS	LOCAL PLANS/ REGULATIONS	STRUCTURAL/ INFRASTRUCTURE	NATURAL SYSTEM PROTECTION	EDUCATION & AWARENESS
Flood	XXX	*****		XXXXXXX
Lightning	х	XXXXXXXXXXX		XXXX
Hurricane	XXXX	XXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXX		XXXXXX
Extreme Heat		XXXXXXX		XXXX
Hail	Х	XXXXXXXXXXX		XXX
Thunderstorm Wind	XX	XXXXXXXXXXXXXXX		XXXX
Tornado	XXX	XXXXXXXXXXXXXXXX		XXXX
Drought		XX		х
Wildfire	XX	XXXXXXX		XXXX
Winter Storm	Х	XXXXXXXXXXXXX		XXXX

BEVIL OAKS: MITIGATION ACTION MATRIX				
		Types of A	ction:	
HAZARDS	LOCAL PLANS/ REGULATIONS	STRUCTURAL/ INFRASTRUCTURE	NATURAL SYSTEM PROTECTION	EDUCATION & AWARENESS
Flood	Х	XXXXXXX		Х
Lightning	XX	XX		х
Hurricane	XX	XXXXXXXX		х
Extreme Heat		XXX		XX
Hail	XX	XX		х
Thunderstorm Wind	XX	XXX		х
Tornado	XX	XXX		х
Drought	х			х
Wildfire		XXXX		х
Winter Storm	XX	XX		Х

#### CHINA: MITIGATION ACTION MATRIX

	Types of Action:			
HAZARDS	LOCAL PLANS/	STRUCTURAL/	NATURAL SYSTEM	EDUCATION &
	REGULATIONS	INFRASTRUCTURE	PROTECTION	AWARENESS
Flood		XXXXXXXXXXXX		XX
Lightning		XXX		х
Hurricane		XXXXXXXXXXX		XX
Extreme Heat		XX		х
Hail		XX		х
Thunderstorm Wind		XXXXX		х
Tornado		XXXXXX		XX
Drought	Х	XX		х
Wildfire		XX		XX
Winter Storm		XXX		Х

Х

Х

Drought

Wildfire

Winter Storm

GROVES: MITIGATION ACTION MATRIX				
	Types of Action:			
HAZARDS	LOCAL PLANS/ REGULATIONS	STRUCTURAL/ INFRASTRUCTURE	NATURAL SYSTEM PROTECTION	EDUCATION & AWARENESS
Flood		XXXXXXXXXXXXXXXXX		Х
Lightning	х	XXXXXXX		Х
Hurricane	Х	XXXXXXXXXXXXX		Х
Extreme Heat		XXXXXXX		Х
Hail	х	XXXXXXX		Х
Thunderstorm Wind	х	XXXXXXXX		Х
Tornado	х	XXXXXXXX		Х
Drought	х	Х		Х
Wildfire		XXXXXXXX		Х
Winter Storm	х	XXXXXXX		Х
	NEDERLAN	D: MITIGATION ACTION	MATRIX	
		Types of A	ction:	
HAZARDS	LOCAL PLANS/ REGULATIONS	STRUCTURAL/ INFRASTRUCTURE	NATURAL SYSTEM PROTECTION	EDUCATION & AWARENESS
Flood		xxxxxxxxxxxxxxxxxx		х
Lightning	Х	XXXXXXXX		Х
Hurricane	х	*****		Х
Extreme Heat		XXXXXXXX		Х
Hail	Х	XXXXXXXX		х
Thunderstorm Wind	Х	XXXXXXXXXXXX		Х
Tornado	Х	XXXXXXXXXXX		Х

Х

XXXXXXXXX

XXXXXXXX

Х

Х

Х

NOME: MITIGATION ACTION MATRIX				
	Types of Action:			
HAZARDS	LOCAL PLANS/ REGULATIONS	STRUCTURAL/ INFRASTRUCTURE	NATURAL SYSTEM PROTECTION	EDUCATION & AWARENESS
Flood		XXXXXXXXXX		XX
Lightning	Х	XXXX		Х
Hurricane	х	XXXXXXXXXXXXXXX		XX
Extreme Heat		XXXX		Х
Hail	х	XXXX		х
Thunderstorm Wind	х	XXXXXXXXX		XX
Tornado	х	XXXXXXX		XX
Drought	х	Х		х
Wildfire		XXXX		XX
Winter Storm	Х	XXXX		XX

PORT ARTHUR: MITIGATION ACTION MATRIX

	Types of Action:			
HAZARDS	LOCAL PLANS/ REGULATIONS	STRUCTURAL/ INFRASTRUCTURE	NATURAL SYSTEM PROTECTION	EDUCATION & AWARENESS
Flood	XX	*****		XX
Lightning	Х	XXXX		XX
Hurricane	XXX	*****		XXX
Extreme Heat		XXX		х
Hail	х	XXXX		XX
Thunderstorm Wind	х	XXXXXXXXXXXXXXXXXXXXXXX		XX
Tornado	х	XXXXXXX		х
Drought	х	Х		х
Wildfire		XXXXX		х
Winter Storm	Х	XXXX		Х

PORT NECHES: MITIGATION ACTION MATRIX				
	Types of Action:			
HAZARDS	LOCAL PLANS/ REGULATIONS	STRUCTURAL/ INFRASTRUCTURE	NATURAL SYSTEM PROTECTION	EDUCATION & AWARENESS
Flood		XXXXXXXXXX		х
Lightning	х	XXXX		Х
Hurricane	х	XXXXXXXXXXXXX		XX
Extreme Heat		XXXX		XX
Hail	х	XXXX		Х
Thunderstorm Wind	х	XXXXXXX		Х
Tornado	х	XXXXXXX		Х
Drought	х			Х
Wildfire		XXXXX		х
Winter Storm	х	XXXX		XX
SETRPC: MITIGATION ACTION MATRIX				
		Types of A	ction:	
HAZARDS	LOCAL PLANS/ REGULATIONS	Types of A STRUCTURAL/ INFRASTRUCTURE	ction: NATURAL SYSTEM PROTECTION	EDUCATION & AWARENESS
HAZARDS	LOCAL PLANS/ REGULATIONS	Types of A STRUCTURAL/ INFRASTRUCTURE XXXXXXXXXXX	ction: NATURAL SYSTEM PROTECTION	EDUCATION & AWARENESS XXXXXXX
HAZARDS Flood Lightning	LOCAL PLANS/ REGULATIONS XXX X	Types of A STRUCTURAL/ INFRASTRUCTURE XXXXXXXXXXX XXX	ction: NATURAL SYSTEM PROTECTION	EDUCATION & AWARENESS XXXXXXX XXXXX
HAZARDS Flood Lightning Hurricane	LOCAL PLANS/ REGULATIONS XXX X XXXX	Types of A STRUCTURAL/ INFRASTRUCTURE XXXXXXXXXXX XXX XXXX	ction: NATURAL SYSTEM PROTECTION	EDUCATION & AWARENESS XXXXXXX XXXXX XXXXX XXXXX
HAZARDS Flood Lightning Hurricane Extreme Heat	LOCAL PLANS/ REGULATIONS XXX X XXXX XXXX XXXX	Types of A STRUCTURAL/ INFRASTRUCTURE XXXXXXXXXXX XXX XXXX XXXXXXXXXXX X	ction: NATURAL SYSTEM PROTECTION	EDUCATION & AWARENESS XXXXXXX XXXXXX XXXXXX XXXXXX XXXXXX
HAZARDS Flood Lightning Hurricane Extreme Heat Hail	LOCAL PLANS/ REGULATIONS XXX X X XXXX XX XX XX XX	Types of A STRUCTURAL/ INFRASTRUCTURE XXXXXXXXXXX XXX XXXXXXXXXXX XXXXXXXXX	ction: NATURAL SYSTEM PROTECTION	EDUCATION & AWARENESS XXXXXXX XXXXXX XXXXXX XXXXXX XXXXXX XXXX
HAZARDS Flood Lightning Hurricane Extreme Heat Hail Thunderstorm Wind	LOCAL PLANS/ REGULATIONS XXX X XXXX XXXX XX XX XX XX	Types of A STRUCTURAL/ INFRASTRUCTURE XXXXXXXXXX XXX XXX XXXXXXXXXXX X XX XX	ction: NATURAL SYSTEM PROTECTION	EDUCATION & AWARENESS XXXXXXX XXXXXX XXXXXX XXXXXX XXXXXX XXXX
HAZARDS Flood Lightning Hurricane Extreme Heat Hail Thunderstorm Wind Tornado	LOCAL PLANS/ REGULATIONS XXX X XXXX XX XX XX XX XX XX XX	Types of A STRUCTURAL/ INFRASTRUCTURE XXXXXXXXXXX XXX XXX XXXXXXXXXXXX X XX XXXX	ction: NATURAL SYSTEM PROTECTION	EDUCATION & AWARENESS XXXXXXX XXXXX XXXXXX XXXXXX XXXXX XXXX XXXX
HAZARDS Flood Lightning Hurricane Extreme Heat Hail Thunderstorm Wind Tornado Drought	LOCAL PLANS/ REGULATIONS XXX X XXXX XX XX XX XX XX XXX XXX XXX	Types of ASTRUCTURAL/ INFRASTRUCTUREXXX	ction: NATURAL SYSTEM PROTECTION	EDUCATION & AWARENESS XXXXXXX XXXXXX XXXXXX XXXXXX XXXXX XXXX
HAZARDS Flood Lightning Hurricane Extreme Heat Hail Thunderstorm Wind Tornado Drought Wildfire	LOCAL PLANS/ REGULATIONS XXX X XXXX XX XX XX XX XXX XXX XXX XX	Types of ASTRUCTURAL/ INFRASTRUCTUREXX	ction: NATURAL SYSTEM PROTECTION	EDUCATION & AWARENESS XXXXXXX XXXXXX XXXXXX XXXXXX XXXXX XXXX

## Jefferson County

	Jefferson County – Action #1
Proposed Action:	To build a structure or structures (including a dome or domes) in coastal (or near coastal) jurisdictions that can withstand 200 mile per hour winds and act as shelters of last resort. This is in conjunction with the Texas Safe Shelter Initiative.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in Jefferson County
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in Port Arthur.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$1.6 Million
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM
Lead Agency/Department Responsible:	Jefferson County Emergency Management
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

COMMENTS

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies) Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 4

	Jefferson County – Action #2
Proposed Action:	Retrofit existing structures and/or construct new structures to act as residential shelters during and after Hurricanes and other severe weather events.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in Jefferson County
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents in Jefferson County.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$2 Million
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM
Lead Agency/Department Responsible:	Jefferson County Emergency Management
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Jefferson County – Action #3
Proposed Action:	Public structure strengthening for Jefferson County Courthouse/S.O./Jail. Actions can include but are not limited to window shutters, roof straps, flood proofing, roll-up door reinforcement (i.e. for fire stations), backup generator power with permanent hook-ups, hail resistant roofing materials, and surge protectors.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Jefferson County Courthouse/S.O./Jail
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Protect structure from damages, protect lives of staff, inmates and occupants, and ensure continuity of emergency services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS		
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat	
Effect on New/Existing Buildings:	Reduce risk to existing structure	
Priority (High, Moderate, Low):	High	
Estimated Cost:	\$1,000,000	
Potential Funding Sources:	HMGP, PDM, local operating budgets	
Lead Agency/Department Responsible:	Jefferson County Public Works	
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding	
Incorporation into Existing Plans:	Emergency Management Plan	

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Jefferson County – Action #4
Proposed Action:	Implement flood protection measures to protect from surge from Hurricanes/Tropical Storms. Actions can include but are not limited to constructing and/or upgrading sea walls, flood barriers, berms and various wet and dry flood proofing measures.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout Jefferson County
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of storm surge damages to structures and infrastructure.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Coastal Erosion
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$50-\$60 Million
Potential Funding Sources:	HMGP, PDM
Lead Agency/Department Responsible:	Jefferson County Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Master Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 4; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

	Jefferson County – Action #5
Proposed Action:	Coordinate and implement construction, expansion and maintenance of flood control structures/barriers for the purpose of mitigating damage and protect fresh water resources from storm surge, sea level rise, and other sources of salt water intrusion.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout Jefferson County
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk to structures and infrastructure through improved flood control measures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Coastal Erosion
Effect on New/Existing Buildings:	Reduce risk to existing structures and infrastructure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$1-\$2 Million
Potential Funding Sources:	HMGP, PDM, USACE, SETRPC
Lead Agency/Department Responsible:	Jefferson County, SETRPC
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Master Drainage Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 4; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

Proposed Action:	Jefferson County – Action #6 Restore sand dunes to protect inland resources during storm surge events.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Coastal locations throughout Jefferson County
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk to structures and infrastructure through fortified/restored dune system.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Natural System Protection

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Coastal Erosion
Effect on New/Existing Buildings:	Reduce risk to existing structures and infrastructure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$60-\$100 Million
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Jefferson County
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Dune Restoration and Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Jefferson County – Action #7
Proposed Action:	Identify and pursue any mitigation activities that would aid/enhance evacuations throughout the region.
BACKGROUND INFORMATION	
Jurisdiction/Location:	County-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Life safety benefits through preparedness.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Education and Awareness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood, Winter Storm, Tornado, Thunderstorm Wind, Wildfire
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$50,000 - \$500,000
Potential Funding Sources:	HMGP, PDM, local operating budget
Lead Agency/Department Responsible:	Jefferson County Emergency Management
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5

	Jefferson County – Action #8
Proposed Action:	Coordinate and work with the Lower Neches Valley
	Authority in order to use an Linva sand pit as a
	10.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Jefferson County south of I-10
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Lifesaving water alternate source and localized flood reduction.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Drought, Flood, Thunderstorm Wind, Hurricane
Effect on New/Existing Buildings:	Reduce damages to existing and future structures
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$75,000 - \$200,000
Potential Funding Sources:	HMGP, PDM, local operating budget, LNVA
Lead Agency/Department Responsible:	Jefferson County
Implementation Schedule:	Within 48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan, Water Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5

	Jefferson County – Action #9
Proposed Action:	Flood proof the Jefferson County courthouse elevators by installing a pump system.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Jefferson County Courthouse
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Protect elevator from flood damages.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing structure
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Jefferson County Public Works
Implementation Schedule:	Within 48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Jefferson County – Action #10
Proposed Action:	Retrofit the LNVA pumping system, which includes a number of pump stations, in order to increase capacity and allow stand-alone service when the Neches River is contaminated or a failure of the primary system occurs.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Jefferson County LNVA pumping system
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Increase capacity and reduce risk of inundation and/ or contamination. Ensure continuity of services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing structure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$3,680,000
Potential Funding Sources:	HMGP, PDM, LNVA operating budgets
Lead Agency/Department Responsible:	Jefferson County Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 4
Proposed Action:	Jefferson County – Action #11 Retrofit the primary diversion point and pumping system in order for the LNVA to fully control isolation and selection of the source of the water which flows into the pumping station and isolate either the Neches River or Pine Island Bayou should contamination occur.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Jefferson County LNVA pumping system
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of inundation and/or contamination. Ensure continuity of services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing structure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$562,000
Potential Funding Sources:	HMGP, PDM, LNVA operating budgets
Lead Agency/Department Responsible:	Jefferson County Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 4

Proposed Action:	Convert an existing 56-acre excavated dirt pit into a localized freshwater storage impoundment which would be used to support water to municipal water plants and industrial users in the Nederland, Port Neches, and Groves area of Jefferson County in the event of an interruption of canal service to the region.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Jefferson County
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Redundant water source and improved storm water management.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Drought
Effect on New/Existing Buildings:	Reduce risk to existing structure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5,468,000
Potential Funding Sources:	HMGP, PDM, LNVA operating budgets
Lead Agency/Department Responsible:	Jefferson County Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 4; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

	Jefferson County – Action #13
Proposed Action:	Enhance existing structures and construct additional water control features to allow the LNVA to isolate segments of canals in the event of contamination or localized bank failures in order to conserve and protect unaffected waters and continue deliveries to as many customers as possible while a clean-up or repair is addressed, rather than having a single event affect all customers of the entire system.
BACKGROUND INFORMATION	
Jurisdiction/Location:	LNVA System
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Ensure continuity of critical services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Drought, Hazardous Materials, Hurricane
Effect on New/Existing Buildings:	Reduce risk to existing structure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$375,000
Potential Funding Sources:	HMGP, PDM, LNVA operating budgets
Lead Agency/Department Responsible:	Jefferson County Public Works, LNVA
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

	Jefferson County – Action #14
Proposed Action:	Adopt and implement land use restrictions in high risk coastal erosion areas
BACKGROUND INFORMATION	•
Site and Location:	Jefferson County coastal areas
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce erosion and impacts through building restrictions
<b>Type of Action:</b> (Local Plans and Regulations, Structure and Infrastructure Projects, Natural Systems Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Coastal Erosion
Effect on new/existing buildings:	Reduce risk to future structures
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$2,000
Potential Funding Sources:	Local Funds
Lead Agency/Department Responsible:	Jefferson County Administration
Implementation Schedule:	Within 36-48 months of plan adoption pending funding
Incorporation into Existing Plans:	Local Ordinance

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies) Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5

	Jefferson County County-Wide – Action #15
Proposed Action:	Storm harden/retrofit critical facilities throughout Jefferson County. Actions can include but are not limited to window shutters, roof straps, flood proofing, roll-up door reinforcement (i.e. for fire stations), backup generator power with permanent hook-ups, hail resistant roofing materials, and surge protectors.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Critical facilities in Jefferson County and all jurisdictions
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Protect critical facilities from damages and ensure continuity of emergency services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$250,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	County and Local Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Jefferson County County-Wide – Action #16
Proposed Action:	Acquire/demolish flood-prone properties with an emphasis on Repetitive and Severe Repetitive Loss properties.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Flood-prone structures in Jefferson County and all jurisdictions
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Eliminate flood damages to flood-prone structures. Reduce burden on emergency services during flood events.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood
Effect on New/Existing Buildings:	Eliminate risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5,000,000
Potential Funding Sources:	HMGP, PDM, FMA
Lead Agency/Department Responsible:	County and City Emergency Management
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable =3 ; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 4; Legal = 4; Economically Sound = 4; and Environmentally Sound = 5

Proposed Action:	Jefferson County County-Wide – Action #17 Storm harden/retrofit regional communication sites and infrastructure throughout Jefferson County.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Regional communication sites and infrastructure in Jefferson County and all jurisdictions
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Protect critical communication infrastructure from damages and ensure continuity of emergency services and communications.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing infrastructure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$300,000
Potential Funding Sources:	HMGP, PDM, local operating budgets, EMPG
Lead Agency/Department Responsible:	County and City Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Jefferson County County-Wide – Action #18
Proposed Action:	Elevate new and existing flood-prone structures and infrastructure throughout Jefferson County
BACKGROUND INFORMATION	
Jurisdiction/Location:	Flood-prone structures in Jefferson County and all jurisdictions
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce flood damages to flood-prone structures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5,000,000
Potential Funding Sources:	HMGP, PDM, FMA
Lead Agency/Department Responsible:	County and City Emergency Management
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Local Ordinance, Comprehensive Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

		Jefferson County County-Wide – Action #19
Proposed Action:		Elevate or upgrade bridges, culverts and other crossings throughout Jefferson County to reduce damages to infrastructure and reduce flooding caused by undersized crossings and culverts.
BACKGROUND INFORMATION		
Jurisdiction/Location:		Undersized bridges and culverts in Jefferson County and all jurisdictions
<b>Risk Reduction Benefit</b> <i>Cost/Losses Avoided</i> ):	(Current	Reduce flood damages to infrastructure and surrounding structures.
<b>Type of Action</b> (Local Regulations, Structure and In projects, Natural System Pro Education and Awareness)	Plans and frastructure otection, or	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing infrastructure and structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$10,000,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	County and City Public Works, TXDOT
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 4

Proposed Action:	Pursue funding and implement drainage improvements throughout Jefferson County. Actions can include but are not limited to installing/upgrading culverts and headwalls as well as enlarging storm water ditches and canals.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout Jefferson County and all jurisdictions
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce flood damages to infrastructure and surrounding structures as a result of undersized drainage.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane
Effect on New/Existing Buildings:	Reduce risk to existing infrastructure and structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5,000,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	County and City Public Works and Engineering, Drainage District
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

	Jefferson County County-Wide – Action #21
Proposed Action:	Coordinate with federal, state, and local partners to provide training opportunities for first responders, including but not limited to HAXMAT, terrorism, Flood, Hurricane, Tornado, and other natural hazards as appropriate.
BACKGROUND INFORMATION	
Jurisdiction/Location:	County-wide and all jurisdictions
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce loss of life through preparedness.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Education and Awareness - Preparedness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood, Tornado, Hail, Lightning, Wildfire, Extreme Heat, Winter Storm, Thunderstorm Wind, Hazardous Material, Terrorism
Effect on New/Existing Buildings:	Reduce risk to existing structures through improved response
Priority (High, Moderate, Low):	High
Estimated Cost:	\$10,000
Potential Funding Sources:	Local operating budgets, HMGP, PDM, TDEM
Lead Agency/Department Responsible:	County and City Emergency Management
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Response Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5

	Jefferson County County-Wide – Action #22
Proposed Action:	Secure, bury, or otherwise harden exposed or vulnerable pipelines including water and sewer.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout Jefferson County, including all jurisdictions
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of pipeline failure through mitigation of exposed or vulnerable lines.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind, Winter Storm, Wildfire, Tornado, Lightning, Extreme Heat, Hail, Hazardous Material, Terrorism
Effect on New/Existing Buildings:	Reduce risk to existing infrastructure and structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5,000,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	County and City Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

		Jefferson County County-Wide – Action #23
Ρ	roposed Action:	Purchase and install backup generator power systems with permanent hook-ups for critical facilities including lift stations, water plants, police stations, EMS, fire stations, and other first responder facilities throughout Jefferson County.
В	ACKGROUND INFORMATION	
Ju	urisdiction/Location:	Critical facilities as indicated above in Jefferson County and all jurisdictions
R C	tisk Reduction Benefit (Current Tost/Losses Avoided):	Protect critical facilities from loss of power and ensure continuity of emergency services.
T R P E	ype of Action (Local Plans and egulations, Structure and Infrastructure rojects, Natural System Protection, or ducation and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$1-\$2 Million
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM
Lead Agency/Department Responsible:	County and Local Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5

	Jefferson County County-Wide – Action #24
Proposed Action:	Construct or upgrade existing detention/ retention ponds where appropriate to collect storm water to reduce flooding.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various Locations TBD throughout Jefferson County and all jurisdictions
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk to structures and infrastructure through improved flood control measures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Thunderstorm Wind, Hurricane
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$1,000,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	County and City Public Works, Drainage District
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 4

	Jefferson County County-Wide – Action #25
Proposed Action:	Retrofit existing structures and/or construct new structures to act as residential shelters during and after Hurricanes and other severe weather events.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout Jefferson County and all jurisdictions
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents in Jefferson County.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$2-\$3 Million
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM
Lead Agency/Department Responsible:	County and City Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

Proposed Action:	Jefferson County County-Wide – Action #26 Construct water retention ponds to collect storm water run-off and use as an alternate water source throughout Jefferson County.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Jefferson County and all jurisdictions
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Lifesaving water alternate source and localized flood reduction.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Drought, Flood, Thunderstorm Wind, Hurricane
Effect on New/Existing Buildings:	Reduce damages to existing and new structures
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$750,000 - \$2,000,000
Potential Funding Sources:	HMGP, PDM
Lead Agency/Department Responsible:	County and City Public Works, Drainage District
Implementation Schedule:	Within 48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Master Drainage Plan, Water Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Jefferson County County-Wide – Action #27
Proposed Action:	<ul> <li>Mitigate damage to utilities in order to maintain function during and after a hazard event. Actions can include but are not limited to: <ul> <li>Bury utility lines underground</li> <li>Provide frangible links/break away connections on utility poles</li> <li>Harden utility poles by converting from wood to concrete or metal utility poles</li> <li>Increase the easement area/clearance of utility lines/poles from tree lines</li> </ul> </li> </ul>
BACKGROUND INFORMATION	
Jurisdiction/Location:	Utility lines throughout Jefferson County and all jurisdictions
<b>Risk Reduction Benefit</b> <i>Cost/Losses Avoided</i> ):	(Current Reduce risk of loss of power and ensure continuity of emergency services.
<b>Type of Action</b> (Local Plan Regulations, Structure and Infras projects, Natural System Protect Education and Awareness)	and Structure and Infrastructure tructure tion, or

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind, Winter Storm, Tornado, Lightning, Hail
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$10,000,000
Potential Funding Sources:	Utility Fees, Federal Grants
Lead Agency/Department Responsible:	County and City Public Works, Entergy
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan, Local Ordinance

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Jefferson County County-Wide – Action #28
Proposed Action:	Develop/construct hardened, joint, multi- jurisdictional EOC.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Site TBD
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Improve coordinated emergency response and ensure continuity of emergency services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind, Winter Storm, Tornado, Lightning, Hail, Hazardous Materials, Terrorism
Effect on New/Existing Buildings:	Reduce risk to new structure
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$2 Million
Potential Funding Sources:	HMGP, PDM, EMPG, Homeland Security Grants
Lead Agency/Department Responsible:	County and City Emergency Management
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 4

	Jefferson County County-Wide – Action #29
Proposed Action:	Provide the public with educational brochures for mitigating damages, planning ahead for disasters and reduce the risk of injury during events including: mitigation measures such as window film, elevated appliances, surge protectors, insulating pipes, drought tolerant landscaping. Education on when to take cover, when to evacuate, locations of local safe rooms, signs of dehydration, and proper storage of flammable materials, or other appropriate materials to mitigate damages and health hazards.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Jefferson County – County-wide and all jurisdictions
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk to residents and structures through education and awareness.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Education and Awareness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood, Thunderstorm Wind, Winter Storm, Wildfire, Tornado, Lightning, Extreme Heat, Hail, Drought
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$50,000
Potential Funding Sources:	HMGP, PDM, local operating budgets, EMGP
Lead Agency/Department Responsible:	County and City Emergency Management
Implementation Schedule:	Within 24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5

	Jefferson County County-Wide – Action #30
Proposed Action:	Retrofit existing structures to act as cooling stations in times of extreme heat.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations (TBD) in Jefferson County and all jurisdictions
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk to vulnerable populations during extreme heat events.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Extreme Heat
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	County and City Administration
Implementation Schedule:	Within 24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Jefferson County County-Wide – Action #31
Proposed Action:	Develop areas of defensible space to prevent damage due to wildfires.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Wildland Urban Interface of Jefferson County and all jurisdictions
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of damages to structures in or near the WUI.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Wildfire
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	County and City Fire Department
Implementation Schedule:	Within 48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Community Wildfire Protection Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 4; Economically Sound = 5; and Environmentally Sound = 5

	Jefferson County County-Wide – Action #32
Proposed Action:	Develop and enact water conservation or drought management plans, ordinances or strategies to be used during times of drought.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Jefferson County – County-wide and all jurisdictions
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risks associated with drought through reduction in water usage during times of drought.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Drought
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$25,000 - \$50,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	County and City Administration
Implementation Schedule:	Within 24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Local Ordinances

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 4; Legal =5 ; Economically Sound = 5; and Environmentally Sound = 5

	Jefferson County County-Wide – Action #33
Proposed Action:	<ul> <li>Minimize damage to structures and infrastructure from falling trees. Actions include but are not limited to the following: <ul> <li>Pursue and coordinate a dangerous tree and limb removal program to protect infrastructure and critical facilities from damage. This includes working with private homeowners for voluntary removal of hazardous trees and limbs on private property.</li> <li>Coordinate contracts to remove and/or trim trees that endanger structures, infrastructure, and vital roadways.</li> </ul> </li> </ul>
BACKGROUND INFORMATION	
Jurisdiction/Location:	Trees near power lines throughout Jefferson County and all jurisdictions
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of loss of power and potential damages to structures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Thunderstorm Wind, Winter Storm, Tornado, Lightning, Hail
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	County and City Administration
Implementation Schedule:	Within 48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan, Local Ordinance

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

	Jefferson County County-Wide – Action #34
Proposed Action:	Develop and implement a public education program to educate residents of the risk of dam failure, actions to reduce risk, and evacuation routes and procedures for residents downstream of the Sam Rayburn Dam and the Toleda Bend Dam in the event of a dam failure.
BACKGROUND INFORMATION	
Jurisdiction/Location:	County-wide, including all jurisdictions
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce loss of life
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Education and Awareness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Prevent or minimize flood damage to structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$2,000 - \$5,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	County OEM
Implementation Schedule:	Within 48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

For purposes of the HMAP, upstream dam failure would affect part of the communities within Jefferson County boundaries. However, the impacts associated with these dams are flood related.

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

# Beaumont

	Beaumont – Action #1
Proposed Action:	Enhance GIS capabilities to develop and maintain a database and identify concentrations of at-risk structures to track community vulnerability to flooding.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce flood risk to structures through understanding risk and vulnerability.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$50,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Beaumont Floodplain Management
Implementation Schedule:	Within 24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan

#### COMMENTS

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 4; Economically Sound = 4; and Environmentally Sound = 5

Proposed Action:	Beaumont – Action #2 Acquire GIS hazard mapping online software of flood prone areas for residents and design professionals.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce flood risk to structures and residents and developers through understanding risk and vulnerability.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Education and Awareness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$50,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Beaumont Floodplain Management
Implementation Schedule:	Within 24 months of plan adoption pending available funding
Incorporation into Existing Plans:	N/A

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

Proposed Action:	Beaumont – Action #3 Enhance Damage Assessment capabilities to develop and maintain a database and identify concentrations of at-risk structures.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of damage to structures through understanding risk and vulnerability.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind, Tornado, Wildfire
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$50,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Beaumont Emergency Management
Implementation Schedule:	Within 24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

Proposed Action:	Beaumont – Action #4 Enhance Damage Assessment trainings to improve public outreach, speed recovery, enhance damage assessments, and reduce risk to first responders and building inspectors.
BACKGROUND INFORMATION	<u> </u>
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury and speed recovery after an event.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Education and Outreach - Response

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Lightning, Hurricane, Hail, Thunderstorm Wind, Wildfire, Winter Storm
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$50,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Beaumont Building Inspectors
Implementation Schedule:	Within 24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Response Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5

	Beaumont – Action #5
Proposed Action:	Retrofit existing city structures or build new structure(s) (including a dome or domes) in this jurisdiction that can withstand 200 mile per hour winds and act as shelters of last resort. This may be in conjunction with the Texas Safe Shelter Initiative.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in Beaumont
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in Beaumont.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$2,000,000
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Beaumont Emergency Management
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Beaumont – Action #6
Proposed Action:	Storm harden/retrofit critical facilities throughout Beaumont. Actions can include but are not limited to window shutters, roof straps, flood proofing, roll-up door reinforcement (i.e. for fire stations, police headquarters, EMS stations 1 and 2, and other critical infrastructure facilities), backup generator power with permanent hook-ups, hail resistant roofing materials, and surge protectors.
BACKGROUND INFORMATION	-
Jurisdiction/Location:	Critical facilities in Beaumont
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Protect critical facilities from damages and ensure continuity of emergency services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$7.4 Million
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Beaumont Administration
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

Proposed Action:	Harden the City of Beaumont Police headquarters at 255 College. Actions include but are not limited to storm shutters, window film, surge protectors, roof straps, hail and fire resistant roofing material, etc. Install generator with permanent hook-ups at the Emergency Operations Center at 700 Orleans.
BACKGROUND INFORMATION	
Jurisdiction/Location:	EOC and Police Headquarters
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Protect critical facilities from damages and ensure continuity of emergency services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$1 Million
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Beaumont Administration
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

Proposed Action:	Wind harden the South Radio Tower at 1550 Pine and Fire Headquarters at 400 Walnut. Actions include but are not limited to roof retrofits, installing storm shutters/screens, installing generators with permanent hook-ups, and hardening of bay doors (specifically fire stations).
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations as noted above
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Protect critical facilities from damages and ensure continuity of emergency services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$3.75 Million
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Beaumont Administration
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies) Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Beaumont – Action #9
Proposed Action:	Wind harden Baptist Hospital, at 3080 College, and Christus St. Elizabeth Hospital at 2830 Calder. Actions include but are not limited to roof retrofits, installing storm shutters/screens, installing generators with permanent hook-ups, and hardening of bay doors (specifically maintenance and facility areas).
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations as noted above
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Protect critical facilities from damages and ensure continuity of emergency services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$8 Million
Potential Funding Sources:	HMGP, PDM, local operating budgets, EMPG
Lead Agency/Department Responsible:	City of Beaumont Administration
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Beaumont – Action #10
Proposed Action:	At Christus St. Elizabeth Hospital install backup generators with permanent hook-ups and elevate key electrical equipment (such as Switchgear and ATS).
BACKGROUND INFORMATION	
Jurisdiction/Location:	St. Elizabeth Hospital
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Protect critical facility from damages and ensure continuity of emergency services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$3.1 Million
Potential Funding Sources:	HMGP, PDM, local operating budgets, EMPG
Lead Agency/Department Responsible:	City of Beaumont Administration, St. Elizabeth Hospital
Implementation Schedule:	Within 12024 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5

	Beaumont – Action #11
Proposed Action:	Upgrade Christus St. Elizabeth Hospitals and Port of Beaumont emergency communication systems to ensure continued communication with outside sources and first responders.
BACKGROUND INFORMATION	•
Jurisdiction/Location:	St. Elizabeth Hospital and Port of Beaumont
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Ensure communications and continuity of emergency services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$62,000
Potential Funding Sources:	HMGP, PDM, local operating budgets, EMPG
Lead Agency/Department Responsible:	City of Beaumont Administration, St. Elizabeth Hospital, Port of Beaumont
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Response Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5
Proposed Action:	Beaumont – Action #12 Install generators and permanent hook-ups for the Beaumont Independent School District at sites including but not limited to Westbrook, Police Building, Administrative Building and the Thomas Educational Support Center which are used for sheltering and emergency operation coordination centers.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Beaumont ISD locations
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Ensure continuity of emergency services and sheltering.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Flood, Winter Storm
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$650,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Beaumont Administration, ISD
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Response Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies) Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5

	Beaumont – Action #13
Proposed Action:	Procure mobile backup generators for the Port of Beaumont. Install permanent quick connections at critical locations.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Port of Beaumont
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Ensure continuity of services during and after events.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Flood, Winter Storm
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$60,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Beaumont Administration, Port of Beaumont
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Port of Beaumont SOP

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Beaumont – Action #14
Proposed Action:	<ul> <li>Mitigate damage to utilities in order to maintain function during and after a hazard event. Actions can include but are not limited to: <ul> <li>Bury utility lines underground</li> <li>Provide frangible links/break away connections on utility poles</li> <li>Harden utility poles by converting from wood to concrete or metal utility poles</li> <li>Increase the easement area/clearance of utility lines/poles from tree lines</li> <li>Relocate and/or update the data communications provided to the Wastewater treatment power supply station to prevent the interruption of operations</li> <li>Relocate and/or update the data communications provided to the Water treatment plant to prevent the interruption of operations.</li> </ul> </li> </ul>
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
RiskReductionBenefit(CurrentCost/Losses Avoided):	Reduce risk of power loss, associated damages, and line repairs, and reduce risk of loss of services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood, Winter Storm, Hail Lightning, Thunderstorm Wind, Tornado
Effect on New/Existing Buildings:	Reduce risk to existing structures and infrastructure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5.5 Million
Potential Funding Sources:	HMGP, PDM, Utility Fees, local operating budgets
Lead Agency/Department Responsible:	Beaumont Public Works, Entergy
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Beaumont – Action #15
Proposed Action:	Hardening Electric Grid and Communications to prevent damage to electric, phone and cable infrastructure for major roadways/ thoroughfares or access routes to critical infrastructure.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of power loss, associated damages, reduce risk of loss of services, and ensure continuity of services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood, Winter Storm, Hail, Lightning, Thunderstorm Wind, Tornado
Effect on New/Existing Buildings:	Reduce risk to existing structures and infrastructure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$10,000,000
Potential Funding Sources:	HMGP, PDM, Utility Fees, local operating budgets
Lead Agency/Department Responsible:	Beaumont Public Works, Entergy
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Response Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Beaumont – Action #16
Proposed Action:	Pursue drainage improvements in the City of Beaumont. Projects include but are not limited to the Tyrell Park Project, Caldwood Outfall, Phelan Boulevard Drainage Project, the Cartwright/Corley Project, and the Brockman Drainage Project.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations as indicated above
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce flood damages to structures and infrastructure due to inadequate drainage.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing structures and infrastructure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$50,000,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Beaumont Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Beaumont – Action #17
Proposed Action:	Implement drainage improvements in the City of Beaumont. Projects include, but are not limited to joint Drainage District 6 and Beaumont projects and Beaumont individual projects, such as Steve's Drive project, and High School Ditch Project (which includes Seventh Street, North Street, Broadway Box projects).
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations as indicated above
Risk Reduction Benefit (Current	Reduce flood damages to structures and infrastructure
Cost/Losses Avoided):	due to inadequate drainage.
Type of Action (Local Plans and	Structure and Infrastructure
Regulations, Structure and Infrastructure	
projects, Natural System Protection, or	
Education and Awareness)	

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing structures and infrastructure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$50,000,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Beaumont Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Beaumont – Action #18
Proposed Action:	Purse drainage improvements throughout the City of Beaumont. Actions can include but are not limited to installing/upgrading culverts and headwalls as well as enlarging storm water ditches and canals.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout the city.
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce flood damages to structures and infrastructure due to inadequate drainage.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing structures and infrastructure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$10,000,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Beaumont Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Beaumont – Action #19
Proposed Action:	Construct water retention ponds to collect storm water run-off and use as an alternate water source for storm control throughout the City of Beaumont.
BACKGROUND INFORMATION	-
Jurisdiction/Location:	TBD sites in Beaumont
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Localized flood reduction.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Drought, Flood, Thunderstorm Wind, Hurricane
Effect on New/Existing Buildings:	Reduce risk to existing structures and infrastructure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$1,000,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Beaumont Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Beaumont – Action #20
Proposed Action:	Complete bank stabilization project at Riverfront Park, Phase II.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Riverfront Park
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Localized flood reduction.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing structures and infrastructure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$3,000,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Beaumont Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Beaumont – Action #21
Proposed Action:	Pursue building a 50 million gallon holding lagoon to store wastewater in case of power outages and plant failure, and removal of sludge build-up in the two lagoons to increase the storage capacity at the Wastewater Treatment Plant in case of power outages and plant failure.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Wastewater Treatment Plant
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Increase the storage capacity at the Wastewater Treatment Plant in case of power outages and plant failure; reduce risk of contamination.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Hail, Lightning, Winter Storm, Thunderstorm Wind, Tornado, Extreme Heat
Effect on New/Existing Buildings:	Reduce risk to existing structures and infrastructure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$3,500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Beaumont Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Wastewater management plan

# Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies) Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

	Beaumont – Action #22
Proposed Action:	Pursue and construct alternate fresh water sources throughout the City of Beaumont. This include, but is not limited to increased capacity to maintain water pressure in case of system failures at the water treatment plant; installation of a 36" water transmission line to provide an alternate water transport method; installation of a raw water pipeline to replace the existing canal to prevent intention or natural pollution of the City's water supply; installation of new chemical feed facilities and building to safely store and use chlorine, etc.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout the City
RiskReductionBenefit(CurrentCost/Losses Avoided):	Reduce loss of potable water or inadequate water supply.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Drought
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5,000,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Beaumont Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Long Term Water Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Beaumont – Action #23
Proposed Action:	At Baptist Hospital, install 1) an on-site filtering system for water well and new pressurized water supply system, and 2) install a Built-in Decontamination System (includes shower, curtain system, hazardous water tank, and drain) in the main facility.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Baptist Hospital
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury to residents and first responders.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hazardous Materials
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$150,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Beaumont Administration, Hermann Memorial Hospital
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Hospital Operations Plan

# Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies) Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 4; Economically Sound = 4; and Environmentally Sound = 5

	Beaumont – Action #24
Proposed Action:	Retrofit the LNVA pumping system at Highway 105 to upgrade pumping capacity. This will allow the system to operate at full stand-alone service in times of contamination of water in the Neches River due to a hazard event.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Highway 105
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of contamination or loss of service.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hazardous Materials, Terrorism
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$3,680,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Beaumont Public Works, LNVA
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Master Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	Retrofit the primary LNVA diversion point and pumping system at 10550 Helbig Rd. to allow the LNVA to fully control, isolate, and section off the source of the water that flows into the pumping station. This would allow the LNVA to isolate either the Neches River or Pine Island Bayou should one suffer any form of contamination.
BACKGROUND INFORMATION	-
Jurisdiction/Location:	LNVA diversion point
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of contamination or loss of service.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hazardous Materials, Terrorism
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$562,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Beaumont Public Works, LNVA
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Master Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies) Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Beaumont – Action #26
Proposed Action:	Develop a database of contact information for first responders, volunteers, and vulnerable populations. This also includes a database of assisted living/nursing homes throughout the City of Beaumont or populations critically dependent on electric service.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk to vulnerable populations during extreme events.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Education and Awareness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Extreme Heat, Winter Storm, Tornado, Hail, Lightning, Thunderstorm Wind, Wildfire
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$15,000
Potential Funding Sources:	EMPG, local operating budgets
Lead Agency/Department Responsible:	Beaumont Administration
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Response Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Beaumont – Action #27
Proposed Action:	Public Awareness and Education of Vulnerable Population through creation of a database and special group in STAN (Southeast Texas Alerting Network regional emergency alerting system) to notify and educate the public of impending hazardous events.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Expedient access to and ability to communicate with those individuals, nursing homes, assisted living centers, hospitals, and others who are most at risk during extreme hazard events.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Education and Awareness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Extreme Heat, Flood, Hurricane, Winter Storm, Thunderstorm Wind, Tornado, Wildfire, Lightning, Hail
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$25,000
Potential Funding Sources:	EMPG, local operating budget
Lead Agency/Department Responsible:	Beaumont Emergency Management
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Operations Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Beaumont – Action #28
Proposed Action:	Plan for the Protection of Vulnerable Populations by identifying at-risk populations and coordinating with home health agencies, medical equipment companies, local churches and neighborhood associations to assist these populations during extreme weather events. Organize strategies for protecting vulnerable populations and develop a plan to expediently activate strategies when need be.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Protection of at-risk populations during extreme weather events.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Education and Awareness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Extreme Heat, Flood, Hurricane, Winter Storm, Thunderstorm Wind, Tornado, Wildfire Lightning
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$10,000
Potential Funding Sources:	Local operating budgets
Lead Agency/Department Responsible:	Beaumont Emergency Management
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Response Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Beaumont – Action #29
Proposed Action:	Coordinate a natural hazards public awareness campaign among agencies and the community. Efforts may include tropical storm/hurricane awareness presentations, shelter-in-place presentations, evacuation maps, floodplain maps, flood control projects, storm tracking maps, safety tips flyers, mitigation articles in local newspapers, and other such information as it relates to natural hazards. Target audiences will include schools, neighborhood watch groups, various civic groups, neighborhood associations, community groups, and industry groups. FEMA publications will also be made available in city hall libraries, municipal courts, police and fire departments, public works departments, public access TV channels, city libraries, and on the SETRPC and jurisdictional websites.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk damages as well as life safety benefits to residents through education and awareness.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Education and Awareness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Extreme Heat, Flood, Hurricanes, Winter Storm, Thunderstorm Wind, Tornado, Wildfire, Lightning, Hail, Drought
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$25,000 - \$50,000
Potential Funding Sources:	EMPG, local operating budgets, HMGP, PDM
Lead Agency/Department Responsible:	Beaumont Emergency Management
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Beaumont – Action #30
Proposed Action:	Coordinate Emergency Management Plans for coastal storms/hurricane events. Specific efforts will include encouraging agencies to install and maintain back-up power at identified facilities, construct and designate emergency operations centers for disaster/ emergency operations and solicit participation in Community Emergency Response Training.
BACKGROUND INFORMATION	-
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Ensure continuity of emergency services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$50,000
Potential Funding Sources:	Local operating budgets
Lead Agency/Department Responsible:	Beaumont Emergency Management
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Response Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Beaumont – Action #31
Proposed Action:	Conduct flood insurance educational seminars for area realtors to increase their knowledge of the National Flood Insurance Program (NFIP) and the benefits to homeowners of securing flood insurance.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce potential uninsured flood losses through education.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Education and Awareness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$10,000
Potential Funding Sources:	Local operating budgets, HMGP, PDM
Lead Agency/Department Responsible:	Beaumont Floodplain Management
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	Beaumont – Action #32 Coordinate with federal, state, and local partners to provide training opportunities for first responders.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce loss of life through education and preparedness.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Education and Awareness - Preparedness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood, Tornado, Hail, Lightning, Wildfire, Extreme Heat, Winter Storm, Thunderstorm Wind
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$15,000
Potential Funding Sources:	Local operating budgets, HMGP, PDM, EMPG
Lead Agency/Department Responsible:	Beaumont Emergency Management
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Beaumont – Action #33
Proposed Action:	Relocations of Fire Headquarters, Fire stations 1, 2, 7 and 11 to improve neighborhood coverage in accordance with the 2005 Pietsch (ISO) study.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce loss of life due to improved neighborhood coverage.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure - Preparedness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood, Tornado, Hail, Lightning, Wildfire, Extreme Heat, Winter Storm, Thunderstorm Wind
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$21,000,000
Potential Funding Sources:	Local operating budgets, HMGP, PDM
Lead Agency/Department Responsible:	Beaumont Emergency Management
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Beaumont – Action #34
Proposed Action:	Install on-site well and new pressurized water supply system to support Christus St. Elizabeth Hospital during loss of potable water.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Christus St. Elizabeth Hospital
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of water contamination and ensure continuity of services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hazardous Material
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$162,000
Potential Funding Sources:	Local operating budgets, HMGP, PDM, EMPG
Lead Agency/Department Responsible:	Beaumont Administration, St. Elizabeth Hospital
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Hospital operations plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Beaumont – Action #35
Proposed Action:	Acquire or elevate flood prone structures throughout the City of Beaumont with an emphasis on current Repetitive Loss and Severe Repetitive Loss properties.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout city
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce or eliminate flood damages to repetitive loss structures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$55,000,000
Potential Funding Sources:	Local operating budgets, HMGP, PDM, RFC
Lead Agency/Department Responsible:	Beaumont Administration
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Beaumont – Action #36
Proposed Action:	Coordinate a consolidated security checkpoint on entry to Plant Road to access industrial and chemical production and storage complexes.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Entry to Plant Road
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of terror attack on vulnerable targets.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Terrorism
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$175,000
Potential Funding Sources:	Local operating budgets, EMPG, Homeland Security Grants
Lead Agency/Department Responsible:	Beaumont Emergency Management
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Anti-Terrorism Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

		Beaumont – Action #37
1	Proposed Action:	Upgrade security system at the Port of Beaumont for access control on all exterior doors for all buildings as well as installing cameras and increased perimeter surveillance capabilities.
I	BACKGROUND INFORMATION	
J	Jurisdiction/Location:	Port of Beaumont
(	<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of terror attack on vulnerable targets.
1   	<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Terrorism
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$175,000
Potential Funding Sources:	Local operating budgets, EMPG, Homeland Security Grants
Lead Agency/Department Responsible:	Beaumont Emergency Management
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Anti-Terrorism Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Beaumont – Action #38
Proposed Action:	Install security systems at the City of Beaumont water utility sites, to include but not limited to, security walls around chemical tanks, water treatment plant and a river pump station, monitoring stations for purity testing at various sites throughout the city; surveillance at six elevated storage tanks and wastewater plant, etc. for access control on all exterior doors for all buildings as well as installing cameras and increased perimeter surveillance capabilities.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Beaumont Water Utility Sites
RiskReductionBenefit(CurrentCost/Losses Avoided):	Reduce risk of terror attack on vulnerable targets.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Terrorism
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$4,500,000
Potential Funding Sources:	Local operating budgets, EMPG, Homeland Security Grants
Lead Agency/Department Responsible:	Beaumont Emergency Management
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Anti-Terrorism Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Beaumont – Action #39
Proposed Action:	Upgrade surveillance capabilities at critical infrastructure sites around the City of Beaumont, to include, fire stations, police, fire and EMS headquarters, lift stations, communication towers and headquarters, etc.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Beaumont Critical Infrastructure Sites
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of terror attack on vulnerable targets.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Terrorism
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$200,000
Potential Funding Sources:	Local operating budgets, EMPG, Homeland Security Grants
Lead Agency/Department Responsible:	Beaumont Emergency Management
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Anti-Terrorism Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Beaumont – Action #40
Proposed Action:	Install generators with permanent hook-ups and harden roofs at sites at Lamar University. Sites include but are not limited to the University Police Station and Soccer Field House which serves as an emergency operations center.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Lamar University
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Ensure continuity of emergency services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood, Winter Storm, Thunderstorm Wind, Tornado
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$1,000,000
Potential Funding Sources:	Local operating budgets, HMGP, PDM
Lead Agency/Department Responsible:	Beaumont Emergency Management, Lamar University
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Operations Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	At Lamar University, replace/upgrade radio repeater/tower, replace/upgrade emergency notification siren tower equipment and tower, cleaning and restoration of tunnel network, and upgrade safety and security lightning throughout campus.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Lamar University
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Ensure continuity of emergency services and reduce risk to students and faculty through early warning.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood, Winter Storm, Thunderstorm Wind, Tornado, Hail, Lightning
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$2,357,000
Potential Funding Sources:	Local operating budgets, HMGP, PDM, EMPG
Lead Agency/Department Responsible:	Beaumont Emergency Management, Lamar University
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Operations Plan

# Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies) Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 4

	Beaumont – Action #42
Proposed Action:	Elevate flood prone properties/structures and key infrastructure and electrical equipment throughout the City of Beaumont.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout the city.
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of flood damages to key structures and infrastructure.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$1,000,000
Potential Funding Sources:	Local operating budgets, HMGP, PDM
Lead Agency/Department Responsible:	Beaumont Public Works
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Beaumont – Action #43
Proposed Action:	<ul> <li>Minimize damage to structures and infrastructure from falling trees. Actions include but are not limited to the following: <ul> <li>Pursue and coordinate a dangerous tree and limb removal program to protect infrastructure and critical facilities from damage. This includes working with private homeowners for voluntary removal of hazardous trees and limbs on private property.</li> <li>Coordinate contract to remove and/or trip trees that endanger structures, infrastructure, and vital roadways.</li> <li>Removal of dangerous trees and limbs (dead, leaners, and hangers). Prevent blockage or damage to infrastructure and/or major roadways/thoroughfares.</li> </ul> </li> </ul>
BACKGROUND INFORMATION	
Jurisdiction/Location:	Trees near power lines throughout Beaumont
RiskReductionBenefit(CurrentCost/Losses Avoided):	Reduce risk of loss of power and potential damages to structures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Thunderstorm Wind, Winter Storm, Tornado, Lightning, Hail
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$1,000,000
Potential Funding Sources:	HMGP, PDM, local operating budgets, Utility Fees
Lead Agency/Department Responsible:	Beaumont Public Works
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan, Local Ordinance

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Beaumont – Action #44
Proposed Action:	Secure and maintain backup information systems to
	store critical information at off-site locations.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
Risk Reduction Benefit (Current	Reduce loss of critical government data and files
Cost/Losses Avoided):	through redundant systems.
Turne of Action (Local Diana and	Chrysteine and Infrastructure
Iype of Action (Local Plans and Regulations Structure and Infrastructure	Structure and infrastructure
projects Natural System Protection or	
Education and Awareness)	
Luucution una Awareness)	

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood, Tornado, Wildfire, Thunderstorm Wind, Lightning, Hail
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$50,000
Potential Funding Sources:	HMGP, PDM, EMPG, local operating budgets
Lead Agency/Department Responsible:	Beaumont Administration
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	N/A

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)
	Beaumont – Action #45
Proposed Action:	Improve quality of local information on vulnerable items (assets and populations) for the purpose of more accurate risk and damage assessments. Work with other agencies in city to get data as up-to-date and complete as possible.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Improve risk and vulnerability assessment.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood, Tornado, Wildfire
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$5,000
Potential Funding Sources:	HMGP, PDM, EMPG, local operating budgets
Lead Agency/Department Responsible:	Beaumont Administration
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	N/A

# Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Beaumont – Action #46
Proposed Action:	Elevate and/or upgrade Marina Drive in the City of Beaumont.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Marina Drive
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of damages to infrastructure and ensure emergency access.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$6,000,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Beaumont Public Works
Implementation Schedule:	Within 48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan

# Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Beaumont – Action #47
Proposed Action:	Provide educational seminars and brochures regarding the voluntary Community Rating System (CRS).
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of flood losses through CRS education and buy-in.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Education and Awareness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$15,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Beaumont Floodplain Management
Implementation Schedule:	Within 48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

# Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Beaumont – Action #48
Proposed Action:	Expand and upgrade security systems at St. Elizabeth Hospital for access control on all exterior doors for all buildings as well as installing cameras and increased perimeter surveillance and safety capabilities.
BACKGROUND INFORMATION	
Jurisdiction/Location:	St. Elizabeth Hospital
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of terror attack on vulnerable targets.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Terrorism
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$350,000
Potential Funding Sources:	Local operating budgets, EMGP, Homeland Security Grants
Lead Agency/Department Responsible:	Beaumont Emergency Management, St. Elizabeth Hospital
Implementation Schedule:	Within 48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Hospital Operations Plan

# Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

# Bevil Oaks

	Bevil Oaks – Action #1
Proposed Action:	Retrofit existing structures and/or construct new structures to act as residential shelters during and after Hurricanes and other severe weather events.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in Bevil Oaks
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents in Bevil Oaks.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$2,000,000
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM
Lead Agency/Department Responsible:	Bevil Oaks City Administration
Implementation Schedule:	Within 48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

# COMMENTS

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	Storm harden/retrofit critical facilities throughout Bevil Oaks. Actions can include but are not limited to window shutters, roof straps, flood proofing, roll-up door reinforcements (i.e. for fire stations), backup generator power with permanent hook-ups, hail resistant roofing materials, and surge protectors.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Critical facilities in Bevil Oaks
<b>Risk Reduction Benefit</b> (Curr Cost/Losses Avoided):	ent Protect critical facilities from damages and ensure continuity of emergency services.
<b>Type of Action</b> (Local Plans of Regulations, Structure and Infrastruct projects, Natural System Protection, Education and Awareness)	and Structure and Infrastructure ure or

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Bevil Oaks, Public Works
Implementation Schedule:	Within 48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

# Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Bevil Oaks – Action #3
Proposed Action:	Acquire/demolish flood-prone properties with an emphasis on Repetitive and Severe Repetitive Loss properties.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Flood-prone structures in Bevil Oaks
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Eliminate flood damages to flood-prone structures. Reduce burden on emergency services during flood events.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood
Effect on New/Existing Buildings:	Eliminate risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5,000,000
Potential Funding Sources:	HMGP, PDM, FMA
Lead Agency/Department Responsible:	City of Bevil Oaks
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan

The City of Bevil Oaks applied for HMGP funding to acquire 4 repetitive loss structures since the last planning cycle. This project will implement the first phase if/when the project is funded. This project is also an extension of the acquisition program to acquire additional structures as they are identified.

# Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Bevil Oaks – Action #4
Proposed Action:	Elevate existing flood-prone structures and infrastructure throughout Bevil Oaks.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Flood-prone structures and infrastructure in Bevil Oaks.
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce flood damages to flood-prone structures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5,000,000
Potential Funding Sources:	HMGP, PDM, FMA
Lead Agency/Department Responsible:	City of Bevil Oaks
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Local Ordinance, Emergency Management Plan

Bevil Oaks updated floodplain ordinance requires new and substantially damaged structures to be elevated 2' above base flood elevation. This project would proactively elevate existing flood-prone structures above the BFE.

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Bevil Oaks – Action #5
Proposed Action:	Elevate or upgrade bridges, culverts and other crossings throughout Bevil Oaks to reduce damages to infrastructure and reduce flooding caused by undersized crossings and culverts.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Undersized and/or inadequate bridges and culverts in Bevil Oaks
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce flood damages to infrastructure and surrounding structures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing infrastructure and structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$1 Million
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Bevil Oaks Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Bevil Oaks – Action #6
Proposed Action:	Pursue funding and implement drainage improvements throughout Bevil Oaks. Actions can include but are not limited to installing/ upgrading culverts and headwalls as well as enlarging storm water ditches and canals.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout Bevil Oaks
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce flood damages to infrastructure and surrounding structures as a result of undersized drainage.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane
Effect on New/Existing Buildings:	Reduce risk to existing infrastructure and structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Bevil Oaks Works in coordination with Jefferson County and Drainage District 6
Implementation Schedule:	Within 12-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

# Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Bevil Oaks – Action #7
Proposed Action:	Provide generators/back-up power systems with permanent hook-ups for lift stations and other critical facilities lacking back-up power throughout Bevil Oaks.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Lift Stations/Critical Facilities in Bevil Oaks
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Protect critical facilities from damages and ensure continuity of services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$200,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Bevil Oaks, Public Works
Implementation Schedule:	Within 48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

# Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Bevil Oaks – Action #8
Proposed Action:	Implement flood protection measures to protect from surge from Hurricanes/Tropical Storms. Actions can include but are not limited to constructing and/or upgrading sea walls, flood barriers, berms and various wet and dry flood proofing measures.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout Bevil Oaks
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of storm surge flooding to structures and infrastructure.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$3,000,000
Potential Funding Sources:	HMGP, PDM
Lead Agency/Department Responsible:	City of Bevil Oaks in coordination with Jefferson County
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

# Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Bevil Oaks – Action #9	
Proposed Action:	Identify and pursue any mitigation activities that would aid/enhance evacuations throughout the Bevil Oaks.	
BACKGROUND INFORMATION	BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide	
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or fatality through improved evacuation routes and procedures.	
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure	

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Wildfire, Flood
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budget
Lead Agency/Department Responsible:	City of Bevil Oaks in coordination with Jefferson County
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Evacuation Plan

# Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Bevil Oaks – Action #10
Proposed Action:	<ul> <li>Mitigate damage to utilities in order to maintain function during and after a hazard event. Actions can include but are not limited to ordinance requirements to: <ul> <li>Bury utility lines underground</li> <li>Provide frangible links/break away connections on utility poles</li> <li>Harden utility poles by converting from wood to concrete or metal utility poles</li> <li>Increase the easement area/clearance of utility lines/poles from tree lines.</li> </ul> </li> </ul>
BACKGROUND INFORMATION	
Jurisdiction/Location:	Utility lines throughout Bevil Oaks
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of loss of power and ensure continuity of emergency services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind, Winter Storm, Tornado, Lightning, Hail
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Bevil Oaks
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan, Local Ordinance

# Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Education and Awareness)

Proposed Action:	Provide the public with educational brochures for mitigating damages, planning ahead for disasters and reducing the risk of injury during events including: Mitigation measures such as window film, elevated appliances, surge protectors, insulating pipes, drought tolerant landscaping. Education on when to take cover, when to evacuate, locations of local safe rooms, signs of dehydration, and proper storage of flammable materials, or other appropriate materials to mitigate damages and health hazards.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
RiskReductionBenefit(CurrentCost/Losses Avoided):	Reduce risk to residents and structures through education and awareness.

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood, Thunderstorm Wind, Winter Storm, Wildfire, Tornado, Lightning, Extreme Heat, Hail, Drought
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$50,000
Potential Funding Sources:	HMGP, PDM, local operating budgets, EMPG
Lead Agency/Department Responsible:	City of Bevil Oaks
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

# COMMENTS

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Bevil Oaks – Action #12
Proposed Action:	Retrofit existing structures to act as cooling stations in times of extreme heat. Educate public on the locations and availability of cooling centers during times of extreme heat.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations TBD
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk to vulnerable populations during extreme heat events.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure Education and Awareness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Extreme Heat
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Bevil Oaks
Implementation Schedule:	Within 36-48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Bevil Oaks – Action #13	
Proposed Action:	Develop areas of defensible space to prevent damage due to wildfires.	
BACKGROUND INFORMATION	BACKGROUND INFORMATION	
Jurisdiction/Location:	Wildland Urban Interface	
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of damages to structures in or near the WUI.	
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure	

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Wildfire
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Bevil Oaks
Implementation Schedule:	Within 48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Community Wildfire Protection Plan

# Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Bevil Oaks – Action #14
Proposed Action:	Develop and enact water conservation or drought management plans, ordinances or strategies to be used during times of drought.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risks associated with drought through reduction in water usage during times of drought.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Drought
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$25,000 - \$50,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Bevil Oaks
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Local Ordinances

# Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Bevil Oaks – Action #15
Proposed Action:	<ul> <li>Minimize damage to structures and infrastructure from falling trees. Actions include but are not limited to the following: <ul> <li>Pursue and coordinate a tree and limb removal program to protect infrastructure and critical facilities from damage. This includes working with private homeowners for voluntary removal of hazardous trees and limbs on private property.</li> <li>Coordinate contracts to remove and/or trim trees that endanger structures, infrastructure, and vital roadways.</li> <li>Update ordinance to require dead tree removal and tree trimming as appropriate to protect structures and infrastructure.</li> </ul> </li> </ul>
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
RiskReductionBenefit(CurrentCost/Losses Avoided):	Reduce risk of loss of power and potential damages to structure and infrastructure.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Thunderstorm Wind, Winter Storm, Tornado, Lightning, Hail
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Bevil Oaks
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan, Local Ordinance

# Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

# China

	China – Action #1
Proposed Action:	Build a structure or structures (including a dome or domes) in coastal (or near coastal) jurisdictions that can withstand 200 mile per hour winds and act as shelters of last resort. This is in conjunction with the Texas Safe Shelter Initiative.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in China
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in China.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$1.6 Million
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM
Lead Agency/Department Responsible:	City of China, Jefferson County Emergency
	Management
Implementation Schedule:	1-3 years
Incorporation into Existing Plans:	Emergency Management Plan, Jefferson County Plan

# COMMENTS

# Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies) Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 4

	China – Action #2
Proposed Action:	Retrofit existing structures and/or construct new structures to act as shelters during and after Hurricanes/Tropical Storms and other severe weather events.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in China
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in China.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$1 - \$2 Million
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM
Lead Agency/Department Responsible:	City of China, Jefferson County Emergency Management, SETRPC
Implementation Schedule:	1-3 years
Incorporation into Existing Plans:	Emergency Management Plan

# Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	Develop and enact water conservation or drought management plans, ordinances or strategies to be used during times of drought.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in China
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in China.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Drought
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$25,000 - \$50,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of China, Jefferson County
Implementation Schedule:	6-12 months
Incorporation into Existing Plans:	Local Ordinances

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	China – Action #4
Proposed Action:	<ul> <li>Minimize damage to structures and infrastructure from falling trees. Actions include but are not limited to the following: <ul> <li>Implement and coordinate a dangerous tree and limb removal program to protect infrastructure and critical facilities from damage. This includes working with private homeowners for voluntary removal of hazardous trees and limbs on private property.</li> <li>Coordinate contracting to remove and/or trim trees that endanger structures, infrastructure, and vital roadways.</li> </ul> </li> </ul>
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in China
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in China.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$50,000 - \$100,000
Potential Funding Sources:	FEMA PA, HMGP, local operating budgets
Lead Agency/Department Responsible:	City of China, Jefferson County
Implementation Schedule:	6-12 months
Incorporation into Existing Plans:	Comprehensive Plan, Local Ordinance

# Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	China – Action #5 Public structure strengthening for City Hall.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in China
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in China.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$10,000 - \$500,000
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets
Lead Agency/Department Responsible:	City of China, Jefferson County
Implementation Schedule:	1-2 years
Incorporation into Existing Plans:	Emergency Management Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	China – Action #6 Purchase and install generators/back-up power systems for critical facilities (including but not limited to lift stations, water plants, police, EMS, Fire and other first responder facilities) throughout China.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in China
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in China.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$50,000 - \$200,000
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM
Lead Agency/Department Responsible:	City of China, Jefferson County
Implementation Schedule:	1-3 years
Incorporation into Existing Plans:	Emergency Management Plan

# Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	China – Action #7
Proposed Action:	Retrofit existing structures or construct new structures to act as safe rooms during tornados or other severe weather events.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in China
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in China.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Tornado, Thunderstorm Wind
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5,000 - \$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of China, Jefferson County
Implementation Schedule:	1-5 years
Incorporation into Existing Plans:	Emergency Management Plan

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	Identify and implement any mitigation activities that would aid/ enhance evacuations throughout China.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in China
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in China.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Education and Awareness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Wildfire, Winter Storm
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$10,000 - \$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of China, Jefferson County
Implementation Schedule:	1-5 years
Incorporation into Existing Plans:	Emergency Management Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	China – Action #9
Proposed Action:	Construct water retention ponds to collect storm water run-off and use as an alternate water source throughout China.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in China
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in China.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Thunderstorm Wind, Flood, Drought
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$75,000 - \$200,000
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets
Lead Agency/Department Responsible:	City of China, Jefferson County
Implementation Schedule:	1-3 years
Incorporation into Existing Plans:	Drainage Plan

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	China – Action #10
Proposed Action:	<ul> <li>Mitigate damage to utilities in order to maintain function during and after a hazard event. Actions can include but are not limited to: <ul> <li>Burying utility lines underground</li> <li>Provide frangible links/break away connections on utility poles</li> <li>Harden utility poles by converting from wood to concrete or metal utility poles</li> <li>Increasing the easement area/clearance of utility lines/ poles from trees lines</li> </ul> </li> </ul>
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in China
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in China.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Lightning, Hail
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$100,000 - \$200,000
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM
Lead Agency/Department Responsible:	City of China, Jefferson County
Implementation Schedule:	1-5 years
Incorporation into Existing Plans:	Comprehensive Plan, Local Ordinance

# Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	Provide the public with educational brochures for the hazards identified as part of the 2017 Plan Update.
BACKGROUND INFORMATION	• •
Jurisdiction/Location:	Various locations in China
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in China.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Education and Awareness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail, Drought
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5,000 - \$50,000
Potential Funding Sources:	HMGP, PDM, local operating budgets, EMPG
Lead Agency/Department Responsible:	City of China, Jefferson County
Implementation Schedule:	6-12 months
Incorporation into Existing Plans:	Emergency Management Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	China – Action #12 Storm harden/retrofit critical facilities throughout China. Actions can include but are not limited to window shutters, roof straps, flood proofing, and roll-up door reinforcement (i.e. for fire stations).
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in China
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in China.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$10,000 - \$500,000
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets
Lead Agency/Department Responsible:	City of China, Jefferson County
Implementation Schedule:	1-2 years
Incorporation into Existing Plans:	Emergency Management Plan

# Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	China – Action #13 Replace and/or upgrade bridges, culverts and other crossings throughout China.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in China
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in China.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing infrastructure and structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$200,000 - \$1 Million
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of China, Jefferson County
Implementation Schedule:	2-5 years
Incorporation into Existing Plans:	Drainage Plan

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	China – Action #14
Proposed Action:	Implement drainage improvements throughout China. Actions can include but are not limited to installing/upgrading culverts and headwalls as well as enlarging storm water ditches and canals.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in China
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in China.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$50,000 - \$500,000
Potential Funding Sources:	HMGP, PDM, FMA, local operating budgets
Lead Agency/Department Responsible:	City of China, Jefferson County
Implementation Schedule:	1-3 years
Incorporation into Existing Plans:	Drainage Plan

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	China – Action #15
Proposed Action:	Implement flood protection measures to protect from surges from Hurricanes/Tropical Storms. Actions can include but are not limited to constructing and/or upgrading sea walls, flood barriers, berms and various wet and dry flood proofing measures.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in China
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in China.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$50,000 - \$3 Million
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL
Lead Agency/Department Responsible:	City of China, Jefferson County
Implementation Schedule:	1-5 years
Incorporation into Existing Plans:	Master Plan

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	Construct or improve existing detention/retention ponds where appropriate to collect storm water to reduce flooding.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in China
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in China.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Thunderstorm Wind, Flood
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$50,000 - \$100,000
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL, local operating budgets
Lead Agency/Department Responsible:	City of China, Jefferson County
Implementation Schedule:	1-3 years
Incorporation into Existing Plans:	Drainage Plan

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)
Proposed Action:	Retrofit existing structures to act as cooling stations in times of extreme heat.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in China
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in China.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Extreme Heat
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of China, Jefferson County
Implementation Schedule:	1-3 years
Incorporation into Existing Plans:	Emergency Management Plan

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	China – Action #18
Proposed Action:	Acquire flood prone properties (including Repetitive Loss and Severe Repetitive Loss Properties).
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in China
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in China.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood
Effect on New/Existing Buildings:	Eliminate risk to existing structures
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$50,000 - \$140,000
Potential Funding Sources:	HMGP, FMA, PDM, RFC, SRL
Lead Agency/Department Responsible:	City of China, Jefferson County
Implementation Schedule:	12 months
Incorporation into Existing Plans:	Comprehensive Plan

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	China – Action #19
Proposed Action:	Elevate new and existing flood prone structures and infrastructure throughout China.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in China
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in China.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$20,000 - \$100,000
Potential Funding Sources:	HMGP, PDM, FMA, RFC, SRL
Lead Agency/Department Responsible:	City of China, Jefferson County
Implementation Schedule:	1-3 years
Incorporation into Existing Plans:	Local Ordinance, Comprehensive Plan

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	Bury underground, secure or otherwise harden exposed or vulnerable pipelines.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in China
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in China.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$100,000 - \$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of China, Jefferson County
Implementation Schedule:	3-5 years
Incorporation into Existing Plans:	Emergency Management Plan

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

		China – Action #21
Proposed Action:		Coordinate and implement construction, expansion and maintenance of flood control structures/barriers for the purpose of mitigating damage and protect fresh water resources from storm surge, sea level rise, and other sources of salt water intrusion.
BACKGROUND INFORMATION		
Jurisdiction/Location:		Various locations in China
<b>Risk Reduction Benefit</b> <i>Cost/Losses Avoided</i> ):	(Current	Reduce risk of injury or death for residents and first responders in China.
<b>Type of Action</b> (Local P Regulations, Structure and Infr projects, Natural System Prot Education and Awareness)	lans and rastructure rection, or	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane
Effect on New/Existing Buildings:	Reduce risk to existing structures and infrastructure
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$1 - \$2 Million
Potential Funding Sources:	HMGP, PDM, USACE, Jefferson County, SETRPC
Lead Agency/Department Responsible:	City of China, Jefferson County
Implementation Schedule:	2-5 years
Incorporation into Existing Plans:	Drainage Plan

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	Develop areas of defensible space to prevent damage due to wildfires.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in China
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in China.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Wildfire
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$50,000 - \$100,000
Potential Funding Sources:	HMGP, EMPG, PDM, local operating budgets
Lead Agency/Department Responsible:	City of China, Jefferson County
Implementation Schedule:	1-3 years
Incorporation into Existing Plans:	Community Wildfire Protection Plan

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

## Groves

	Groves – Action #1
Proposed Action:	Retrofit existing structures and/or construct new structures to act as residential shelters during and after Hurricanes and other severe weather events.
BACKGROUND INFORMATION	-
Jurisdiction/Location:	Various locations in Groves
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents in Groves.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$2 Million
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM
Lead Agency/Department Responsible:	Groves Emergency Management
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

## COMMENTS

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:		Storm harden/retrofit critical facilities throughout Groves. Actions can include but are not limited to window shutters, roof straps, flood proofing, roll-up door reinforcement (i.e. for fire stations), backup generator power with permanent hook-ups, hail resistant roofing materials, and surge protectors.
BACKGROUND INFORMAT	ION	
Jurisdiction/Location:		Critical facilities in Groves
<b>Risk Reduction Ben</b> Cost/Losses Avoided):	efit (Current	Protect critical facilities from damages and ensure continuity of emergency services.
<b>Type of Action</b> (Loca Regulations, Structure and projects, Natural System Education and Awareness)	nl Plans and d Infrastructure Protection, or	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Groves Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Groves – Action #3
Proposed Action:	Identify and pursue any mitigation activities that would aid/enhance evacuations throughout Groves.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or fatality through improved evacuation routes and procedures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Wildfire, Flood
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Groves Emergency Management
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Groves – Action #4
Proposed Action:	Storm harden/retrofit City Hall, Police Station, and Activity Center Complex. Actions can include but are not limited to window shutters, roof straps, flood proofing, roll-up door reinforcement (i.e. for fire stations), backup generator power with permanent hook-ups, hail resistant roofing materials, and surge protectors.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City Hall, Police Station, and Activity Center Complex
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Protect public facilities from damages and ensure continuity of services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Groves Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	Storm harden/retrofit Groves Fire Station. Actions can include but are not limited to window shutters, roof straps, flood proofing, roll-up door reinforcement (i.e. for fire stations), backup generator power with permanent hook-ups, hail resistant roofing materials, and surge protectors.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Groves Fire Station
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Protect facility and fire trucks from damages and ensure continuity of emergency services. Life Safety benefits for first responders.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Groves Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	Groves – Action #6 Storm harden/retrofit Public Works Complex. Actions can include but are not limited to window shutters, roof straps, flood proofing, roll-up door
	reinforcement (i.e. for fire stations), backup generator power with permanent hook-ups, hail resistant roofing materials, and surge protectors.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Groves Public Works Complex
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Protect public works facilities from damages and ensure continuity of emergency services. Life Safety benefits for first responders.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Groves Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies) Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

Proposed Action:		Groves – Action #7 Storm harden/retrofit Wastewater Treatment Plant. Actions can include but are not limited to window shutters, roof straps, flood proofing, roll-up door reinforcement (i.e. for fire stations), backup generator power with permanent hook-ups, hail resistant roofing materials, and surge protectors.
BACKGROUND INF	ORMATION	
Jurisdiction/Locati	on:	Groves Wastewater Treatment Plant
<b>Risk Reduction</b> Cost/Losses Avoide	<b>Benefit</b> (Current d):	Protect critical facility from damages and ensure continuity of services.
<b>Type of Action</b> Regulations, Struct projects, Natural Education and Awa	(Local Plans and ture and Infrastructure System Protection, or areness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Groves Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Groves – Action #8
Proposed Action:	Provide the public with educational brochures for mitigating damages, planning ahead for disasters and reducing the risk of injury during events including: Mitigation measures such as window film, elevated appliances, surge protectors, insulating pipes, drought tolerant landscaping. Education on when to take cover, when to evacuate, locations of local safe rooms, signs of dehydration, and proper storage of flammable materials, or other appropriate materials to mitigate damages and health hazards.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
RiskReductionBenefit(CurrentCost/Losses Avoided):	Reduce risk to residents and structures through education and awareness.
Type of Action (Local Plans and	Education and Awareness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood, Thunderstorm Wind, Winter Storm, Wildfire, Tornado, Lightning, Extreme Heat, Hail, Drought
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$50,000
Potential Funding Sources:	HMGP, PDM, local operating budgets, EMPG
Lead Agency/Department Responsible:	Groves Emergency Management
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	Groves – Action #9 Elevate or upgrade bridges, culverts and other crossings throughout Groves to reduce damages to infrastructure and reduce flooding caused by undersized crossings and culverts.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Undersized bridges and culverts in Groves
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce flood damages to infrastructure and surrounding structures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing infrastructure and structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$1 Million
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Groves Public Works
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Groves – Action #10
Proposed Action:	Pursue funding and implement drainage improvements throughout Groves. Actions can include but are not limited to installing/upgrading culverts and headwalls as well as enlarging storm water ditches and canals.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout Groves
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce flood damages to infrastructure and surrounding structures as a result of undersized drainage.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane
Effect on New/Existing Buildings:	Reduce risk to existing infrastructure and structures
Priority (High, Moderate, Low): Moderate	
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Groves Public Works
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Groves – Action #11
Proposed Action:	Implement flood protection measures to protect from surge from Hurricanes/Tropical Storms. Actions can include but are not limited to constructing and/or upgrading sea walls, flood barriers, berms and various wet and dry flood proofing measures.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout Groves
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of storm surge flooding to structures and infrastructure.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS		
Hazard(s) Addressed:	Hurricane	
Effect on New/Existing Buildings:	Reduce risk to existing and future structures	
Priority (High, Moderate, Low):	Moderate	
Estimated Cost:	\$3,000,000	
Potential Funding Sources:	HMGP, PDM	
Lead Agency/Department Responsible:	City of Groves Public Works	
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding	
Incorporation into Existing Plans:	Drainage Plan	

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	Coordinate and implement construction, expansion and maintenance of flood control structures/barriers for the purpose of mitigating damage and protect fresh water resources from storm surge, sea level rise and other sources of salt water intrusion.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk to structures and infrastructure through improved flood control measures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$2,000,000
Potential Funding Sources:	HMGP, PDM, USACE, SETRPC
Lead Agency/Department Responsible:	Groves Public Works, SETRPC
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Groves – Action #13
Proposed Action:	Develop and enact water conservation or drought management plans, ordinances or strategies to be used during times of drought.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risks associated with drought through reduction in water usage during times of drought.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Drought
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$25,000 - \$50,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Groves Administration
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Local Ordinances

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Groves – Action #14
Proposed Action:	Construct or upgrade existing detention/retention ponds where appropriate to collect storm water to reduce flooding and for use as an alternate water source throughout Groves.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various Locations TBD
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk to structures and infrastructure through improved flood control measures. Water resource for irrigation use during drought.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Drought
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Groves Public Works
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Groves – Action #15
Proposed Action:	<ul> <li>Mitigate damage to utilities in order to maintain function during and after a hazard event. Actions can include but are not limited to: <ul> <li>Bury utility lines underground</li> <li>Provide frangible links/break away connections on utility poles</li> <li>Harden utility poles by converting from wood to concrete or metal utility poles</li> <li>Increase the easement area/clearance of utility lines/poles from tree lines</li> </ul> </li> </ul>
BACKGROUND INFORMATION	
Jurisdiction/Location:	Utility lines throughout Groves
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of loss of power and ensure continuity of emergency services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind, Winter Storm, Tornado, Lightning, Hail
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$200,000
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM
Lead Agency/Department Responsible:	Groves Public Works, Entergy
Implementation Schedule:	Within 48-60 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan, Local Ordinance

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	Groves – Action #16 Retrofit existing structures to act as cooling stations in times of extreme heat.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations TBD
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk to vulnerable populations during extreme heat events.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Extreme Heat
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Groves Administration
Implementation Schedule:	Within 48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Groves – Action #17
Proposed Action:	Develop areas of defensible space to prevent damage due to wildfires.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Wildland Urban Interface
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of damages to structures in or near the WUI.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Wildfire
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Groves Fire Department
Implementation Schedule:	Within 48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Community Wildfire Protection Plan

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

		Groves – Action #18
	Proposed Action:	<ul> <li>Minimize damage to structures and infrastructure from falling trees. Actions include but are not limited to the following: <ul> <li>Pursue and coordinate a dangerous tree and limb removal program to protect infrastructure and critical facilities from damage. This includes working with private homeowners for voluntary removal of hazardous trees and limbs on private property.</li> <li>Coordinate contracts to remove and/or trim trees that endanger structures, infrastructure, and vital roadways.</li> </ul> </li> </ul>
Ī	BACKGROUND INFORMATION	
	Jurisdiction/Location:	Trees near power lines throughout Groves.
	Risk         Reduction         Benefit         (Current           Cost/Losses Avoided):	Reduce risk of loss of power and potential damages to structures.
	<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Thunderstorm Wind, Winter Storm, Tornado, Lightning, Hail
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Groves Public Works
Implementation Schedule:	Within 48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan, Local Ordinances

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Groves – Action #19
Proposed Action:	Upgrade drainage system to reduce flooding on Van Buren from Wilson to Grant.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Van Buren from Wilson to Grant
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of flood damages to structures and infrastructure due to inadequate drainage.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing structures and infrastructure
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Groves Public Works
Implementation Schedule:	Within 48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Groves – Action #20
Proposed Action:	Upgrade drainage system to reduce flooding issues on 34 <sup>th</sup> Street and the south end of Franklin Street.
BACKGROUND INFORMATION	
Jurisdiction/Location:	34 <sup>th</sup> Street and the south end of Franklin Street
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of flood damages to structures and infrastructure due to inadequate drainage.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing structures and infrastructure
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Groves Public Works
Implementation Schedule:	Within 48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Groves – Action #21
Proposed Action:	Acquire and demolish or relocate flood-prone properties with an emphasis on Repetitive and Severe Repetitive Loss Properties.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Flood-prone structures in Groves
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Eliminate flood damages to flood-prone structures. Reduce burden on emergency services during flood events.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood
Effect on New/Existing Buildings:	Eliminate risk to existing structures
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$5,000,000
Potential Funding Sources:	HMGP, PDM, FMA
Lead Agency/Department Responsible:	Groves Emergency Management
Implementation Schedule:	Within 48-60 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Groves – Action #22
Proposed Action:	Elevate new and existing flood-prone structures and infrastructure throughout Groves.
BACKGROUND INFORMATION	-
Jurisdiction/Location:	Flood-prone structures and infrastructure in Groves
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce flood damages to flood-prone structures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$5,000,000
Potential Funding Sources:	HMGP, PDM, FMA
Lead Agency/Department Responsible:	Groves Emergency Management
Implementation Schedule:	Within 48-60 months of plan adoption pending available funding
Incorporation into Existing Plans:	Local Ordinance, Emergency Management Plan

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Groves – Action #23
Proposed Action:	Secure, bury or otherwise harden exposed or vulnerable pipelines including water, sewer, liquid petroleum, and natural gas.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout Groves
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of pipeline failure through mitigation of exposed or vulnerable lines.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind, Winter Storm, Wildfire, Tornado, Lightning, Extreme Heat, Hail, Pipeline Failure, Hazardous Materials
Effect on New/Existing Buildings:	Reduce risk to existing infrastructure and structures
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Groves Public Works
Implementation Schedule:	Within 48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

# Nederland

	Nederland – Action #1
Proposed Action:	Build a structure or structures (including a dome or domes) in coastal (or near coastal) jurisdictions that can withstand 200 miles per hour winds and act as shelters of last resort. This is in conjunction with the Texas Safe Shelter Initiative.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in Nederland
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in Nederland.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$1.6 Million
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM
Lead Agency/Department Responsible:	City of Nederland Emergency Management
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

COMMENTS

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies) Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 4

Proposed Action:	Nederland – Action #2 Retrofit existing structures and/or construct new structures to act as residential shelters during and after Hurricanes and other severe weather events.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in Nederland
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents in Nederland.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$2 Million
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM
Lead Agency/Department Responsible:	City of Nederland Emergency Management
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	Storm harden/retrofit critical facilities throughout Nederland. Actions can include but are not limited to window shutters, roof straps, flood proofing, roll-up door reinforcement (i.e. for fire stations), backup generator power with permanent hook-ups, hail resistant roofing materials, and surge protectors.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Critical facilities in Nederland
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Protect critical facilities from damages and ensure continuity of emergency services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Nederland Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Nederland – Action #4
Proposed Action:	Acquire and demolish or relocate flood-prone properties with an emphasis on Repetitive and Severe Repetitive Loss properties.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Flood-prone structures in Nederland
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Eliminate flood damages to flood-prone structures. Reduce burden on emergency services during flood events.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood
Effect on New/Existing Buildings:	Eliminate risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5,000,000
Potential Funding Sources:	HMGP, PDM, FMA
Lead Agency/Department Responsible:	City of Nederland Emergency Management
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Nederland – Action #5
Proposed Action:	Elevate new and existing flood-prone structures and infrastructure throughout Nederland.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Flood-prone structures and infrastructure in Nederland
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce flood damages to flood-prone structures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5,000,000
Potential Funding Sources:	HMGP, PDM, FMA
Lead Agency/Department Responsible:	City of Nederland Emergency Management
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Local Ordinance, Emergency Management Plan

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Nederland – Action #6
Proposed Action:	Elevate or upgrade bridges, culverts and other crossings throughout Nederland to reduce damages to infrastructure and reduce flooding caused by undersized crossings and culverts.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Undersized bridges and culverts in Nederland
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce flood damages to infrastructure and surrounding structures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing infrastructure and structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$1 Million
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Nederland Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Nederland – Action #7
Proposed Action:	Pursue funding and implement drainage improvements throughout Nederland. Actions can include but are not limited to installing/ upgrading culverts and headwalls as well as enlarging storm water ditches and canals.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout Nederland
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce flood damages to infrastructure and surrounding structures as a result of undersized drainage.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane
Effect on New/Existing Buildings:	Reduce risk to existing infrastructure and structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Nederland Public Works
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)
	Nederland – Action #8
Proposed Action:	Secure, bury or otherwise harden exposed or vulnerable pipelines including water, sewer, liquid petroleum, and natural gas.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout Nederland
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of pipeline failure through mitigation of exposed or vulnerable lines.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind, Winter Storm, Wildfire, Tornado, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing infrastructure and structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Nederland Public Works
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

	Nederland – Action #9
Proposed Action:	Implement flood protection measures to protect from surge from Hurricanes/Tropical Storms. Actions can include but are not limited to constructing and/or upgrading sea walls, flood barriers, berms and various wet and dry flood proofing measures.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout Nederland
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of storm surge flooding to structures and infrastructure.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$3,000,000
Potential Funding Sources:	HMGP, PDM
Lead Agency/Department Responsible:	City of Nederland Public Works
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Master Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 4; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

	Nederland – Action #10
Proposed Action:	Coordinate and implement construction, expansion and maintenance of flood control structures/barriers for the purpose of mitigating damage and protect fresh water resources from storm surge, sea level rise and other sources of salt water intrusion.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk to structures and infrastructure through improved flood control measures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$2,000,000
Potential Funding Sources:	HMGP, PDM, USACE, SETRPC
Lead Agency/Department Responsible:	City of Nederland Public Works, SETRPC
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 4; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

	Nederland – Action #11
Proposed Action:	Construct or upgrade existing detention/ retention ponds where appropriate to collect storm water to reduce flooding.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various Locations TBD
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk to structures and infrastructure through improved flood control measures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Nederland Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

	Nederland – Action #12
Proposed Action:	Retrofit existing structures or construct new structures to act as residential safe rooms during tornados or other severe weather events.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various Locations TBD
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5,000 per site
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Nederland Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan, Local Ordinance

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Nederland – Action #13
Proposed Action:	Identify and pursue any mitigation activities that would aid/enhance evacuations throughout Nederland.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or fatality through improved evacuation routes and procedures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Wildfire, Flood
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Nederland Emergency Management
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5

	Nederland – Action #14
Proposed Action:	Construct water retention ponds to collect storm water run-off, reduce flooding and use as an alternate water source throughout Nederland.
BACKGROUND INFORMATION	
Jurisdiction/Location:	TBD site in Nederland
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Lifesaving water alternate source and localized flood reduction.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Drought, Flood, Thunderstorm Wind, Hurricane
Effect on New/Existing Buildings:	Reduce damages to existing and new structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$200,000
Potential Funding Sources:	HMGP, PDM, SETRPC
Lead Agency/Department Responsible:	City of Nederland Public Works
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Master Drainage Plan, Water Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

	Nederland – Action #15
Proposed Action:	<ul> <li>Mitigate damage to utilities in order to maintain function during and after a hazard event. Actions can include but are not limited to: <ul> <li>Bury utility lines underground</li> <li>Provide frangible links/break away connections on utility poles</li> <li>Harden utility poles by converting from wood to concrete or metal utility poles</li> <li>Increase the easement area/clearance of utility lines/poles from tree lines</li> </ul> </li> </ul>
BACKGROUND INFORMATION	
Jurisdiction/Location:	Utility lines throughout Nederland
RiskReductionBenefit(CurrentCost/Losses Avoided):	Reduce risk of loss of power and ensure continuity of emergency services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind, Winter Storm, Tornado, Lightning, Hail
Effect on New/Existing Buildings:	Reduce risk to existing and new structure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$200,000
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM
Lead Agency/Department Responsible:	City of Nederland Public Works, Entergy
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan, Local Ordinance

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Nederland – Action #16
Proposed Action:	Storm harden/retrofit Nederland Water Treatment Plant. Actions can include but are not limited to window shutters, roof straps, flood proofing, roll-up door reinforcement (i.e. for fire stations), backup generator power with permanent hookups, hail resistant roofing materials, and surge protectors.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Nederland Water Treatment Plant
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Protect critical facility from damages and ensure continuity of services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Nederland Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Nederland – Action #17	
Proposed Action:	Storm harden/retrofit Nederland Service Center. Actions can include but are not limited to window shutters, roof straps, flood proofing, roll-up door reinforcement (i.e. for fire stations), backup generator power with permanent hookups, hail resistant roofing materials, and surge protectors.	
BACKGROUND INFORMATION	BACKGROUND INFORMATION	
Jurisdiction/Location:	Nederland Service Center	
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Protect critical facility from damages and ensure continuity of services.	
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure	

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Nederland Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

### Additional Considerations:

Proposed Action:	Storm harden/retrofit Wastewater Treatment Plant. Actions can include but are not limited to window shutters, roof straps, flood proofing, roll-up door reinforcement (i.e. for fire stations), backup generator power with permanent hookups, hail resistant roofing materials, and surge protectors.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Wastewater Treatment Plant at 515 Hardy Avenue
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Protect critical facility from damages and ensure continuity of services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Nederland Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

### Additional Considerations:

Proposed Action:	Storm harden/retrofit Hughes Library. Actions can include but are not limited to window shutters, roof straps, flood proofing, roll-up door reinforcement (i.e. for fire stations), backup generator power with permanent hookups, hail resistant roofing materials, and surge protectors.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Hughes Library
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Protect critical facility from damages and ensure continuity of services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Nederland Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

### Additional Considerations:

	Nederland – Action #20
Proposed Action:	Provide the public with educational brochures for mitigating damages, planning ahead for disasters and reducing the risk of injury during events including: Mitigation measures such as window film, elevated appliances, surge protectors, insulating pipes, drought tolerant landscaping. Education on when to take cover, when to evacuate, locations of local safe rooms, signs of dehydration, and proper storage of flammable materials, or other appropriate materials to mitigate damages and health hazards.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
RiskReductionBenefit(CurrentCost/Losses Avoided):	Reduce risk to residents and structures through education and awareness.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects Natural System Protection, or	Education and Awareness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood, Thunderstorm Wind, Winter Storm, Wildfire, Tornado, Lightning, Extreme Heat, Hail, Drought
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$50,000
Potential Funding Sources:	HMGP, PDM, local operating budgets, EMPG
Lead Agency/Department Responsible:	City of Nederland Emergency Management
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

## Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5

	Nederland – Action #21
Proposed Action:	Retrofit existing structures to act as cooling stations in times of extreme heat.
BACKGROUND INFORMATION	•
Jurisdiction/Location:	Various locations TBD
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk to vulnerable populations during extreme heat events.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Extreme Heat
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Nederland Administration
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Nederland – Action #22
Proposed Action:	Develop areas of defensible space to prevent damage due to wildfires
BACKGROUND INFORMATION	
Jurisdiction/Location:	Wildland Urban Interface
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of damages to structures in or near the WUI.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Wildfire
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Nederland Fire Department
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Community Wildfire Protection Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Nederland – Action #23
Proposed Action:	Develop and enact water conservation or drought management plans, ordinances or strategies to be used during times of drought.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risks associated with drought through reduction in water usage during times of drought.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Drought
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$25,00 - \$50,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Nederland Administration
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Local Ordinances

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 4; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5

		Nederland – Action #24
	Proposed Action:	<ul> <li>Minimize damage to structures and infrastructure from falling trees. Actions include but are not limited to the following: <ul> <li>Pursue and coordinate a dangerous tree and limb removal program to protect infrastructure and critical facilities from damage. This includes working with private homeowners for voluntary removal of hazardous trees and limbs on private property.</li> <li>Coordinate contracts to remove and/or trim trees that endanger structures, infrastructure, and vital roadways.</li> </ul> </li> </ul>
Ī	BACKGROUND INFORMATION	
	Jurisdiction/Location:	Trees near power lines throughout Nederland
	RiskReductionBenefit(CurrentCost/Losses Avoided):	Reduce risk of loss of power and potential damages to structures.
	<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Thunderstorm Wind, Winter Storm, Tornado, Lightning, Hail
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Nederland Public Works
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan, Local Ordinance

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

	Nederland – Action #25
Proposed Action:	Public structure strengthening by replacing drainage tile main feeders.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Drainage main and feeders
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of flood damages to structures and infrastructure due to inadequate drainage.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing and new structures and infrastructure
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Nederland Public Works
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Master Drainage Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

Proposed Action:	Storm harden/retrofit D. Bob Henson Building. Actions can include but are not limited to window shutters, roof straps, flood proofing, roll-up door reinforcement (i.e. for fire stations), backup generator power with permanent hookups, hail resistant roofing materials, and surge protectors.
BACKGROUND INFORMATION	
Jurisdiction/Location:	D. Bob Henson Building
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Protect critical facility from damages and ensure continuity of services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structure
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Nederland Public Works
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

### Additional Considerations:

	Nederland – Action #27
Proposed Action:	Improve underground storm sewer culvert size on Detroit Avenue.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Detroit Avenue
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of flood damages to structures and infrastructure due to inadequate drainage.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing and new structures and infrastructure
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$3,000,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Nederland Public Works
Implementation Schedule:	Within 48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Master Drainage Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

# Nome

Proposed Action:	Nome – Action #1 Build a structure or structures (including a dome or domes) in coastal (or near coastal) jurisdictions that can withstand 200 mile per hour winds and act as shelters of last resort. This is in conjunction with the Texas Safe Shelter Initiative.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in Nome
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in Nome.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$1.6 Million
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM
Lead Agency/Department Responsible:	City of Nome
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan, Jefferson County Plan

# COMMENTS

### Additional Considerations:

Proposed Action:	Nome – Action #2 Retrofit existing structures and/or construct new structures to act as residential shelters during and after Hurricanes and other severe weather events.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in Nome
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents in Nome.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$2 Million
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM
Lead Agency/Department Responsible:	City of Nome
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Nome – Action #3
Proposed Action:	Storm harden/retrofit critical facilities throughout Nome. Actions can include but are not limited to window shutters, roof straps, flood proofing, roll-up door reinforcement (i.e. for fire stations), backup generator power with permanent hook-ups, hail resistant roofing materials, and surge protectors.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Critical facilities in Nome
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Protect critical facilities from damages and ensure continuity of emergency services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$3,000,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Nome Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

### Additional Considerations:

	Nome – Action #4
Proposed Action:	Acquire/demolish flood-prone properties with an emphasis on Repetitive and Severe Repetitive Loss properties.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Flood-prone structures in Nome
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Eliminate flood damages to flood-prone structures. Reduce burden on emergency services during flood events.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood
Effect on New/Existing Buildings:	Eliminate risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5,000,000
Potential Funding Sources:	HMGP, PDM, FMA
Lead Agency/Department Responsible:	City of Nome
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable =3 ; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 4; Legal = 4; Economically Sound = 4; and Environmentally Sound = 5

Proposed Action:	Nome – Action #5 Elevate new and existing flood-prone structures and infrastructure throughout Nome.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Flood-prone structures in Nome
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce flood damages to flood-prone structures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5,000,000
Potential Funding Sources:	HMGP, PDM, FMA
Lead Agency/Department Responsible:	City of Nome
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Local Ordinance, Comprehensive Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

	Nome – Action #6
Proposed Action:	Elevate or upgrade bridges, culverts and other crossings throughout Nome to reduce damages to infrastructure and reduce flooding caused by undersized crossings and culverts.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Undersized bridges and culverts in Nome
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce flood damages to infrastructure and surrounding structures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing infrastructure and structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$10,000,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Nome
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 4

Proposed Action:	Pursue funding and implement drainage improvements throughout Nome. Actions can include but are not limited to installing/upgrading culverts and headwalls as well as enlarging storm water ditches and canals.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout Nome
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce flood damages to infrastructure and surrounding structures as a result of undersized drainage.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane
Effect on New/Existing Buildings:	Reduce risk to existing infrastructure and structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Nome
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

	Nome – Action #8
Proposed Action:	Secure, bury, or otherwise harden exposed or vulnerable pipelines including water and sewer.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout Nome
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of pipeline failure through mitigation of exposed or vulnerable lines.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind, Winter Storm, Wildfire, Tornado, Lightning, Extreme Heat, Hail, Hazardous Material, Terrorism
Effect on New/Existing Buildings:	Reduce risk to existing infrastructure and structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Nome
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

	Nome – Action #9
Proposed Action:	Purchase and install backup generator power systems with permanent hook-ups for critical facilities including lift stations, water plants, police stations, EMS, fire stations, and other first responder facilities throughout Nome.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Critical facilities as indicated above in Nome
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Protect critical facilities from loss of power and ensure continuity of emergency services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$1-\$2 Million
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM
Lead Agency/Department Responsible:	City of Nome
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5

	Nome – Action #10
Proposed Action:	Implement flood protection measures to protect from surge from Hurricanes/Tropical Storms. Actions can include but are not limited to constructing and/or upgrading sea walls, flood barriers, berms and various wet and dry flood proofing measures.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout Nome
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of storm surge damages to structures and infrastructure.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$3 Million
Potential Funding Sources:	HMGP, PDM
Lead Agency/Department Responsible:	City of Nome
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Master Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 4; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

	Nome – Action #11
Proposed Action:	Coordinate and implement construction, expansion and maintenance of flood control structures/barriers for the purpose of mitigating damage and protect fresh water resources from storm surge, sea level rise, and other sources of salt water intrusion.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout Nome
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk to structures and infrastructure through improved flood control measures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane
Effect on New/Existing Buildings:	Reduce risk to existing structures and infrastructure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$1-\$2 Million
Potential Funding Sources:	HMGP, PDM, USACE, SETRPC
Lead Agency/Department Responsible:	City of Nome
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Master Drainage Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 4; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

Proposed Action:	Nome – Action #12 Construct or upgrade existing detention/ retention ponds where appropriate to collect storm water to reduce flooding.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various Locations TBD throughout Nome
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk to structures and infrastructure through improved flood control measures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Thunderstorm Wind, Hurricane
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Nome
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 4

	Nome – Action #13
Proposed Action:	Retrofit existing structures and/or construct new structures to act as residential shelters during and after Hurricanes and other severe weather events.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations TBD
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5,000 - \$100,000 per site
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM
Lead Agency/Department Responsible:	City of Nome
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Nome – Action #14
Proposed Action:	Identify and pursue any mitigation activities that would aid/enhance evacuations throughout Nome.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Life safety benefits through preparedness.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Education and Awareness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood, Winter Storm, Tornado, Thunderstorm Wind, Wildfire
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budget
Lead Agency/Department Responsible:	City of Nome in coordination with Jefferson County
Implementation Schedule:	Within 12-60 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5

Proposed Action:	Nome – Action #15 Construct water retention ponds to collect storm water run-off and use as an alternate water source throughout Nome.
BACKGROUND INFORMATION	
Jurisdiction/Location:	TBD site in Nome
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Lifesaving water alternate source and localized flood reduction.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Drought, Flood, Thunderstorm Wind, Hurricane
Effect on New/Existing Buildings:	Reduce damages to existing and new structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$200,000
Potential Funding Sources:	HMGP, PDM
Lead Agency/Department Responsible:	City of Nome
Implementation Schedule:	Within 12-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Master Drainage Plan, Water Plan

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Nome – Action #16
Proposed Action:	<ul> <li>Mitigate damage to utilities in order to maintain function during and after a hazard event. Actions can include but are not limited to: <ul> <li>Bury utility lines underground</li> <li>Provide frangible links/break away connections on utility poles</li> <li>Harden utility poles by converting from wood to concrete or metal utility poles</li> <li>Increase the easement area/clearance of utility lines/poles from tree lines</li> </ul> </li> </ul>
BACKGROUND INFORMATION	
Jurisdiction/Location:	Utility lines throughout Nome
RiskReductionBenefit(CurrentCost/Losses Avoided):	Reduce risk of loss of power and ensure continuity of emergency services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind, Winter Storm, Tornado, Lightning, Hail
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$20,000,000
Potential Funding Sources:	Utility Fees, Federal Grants
Lead Agency/Department Responsible:	City of Nome, Entergy
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan, Local Ordinance

### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5
	Nome – Action #17
Proposed Action:	Provide the public with educational brochures for mitigating damages, planning ahead for disasters and reduce the risk of injury during events including: mitigation measures such as window film, elevated appliances, surge protectors, insulating pipes, drought tolerant landscaping. Education on when to take cover, when to evacuate, locations of local safe rooms, signs of dehydration, and proper storage of flammable materials, or other appropriate materials to mitigate damages and health hazards.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
RiskReductionBenefit(CurrentCost/Losses Avoided):	Reduce risk to residents and structures through education and awareness.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Education and Awareness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood, Thunderstorm Wind, Winter Storm, Wildfire, Tornado, Lightning, Extreme Heat, Hail, Drought
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$50,000
Potential Funding Sources:	HMGP, PDM, local operating budgets, EMGP
Lead Agency/Department Responsible:	County and City Emergency Management
Implementation Schedule:	Within 24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	Nome – Action #18 Retrofit existing structures to act as cooling stations in times of extreme heat.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations (TBD) in Nome
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk to vulnerable populations during extreme heat events.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Extreme Heat
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Nome
Implementation Schedule:	Within 24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	Nome – Action #19 Develop areas of defensible space to prevent damage due to wildfires.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Wildland Urban Interface of Nome
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of damages to structures in or near the WUI.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Wildfire
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Nome
Implementation Schedule:	Within 48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Community Wildfire Protection Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Nome – Action #20
Proposed Action:	Develop and enact water conservation or drought management plans, ordinances or strategies to be used during times of drought.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risks associated with drought through reduction in water usage during times of drought.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Drought
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$25,000 - \$50,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Nome
Implementation Schedule:	Within 24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Local Ordinances

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Nome – Action #21
Proposed Action:	<ul> <li>Minimize damage to structures and infrastructure from falling trees. Actions include but are not limited to the following: <ul> <li>Pursue and coordinate a dangerous tree and limb removal program to protect infrastructure and critical facilities from damage. This includes working with private homeowners for voluntary removal of hazardous trees and limbs on private property.</li> <li>Coordinate contracts to remove and/or trim trees that endanger structures, infrastructure, and vital roadways.</li> </ul> </li> </ul>
BACKGROUND INFORMATION	
Jurisdiction/Location:	Trees near power lines throughout Nome
RiskReductionBenefit(CurrentCost/Losses Avoided):	Reduce risk of loss of power and potential damages to structures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Thunderstorm Wind, Winter Storm, Tornado, Lightning, Hail
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Nome
Implementation Schedule:	Within 48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan, Local Ordinance

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

# Port Arthur

	Port Arthur – Action #1
Proposed Action:	Build a structure or structures (including a dome or domes) in coastal (or near coastal) jurisdictions that can withstand 200 mile per hour winds and act as shelters of last resort. This is in conjunction with the Texas Safe Shelter Initiative.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in Port Arthur
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in Port Arthur.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$1.6 Million
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM
Lead Agency/Department Responsible:	City of Port Arthur
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan, Jefferson County Plan

#### COMMENTS

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies) Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 4

	Port Arthur – Action #2
Proposed Action:	Retrofit existing structures and/or construct new structures to act as residential shelters during and after Hurricanes and other severe weather events.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in Port Arthur
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents in Port Arthur.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$2 Million
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM
Lead Agency/Department Responsible:	City of Port Arthur
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	Storm harden/retrofit critical facilities throughout Port Arthur. Actions can include but are not limited to window shutters, roof straps, flood proofing, roll-up door reinforcement (i.e. for fire stations), backup generator power with permanent hookups, hail resistant roofing materials, and surge protectors.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Critical facilities in Port Arthur
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Protect critical facilities from damages and ensure continuity of emergency services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Port Arthur Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Port Arthur – Action #4
Proposed Action:	Acquire/demolish flood-prone properties with an emphasis on Repetitive and Severe Repetitive Loss properties.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Flood-prone structures in Port Arthur
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Eliminate flood damages to flood-prone structures. Reduce burden on emergency services during flood events.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood, Thunderstorm Wind
Effect on New/Existing Buildings:	Eliminate risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5,000,000
Potential Funding Sources:	HMGP, PDM, FMA
Lead Agency/Department Responsible:	City of Port Arthur
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Port Arthur – Action #5
Proposed Action:	Elevate new and existing flood-prone structures and infrastructure throughout Port Arthur.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Flood-prone structures and infrastructure in Port Arthur
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce flood damages to flood-prone structures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5,000,000
Potential Funding Sources:	HMGP, PDM, FMA
Lead Agency/Department Responsible:	City of Port Arthur
Implementation Schedule:	Within 12-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Local Ordinance, Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Port Arthur – Action #6
Proposed Action:	Elevate or upgrade bridges, culverts and other crossings throughout Port Arthur to reduce damages to infrastructure and reduce flooding caused by undersized crossings and culverts.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Undersized bridges and culverts in Port Arthur
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce flood damages to infrastructure and surrounding structures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing infrastructure and structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$1 Million
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Port Arthur Public Works
Implementation Schedule:	Within 12-60 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	Port Arthur – Action #7 Pursue funding and implement drainage improvements throughout Port Arthur. Actions can include but are not limited to installing/ upgrading culverts and headwalls as well as enlarging storm water ditches and canals.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout Port Arthur
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce flood damages to infrastructure and surrounding structures as a result of undersized drainage.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind
Effect on New/Existing Buildings:	Reduce risk to existing infrastructure and structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Port Arthur Public Works
Implementation Schedule:	Within 12-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Port Arthur – Action #8
Proposed Action:	Secure, bury or otherwise harden exposed or vulnerable pipelines.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout Port Arthur
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of pipeline failure through mitigation of exposed lines.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind, Winter Storm, Wildfire, Tornado, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing infrastructure and structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Port Arthur Public Works
Implementation Schedule:	Within 12-60 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Port Arthur – Action #9
Proposed Action:	Implement flood protection measures to protect from surge from Hurricanes/Tropical Storms. Actions can include but are not limited to constructing and/or upgrading sea walls, flood barriers, berms and various wet and dry flood proofing measures.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout Port Arthur
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of storm surge flooding to structures and infrastructure
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Thunderstorm Wind, Flood
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$3,000,000
Potential Funding Sources:	HMGP, PDM,
Lead Agency/Department Responsible:	City of Port Arthur Public Works
Implementation Schedule:	Within 12-60 months of plan adoption pending available funding
Incorporation into Existing Plans:	Master Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Port Arthur – Action #10
Proposed Action:	Coordinate and implement construction, expansion and maintenance of flood control structures/barriers for the purpose of mitigating damage and protect fresh water resources from storm surge, sea level rise, and other sources of salt water intrusion.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk to structures and infrastructure through improved flood control measures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$2,000,000
Potential Funding Sources:	HMGP, PDM, USACE, SETRPC
Lead Agency/Department Responsible:	City of Port Arthur Public Works, SETRPC
Implementation Schedule:	Within 12-60 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Port Arthur – Action #11
Proposed Action:	Construct or upgrade existing detention/ retention ponds where appropriate to collect storm water to reduce flooding.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations TBD
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk to structures and infrastructure through improved flood control measures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budget
Lead Agency/Department Responsible:	City of Port Arthur Public Works
Implementation Schedule:	Within 12-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Port Arthur – Action #12
Proposed Action:	Retrofit existing structures or construct new structures to act as residential safe rooms during tornados or other severe weather events.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various Locations TBD
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5,000 - \$100,000 per site
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Port Arthur Public Works
Implementation Schedule:	Within 12-60 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan, Local Ordinance

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Port Arthur – Action #13
Proposed Action:	Identify and pursue any mitigation activities that would aid/enhance evacuations throughout Port Arthur.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or fatality through improved evacuation routes and procedures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Wildfire, Flood
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Port Arthur in coordination with Jefferson County
Implementation Schedule:	Within 12-60 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Port Arthur – Action #14	
Proposed Action:	Construct water retention ponds to collect storm water run-off, reduce flooding and use as an alternate water source throughout Port Arthur.	
BACKGROUND INFORMATION	BACKGROUND INFORMATION	
Jurisdiction/Location:	TBD site in Port Arthur	
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Lifesaving water alternate source and localized flood reduction.	
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure	

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Drought, Flood, Thunderstorm Wind, Hurricane
Effect on New/Existing Buildings:	Reduce damages to existing and new structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$200,000
Potential Funding Sources:	HMGP, PDM, SETRPC
Lead Agency/Department Responsible:	Port Arthur Public Works
Implementation Schedule:	Within 12-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Master Drainage Plan, Water Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Port Arthur – Action #15
Proposed Action:	<ul> <li>Mitigate damage to utilities in order to maintain function during and after a hazard event. Actions can include but are not limited to: <ul> <li>Bury utility lines underground</li> <li>Provide frangible links/break away connections on utility poles</li> <li>Harden utility poles by converting from wood to concrete or metal utility poles</li> <li>Increase the easement area/clearance of utility lines/poles from tree lines</li> </ul> </li> </ul>
BACKGROUND INFORMATION	
Jurisdiction/Location:	Utility lines throughout Port Arthur
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of loss of power and ensure continuity of emergency services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind, Winter Storm, Tornado, Lightning, Hail
Effect on New/Existing Buildings:	Reduce risk to existing and new structure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$200,000
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM
Lead Agency/Department Responsible:	City of Port Arthur Public Works, Entergy
Implementation Schedule:	Within 12-60 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan, Local Ordinance

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Education and Awareness)

Proposed Action:	Provide the public with educational brochures for mitigating damages, planning ahead for disasters and reducing the risk of injury during events including: Mitigation measures such as window film, elevated appliances, surge protectors, insulating pipes, drought tolerant landscaping. Education on when to take cover, when to evacuate, locations of local safe rooms, signs of dehydration, and proper storage of flammable materials, or other appropriate materials to mitigate damages and health hazards.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
RiskReductionBenefit(CurrentCost/Losses Avoided):	Reduce risk to residents and structures through education and awareness.

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood, Thunderstorm Wind, Winter Storm, Wildfire, Tornado, Lightning, Extreme Heat, Hail, Drought
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$50,000
Potential Funding Sources:	HMGP, PDM, local operating budgets, EMPG
Lead Agency/Department Responsible:	City of Port Arthur
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

## COMMENTS

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Port Arthur – Action #17
Proposed Action:	Retrofit existing structures to act as cooling stations in
	times of extreme heat.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations TBD
Risk Reduction Benefit (Current	Reduce risk to vulnerable populations during extreme
Cost/Losses Avoided):	heat events.
Type of Action (Local Plans and	Structure and Infrastructure
Regulations, Structure and Infrastructure	
projects, Natural System Protection, or	
Education and Awareness)	

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Extreme Heat
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Port Arthur
Implementation Schedule:	Within 12-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Port Arthur – Action #18
Proposed Action:	Develop areas of defensible space to prevent damage due to wildfires.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Wildland Urban Interface
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of damages to structures in or near the WUI.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Wildfire
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Port Arthur Fire Department
Implementation Schedule:	Within 12-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Community Wildfire Protection Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Port Arthur – Action #19
Proposed Action:	Develop and enact water conservation or drought management plans, ordinances or strategies to be used during times of drought.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risks associated with drought through reduction in water usage during times of drought.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Drought
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$25,000 - \$50,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Port Arthur
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Local Ordinances

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Port Arthur – Action #20
Proposed Action:	<ul> <li>Minimize damage to structures and infrastructure from falling trees. Actions include but are not limited to the following: <ul> <li>Pursue and coordinate a dangerous tree and limb removal program to protect infrastructure and critical facilities from damage. This includes working with private homeowners for voluntary removal of hazardous trees and limbs on private property.</li> <li>Coordinate contracts to remove and/or trim trees that endanger structures, infrastructure, and vital roadways.</li> </ul> </li> </ul>
BACKGROUND INFORMATION	
Jurisdiction/Location:	Trees near power lines throughout Port Arthur
RiskReductionBenefit(CurrentCost/Losses Avoided):	Reduce risk of loss of power and potential damages to structures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Thunderstorm Wind, Winter Storm, Tornado, Lightning, Hail
Effect on New/Existing Buildings:	Reduce risk to existing and new structure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Port Arthur Public Works
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan, Local Ordinance

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Port Arthur – Action #21
Proposed Action:	Increase channel capacity and improve multiple culvert crossings of Drainage Channel Main B.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Drainage Channel Main B
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of flood damages to structures and infrastructure due to inadequate channel capacity and undersized culverts.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind
Effect on New/Existing Buildings:	Reduce risk to existing and new structures and infrastructure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Port Arthur Public Works
Implementation Schedule:	Within 12-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Master Drainage Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	Port Arthur – Action #22 Improve channel capacity and crossings in Lateral 3 of the Drainage Channel Main A system.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Drainage Channel Main A
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of flood damages to structures and infrastructure due to inadequate channel capacity and undersized crossings.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Port Arthur Public Works, and Drainage District #7
Implementation Schedule:	Within 12-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Master Drainage Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Port Arthur – Action #23
Proposed Action:	Improve culvert crossings in the Lakeview Drainage system.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Culvert crossings in Lakeview Drainage system
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of flood damages to structures and infrastructure due to inadequate undersized culvert crossings.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Port Arthur Public Works
Implementation Schedule:	Within 12-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Master Drainage Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Port Arthur – Action #24
Proposed Action:	Increase drainage capacity to reduce flooding on Westside.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Port Arthur Westside
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of flood damages to structures and infrastructure due to inadequate drainage capacity.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Port Arthur Public Works
Implementation Schedule:	Within 12-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Master Drainage Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Port Arthur – Action #25
Proposed Action:	Upgrade concrete lining of the El Vista Pump Station.
BACKGROUND INFORMATION	
Jurisdiction/Location:	El Vista Pump Station
Risk Reduction Benefit (Current	Reduce risk of flood damages to pump station and
Cost/Losses Avoided):	channel.
Type of Action (Local Plans and	Structure and Infrastructure
Regulations, Structure and Infrastructure	
projects, Natural System Protection, or	
Education and Awareness)	

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind
Effect on New/Existing Buildings:	Reduce risk to existing structures and infrastructure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Port Arthur Public Works
Implementation Schedule:	Within 12-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Master Drainage Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Port Arthur – Action #26
Proposed Action:	Upgrade existing storm sewer in the Port Acres area and along Procter Street.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Port Acres area and along Procter Street
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of flood damages to structures and infrastructure due to inadequate storm sewer capacity.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind
Effect on New/Existing Buildings:	Reduce risk to existing structures and infrastructure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Port Arthur Public Works
Implementation Schedule:	Within 12-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Master Drainage Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Port Arthur – Action #27
Develop/implement shelter-in-place presentations.
City-wide
Reduce risk of death or injury to residents through
education.
Education and Awaronass
Education and Awareness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind, Lightning, Hail
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Port Arthur Emergency Management
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Response Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Port Arthur – Action #28
Proposed Action:	Develop/implement emergency first responder teams with Sabine Neches Chief's Association.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Regional
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of damages, death or injury through improved emergency response.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations – Emergency Response

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind, Lightning, Hail, Winter Storm, Tornado
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$10,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Port Arthur Emergency Management
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Response Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Port Arthur – Action #29
Proposed Action:	Develop/implement coastal storm presentations to public, groups, schools, etc. Educate residents on risk mitigation techniques, early mitigation strategies, emergency kits, evacuation routes, and other appropriate materials.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of damages and death or injury to residents through education and awareness.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Education and Awareness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Port Arthur Emergency Management
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Port Arthur – Action #30
Proposed Action:	Adopt additional freeboard in the local floodplain ordinance. Require all new construction to meet/exceed minimum established flood elevations.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of damages to new structures through higher elevation requirements.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane
Effect on New/Existing Buildings:	Reduce risk to new structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$30,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Port Arthur Floodplain Management
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Floodplain Ordinance

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	Port Arthur – Action #31
Proposed Action:	Receive maximum credit for the NFIP CRS.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
RiskReductionBenefit(CurrentCost/Losses Avoided):	Reduce risk of damages to new structures through higher construction standards and restrictions on floodplain development.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane
Effect on New/Existing Buildings:	Reduce risk to new structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$30,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Port Arthur Floodplain Management
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Floodplain Ordinance

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)
	Port Arthur – Action #32
Proposed Action:	Develop/implement bus transportation for hurricane evacuations.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk to residents through coordinated evacuation assistance.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations – Emergency Response

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$300,000
Potential Funding Sources:	Local operating budgets
Lead Agency/Department Responsible:	City of Port Arthur Emergency Management
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Response Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Port Arthur – Action #33
Proposed Action:	Develop/implement transportation plan for special needs populations.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk to special needs populations through coordinated evacuation assistance.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations – Emergency Response

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Tornado, Winter Storm
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$300,000
Potential Funding Sources:	Local operating budgets
Lead Agency/Department Responsible:	City of Port Arthur Emergency Management
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Response Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Port Arthur – Action #34
Proposed Action:	Assist in implementation of 211 TX Linkage Access Service.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk to residents through Texas Health and Services information access.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations – Preparedness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$50,000 - \$500,000
Potential Funding Sources:	Local operating budgets, HMGP, PDM
Lead Agency/Department Responsible:	City of Port Arthur Emergency Management
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

Proposed Action:	Port Arthur – Action #35 Improve storm water runoff by installing upgraded 8- foot by 7-foot concrete box in Tiger Bayou.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Tiger Bayou
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of flooding through improved drainage.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind
Effect on New/Existing Buildings:	Reduce risk to existing structures and infrastructure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Port Arthur Public Works
Implementation Schedule:	Within 12-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Master Drainage Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 4

	Port Arthur – Action #36
Proposed Action:	Improve Hwy. 365 from Hwy. 69 to Rhodair Gully for improved emergency access.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Hwy. 365 from Hwy. 69 to Rhodair Gully
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Improve emergency access through road improvement.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind, Hail, Wildfire, Lightning, Winter Storm, Tornado
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Port Arthur Public Works
Implementation Schedule:	Within 12-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Capital Improvement Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

	Port Arthur – Action #37
Proposed Action:	Decrease floodplain width in North Port Acres Ditch through improved drainage.
BACKGROUND INFORMATION	
Jurisdiction/Location:	North Port Acres Ditch
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce flood risk to structures and infrastructure through improved drainage and reduction of the floodplain.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$300,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	City of Port Arthur Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Master Drainage Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

# Port Neches

	Port Neches – Action #1
Proposed Action:	Build a structure or structures (including a dome or domes) in coastal (or near coastal) jurisdictions that can withstand 200 mile per hour winds and act as shelters of last resort. This is in conjunction with the Texas Safe Shelter Initiative.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in Port Neches
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents and first responders in Port Neches.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$1.6 Million
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM
Lead Agency/Department Responsible:	Port Neches Emergency Management
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

COMMENTS

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies) Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 4

	Port Neches – Action #2
Proposed Action:	Retrofit existing structures and/or construct new structures to act as residential shelters during and after Hurricanes and other severe weather events.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations in Port Neches
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents in Port Neches.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$2 Million
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM
Lead Agency/Department Responsible:	Port Neches Emergency Management
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

Proposed Action:	Storm harden/retrofit critical facilities throughout Port Neches. Actions can include but are not limited to window shutters, roof straps, flood proofing, roll- up door reinforcement (i.e. for fire stations), backup generator power with permanent hook-ups, hail resistant roofing materials, and surge protectors.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Critical facilities in Port Neches
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Protect critical facilities from damages and ensure continuity of emergency services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Flood, Winter Storm, Wildfire, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Port Neches Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Port Neches – Action #4
Proposed Action:	Elevate new and existing flood-prone structures and infrastructure throughout Port Neches.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Flood-prone structures and infrastructure in Port Neches
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce flood damages to flood-prone structures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood
Effect on New/Existing Buildings:	Reduce risk to existing and future structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5,000,000
Potential Funding Sources:	HMGP, PDM, FMA
Lead Agency/Department Responsible:	Port Neches Emergency Management
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Local Ordinance, Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

	Port Neches – Action #5
Proposed Action:	Elevate or upgrade bridges, culverts and other crossings throughout Port Neches to reduce damages to infrastructure and reduce flooding caused by undersized crossings and culverts.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Undersized bridges and culverts in Port Neches
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce flood damages to infrastructure and surrounding structures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing infrastructure and structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$1 Million
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Port Neches Public Works
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 4

Proposed Action:	Port Neches – Action #6 Pursue finding and implement drainage improvements throughout Port Neches. Actions can include but are not limited to installing/ upgrading culverts and headwalls as well as enlarging storm water ditches and canals.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout Port Neches
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce flood damages to infrastructure and surrounding structures as a result of undersized drainage.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane
Effect on New/Existing Buildings:	Reduce risk to existing infrastructure and structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Port Neches Public Works
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

	Port Neches – Action #7
Proposed Action:	Secure, bury or otherwise harden exposed or vulnerable pipelines.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout Port Neches
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of pipeline failure through mitigation of exposed lines.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind, Winter Storm, Wildfire, Tornado, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing infrastructure and structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Port Neches Public Works
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

	Port Neches – Action #8
Proposed Action:	Provide generators/back-up power systems with permanent hook-ups for critical facilities (including but not limited to lift stations, water plants, police, EMS, fire and other first responder facilities) throughout Port Neches.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations throughout Port Neches
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of loss of power and ensure continuity of emergency services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind, Winter Storm, Wildfire, Tornado, Lightning, Extreme Heat, Hail
Effect on New/Existing Buildings:	Reduce risk to existing infrastructure and structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$50,000 - \$200,000 per site
Potential Funding Sources:	HMGP, PDM, local operating budgets, TDEM
Lead Agency/Department Responsible:	Port Neches Emergency Management
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5

	Port Neches – Action #9
Proposed Action:	<ul> <li>Mitigate damage to utilities in order to maintain function during and after a hazard event. Actions can include but are not limited to: <ul> <li>Bury utility lines underground</li> <li>Provide frangible links/break away connections on utility poles</li> <li>Harden utility poles by converting from wood to concrete or metal utility poles</li> <li>Increase the easement area/clearance of utility lines/poles from tree lines</li> </ul> </li> </ul>
BACKGROUND INFORMATION	
Jurisdiction/Location:	Utility lines throughout Port Neches
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of loss of power and ensure continuity of emergency services.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Thunderstorm Wind, Winter Storm, Tornado, Lightning, Hail
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$200,000
Potential Funding Sources:	HMPG, PDM, local operating budgets, TDEM
Lead Agency/Department Responsible:	Port Neches Emergency Management, Entergy
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan, Local Ordinance

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Port Neches – Action #10
Proposed Action:	<ul> <li>Minimize damage to structures and infrastructure from falling trees. Actions include but are not limited to the following: <ul> <li>Pursue and coordinate a dangerous tree and limb removal program to protect infrastructure and critical facilities from damage. This includes working with private homeowners for voluntary removal of hazardous trees and limbs on private property.</li> <li>Coordinate contracts to remove and/or trim trees that endanger structures, infrastructure, and vital roadways.</li> </ul> </li> </ul>
BACKGROUND INFORMATION	
Jurisdiction/Location:	Trees near power lines throughout Port Neches
RiskReductionBenefit(CurrentCost/Losses Avoided):	Reduce risk of loss of power and potential damages to structures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Thunderstorm Wind, Winter Storm, Tornado, Lightning, Hail
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Port Neches Public Works, Entergy
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan, Local Ordinance

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

	Port Neches – Action #11
Proposed Action:	Coordinate public-private partnerships to ensure special needs population are protected from extreme temperatures by establishing and promoting accessible heating or cooling centers in the community.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Local business and identified public facilities
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk to vulnerable populations during extreme temperatures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Education and Awareness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Winter Storm, Extreme Heat
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$50,000
Potential Funding Sources:	HMGP, PDM, local operating budgets, local business
	partners
Lead Agency/Department Responsible:	Port Neches Emergency Management
Implementation Schedule:	Within 12 months of plan adoption pending available
	funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5

Proposed Action:	Port Neches – Action #12 Conduct coastal storm presentations to educate the public on evacuation procedures, mitigation techniques, and risk associated with coastal storms.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk to residents and structures through education and awareness.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Education and Awareness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Port Neches Emergency Management
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5 Education and Awareness)

	For Action #13
Proposed Action:	Provide the public with educational brochures for mitigating damages, planning ahead for disasters and reducing the risk of injury during events including: Mitigation measures such as window film, elevated appliances, surge protectors, insulating pipes, drought tolerant landscaping, education on when to take cover, when to evacuate, locations of local safe rooms, signs of dehydration, and proper storage of flammable materials, or other appropriate materials to mitigate damages and health hazards.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
RiskReductionBenefit(CurrentCost/LossesAvoided):	Reduce risk to residents and structures through education and awareness.

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood, Thunderstorm Wind, Winter Storm, Wildfire, Tornado, Lightning, Extreme Heat, Hail, Drought
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$50,000
Potential Funding Sources:	HMGP, PDM, local operating budgets, EMPG
Lead Agency/Department Responsible:	Port Neches Emergency Management
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

### COMMENTS

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5

Jefferson County | Hazard Mitigation Plan Update 2017 | Page 251

Proposed Action:	Port Neches – Action #14 Coordinate and implement construction, expansion and maintenance of flood control structures/barriers for the purpose of mitigating damage and protect fresh water resources from storm surge, sea level rise and other sources of salt water intrusion.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk to structures and infrastructure through improved flood control measures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$2,000,000
Potential Funding Sources:	HMGP, PDM, USACE, SETRPC
Lead Agency/Department Responsible:	Port Neches Public Works, SETRPC
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 4; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

	Port Neches – Action #15
Proposed Action:	Construct or upgrade existing detention/ retention ponds where appropriate to collect storm water to reduce flooding.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations TBD
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk to structures and infrastructure through improved flood control measures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budget
Lead Agency/Department Responsible:	Port Neches Public Works
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

	Port Neches – Action #16
Proposed Action:	Retrofit existing structures or construct new structures to act as residential safe rooms during tornados or other severe weather events.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations TBD
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or death for residents.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Port Neches Public Works
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan, Local Ordinance

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Port Neches – Action #17
Proposed Action:	Identify and pursue any mitigation activities that would aid/enhance evacuations throughout Port Neches.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of injury or fatality through improved evacuation routes and procedures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Wildfire, Flood
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$500,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Port Neches Emergency Management
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5

	Port Neches – Action #18
Proposed Action:	Acquire/demolish flood-prone properties with an emphasis on Repetitive and Severe Repetitive Loss properties.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Flood-prone structures in Port Neches
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Eliminate flood damages to flood-prone structures. Reduce burden on emergency services during flood events.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood
Effect on New/Existing Buildings:	Eliminate risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5,000,000
Potential Funding Sources:	HMGP, PDM, FMA
Lead Agency/Department Responsible:	Port Neches Emergency Management
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 3; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 4; Legal = 4; Economically Sound = 4; and Environmentally Sound = 5

	Port Neches – Action #19
Proposed Action:	Pursue flood protection measures to protect from surge from Hurricanes. Actions can include but are not limited to constructing and/ or upgrading sea walls, flood barriers, berms and various wet and dry flood proofing measures.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk to structures and infrastructure through improved flood control measures.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$3,000,000
Potential Funding Sources:	HMGP, PDM, USACE, SETRPC
Lead Agency/Department Responsible:	Port Neches Public Works
Implementation Schedule:	Within 48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 4; Legal = 4; Economically Sound = 4; and Environmentally Sound = 4

	Port Neches – Action #20
Proposed Action:	Retrofit existing structures to act as cooling stations in
	times of extreme heat.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Various locations TBD
Risk Reduction Benefit (Current	Reduce risk to vulnerable populations during extreme
Cost/Losses Avoided):	heat events.
Type of Action (Local Plans and	Structure and Infrastructure
Regulations, Structure and Infrastructure	
projects, Natural System Protection, or	
Education and Awareness)	

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Extreme Heat
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Port Neches Administration
Implementation Schedule:	Within 48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 4; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Port Neches – Action #21
Proposed Action:	Develop areas of defensible space to prevent damage due to wildfires.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Wildland Urban Interface
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risk of damages to structures in or near the WUI.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Wildfire
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets, EMPG
Lead Agency/Department Responsible:	Port Neches Fire Department
Implementation Schedule:	Within 48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Community Wildfire Protection Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 4; and Environmentally Sound = 5

	Port Neches – Action #22
Proposed Action:	Develop and enact water conservation or drought management plans, ordinances or strategies to be used during times of drought.
BACKGROUND INFORMATION	
Jurisdiction/Location:	City-wide
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce risks associated with drought through reduction in water usage during times of drought.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Drought
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	Low
Estimated Cost:	\$25,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	Port Neches Administration
Implementation Schedule:	Within 48 months of plan adoption pending available funding
Incorporation into Existing Plans:	Local Ordinances

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 4; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 4; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5

# SETRPC

	SETRPC – Action #1
Proposed Action:	Plan for the Protection of Vulnerable Populations by identifying at-risk populations and coordinating with home health agencies, medical equipment companies, local churches and neighborhood associations to assist these populations during extreme weather events. Organize strategies for protecting vulnerable populations and develop a plan to expediently activate strategies when need be.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Southeast Texas Region of Hardin, Jefferson and Orange Counties
RiskReductionBenefit(CurrentCost/Losses Avoided):	Protection of at-risk populations during extreme weather events.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Education and Awareness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Extreme Heat, Flood, Hurricane, Winter Storm, Thunderstorm Wind, Tornado, Wildfire, Lightning
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$25,000 - \$100,000
Potential Funding Sources:	HMGP, PDM
Lead Agency/Department Responsible:	SETRPC, Jefferson County Emergency Management
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

## COMMENTS

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5

Proposed Action:	Public Awareness and Education of Vulnerable Population through creation of a database and special group in STAN (Southeast Texas Alerting Network regional emergency alerting system) whereby public information protection actions can be disseminated.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Southeast Texas Region of Hardin, Jefferson and Orange Counties
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Expedient access to and ability to communicate with those individuals, nursing homes, assisted living centers, hospitals and others who are most at risk during extreme hazard events.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Education and Awareness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Extreme Heat, Flood, Hurricane, Winter Storm, Thunderstorm Wind, Tornado, Wildfire, Lightning, Hail
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$50,000 - \$100,000
Potential Funding Sources:	Area Agency on Aging and Disabilities, PDM, HMGP
Lead Agency/Department Responsible:	SETRPC
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Operations Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5

	SETRPC – Action #3
Proposed Action:	Plan for future drought in the Southeast Texas region of Hardin, Jefferson, and Orange Counties by developing a drought emergency plan, developing criteria or triggers for drought-related actions, enact water conservation measures during drought conditions, and develop a drought communication plan and early warning system to facilitate timely communication of relevant information.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Southeast Texas Region of Hardin, Jefferson and Orange Counties
RiskReductionBenefit(CurrentCost/Losses Avoided):	Reduction of risk due to drought for residents in the Southeast Texas region.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Drought
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$75,000 - \$100,000
Potential Funding Sources:	HMGP, PDM, local operating budgets
Lead Agency/Department Responsible:	SETRPC
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Local water plans and ordinances

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5

	SETRPC – Action #4
Proposed Action:	Pursue the identification and construction of alternate fresh water resources.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Southeast Texas Region of Hardin, Jefferson and Orange Counties
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce health risks associated with a lack of clean, uncontaminated water available to local residents.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Drought
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$100,000 - \$200,000
Potential Funding Sources:	HMGP, PDM
Lead Agency/Department Responsible:	SETRPC
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Local Water Management Plans

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 3; Technically Feasible = 3; Administratively Possible = 5; Politically Acceptable = 3; Legal = 3; Economically Sound = 3; and Environmentally Sound = 3

	SETRPC – Action #5
Proposed Action:	Facilitate use of all mass notification systems including but not limited to the Southeast Texas Alerting Network (STAN), to notify and educate the public of impending hazardous events.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Southeast Texas Region of Hardin, Jefferson and Orange Counties
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce the loss of life and property as a result of a drought, extreme heat, flood, hurricane, lightning, thunderstorm wind, tornado, winter storm, or wildfire.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Education and Awareness

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Drought, Extreme Heat, Flood, Hurricane, Tornado,
	which re, munderstorm wind, winter storm, Lightning
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$500,000
Potential Funding Sources:	PSGP, PDM, HMGP
Lead Agency/Department Responsible:	SETRPC
Implementation Schedule:	Within 12 months of plan adoption pending available
	funding
Incorporation into Existing Plans:	EOPs

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5

Proposed Action:	SETRPC – Action #6 Retrofit existing structures and/or construct new structures to act as shelters during and after Hurricanes, Floods and other severe weather events.	
BACKGROUND INFORMATION	BACKGROUND INFORMATION	
Jurisdiction/Location:	Inland at strategic points outside the floodplain throughout the Southeast Texas Region of Hardin, Jefferson and Orange Counties	
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce loss of life as a result of a natural disaster by providing a safe structure in which residents can temporarily be housed.	
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure	

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Flood, Thunderstorm Wind
Effect on New/Existing Buildings:	Reduce damages to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$1.5 - \$5 Million
Potential Funding Sources:	Texas Safe Shelter Initiative, HMGP, PDM
Lead Agency/Department Responsible:	SETRPC, Hardin County, Jefferson County, Orange County
Implementation Schedule:	Within 12-24 months of plan adoption pending funding
Incorporation into Existing Plans:	Emergency Management Plan

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 3; Legal = 5; Economically Sound = 3; and Environmentally Sound = 3

Proposed Action:	Retrofit/harden SETRPC building that serves as an alternate 911 Public Safety Answering Point (PSAP) site for local jurisdictions within the Southeast Texas region during times of natural disaster.
BACKGROUND INFORMATION	
Jurisdiction/Location:	2210 Eastex Freeway Beaumont, Texas 77703
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduction of the loss of life and property through the continuity of operations of Public Safety Answering Points (PSAP) in the Southeast Texas region in the event a natural disaster necessitates PSAP relocation from the primary local jurisdiction site.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane, Winter Storm, Thunderstorm Wind, Tornado, Wildfire, Lightning
Effect on New/Existing Buildings:	Reduce risk to existing facility
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$10,000 - \$2,500,000
Potential Funding Sources:	HMGP, PDM,
Lead Agency/Department Responsible:	SETRPC
Implementation Schedule:	Within 24-36 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Operations Plans/Local PSAP Standard Operating Procedures

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5

Proposed Action:	Coordinate Emergency Management Plans for coastal storm/hurricane events. Specific efforts include encouraging jurisdictions to install and maintain back up power at identified facilities, construct and designate emergency operations centers for disaster/emergency operations, and solicit participation in Community Emergency Response training.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Southeast Texas Region of Hardin, Jefferson and Orange Counties
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Continuity of emergency operations which saves lives and property.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$5,000 - \$50,000
Potential Funding Sources:	HMGP, EMPG
Lead Agency/Department Responsible:	SETRPC, Hardin, Jefferson and Orange Counties/ Cities within
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Response Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5
Proposed Action:	SETRPC – Action #9 Conduct flood insurance educational seminars for area realtors to increase their knowledge of National Flood Insurance Program (NFIP) and the benefits to homeowners in security flood insurance.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Southeast Texas Region of Hardin, Jefferson and Orange Counties
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	All homeowners in the Southeast Texas region could benefit from owning flood insurance. Through education with realtors, who homeowners often turn to first for advice when purchasing a home, this important message could be conveyed and the public educated on the importance of flood insurance; whether or not a home is in the floodplain.
<b>Type of Action</b> (Local Plans and Regulations Structure and Infrastructure	Education and Awareness
projects, Natural System Protection, or	
Education and Awareness)	

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood, Hurricane
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$20,000 - \$50,000
Potential Funding Sources:	HMGP, PDM
Lead Agency/Department Responsible:	SETRPC, Hardin Jefferson and Orange Counties/ Cities within
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	N/A

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	SETRPC – Action #10
Proposed Action:	Coordinate public/private partnerships to ensure special needs populations are protected from health risks due to extreme weather conditions. Actions will be targeted toward citizens with physical limitations and others who may be unable to reach safety in times of severe weather. Volunteer groups may be available to assist by visiting special needs groups to ensure their safety and comfort during extreme weather events or assist when evacuations are necessary.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Southeast Texas Region of Hardin, Jefferson and Orange Counties
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce loss of life to most vulnerable populations.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Drought, Extreme Heat, Flood, Hurricane, Winter Storm, Thunderstorm Wind, Tornado, Wildfire, Earthquake, Geologic Hazards, Tsunami, Water Contamination
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$50,000 - \$100,000
Potential Funding Sources:	HMGP, PDM
Lead Agency/Department Responsible:	SETRPC, Hardin Jefferson and Orange Counties/ Cities within
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Response Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	SETRPC – Action #11
Proposed Action:	Coordinate and implement a natural hazards public awareness campaign among the jurisdictions. Efforts may include tropical storm/ hurricane awareness presentations, shelter-in-place presentations, evacuation maps, floodplain maps, mitigation damages, flood control projects, storm tracking maps, health and safety tips flyers, mitigation articles in local newspapers, and other such information as it relates to natural hazards. Target audiences will include schools, neighborhood watch groups, various civic groups, neighborhood associations, community groups, and industry groups. FEMA publications will also be made available in city hall libraries, municipal courts, police and fire departments, public works departments, public access TV channels, city libraries and on the SETRPC and jurisdictional websites.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Southeast Texas Region of Hardin, Jefferson and Orange Counties
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce damages to structures through mitigation education. Reduce risk of injury or loss of life to area residents.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awarapass)	Education and Awareness

# Section 18: Mitigation Actions

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Drought, Extreme Heat, Hail, Flood, Hurricane, Winter Storm, Thunderstorm Wind, Tornado, Lightning, Wildfire
Effect on New/Existing Buildings:	Reduce risk to existing structures through mitigation education and early preparation
Priority (High, Moderate, Low):	High
Estimated Cost:	\$25,000 - \$75,000
Potential Funding Sources:	HMGP, PDM
Lead Agency/Department Responsible:	SETRPC, Hardin Jefferson and Orange Counties/ Cities within
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### COMMENTS

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies) Socially Acceptable = 5; Technically Feasible = 5; Administratively Possible = 5; Politically Acceptable = 5; Legal = 5; Economically Sound = 5; and Environmentally Sound = 5

Proposed Action:	SETRPC – Action #12 Construct water retention ponds to collect storm water run-off and use as an alternate water source for agricultural resources throughout the Southeast Texas region.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Southeast Texas Region of Hardin, Jefferson and Orange Counties
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Lifesaving water alternate source and localized flood reduction.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Drought, Flood, Thunderstorm Wind, Hurricane
Effect on New/Existing Buildings:	Reduce damages to existing and new structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$100,000 - \$250,000
Potential Funding Sources:	HMGP, PDM
Lead Agency/Department Responsible:	SETRPC, Hardin Jefferson and Orange Counties/ Cities within
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Master Drainage Plan, Water Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	SETRPC – Action #13
Proposed Action:	<ul> <li>Minimize damage to structures and infrastructure from falling trees. Actions include but are not limited to the following: <ul> <li>Pursue and coordinate a dangerous tree and limb removal program to protect infrastructure and critical facilities from damage. This includes working with private homeowners for voluntary removal of hazardous trees and limbs on private property.</li> <li>Coordinate contracts to remove and/or trim trees that endanger structures, infrastructure, and vital roadways.</li> </ul> </li> </ul>
BACKGROUND INFORMATION	-
Jurisdiction/Location:	Trees near power lines throughout Southeast Texas Region of Hardin, Jefferson and Orange Counties
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce loss of life and property due to substantial damage from falling trees resulting in failing structures and reduce risk of power outages.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Tornado, Thunderstorm Wind, Hail, Lightning, Winter Storm
Effect on New/Existing Buildings:	Reduce risk to existing and new structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$50,000 - \$100,000
Potential Funding Sources:	HMGP, PDM, local Utility Fees
Lead Agency/Department Responsible:	SETRPC, Hardin Jefferson and Orange Counties/ Cities within
Implementation Schedule:	Within 24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Public Works SOP

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	SETRPC – Action #14
Proposed Action:	Secure and maintain backup information systems to store critical information at off-site locations.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Southeast Texas Region of Hardin, Jefferson and Orange Counties
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce loss of critical government data and files through redundant systems.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood, Tornado, Wildfire
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$50,000 - \$350,000
Potential Funding Sources:	HMGP, PDM
Lead Agency/Department Responsible:	SETRPC, Hardin Jefferson and Orange Counties/ Cities within
Implementation Schedule:	Within 24 months of plan adoption pending available funding
Incorporation into Existing Plans:	N/A

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	SETRPC – Action #15 Coordinate with county and municipal governments to allow the SETRPC to maintain a copy of all local ordinances relevant to mitigation activities.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Southeast Texas Region of Hardin, Jefferson and Orange Counties
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce loss of local ordinances through redundant systems.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood, Drought, Extreme Heat, Tornado, Winter Storm, Wildfire
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$1,000 - \$15,000
Potential Funding Sources:	HMGP, PDM
Lead Agency/Department Responsible:	SETRPC, Hardin Jefferson and Orange Counties/ Cities within
Implementation Schedule:	Within 24 months of plan adoption pending available funding
Incorporation into Existing Plans:	N/A

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

Proposed Action:	SETRPC – Action #16 Acquire flood-prone properties throughout the region with a focus on Repetitive and Severe Repetitive Loss properties.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Southeast Texas Region of Hardin, Jefferson and Orange Counties
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduction in repetitive loss payments from FEMA for structures known to have flooded on more than one occasion.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood
Effect on New/Existing Buildings:	Eliminate risk to existing structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$90,000 - \$10,000,000 (cost varies per structure)
Potential Funding Sources:	HMGP, PDM, FMA
Lead Agency/Department Responsible:	SETRPC, Hardin Jefferson and Orange Counties/ Cities within
Implementation Schedule:	Within 24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	SETRPC – Action #17
Proposed Action:	Elevate flood-prone properties throughout the region.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Southeast Texas Region of Hardin, Jefferson and Orange Counties
Risk Reduction Benefit (Current	Reduction in repetitive loss payments from FEMA for
Cost/Losses Avoided):	structures known to lie in a floodplain or that have flooded.
<b>Type of Action</b> (Local Plans and Regulations Structure and Infractructure)	Structure and Infrastructure
projects Natural System Protection or	
Education and Awareness)	

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	Moderate
Estimated Cost:	\$90,000 - \$2,000,000 (cost varies per structure)
Potential Funding Sources:	HMGP, PDM
Lead Agency/Department Responsible:	SETRPC, Hardin Jefferson and Orange Counties/ Cities within
Implementation Schedule:	Within 24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	SETRPC – Action #18
Proposed Action:	Storm harden/retrofit critical facilities throughout the Southeast Texas Region. Actions can include but are not limited to window shutters, roof straps, hail resistant roofing, flood proofing, roll-up door reinforcement (i.e. fire stations), emergency backup/generator power with permanent hook-ups, fire resistant construction materials, window film, and surge protectors.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Southeast Texas Region of Hardin, Jefferson and Orange Counties
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Avoid loss of property and, due to the critical nature of the facilities included in this action, potential loss of life.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood, Thunderstorm Wind, Hail, Lightning, Winter Storm, Wildfire, Extreme Heat
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$10,000 - \$2,000,000
Potential Funding Sources:	HMGP, PDM
Lead Agency/Department Responsible:	SETRPC, Hardin Jefferson and Orange Counties/ Cities within
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Emergency Management Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	SETRPC – Action #19
Proposed Action:	Pursue drainage improvements throughout the regions. Actions can include but are not limited to installing/upgrading culverts and headwalls as well as enlarging storm water ditches and canals.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Southeast Texas Region of Hardin, Jefferson and Orange Counties
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduction in property losses/repetitive loss buy outs and infrastructure damage.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	Reduce risk to existing structures
Priority (High, Moderate, Low):	High
Estimated Cost:	\$100,000 - \$5,000,000
Potential Funding Sources:	HMGP, PDM
Lead Agency/Department Responsible:	SETRPC, Hardin Jefferson and Orange Counties/ Cities within
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Drainage Plans

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	SETRPC – Action #20
Proposed Action:	<ul> <li>Mitigate damage to utilities in order to maintain function during and after a hazard event. Actions can include but are not limited to: <ul> <li>Bury utility lines underground</li> <li>Provide frangible links/break away connections on utility poles</li> <li>Harden utility poles by converting from wood to concrete or metal utility poles</li> <li>Increase the easement area/clearance of utility lines/poles from tree lines</li> </ul> </li> </ul>
BACKGROUND INFORMATION	
Jurisdiction/Location:	Southeast Texas Region of Hardin, Jefferson and Orange Counties
RiskReductionBenefit(CurrentCost/Losses Avoided):	Reduce risk of power loss, associated damages, and line repairs.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood, Winter Storm, Hail, Lightning, Thunderstorm Wind, Tornado
Effect on New/Existing Buildings:	Reduce risk to existing structures and infrastructure
Priority (High, Moderate, Low):	High
Estimated Cost:	\$150,000 - \$600,000
Potential Funding Sources:	HMGP, PDM, Utility Fees
Lead Agency/Department Responsible:	SETRPC, Hardin Jefferson and Orange Counties/ Cities within
Implementation Schedule:	Within 12-24 months of plan adoption pending available funding
Incorporation into Existing Plans:	Comprehensive Plan

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	SETRPC – Action #21
Proposed Action:	Coordinate with federal, state and local partners to provide all hazards, ICS, and specialized training that may enhance preparedness for first responders.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Southeast Texas Region of Hardin, Jefferson and Orange Counties
Risk Reduction Benefit (Current	Increase in public safety knowledge base through
Cost/Losses Avoided):	education and awareness.
Type of Action (Local Plans and	Education and Awareness - Preparedness
Regulations, Structure and Infrastructure	
projects, Natural System Protection, or	
Education and Awareness)	

MITIGATION ACTION DETAILS		
Hazard(s) Addressed:	Hurricane, Flood, Drought, Extreme Heat, Winter Storm, Tornado, Lightning, Thunderstorm Wind, Wildfire, Hail	
Effect on New/Existing Buildings:	N/A	
Priority (High, Moderate, Low):	High	
Estimated Cost:	\$1,000 - \$50,000	
Potential Funding Sources:	HMGP, PDM	
Lead Agency/Department Responsible:	SETRPC, Hardin Jefferson and Orange Counties/ Cities within	
Implementation Schedule:	Within 12 months of plan adoption pending available funding	
Incorporation into Existing Plans:	Emergency Operations Plan	

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	SETRPC – Action #22
Proposed Action:	Identify and pursue any mitigation activities that would aid/enhance evacuations throughout the region.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Southeast Texas Region of Hardin, Jefferson and Orange Counties
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Life safety benefits through education and awareness.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Education and Awareness

MITIGATION ACTION DETAILS		
Hazard(s) Addressed:	Hurricane, Flood, Winter Storm, Tornado, Thunderstorm Wind, Wildfire	
Effect on New/Existing Buildings:	N/A	
Priority (High, Moderate, Low):	High	
Estimated Cost:	\$50,000 - \$500,000	
Potential Funding Sources:	HMGP, PDM	
Lead Agency/Department Responsible:	SETRPC, Hardin Jefferson and Orange Counties/ Cities within	
Implementation Schedule:	Within 12 months of plan adoption pending available funding	
Incorporation into Existing Plans:	Emergency Management Plan	

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	SETRPC – Action #23
Proposed Action:	Continually review, revise, update, and systematically maintain floodplain data and maps of flood prone areas throughout the Southeast Texas Region.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Southeast Texas Region of Hardin, Jefferson and Orange Counties
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce flood risk to structures through understanding risk and vulnerability.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Local Plans and Regulations

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Flood
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$1,000 - \$5,000
Potential Funding Sources:	HMGP, PDM
Lead Agency/Department Responsible:	SETRPC, Hardin Jefferson and Orange Counties/ Cities within
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Floodplain Ordinance

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	SETRPC – Action #24
Proposed Action:	Pursue the coordination, construction, expansion and maintenance of flood control structures/barriers for the purpose of mitigating damage from storm surge and sea level rise and other sources of salt water intrusion.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Southeast Texas Region of Hardin, Jefferson and Orange Counties
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce damages to structures and infrastructure from storm surge and flooding.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS		
Hazard(s) Addressed:	Hurricane, Flood	
Effect on New/Existing Buildings:	Reduce risk to existing and new structures	
Priority (High, Moderate, Low):	High	
Estimated Cost:	\$10,000,000	
Potential Funding Sources:	HMGP, PDM	
Lead Agency/Department Responsible:	SETRPC, Hardin Jefferson and Orange Counties/ Cities within	
Implementation Schedule:	Within 12 months of plan adoption pending available funding	
Incorporation into Existing Plans:	Master Drainage Plan	

#### Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

	SETRPC – Action #25
Proposed Action:	Pursue the coordination, construction, expansion, and maintenance of flood control structures/barriers for the purpose of protecting potable water sources and agricultural resources from water contamination and salt water intrusion.
BACKGROUND INFORMATION	
Jurisdiction/Location:	Southeast Texas Region of Hardin, Jefferson and Orange Counties
<b>Risk Reduction Benefit</b> (Current Cost/Losses Avoided):	Reduce loss of potable water or inadequate water supply.
<b>Type of Action</b> (Local Plans and Regulations, Structure and Infrastructure projects, Natural System Protection, or Education and Awareness)	Structure and Infrastructure

MITIGATION ACTION DETAILS	
Hazard(s) Addressed:	Hurricane, Flood
Effect on New/Existing Buildings:	N/A
Priority (High, Moderate, Low):	High
Estimated Cost:	\$10,000,000
Potential Funding Sources:	HMGP, PDM
Lead Agency/Department Responsible:	SETRPC, Hardin Jefferson and Orange Counties/ Cities within
Implementation Schedule:	Within 12 months of plan adoption pending available funding
Incorporation into Existing Plans:	Master Drainage Plan

Additional Considerations:

The following STAPLEE criteria were evaluated on a scale of 1 to 5 indicating the extent to which this action satisfies each consideration. (1= Does Not Satisfy 3 = Moderately Satisfies 5 = Strongly Satisfies)

# **SECTION 19: PLAN MAINTENANCE**

Plan Maintenance Procedures	.1
Incorporation	.1
Process of Incorporation	.1
Monitoring and Evaluation	.5
Monitoring	.5
Evaluation	.6
Updating	.6
Plan Amendments	.6
Five (5) Year Review	.6
Continued Public Involvement	.7

# **Plan Maintenance Procedures**

The following is an explanation of how Jefferson County, participating jurisdictions, and the general public will be involved in implementing, evaluating, and enhancing the Plan over time. The sustained hazard mitigation planning process consists of four main parts:

- Incorporation
- Monitoring and Evaluation
- Updating
- Continued Public Involvement

# Incorporation

Jefferson County and participating jurisdictions will be responsible for further development and implementation of mitigation actions. Each action has been assigned to a specific department within the County and participating jurisdictions. The following describes the process by which Jefferson County will incorporate elements of the mitigation plan into other planning mechanisms.

# **Process of Incorporation**

Once the Plan is adopted, Jefferson County, SETRPC and participating jurisdictions will implement actions based on priority and the availability of funding. The County currently implements policies and programs to reduce loss to life and property from hazards. The mitigation actions developed for this Plan Update enhance this ongoing effort and will be implemented through other program mechanisms where possible.

The potential funding sources listed for each identified action may be used when the jurisdiction seeks funds to implement actions. An implementation time period or a specific implementation date has been

assigned to each action as an incentive for completing each task and gauging whether actions are implemented in a timely manner.

Jefferson County and participating jurisdictions will integrate implementation of their mitigation actions with other plans and policies such as construction standards and emergency management plans, and ensure that these actions, or proposed projects, are reflected in other planning efforts. Coordinating and integrating components of other plans and policies into goals and objectives of the Plan will further maximize funding and provide possible cost-sharing of key projects, thereby reducing loss of lives and property, and mitigating hazards affecting the area.

Upon formal adoption of the Plan Update, planning team members from each participating jurisdiction will work to integrate the hazard mitigation strategies into other plans and codes, as they are developed. Participating team members will conduct periodic review of plans and policies, once per year at a minimum, and analyze the need for amendments in light of the approved Plan Update. The planning team will review all comprehensive land use plans, capital improvement plans, annual budget reviews, emergency operations or management plans, transportation plans, and any building codes to guide and control development. Participating jurisdictions will ensure that capital improvement planning in the future will also contribute to the goals of this hazard mitigation Plan Update to reduce the long-term risk to life and property from all hazards. Within one year of formal adoption of the hazard mitigation Plan Update, existing planning mechanisms will be reviewed by each jurisdiction.

Jefferson County is committed to supporting the cities, communities, and participating jurisdictions as they implement their mitigation actions. Jefferson County and participating planning team members will review and revise, as necessary, the long-range goals and objectives in strategic plan and budgets to ensure that they are consistent with this mitigation action plan. Additionally, the County will work to advance the goals of this hazard mitigation plan through its routine, ongoing, long-range planning, budgeting, and work processes.

Planning Mechanism	Department/Title Responsible	Incorporation of Plan
Grant Applications	Jefferson County: Emergency Management Coordinator Beaumont: Emergency Management Coordinator Bevil Oaks: Mayor China: Mayor Groves: Emergency Management	The Plan Update will be evaluated by Jefferson County and participating jurisdictions when grant funding is sought for mitigation projects. If a project is not in the Plan Update, an amendment may be necessary to include the action in the Plan Update.
	coordinator	

# Table 19-1. Methods of Incorporation of the Plan

Planning Mechanism	Department/Title Responsible	Incorporation of Plan
	Nederland: Emergency Management Coordinator Nome: Mayor Port Arthur: Senior Planner Port Neches: Emergency Management	
	Coordinator SETRPC: Regional Emergency Planner	
Annual Budget Review	Jefferson County: EmergencyManagement CoordinatorBeaumont: Emergency ManagementCoordinatorBevil Oaks: MayorChina: MayorGroves: Emergency ManagementCoordinatorNederland: Emergency ManagementCoordinatorNome: MayorPort Arthur: Senior PlannerPort Neches: Emergency ManagementCoordinatorSETRPC: Regional Emergency Planner	Various departments and key personnel that participated in the planning process for Jefferson County and participating jurisdictions will review the Plan Update and mitigation actions therein when conducting their annual budget review. Allowances will be made in accordance with grant applications sought, and mitigation actions that will be undertaken, according to the implementation schedule of the specific action.
Regulatory Plans	Jefferson County: Emergency Management Coordinator Beaumont: Emergency Management Coordinator Bevil Oaks: Mayor China: Mayor Groves: Emergency Management Coordinator	Currently, Jefferson County and participating jurisdictions have regulatory plans in place, such as Emergency Management Plans, Continuity of Operations Plans, Economic Development, and Evacuation Plans. The Plan Update will be consulted when County and City departments review or revise their current regulatory planning mechanisms, or in the development of

Planning Mechanism	Department/Title Responsible	Incorporation of Plan
	Nederland: Emergency Management Coordinator Nome: Mayor Port Arthur: Senior Planner Port Neches: Emergency Management Coordinator SETRPC: Regional Emergency Planner	regulatory plans that are not currently in place.
Capital Improvement Plans	Jefferson County: EmergencyManagement CoordinatorBeaumont: Emergency ManagementCoordinatorBevil Oaks: MayorChina: MayorGroves: Emergency ManagementCoordinatorNederland: Emergency ManagementCoordinatorNome: MayorPort Arthur: Senior PlannerPort Neches: Emergency ManagementCoordinatorSETRPC: Regional Emergency Planner	Jefferson County and participating jurisdictions have a Capital Improvement Plan (CIP) in place. Prior to any revisions to the CIP, County and City departments will review the risk assessment and mitigation strategy sections of the HMAP, as limiting public spending in hazardous zones is one of the most effective long- term mitigation actions available to local governments.
Floodplain Management Plans	Jefferson County: Floodplain Manager Beaumont: Floodplain Manager Bevil Oaks: Floodplain Manager China: Mayor Groves: Floodplain Manager Nederland: Floodplain Manager Nome: Mayor Port Arthur: Floodplain Manager	Floodplain management plans include preventative and corrective actions to address the flood hazard. Therefore, the actions for flooding, and information found in Section 5 of this Plan Update discussing the people and property at risk to flood, will be reviewed and revised when Jefferson County updates their management plans or develops new plans.

Planning Mechanism	Department/Title Responsible	Incorporation of Plan
	Port Neches: Floodplain Manager	
	SETRPC: Floodplain Manager	

# Monitoring and Evaluation

Periodic revisions of the Plan Update are required to ensure that goals, objectives, and mitigation actions are kept current. Revisions may be required to ensure the Plan Update is in compliance with federal and state statutes and regulations. This section outlines the procedures for completing Plan revisions, updates, and review. Table 19-2 indicates the department and title of the party responsible for Plan monitoring, updating, and review of the Plan.

# Table 19-2. Team Members Responsible for Plan Monitoring, Evaluating, Reviewing and Updating ofthe Plan

JURISDICTION	TITLE
Jefferson County	Emergency Management Coordinator
Beaumont	Emergency Management Coordinator
Bevil Oaks	Mayor
China	Mayor
Groves	Emergency Management Coordinator
Nederland	Emergency Management Coordinator
Nome	Mayor
Port Arthur	Senior Planner
Port Neches	Emergency Management Coordinator
SETRPC	Regional Emergency Planner

# Monitoring

Designated Planning Team members are responsible for monitoring, updating, and reviewing the Plan Update, as shown in Table 19-2. Individuals holding the title listed in Table 19-2 will be responsible for monitoring the Plan Update on an annual basis. Plan monitoring, includes reviewing and incorporation into the Plan other existing planning mechanisms that relate or support goals and objectives of the Plan; monitoring the incorporation of the Plan into future updates of other existing planning mechanisms as appropriate; reviewing mitigation actions submitted and coordinating with various County and City departments to determine if mitigation actions need to be re-evaluated and updated; evaluating and updating the Plan as necessary; and monitoring plan maintenance to ensure that the process described is being followed, on an annual basis, throughout the planning process. The Planning Team will develop a

brief report that identifies if changes to the Plan Update are needed, such as recommending an action for funding. A summary of meeting notes will report the particulars involved in developing an action into a project.

# Evaluation

As part of the evaluation process, the Planning Team will assess changes in risk; determine whether the implementation of mitigation actions is on schedule; determine whether there are any implementation problems, such as technical, political, legal, or coordination issues; and identify changes in land development or programs that affect mitigation priorities for each respective department or organization.

The Planning Team will meet on an annual basis to evaluate the Plan and identify any needed changes. The annual evaluation process will help to determine if any changes are necessary.

# Updating

# Plan Amendments

At any time, minor technical changes may be made to update the Jefferson County Hazard Mitigation Plan Update. Material changes to mitigation actions or major changes in the overall direction of the Plan Update or the policies contained within it, must be subject to formal adoption by the County and participating jurisdictions.

The County will review proposed amendments and vote to accept, reject, or amend the proposed change. Upon ratification, the amendment will be transmitted to TDEM.

In determining whether to recommend approval or denial of a Plan Update amendment request, the County will consider the following factors:

- Errors or omissions made in the identification of issues or needs during the preparation of the Plan Update;
- New issues or needs that were not adequately addressed in the Plan Update; and
- Changes in information, data, or assumptions from those on which the Plan Update was based.

# Five (5) Year Review

The Plan will be thoroughly reviewed by the Planning Team at the end of three years from the approval date, to determine whether there have been significant changes in the planning area that necessitate changes in the types of mitigation actions proposed. Factors that may affect the content of the Plan include new development in identified hazard areas, increased exposure to hazards, disaster declarations, increase or decrease in capability to address hazards, and changes to federal or state legislation.

The Plan review process provides the County and participating jurisdictions an opportunity to evaluate mitigation actions that have been successful, identify losses avoided due to the implementation of specific mitigation measures, and address mitigation actions that may not have been successfully implemented as assigned.

It is recommended that the full Advisory Planning Team (Section 2, Table 2-2) meet to review the Plan at the end of three years because grant funds may be necessary for the development of a five-year update. Reviewing planning grant options in advance of the five-year Plan update deadline is recommended considering the timelines for grant and planning cycles can be in excess of a year.

Following the Plan review, any revisions deemed necessary will be summarized and implemented according to the reporting procedures and Plan amendment process outlined herein. Upon completion of the review, update, and amendment process the revised Plan will be submitted to TDEM for final review and approval in coordination with FEMA.

# **Continued Public Involvement**

Public input was an integral part of the preparation of this Plan and will continue to be essential for Plan updates. The Public will be directly involved in the annual review and cyclical updates. Changes or suggestions to improve or update the Plan will provide opportunities for additional public input.

The public can review the Plan Update on Jefferson County's website where officials and the public are invited to provide ongoing feedback, via email to the County's Assistant Emergency Management Coordinator at mwhite@co.jefferson.tx.us. Additionally, hard copies will be kept at the SETRPC's office and the County's Engineering Office.

The Planning Team may also designate voluntary citizens from the County, or willing stakeholder members from the private sector businesses that were involved in the Plan's development to provide feedback on an annual basis. It is important that stakeholders and the immediate community maintain a vested interest in preserving the functionality of the planning area as it pertains to the overall goals of the mitigation plan. The Planning team is responsible for notifying stakeholders and community members on an annual basis, and maintaining the Plan.

Media, including local newspaper and radio stations, will be used to notify the public of any maintenance or periodic review activities during the implementation, monitoring, and evaluation phases. Additionally, local news media will be contacted to cover information regarding Plan updates, status of grant applications, and project implementation. Local and social media outlets, such as Facebook and Twitter, will keep the public and stakeholders apprised of potential opportunities to fund and implement mitigation projects identified in the Plan Update.

# APPENDIX A: LOW RISK AND MANMADE HAZARDS

Overview	.1
Study Area Definition	.1
lazard Profiles, Vulnerability, and Impact	.3
Geologic Hazard	.3
Tsunami	.4
Earthquake	.5
Water Contamination	.6
Dam Failure	.8
Hazardous Materials Incident (Fixed and Mobile)	11
Terrorism	16

# Overview

During the early stages of the planning process the team analyzed several natural hazards that were considered low risk. These hazards include Earthquake, Tsunami, and Geologic Hazards, and Dam Failure. In addition, the team reviewed technological hazards including Hazardous Material Incidents, Terrorism, and Water Contamination. A description of the hazard and Jefferson County's overall vulnerability to that hazard was developed. Annualized loss data is provided where available and impact is addressed looking at the warning time or potential speed of onset of the hazard.

None of these hazards have had reported damages to any of the critical facilities for the Jefferson County planning area, therefore the planning area has not had any impact due to these hazards nor do they pose a risk to the critical services provided. In the intent of 44 CFR 201.6(c)(2)(i) & 44 CRF 201.6(c)(2)(iii) the intent is to, "To understand the potential and chronic hazards affecting the planning area in order to identify which hazard risks are most significant (...),". Based on the intent, it is the participating jurisdictions belief that earthquakes, tsunamis, geologic hazards, and dam failure are not hazards that are most significant to the jurisdiction. During public outreach none of these hazards were a concern of the public population.

# **Study Area Definition**

All areas of Jefferson County and participating jurisdictions and entities are included. Figure A-1 shows the study area for the Jefferson County HMAP Update 2017.



#### Figure A-1. Jefferson County Study Area

# Hazard Profiles, Vulnerability, and Impact

Each low risk natural hazard includes a description of the hazard and a summary of the planning area's risk. For each of the three technological hazards, a description of the hazard and Jefferson County's overall vulnerability to that hazard was developed. Impact is addressed looking at the warning time or potential speed of onset of the hazard. Impact statements are defined in Table A-1 below.

POTENTIAL SEVERITY	DESCRIPTION
Substantial	Multiple deaths. Complete shutdown of facilities for 30 days or more. More than 50 percent of property destroyed or with major damage.
Major	Injuries and illnesses resulting in permanent disability. Complete shutdown of critical facilities for at least two weeks. More than 25 percent of property destroyed or with major damage.
Minor	Injuries and illnesses do not result in permanent disability. Complete shutdown of critical facilities for more than one week. More than 10 percent of property destroyed or with major damage.
Limited	Injuries and illnesses are treatable with first aid. Shutdown of critical facilities and services for 24 hours or less. Less than 10 percent of property destroyed or with major damage.

Technological hazards refers to the origins of incidents that can arise from human activities, such as the construction and maintenance of dams. Technological hazards are distinct from natural hazards primarily because they originate from human activity. Whereas the risks presented by natural hazards may be increased or decreased as a result of human activity, they are not inherently human-induced. Therefore, dam failure is classified as a quasi-technological hazard.

For the purposes of this risk assessment, technological hazards are events or incidents associated with the use of gas and oil pipeline and their manufacture, transportation, and storage. Water contamination, acts of terrorism, and the use of hazardous materials across all industries are also considered technological hazards.

The scope of this risk assessment assumes that hazardous material incidents and water contamination events addressed in this section would be accidental in nature and that their consequences are unplanned and unintended.

# **Geologic Hazard**

A geologic hazard is a natural geologic event that can endanger human lives and threaten property and infrastructure. While geologic hazards are by definition a natural event, they can be caused or exacerbated by human activities. For the purpose of this hazard mitigation action plan update for Jefferson County, included in this hazard type are riverine erosion, landslides, and land subsidence (sinkholes). The U.S. Geological Survey (USGS) serves as the primary data and forecasting source for geologic hazards.

Riverine erosion is defined as downstream flow, shifting, or removal of sediment from a watershed. Caving river and stream banks are common associations with the migration of river channel alignment, and can threaten structures, undermine bridge foundations, and pose public safety risk.

Landslide is a general term used to describe the process of movement of material (i.e., soil, rock, mud, etc.) down a slope by falling, sliding or flowing under the force of gravity. The major causes of landslides are earthquakes, volcanic eruptions, or extreme rain events. Landslides are commonly associated with areas of steep slopes, but can also occur in relatively level topography on un-retained constructed slopes and dirt embankments. Sloughing fill material can cause property and infrastructure damage, and indirectly threaten public safety.

Land Subsidence can occur either gradually or dramatically (as in sinkhole occurrence), and refers to the loss of surface elevation due to remove of subsurface support. Land subsidence can be caused by crustal deformation; sediment compaction; withdrawal of groundwater, hydrocarbons (crude oil and natural gas), geothermal fluids or minerals (Sulphur); or increased surface load associated with high-rise buildings.

All three geologic hazards were researched for previous occurrences. Impacts of geologic hazards in Jefferson County are not widespread, and historically have been limited to minor land loss along waterways, Sabine Lake and the banks of the Gulf Inter-Coastal Water Way. Probability of future events is considered unlikely. Due to relatively isolated occurrence of impacts and no recorded occurrence of damages, injuries or fatalities, the hazard is considered to have a negligible impact on the planning area and is therefore considered a nuisance.

# Tsunami

The National Oceanic and Atmospheric Administration (NOAA) describes a tsunami as a series of ocean waves generated by sudden displacements in the sea floor, landslides, volcanic activity or other large, abrupt disturbance of the sea-surface. Tsunamis have reached heights of more than 100 feet. As the waves approach shallow coastal waters, they appear normal and the speed decreases. If the disturbance is close to the coastline, tsunamis can demolish coastal communities within minutes, and a large disturbance can cause inundation and destruction thousands of miles away from its epicenter.

The USGS monitors earthquakes through network of seismic detectors. This information is critical to understand when a tsunami wave might be generated. The USGS and NOAA's National Ocean Service has the responsibilities for providing ocean bathymetry, coastlines and topography. The information is critical to understand how and where a tsunami wave will come ashore. NOAA research develops models that forecast tsunami impacts and create inundation maps of modeled events. NOAA research provides the forecast models to the NOAA's Weather Service forecasters and the inundation models and maps to state and national planner and emergency managers. NOAA monitors sea height through a network of buoys and tide gauges. This information is critical to understand the height a tsunami wave may be when it comes ashore. NOAA completed the original 6-buoy operational array in 2001 and expanded to a full network of 30 stations in March 2008 which includes the Gulf of Mexico.

According to the National Oceanic and Atmospheric Administration (NOAA), since 1900, over 200 tsunami events have affected the coasts of the United States and its territories, causing more than 500 deaths. Tsunami events are well documented in the Pacific Ocean Basin. Tsunamis have also occurred in the Gulf of Mexico. In 1991, a magnitude 7.6 earthquake in Costa Rica produced a six foot high tsunami that flooded nearly 1,000 feet inland on the Caribbean side of the country. The Caribbean also has a number

of active submarine volcanoes and fault systems that are capable of producing large earthquakes like that in Haiti, which could generate a tsunami. There are no recorded occurrences of tsunami impacts in Jefferson County.

The National Tsunami Hazard Mitigation Program produced an assessment in August 2008 that assigned a "very low" hazard classification for the U.S. Gulf Coast based on previous frequency and local earthquake probability. Probability of future events is considered unlikely. Overall vulnerability to tsunami is considered very low based on the remote potential for causal.

# Earthquake

An earthquake is a sudden motion or trembling of the earth caused by an abrupt release of stored energy in the rocks beneath the earth's surface. The energy released results in vibrations known as seismic waves that are responsible for the trembling and shaking of the ground during an earthquake. Ground motion is expressed as peak ground acceleration (PGA). PGA is expressed as a percent of gravity or "g".

Earthquakes are typically described in terms of magnitude and intensity. The traditional measurement of amplitude of the seismic wave through the assignment of a single number to quantify the amount of seismic energy released by an earthquake is the Richter scale. The intensity of how strong the shock was felt at a particular location is the Modified Mercalli Intensity (MMI) scale. The scale quantifies the effects of an earthquake on the Earth's surface, humans, objects of nature and man-made structures. Table A-2 below is a combined earthquake magnitude and intensity comparison from the United States Geological Survey.

PGA (% g)	Magnitude (Richter)	Intensity (MMI)	Description
<0.17	1.0 - 3.0	I	I. Not felt except by a very few under especially favorable conditions.
0.17 - 1.4	3.0 - 3.9	11 - 111	<ul> <li>II. Felt only by a few persons at rest, especially on upper floors of buildings.</li> <li>III. Felt quite noticeably by persons indoors, especially on upper floors of buildings. Many people do not recognize it as an earthquake. Standing motor cars may rock slightly. Vibrations similar to the passing of a truck. Duration estimated.</li> </ul>
1.4 - 9.2	4.0 - 4.9	IV - V	<ul> <li>IV. Felt indoors by many, outdoors by few during the day.</li> <li>At night, some awakened. Dishes, windows, doors disturbed; walls make cracking sound. Sensation like heavy truck striking building. Standing motor cars rocked noticeably.</li> <li>V. Felt by nearly everyone; many awakened. Some dishes, windows broken. Unstable objects overturned. Pendulum clocks may stop.</li> </ul>

## Table A-2. Earthquake Magnitude/Intensity Comparison<sup>1</sup>

<sup>&</sup>lt;sup>1</sup> Source: Wald, D., et al., 1999, "Relationship between Peak Ground Acceleration, Peak Ground Motion, and Modified Mercalli Intensity in California," *Earthquake Spectra*, v. 15, p. 557 – 564. USGS Magnitude/Intensity Comparison http://earthquake.usgs.gov/learn/topics/mag\_vs\_int.php

PGA (% g)	Magnitude (Richter)	Intensity (MMI)	Description
9.2 - 34	5.0 - 5.9	VI - VII	<ul> <li>VI. Felt by all, many frightened. Some heavy furniture moved; a few instances of fallen plaster. Damage slight.</li> <li>VII. Damage negligible in buildings of good design and construction; slight to moderate in well-built ordinary structures; considerable damage in poorly built or badly designed structures; some chimneys broken.</li> </ul>
34 - 124	6.0 - 6.9	VII - IX	<ul> <li>VIII. Damage slight in specially designed structures; considerable damage in ordinary substantial buildings with partial collapse. Damage great in poorly built structures. Fall of chimneys, factory stacks, columns, monuments, walls. Heavy furniture overturned.</li> <li>IX. Damage considerable in specially designed structures; well-designed frame structures thrown out of plumb. Damage great in substantial buildings, with partial collapse. Buildings shifted off foundations.</li> </ul>
>124	7.0 and higher	VIII or higher	<ul> <li>X. Some well-built wooden structures destroyed; most masonry and frame structures destroyed with foundations. Rails bent.</li> <li>XI. Few, if any (masonry) structures remain standing. Bridges destroyed. Rails bent greatly</li> <li>XII. Damage total. Lines of sight and level are distorted. Objects thrown into the air.</li> </ul>

There are no recorded earthquakes with epicenters in Jefferson County, and the planning area is roughly 250 miles from the region of recent (minor) seismic activity in Northeast Texas. The annual probability for earthquakes capable of structural damage in the planning area is considered very low. The magnitude or intensity of a potential earthquake in the planning area based on historical data is an Intensity level of I or II (Table A-2). Based on the probability of future occurrences and magnitude/severity the overall vulnerability is considered low and the hazard is considered to have a negligible impact on the planning area.

# Water Contamination

# Hazard Profile

Water Contamination is the introduction of point and non-point source pollutants into public ground and/or surface water supplies. Microbiological and chemical contaminants can enter water supplies. Chemicals can leach through soils from leaking underground storage tanks, feedlots and waste disposal sites. Human wastes and pesticides can also be carried into surface waters during high water events.

The Environmental Protection Agency (EPA) is the federal agency authorized to protect the environment and public health. Congress writes the laws and the President signs them into law. The EPA is a regulatory agency with the duty to prepare administrative rules and procedures on how these laws and Presidential Executive Orders will be implemented and enforced.

The Clean Water Act establishes the basic structure for regulating discharges of pollutants into the waters of the United States and regulating quality standards for surface waters. Under the Clean Water Act, the

EPA has implemented pollution control programs. The Clean Water Act made it unlawful to discharge any pollutant from a point source into navigable waters, unless a permit was obtained. The EPA's National Pollution Discharge Elimination System (NPDES) permit program controls discharges.

Further, the EPA is the federal authority to protect drinking water. The Safe Water Drinking Act was established to protect the quality of drinking water in the U.S. The law focuses on all water actually or potentially designed for drinking use, whether from above ground or underground sources. The Act authorizes the EPA to establish minimum standards to protect tap water and requires all owners or operators of public water systems to comply with these primary health related standards<sup>2</sup>.

States must adopt rules that are at least as restrictive as the Clean Water Act and the Safe Water Drinking Act standards. The Texas Commission on Environmental Quality establishes State rules and regulations for public water systems and also specifies construction and operational standards for public water supply systems.

Disasters such as hurricanes and floods can disrupt drinking water supply and wastewater disposal systems. The Texas Commission on Environmental Quality provides guidance on remediation of public water supply systems after potential contamination due to natural disasters. Further, the Jefferson County Emergency Management Plan provides guidance regarding emergency water supplies after a disaster.

# Location

Potential and ongoing water contamination is present along all waterways and in the groundwater supply. Per a 1990 report by the Texas Water Development Board, surface water supplies the majority of municipal and industrial demands, which make up the largest portion of total water use in the planning area. Ground-water needs, including all municipal requirements in Jefferson County, were met almost entirely from the lower Chicot Aquifer.

# Extent

In general, levels of water contamination can influence community health when considered severe. Accordingly, magnitude and severity of water contamination is considered Critical by the Team, with potential public safety risks present and the potential for extended loss of function for water processing facilities. The high concentration of hazardous materials processing and shipping facilities in the planning area, low topographic gradient influencing river discharge rates and levels of dissolved oxygen, and relatively high total maximum daily load readings (TMDLs) in monitored surface water, all contribute to the magnitude and severity assessment by the Team.

# **Previous Occurrences**

According to the Evaluation of Water Resources of Orange and Eastern Jefferson Counties (Texas Water Development Board, 1990), the main ground-water quality problem is elevated chloride concentrations caused by saline-water encroachment in areas of concentrated pumpage, although from the late 1970's to 1988, chloride concentrations have not changed significantly due to decreased ground-water withdrawals.

<sup>&</sup>lt;sup>2</sup> Source: EPA

# Probability of Future Events

Considering ongoing problems and previous water quality monitoring results, probability of future occurrence is considered highly likely.

# Vulnerability and Impact

Water contamination can have a "substantial" impact. Overall vulnerability for the planning area could result in multiple deaths during extreme contamination events.

# Dam Failure

# Hazard Profile

Dams are water storage, control, or diversion structures that impound water upstream in reservoirs. Dam failure can take several forms, including a collapse of or breach in the structure. While most dams have storage volumes small enough that failures have few or no repercussions, dams storing large amounts can cause significant flooding downstream. Dam failures can result from any one or a combination of the following causes:

- Prolonged periods of rainfall and flooding, which cause most failures;
- Inadequate spillway capacity, resulting in excess overtopping of the embankment;
- Internal erosion caused by embankment or foundation leakage or piping;
- Improper maintenance, including failure to remove trees, repair internal seepage problems, or maintain gates, valves, and other operational components;
- Improper design or use of improper construction materials;
- Failure of upstream dams in the same drainage basin;
- Landslides into reservoirs, which cause surges that result in overtopping;
- High winds, which can cause significant wave action and result in substantial erosion;
- Destructive acts of terrorism; and,
- Earthquakes, which typically cause longitudinal cracks at the tops of the embankments, leading to structural failure.

Benefits provided by dams include water supplies for drinking, irrigation and industrial uses, flood control, hydroelectric power, recreation, and navigation. At the same time, dams also represent a risk to public safety. Dams require ongoing maintenance, monitoring, safety inspections, and sometimes even rehabilitation to continue safe service.

# Location

For dams in Jefferson County, location, volume, elevation, condition, and classification information were factored into the risk ranking in Figure A-2, which illustrates general locations for each dam in the area. Currently, there are 3 dams located in Jefferson County and all 3 are classified as "low-hazard" dams. The dams are listed in Table A-3, along with regulation information.



### Figure A-2. Dam Locations in Jefferson County

Jefferson County | Hazard Mitigation Plan Update 2017 | Page 9

JURISDICTION	DAM NAME	HEIGHT (ft.)	STORAGE (Acre ft.)	POTENTIAL HAZARD CLASSIFICATION	EXTENT
Port Arthur	Port Arthur Raw Water Reservoir Levee	14	300	Low	No Impact
Jefferson County	McBride Lake Levee	7	450	Low	No Impact
Jefferson County	Spindletop Weir Saltwater Barrier	12	375	Low	No Impact

#### Table A-3. Jefferson County Dam Survey

# Extent

The extent or magnitude of a dam failure event is described in terms of the classification of damages that could result from a dam's failure, not the probability of failure. Table A-4 represents the average extent or magnitude of a dam failure event that could be expected for the Jefferson County planning area, including all participating jurisdictions. The "Extent Classification" column was determined by taking the average of dams in the jurisdiction and weighing low hazard dams as a 1, significant hazard dams as a 2, and high hazard dams as a 3 based on the potential severity, warning time, and duration.

JURISDICTION	DAMS & CLASSIFICATION	EXTENT CLASSIFICATION	LEVEL OF INTENSITY TO MITIGATE
Jefferson County	2 – Low	None	The county has 2 low hazard dams with limited storage capacity. Loss of life is not expected and any economic loss would be negligible.
Beaumont	None	None	There are no dams or inundation areas located within the city limits.
Bevil Oaks	None	None	There are no dams or inundation areas located within the city limits.
China	None	None	There are no dams or inundation areas located within the city limits.
Groves	None	None	There are no dams or inundation areas located within the city limits.
Nederland	None	None	There are no dams or inundation areas located within the city limits.
Nome	None	None	There are no dams or inundation areas located within the city limits.
Port Arthur	1 – Low	None	The city has 1 low hazard dam with limited storage capacity. Loss of life is not expected and any economic loss would be negligible.
Port Neches	None	None	There are no dams or inundation areas located within the city limits.
### **Historical Occurrences**

The State of Texas has not experienced loss of life or extensive economic damage due to a dam failure since the first half of the 20<sup>th</sup> century. However, there may be many incidents that are not reported and, therefore, the actual number of incidents is likely to be greater.

There has not been a recorded dam failure event for the entire Jefferson County planning area, including all participating jurisdictions.

### Probability of Future Events

No historical events of dam failure have been recorded in the Jefferson County planning area, though the risk of dam failure is monitored closely. Due to the lack of historical occurrences, the probability of a future event is unlikely, meaning an event is possible in the next 10 years.

### Vulnerability and Impact

There are 3 dams in the Jefferson County planning area, and all 3 dams are considered low hazard dams. Low hazard dams are those at which failure or mis-operation probably would not result in loss of human life and cause limited economic and/or environmental losses. Damage to agriculture and structures near both dams is considered negligible due to the small size and limited capacity of each dam.

The potential severity of a dam failure in the planning area, including the SETRPC and all participating jurisdictions, would be "Limited." As a result, a dam breach could result in injuries that are treatable with first aid, with facilities being shut down for 24 hours or less, and less than 10 percent of property destroyed or damaged.

## Hazardous Materials Incident (Fixed and Mobile)

### Hazard Profile

In a hazardous materials incident, solid, liquid, and/or gaseous contaminants may be released from fixed or mobile containers, although this profile focuses on fixed sites. Weather conditions will directly affect how the hazard develops.

The location of the most concentrated and potentially hazardous materials in the planning area are: fixed industrial facilities including oil and gas wells and storage facilities, pipelines, large and small industrial complexes that use or process chemicals or petroleum products, highways, and railroads. Numerous other sources are also present across the planning area, including storage areas for insecticides, herbicides, and fertilizers, wrecking yards, retail fueling stations, and abandoned industrial facilities. Within regard to pipeline locations, roughly one third (1/3) of the 367,000 linear miles of pipelines transporting hazardous materials in the State of Texas are located in the southeast region of the state. This concentration of pipelines in the region that includes Jefferson County relates to a corresponding high probability of hazardous material transport accidents.

The Toxics Release Inventory (TRI) is a publicly available database from the federal Environmental Protection Agency (EPA) that contains information on toxic chemical releases and other waste management activities reported annually by certain covered industry groups, as well as federal facilities. This inventory was established under the Emergency Planning and Community Right-to-Know Act of 1986 (EPCRA) and expanded by the Pollution Prevention Act of 1990. Each year, facilities that meet certain activity thresholds must report their releases and other waste management activities for listed toxic

chemicals to EPA and to their state or tribal entity. A facility must report if it meets the following three criteria:

- The facility falls within one of the following industrial categories: manufacturing; metal mining; coal mining; electric generating facilities that combust coal and/or oil; chemical wholesale distributors; petroleum terminals and bulk storage facilities; RCRA Subtitle C treatment, storage and disposal (TSD) facilities; and solvent recovery services.
- Have 10 or more full-time employee equivalents.
- Manufactures or processes more than 25,000 pounds or otherwise uses more than 10,000 pounds of any listed chemical during the calendar year. Persistent, bioaccumulative and toxic (PBT) chemicals are subject to different thresholds of 10 pounds, 100 pounds or 0.1 grams, depending on the chemical.

Tier 2 data is a publicly available database from the Texas Department of State Health Services Tier 2 Chemical Reporting Program. Under the community right-to-know program laws upheld at the state and federal level, all facilities which store significant quantities of hazardous chemicals must share this information with state and local emergency responders and planners. Facilities in Texas share this information by filing annual hazardous chemical inventories with the state, with Local Emergency Planning Committees (LEPCs) and with local fire departments. The Texas Tier 2 Reports contain facility identification information and detailed chemical data about hazardous chemicals stored at the facility.

A facility must report if it meets the following criteria:

- Any company using chemicals that could present a physical or health hazard must report them, according to Tier 2 requirements.
- If an industry has an OSHA deemed hazardous chemical that exceeds the appropriate threshold at a certain point in time, that chemical must be reported. These chemicals may be on the list of 356 Extremely Hazardous Substances (EHS) or could be one of the 650,000 reportable hazardous substances (not on the EHS list). This reporting format is for a "snapshot in time." EHS chemicals have to be reported if the quantity is either greater than 500 pounds, or if the Threshold Planning Quantity (TPQ) amount is less than 500 pounds.

### Location

The locations of available TRI and Tier 2 toxic sites in the Jefferson County planning area are shown below in Table A-5.

JURISDICTION	FACILITY NAME	ADDRESS	NUMBER OF CHEMICALS
JEFFERSON COUNTY	CHEMTREAT INC	4200 TWIN CITY HWY	2
JEFFERSON COUNTY	CHEMOURS BEAUMONT PLANT	5470 N TWIN CITY HWY	14
JEFFERSON COUNTY	LUCITE INTERNATIONAL INC	6350 N TWIN CITY HWY	11

#### Table A-5. Toxic Sites in Jefferson County<sup>3</sup>

<sup>&</sup>lt;sup>3</sup> Source: EPA Toxic Release Inventory

JURISDICTION	FACILITY NAME	ADDRESS	NUMBER OF CHEMICALS
JEFFERSON COUNTY	SASOL CHEMICALS (USA) LLC	HWY 124 & ROLLINS RD	2
JEFFERSON COUNTY	PHILLIPS 66 PIPELINE LLC- BEAUMONT TERMINAL	HWY 366 1/2 MILE E OF INTERSECTION W HWY 347	14
BEAUMONT	MARTIN OPERATING PTNR-NECHES	1 GULF STATES RD	4
BEAUMONT	MARTIN OPERATING PTNR- STANOLIND	10 SULFUR PLANT RD	4
BEAUMONT	EASTHAM FORGE INC.	1050 NECHES ST.	5
BEAUMONT	GE WATER & PROCESS TECHNOLOGIES BEAUMONT FACILITY	10658 HWY 90	12
BEAUMONT	GOODYEAR TIRE & RUBBER CO	11241 INTERSTATE HWY 10	18
BEAUMONT	MOBIL CHEMICAL CO BEAUMONT POLYETHYLENE PLANT	11440 HWY 90	8
BEAUMONT	TOTAL PETROCHEMICALS & REFINING USA INC-BEAUMONT PLANT	11455 INTERSTATE HWY 10	2
BEAUMONT	TXI OPERATIONS LP-SOUTH READY MIX PLANT	1250 E FLORIDA ST	1
BEAUMONT	CHEMTRADE REFINERY SERVICES INC	1400 OLIN RD	2
BEAUMONT	BASF CORP - BEAUMONT	14385 W PORT ARTHUR RD	17
BEAUMONT	COLONIAL TANK FARM	14713 W PORT ARTHUR RD	9
BEAUMONT	GERDAU AMERISTEEL US INC- BEAUMONT WIRE OPERATIONS	220 AVE A	1
BEAUMONT	TXI OPERATIONS LP-DOLLINGER READY MIX PLANT	2525 DOLLINGER	1
BEAUMONT	DRAGON WESPINE FACILITY	2609 WESPINE RD	1
BEAUMONT	LNVA-NORTH REGIONAL TREATMENT PLANT	2655 GULF STATES RD	24
BEAUMONT	EXXONMOBIL OIL CORP BEAUMONT CHEMICAL PLANT	2775 GULF STATES RD	43
BEAUMONT	ARKEMA INC	2810 GULF STATE RD	11
BEAUMONT	AZZ GALVANIZING SERVICES- BEAUMONT	500 INDUSTRIAL RD	2
BEAUMONT	FEDERAL CORRECTIONAL COMPLEX	5430 KNAUTH RD	1
BEAUMONT	NOV XL SYSTEMS BEAUMONT	5780 HAGNER RD	2
BEAUMONT	COASTAL CHEMICAL CO LLC	6534 INDUSTRIAL ROAD	5
BEAUMONT	GULFCO FORGE & MACHINE	6817 INDUSTRIAL RD	3
BEAUMONT	CB&I-BEAUMONT	850 PINE ST	7

JURISDICTION	FACILITY NAME	ADDRESS	NUMBER OF CHEMICALS
BEAUMONT	OHMSTEDE LTD	895 N MAIN ST	4
BEAUMONT	EXXONMOBIL OIL BEAUMONT REFINERY	E END OF BURT ST	43
PORT ARTHUR	AIR PRODUCTS LLC	1801 S GULFWAY DR	43
PORT ARTHUR	TXI PORT ARTHUR READY MIX	2.36 M FROM THE INTERX OF GULFWAY DR & HWY 82	1
PORT ARTHUR	CHEVRON PHILLIPS CHEMICAL CO	2001 S GULFWAY DR	21
PORT ARTHUR	STANDARD ALLOYS INC (PORT ARTHUR)	201 W LAKESHORE DR	2
PORT ARTHUR	KMTEX LLC	2450 S GULFWAY DR	16
PORT ARTHUR	MOTIVA-PORT ARTHUR REFINERY	2555 SAVANNAH AVE	31
PORT ARTHUR	OXBOW CALCINING LLC	3901 COKE DOCK RD	7
PORT ARTHUR	MOTIVA-PORT ARTHUR TERMINAL	3901 TEXACO ISLAND RD	11
PORT ARTHUR	FLINT HILLS RESOURCES PORT ARTHUR LLC	4241 SAVANNAH AVE	18
PORT ARTHUR	TEAM FABRICATORS	650 MAIN AVENUE	1
PORT ARTHUR	TOTAL PETROCHEMICALS & REFINING USA INC-PORT ARTHUR REFI	7600 32ND ST	24
PORT ARTHUR	VEOLIA ES TECHNICAL SOLUTIONS LLC PORT ARTHUR FACILITY	HWY 73, 3.5 MILES W OF TAYLOR BAYOU	207
PORT ARTHUR	BASF TOTAL PETROCHEMICALS LLC	NE OF INTERSECTION OF HWY 73 & HWY 366	24
PORT NECHES	LION ELASTOMERS LLC	1615 MAIN ST	5
PORT NECHES	TPC GROUP	2102 SPUR 136	14
PORT NECHES	AIR LIQUIDE LARGE INDUSTRIES US LP PORT NECHES	2121 PARK ST	1
PORT NECHES	CALABRIAN CORP	5500 HWY 366	1
PORT NECHES	HUNTSMAN PETROCHEMICAL LLC PORT NECHES FACILITY	6001 HWY 366	35
PORT NECHES	MOTIVA PORT NECHES TERMINAL	CORNER SPUR 136 & GRIGSBY DR	7

#### Extent

From a hazardous materials incident, the micro-meteorological effects of the buildings and terrain can alter travel and duration of agents. Shielding in the form of sheltering-in-place can protect people and property from harmful effects. Non-compliance with fire and building codes, as well as failure to maintain existing fire and containment features can substantially increase the damage from a hazardous materials

release. The duration of a hazardous materials incident can range from hours to days. Warning time for hazardous materials incidents is minimal to none.

### **Previous Occurrences**

Hazardous materials are substances which if released or misused can cause death, serious injury, longlasting health effects, and damage to structure and other properties as well as to the environment. Many products containing hazardous chemicals are used and stored in homes routinely. These products are also shipped daily on the nation's highways, railroads, waterways, and pipelines.

A total of 126 transportation incidents have been reported in the Jefferson County planning area over the last 67 years. The data collected is from 1950 to 2016 and identifies the hazardous materials transportation incidents as in-transit, loading, and unloading of transport vehicles. A summary of reported events are listed in Table A-6 below by jurisdiction.

JURISDICTION	NUMBER OF INCIDENTS	INJURIES	FATALITIES	PROPERTY AND CROP DAMAGE
Beaumont	1	0	0	\$0
Bevil Oaks	0	0	0	\$0
China	1	0	0	\$440
Groves	0	0	0	\$0
Nederland	18	0	0	\$99
Nome	2	0	0	\$70,585
Port Arthur	169	23	0	\$372,529
Port Neches	21	9	1	\$7,612
Jefferson County	11	0	0	\$135,270
TOTAL LOSSES	223	32	1	\$586,535

#### Table A-6. Jefferson County Hazardous Material Incident Events by Jurisdiction<sup>4</sup>

### Probability of Future Events

Based on the historic incident records, the frequency of occurrence is highly likely and an event is probable in the next year in the Jefferson County planning area.

### Vulnerability and Impact

Hazardous materials or toxic releases can have a "substantial" impact. Such events can cause multiple deaths, completely shut down facilities for 30 days or more, and cause more than 50 percent of affected properties to be destroyed or suffer major damage.

<sup>&</sup>lt;sup>4</sup> Damages reported in 2016 dollars.

## Terrorism

### Hazard Profile

The Federal Bureau of Investigation (FBI) categorizes terrorism in the United States as one of two types domestic terrorism or international terrorism. Domestic terrorism involves groups or individuals whose terrorist activities are directed at elements of our government or population without foreign direction. International terrorism involves groups or individuals whose terrorist activities are foreign-based and/or directed by countries or groups outside of the United States, or whose activities transcend their national boundaries.

A terrorist attack can take several forms, depending on the technological means available to the terrorist, the nature of issue motivating the attack, and the points of weakness of the terrorist's target. Bombings are the most frequently used terrorist method in the United States. A terrorist using a chemical or biological weapon is of particular concern to officials. Special training and equipment is needed in order to safely manage a Weapons of Mass Destruction incident.

Biological agents are infectious microbes or toxins used to produce illness or death in people, animals or plants. Biological agents can be dispersed as aerosols or airborne particles. Terrorists may use biological agents to contaminate food or water, as they are extremely difficult to detect.

Chemical agents kill or incapacitate people, destroy livestock, or ravage crops. Some chemical agents are odorless and tasteless and are therefore difficult to detect. These chemical agents can have an immediate effect (a few seconds to a few minutes) or a delayed effect (several hours to several days).

The Department of Defense estimates that as many as 26 nations may possess chemical agents and/or weapons, and an additional 12 may be seeking to develop them. The Central Intelligence Agency reports that at least 10 countries are believed to possess or are currently conducting research on biological agents for weaponization.

Terrorist incidents – as with other natural and technological disasters – involve the application of one or more modes of harmful force to the built environment. These modes include contamination (as in the case of chemical, biological radiological or nuclear hazards), energy (explosives, arson, and even electromagnetic waves), or denial of service (sabotage, infrastructure breakdown, and transportation service disruption).

### Location

There is no distinct geographic boundary to the threat of terrorism. An event is possible throughout the Jefferson County planning area.

### Extent

The Homeland Security Advisory System, issued by the U. S. Department of Homeland Security, previously used a color-coded terrorism warning system that identified five threat levels. In 2011, the Department of Homeland Security (DHS) replaced the color-coded alerts of the Homeland Security Advisory System (HSAS) with the National Terrorism Advisory System (NTAS), designed to more effectively communicate information about terrorist threats by providing timely, detailed information to the American public.

NTAS now consists of two types of advisories: Bulletins and Alerts. DHS has added Bulletins to the advisory system to be able to communicate current developments or general trends regarding threats of terrorism.

NTAS Bulletins permit the Secretary to communicate critical terrorism information that, while not necessarily indicative of a specific threat against the United States, can reach homeland security partners or the public quickly, thereby allowing recipients to implement necessary protective measures. Because DHS may issue NTAS Bulletins in circumstances not warranting a more specific warning, NTAS Bulletins provide the Secretary with greater flexibility to provide timely information to stakeholders and members of the public.

When there is specific, credible information about a terrorist threat against the United States, DHS will share an NTAS Alert with the American public when circumstances warrant doing so. The Alert may include specific information, if available, about the nature of the threat, including the geographic region, mode of transportation, or critical infrastructure potentially affected by the threat, as well as steps that individuals and communities can take to protect themselves and help prevent, mitigate or respond to the

threat. The Alert may take one of two forms: Elevated, if there is credible threat information, but only general information about timing and target such that it is reasonable to recommend implementation of protective measures to thwart or mitigate against an attack; or Imminent, if the threat is believed credible, specific, and impending in the very near term. Terrorism Advisory System Alerts are described in Figure A-3.<sup>5</sup>

The Red Cross also issues Advisory System Recommendations for individuals, families, neighborhoods, schools and businesses for each alert level. These may be found at: *www.redcross.org*.

Heightened periods for terrorism risk are based on intelligence and other information. A potential terrorist event could devastate the community physically, economically and psychologically for many years to come. Warning time for terrorism is minimal to none.

#### **Previous Occurrences**

The history of terrorism on United States soil includes the attacks of September 11, 2001, on the World Trade Center in New York and the Pentagon in Washington, D.C. and the ensuing anthrax attacks; the 1995 bombing of the Murrah Federal Building in Oklahoma City; and the bombing of the World Trade Center in 1993.

Jefferson County has not experienced a terrorist act. While complete prevention of an attack may not be attainable, the County can lessen the likelihood and/or the potential effects of an incident. The County continues to improve its readiness to respond to a terrorist incident through participation in state and federal programs that provide training and equipment for agencies that would respond to a local terrorist incident, and in exercises that help to improve agency coordination and test local response plans.

#### Figure A-3. National Terrorism Advisory System



<sup>&</sup>lt;sup>5</sup> Source: Department of Homeland Security, https://www.dhs.gov/national-terrorism-advisory-system

### Probability of Future Events

The types, frequencies, and locations of many natural hazards are identifiable and, even in some cases, predictable, as the laws of physics and nature govern them. Malevolence, however, cannot be forecast with any accuracy. There is, therefore, some potential for most, if not all, types of intentional terrorist acts to occur anywhere and at any time.

### Vulnerability and Impact

There is no defined geographic boundary for a terrorist event. All of the population, buildings, critical facilities, infrastructure and lifelines and hazardous materials facilities are considered exposed to the hazards of terrorism and could potentially be affected.

There are no past local events. Therefore, all assets and facilities are potentially at risk to damages that may, for the most part, be secondary.

Terrorist events can have a "substantial" severity of impact. They can cause multiple deaths, completely shut down facilities for 30 days or more, and cause more than 50 percent of affected properties to be destroyed or suffer major damage.

# **APPENDIX B: PLANNING TEAM**

Planning Team Members	.1
Stakeholders	. 2

# Planning Team Members

The Jefferson County Hazard Mitigation Plan Update 2017 (Plan or Plan Update), was organized using a direct representative model. An Executive Planning Team from Jefferson County and participating jurisdictions, shown in Table B-1, was formed to coordinate planning efforts and request input and participation in the planning process. Table B-2 reflects the Advisory Planning Team, consisting of representatives from area organizations and departments of the jurisdictions that participated throughout the planning process. Table B-3 is comprised of member Stakeholders who were invited to attend meetings to provide Plan Update input. The public were also invited to participate throughout the planning process. Public outreach efforts and meeting documentation is provided in Appendix E.

ORGANIZATION/JURISDICTION	TITLE
Jefferson County	Emergency Management Coordinator
City of Beaumont	Emergency Management Coordinator
City of Bevil Oaks	Mayor/ Floodplain Manager
City of China	Mayor
City of Groves	Emergency Management Coordinator
City of Nederland	Emergency Management Coordinator
City of Nome	Mayor
City of Port Arthur	Senior Planner
City of Port Neches	Emergency Management Coordinator
SETRPC	Homeland Security and Emergency Management Planning Director

#### Table B-1. Executive Planning Team

#### Table B-2. Advisory Planning Team

ORGANIZATION/JURISDICTION	TITLE
City of Beaumont	Communications Manager
City of Beaumont	Emergency Management Assistant

### Appendix B: Planning Team

ORGANIZATION/JURISDICTION	TITLE
City of Beaumont	Emergency Management Specialist
City of Beaumont	Director of Technology Services
City of Beaumont Police Department	Assistant Chief
City of China	City Secretary
City of Nederland Police Department	Assistant Chief
City of Nome	City Secretary
City of Port Arthur	Senior Planner
City of Port Arthur Development Services	Director
City of Port Arthur Fire Department	Emergency Management Coordinator
City of Port Arthur Police Department	Emergency Management Coordinator
City of Port Neches Fire Department	Emergency Management Coordinator
City of Taylor Landing	Mayor
Jefferson County	Assistant Emergency Management Coordinator
SETRPC	Regional Emergency Planner

# Stakeholders

The following groups listed in Table B-3 represent a list of Jefferson County organizations who attended meetings, public meetings and workshops throughout the planning process. Invited organizations and stakeholders participated and were integral to providing comments and data for the Plan Update. For a list of attendees at meetings, please see Appendix F<sup>1</sup>.

#### Table B-3. Stakeholders

AGENCY/ORGANIZATION	TITLE
Colonial Pipeline	Manager
Lamar University	Assistant Professor
Local Emergency Planning Committee	Chairperson
Jefferson County Drainage District 7	Graduate Engineer
Jefferson County Drainage District 7	Supervisor

<sup>&</sup>lt;sup>1</sup> Information contained in Appendix F is exempt from public release under the Freedom of Information1 Act (FOIA).

# Appendix B: Planning Team

AGENCY/ORGANIZATION	TITLE
RPS	Senior Consulting Engineer
South East Texas Disaster Recovery Group	Executive Director
Texas House of Representatives	Texas US Representative
Texas State Senate	Texas State Senator
United Way	Executive Director
City of Kountze	Emergency Management Coordinator
City of Lumberton	City Manager
City of Rose Hill Acres	Mayor
City of Silsbee	Emergency Management Coordinator
City of Silsbee	Assistant Emergency Management Coordinator
City of Sour Lake	City Manager
City of Sour Lake	Police Chief
Hardin County	Emergency Management Coordinator
Hardin County	Floodplain Administrator
South East Texas Regional Planning Commission	Homeland Security and Emergency Management Planning Director
City of Bridge City	Emergency Management Coordinator
City of Orange	Deputy Chief/Emergency Management Coordinator
City of Pinehurst	Emergency Management Coordinator
City of Pine Forest	Emergency Management Coordinator
City of Rose City	City Secretary
City of Vidor Police Department	Emergency Management Coordinator
City of West Orange	Emergency Management Coordinator
Orange County	Tax Assessor-Collector
Orange County Office of Emergency Management	Deputy Emergency Management Coordinator
Orange County Office of Emergency Management	Emergency Management Coordinator

# **APPENDIX C: PUBLIC SURVEY RESULTS**

Overview	1
Public Survey Results	2

# Overview

Jefferson County prepared a public survey that requested public opinion on a wide range of questions relating to natural hazards. The survey was made available on websites including the Jefferson County Office of Emergency Management Facebook page as well as the SETRPC webpage. This survey link was also distributed at public meetings throughout the planning process.

A total of 69 surveys were collected, the results of which are analyzed in Appendix C. The purpose of the survey was twofold: 1) to solicit public input during the planning process, and 2) to help the jurisdictions identify any potential actions or problem areas.

The following survey results depict the percentage of responses for each answer. Similar responses have been summarized for questions that did not provide a multiple-choice answer or that required an explanation.

# Public Survey Results

1. Please state the jurisdiction (city and community) where you reside.



2. A. Have you ever experienced or been impacted by a disaster?



2. B. If "Yes", please explain:



3. How concerned are you about the possibility of your community being impacted by a disaster?



4. Please select the one hazard you think is the highest threat to your neighborhood:



5. Please select the one hazard you think is the second highest threat to your neighborhood:



6. A. Are there hazards not listed above that you think is a wide-scale threat to your neighborhood?



6. B. If "Yes", please explain:



7. Is your home located in a floodplain?



8. Do you have flood insurance?



9. If you do not have flood insurance, why not?



10. A. Have you taken any actions to make your home or neighborhood more resistant to hazards?



10. B. What have you done?



11. Are you interested in making your home or neighborhood more resistant to hazards?





12. A. What is the most effective way for you to receive information about how to make your home and neighborhood more resistant to hazards?

12. B. If other, please specify.



13. In your opinion, what are some steps your local government could take to reduce or eliminate the risk of future hazard damages in your neighborhood?



14. Are there any other issues regarding the reduction of risk and loss associated with hazards or disasters in the community that you think are important?



15. A number of community-wide activities can reduce our risk from hazards. In general, these activities fall into one of the following six broad categories. Please tell us how important you think each one is for your community to consider pursuing.



Prevention / Local Plans & Regulations - Administrative or regulatory actions that influence the way land is developed and buildings are built. Examples include planning and zoning, building codes, open space preservation, and floodplain regulations.

Property Protection - Actions that involve the modification of existing buildings to protect them from a hazard or removal from the hazard area. Examples include acquisition, relocation, elevation, structural retrofits, and storm shutters.

Natural Resource Protection - Actions that in addition to minimizing hazard losses also preserve or restore the functions of natural systems. Examples include: floodplain protection, habitat preservation, slope stabilization, riparian buffers, and forest management.

Structural Projects - Actions intended to lessen the impact of a hazard by modifying the natural progression of the hazard. Examples include dams, levees, seawalls detention / retention basins, channel modification, retaining walls and storm sewers.

Emergency Services - Actions that protect people and property during and immediately after a hazard event. Examples include warning systems, evacuation planning, emergency response training, and protection of critical facilities or systems.

Public Education and Awareness - Actions to inform citizens about hazards and techniques they can use to protect themselves and their property. Examples include outreach projects, school education programs, library materials and demonstration events.

# **APPENDIX D: CRITICAL FACILITIES**

Overview	1
Critical Facilities	2

# Overview

This Appendix is **For Official Use Only (FOUO)** and may be exempt from public release under Freedom of Information Act (FOIA). Figures D-1 through D-9 locates all critical facilities that were included in the risk assessment. Mapped facilities were provided by Jefferson County Planning Team members. Table D-1 notes the critical facilities by type.

# **Critical Facilities**



Figure D-1. Critical Facilities in Jefferson County

ТҮРЕ	NUMBER		
Fire Stations	10		
Police Stations	6		
Hospitals	9		
Schools	68		
Port/Authority	4		
SETRPC	1		
Airports	1		

### Table D-1. Critical Facilities by Type in Jefferson County



Figure D-2. Critical Facilities in City of Beaumont



Figure D-3. Critical Facilities in City of Bevil Oaks



#### Figure D-4. Critical Facilities in City of China



Figure D-5. Critical Facilities in Groves



Figure D-6. Critical Facilities in City of Nederland



Figure D-7. Critical Facilities in Nome







Figure D-9. Critical Facilities in City of Port Neches

# **APPENDIX E: DAM LOCATIONS**

Overview	.1
Dam Locations	.1

## Overview

Appendix E is **For Official Use Only (FOUO)** and may be exempt from public release under the Freedom of Information Act (FOIA).

Table E-1 below reflects all dams that are located in Jefferson County. This list includes High, Significant, and Low Hazard Dams.

## **Dam Locations**

JURISDICTION	LATITUDE	LONGITUDE	HEIGHT (Ft.)	STORAGE (Acre Feet)
Jefferson County	29.90444	-93.96833	14	300
Jefferson County	29.67051	-94.34415	7	450
Jefferson County	29.7146	-94.31468	12	375

#### Table E-1. Listing of Jefferson County Dam Locations and Storage Capacities

# **APPENDIX F: MEETING DOCUMENTATION**

Workshop Documentation	1
Public Meeting Documentation	8
Public Notices	9

# Workshop Documentation

Appendix F is **For Official Use Only (FOUO)** and may be exempt from public release under the Freedom of Information Act (FOIA).

Jefferson County held a series of Planning Team workshops: a Kickoff Workshop on March 30, 2016, a Risk Assessment Workshop on June 1, 2016, and a Mitigation Workshop on August 24, 2016. At each of these workshops members of the Planning Team were informed of the planning process, expressed opinions, and volunteered information. SETPRC hosted three public meetings (one following each workshop). The sign-in sheets for each workshop and public meeting are included below. For more details on the workshops and planning process, see Section 2.

#### Figure F-1. SETPRC Kickoff Workshop, 03.30.16



#### REGISTRATION

MEETING: Mitigation Action Plan Kick-Off Meeting

LOCATION: SETRPC – Homer E Nagel Conference Room 2210 Eastex Freeway Beaumont, Texas 77703 DATE: Wednesday, March 30, 2016 1:30pm

#### PLEASE PRINT AND WRITE LEGIBLY

NAME:	TITLE:	AGENCY:	ADDRESS:	E-MATL ADDRESS	
1111	Depity Firecly	PortArshur		L-MAIL ADDRESS	PHONE NUMBER:
Robert 2 14Vens	Sm '	Fire Dept	300 WACD	vhavens @ portstolly was	169 5825724
June Guard	MANCASER	Sourchake	625 HWW 103W SOUK	1 PROVER & Coty of Sour Like Com	-01 105015
DAVE L. SHOLD	Chief	Vider PD	1 - ponc	del aus Raitatil	409-287-35/3
thrisserves	Director	ocuc1	460 EBNINGE VIEOR TY	Kerrectional and	X07-167-406/
Jacob Rollins	Graduate Engineer	DD7		irollins @ tia provingering com	409-659-2669 EXT2
JON SHERWIN	PUBLIC WORKS SUPERVISOR	CITY OF WEST ORANGE	2700 AUSTIN AVE. WEST DRAWLE 77622	isherwice a last and a second	101 057 1447
Amanda Young	Hardin County	HardinCounty	300 Mourse Kount 20TH.	6 Cano de Vanalia listi des	(409) 639-4992 (162 666-889)
Glenda Piazza	Eng Mont Asst	Beaumont?	700 Orleans 11701	POinzza BBernan ITan	407-091-2070
LeeAnne Brown	Deputy Chiefler.	cit of Drange	Eal in TH De i sain	JE IL OW - INAUMONT VERAS. GOV	407-780-1280
			501 10. 1. Vrange 77670	lbrownporange fd. com	409.988.3323

South East Texas Regional Planning Commission – Registration Sign-In Sheet

NAME:	TITLE:	AGENCY:	ADDRESS:		
John Allans	Deputy Chief	P. JAH DI	645 Fourth Street	E-MAIL ADDRESS:	PHONE NUMBER:
- cc 1 A .	OT Police FEAC	VOST METHUS PL	P.A. TX, 77641	jowens@postarthurpd.com	409-983-8616
Jett halonb	EME	City of Kout	1025 W Pine 11	SIKCh @Shach but	1 700-8990
Tod MSDAWAU	Capt/Emc		2290 TCX45 AVE.	VINCED DE SIDDE M	110 017-
P , M		City of Bridge	BR. Jec C. H, D	Imcdowell DERidgeeinged. Lon	408-725-5025
Soyact Chypang	Fre Chief	OCESD#7	Vin 1/2	chief 2@ ocesd 1. com	409-769-8294
Kick FHomisee	Mayor	Rose VIII Acres	100 Jordan Ad	- 1+ hemisea Byma	454 6593
Larry Squrage	Chier	Sour Lake PD	625 Hay 105W	al la	907 755 4740
and Bu (	607	PIA II	UL ACL DA	Chief 301 @ Oitget Sourfale - Con	409 287 2059
LODY L. CL	OK. Monner	lor1 Arthur	444-4-57 14 77640	pzel proun portorthurt	Goi 783-8139
ent LAFleur	Comm MANJOR	C.T. of BMI-		TIGFIEUR ChODINGATTENAS	1 4106737126
naurenninoles	EN STOR	Citing Bard	100 Orleans	TETTICAN COEAGINIONT TE XTTS. GOL	10701)1020
D T II	on spec.	Ligg KMIT		SNobles@beaumontteras.gov	409-980-7285
Dart Dartkowiak	CTO	COR	801 Man	bbatkunzk@ba Ha	\$80-7767
Tim Ocurs diel	Enc	Bernart	700 polana	- LIC /	gov 000-313C
neresa Wicler	Emc.	Hardin Co	300 W. Momoe	Theresa Wicker @	1900 F50 -3804
1 111	2	r an later of	Koutzetx	eo.hardin.tx.us	246-5119
Son 161/13	Em	NER	12462 Box Noly	ghalling al she in	200
Seg Fountul	EMC	Jeffenn	1149 Poor 57.	Come Dick And Consert The St	123 1529
La Puelle	h	TIL	wrut, 1x 11701	gtoundain@ co Jeffessu. Tx.us	625-8750
0 11	"hir,	Morting			
raul NU/San	Fire well For	Port Helling ED	box When 1 1 1.		608-72 -

South East Texas Regional Planning Commission - Registration Sign-In Sheet
NAME:	TITLE:	AGENCY:	ADDRESS:	F-MATL ADDRESS	
Jim Clay	Asst Chief	Bont PD	255 Collige Bont, Th	jelac @ Beaument Texas.	900 409-880-3881
Kon Bullion	Diretin De	Servy City of	Port Arthur 444 4th Sha	Row butow & Port port	12. Jay 409-983-8136
			119 15 FM 1472 Wange	(1 moorela)-LO .Drg. 281X.VS	4046704135

South East Texas Regional Planning Commission – Registration Sign-In Sheet



1

H<sub>2</sub>O PARTNERS

#### SETRPC HAZARD MITIGATION PLAN KICKOFF Planning / Stakeholder Workshop South East Texas Regional Planning Commission, Beaumont, TX March 30, 2016

Name/Title	Department	Phone	Email
Rachel Andrews	HZO Parmers "	512-983-0092	rache 10/2000 trensusa.co.
CANY PORTAN ACON	Hederland PD	409-723-1517	Sporte a complete us
Kyan Peaking Emic	Orayz Contgo EM	407 228 0013	- peakody a to
AIKENAITE	JEFFORSON COUNTY	635-8757	MINE ITERCO- 6TTERSOND
Hope apon	City of Vidac	409-769-4561	alapenEcityofidoccom
Jerry Parket	Vidor ISD	409-951-8966	Sparker @ Vidorisd. are
Heidi Watson	H20 Pourtners	512 568 2759	heidi@hzopartnersusa.com

#### Figure F-2. SETPRC Risk Assessment Workshop, 06.01.16



SOUTH EAST TEXAS REGIONAL PLANNING COMMISSION

#### PLEASE PRINT AND WRITE LEGIBLY

#### REGISTRATION

MEETING: SETRPC Hazard Mitigation Planning Team Meeting

LOCATION: SETRPC – Homer E Nagel Conference Room 2210 Eastex Freeway Beaumont, Texas 77703 DATE: Wednesday, June 1, 2016 1:30pm

Risck Assessment Libritshop

Name	Title	Адепсу	Address	Email Address	Phone Number
Aleta Cappen	City of Vidor PD	City of Vidor	695 East Railroad, Vidor 77662	acappen@citvofvidor.com	409-769-4561
Amanda Moore	OCPH-PHEP	OCPH	11475 FM 1442 Orange	amoore@co.orange.tx.us	409-670-4135
Amanda Young (1/1/)	Hardin County Floodplain	Hardin County	300 Monroe Kountze 77625	amanda.voung@co.hardin.tx.us	409-651-8890
Bart Bartkowiak	СТО	City of Beaumont	801 Main	bbartkowiak@beaumonttexas.gov	880-3752
Bryant Champagne	Fire Chief	OCESD#1	2351 Hwy 12 Vidor	chief1@ocesd1.com	409-769-8294
Chris Serres	Finance Director	OCWC1	460 E Bolivar, Vidor TX	cserres@ocwc1.com	409-769-2269 ext.234
-Dave L. Shows	Chief	Vidor P.D.		dishows@cityofvidor.com	409-769-4561
Gary Collins	Fire Chief/EMC	Nederland	1404 Boston Ave. Neaderland	gcollins@ci.nederland.tx.us	723-1529
Gary Porter	Assistant Chief	Nederland PD		gporter@ci.nederland.tx.us	409-723-1517
Glenda Piazza	City of Bmt Emergency Management Assistant	Beaumont	700 Orleans 77701	gpiazza@BeaumontTexas.gov	409-980-7280
Greg Fountain	EMC	Jefferson County	1149 Pearl St. Beaumont TX 77701	gfountain@co.jefferson.tx.us	835-8757
Jack Provost	City Manager	City of Sour Lake	625 Hwy 105 W Sour Lake	jprovost@cityof sourlake.com	409-287-3573
Jacob Rollins	Graduate Engineer	DD7		jrollins@ljaengineering.com	409-659-1947
Jeff LaComb	EMC	City of Kountze	1025 W. Pine	jlkeh@sbcGlobal.net	790-8990
Jerry Parker	Chief	Vidor ISD	120 E Bolivar	iparker@vidorisd.org	409-951-8966
Jim Clay	Assistant Chief	Beaumont PD	225 College St. Beaumont TX	jclay@beaumonttexas.gov	409-880-3881

South East Texas Regional Planning Commission – Registration Sign-In Sheet

John Durkay	Mayor	Taylor Landing			527-3349
John Owens	Deputy Chief of Police/EMC	Port Arthur P.D.	645 Fourth Street, P.A. TX 77641	jowens@portarthurpd.com	409-983-8616
1			2700 Austin Ave, West Orange		
ion Sherwin	Public Works Supervisor	City of West Orange	77630	jsherwin@cityofwestorange.com	409-659-4992
Larry saurge	Chief	Sour Lake P.D.	625 Hwy 105 W Sour Lake	chief301@citvofsourlake.com	409-287-2059
LeeAnne Brown	Deputy Chief/EMC	City of Orange	501 N. 7th Orange 77630	lbrown@orangefd.com	409-988-3323
Mike White		Jefferson County		mwhite@co.jefferson.tx.us	835-8757
Paul Borwn	Sr. Planner	Port Arthur	444 4th St. P.A. 77640	paul.brown@portarthurtx.gov	983-8139
Paul Nelson	Fire Chief/EMC	Port Neches	606 Magnolia Ave	pnelson@ci.port-neches.tx.us	608-770-1529
Rick Thomisee	Mayor	Rose Hill Acres	100 Jordan Road	rithomisee@gmail.com	409-755-4940
Robert L. Havens	Deputy Fire Chief / EMC	Port Arthur Fire Department	300 Waco	rhavens@portarthur.com	409-583-8734
Ron Burton	Director	City of Port Arthur	444 4th 5t. P.A. 77640	Ron, burton@portarthur.tx.gov	409-983-8136
Ryan Peabody	EMC	Orange County OEM		rpeabody@co.orange.tx.us	409-228-0013
Shaqueena Nobles	EM Speer O	City of Beaumont	700 Orleans 77701	Snobles@beaumonttexas.gov	409-980-7285
Terry LaFleur	Community Manager	City of Beaumont		Tlafleur@beaumontTexas.gov	409-673-7636
Theresa Wigley	EMC	Hardin County	300 W. Monroe, Kountze TX	Theresa, Wigley@co.hardin.tx.us	246-5119
Tim Ocnaschek	EMC	Beaumont	700 Orleans 77701	tocnaschek@boaumonttexas.gov	880-3804
Tod McDowell TA	Capt. / EMC	City of Bridge City	2290 Texas Ave, Bridge City TX	tmcdowell@bridgecitypd.com	409-735-5028
A alo Jackson	Fore Chill / Emy	City of GOVES			100 100 3020
MIKEWHIE	DEMC	JEFFERSON COUNT	1149 PEARL ST	MUNTED CO HEARING	1-5-8757
Broken Bocker	Ry Chiop	Pauser land FSP #2	330 Black Bude Sta Tre	Resdachard at 201	Hoc - 652 - 100 1
MOGI HUBBINI	BREVE EMC	Orthog. County	1835 Jubs Minino	molthy I DA ARUN -Y IS	IN OR FAN
LARPY RICHAM	FIRE CHILDP	Port Operation	PEROKIOBE DA FRUM	wich a Que Tutrontela (car	4-1-11-11(1)
-Revised Appiveus	120 Poster Mithodom Sort Sist	· · · · · · · · · · · · · · · · · · ·	260 House Pour Rd - Austra	og let læk inger poercie a	NA 18-24082
LauerBrown	11.5 Contriun Palin	anna Countis	120W. Exponttre.	6 no how a Que I have no	1104 182 -5117
LISA KIEVES	@ IT	Orange County	SOI W. Division	WERTS P. M. Dranca to 115	409-771-67224
Rean Fuller	City of West Orange	Ocume County	2700 Western Ave.	dfulleral to at which are car	ung. w/2. 3/14
DON HARMON	DEANGE CO SHERIFF'S CORRE (CAPTAIN	ORANGE Co. SHEDIGE	295 BURDER DURANTE TO	Abaran Bar an art Thur	LAC 122 612-
Jessica Hill	Exec. Diva	DCEDL	123 S. Letter St. DAWON	in Conversition	412 -411-9-25
KERVIN KNAUP	CITY OF OF MAGE PLNG. DIE	CITY OF PERNICE	303N, 8TH, DEMES	2 trouge rowardy org	MAR 588-1120
Ronnie Hollien.	Jell ( PD / Supervisie.	Self. G. DD1	440194 Are	Thallier & dd Train	409 995 4019
REBERTSMETH	-DRANGE (U. F.SO. 1- 1 Spiller	2 OCES02	2351 Hurs 12	All la Conce full Com	1-9-716 0-61
	-			- mile + Ch ULE. 30 1. L. 3 M	40-1- (67)-DENA

South East Texas Regional Planning Commission ~ Registration Sign-In Sheet



SOUTH EAST TEXAS REGIONAL PLANNING COMMISSION

#### PLEASE PRINT AND WRITE LEGIBLY

### REGISTRATION

MEETING: SETRPC Hazard Mitigation Planning Team Meeting

LOCATION: SETRPC – Homer E Nagel Conference Room 2210 Eastex Freeway Beaumont, Texas 77703 DATE: Wednesday, June 1, 2016 1:30pm

Risk Assessment Warkshop

Name	Title	Agency	Address	Email Address	Phone Number
Berna Toligoz	Assistant Prof.	Lanor Vivria	FLAGAL PROPERTY BONZODZ	he con to have a langer chy	(109 480 Pais
SHAWN BE ASKER !!	ASSISTANT POLICE CHIEF	SILSOCE O	WAY CALER COU	Shlack will De la Calalia	49-1000005
MIKE OUSTRA	DER. FACA	BEAUMANT FOC	700 NO EASS FORT	Encustre (a, b any at low	a ana ana Cont
LUNDO COUNSTRIC	in ERANGELD Tax Assess-Coll	GEANGELS	PILBOX KLE PRANGE	CULUSTORED CO (b Ad TY US-	UNC 202-1929
MARK DAVIS	CHIEF & EMC.	SILSBEE	Company -	garant and Containing	104 00104 7 18 7
Toci Ardoin	Director Equipromond Heals	Remark Conn Ve	11474 Em 1442 Drover	ando in to co scence trees	429-745-144.2
BECKY FORD	MAYOR/CEM	City on Beard Oak	13378 ALASKAN DR -Pont	h ford 630 as here	414-1155-1475
KAY ANAVERIERE	REGIONA Emergence Planer	SETERC	2910 FAREFRILLY Brit	KASE with estrances	416-924-2512
Heich Watson	Mitigation Decisionist	120 Partiers	<u> </u>	hadre heardners in in	912-568-2259
	, i			interest in the second	
					~

South East Texas Regional Planning Commission - Registration Sign-In Sheet



### REGISTRATION

MEETING: SETRPC Hazard Mitigation Planning Team Meeting

LOCATION:	SETRPC – Homer E Nagel Co 2210 Fastex Freeway	nference Room
	Beaumont, Texas 77703	
DATE:	Wednesday, June 1, 2016	1:30pm

#### PLEASE PRINT AND WRITE LEGIBLY

Name	Title	Agency	Address	Email Address	Phone Number
or Ardoin	HR Director	Drange.	12352th St Dualog le	17.30 Jarder alan Alan at	V 11 11 11 11 12 70 1
			100000000000	1100 Carrier Carpinary	145 -141-061-276
<u>.</u>					
				1	
				· · · · · · · · · · · · · · · · · · ·	
	· · · · · · · · · · · · · · · · · · ·				
	1				
	1				r

South East Texas Regional Planning Commission - Registration Sign-In Sheet

### Figure F-3. SETPRC Mitigation Actions Workshop, 08.24.16





#### SETRPC HAZARD MITIGATION PLAN MITIGATION ACTIONS Planning / Stakeholder Workshop South East Texas Regional Planning Commission, Beaumont, TX August 24, 2016

Name/Title	Department	Phone	Email
Delaut 1 thereas	Deplenc Port Arthur Fis	4093441021	phavenes @ fortAothur. com
John Owens	Deputy Chief, EMC Port Arthur F.	20. 409-983-8616	Jowens Opertarthurpd. com
MICHAEL R. WHITE	DEMC JEFFERSON GUNTY	4098353757	MWHITE CO. LE FFERSON, TY.US
MIKE CUSTER I	DEMC CITY OF BEAUMONT	4092398233	mouster @ Bowmont Exas gov
Shoqueen Nobles	EMSPEC. City M. BIALMON	1 409-980-7285	Snobles Braumont + exas gov
TIMOCARASchele EN	K City of Bannat	880-3804 T	senschel @ bermant Terns, 900
DaleJackson E	MC City of Groves	960-5761	diackson Digiolesty, com
· Cons Celling 59	me Prog of Nechotik	723-1531 9	Collins on Li, Medenterd, TX, US
Aleta Cappos 5	at/Emc City of Vidor	409-238-4448 0	ill ppen acity of Vidor com
(HARLISS Cox	ORANGE COUNTY EMC	409-238-9169	CCOX D CO. ORANGE. TX, US
Connie Hollie	n Lott Co. DD7	409-985-4369	rholliereddiorg
Lee Anne Brown	City of Orange	409.8-409.988.7359	Ibrownoorangefd.com

SETRPC

1



#### SETRPC HAZARD MITIGATION PLAN RISK ASSESSMENT Planning / Stakeholder Workshop South East Texas Regional Planning Commission, Beaumont, TX August 24, 2016

Name/Title	Department	Phone 409 - 781 - 4440 cell 409 - 78 - 11175 cell	Email
Greg Foundain EMC	Jefferson County	409-835-8757	Spantwite Co Jette Str. Tx. M-
· Jan to Dynut City	NAME City of Sourchake	409-781-2525	IPROVOSTOCHYOS Souchate con
Rogald Kon Byéi	in City of Part with	409-983-8136	Von burton & portan tu 1x. gov
·	per ferrig mean		
			· · · · · · · · · · · · · · · · · · ·
·			
-			
2			

SETRPC



SETRPC HAZARD MITIGATION PLAN RISK ASSESSMENT Planning / Stakeholder Workshop South East Texas Regional Planning Commission, Beaumont, TX August 24, 2016

Name/Title	Department	Phone	Email
MURIC DAVIS POLICEC	hief-EMC City of Silsb	ee 385:3714	Mdavis City of S. Isbee. com
Shawn Blackwell	ASST CHIEF ASST ENC City of S.B.	bee 385-3714	sblackwell Catyofsilsbeeron
. Steven Stafford/Engr-	Supristendent Jefferron G. Engineering	835-8584	SStafford Co. jefferson tx. us
STEVE CLORK-C	ty of Lumisorion	755-0031	Sclarkegtbiz class, com
RhowDA Musphy	H20 Partwers	512-571-2088	RMUrphy@H20partwersUSH, can



3

H<sub>2</sub>O PARTNERS

SETRPC HAZARD MITIGATION PLAN RISK ASSESSMENT Planning / Stakeholder Workshop South East Texas Regional Planning Commission, Beaumont, TX August 24, 2016

Name/Title	Department	Phone	Email
Jeffery h. hylemb	EMC City of Kountze	409-790-8990	JIKCL OSbCG/ob-nl. Not
Shorasa Wigley, Er	nc. Handin County	409-246-5119	Theresa. Wigley@
Fick THompisce	- Mayor/EMC RoseHilliAc	105 409-755-4940	rlthomisee Cymanicom us
CHRIS SERRES	FINANCE Director DCU	1. #1. 409-769-2669	CSErres@ocwc1.com
SUMMIN BLACKWELL	SILSISEE, Ging OF	409-651-8668	sblackwell Certyofs, Is beer con
Rochel Andrews	FIZO Partners	512-983-0092	racherah 2000 ther, usa
Heidi Watson	H20 Partners	512.568.2259	heidi Ch20partnors USa. Com
	anna ann ann an an t-airte ann ann an t-airte		1
		р.	
4			

# **Public Meeting Documentation**

As discussed in Section 2, a series of three public meetings were held in conjunction with each of the SETRPC workshops. Documentation in the form of sign-in sheets for each of the meetings follows.

CET		5/			REGISTRATIO	N
				MEETING: OCATION:	Mitigation Action Plan Public Me SETRPC – Homer E Nagel Confe	eeting erence Room
ODOLII PAOL IPVAO IV	LEUIUNAL PLAN	MING COMMI22	IUN	DATE.	Beaumont, Texas 77703	
				DATE:	wednesday, March 30, 2016	5:30pm
PLEASE PRINT AND W	RITE LEGIBLY					
PLEASE PRINT AND W	RITE LEGIBLY	AGENCY:	ADDRESS:		E-MAIL ADDRESS	PHONE NUMBER
PLEASE PRINT AND W VAME: Rachel Andrews	RITE LEGIBLY TITLE: NikoptonSpeci	AGENCY: Shist HZG Partners	ADDRESS:		E-MAIL ADDRESS	PHONE NUMBER:
PLEASE PRINT AND W NAME: Rechel Andrews Heidi Watson	RITE LEGIBLY TITLE: NikopkonSpeci MikoptionSpeciales	AGENCY: Sir Itzo Partners H H207Brthers	ADDRESS: Avstry, TX		E-MAIL ADDRESS radied ghao partnery heidi@hzopartnersvs.com	PHONE NUMBER: 5 12 983-009 2
HEASE PRINT AND W VAME: Rachel Andrews Heidi Watson Die Bandy	RITE LEGIBLY TITLE: Mikgston Species Mikgston Species Directur	AGENCY: XIST HEO Portnors H HEO Portners SETPPC	ADDRESS: Avstry, TX Austrin TX	Talmos	E-MAIL ADDRESS roll of Department bed Orzopathers va. com	PHONE NUMBER: 5 12 983-009 2 512 568-2259

Figure F-4. SETPRC Public Kickoff Workshop, 03.30.16

Figure F-5.	SETPRC Public Risk	Assessment V	Vorkshon	06.01.16
inguici -J.	SETTING FUSING MISK /		vorksnop,	00.01.10



1

H<sub>2</sub>O PARTNERS

SETRPC HAZARD MITIGATION PLAN Public Meeting Hardin County Courthouse Commissioners' Courtroom, Kountze, TX June 1, 2016

Name/Title	Department . County Judge - Havid	Phone 404-246- 5120 ملط	Email Wayne. <i>M</i> c Danie/I
Shore sc. Wigley	Ermeng Mgt Coon - He	udin Cru 404-246-	Theresa. Wigky @
Mile water	BEAUMOUST EOC/PD	409-239-8233 mCust	Slad march to Dr.
Richel Anotows	120 Portners	572 983-0092	Cachelah2opertrosusa
Hadi Watson	Azo Partuers	512 568 2259	keidi a.h.Zupar-Hursusa. com
			·
			14 11 VIII
<u></u>		a. , , , ,	

SETRPC			H <sub>2</sub> O partners
	SETRPC HAZARD MITIGA Public Meeting Orange County Expo Center August 24, 2016	TION PLAN , Orange, TX	
Name/Title	Department	Phone	Email
Rochel Andrews / Mutig	ation Specialist H20 Partner	z 512-983-0092	rachelohaopantersura.com
Heidi Watson / Mitgate	aspecialist H20Particers	512 568 2259	heidile hzopartnersusa.com
MARK LANdry	LEPC OFANGE CO	337-304-7698	MARK.W. Lands y-20 dulout cont
CHEAY FRAGOMEN	I Compered	409 745-0400	Fegcher tol. com.
D USTINHARRIS	COLONIAL REPELINE	409984 3560	d harris @ colpire.com
KILEY WILLIAMS	Calonia Pretino	(109) (5:2-3)47	KWILLIAM COUPIPECOM
Aleta Cappen	City of Widor	4097694561	acapen & city of Vidor.com
CLORK SLOKIEM	ORANGE COUNTY	409-882-7905	CSLACISM R CO. ORANGE, TX.US
Laura Slacy	m		
Kaulan Arendure	SETTAPL	429-924-7513	KARELAHE CEUPC. ONG
CHAMLES Los	ORANGE COUNTY DEM	409-298-9169	CCONA CON ORMUGETY, US
MISSY PILISA VY	OTANIQ COUNTY DEM	409-745-9718	mpilister @co. pravige tx US

### Figure F-6. SETPRC Public Mitigation Actions Workshop, 08.24.16

## **Public Notices**

1

Invitations to take the public survey and public notices to announce Jefferson County's participation in the Plan Update development process were posted on various websites and on Facebook as shown in Figures F-7 through F-12.



### Figure F-7. Public Notice, Jefferson County Drainage District No. 7 Web Page, HMAP Update

### Figure F-8. Public Notice, Jefferson County Office of Emergency Management Facebook Page, Survey Posting



Figure F-9. Public Notice, South East Texas Regional Planning Commission Web Page,



03.30.16 Public Meeting

### Figure F-10. Public Notice, South East Texas Regional Planning Commission Calendar Posting, 06.01.16 Public Meeting and Survey Invitations



DATE	EVENT	LOCATION	TIME:
MAY			
05/19	South East Texas Foster Grandparent In-service	Appomattox Club Harriot Street	8:30 am - 12:30 pm
05/26	Jefferson-Orange-Hardin Regional Transportation Study (JOHRTS) Transportation Planning Committee	Homer E. Nagel Room	10:00 am
05/30	Memorial Day	SETRPC Offices Closed	
JUNE			
06/01	Mitigation Action Plan Public Meeting	Hardin County Courthouse 300 West Monroe	5:30 pm
06/08	Regional Coordination Steering Committee	MCM Elegante'	1:30 pm Ballroom D
06/09	Jefferson-Orange-Hardin Regional Transportation Study (JOHRTS) Technical Committee	Transportation Room	10:00 am
06/15	Executive Committee	Homer E. Nagel Room	12:00 pm
06/16	South East Texas Coalition for the Homeless	Homer E. Nagel Room	3:00 pm
06/23	South East Texas Foster Grandparent In-service	Appomattox Club Harriot Street Beaumont	8:30 am – 12:30 pm
JULY	NO EXECUTIVE COMMITTEE SCHEDULED		
07/04	Independence Day	SETRPC Offices Closed	
07/06	Community Services Block Grant Advisory Board	Homer E. Nagel Room	3:00 pm - 5:00 pm
07/21	South East Texas Coalition for the Homeless	Homer E. Nagel Room	3:00 pm
07/28	Jefferson-Orange-Hardin Regional Transportation Study (JOHRTS) Transportation Planning Committee ( <i>Previously scheduled for 7/21/16</i> )	Homer E. Nagel Room	10:00 am

President – Kirk Roccaforte, Bridge City I 1ª VP – Eddie Arnold, Jefferson County I 2<sup>nd</sup> VP – Chris Kirkendall, Hardin County 3<sup>rd</sup> VP – Barry Burton, Orange County I Treasurer – Rebecca Ford, Bevil Oaks I Secretary – Mary Adams, Kountze Executive Director - Shaun P. Davis 2210 Eastex Freeway Beaumont, Texas 77703-4929 (409) 899-8444 I (409) 347-0138 fax <u>setrpc@setrpc.org</u> I http://www.setrpc.org

Jefferson County | Hazard Mitigation Plan Update 2017 | Page 13



#### Public Participation in Hazard Mitigation Action Plan

The South East Texas Regional Planning Commission (SETRPC) will hold the second Mitigation Action Plan public meeting on Wednesday, June 1, 2016 beginning at 5:30pm at the Hardin County Courthouse, Commissioner's Courtroom located at 300 West Monroë, Koünize, Texas 77625. The meeting will encourage the public's participation in the risk assessment portion of updating the Hazard Mitigation Action Plans for the South East Texas TRagion of Hardin, Jefferson and Orange Counties. The Federal Emergency Management Administration (FEMA) requires the plans be updated every five (5) years. The goal of the Mitigation Action Plans is to outline projects within the region that can mitigate/reduce/prevent the loss of life and property as a result of a natural disaster. The public is encouraged to attend. For more information, or if special accommodations are needed, please contact Sue Landry at slandry@setrpc.org or (409) 924-7514.

You can also participate by completing an online survey by going to:

https://www.surveymonkey.com/r/SETRPC HMAP

President – Kirk Roccaforte, Bridge City I 1<sup>st</sup> VP – Eddie Arnold, Jefferson County I 2<sup>nd</sup> VP – Chris Kirkendall, Hardin County 3<sup>rd</sup> VP – Barry Burton, Orange County I Treasurer – Rebecca Ford, Bevil Oaks, I Secretary – Mary Adams, Kountze

Executive Director - Shaun P. Davis 2210 Eastex Freeway Beaumont, Texas 77703-4929 (409) 899-8444 | (409) 347-0138 fax <u>setrpc@setrpc.org</u> | http://www.setrpc.org

### Figure F-11. Public Notice, South East Texas Regional Planning Commission Web Page,

#### 06.01.16 Public Meeting and Survey Invitations



http://www.setrpc.org/

5/3/2016

Figure F-12. Public Notice, SETRPC Webpage Posting, 08.24.16 Public Meeting and Survey Invitations



APPENDIX G: CAPABILITY ASSESSMENT

Overview	.1

Jefferson County Multi-Jurisdictional Capability Assessment ......2

# Overview

The Planning Team completed a Capability Assessment Survey at the beginning of the planning process. The completed Capability Assessment Checklist, included in Appendix G, provides information on existing policies, plans, and regulations for Jefferson County and the participating jurisdictions.

A Capability Assessment is an integral component of the Plan Update development process. The Capability Assessment serves to evaluate a community's existing planning and regulatory capabilities to support implementation of the Plan's Mitigation Strategy Objectives.

Each community has a unique set of capabilities including policies, programs, staff, funding, and other resources available to accomplish hazard mitigation objectives and reduce long-term vulnerability. The Planning Team identified existing capabilities in each jurisdiction that currently reduce disaster losses or could be used to reduce losses in the future, and capabilities that inadvertently increase risks in the community.

# Jefferson County Multi-Jurisdictional Capability Assessment

CAPABILITY CHECKLIST	Jefferson County	City of Beaumont	City of Bevil Oaks	City of China	City of Groves	City of Nederland	City of Nome	City of Port Arthur	City of Port Neches	SETRPC
Plan	ning/	Regula	itory T	ool						
Hazard Mitigation Plan	х	х	х	х	х	х	х	х	х	х
Comprehensive Land Use Plan		х		х	х	х		х		
Stormwater Management Plan/Ordinance	х	х		х	х	х		х	х	
Emergency Operations Plan		х		х		х		х	х	х
Capital Improvements Plan	х	х		х		х		х	х	
Floodplain Management Plan	х	х	х	х	х	х		х	х	
Flood Response Plan	х	х		х	х				х	
Historic Preservation Plan	х	х						х		
Continuity of Operations Plan	х			х	х	х		х		х
Evacuation Plan	х	х		х	х	х		х		х
National Flood Insurance Program (NFIP)	х	х	х	х	х	х	х	х	х	
NFIP Community Rating System		х	х					х		
NFIP Floodplain Ordinance	х	х	х	х	х	х	х	х	х	
Building Code	х	х	х	х	х	х		х	х	
Fire Code		х	х	х	х	х		х	х	
Other Plans										
Administra	tive aı	nd Tec	hnical	Capab	ility					
Planners		Х						Х	Х	Х

# Appendix G: Capability Assessment

CAPABILITY CHECKLIST	Jefferson County	City of Beaumont	City of Bevil Oaks	City of China	City of Groves	City of Nederland	City of Nome	City of Port Arthur	City of Port Neches	SETRPC
Engineers	х	х	х			х		х	х	
Emergency Manager	х	х	х		х	х		х	х	
Floodplain Manager	х	х	х		х	х		х	х	
Personnel skilled in Geographic Information Systems (GIS)	х	х						х		х
Resource development staff or grant writers	х	Х						х	Х	
F	inanci	al Res	ources	;						
Capital Improvement Programming	х	х							х	
Community Development Block Grants (CDBG)	х	х		х		х		х	х	
Stormwater Utility Fees				х						
Development Impact Fees										
Partnering Agreements or Intergovernmental Agreements		х		х				х		
Other										

## APPENDIX

## **City of Taylor Landing Mitigation Plan Participation**

NOTE: Taylor Landing is a Type C general law municipality located entirely within Jefferson County and can best plan for hazard mitigation by partnering on an interlocal basis with the County. It has a population of approximately 275 people, maintains a sewage treatment plant TCEQ licensed, a road system, a drainage system, and a garbage collection service. Since the incorporation of the City, it has done all of its hazard mitigation planning in cooperation with, under interlocal contract with, and subject to oversight by Jefferson County. In 2006, the City of Taylor Landing and Jefferson County partnered to update the County's Local Mitigation Plan. This plan was adopted at the end of 2006. It also did so in 2012 when the plan was revised and in 2016 when the plan was revised again. Representatives of the City fully participated in this update process under the oversight of the South East Texas Regional Planning Commission. Therefore, the City seeks to be "added" to the Jefferson County Local Mitigation Plan Update.

According to the Federal Emergency Management Agency (FEMA), "Additional jurisdictions may be added to an existing, FEMA-approved, mitigation plan, only if the conditions below are met (if all three conditions are not met, the jurisdiction may develop its own mitigation plan).

- 1. The jurisdiction(s) asking to be included is located within the boundaries of, or adjacent to, the area covered by the multi-jurisdictional mitigation plan.
- 2. The organization that was responsible for preparing and submitting the multi-jurisdictional mitigation plan to the State and FEMA agrees with the addition of the requesting jurisdiction(s) to the mitigation plan.
- 3. When the multi-jurisdictional mitigation plan was developed, the risk assessment included an analysis of the major hazards, specifically the natural hazards that have the potential to impact the additional jurisdiction(s).

If these conditions can be met, there are two options that can be used to add additional jurisdictions to a FEMA-approved mitigation plan. Regardless of the option chosen, each jurisdiction joining a multihazard planning process and seeking to receive credit from FEMA for an approved mitigation plan must satisfy all of the Local Mitigation Plan requirements identified at 44 CFR §201.6.

The following actions must be taken to add new jurisdictions to the existing multi-jurisdictional mitigation plan and enable them to receive approval as part of the mitigation plan:

- 1. The requesting jurisdiction(s) must review the multi-jurisdictional hazard analysis and determine if there are any additional hazards that have not been addressed and threaten the jurisdiction(s). If none exist, the jurisdiction(s) must document their review process and state that no additional hazards exist. If the review reveals additional hazards, the jurisdiction(s) must analyze the risks it/they face associated with those hazards and include this analysis in their written appendix to the multi-jurisdictional mitigation plan. The existing risk assessment cannot be resubmitted without this additional documentation.
- 2. The requesting jurisdiction(s) must document agreement with the stated mitigation goals of the multi-jurisdictional mitigation plan. Additional goals specific to the requesting jurisdiction may be added. Each additional jurisdiction must also develop a list of proposed mitigation actions appropriate for that jurisdiction. These can include the common actions outline in the multi-jurisdictional mitigation plan, but must include specific mitigation actions for each profiled hazard the jurisdiction itself [sic].
- 3. The requesting jurisdiction(s) must document the involvement of both the general and the local government in the planning process in accordance with 44 CFR §201.6. The level of participation in the multi-jurisdictional mitigation plan must be consistent for the additional jurisdictions.

4. The annex or appendix, along with the multi-jurisdictional mitigation plan, and a letter of concurrence from the agency or organization responsible for the mitigation plan, must be submitted to the State for formal review. When the State finds the mitigation plan approvable, it will forward it to FEMA. When FEMA's review finds the mitigation plan 'approvable pending adoption,' the new jurisdiction can formally adopt the full mitigation plan and its jurisdiction-specific annex or appendix and submit the mitigation plan in final form through the State to FEMA for approval.

The mitigation plan approval date for the added jurisdictions will continue to be the date given by FEMA for the multi-jurisdictional mitigation plan. This means that the additional jurisdictions will have less than the entire 5-year plan approval window before they will need to engage in the required update for the full multi-jurisdictional mitigation plan" (FEMA, 2008, pp. 21-23).

The City believes its most important hazard mitigation step is to comply with FEMA NFIP building code restrictions for construction of structures. All structures of significance in the City are single family private residences and since 2009 all constructed structures have been required to obtain a building permit with an elevation certificate showing first level of habitation to be eighteen inches over BFE. No structure has been built since then without such a permit. The City maintains a flood plain administrator and building permit issue activity under the office of the Mayor.

The City owns and operates a sewage treatment plant, damaged by high waters in Ike in 2008. It has since kept all water sensitive components high and during Harvey damage was significantly reduced.

The City has since 2009 actively participated in the STEAR 211 program registering all individuals requiring evacuation assistance. There is a STEAR designated coordinator who has completed the State training program.

The City has regularly attended meetings of the Sabine Neches Chiefs Association to ensure coordination with regional emergency management. It has executed an interlocal agreement with Jefferson County whereby the Jefferson County EMC is also the City EMC and the City has adopted the Jefferson County Emergency Management Plan.

The City has promulgated all emergency management communications to its residents to ensure they are timely informed of potential emergencies.

All coordination activities with the County and with the SETRPC have been delegated to the Mayor who has accomplished all such coordination. The Mayor as Emergency Management Director has completed all NIMS training and has ensured full NIFP compliance consistent with adopted City ordinances.

These activities have included in County-wide and City specific risk analysis, the primary risk to the City being rising Taylor Bayou because of either rain events or hurricane surge. The surge analysis was the experience of the City during Ike, the largest surge event in recent Texas history and the rain analysis

was proven accurate after Harvey with no substantial damage to any City structure. Harvey did not produce a repeated substantial damage event.

All other risks possibly applicable to the City have been analyzed by the County and the SETRPC and the City has studied these risks for applicability to the City.

The City will maintain this Appendix by continued meeting with, coordination with and discussion with Jefferson County Emergency Management and the SETRPC. Any actions required by those plans applicable to the City will be effected.

During Hurricane Ike, September 2008, the City experienced one severe NFIP loss and that structure was demolished. During Hurricane Harvey, September 2017, the City experienced no severe and no severe repetitive NFIP losses.

The City has commenced steps necessary for Texas to review this adoption and to recommend that FEMA recognize the adoption.